

# On Charged Lifshitz Holography

Emil Have



September 2017 Master's Thesis SUPERVISORS: Cindy Keeler and Niels Obers

Emil Have: *On Charged Lifshitz Holography*, Master's thesis, © September 2017. Cover Illustration: *Circles in a Circle* by Wassily Kandinsky (1923), Oil on canvas, Philadelphia Museum of Art.

I just think too many nice things have happened in string theory for it to be all wrong. Humans do not understand it very well, but I just don't believe there is a big cosmic conspiracy that created this incredible thing that has nothing to do with the real world.

— Edward Witten

We begin by studying holographic renormalization of a free Einstein-Maxwell-Dilaton theory, which we call the electromagnetic uplift. Imposing asymptotically locally AdS boundary conditions, we employ a generalized version of the Hamilton-Jacobi approach to holographic renormalization and find a novel counterterm for d = 4. We construct the associated Fefferman-Graham expansions recursively and identify the Ward identities. We also comment on a subtlety regarding holographic renormalization of *p*-form fields and provide a conjecture based on our results for one-forms. We present a simple method to determine the counterterm action for massless *p*-form fields in  $AdS_{p+2}$  and illustrate the approach for d = 2.

We then develop charged Lifshitz holography for z = 2 by performing a Scherk-Schwarz reduction of the electromagnetic uplift. The Lifshitz space-time in the reduced theory is shown to correspond to a z = 0 Schrödinger geometry in the electromagnetic uplift. The sources are identified as the leading components of the bulk fields in a vielbein formalism and are shown to transform under a U(1)-extended Schrödinger group. We show that the new sources can be identified with the fields of Galiliean Electrodynamics (GED). Since the Scherk-Schwarz reduction becomes null on the boundary, we are directly lead to the conclusion that the boundary geometry becomes torsional Newton-Cartan (TNC), which we also explore from the perspective of gauging the Schrödinger algebra. We then determine the VEVs along with all Ward identities, and we show that the integrated Weyl anomaly becomes an action describing Hořava-Lifshitz gravity coupled to GED on a Newton-Cartan geometry. Finally, based on a dimensional analysis of GED on anisotropic backgrounds, we provide a conjecture for the extension of charged Lifshitz holography to general values of z.

We provide the required background in holography, Newton-Cartan geometry and holographic renormalization. We also provide a survey of pure Lifshitz holography for arbitrary values of z.

We are all agreed that your theory is crazy. The question which divides us is whether it is crazy enough to have a chance of being correct. My own feeling is that it is not crazy enough.

- Niels Bohr

### ACKNOWLEDGMENTS

First and foremost, I want to thank my advisors Cindy and Niels, whose guidance and encouragement have been instrumental during the preparation of this thesis.

I wish to express my gratitude to Troels Harmark, Mathias Wilhelm, Jacob Bourjaily, Lárus Thorlacius, Jens Paaske, and, in particular, Jelle Hartong for useful discussions. I also thank Morten Holm Christensen, Blaise Rollier, Koenraad Schalm, Kostas Skenderis and Ioannis Papadimitriou for useful correspondences.

I also extend my thanks to the entire HET group—in particular fellow Master students Bjarke Todbjerg Nielsen and Marieke van Beest—for interesting discussions and for providing a pleasant work environment.

I thank Bjarke Todbjerg Nielsen, Marieke van Beest, and Dennis Hansen for comments and corrections on an early draft. There may still be the odd<sup>1</sup> sign mistake here and there, which is of course entirely on me.

Finally, I want to thank my family for their love and support through the years.

Special thanks goes to Nanna. You mean the world to me.

<sup>1</sup> Although, I'd prefer it to be even.

CONTENTS

1	INTRODUCTION		1
	1.1	Outline & Summary	2
2	ASP	PECTS OF THE ADS/CFT CORRESPONDENCE	4
	2.1	The Holographic Principle	4
	2.2	Motivation of the Correspondence: Geometrization of Renormalization Group Flow	5
	2.3	The $AdS_5/CFT_4$ Correspondence	5
		2.3.1 Counting Degrees of Freedom and the UV/IR Connection	7
	2.4	The Holographic Dictionary for AdS/CFT I: The Field/Operator Map	7
		2.4.1 Boundary Asymptotics	3
	2.5	The Holographic Dictionary for AdS/CFT II: The GKPW Rule	С
	2.6	Witten Diagrams	2
	2.7	Outlook	3
3	ног	LOGRAPHIC RENORMALIZATION 15	5
	3.1	Holographic Renormalization as Hamilton-Jacobi Theory	5
		3.1.1 Hamilton-Jacobi Theory in a Nutshell	5
		3.1.2 Hamiltonian Gravity and the Hamilton-Jacobi Equation	7
		3.1.3 Recursive Solution of the HJ Equation	С
		3.1.4 The Induced Metric Expansion	1
	3.2	The $\delta_D$ Expansion & Holographic Renormalization of the Electromagnetic Uplift 22	2
		3.2.1 Holographic Renormalization of the Electromagnetic Uplift	2
		3.2.2 Renormalized Ward Identities & One-Point Functions	С
		3.2.3 Generating the $d = 4$ Fefferman-Graham Expansions	1
	3.3	Holographic Renormalization of $p$ -forms $\mathcal{E}$ the Scalar-Vector Duality $\dots \dots \dots \dots 33$	3
		3.3.1 From the Maxwell One-Form to <i>p</i> -Forms: A Conjecture	3
		3.3.2 The $d = 2$ Scalar-Vector Duality & Its Generalization	4
	3.4	Outlook	5
4	NEV	NTON-CARTAN GEOMETRY 37	7
	4.1	Warm-Up: Gauging the Poincaré Algebra to Obtain General Relativity	7
	4.2	Non-Relativistic Space-Times & Newton-Cartan Geometry	1
		4.2.1 Non-Relativistic Space-Times	1
		4.2.2 Newton-Cartan Geometry from Local Algebras	2
		4.2.3 Gauging Galilei	3
		4.2.4 Extension to Bargmann and the Affine Connection	3
	4.3	The Freezeward Manual Reduction	7
	4.4	Outlash	5
_	4.5	Outlook	)
5	(PU	RE) LIFSHITZ HOLOGRAPHY   51     The Oversteen Lifebite Medal   51	L
	5.1	Lifshitz Halageraphy	1
	5.2		2
		5.2.1 The EFD Model	2
		5.2.2 AILII boundary Conditions of the Sources	3 R
		5.2.3 More Eocal Hansformations of the Sources	י ר
		5.2.4 Entrigence of Tive Geometry O Schlounger Symmetry	י ר
	ΕC	VEVs and Covariant Ward Identities in Lifshitz Holography	- ว
	5.5	5.2.1 VEVs & Circumvention of Renormalization	- 2
		5.3.2 Boundary Energy-Momentum Tensor and Mass Current	4
		5.3.3 Ward Identities	† 1
	5.4	Outlook	т б
6	сни	ARGED LIFSHITZ HOLOGRAPHY $\mathscr{E}$ GALILEAN ELECTRODYNAMICS $6^{r}$	7
2	6.1	Galilean Electrodynamics	3
		6.1.1 Non-Relativistic Electrodynamics & the Unification of Two Limits	3
		6.1.2 Null Reduction of Maxwell Electrodynamics on a Curved Background	9
		6.1.3 Dimensional Analysis & Scaling Weights for Anisotropic Backgrounds	5
	6.2	The Electromagnetic Uplift	1

		6.2.1 The Unlift & $z = 2$ Pure Lifshitz Holography 71
		6.2.2 Scherk-Schwarz Reduction of the Electromagnetic Unlift 71
		6.2.2 Variation of the Action
		0.2.3 variation of the Action $$
	( -	$6.2.4$ Oxidation to a $2 = 0$ schrödinger Geometry $\ldots \ldots $
	6.3	Sources & Their Transformations
		6.3.1 Defining the Sources
		6.3.2 Iransformation Properties
	6.4	Boundary Geometry & $\mathfrak{sch}_2(2,1)$
		6.4.1 Null Reduction on the Boundary
		6.4.2 Gauging $\mathfrak{sch}_2(2,1)$
	6.5	The Anomaly & Hořava-Lifshitz Gravity
		6.5.1 Reduction of the Anomaly
	6.6	Holographic Reconstruction or Mock FG Expansions
	6.7	The VEVs
		6.7.1 Definition & Near-Boundary Behaviour
		6.7.2 Relation to Five-Dimensional VEVs Via Null Reduction
		6.7.3 Local Transformations of the VEVs 90
	6.8	The Ward Identities & Boundary Stress Tensor
		6.8.1 The HIM Boundary Energy-Momentum Tensor & Augmentations
		6.8.2 Ward Identities
		6.8.3 Ward Identities From Null Reduction
	6.9	General–z Conjecture
	0.)	6.9.1 The Action
		6 o 2 Sources & VEVs
		6.0.2 Ward Identities
_	<b>CON</b>	(0.9.5)  ward recharges  (0.9.5)  ward recharges
7	CON	CLUSION & OUTLOOK 99
Ar	nenc	ix 101
11 1	CEO	
A	GEU	AdS Space Times
	A.1	Aus Space-Innes
		A.1.1 Demittion and Oseful Cooldinates
		A.1.2 Conformally Compact Einstein Manifolds & The Boundary
		A.1.3 The boundary: There and back Again
	A.2	AIAdS Space-limes & the Fefferman-Graham Theorem
В	CON	FORMAL FIELD THEORY 109
	B.1	CFT and the Conformal Algebra
	B.2	The Superconformal Algebra
С	MAX	IMALLY SUPERSYMMETRIC YANG MILLS THEORY 115
	C.1	Supersymmetric Gauge Theory
		C.1.1 The Planar Limit of $SU(N)$ Gauge Theories
	C.2	Type IIB Supergravity
D	THE	ADS/CFT CORRESPONDENCE FROM STRING THEORY 121
	D.1	D <sub>3</sub> Duality and $AdS_5/CFT_4$
		D.1.1 Open Strings
		D.1.2 Closed Strings
Б	тнғ	HOLOGRAPHIC SUPERCONDUCTOR
L	F 1	Holography at Finite Temperature Density & Chemical Potential $125$
	L.1	Fill Cravity Dual of Finite Temperature Field Theory J: Black Da-Branes
		E.1.1 Gravity Dual of Finite Temperature Field Theory II: AdS Schwarzschild Black
		Holo
		The Da / Da Brane Configuration and the Add Daisener Mondetröm Plack Using the
	П.С	E.1.3 The $D_3/D_7$ brane Configuration and the Aub Keissner-INORdstrom black Hole 127 The a Wave Holographic Superconductor
	Е.2	The s-vvave molographic superconductor $\dots \dots \dots$
		<b>E.2.1</b> Holographic Superconductivity in the Probe Limit $(a = 3)$
		E.2.2 Conductivity in the Probe Limit
F	HOL	OGRAPHIC RENORMALIZATION: SUPPLEMENTARY MATERIAL 136
	F.1	Fetterman-Graham Holographic Renormalization
		F.1.1 FG Holographic Renormalization of Pure Gravity in $AlAdS_5$
	F.2	The dBVV Ansatz Approach
		F.2.1 Setting up the ADM Hamiltonian and the Hamilton-Jacobi Equation141

	F.2.	2 Counterterm Action Ansatz	. 142
	F.2.	3 The DeWitt Bracket	. 143
	F.2.	4 Holographic Weyl Anomalies	. 144
G	HOLOGI	APHIC RENORMALIZATION OF LIFSHITZ SPACE-TIMES	146
	G.1 Ho	lographic Renormalization of Einstein-Proca Theory with Lifshitz Boundary Condi-	
	tion	ıs	. 146
	G.1	1 The Einstein-Proca ADM Hamiltonian	. 146
	G.1	2 Boundary Conditions & the $\delta_D$ Expansions	. 148
	G.1	3 Finding the Counterterms	. 153
н	DIMENS	IONAL REDUCTION	154
	н.1 Kal	uza-Klein Reduction	. 154
	H.1	1 Kaluza-Klein Reduction of Pure Gravity & the EMD Model	. 155
	н.2 Sch	erk-Schwarz Reduction	. 156
	H.2	1 Scherk-Schwarz Reduction of the Uplift	. 157
BI	BLIOGRA	РНҮ	161

## LIST OF FIGURES

Figure 2.1	Emergent AdS and renormalization group flow
Figure 2.2	Witten diagram for holographic computation of the three-point function 13
Figure 4.1	Foliation af a TTNC space-time
Figure A.1	Conformal Diagram of AdS
Figure C.1	On-shell diagrams: an intersting development in the description of $\mathcal{N} = 4$ SYM. 115
Figure C.2	Propagators and vertices for Yang-Mills theory
Figure C.3	Illustration of planar vs. non-planar diagrams
Figure E.1	Condensates and phase transitions of the holographic superconductor
Figure E.2	Conductivities for the holographic superconductor.

## LIST OF TABLES

Table 2 1	Versions of the AdS/CET Correspondence, grouped by strength
Table 2.1	Relation between mass and conformal dimension for various fields
	Relation between mass and conformal dimension for various news
Table 2.3	The AdS/CFT dictionary
Table 2.4	Feynman rules for Witten Diagrams
Table 4.1	Overview of the gauging procedure for the Galilean algebra
Table 4.2	The three Newton-Cartan geometries
Table 5.1	Sources and their scaling dimensions. Note that the sets $(\tau_{\mu}, e_{\mu}^{a})$ and $(v^{\mu}, e_{a}^{\mu})$
	are not independent, one must choose to work with one of them
Table 5.2	Newton-Cartan complex and associated invariances
Table 5.3	Tangent space projections of $T^{\mu}{}_{\nu}$ and $T^{\mu}$ with associated scaling weights. Quan-
	tities in parentheses denote the corresponding symbol in [26]
Table 6.1	The two limits of Maxwellian electrodynamics
Table 6.2	Dictionary between [58] and this work
Table 6.3	Scaling Weights of the GED fields
Table 6.4	Scaling Weights of the GED VEVs
Table 6.5	The sources and their scaling dimensions
Table 6.6	Overview of the gauging procedure for the Schrödinger algebra
Table 6.7	The VEVs and their scaling dimensions
Table B.1	Properties of superconformal BPS multiplets

### NOTATION & CONVENTIONS

We use the mostly positive Minkowski metric—for  $\mathbb{R}^{d,1}$ ,

$$\eta_{\mu\nu} = \operatorname{diag}(-1, \underbrace{1, \dots, 1}_{d \text{ entries}}).$$
(i)

Further, we make use of the summation convention, i.e. all repeated indices (of any kind!) are summed over unless explicitly stated. In the chiral—or Weyl—representation the four-dimensional Dirac matrices are given by:

$$\gamma^{\mu} = \begin{pmatrix} 0 & (\sigma^{\mu})_{\alpha\dot{\beta}} \\ (\bar{\sigma}^{\mu})^{\dot{\alpha}\beta} & 0 \end{pmatrix},$$
(ii)

with Pauli matrices

$$(\sigma^{\mu})_{\alpha\dot{\beta}} = (1,\sigma^{i})_{\alpha\dot{\beta}}, \ \ (\bar{\sigma}^{\mu})^{\dot{\alpha}\beta} = (1,-\sigma^{i})^{\dot{\alpha}\beta}.$$
(iii)

#### X ACRONYMS

They satisfy the Clifford algebra,  $\{\gamma^{\mu}, \gamma^{\nu}\} = -2\eta^{\mu\nu}\mathbb{1}$ . Objects contracted with  $\gamma$ -matrices may be written in Feynman slash notation, e.g.  $\partial := \gamma^{\mu}\partial_{\mu}$ . The van der Waerden indices are raised and lowered using the two dimensional Levi-Civita symbol

$$\varepsilon^{\alpha\beta} = \varepsilon^{\dot{\alpha}\dot{\beta}} = \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix} = -\varepsilon_{\alpha\beta} = -\varepsilon_{\dot{\alpha}\dot{\beta}}.$$
(iv)

Further, we can define representations of generators of Lorentz transformations in terms of the Pauli matrices,

$$(\sigma^{\mu\nu})_{\alpha}{}^{\beta} = \frac{i}{4} \left( \sigma^{\mu}_{\alpha\dot{\alpha}} \bar{\sigma}^{\nu\dot{\alpha}\beta} - \sigma^{\nu}_{\alpha\dot{\alpha}} \bar{\sigma}^{\mu\dot{\alpha}\beta} \right), \tag{v}$$

$$(\bar{\sigma}^{\mu\nu})^{\dot{\alpha}}{}_{\dot{\beta}} = \frac{i}{4} \left( \bar{\sigma}^{\mu\dot{\alpha}\alpha} \sigma^{\nu}_{\alpha\dot{\beta}} - \bar{\sigma}^{\nu\dot{\alpha}\alpha} \sigma^{\mu}_{\alpha\dot{\beta}} \right).$$
(vi)

We denote antisymmetrization of indices with "[]" and symmetrization with "()", e.g.

$$T_{[\mu\dots\mu_n]} = \frac{1}{n!} \sum_{\sigma \in S_n} \operatorname{sgn}(\sigma) T_{\mu_{\sigma(1)}\dots\mu_{\sigma(n)}}$$
(vii)

where  $S_n$  is the symmetric group on *n* symbols. The Riemann tensor is defined via

$$[\nabla_{\mu}, \nabla_{\nu}]X_{\rho} = R_{\mu\nu\sigma}{}^{\rho}X_{\rho} - 2\Gamma^{\rho}_{[\mu\nu]}\nabla_{\rho}X_{\sigma}, \qquad (\text{viii})$$

for an arbitrary one-form  $X_{\rho}$ , where

$$R_{\mu\nu\sigma}{}^{\rho}X_{\rho} = -\partial_{\mu}\Gamma^{\rho}_{\nu\sigma} + \partial_{\nu}\Gamma^{\rho}_{\mu\sigma} - \Gamma^{\rho}_{\mu\lambda}\Gamma^{\lambda}_{\nu\sigma} + \Gamma^{\rho}_{\nu\lambda}\Gamma^{\lambda}_{\mu\sigma}, \qquad (ix)$$

from which the Ricci tensor obtained via the contraction  $R_{\mu\nu} = R_{\mu\sigma\nu}^{\sigma}$ . Given a metric  $g_{\mu\nu}$ , the Christof-fel symbols are given by

$$\Gamma^{\lambda}_{\mu\nu} = \frac{1}{2} g^{\lambda\rho} \left( \partial_{\mu} g_{\nu\rho} + \partial_{\nu} g_{\rho\mu} - \partial_{\rho} g_{\mu\nu} \right). \tag{x}$$

We will, unless stated otherwise, adhere to the following index convention

 $\mu$ ,  $\nu$ ,...: General space-time indices. In the context of Lifshitz holography in chapters 5 and 6, the three-dimensional indices on radial hypersurfaces.

*i*, *j*, . . . : Spatial part of the general space-time indices. The time component will be denoted *t*.

<u>*a*</u>, <u>*b*</u>, . . . : Tangent space indices including time.

 $a, b, \ldots$ : Tangent space indices excluding time. The time component is denoted "0".

 $\mathcal{M}, \mathcal{N}, \ldots$ : Five-dimensional space-time indices. Used in bulk models for the uplift in chapter 6.

*u*: The compact direction in five-dimensional models.

*A*, *B*, ...: Four-dimensional space-time indices. The same as M, N, ..., but excludes the radial direction.

 $M, N, \ldots$ : Four-dimensional space-time indices. The same as  $M, N, \ldots$ , but excludes the compact direction *u*.

We point out that in chapter 3, the index structure is different: there, we use  $\mu, \nu, \ldots$  to denote (d + 1)-dimensional space-time indices, while index  $i, j, \ldots$  are used for *d*-dimensional indices that exclude the radial direction. We also point out that tangent space indices  $a, b, \ldots$  in this chapter incluces time.

Note also that in chapter 2, we use *z* to denote the radial coordinate in Poincaré coordinates, while in later chapters—notably chapters 3, 5, and 6, we use *r*. In particular, since in the context of Lifshitz holography, *z* is the symbol for the dynamic exponent, we hope that this does not cause any confusion.

Throughout the thesis, we will use  $\simeq$  to indicate the leading term. For example, if a quantity *X* has an expansion in a parameter *r* near r = 0 of the form  $X = X_{(0)}r^{-2} + X_{(1)}r + X_{(2)}r^{75} + \cdots$ , we write  $X \simeq X_{(0)}r^{-2}$ .

Unless stated otherwise, we work in natural units, where  $\hbar = c = k = 1$ .

#### INTRODUCTION

The seminal paper by Maldacena in 1997 on the AdS/CFT correspondence [1] sparked a veritable revolution in theoretical physics. The correspondence provides a concrete realization of the holographic principle put forth by 't Hooft and Susskind in [2, 3], and was the culmination of the second superstring revolution initiated by Witten in [4].

While the most general version of the AdS/CFT correspondence posits an exact equivalence between type IIB string theory on  $AdS_5 \times S^5$  and  $\mathcal{N} = 4$  super-Yang-Mills theory, a more moderate and well-established version of the correspondence relates weakly coupled gravitational theories in asymptotically locally AdS—the *bulk*—to strongly coupled field theories on the boundary (usually of either flat or spherical topology). This feature of the duality is extremely useful: it offers tantalizing opportunities of describing strongly coupled field theories using *weakly coupled*—that is to say classical—gravitational theories, and many of the most interesting not-yet well understood physical phenomena belong in this class of theories, e.g. high  $T_c$  superconductivity—explored holographically in e.g. [5–8]—and the  $\mathcal{N} = 4$  quark-gluon plasma, for which the viscosity to entropy ratio was calculated holographically in the seminal work [9] (see [10] for a review). The attempt to describe strongly coupled condensed matter systems using the AdS/CFT correspondence has garnered much attention in recent years [11–14] following the groundbreaking work of Sachdev and Herzog in [15].

However, most strongly coupled condensed matter systems are non-relativistic in nature. This is an inherent problem and would seem to render the application of holography in the context of such systems void, as succinctly pointed out by Nobel laureate Phil Anderson in [16]:

"As a very general problem with the AdS/CFT approach in condensed-matter theory, we can point to those telltale initials "CFT"—conformal field theory. Condensed-matter problems are, in general, neither relativistic nor conformal. Near a quantum critical point, both time and space may be scaling, but even there we still have a preferred coordinate system and, usually, a lattice. There is some evidence of other linear-T phases to the left of the strange metal about which they are welcome to speculate, but again in this case the condensed-matter problem is overdetermined by experimental facts." —Philip W. Anderson.

This is of course a legitimate point of critique. While holographic systems in an AdS/CFT context are conformal, relativistic and, to make matters worse, supersymmetric, at short distances, many of these symmetries are broken at large distances in the presence of a finite chemical potential or temperature. This long distance behaviour can, in fact, be captured by gravity duals, showcasing the power of the fluid/gravity correspondence [10, 11, 17, 18].

Another—and more recent—way forward is to directly consider bulk space-times, whose asymptotic behaviours are different from AdS as emphasized in [19–22]. Such space-times include Schrödinger, Lifshitz and hyperscaling violating geometries, which are all characterized by a so-called dynamical exponent *z*, which expresses the anisotropy between space and time on the boundary. Developing such notions of non-relativistic holography turns out to be extremely challenging, and despite a flurry of research (see e.g. [23–29]), this is still very much a work in progress, and many aspects remain poorly understood. It is important to emphasize in this context that there are no conventional tools available to the condensed matter theorist that allows him/her to attack such problems, and as such these holographic approaches are unique in their scope.

In addition to the applications of non-relativistic holography to condensed matter problems, it is also of intense theoretical interest to ascertain the validity of the holographic principle in concrete settings that go beyond the original AdS setting, and circumstantial evidence for the principle in the form of exotic gravity/field-theory dualities constitute important milestones on the road towards quantum gravity.

For Lifshitz bulk geometries, which will be the focus of this thesis, such a non-relativistic holographic correspondence for z = 2 was developed in [23, 24], where it was shown that this model could be embedded in a higher-dimensional model with conventional AdS geometry. By considering perturbations around a z = 0 Schrödinger geometry (which has AdS boundary conditions) in the higher-dimensional theory, it was shown that a Scherk-Schwarz reduction produced a corresponding z = 2 Lifshitz geometry with associated perturbations, and this mapping allowed for a complete identification of the holographic dictionary for z = 2 Lifshitz holography. A crucial observation was the realization that the boundary is described by a novel extension of Newton-Cartan (NC) geometry (originally developed by Cartan in [30, 31]) called torsional Newton-Cartan (TNC) geometry. This analysis was subsequently extended to generic values of z > 1 in [29] (see also [32, 33]). The sources and VEVs of Lifshitz holography possess a Schrödinger symmetry, and it was shown in [34] that TNC geometry as it appears in Lifshitz holography can be obtained by gauging the Schrödinger algebra for suitable values of z.

Concretely, it was recently shown in [35, 36] (see also [37, 38]) that TNC provides a framework for field theory analyses of the (fractional) quantum Hall effect. It has also been used in connection with other problems involving strongly correlated electrons [39, 40]. In these approaches, the symmetry of the problem is used as a guiding principle, and various quantities such as the Hall viscosity and the Hall conductance can be computed as responses to the geometric data.

#### 1.1 OUTLINE & SUMMARY

Below, we provide an outline of the contents of each chapter. Note that chapters 3, and 6 in particular, contain mainly new material—these results will be the subject of [41], to appear. We also remark that each chapter has its own outlook section, where we comment on interesting extensions of the analyses provided in the chapter.

In chapter 2—along with the companion appendix D, where a string theoretic derivation is presented we provide a review of the AdS/CFT correspondence. This includes an investigation of the holographic principle in section 2.1 and a motivation of the correspondence based on the interpretation of the radial AdS direction as a renormalization group scale in section 2.2. We then formulate a precise version of the correspondence in section 2.3 and explore the holographic dictionary in sections 2.4 and 2.5. These sections review material found mainly in [11, 12, 42–46], and provides a foothold in holography.

We conclude in section 2.6 with a brief discussion of Witten diagrams and provide a concrete computation of a three-point function, following the approach of [47].

In chapter 3, we provide a detailed survey of holographic renormalization for AdS space-times. In the supplementary appendix F, we—following mainly the review [48]—review the original Fefferman-Graham (FG) approach, which, although conceptually straightforward, is computationally forbidding. We also provide a survey of the de Boer-Verlinde-Verlinde (dBVV) method following [49], which relies on an ansatz and a bilinear operation that we have named the deWitt bracket<sup>1</sup>. We then show in section 3.1 how holographic renormalization can be understood in terms of Hamilton-Jacobi (HJ) theory, and discuss how the solution of the HJ equation conveniently involves the introduction of a suitably chosen operator  $\delta$  in terms of which the (bare) on-shell action and other relevant quantities can be expanded in eigen-modes. The discussion is based on [50, 51]. In section 3.2, we discuss the renormalization of the free Einstein-Maxwell-Dilaton model (EMD)-which we will call the electromagnetic uplift in chapter 6—and derive a set of counterterms, which has not previously appeared in the literature. We also construct novel FG expansions and determine the new Ward identities satisfied by the VEVs. Finally, in section 3.3, we discuss renormalization of *p*-form fields based on our results for the Maxwell field and provide a conjecture. These results do not exist in the literature, but are under investigation by other people, and will be the subject of an upcoming paper<sup>2</sup> [52]. We conclude with the observation that for d = 2, we can renormalize an EMD model by hodge dualization, and we generalize this observation to an easy way of obtaining the counterterm action for free *p*-form fields in  $AdS_{p+2}$ . To our knowledge, this has not appeared in the literature previously.

In chapter 4, we tell the tale of Newton-Cartan geometry from the perspective of *gauging algebras*. We start in section 4.1 with an introductory calculation that shows how Riemannian geometry—the arena of general relativity—is obtained by gauging the Poincaré group in a process analogous how one obtains gauge theories from a quantum field theory perspective. This section is based on appendix A in [53] and [54, 55], but is significantly more detailed. In section 4.2, we then discuss generalities of non-relativistic space-times and their relation Newton-Cartan geometry, before showing in section 4.2.2 that the gauging procedure applied to the Galilei and Bargmann algebras precisely gives TNC geometry. These sections are based on [34, 53, 56] and summarize the results therein. In section 4.3, we showcase how TNC geometry can be obtained from Lorentzian geometry via null reduction, and in the following section 4.4, we explore how non-relativistic field theories couple to TNC geometry from the perspective of null reduction. Both the preceding sections are based mainly on [57].

<sup>1</sup> This object, seemingly, did not have a name. Since it involves the deWitt metric with parameter d - 1, we have named this operation the deWitt bracket.

<sup>2</sup> I thank Kostas Skenderis for telling me about this.

In chapter 5, we investigate Lifshitz holography as developed by Hartong, Kiritsis and Obers in [29, 32] for generic values of the critical exponent *z*. We begin by a lighting review of Lifshitz field theory in section 5.1, before we turning to the actual holographic analysis in section 5.2. The bulk consists of an Einstein-Proca-Dilaton (EPD) model, which is shown to support Lifshitz solutions for generic values of *z*. We then identify the sources as the leading parts of the bulk fields in a vielbein formalism and see that they transform under the Schrödinger algebra, and we discuss how the boundary geometry becomes TNC in section 5.2.4. We identify the general properties of the VEVs by assuming holographic renormalizability in section 5.3, which also allows us to determine the TNC covariant Ward identities. The analysis of this chapter is a much more detailed version of [29].

We end the thesis in chapter 6, where we develop charged Lifshitz holography for z = 2 by generalizing the results of [23, 24, 57]. In section 6.1, we begin with a discussion of Galilean Electrodynamics (GED) coupled to TNC geometries, which was recently developed in [58]. We then provide a review of pure z = 2 Lifshitz holography and relate it to the uplift (see also appendix H), after which we Scherk-Schwarz reduce the electromagnetic uplift that we renormalized in chapter 3, which gives rise to a new reduced EPD-Maxwell-scalar model acting as the bulk theory for charged Lifshitz holography. In section 6.3, we show that this allows for a complete identification of the sources of charged Lifshitz holography, and we show that the new sources transform as the fields of GED under local transformations. This novel result is one the main contributions of this thesis. We then demonstrate explicitly that the reduction employed becomes null on the boundary, which is also discussed in [23, 24, 57]. This means that the results of chapter 4 directly leads us to conclude that the boundary geometry becomes TNC. In section 6.4.2, we, following [34], explore how the boundary geometry emerges from from gauging the z = 2 Schrödinger algebra. In section 6.5, we show that the integrated dimensionally reduced Weyl anomaly takes the form of an action for Hořava-Lifshitz (HL) gravity coupled to GED on TNC geometry, which is also a new result. We then consider in section 6.6 how the higherdimensional FG expansions that we generated in chapter 3 lead to expansions for the Lifshitz bulk fields, which will allow us to see the source structure of section 6.3 appear explicitly, which is a new analysis. We then work out all the new VEVs corresponding to the sources and determine novel Ward identities. Finally, we provide a conjecture for general-z charged Lifshitz holography based on a dimensional analysis of GED coupled to anisotropic backgrounds.

In this chapter, we provide background in the AdS/CFT correspondence which will be useful for the understanding of Lifshitz holography. This material contained here is based on a motley collection of sources, chiefly [11, 42–45].

We begin by reviewing the holographic principle in section 2.1, which, in its most general sense, roughly states that a volume of spacetime is equivalently described by the codimension one boundary of the volume, as argued by 't Hooft and Susskind, and we briefly mention a generalization known as the covariant entropy bound. This section is a review of the discussions in [2, 3, 46].

Based on [12], we then motivate the AdS/CFT correspondence in section 2.2 by interpreting the radial AdS direction as a renormalization scale, which naturally leads to AdS geometry.

Next, in section 2.3, we provide an overview of the AdS/CFT correspondence and discuss the relation to string theory and various supersymmetric field theories. This analysis follows mainly [42–44]. In the companion appendix D, we present a fairly detailed "stringy" derivation of the correspondence from D<sub>3</sub> branes, which are considered from the point of view of both open and closed strings, which, when unified, give the correspondence.

We then proceed to consider the holographic dictionary in its various manifestations, starting with the field-operator correspondence in section 2.4 and culminating with the GKPW rule in section 2.5 as well as a somewhat detailed summary of the dictionary in table 2.3. These analyses are based primarily on [11, 43]

We end the chapter in section 2.6 with a small discussion of Witten diagrams and showcase the computation of a concrete three-point diagram. The calculation follows [47].

#### 2.1 THE HOLOGRAPHIC PRINCIPLE

The idea of holography originates with 't Hooft's exposition in [2]: given some quantum theory, we can—invoking an analogue of the third law of thermodynamics—relate the entropy *S* to the total number of degrees of freedom; in particular, if<sup>1</sup>  $\mathcal{N}$  is the dimension of the Hilbert space (i.e. the number of states), the following relation holds

$$e^{S} = \mathcal{N}. \tag{2.1.1}$$

By the the covariant entropy bound, or simply the Bekenstein bound<sup>2</sup>, the entropy of some system cannot exceed that of a black hole which is given by the Bekenstein-Hawking relation [60]

$$S \le S_{\rm BH} = \frac{A}{4G},\tag{2.1.2}$$

where *A* is the area of the black hole and *G* is Newton's constant. Combining the spherical entropy bound with the relation between entropy and degrees of freedom (2.1.1) leads us to conclude that the number of states is bounded by  $\mathcal{N} \leq e^{\frac{A}{4G}}$ . The ideas put forward by 't Hooft inspired Susskind to write the iconic article *The World as a Hologram* [3], where he proffers the following bit of reasoning: supposing that the world—which we can take to be *d*-dimensional—is a lattice of binary quantum degrees of freedom (i.e. "spin-like") and assume that the lattice spacing is the Planck length,  $\ell_p$ , since smaller distances cannot be resolved in quantum gravity. Thus, the number of quantum states in a volume *V* is  $\mathcal{N}(V) = 2^n$ , where  $n = \frac{V}{\ell_p^d}$  denotes the number of lattice sites in *V*, which—in contradistinction to what we found above—implies the following entropy bound:

$$S \le \log \mathcal{N}(V) = \frac{V}{\ell_p^d} \log 2. \tag{2.1.3}$$

Rather than being bounded by the area A, the largest possible entropy scales as the *volume* V! As long as the system is larger than the Planck scale, it holds that  $V \ge A$ —and so this "field-theoretic" derivation predicts a larger entropy bound.

<sup>1</sup> Although standard in this context, it is unfortunate that the dimensionality of the Hilbert space is denoted by  $\mathcal{N}$ —a symbol usually reserved to denote to the number of supercharges in a given theory. We hope this does not cause confusion.

<sup>2</sup> This bound relies on certain assumptions [59]: the system must be of constant, finite size and have limited self-gravity—that is, gravity must be weak compared to the other forces acting in the system.

To see this, we invoke a basic principle of quantum mechanics: unitarity. Suppose that the field-theoretic entropy estimate holds. Then by the bound (2.1.1), the dimensionality of the Hilbert space describing the region is  $\mathcal{N} \sim e^V$  for V the volume. But, supposing that the region was converted into a black hole, the Bekenstein-Hawking entropy relation implies that the region is now described by a Hilbert space of dimension  $e^{A/4G}$ : the number of states has decreased and the Hilbert spaces are no longer isomorphic, violating unitarity. Insisting on unitary quantum mechanical evolution, we are let to conclude that the Hilbert space must have had dimensionality  $e^{A/4G}$  to start with. This observation leads to a preliminary version of the holographic principle:

**Susskind-'t Hooft Holographic Principle:** A region of spacetime with boundary of area A is fully described by at most A/4G degrees of freedom.

The spherical entropy bound, however, turns out to be violated in some instances—the simplest of which is the following: consider a system in the midst of a gravitational collapse. Before the system is destroyed on the black hole singularity, its surface area becomes arbitrarily small, and since entropy cannot decrease, the bound is violated. This spurred the discovery of a more universal<sup>3</sup> entropy bound: *the covariant entropy bound*, put forward by Bousso in [59]. It is worth noting, however, that no known fundamental derivation of the covariant entropy bound exists; if it holds true it must eventually be explained by a theory unifying gravity and quantum mechanics.

Since its proposal, several circumstances under which the bound holds has been uncovered by Flanagan *et al.* in [62], and further proof of its validity for free matter fields in the limit of weak gravitational back-reaction was given in [63]. Also, the covariant entropy bound has been shown [64] to reduce to other entropy bounds, which were observed to hold in some settings—in particular the spherical entropy bound. For more details, we refer the reader to [46, 63, 64].

Taking the holographic principle at face value, then, we are lead to conclude that if we have a theory of quantum gravity on some manifold  $\mathcal{M}$ , the theory will be entirely determined by some other theory living on the boundary  $\partial \mathcal{M}$ , and the theories are said to be *dual*.

## 2.2 MOTIVATION OF THE CORRESPONDENCE: GEOMETRIZATION OF RENORMALIZATION GROUP FLOW

Although the AdS/CFT has its origins in string theory—as we demonstrate in appendix D—it is possible to motivate the correspondence without explicit reference to string theory. The following line of reasoning was put forth by Horowitz and Polchinski in [65] (see also [11, 12, 66]): since any quantum theory of gravity contains a massless spin-two graviton<sup>4</sup>, one could theorize that the graviton somehow arises as a composite of two spin-one gauge bosons<sup>5</sup>. A priori, this seems to be excluded by the Weinberg-Witten theorem [68], which states that

**Theorem (Weinberg-Witten)** A quantum field theory with a Poincaré covariant and conserved energymomentum tensor  $T^{\mu\nu}$  forbids massless particles of spin j > 1 which carry momentum<sup>6</sup>.

General relativity circumvents this theorem by having either a vanishing energy-momentum tensor<sup>7</sup> or by having a reparametrization non-invariant *matter stress tensor* when additional fields are present.

However, the crucial assumption made in the Weinberg-Witten theorem is that the graviton moves in the same spacetime as the gauge boson, so by making the graviton live in a higher dimensional space, for example, we can again circumvent the theorem. As we saw above, the holographic principle constrains the entropy of any system to be at most that of a black hole occupying the space of the system, implying that the theory of quantum gravity lives in one dimension higher than the gauge theory. The extra dimension in the quantum gravity, as we now demonstrate, has a nice interpretation

<sup>3</sup> Although a recent paper [61] suggests speculative scenarios in which this new bound might be broken.

<sup>4</sup> The usual way of realizing this is by linearizing Einsteins equations, gauge fixing and then noting that the resulting polarization tensor for the perturbation (which we think of as the graviton) is a spin-two representation.

<sup>5</sup> Interestingly, in perturbative quantum gravity, this composite behavior is actually observed at the amplitude-level: so-called KLT (Kawai, Lewellen, Tye) relations relate gravity amplitudes to products of Yang-Mills amplitudes; this *double-copy* structure has led to the paradigm (Gravity) = (Yang-Mills)<sup>2</sup>; see e.g. [67] for details.

<sup>6</sup> This momentum would then be given by  $P^{\mu} = \int d^d x T^{0\mu}$ .

<sup>7</sup> By the metric equation of motion,  $0 = \frac{\delta S}{\delta g_{\mu\nu}} \sim T^{\mu\nu}$ .

in terms of the Wilsonian renormalization energy scale u of the gauge theory. In particular, the RG equations for a generic coupling constant  $\lambda$  are *local*,

$$u\partial_u\lambda(u) = \beta(\lambda(u)), \tag{2.2.1}$$

where  $\partial_u = \frac{\partial}{\partial u}$  and  $\beta$  is the beta-function which encodes the energy-scale dependency of the coupling (see e.g. [69, 70]). Furthermore, if we want gravity from pure Yang-Mills, we must require that it be strongly coupled so that quantum effects are dominant which are not well understood—classical Yang-Mills and general relativity are certainly not the same thing. To simplify our considerations, we take the simplest possible RG flow of vanishing  $\beta$ -function, leading to a conformal field theory. This implies that—in a Lorentz-invariant theory, which we tacitly assume to be dealing with—the scale transformation  $x^{\mu} \rightarrow \lambda x^{\mu}$ , with  $\mu = 1, \ldots, d$  and  $\lambda \in \mathbb{R}$ , is a symmetry, and, if u is an energy scale, it must behave under scale transformations as  $u \rightarrow u/\lambda$  by dimensional analysis, giving rise to a SO(1,1) symmetry. Writing down a (d + 1)-dimensional metric with u as the extra direction, which is Poincaré invariant and respects SO(1,1) symmetry, we are invariably led to the result  $ds^2 = \frac{u^2}{l^2}\eta_{\mu\nu}dx^{\mu}dx^{\nu} + \frac{L^2}{u^2}du^2$ , which, on making the change of coordinates  $z := L^2/u$  takes the form

$$ds^{2} = \frac{L^{2}}{z^{2}} \left( \eta_{\mu\nu} dx^{\mu} dx^{\nu} + dz^{2} \right), \qquad (2.2.2)$$

for *L* the AdS radius, which we recognize as the metric (of the Poincaré patch) of  $AdS_{d+1}$  (see Appendix A). This beautiful relation between the renormalization group (RG) equations and geometry (i.e. general relativity (GR)) is occasionally stylized as

$$RG = GR. \tag{2.2.3}$$



Figure 2.1: The left figure is an illustration of how the extra "radial" direction of the bulk acts as the resolution scale of the field theory. This course-graining naturally leads to AdS space in the Poincaré patch, as shown on the right, which has boundary  $\mathbb{R}^{d-1,1}$ .

The metric (2.2.2) is a solution to a large class of dynamical theories fulfilling the "Landau criterion" of being invariant under diffeomorphisms while at the same time having a minimal number of derivatives; these have the schematic Einstein-Hilbert form

$$S = \frac{1}{16\pi G} \int d^{d+1}x \sqrt{-g} \left[ R - 2\Lambda + \dots \right], \qquad (2.2.4)$$

where  $2\Lambda = -d(d-1)/L^2$  in order for the AdS metric (2.2.2) to be a solution. The AdS length scale L represents the radius of curvature, i.e.  $R \sim L^{-2}$ , implying that the gravity theory is *classical* (and in particular "weakly coupled") in the regime  $L \gg \ell_p$ . As we shall see, the dual CFT is precisely *strongly coupled* in this limit. We now turn to a more formal discussion of the AdS/CFT correspondence.

#### 2.3 The $ADS_5/CFT_4$ correspondence

While there are in fact various incarnations of the AdS/CFT correspondence<sup>8</sup>, we focus our exposition on the *vanilla* version: AdS<sub>5</sub>/CFT<sub>4</sub>, which relates the dynamics of (3 + 1)-dimensional  $\mathcal{N} = 4$ 

<sup>8</sup> For example, AdS<sub>7</sub>/CFT<sub>6</sub>, which involves a horizon limit of M<sub>5</sub> branes, where the CFT is a 6d (2,0)-superconformal field theory; see [42] for a review. This field theory has extremely interesting ties to the geometric Langlands duality, see [71].

super Yang-Mills theory (SYM)—see appendix C for a short review—to type IIB superstring theory on  $AdS_5 \times S^5$ . The *derivation* for this correspondence—which is, admittedly, somewhat heuristic<sup>9</sup>—is considered in appendix D and involves two perspectives of D<sub>3</sub> branes. We state the correspondence below:

**The AdS**<sub>5</sub>/**CFT**<sub>4</sub> **Conjecture:**  $\mathcal{N} = 4$  SYM with gauge group SU(N) and coupling  $g_{YM}$  is dynamically equivalent to type IIB superstring theory with string length  $l_s = \sqrt{\alpha'}$  and string coupling  $g_s$  on  $AdS_5 \times S^5$  with AdS radius L and N units of  $F_{(5)}$  flux on  $S^5$ . The parameters of each side of the correspondence are mapped to each other via

$$g_{YM}^2 = 4\pi g_s, \quad and \quad 2g_{YM}^2 N = \frac{L^4}{(\alpha')^2}.$$
 (2.3.1)

In particular, the results of appendix **D** allow us identify three versions of the conjecture (see also e.g. [43]):

Table 2.1: Incarnations of the AdS<sub>5</sub>/CFT<sub>4</sub> Correspondence. Here,  $\lambda = g_{YM}^2 N$  is the t'Hooft coupling (cf. appendices C and D).

	$\mathcal{N}=4~\mathrm{SYM}$	IIB on $AdS_5 \times S^5$
Strongest form	any $N$ and $\lambda$	Quantum string theory, $g_s \neq 0$ , $\alpha'/L^2 \neq 0$
Strong form	$N \rightarrow \infty$ , $\lambda$ fixed but arbitrary	Classical string theory, $g_s \rightarrow 0$ , $\alpha'/L^2 \neq 0$
Weak form	$N \rightarrow \infty$ , $\lambda$ large	Classical supergravity, $g_s \rightarrow 0, \alpha'/L^2 \rightarrow 0$

The derivation found in appendix D presents a heuristic derivation of the weakest form of the conjecture as described above.

#### 2.3.1 Counting Degrees of Freedom and the UV/IR Connection

To furnish a realization of the holographic principle, the Bekenstein-Hawking relation (2.1.2) should be satisfied, that is; the area of the boundary A should equal the number of degrees of freedom  $N_d$ , or, in other words, the maximum entropy, which we schematically write as

$$\frac{A}{4G} = N_d. \tag{2.3.2}$$

Naïvely, both sides of the above equality are infinite: on the boundary  $\mathcal{N} = 4$  SYM, the entropy is infinite since the theory is conformal, which means that it has degrees of freedom at arbitrarily small scales, while the boundary area of AdS has infinite area. To verify (2.3.2), we therefore require regularization, which, following [12, 73], is achieved for the boundary by replacing it with a sphere just inside the boundary at  $z = \delta$ , the resulting area of which is given by  $A \sim L^3/\delta^3$ . The total number of *cells* making up the sphere is  $\sim \delta^{-3}$ , while the number of field degrees of freedom for SU(N) $\mathcal{N} = 4$  SYM is  $\sim N^2$ , implying that (up to numerical factors)

$$N_d \sim \frac{N^2}{\delta^3} \sim \frac{AN^2}{L^3} \sim \frac{AL^5}{(\alpha')^2 g_s^2} \sim A/G_5,$$
 (2.3.3)

where we have used (D.1.26) and identified the five-dimensional Newton constant  $G_5 = (\alpha')^2 g_s^2 L^{-5}$ . This is precisely the desired result.

Now, from the discussion in section 2.2, we immediately infer what has become known as the UV/IR connection: when the energy scale u is small—corresponding to the IR of the field theory—the corresponding AdS space has a large value of z, i.e. we are in the deep interior; the UV—and *vice versa*. This is precisely the UV/IR connection of [73] (see also [72, 74]), where they argue for it by using the behaviours of certain correlators in supergravity and Yang-Mills theory.

#### 2.4 THE HOLOGRAPHIC DICTIONARY FOR ADS/CFT I: THE FIELD/OPERATOR MAP

The AdS/CFT correspondence provides a map between operators of  $\mathcal{N} = 4$  SYM in certain representations of PSU(2,2|4) and supergravity fields showing up in the Kaluza-Klein tower of type IIB

<sup>9</sup> But the correspondence has so far passed all tests; many of which have been very non-trivial [43, 72].

SUGRA on AdS<sub>5</sub> reduced on  $S^5$ . The principal<sup>10</sup> field theory operators involved in the mapping are gauge invariant<sup>11</sup> 1/2 BPS primary operators  $\mathcal{O}_{\Delta_i}$  of conformal weight  $\Delta_i$ , i.e. the operators satisfy  $[D, \mathcal{O}_{\Delta_i}(0)] = -i\Delta_i \mathcal{O}_{\Delta_i}(0), [K_{\mu}, \mathcal{O}_{\Delta_i}(0)] = 0$ , as well as  $[S_{a\alpha}, \mathcal{O}_{\Delta}] = 0 = [\bar{S}^a_{\dot{\alpha}}, \mathcal{O}_{\Delta}]$  for all  $a = 1, ..., \mathcal{N}$  and all  $\alpha, \dot{\alpha} = 1, 2$ , as well as  $[\mathcal{Q}^a_{\alpha}, \mathcal{O}] = 0$  for at least one of the Poincaré supercharges (see also appendix **B**). The 1/2 BPS operators of conformal weight  $\Delta = k$  all have the form<sup>12</sup> [43]

$$\mathcal{O}_{\Delta}(x) = \operatorname{Str}\left[\phi^{\{i_1}\cdots\phi^{i_k\}}\right],\tag{2.4.1}$$

where the  $\phi^i$  are the scalar fields of  $\mathcal{N} = 4$ . Operators of this form are dual to the single particle (elementary) fields of type IIB SUGRA on  $AdS_5 \times S^5$ , while higher trace operators are dual to bound states of one-particle states. A concrete mapping can be found in [45]. In particular, to actually derive the mapping, type IIB SUGRA is Kalaza-Klein compactified on  $S^5$ , leading to an expansion of the SUGRA fields in spherical harmonics  $Y_{\Delta}$  on the sphere, which are labelled by the rank  $\Delta$  of the totally symmetric traceless representations of  $\mathfrak{so}(6) \simeq \mathfrak{su}(4)$ . Taking y to be the coordinates on  $S^5$ , we get

$$\varphi(x,y) = \sum_{\Delta=0}^{\infty} \varphi_{\Delta}(z) Y_{\Delta}(y).$$
(2.4.2)

The reduced fields may acquire mass from the compactification process, and an involved calculation shows that the conformal dimension of the field theory operator  $\mathcal{O}_{\Delta}$  is mapped to the mass as depicted in the table below (generalized to the AdS<sub>*d*+1</sub>/CFT<sub>*d*</sub> correspondence) [43].

Type of field	Relation between $m$ and $\Delta$
Scalars, massive spin-2 fields	$m^2 L^2 = \Delta(\Delta - d)$
Massless spin-2 fields	$\Delta = d$ ,
<i>p</i> -form fields	$m^2 L^2 = (\Delta - p)(\Delta + p - d)$
Spin-1/2, spin-3/2	$ m L = \Delta - d/2$
Rank <i>s</i> symmetric traceless tensor	$m^2L^2 = (\Delta + s - 2)(\Delta - s + 2 - d)$

Table 2.2: Relation between mass and conformal dimension for various fields.

Unprotected non-BPS operators, such as the Konishi operator [75], are believed to be dual to massive type IIB string modes not present in the low energy SUGRA description.

#### 2.4.1 Boundary Asymptotics

Our considerations above were motivated by symmetry arguments alone. We can, however, be more explicit by examining the boundary behaviour of the SUGRA fields [12, 76]. For simplicity, we consider a toy model consisting of a massive scalar  $\varphi$  in the bulk

$$S = -\frac{1}{2} \int_{\text{AdS}} dz d^d x \sqrt{-g} \left[ g^{mn} \partial_m \varphi \partial_m \varphi + m^2 \varphi^2 \right].$$
(2.4.3)

The scalar equation of motion becomes the usual Klein-Gordon equation,

$$\left(\Box_g - m^2\right)\varphi = 0, \tag{2.4.4}$$

where the Laplacian reads

$$\Box_g \varphi = \frac{1}{\sqrt{-g}} \partial_m \left( \sqrt{-g} g^{mn} \partial_n \varphi \right), \qquad (2.4.5)$$

<sup>10</sup> The single trace 1/2 BPS operators are the principal fields in the sense that higher trace operators may be constructed in terms of them using the OPE [45].

<sup>11</sup> There is no gauge group on the SUGRA side of the correspondence, so the operators on the field theory side must be gauge invariant.

<sup>12</sup> Gauge invariance and the requirement of superconformal primarity constrains the single trace 1/2 BPS operators to involve only scalars, see [45] for details.

with  $g_{mn}$  the AdS<sub>*d*+1</sub> metric in Poincaré coordinates,  $ds^2 = g_{mn}dx^m dx^n = \frac{L^2}{z^2} (dz^2 + \eta_{\mu\nu}dx^{\mu}dx^{\nu})$ , so that

$$\Box_{g} = \frac{z^{d+1}}{L^{d+1}} \partial_{m} \left( \frac{L^{d+1}}{z^{d+1}} g^{mn} \partial_{n} \right) = \frac{z^{d+1}}{L^{d+1}} \partial_{z} \left( \frac{L^{d-1}}{z^{d-1}} \partial_{z} + \frac{L^{d-1}}{z^{d-1}} \partial^{2} \right)$$
(2.4.6)

$$= \frac{z^2}{L^2} \left( \partial_z^2 - (d-1)z^{-1}\partial_z + \partial^2 \right), \qquad (2.4.7)$$

where  $\partial^2 = \eta_{\mu\nu}\partial^{\mu}\partial^{\nu}$ . Employing a plane wave ansatz  $\varphi(z, x) = e^{ip_{\mu}x^{\mu}}\varphi_p(z)$ , the Klein-Gordon equation becomes a Bessel-like ODE,

$$z^{2}\partial_{z}^{2}\varphi_{p}(z) - (d-1)z\partial_{z}\varphi_{p}(z) - (m^{2}L^{2} + p^{2}z^{2})\varphi_{p}(z) = 0,$$
(2.4.8)

which can be solved with e.g. *Mathematica*—the solution is a superposition of two Bessel functions (each of which corresponds to a solution itself), which gives rise to the  $z \rightarrow 0$  boundary expansion (to lowest order in z)

$$\varphi(z,x) \sim \varphi_{(0)}(x) z^{\Delta_{-}} + \varphi_{(+)}(x) z^{\Delta_{+}},$$
(2.4.9)

where  $\Delta_{\pm} = \frac{d}{2} \pm \frac{1}{2}\sqrt{d^2 + 4m^2L^2}$ , and where the fields  $\varphi_{(0)}$  and  $\varphi_{(+)}$  live on the boundary—i.e. they are a function only of the boundary coordinates. The solution that goes as  $\sim z^{\Delta_+}$  is known as the normalizable solution, while the solution that behaves as  $\sim z^{\Delta_-}$  is called the non-normalizable solution<sup>13</sup>. Requiring the quantity in the square root to be positive in order to avoid imaginary fields, we get the Breitenholmer-Freedman bound [77],

$$m^2 L^2 \ge -\frac{d^2}{4},$$
 (2.4.10)

which implies that AdS is stable in the presence of negative mass-squared scalars as long as the bound above is not violated<sup>14</sup>. Note further that  $\Delta_+ \ge \Delta_-$  as well as  $\Delta_- = d - \Delta_+$ , which implies that under boundary conformal rescalings  $x \to x' = \lambda x$ , the boundary field  $\varphi_{(0)}(x)$  transforms as (defining  $z' = \lambda z$ )

$$\varphi'_{(0)}(\lambda x) = \lim_{z' \to 0} (z')^{-\Delta_{-}} \varphi'(z', x') = \lambda^{-\Delta_{-}} \lim_{z \to 0} z^{-\Delta_{-}} \varphi(z, x) = \lambda^{d-\Delta_{+}} \phi_{(0)}(x),$$
(2.4.11)

where we have used invariance of the bulk field  $\varphi(z, x)$  under the AdS isometry  $(z, x) \rightarrow (z', x') = (\lambda z, \lambda x)$ , i.e.  $\varphi'(z', x') = \varphi(z, x)$ . From this we infer that  $\varphi_{(0)}$  transforms as a source for a conformal primary operator with dimension  $\Delta_+$ , leading us to identify the boundary field  $\varphi_{(0)}$  as a source for a dual field theory operator  $\mathcal{O}_{\Delta_+}$ ; and a similar analysis reveals that  $\varphi_{(+)}$  is the vacuum expectation value (VEV) of  $\mathcal{O}_{\Delta_+}$ .

A careful analysis reveals, however, that this identification of source and VEV is only true for the mass range  $-d^2/4 \le m^2L^2 < 0$  with  $\Delta < d$ . For  $(d-2)/2 \le \Delta < d/2$ , it turns out that we have to identify the conformal dimension of the field with  $\Delta_-$ , so on the overlap  $-d^2/4 < m^2L^2 \le -d^2/4 + 1$ , the identification of VEV and source of the field theory operator can be interchanged (see [43] for details.)

For completeness, we bring a version of the AdS/CFT dictionary:

<sup>13</sup> This nomenclature comes from the fact that the action evaluated on the normalizable solution is finite, while the action evaluated on the other is not.

<sup>14</sup> In fact, violating this bound will give rise to tachyons, just as  $m^2 < 0$  in ordinary field theory. Such violations allow us to encode spontaneous symmetry breaking in the boundary field theory from a bulk perspective [11].

Boundary:	Bulk:
Field theory	Supergravity
Scalar operator/order parameter $\mathcal{O}$	Scalar field
Source of operator	Boundary value of field
	(leading coefficient of non-normalizable solution)
	Boundary value of radial momentum of the field
VEV of operator	(leading coefficient of the normalizable solution;
	sub-leading to the non-normalizable solution)
Conformal dimension of operator	Mass of field (see table 2.2)
spin/charge of operator	Spin/charge of field
Energy momentum tensor $T^{\mu\nu}$	Metric field $g_{mn}$
Global internal symmetry current $J^{\mu}$	Maxwell field $A_m$
Fermionic operator $\mathcal{O}_\psi$	Dirac field $\psi$
RG flow	Evolution in the radial AdS direction
No. of degrees of freedom	Radius L of AdS space
Global spacetime symmetry	Local isometry
Global internal symmetry	local gauge symmetry
Finite temperature	Black hole Hawking temperature
	(or radius of compact Euclidean time circle)
Chemical potential/charge density	Boundary values of electrostatic potential $A_t$
	(time component)
Free energy	On-shell value of action
Phase transition	Instability of black holes
Wilson line along ${\cal C}$	String worldsheet with endpoints on ${\cal C}$
Entanglement entropy of area ${\cal A}$	Minimal surface $\Sigma$ with boundary $\partial \Sigma = \mathcal{A}$
Quantum anomalies	Chern-Simons terms

Table 2.3: The basic dictionary for the AdS/CFT correspondence. Adapted from [11].

#### 2.5 THE HOLOGRAPHIC DICTIONARY FOR ADS/CFT II: THE GKPW RULE

The GKPW rule—named after Gubser, Klebanov, Polyakov and Witten, who discovered it in 1998 [78, 79]—links the partition functions of the two sides of the AdS/CFT correspondence. On the field theory side, the  $\varphi_{(0)}$  acts as a source for the operator  $\mathcal{O}_{\Delta}$ —as we argued above—so the partition function (in Euclidean signature) for the CFT takes the form

$$Z_{\text{CFT}}[\varphi_{(0)}] = e^{-W[\varphi_{(0)}]} = \left\langle \exp\left(\int d^d x \; \varphi_{(0)}(x) \mathcal{O}_{\Delta}\right) \right\rangle_{\text{CFT}}.$$
(2.5.1)

On the AdS side—for the weak form of the conjecture<sup>15</sup>—the partition function is given in terms of  $S_{SUGRA}^{on-shell}[\varphi]$ , where  $\varphi$  are fields (with possible indices suppressed) on AdS<sub>5</sub> reduced on  $S^5$ . The Ad-S/CFT correspondence, then, translates into the following statement relating the generating functional  $W[\varphi_{(0)}]$  of connected diagrams to the SUGRA action

$$W[\varphi_{(0)}] = S_{\text{SUGRA}}^{\text{on-shell}}[\varphi] \Big|_{\substack{z \to 0 \\ z \to 0}} z^{\Delta - d} \varphi(z, x) = \varphi_{(0)}(x)}.$$
(2.5.3)

However, the on-shell value of the SUGRA action is fraught with IR divergences due to the infinite volume of AdS, while the field theory action will contain UV divergences. The method of holographic

$$\left\langle \exp\left(\int \mathrm{d}^{d}x \; \varphi_{(0)}(x)\mathcal{O}_{\Delta}\right) \right\rangle_{\mathrm{CFT}} = \left. Z_{\mathrm{IIB}\;\mathrm{String}} \right|_{\substack{z \to 0 \\ z \to 0}} z^{\Delta - d} \varphi(z, x) = \varphi_{(0)}(x) \;.$$
(2.5.2)

The string partition function is not known. The weak form of the correspondence, which we expound on presently, amounts to a saddle point approximation of the string partition function.

<sup>15</sup> For the strongest form of the conjecture, the correspondence relates the partition function of the CFT to the partition function of the full (type IIB) string theory. Letting the generalized quantities (i.e. not necessarily scalars)  $\mathcal{O}_{\Delta}$  and  $\varphi$  be dual to each other in the usual sense, the strongest form can be expressed as

renormalization (see chapter 3) can then be applied to subtract divergences from the bulk action, which, by the AdS/CFT correspondence, simultaneously renormalizes the field theory. When this process has been carried out, we can compute *n*-point correlation functions as follows,

$$\left\langle \mathcal{O}_{\Delta_{1}}(x_{1})\dots\mathcal{O}_{\Delta_{n}}(x_{n})\right\rangle_{\text{Connected}} = (-1)^{n} \frac{\delta S_{\text{SUGRA}}^{\text{on-shell,ren}}[\varphi_{\Delta_{i}}]|_{\substack{\lim_{z\to 0} z^{\Delta-d}\varphi_{\Delta_{i}}(z,x)=\varphi_{(0)\Delta_{i}}(x)}}{\delta\varphi_{(0)\Delta_{1}}(x_{1})\cdots\delta\varphi_{(0)\Delta_{n}}(x_{n})}\right|_{\varphi_{(0)\Delta_{i}}=0}.$$

$$(2.5.4)$$

Remarkably, holographic renormalization is not required for the two-point function for a massless scalar<sup>16</sup> [44]; so without having to deal with the hassle that is holographic renormalization just yet, we may apply the formalism above to compute the two-point function for a massless scalar in the CFT. Using Euclidean  $AdS_{d+1}$  (Lobachevski space) with metric  $ds^2 = \frac{L^2}{z^2} (dz^2 + \delta_{\mu\nu} dx^{\mu} dx^{\nu})$ , it is now convenient to introduce the bulk-to-boundary propagator,  $K_{\Delta}$ , which relates the bulk field  $\varphi_{\Delta}(z, x)$ —subject to the boundary condition  $\varphi(z, x) \sim \varphi_{(0),\Delta}(x) z^{d-\Delta}$  for  $z \to 0$ —to the boundary field  $\varphi_{(0),\Delta}(x)$ ,

$$\varphi_{\Delta}(z,x) = \oint_{\partial \text{AdS}} d^d y \ K_{\Delta}(z,x;y) \varphi_{(0),\Delta}(y).$$
(2.5.5)

In order to actually determine the bulk-to-boundary propagator, we first identify the bulk-to-bulk propagator  $G_{\Delta}(z, x; w, y)$ . Now, the Klein-Gordon equation for a massless scalar with a source reads  $\Box_g \varphi_{\Delta}(z, x) = J(z, x)$ , the solution to which can be written as

$$\varphi_{\Delta}(z,x) = \int_{\text{AdS}} dw d^d y \sqrt{g} G_{\Delta}(z,x;w,y) J(w,y), \qquad (2.5.6)$$

implying that the bulk-to-bulk propagator satisfies,

$$\Box_g G_\Delta(z, x; w, y) = \frac{\delta(z - w)\delta^d(x - y)}{\sqrt{g}}.$$
(2.5.7)

The solution to this differential equation is given by a hypergeometric function in the chordal distance  $\xi$  of the geodesic connecting the points (z, x) and (w, y) [45]. From  $G_{\Delta}(z, x; w, y)$ , we can get the bulk-to-boundary propagator by taking the limit  $w \to 0$  in an appropriate way<sup>17</sup>; the result is<sup>18</sup>

$$K_{\Delta}(z,x;y) = \underbrace{\frac{\Gamma(d)}{\pi^{d/2}\Gamma(d/2)}}_{=:C_d} \left(\frac{z}{z^2 + (x-y)^2}\right)^d.$$
(2.5.9)

On the boundary,  $z \rightarrow 0$ , the Lorentzian form of the propagator implies that

$$\lim_{z \to 0} K_{\Delta}(z, x; y) = \overbrace{z^{\Delta - d} \delta^d(x - y)}^{\text{General case for massive scalar}} = \delta^d(x - y)$$
(2.5.10)

Equipped with the bulk-to-boundary propagator, and its boundary behavior, we're ready to calculate the massless two-point function. The on-shell (Euclidean) action reads

$$S_{\text{on-shell}} = -\frac{1}{2} \int_{\text{AdS}} dz d^d x \sqrt{g} g^{mn} \partial_m \varphi \partial_m \varphi$$
(2.5.11)

$$=\underbrace{\frac{1}{2}\int_{\mathrm{AdS}}\mathrm{d}z\mathrm{d}^{d}x\;\sqrt{g}\varphi\Box_{g}\varphi}_{=0}+\frac{1}{2}\oint_{\partial\mathrm{AdS}}\mathrm{d}^{d}x\;\sqrt{h}\varphi\hat{\partial}\varphi\qquad(2.5.12)$$

$$= \frac{1}{2} \oint_{\partial AdS} d^d x \sqrt{h} \varphi \hat{\partial} \varphi, \qquad (2.5.13)$$

16 By table 2.2, masslessness implies that  $\Delta = d$ .

17 An illuminating way of deriving the precise relation between  $K_{\Delta}$  and  $G_{\Delta}$  involves Green's second identity,

$$\int_{\mathcal{M}} \mathrm{d}^{d+1}x \sqrt{-g} \left( \phi(\Box - m^2)\psi - (\phi \leftrightarrow \psi) \right) = \int_{\partial \mathcal{M}} \mathrm{d}^d x \sqrt{-\gamma} \left( \phi n \cdot \partial \psi - (\phi \leftrightarrow \psi) \right),$$

where *n* is the normal to  $\partial M$  and then setting  $\phi = G_{\Delta}$  and  $\psi = K_{\Delta}$ ; see e.g. [43] for details.

18 For a massive scalar with conformal dimension  $\Delta$ , the result reads

$$K_{\Delta}(z,x;y) = \underbrace{\frac{\Gamma(\Delta)}{\pi^{d/2}\Gamma(\Delta - d/2)}}_{=:C_{\Delta}} \left(\frac{z}{z^2 + (x-y)^2}\right)^{\Delta}.$$
(2.5.8)

where  $h_{\mu\nu}$  is the induced metric on the boundary,  $z \to 0$ , which gives  $h_{\mu\nu} = \frac{L^2}{z^2} \eta_{\mu\nu} dx^{\mu} dx^{\nu}$ , and where  $\hat{\partial} = n^m \partial_m$  is the normal derivative outward directed from the boundary, which thus clearly points in the *z* direction,  $n^m = (n, 0, ..., 0)$  while normality dictates that  $1 = g_{mn}n^m n^n = n^2 \frac{L^2}{z^2}$ , so  $n = \frac{z}{L}$ . Hence, since the leading behavior as  $z \to 0$  of the field is  $\varphi(z, x) \sim z^{d-\Delta}\varphi_{(0)}(x) = \varphi_{(0)}(x)$ ,

$$\sqrt{h}\varphi(z,x)\hat{\partial}\varphi(z,x) = \left(\frac{L}{z}\right)^{d-1}\varphi(z,x)\partial_z\varphi(z,x) = \left(\frac{L}{z}\right)^{d-1}\varphi(z,x)\partial_z\oint_{\partial \mathrm{AdS}}\mathrm{d}^d y \ K_{\Delta}(z,x;y)\varphi_{(0)}(y)$$
(2.5.14)

$$= \oint_{\partial \mathrm{AdS}} \mathrm{d}^{d} y \ C_{d} \left(\frac{L}{z}\right)^{d-1} \varphi(z, x) \frac{d\left((x-y)^{2}-z^{2}\right) z^{d-1}}{\left((x-y)^{2}+z^{2}\right)^{d+1}} \varphi_{(0)}(y)$$
(2.5.15)

$$\stackrel{z \to 0}{\longrightarrow} C_d dL^{d-1} \oint_{\partial \text{AdS}} \mathrm{d}^d y \; \frac{\varphi_{(0)}(x)\varphi_{(0)}(y)}{(x-y)^{2d}},\tag{2.5.16}$$

which means that the on-shell action takes the form,

$$S_{\text{on-shell}} = \frac{C_d dL^{d-1}}{2} \oint_{\partial \text{AdS}} \oint_{\partial \text{AdS}} d^d x d^d y \; \frac{\varphi_{(0)}(x)\varphi_{(0)}(y)}{(x-y)^{2d}}.$$
(2.5.17)

With this, the two-point function for scalar conformal operators  $\mathcal{O}_d$  of dimension  $\Delta = d$  can be determined using the GKPW rule (2.5.4):

$$\langle \mathcal{O}_d(x_1)\mathcal{O}_d(x_2)\rangle = \left.\frac{\delta S_{\text{on-shell}}}{\delta \varphi_{(0)}(x_1)\delta \varphi_{(0)}(x_2)}\right|_{\varphi_{(0)}=0} = \frac{C_d dL^{d-1}}{(x_1 - x_2)^{2d}},$$
(2.5.18)

in agreement with the conformal result (B.1.16). For a massive scalar dual to operators of weight  $\Delta$  (the operators need to have the same conformal dimension, otherwise the two-point function vanishes identically, see appendix B), the result is [43]

$$\langle \mathcal{O}_{\Delta}(x_1)\mathcal{O}_{\Delta}(x_2)\rangle = L^{d-1}C_{\Delta}\frac{2\Delta - d}{(x_1 - x_2)^{2\Delta}},\tag{2.5.19}$$

which reduces to our result for  $\Delta = d$ . Naïvely, the procedure for the massless scalar should work for the massive scalar case as well, but it does not give the correct prefactor: in that case, we have to regulate the bulk-to-boundary propagator using holographic renormalization, which will be the topic of chapter 3.

#### 2.6 WITTEN DIAGRAMS

Introduced by Witten in [79], Witten diagrams are pictorial representations of the GKPW relation (2.5.1) and (2.5.4)—i.e. a way to obtain CFT correlation functions from diagrams in AdS. In the supergravity approximation, (2.5.3)—corresponding to the large-*N* limit of the CFT—only tree diagrams contribute, with loop diagrams inducing 1/N corrections to the boundary correlators. Pictorially, we may represent AdS as a sphere and draw the supergravity fields in a manner similar to the one employed when drawing Feynman diagrams (see e.g. [69, 70] for standard references), but with external lines connecting to the boundary. For concreteness, consider a scalar operator of dimension  $\Delta$ , which is dual to the bulk field  $\varphi$  with interaction

$$S_{\rm int} = \int d^{d+1}x \,\sqrt{-g} \left(-\frac{\lambda}{3!}\varphi^3\right). \tag{2.6.1}$$

The Feynman rules for Witten diagrams are summarized in table 2.4 below.

	0
Internal line	Bulk-to-bulk propagator $G_{\Delta}(z, x; w, y)$ (see (2.5.7))
Vertex	$\int dz d^d x \sqrt{-g} \lambda$ , i.e. we integrate over <i>all</i> of AdS
External line (i.e. connected to b'dary)	A bulk-to-boundary propagator $K_{\Delta}(z, x; y)$ (see (2.5.8))

Table 2.4: Feynman rules for Witten Diagrams.

Since their introduction, many exciting developments involving Witten diagrams have taken place. For example, in [80] (see also [81]), it is proposed that the holographic dual of conformal blocks are so-called *geodesic Witten diagrams*, that is, Witten diagrams where vertices are integrated—not over all of AdS—but rather over geodesics connecting each pair of boundary points. Other interesting developments (see e.g. [82, 83]) include the use of the Mellin transform to deduce a more useful set of Feynman rules in "Mellin space"—leading to the concept of *Mellin amplitudes*—as well as an implementation of BCFW-like (see fig. C.1) recursive techniques applicable for Witten diagrams [84].

2.6.0.1 Holographic Computation of the CFT Three-Point Correlator



Figure 2.2: Witten diagram for  $\langle \mathcal{O}_{\Delta} \mathcal{O}_{\Delta} \mathcal{O}_{\Delta} \rangle$  due to the interaction (2.6.1).

To showcase the power of the formalism, we now proceed to calculate the three-point function using Witten diagrams (see also [47, 85]). We already know the result: it is given by (B.1.17). Taking the interaction for simplicity to be (2.6.1), where  $\varphi$  is dual to an operator  $\mathcal{O}_{\Delta}$  of dimension  $\Delta$ , the leading order contribution to the correlation function  $\langle \mathcal{O}_{\Delta} \mathcal{O}_{\Delta} \mathcal{O}_{\Delta} \rangle$  is given by the single diagram of figure 2.2. By use of the Feynman rules of table 2.4, the value of this diagram is readily obtained<sup>19</sup>:

$$\langle \mathcal{O}(x_1)_{\Delta} \mathcal{O}_{\Delta}(x_2) \mathcal{O}_{\Delta}(x_3) \rangle = \int_{\text{AdS}} dz d^d x \ \sqrt{-g} \lambda \prod_{i=1}^3 K_{\Delta}(z, x; x_i).$$
(2.6.2)

To tackle this integral, we—following [47]—use Feynman parametrization, allowing us to perform the integral over AdS to obtain<sup>20</sup>

$$\langle \cdots \rangle = C_{\Delta}^{3} \lambda \frac{\pi^{d/2} \Gamma \left( 3\Delta/2 - d/2 \right) \Gamma \left( 3\Delta/2 \right)}{2\Gamma \left( \Delta \right)^{3}} \int_{0}^{\infty} \left( \prod_{i} d\alpha_{i} \right) \delta \left( \sum_{j} \alpha_{j} - 1 \right) \frac{\prod_{k} \alpha_{k}^{\Delta - 1}}{\left[ \sum_{\ell < q} \alpha_{\ell} \alpha_{q} x_{\ell q}^{2} \right]}, \quad (2.6.3)$$

where we have defined  $x_{ij} = x_i - x_j$ . Changing integration variables to  $\beta_i$  defined via  $\alpha_1 = \beta_1$  and  $\alpha_i = \beta_1 \beta_i$  for  $i \in \{2, 3\}$ , the integral over the  $\beta_i$  takes the form

$$\langle \cdots \rangle = C_{\Delta}^{3} \lambda \frac{\pi^{d/2} \Gamma \left( 3\Delta/2 - d/2 \right) \Gamma \left( 3\Delta/2 \right)}{2\Gamma \left( \Delta \right)^{3}} \int_{0}^{\infty} \left( \prod_{i>1} d\beta_{i} \right) \frac{\prod_{k>1} \beta_{k}^{\Delta-1}}{\left[ \sum_{\ell>1} \beta_{\ell} \left( x_{1\ell}^{2} + \sum_{q>\ell} \beta_{q} x_{\ell q}^{2} \right) \right]}$$
(2.6.4)  
$$= -\frac{\lambda \Gamma \left( (3\Delta - d)/2 \right)}{2\pi^{d}} \left( \frac{\Gamma (\Delta/2)}{\Gamma \left( \Delta - d/2 \right)} \right)^{3} \frac{1}{(x_{12} x_{13} x_{23})^{\Delta}},$$
(2.6.5)

in agreement with the result obtained from conformal field theory, cf. (B.1.17).

2.7 OUTLOOK

The AdS/CFT correspondence remains a veritable cornucopia for new, interesting physics and is an extremely active area of research. Although still not fully understood, significant progress has been achieved in the years following Maldacena's foundational work, and many formal aspects are now quite well understood.

<sup>19</sup> Note in particular that there is no contribution from the kinetic term, see e.g. [72] for details.

<sup>20</sup> The standard method to compute the three-point function exploits inversion symmetry to constrain the spatial dependence, see e.g. [43, 45].

It has been successfully applied to explain the low viscosity of the quark gluon plasma (see [9]), and a whole program exploring the application of holography in strongly coupled QCD—known as AdS/QCD—has since flourished.

It has also been applied to strongly coupled field theories that arise in condensed matter theories (this endeavour is known as AdS/CMT, see [11] for a recent review), notably holographic superconductors [5, 6] (these are the subject of appendix E).

The AdS/CFT correspondence has been likened to the "hydrogen atom of holography": as a specific realization of the holographic principle, it may well serve as a stepping stone to a much grander array of holographic correspondences, and, in particular, as the most promising route towards an understanding of quantum gravity. Already many other holographic dualities have been proposed, and many of these are independent of string theory. The main topic of this thesis is one such realization of the holographic principle: Lifshitz holography. This chapter—along with the supplementary appendix F—explores many facets of holographic renormalization. This procedure is required to remove the divergences due to the infinite volume of AdS space—corresponding to UV divergences in the dual field theory (recall our interpretation of the radial direction as an energy scale in section 2.3.1).

There are many approaches to this endeavour, and in section **F.1** of the the supplementary appendix **F**, we review the original approach: the *Fefferman-Graham* (FG) approach, developed by de Haro et al. in [86], which relies on the FG theorem of differential geometry [87]. This theorem roughly states that the metric for certain classes of geometries has a certain universal expansion in an appropriate radial coordinate. We introduce the method and provide a detailed exposition of the renormalization of pure gravity in five-dimensional asymptotically locally AdS in section **F.1.1**. Since the FG approach is computationally involved, we will occasionally make use of the symbolic differential geometry package *xAct* for Mathematica [88].

In section F.2 in the same appendix, we review the de Boer-Verline-Verlinde (dBVV) ansatz method, where one writes down an ansatz for the counterterm. This approach involves a bilinear and symmetric operation that we have named the deWitt bracket, which allows for a recursive determination of the unknown coefficients in the counterterm ansatz. This analysis is based on [49, 89].

We start this chapter with a brief review (following [90–92]) of Hamilton-Jacobi (HJ) theory in section 3.1.1, serving as an appetizer to the HJ approach to holographic renormalization presented in section 3.1.2, which is based on [50, 51, 93]. This method requires us to determine the Hamiltonian of the model we consider, and to this end we employ a radial ADM formalism and illustrate the principles using a generic model consisting of a scalar  $\phi$  with some potential  $V(\phi)$  coupled to gravity. The HJ equation implies a set of constraints, which can be solved recursively which allows the identification of the divergent parts of the action. In order to solve the constraints order by order, we need a good way to define what we mean by "order": to that end, we introduce an operator  $\delta$ —which is conveniently (for our purposes) chosen to be the dilatation operator  $\delta_D$ —that we use to sort various quantities such as the action and the Hamiltonian in eigenmodes.

In section 3.2, we present the main new result of this chapter: using the HJ approach, we calculate a novel counterterm for a five-dimensional free Einstein-Maxwell-Dilaton (EMD) model—which we call the electromagnetic uplift due to its rôle as an upliftable model for charged Lifshitz holography in chapter 6. We also work out the novel Ward identities satisfied by the VEVs and construct the new FG expansions, which is achieved via the so-called flow equations; a natural ingredient in the HJ approach.

We also comment on a new subtlety regarding the renormalization of p-form fields. This has gone unnoticed in the literature, and will be the topic of an upcoming article by Skenderis and Papadimitriou [52]. Based on insights from our renormalization of the electromagnetic uplift, which contains a one-form, we provide a conjecture for the general p-form case: in particular we present four new regions of parameter space, where holographic renormalization works differently depending on the values of d and p.

Finally, in section 3.3, we illustrate a manifestation of this subtlety by considering hodge dualization of a biscalar theory to an EMD model in d = 2 and generalize the approach to a simple way of determining the counterterm action for a massless *p*-form in  $AdS_{p+2}$ . This has, to our knowledge, not appeared in the literature previously.

#### 3.1 HOLOGRAPHIC RENORMALIZATION AS HAMILTON-JACOBI THEORY

In this section, we describe a method originally devised by Papadimitriou and Skenderis in [94, 95] and further refined in [50, 51]. This method was used by Ross in [27] (see also [96]) to renormalize a particular Einstein-Proca model which supports Lifshitz solutions; this we review in appendix G. For expositional reasons, we consider free scalars coupled to gravity, while our final application of the procedure concerns the electromagnetic uplift, the renormalization of which is new.

#### 3.1.1 Hamilton-Jacobi Theory in a Nutshell

We begin with a review of Hamilton-Jacobi theory as it appears in classical mechanics. Our presentation is based on [90–92].

Consider a mechanical system described by the action functional,

$$S = \int_{\gamma} \mathrm{d}t' \ L(q, \dot{q}; t'), \tag{3.1.1}$$

where  $\dot{q}^i = \partial_t q^i$  and  $\gamma$  is a curve connecting the initial and final positions,  $(q_0, t_0)$  and  $(q_1, t_1)$ , in the configuration space *V*, which is an *n*-dimensional manifold. The coordinates on *V* are the generalized coordinates  $q^i$ , and to each generalized coordinate  $q_i$  corresponds a canonical momentum, given by  $p_i = \frac{dL}{dq^i}$ . The  $p_i$  and the  $q^i$  are independent variables and are the natural coordinates of the Hamiltonian formalism, where we trade the Lagrangian for the Hamiltonian via the Legendre transformation,

$$H(q, p; t) = p_i \dot{q}^i(q, p) - L(p, q; t), \qquad (3.1.2)$$

which dictates the time evolution through Hamilton's equations,

$$\dot{q}^i = \frac{\partial H}{\partial p_i}, \qquad \dot{p}_i = -\frac{\partial H}{\partial q^i}.$$
 (3.1.3)

Note that these imply an additional "Hamilton equation" for explicitly time dependent systems: since  $\dot{H} = \frac{\partial H}{\partial p_i} \dot{p}_i + \frac{\partial H}{\partial q^i} \dot{q}^i + \frac{\partial H}{\partial t} = \frac{\partial H}{\partial t}$ , we infer that -H is the canonical momentum conjugate to t.

When H(p,q) does not explicitly depend on time, the set  $\{p_i, q^i\}$  combine to form the 2*n*-dimensional *phase space* of our system, which is isomorphic to the cotangent bundle of the configuration space,  $T^*V$ .

If the Hamiltonian *does* depend explicitly on time, we must include time as a generalized coordinate, and our extended configuration space  $\overline{V} = V \times \mathbb{R}$  becomes (n + 1)-dimensional, with a corresponding extended phase space  $T^*\overline{V}$ . Note that the Hamiltonian is a mapping giving some real number for each point  $(p_i, q^i)$  in phase space, i.e.  $H : T^*\overline{V} \to \mathbb{R}$ . The (extended) cotangent bundle is naturally endowed with a symplectic structure (see [97] for a review on symplectic geometry), i.e. a closed two-form  $\omega^2$  given by

$$\omega^2 = \mathrm{d}p_i \wedge \mathrm{d}q^i - \mathrm{d}H \wedge \mathrm{d}t, \qquad (3.1.4)$$

which, due to closure, is locally expressible as

$$\omega^2 = \mathrm{d}\lambda,\tag{3.1.5}$$

where  $\lambda = p_i dq^i - H dt$  is the presymplectic one-form<sup>1</sup>. We remark that the presymplectic form is equal to the exterior derivative of the action (3.1.1) considered as a function of final position (i.e, with the initial position fixed) [92]. Consider now a diffeomorphism on  $T^*\overline{V}$  defining a set of new coordinates  $\{P_i(q^i, p_i; t), Q_I(q^i, p_i; t)\}$  (leaving *t* untouched). By demanding that the Lagrangian changes only up to a total derivative<sup>2</sup>,

$$p_i \dot{q}^i - H(p,q;t) = P_i \dot{Q}^i - \tilde{H}(P,Q;t) + \frac{d}{dt} F(q,Q;t), \qquad (3.1.6)$$

we can ensure that the principle of stationary action again produces Hamilton's equations (3.1.3) in the new coordinates,

$$\dot{Q}^{i} = \frac{\partial \widetilde{H}}{\partial P_{i}}, \qquad \dot{P}_{i} = -\frac{\partial \widetilde{H}}{\partial Q^{i}}.$$
(3.1.7)

A diffeomorphism satisfying the above is called *canonical*. Now, observe that  $\frac{d}{dt}F(q,Q;t) = \frac{\partial F}{\partial q^i}\dot{q}^i + \frac{\partial F}{\partial O^i}\dot{Q}^i + \frac{\partial F}{\partial t}$ , which we can plug into the requirement (3.1.6) to obtain

$$\left(p_i - \frac{\partial F}{\partial q^i}\right)\dot{q}^i - H(p,q;t) = \left(P_i + \frac{\partial F}{\partial Q^i}\right)\dot{Q}^i - \widetilde{H}(P,Q;t) + \frac{\partial F}{\partial t}.$$
(3.1.8)

<sup>1</sup> The presymplectic form is also known as the Liouville form, and, occasionally, as the tautological form [97].

<sup>2</sup> Note that we have a lot of freedom in choosing the variables occurring in *F*. In [91], the functions are numbered according to their dependence; for example our *F* here is called  $F_1$ .

Now, the old and the new coordinates are independent, implying that the identity above is true identically only if the coefficients of  $\dot{q}^i$  and  $\dot{Q}^i$  each vanish, leaving us with

$$p_i = \frac{\partial}{\partial q^i} F(q, Q; t), \qquad P_i = \frac{\partial}{\partial Q^i} F(q, Q; t), \qquad \widetilde{H}(P, Q; t) = H(p, q; t) + \frac{\partial}{\partial t} F(q, Q; t).$$
(3.1.9)

In particular, we see that any such *F* generates some canonical function—consequently it is referred to as the generating function. Now, write

$$F(q,Q;t) = -P_i Q^i + \mathcal{S}.$$
(3.1.10)

By taking the exterior derivative of F, which is a section of  $T^*\overline{V}$ , it is easy to show that S = S(q, P; t)—that is, the rewriting (3.1.10) is nothing but a Legendre transformation from the set of coordinates (q, Q; t) to (q, P; t)— and that

$$p_i = \frac{\partial S}{\partial q^i}, \qquad Q^i = \frac{\partial S}{\partial P_i}, \qquad \frac{\partial S}{\partial t} = \frac{\partial F}{\partial t}.$$
 (3.1.11)

As before, *any*  $\mathcal{S}$  generates some canonical transformation, and we choose S such that  $\tilde{H}$  is identically zero, which means that  $\dot{Q}^i = 0 = \dot{P}_i$ , i.e. they are constants of motion, and we write  $P_i = \alpha_i$  and  $Q^i = \beta^i$ . The condition  $\tilde{H} = 0$  is can now be recast in the form

$$H\left(\frac{\partial S}{\partial q}, q; t\right) + \frac{\partial S}{\partial t} = 0, \qquad (3.1.12)$$

where we have used the third relation of (3.1.9) as well as the expression for  $p_i$  in (3.1.11). This is the Hamilton-Jacobi equation. It is a first-order partial differential equation for S, known as *Hamilton's principal function*. Now, note that we can write  $S = S(q, \alpha)$  as well inverting F to obtain  $q^i = q^i(\alpha, \beta; t)$ , which is solution to the mechanical problem in the original coordinates and thus leads to a trajectory  $\gamma$  along which the action is stationary. Along  $\gamma$ , where  $P_i = \alpha_i$ , we can compute the time derivative of S:

$$\frac{\mathrm{d}\mathcal{S}}{\mathrm{d}t} = \frac{\partial\mathcal{S}}{\partial q^i}\dot{q}^i + \frac{\partial\mathcal{S}}{\partial t} = p_i\dot{q}^i - H = L, \qquad (3.1.13)$$

where we have used the relation for  $p_i$  in (3.1.11) and the Hamilton-Jacobi equation (3.1.12), implying that

$$\mathcal{S}(t) = \mathcal{S}(q(t), \alpha, t) = \int_{\gamma(t_0, t)} dt' \ L(t') + \mathcal{S}(t_0), \tag{3.1.14}$$

where  $\gamma(t_0, t)$  is the part of the trajectory  $\gamma$  starting at  $t_0$  and ending at t. Thus, we conclude that *Hamilton's principal function is the on-shell action*. This insight is the crucial ingredient in the Hamilton-Jacobi approach to holographic renormalization.

#### 3.1.2 Hamiltonian Gravity and the Hamilton-Jacobi Equation

In this section, we set up a HJ description of holographic renormalization, where *r* plays the rôle of time. Consider gravity coupled to (possibly massive) free scalars in a (d + 1)-dimensional non-compact manifold  $\mathcal{M}$ , which is described by the action<sup>3</sup>

$$S = -\int_{\mathcal{M}} d^{d+1}x \sqrt{g} \left( R^{(g)} - \frac{1}{2} \partial_{\mu} \partial^{\mu} \phi + V(\phi) \right) + \int_{\partial \mathcal{M}} d^{d}x \sqrt{\gamma} 2K.$$
(3.1.15)

The next ingredient is the identification of a radial coordinate r such that  $r \to \infty$  corresponds to the boundary  $\partial \mathcal{M}$  of  $\mathcal{M}$ , which we require only to cover an open chart  $\mathcal{M}_{\varepsilon}$  in the vicinity of  $\partial \mathcal{M}$ . There are additional subtleties involved if the boundary  $\partial \mathcal{M}$  consists of several disconnected pieces; in that case, a different radial coordinate will have to be used for each component [50]. When such complications have been dealt with, we apply a radial ADM formalism (for a review of the ADM

<sup>3</sup> Note that we'll work in Euclidean signature, but it is straightforward to generalize this to Lorentzian signature [48]; in particular, we can interchange the two with no complications, so we will not be particularly careful about this.

formalism, see [98]), where the metric is parametrized in terms of the lapse *N*, the shift  $N_i$  and the induced metric  $\gamma_{ij}$  on radial leaves  $\Sigma_r$  (i.e. hypersurfaces of constant *r*),

$$ds^{2} = (N^{2} + N_{i}N^{i})dr^{2} + 2N_{i}drdx^{i} + \gamma_{ij}dx^{i}dx^{j}, \qquad (3.1.16)$$

implying that the metric  $g_{\mu\nu}$  is equivalently described by the triplet  $\{N, N_i, \gamma_{ij}\}$ . The curvatures pertaining to  $g_{\mu\nu}$  are then expressible in terms of the curvatures on  $\Sigma_r$  and the extrinsic curvature  $K_{ij}$ , which is given by

$$K_{ij} = \frac{1}{2} \pounds_n g_{ij} = \frac{1}{2N} \left( \dot{\gamma}_{ij} - D_i N_j - D_j N_i \right), \qquad (3.1.17)$$

where  $\dot{\gamma}_{ij} = \partial_r \gamma_{ij}$ ,  $\mathcal{L}_n$  is the Lie derivative in the direction of the normal vector n, and  $D_i$  is the covariant derivative with respect to the induced metric  $\gamma_{ij}$ . The unit normal vector  $n^{\mu}$  to  $\Sigma_r$  is given by

$$n^{\mu} = \left(1/N, -N^{i}/N\right). \tag{3.1.18}$$

Following [50], we now consider some useful identities in the radial ADM formalism that we are employing. Note first that

$$g = \begin{pmatrix} N^2 + N_i N^i & N_i \\ N_i & \gamma_{ij} \end{pmatrix}, \qquad g^{-1} = \begin{pmatrix} 1/N^2 & -N^i/N^2 \\ -N^i/N^2 & \gamma^{ij} + N^i N^j/N^2 \end{pmatrix},$$
(3.1.19)

which means that

$$\sqrt{g} = N\sqrt{\gamma},\tag{3.1.20}$$

which follows from the useful identity

$$\det \begin{pmatrix} A & B \\ C & D \end{pmatrix} = \det (D) \det \left( A - BD^{-1}C \right).$$
(3.1.21)

From the metric (3.1.19), we get the following Christoffel symbols,

$$\Gamma_{rr}^{r} = N^{-1} \left( \dot{N} + N^{i} \partial_{i} N - N^{i} N^{j} K_{ij} \right), \qquad (3.1.22)$$

$$\Gamma_{ri}^{r} = N^{-1} \left( \partial_{i} N - N^{j} K_{ij} \right), \qquad (3.1.23)$$

$$\Gamma_{ij}^r = N^{-1} K_{ij},$$
 (3.1.24)

$$\Gamma^{i}_{rr} = -N^{-1}N^{i}\dot{N} - ND^{i}N - N^{-1}N^{i}N^{j}\partial_{j}N + \dot{N}^{i} + N^{j}D_{j}N^{i} + 2NN^{j}K^{i}_{j} + N^{-1}N^{i}N^{k}N^{l}K_{kl},$$
(3.1.25)

$$\Gamma_{rj}^{i} = -N^{-1}N^{i}\partial_{j}N + D_{j}N^{i} + N^{-1}N^{i}N^{k}K_{kj} + NK_{j}^{i}, \qquad (3.1.26)$$

$$\Gamma_{ij}^{k} = \Gamma_{ij}^{(\gamma)k} + N^{-1} N^{k} K_{ij}.$$
(3.1.27)

In terms of the ADM variables, the Ricci scalar of g is decomposed in the following manner,

$$R^{(g)} = R^{(\gamma)} + K^2 - K_{ij}K^{ij} + \nabla_{\mu}\zeta^{\mu}, \qquad (3.1.28)$$

where  $\zeta^{\mu} = -2Kn^{\mu} + 2n^{\rho}\nabla_{\rho}n^{\mu}$ , which implies that

$$\zeta^{r} = -2Kn^{r} + 2n^{\rho}\nabla_{\rho}n^{r} = -2K/N + 2n^{\rho}\left(\partial_{\rho}n^{r} + \Gamma^{r}_{\rho\lambda}n^{\lambda}\right)$$

$$=0$$
(3.1.29)

$$= -2K/N + 2n^{r} \left(\partial_{\rho} n^{r} + \Gamma_{rr}^{r} n^{r}\right) + 2n^{i} \left(\partial_{i} n^{r} + \Gamma_{ir}^{r} n^{r} + \Gamma_{ij}^{r} n^{j}\right)$$
(3.1.30)

$$= -2K/N,$$
 (3.1.31)

which means that the Gibbons-Hawking term precisely cancels the total derivative of the Ricci scalar (3.1.28); see also [99]. The total action is then expressible as

$$S = \int \mathrm{d}r \ L,\tag{3.1.32}$$

where

$$L = \int_{\Sigma_r} \mathrm{d}^d x \sqrt{\gamma} N\left(R^{(\gamma)} + K^2 - K_{ij}K^{ij} - \frac{1}{2N^2}\left(\dot{\phi} - N^i\partial_i\phi\right)^2 - \frac{1}{2}\gamma^{ij}\partial_i\phi\partial_j\phi - V(\phi)\right), \quad (3.1.33)$$

where we have used the metric (3.1.19) to write

$$\frac{1}{2}\partial_{\mu}\phi\partial^{\mu}\phi = \frac{1}{2}g^{\mu\nu}\partial_{\mu}\phi\partial_{\nu}\phi = \frac{1}{2}\gamma^{ij}\partial_{i}\phi\partial_{j}\phi + \frac{1}{2N^{2}}\left(\dot{\phi} - N^{i}\partial_{i}\phi\right)^{2}.$$
(3.1.34)

From the Lagrangian (3.1.33), we can read off—using the extrinsic curvature (3.1.17)—the conjugate momenta to the induced metric and the scalar,

$$\pi^{ij} = \frac{1}{\sqrt{\gamma}} \frac{\delta L}{\delta \dot{\gamma}_{ij}} = K^{ij} - K \gamma^{ij}, \qquad (3.1.35)$$

$$\pi_{\phi} = \frac{1}{\sqrt{\gamma}} \frac{\delta L}{\dot{\phi}} = N^{-1} \left( \dot{\phi} - N^{i} \partial_{i} \phi \right), \qquad (3.1.36)$$

so we can apply the Legendre transform to our Lagrangian (3.1.33) to obtain the Hamiltonian on the radial hypersurface  $\Sigma_r$ ,

$$H = \int_{\Sigma_r} \mathrm{d}^d x \,\sqrt{\gamma} \left( \pi^{ij} \dot{\gamma}_{ij} + \pi_{\phi} \dot{\phi} \right) - L = \int_{\Sigma_r} \mathrm{d}^d x \,\sqrt{\gamma} \left( N \mathcal{H} + N_i \mathcal{H}^i \right), \tag{3.1.37}$$

where

$$\mathcal{H} = \pi_{ij}\pi^{ij} - \frac{1}{d-1}\pi^2 + \frac{1}{2}\pi_{\phi}^2 + R^{(\gamma)} - \frac{1}{2}\partial_i\phi\partial^i\phi - V(\phi), \qquad (3.1.38)$$

$$\mathcal{H}^{i} = -2D_{j}\pi^{ij} + \pi_{\phi}\partial^{i}\phi, \qquad (3.1.39)$$

where we have used the extrinsic curvature (3.1.17) and the momenta (3.1.35)–(3.1.36) to get  $\mathcal{H}$ , whereas we have integrated (covariantly) by parts to get  $\mathcal{H}^i$ . From the Hamiltonian (3.1.37), the equations of motion for the Lagrange multiplier fields N and  $N^i$  immediately give the constraints

$$\mathcal{H} = 0, \quad \mathcal{H}^i = 0, \tag{3.1.40}$$

known as the Hamiltonian and momentum constraint, respectively. The constraints (3.1.40) imply that the Hamiltonian (3.1.37) vanishes on the constraint surface; this is a consequence of diffeomorphism invariance and is, as we shall see, closely related to the Hamilton-Jacobi equation. The constraints of (3.1.40) are so-called *first-class constraints*<sup>4</sup> [100], which generate diffeomorphisms along the radial direction and along  $\Sigma_r$ , respectively, via the Poisson bracket. Locally, we can use diffeomorphism invariance to gauge fix the shift and lapse in the following manner [89],

$$N = 1, \qquad N' = 0. \tag{3.1.41}$$

Henceforth, we will adopt the gauge choice of (3.1.41). Now, from the Hamiltonian densities (3.1.38) and (3.1.39), it is clear the Hamiltonian (3.1.37) has no explicit radial dependence, so the radial Hamilton-Jacobi equation assumes the form

$$H = 0,$$
 (3.1.42)

which is the same as the constraints (3.1.40); crucially, however, the canonical momenta are here determined in terms of Hamilton's principal function \$ (compare with the expressions in (3.1.11)),

$$\pi^{ij} = \frac{1}{\sqrt{\gamma}} \frac{\delta \mathcal{S}}{\delta \gamma_{ij}}, \quad \pi_{\phi} = \frac{1}{\sqrt{\gamma}} \frac{\delta \mathcal{S}}{\delta \phi}. \tag{3.1.43}$$

The prescription (3.1.43) then implies that the constraints (3.1.40) become functional differential equations for S, and the Hamiltonian constraint thus reads

$$\left[\gamma_{ik}\gamma_{jl} - \frac{1}{d-1}\gamma_{ij}\gamma_{kl}\right]\frac{1}{\gamma}\frac{\delta\mathcal{S}}{\delta\gamma_{ij}}\frac{\delta\mathcal{S}}{\delta\gamma_{kl}} + \frac{1}{2\gamma}\left(\frac{\delta\mathcal{S}}{\delta\phi}\right)^2 + R^{(\gamma)} - \frac{1}{2}\partial_i\phi\partial^i\phi - V(\phi) = 0, \quad (3.1.44)$$

<sup>4</sup> In the sense that their Poisson brackets with the constraints (3.1.40) vanishes weakly, for details see [100].

which is the starting point of the dBVV-DeWitt<sup>5</sup> approach to holographic renormalization, which we review in appendix **F**. As we saw in section (3.1.1), a solution *S* to the HJ equation provides a solution to Hamilton's equations—in fact, it is be to identified with the on-shell action—and thus it also solves the second order equations of motion (i.e. the Euler-Lagrange equations); in particular, Hamilton's first equation  $\dot{\mathbf{q}} = \frac{\delta H}{\delta \mathbf{p}}$  leads to the first order *flow equations* 

$$\dot{\gamma}_{ij} = 2\left(\gamma_{ik}\gamma_{jl} - \frac{1}{d-1}\gamma_{ij}\gamma_{kl}\right)\frac{1}{\sqrt{\gamma}}\frac{\delta S}{\delta \gamma_{kl}},\tag{3.1.45}$$

$$\dot{\phi} = \frac{1}{\sqrt{\gamma}} \frac{\delta S}{\delta \phi}.$$
(3.1.46)

Equivalently, the equations above may be obtained by inserting the momenta (3.1.43) into their representations (3.1.35) and (3.1.36) with the gauge choice (3.1.41); this is straightforward for  $\dot{\phi}$ , but  $\dot{\gamma}_{ij}$  is related to the extrinsic curvature as detailed in (3.1.17), which implies that  $\frac{1}{2}\dot{\gamma}_{ij} = K_{ij}$ , and thus the relation for  $\pi^{ij}$  (3.1.35) leads to  $\dot{\gamma}_{ij} = 2\pi_{ij} - \frac{2}{1-d}\pi$ , which is the same as (3.1.45).

In asymptotically locally AdS (AlAdS) geometries (see appendix A) in particular<sup>6</sup>, these flow equations allow one to reconstruct the Fefferman–Graham expansions of the fields under scrutiny, which we pursue in section 3.2.3.

The application of Hamilton-Jacobi theory in holographic renormalization hinges on the fact that Hamilton's principal function on  $\Sigma_r$  is equivalent to the on-shell action *S* on  $\Sigma_r$ , as we showed in section 3.1.1. More explicitly, the regularized action, i.e. the on-shell action evaluated on  $\Sigma_r$ , which is given by

$$S_{\rm reg}[\gamma(r,x),\phi(r,x)] = \int^r dr' L_{\rm os},$$
(3.1.47)

satisfies the HJ equation and may thus be identified with S. So, if we can determine the principal function, we can identify the divergences of the on-shell action and thus construct the appropriate counterterm; denoting the divergent part of the principal function  $S_{loc}$ , which is a local functional of the data on  $\Sigma_r$ , the counterterm takes the form

$$S_{\rm ct} = -S_{\rm loc}.$$
 (3.1.48)

It is worth noting that it is possible to add finite local terms to the bulk action<sup>7</sup>, corresponding to choosing a specific renormalization scheme, i.e.

$$S_{\rm ct} = -\left(\mathcal{S}_{\rm loc} + \mathcal{S}_{\rm scheme}\right). \tag{3.1.49}$$

In the present work, we exclusively use minimal subtraction, i.e. we subtract only the divergences of the bare action. Once the counterterm has been obtained, the renormalized action on-shell on  $\Sigma_r$  is given by

$$S_{\rm ren} = S_{\rm reg} + S_{\rm ct} = \int d^d x \,\sqrt{\gamma} \left(\gamma_{ij} \Pi^{ij} + \phi \Pi_{\phi}\right), \qquad (3.1.50)$$

where the renormalized canonical momenta  $\Pi^{ij}$  and  $\Pi_{\phi}$  are closely related to the renormalized onepoint functions of the operators dual to the fields.

#### 3.1.3 Recursive Solution of the HJ Equation

In order to provide a general algorithm for determining the principal function S, we write S as an expansion of some functional operator  $\delta$ , i.e.

$$S = S_{(\alpha_0)} + S_{(\alpha_1)} + S_{(\alpha_2)} + \cdots,$$
 (3.1.51)

<sup>5</sup> The quantity in square brackets in (3.1.44) bears a striking resemblance to the standard Wheeler-DeWitt metric [101]; in fact, it is the deWitt metric with parameter d - 1 (see [102]).

<sup>6</sup> It is also possible for more general backgrounds to generate analogues of the FG expansions using this method, but then one has to solve the differential equations. For AlAdS boundary conditions, the FG theorem provides a highly useful ansatz.

<sup>7</sup> This corresponds to a special choice of integration constants in the complete integral of the HJ equation.

with

$$\delta \mathcal{S}_{(\alpha_k)} = \lambda_k \mathcal{S}_{(\alpha_k)}, \tag{3.1.52}$$

that is, every term in the expansion (3.1.51) is an eigenfunction of  $\delta$  with eigenvalue  $\lambda_k$ , where all the  $\lambda_k$ 's are distinct. It is now convenient to introduce a density  $\mathscr{L}$  such that the principal function reads

$$\mathcal{S} = \int_{\Sigma_r} \mathrm{d}^d x \,\sqrt{\gamma} \mathcal{L}\,,\tag{3.1.53}$$

where we *require that*  $\sqrt{\gamma}$  be an eigenfunction of our operator  $\delta$  with eigenvalue  $\lambda_{\gamma}$ ; this is a constraint on the possible form of  $\delta$ . This implies that we have an expansion of the form

$$\mathscr{L} = \mathscr{L}_{(\alpha_0)} + \mathscr{L}_{(\alpha_1)} + \mathscr{L}_{(\alpha_2)} + \cdots, \qquad (3.1.54)$$

where

$$\mathcal{S}_{(\alpha_k)} = \int_{\Sigma_r} \mathrm{d}^d x \,\sqrt{\gamma} \mathscr{L}_{(\alpha_k)},\tag{3.1.55}$$

implying that the  $\sqrt{\gamma} \mathscr{L}_{(\alpha_k)}$  are only defined up to a total divergence, which is not generically an eigenfunction of  $\delta$ . To remedy this, we note first that for arbitrary variations of the on-shell action, the momentum relations (3.1.43) imply that

$$\pi^{ij}\delta\gamma_{ij} + \pi_{\phi}\delta\phi = \frac{1}{\sqrt{\gamma}}\delta\left(\sqrt{\gamma}\mathscr{L}\right) + \frac{1}{\sqrt{\gamma}}\partial_{i}v^{i}(\delta\gamma,\delta\phi), \qquad (3.1.56)$$

for some vector field  $v^i(\delta\gamma,\delta\phi)$ . Applying this for  $\delta$ , we find that

$$\pi^{ij}_{(\alpha_k)} \delta\gamma_{ij} + \pi_{\phi(\alpha_k)} \delta\phi = \frac{1}{\sqrt{\gamma}} \delta\left(\sqrt{\gamma} \mathscr{L}_{(\alpha_k)}\right) + \frac{1}{\sqrt{\gamma}} \partial_i \tilde{v}^i (\delta\gamma, \delta\phi)$$
(3.1.57)

$$= (\lambda_{\gamma} + \lambda_{k}) \mathscr{L}_{(\alpha_{k})} + \frac{1}{\sqrt{\gamma}} \partial_{i} \tilde{v}^{i}(\delta\gamma, \delta\phi), \qquad (3.1.58)$$

where we have used the Leibniz property of  $\delta$ , and where  $\tilde{v}_{(\alpha_k)}^i$  is in general different from  $v_{(\alpha_k)}^i$ , since  $\delta$  acting on  $\sqrt{\gamma} \mathscr{L}$  may involve a total derivative. However, since we are ultimately interested in the *action*, these total derivatives are mere "artefacts", and we may choose—without loss of generality—the total derivative such that the following extremely useful relation holds:

$$\pi^{ij}_{(\alpha_k)} \delta \gamma_{ij} + \pi_{\phi(\alpha_k)} \delta \phi = (\lambda_\gamma - \lambda_k) \mathscr{L}_{(\alpha_k)}, \tag{3.1.59}$$

which is crucial since it allows us to relate the Hamiltonian constraint (3.1.44) directly to the on-shell action via the  $\mathscr{L}_{(\alpha_k)}$ . We now proceed to discuss two concrete realizations of the operator  $\delta$ .

#### 3.1.4 The Induced Metric Expansion

Used in [51] to renormalize general dilaton-axion gravity, this approach relies on choosing  $\delta$  to be

$$\delta_{\gamma} = \int_{\Sigma_r} \mathrm{d}^d x \; 2\gamma_{ij} \frac{\delta}{\delta \gamma_{ij}},\tag{3.1.60}$$

which satisfies both the Leibniz property and has  $\sqrt{\gamma}$  as an eigenfunction, since

$$\delta_{\gamma}\sqrt{\gamma} = \gamma_{ij}\gamma^{ij}\sqrt{\gamma} = d\sqrt{\gamma}, \qquad (3.1.61)$$

i.e.  $\lambda_{\gamma} = d$ . The expansion in eigenmodes of  $\delta_{\gamma}$  treats the scalar field non-perturbatively, so we do not need to specify the potential explicitly, which makes this method quite general—in particular, this method applies also to asymptotically non-AdS backgrounds, e.g. non–conformal branes [103]. For the gravity-scalar system we consider, the resulting expansion is a derivative expansion<sup>8</sup>, and each factor of the metric is associated with two derivatives, i.e. in terms of the form  $\partial_i \phi \partial^i \phi$ , and therefore

<sup>8</sup> This is no longer the case if we include a Maxwell field  $A_i$ , since then we have terms of the schematic form  $\gamma^{ij}A_iA_j$ .—we will explore this scenario in section 3.2.1.

we choose the label  $\alpha_k = 2k$  to count derivatives. The statement corresponding to (3.1.58) with  $\delta$  given by (3.1.60) thus takes the form

$$2\pi_{i(2k)}^{i} = (d - 2k)\mathcal{L}_{(2k)}.$$
(3.1.62)

Now, the zeroth order contribution contains no derivatives and is therefore solely expressed in terms of  $\phi$ . We define the *superpotential*  $U(\phi) := \mathcal{L}_{(0)}$ , so that

$$\mathcal{S}_{(0)} = \int_{\Sigma_r} \mathrm{d}^d x \,\sqrt{\gamma} U(\phi). \tag{3.1.63}$$

Plugging the above into the functional Hamiltonian constraint of (3.1.44) and collecting terms with no derivatives, we get

$$2(U'(\phi))^2 - \frac{d}{(d-1)}U(\phi)^2 - 4V(\phi) = 0, \qquad (3.1.64)$$

where we have used that  $\frac{\delta \delta_{(0)}}{\delta \gamma_{ij}} = \frac{1}{2} \sqrt{\gamma} \gamma^{ij} U(\phi)$ . It is important to note that we need only solve this equation near the boundary, which simplifies the analysis substantially. Given  $U(\phi)$ , we can insert the  $\delta_{\gamma}$ -eigenmode expansion of the principal function into (3.1.44) to obtain the following first order linear inhomogeneous recursive functional differential equations

$$2U'(\phi)\frac{1}{\sqrt{\gamma}}\frac{\delta}{\delta\phi}\int_{\Sigma_r} \mathrm{d}^d x \,\sqrt{\gamma}\mathscr{L}_{(2n)} - \frac{d-2n}{d-1}U(\phi)\mathscr{L}_{(2n)} = \mathcal{R}_{(2n)}, \quad n > 0, \tag{3.1.65}$$

where

$$\mathcal{R}_{(2)} = \frac{1}{2} \partial_i \phi \partial^i \phi - R^{(\gamma)}, \qquad (3.1.66)$$

$$\mathcal{R}_{(2n)} = -\sum_{m=1}^{n-1} \left( \pi_{(2m)}^{ij} \pi_{(2(n-m))ij} - \frac{1}{d-1} \pi_{(2m)} \pi_{(2(n-m))} + \frac{1}{2} \pi_{\phi(2m)} \pi_{\phi(2(n-m))} \right).$$
(3.1.67)

Solving these equations is, however, tricky, and it is often advantageous to choose  $\delta$  to be the dilatation operator (which, as we shall see, occasionally coincides with  $\delta_{\gamma}$ ), which, in this context, is written as  $\delta_D$ , since in many cases it reduces the functional differential equations to algebraic equations which are much easier to solve. We explore the consequences of this choice in the section below.

## 3.2 The $\delta_D$ expansion & holographic renormalization of the electromagnetic uplift

In this section, we perform a novel renormalization of an EMD model where the dilaton does not couple to the electromagnetic field nor the cosmological constant: we refer to this model as the electromagnetic uplift. We need the renormalized action of this model when considering charged Lifshitz holography via Sherk-Schwarz reduction, which will be the topic of chapter 6. As remarked above, choosing  $\delta$  to be the dilatation operator  $\delta_D$  is often a good choice, and one we will make in this section. The addition of a free Maxwell field leads to an extra constraint encoding gauge invariance of the action in addition to those encountered in (3.1.40). This constraint is not trivially satisfied even after fixing the gauge to an axial gauge  $A_r = 0$  (which supplements the ADM gauge (3.1.41)), since this choice leaves residual gauge transformations, under which we must also require invariance.

Some aspects of holographic renormalization for EMD models has been explored in previous literature: it was considered in a restricted setting in [104], while the general principles were laid down in [93].

Holography including renormalization for a different class of EMD models from the perspective of dimensional reduction was considered [105]; this approach somewhat reminiscent of the structure we will discuss in chapter 6.

#### 3.2.1 Holographic Renormalization of the Electromagnetic Uplift

The model we consider is

$$S = \frac{1}{2\kappa^2} \int_{\mathcal{M}} d^{d+1}x \,\sqrt{-g} \left( R^{(g)} + d(d-1) - \frac{1}{2} \partial_{\mu} \phi \partial^{\mu} \phi - \frac{1}{4} F_{\mu\nu} F^{\mu\nu} \right) + S_{\text{GH}},\tag{3.2.1}$$

where  $S_{\text{GH}}$  is the Gibbons-Hawking boundary term, and where the field strength is defined in the usual manner  $F_{ij} = 2\partial_{(i}A_{j)}$ . Since we will eventually set d = 4, we will assume that d is even. The equations of motion for this action read

$$G_{\mu\nu} = \frac{1}{2} \left( \partial_{\mu}\phi \partial_{\nu}\phi - \frac{1}{2} (\partial\phi)^2 g_{\mu\nu} \right) + \frac{d(d-1)}{2} + \frac{1}{2} \left( F_{\mu\rho}F_{\nu}{}^{\rho} - \frac{1}{4}F^2 g_{\mu\nu} \right),$$
(3.2.2)

$$0 = \frac{1}{\sqrt{-g}} \partial_{\mu} \left( \sqrt{-g} \partial^{\mu} \phi \right), \qquad \frac{1}{\sqrt{-g}} \partial_{\mu} \left( \sqrt{-g} F^{\mu\nu} \right) = 0, \tag{3.2.3}$$

where  $G_{\mu\nu}$  is the Einstein tensor. Performing an ADM-decomposition of the metric,

$$ds^{2} = g_{\mu\nu} dx^{\mu} dx^{\nu} = N^{2} dr^{2} + \gamma_{ij} \left( dx^{i} + N^{i} dr \right) \left( dx^{j} + N^{j} dr \right), \qquad (3.2.4)$$

and using the results of section 3.1.2, the total action may be written as

$$S = \int \mathrm{d}r \ L, \tag{3.2.5}$$

where [28]

$$L = \frac{1}{2\kappa^2} \int_{\Sigma_r} d^d x \sqrt{-\gamma} N\left(K^2 - K^{ij} K_{ij} - \frac{1}{2N^2} \left(\dot{\phi} - N^i \partial_i \phi\right)^2 - \frac{1}{2N^2} \left(F_{ri} - N^k F_{ki}\right) \left(F_r^{\ i} - N^l F_l^{\ i}\right)$$
(3.2.6)

$$+ R^{(\gamma)} - \frac{1}{2} \partial_i \phi \partial^i \phi - \frac{1}{4} F_{ij} F^{ij} + d(d-1) \bigg),$$
(3.2.7)

where we note that  $A_r$  is not dynamical, i.e. it acts as a Lagrange multiplier. From the Lagrangian (3.2.7), we may immediately determine the canonical momenta

$$\pi^{ij} = \frac{1}{\sqrt{-\gamma}} \frac{\delta L}{\delta \dot{\gamma}_{ij}} = \frac{1}{2\kappa^2} \left( K \gamma^{ij} - K^{ij} \right), \qquad (3.2.8)$$

$$\pi^{i} = \frac{1}{\sqrt{-\gamma}} \frac{\delta L}{\delta \dot{A}_{i}} = -\frac{1}{2\kappa^{2}} N^{-1} \left( F_{r}^{i} - N^{k} F_{k}^{i} \right), \qquad (3.2.9)$$

$$\pi_{\phi} = \frac{1}{2\kappa^2} N^{-1} \left( \dot{\phi} - N^i \partial_i \phi \right). \tag{3.2.10}$$

These can be inverted to obtain expressions for the generalized velocities in terms of the momenta; this produces the flow equations that we encountered in (3.1.45)-(3.1.46),

$$\dot{\gamma}_{ij} = -4\kappa^2 N\left(\pi_{ij} - \frac{1}{d-1}\pi\gamma_{ij}\right) + D_i N_j + D_j N_i,$$
(3.2.11)

$$\dot{A}_i = -2\kappa^2 N\pi_i + \partial_i A_r + N^k F_{ki}, \qquad (3.2.12)$$

$$\dot{\phi} = -2\kappa^2 N \pi_{\phi} + N^i \partial_i \phi, \qquad (3.2.13)$$

which allow us to determine the Hamiltonian via the Legendre transform,

,

$$H = \int_{\Sigma_r} \mathrm{d}^d x \,\sqrt{-\gamma} \left(\dot{\gamma}_{ij} \pi^{ij} + \dot{A}_i \pi^i + \dot{\phi} \pi_\phi\right) - L = \int_{\Sigma_r} \mathrm{d}^d x \,\sqrt{-\gamma} \left(N\mathcal{H} + N_i \mathcal{H}^i + A_r \mathcal{F}\right),\tag{3.2.14}$$

with the densities in front of the Lagrange multipliers  $\{N, N_i, A_r\}$  are given by

$$\mathcal{H} = -2\kappa^2 \left( \pi^{ij} \pi_{ij} - \frac{1}{d-1} \pi^2 + \frac{1}{2} \pi_{\phi}^2 + \frac{1}{2} \pi^i \pi_i \right) + \frac{1}{2\kappa^2} \left( -R^{(\gamma)} + \frac{1}{2} \partial_i \phi \partial^i \phi + \frac{1}{4} F_{ij} F^{ij} - d(d-1) \right),$$
(3.2.15)

$$\mathcal{H}^{i} = -2D_{j}\pi^{ji} + F^{ij}\pi_{j} + \pi_{\phi}\partial^{i}\phi, \qquad (3.2.16)$$

$$\mathcal{F} = -D_i \pi^i. \tag{3.2.17}$$

The equations of motion for the Lagrange multipliers  $\{N, N_i, A_r\}$  impose the following first-class constraints,

$$\mathcal{H} = 0, \quad \mathcal{H}^i = 0, \quad \mathcal{F} = 0.$$
 (3.2.18)

#### 24 HOLOGRAPHIC RENORMALIZATION

These are equivalent to the Hamilton-Jacobi equation H = 0, where the canonical momenta in H are expressed in the usual way as functional derivatives of Hamilton's principal function  $S[\gamma, A, \phi]$ , leading to a functional differential equation. However, the momentum constraint  $\mathcal{H}^i = 0$  simply requires S to be diffeomorphism invariant on constant-r leaves  $\Sigma_r$ , whereas  $\mathcal{F} = 0$  amounts to the statement that S is gauge invariant—something we will explicitly see later. In order to proceed, we adopt the gauge choice

$$N = 1, \quad N^{i} = 0, \quad A_{r} = 0, \tag{3.2.19}$$

which simplifies the analysis substantially. On AlAdS backgrounds, the equations of motion (3.2.2)-(3.2.3) imply that the fields involved in our model have the following asymptotic behaviour<sup>9</sup>

$$\gamma_{ij} \simeq e^{2r} g_{(0)ij}(x), \quad \phi \simeq \phi_{(0)}(x), \quad A_i \simeq A_{(0)i}(x), \quad (3.2.20)$$

where we use domain-wall coordinates (cf. (A.1.9)). The radial derivative, given by

$$\delta_r = \int_{\Sigma_r} \mathrm{d}^d x \, \left( \dot{\gamma}_{ij} \frac{\delta}{\delta \gamma_{ij}} + \dot{\phi} \frac{\delta}{\delta \phi} + \dot{A}_i \frac{\delta}{\delta A_i} \right), \tag{3.2.21}$$

asymptotes—since by (3.2.20)  $\dot{\gamma}_{ij} \simeq 2\gamma_{ij}$ ,  $\dot{\phi} \simeq 0$  and  $\dot{A}_i \simeq 0$ —the dilatation operator,

$$\delta_r \simeq \int_{\Sigma_r} \mathrm{d}^d x \, 2\gamma_{ij} \frac{\delta}{\delta \gamma_{ij}} = \delta_D.$$
 (3.2.22)

Note that for this model, the dilatation operator in fact coincides with the induced metric operator  $\delta_{\gamma}$  of (3.1.60) to lowest order. The next step is to express the principal function as

$$S = \int_{\Sigma_r} \mathrm{d}^d x \, \sqrt{-\gamma} \mathcal{L}, \qquad (3.2.23)$$

where  $\mathscr{L}$  is expanded in eigenmodes of  $\delta_D$  in the following manner, where, due to the fact that *d* is even, we have anticipated logarithmic terms in accordance with the general FG scheme applicable for AlAdS boundary conditions,

$$\mathscr{L} = \dots + \mathscr{L}_{(-1)} + \mathscr{L}_{(0)} + \mathscr{L}_{(1)} + \mathscr{L}_{(2)} + \dots + \widetilde{\mathscr{L}}_{(d)} \log e^{-2r} + \mathscr{L}_{(d)} + \dots, \qquad (3.2.24)$$

where

$$\delta_D \mathscr{L}_{(w)} = -w \mathscr{L}_{(w)}, \quad w < d, \tag{3.2.25}$$

$$\delta_D \mathscr{L}_{(d)} = -d \mathscr{L}_{(d)}, \tag{3.2.26}$$

$$\delta_D \mathscr{L}_{(d)} = -d \mathscr{L}_{(d)} - 2 \mathscr{L}_{(d)}. \tag{3.2.27}$$

The transformation (3.2.27) is not entirely obvious, so let us derive it. Due to diffeomorphism invariance, the on-shell action does not—as we have seen—depend explicitly on the cut-off r, so the only dependence enters via the fields. This implies that the radial derivative of the bare on-shell action is given by  $\delta_r$ , as given in (3.2.21), which to leading order is equivalent to the dilatation operator as demonstrated in (3.2.22). Thus, we find that

$$\delta_r \left( \widetilde{\mathscr{L}}_{(d)} \log e^{-2r} + \mathscr{L}_{(d)} \right) = \delta_r \left( \widetilde{\mathscr{L}}_{(d)} \right) \log e^{-2r} - 2 \widetilde{\mathscr{L}}_{(d)} + \delta_r \mathscr{L}_{(d)}, \tag{3.2.28}$$

which, when set equal to  $\log e^{-2r} \delta_D \widehat{\mathcal{Z}}_{(d)} + \delta_D \mathcal{L}_{(d)}$ , produces the transformation property (3.2.27). The expansion of  $\mathcal{L}$  in (3.2.24) leads to the following expansions of the momenta,

$$\pi^{ij} = \frac{1}{\sqrt{-\gamma}} \frac{\delta}{\delta \gamma_{ij}} \int_{\Sigma_r} \mathrm{d}^d x \, \sqrt{-\gamma} \mathscr{L} = \dots + \pi^{ij}_{(1)} + \pi^{ij}_{(2)} + \pi^{ij}_{(3)} + \pi^{ij}_{(4)} + \dots + \widetilde{\pi}^{ij}_{d+2} \log e^{-2r} + \pi^{ij}_{(d+2)} + \dots$$
(3.2.29)

$$\pi_{\phi} = \frac{1}{\sqrt{-\gamma}} \frac{\delta}{\delta \phi} \int_{\Sigma_r} \mathrm{d}^d x \, \sqrt{-\gamma} \mathcal{L} = \dots + \pi_{\phi(-1)} + \pi_{\phi(0)} + \pi_{\phi(1)} + \dots \widetilde{\pi}_{\phi(d)} \log e^{-2r} + \pi_{\phi(d)} + \dots ,$$
(3.2.30)

$$\pi^{i} = \frac{1}{\sqrt{-\gamma}} \frac{\delta}{\delta A_{i}} \int_{\Sigma_{r}} \mathrm{d}^{d} x \, \sqrt{-\gamma} \mathscr{L} = \dots + \pi^{i}_{(-1)} + \pi^{i}_{(0)} + \dots \widetilde{\pi}^{i}_{(d)} \log e^{-2r} + \pi^{i}_{(d)} + \dots$$
(3.2.31)

<sup>9</sup> This statement is only valid for  $d \ge 2$ , which has interesting consequences that we will explore in sections 3.2.2 and 3.2.3 and in particular in section 3.3. We will also briefly comment on the generalization for *p*-forms.

Applying the dilatation operator (3.2.22) to the principal function (3.2.23), we get the following useful relation for w < d:

$$\left[2\gamma_{ij}\pi^{ij}\right]_{(w)} = (d-w)\mathscr{L}_{(w)}, \quad w < d$$
(3.2.32)

$$\Rightarrow 2\gamma_{ij}\pi^{ij}_{(w+2)} = 2\pi_{(w)} = (d-w)\mathscr{L}_{(w)}, \quad w < d,$$
(3.2.33)

where  $[XY]_{(w)}$  means the piece in the  $\delta_D$  expansion of XY with dilatation weight w, i.e.  $\delta_D[XY]_{(w)} = -w[XY]_{(w)}$ , and where  $\pi_{(w)} = \gamma_{ij}\pi^{ij}_{(w+2)}$  is the trace of the momentum, which, due to the asymptotic behaviour of the fields (3.2.20), has scaling weight w. For w = d, we find, by use of the scaling relations for the eigenmodes (3.2.26) and (3.2.27), the following relation,

$$2\pi_{(d)} = \frac{1}{\sqrt{-\gamma}} \delta_D \int_{\Sigma_r} \mathrm{d}^d x \, \sqrt{-\gamma} \left( \widetilde{\mathscr{L}}_{(d)} \log e^{-2r} + \mathscr{L}_{(d)} \right) \tag{3.2.34}$$

$$=-2\widetilde{\mathscr{L}}_{(d)},\tag{3.2.35}$$

which is closely related to the conformal anomaly. Note in particular, we don't need to care about w > d, since these terms will vanish as  $r \to \infty$ . Now let's rewrite the Hamiltonian

$$\mathcal{H} = -2\kappa^2 \left( \pi^{ij}\pi_{ij} - \frac{1}{d-1}\pi^2 + \frac{1}{2}\pi_{\phi}^2 + \frac{1}{2}\pi^i\pi_i \right) + \frac{1}{2\kappa^2} \left( -R^{(\gamma)} + \frac{1}{2}\partial_i\phi\partial^i\phi + \frac{1}{4}F_{ij}F^{ij} - d(d-1) \right)$$
(3.2.36)

$$= K_{ij}\pi^{ij} - 2\kappa^2 \left(\frac{1}{2}\pi_{\phi}^2 + \frac{1}{2}\pi^i\pi_i\right) + \frac{1}{2\kappa^2} \left(-R^{(\gamma)} + \frac{1}{2}\partial_i\phi\partial^i\phi + \frac{1}{4}F_{ij}F^{ij} - d(d-1)\right),$$
(3.2.37)

where we have used our expression for  $\pi^{ij}$  (3.2.8) to rewrite the gravitational kinetic term. We can also expand this in dilatation weights,

$$\mathcal{H} = \sum_{w} \mathcal{H}_{(w)}, \quad \delta_D \mathcal{H}_{(w)} = -w \mathcal{H}_{(w)}, \tag{3.2.38}$$

and we then impose the Hamiltonian constraint  $\mathcal{H} = 0$ , which, due to unitarity of  $\delta_D$  implies that each  $\mathcal{H}_{(w)}$  vanishes individually,

$$\mathcal{H}_{(w)} = 0, \quad \forall w. \tag{3.2.39}$$

The next step is to introduce vielbeine<sup>10</sup>,  $\gamma_{ij} = e_i^a e_j^b \eta_{ab}$ . This is advantageous since quantities with flat indices retain their scaling weight when indices are raised and lowered—in contrast to quantities carrying curved indices. This does, however, change the scaling weights of quantities with indices (but does so once and for all). Clearly,  $e_i^a$  has scaling weight -1, while  $e_a^i$  has scaling weight 1, so  $A_a = A_i e_a^i$  also has scaling weight 1. The expansions of the momenta (3.2.29) and (3.2.31) in flat indices become (the expansion for  $\pi_{\phi}$  carries no indices and is unchanged)

$$\pi^{ab} = e_i^a e_j^b \pi^{ij} = \dots + \pi^{ab}_{(-1)} + \pi^{ab}_{(0)} + \pi^{ab}_{(1)} + \pi^{ab}_{(2)} + \dots + \widetilde{\pi}^{ab}_{(d)} \log e^{-2r} + \pi^{ab}_{(d)} + \dots , \quad (3.2.40)$$
  
$$\pi^a = e_i^a \pi^i = \dots + \pi^a_{(-2)} + \pi^a_{(-1)} + \dots \widetilde{\pi}^a_{(d-1)} \log e^{-2r} + \pi^a_{(d-1)} + \dots \quad (3.2.41)$$

The asymptotic behaviours of the fields (3.2.20) determine—via the momentum relations (3.2.8)–(3.2.10)—the scaling weight at which the asymptotic expansions (3.2.29)–(3.2.31) begin, which in turn, through the useful relation (3.2.33), determines the scaling weight at which the expansion (3.2.24) of  $\mathscr{L}$  begins. Similarly, they determine where the expansion (3.2.38) of  $\mathcal{H}$  begins. We find that the asymptotic identification  $\partial_r \simeq \delta_D$  implies

$$\pi^{ij} = \frac{1}{2\kappa^2} \left( K\gamma^{ij} - K^{ij} \right) = \frac{1}{2\kappa^2} \left( \frac{1}{2} \dot{\gamma}_{lk} \gamma^{lk} \gamma^{ij} - \frac{1}{2} \dot{\gamma}_{lk} \gamma^{li} \gamma^{kj} \right)$$
(3.2.42)

$$\stackrel{\partial_r \simeq \delta_D}{\simeq} \frac{d-1}{2\kappa^2} \gamma^{ij},\tag{3.2.43}$$

so the expansion of  $\pi^{ij}$  starts at scaling weight two, which in turn means that  $\pi^{ab}$  starts at scaling weight zero. Similarly, we find for  $\pi_{(\phi)}$  that the lowest term has the same scaling weight as  $\phi$  itself

<sup>10</sup> This is not necessary and is mainly an "aesthetic choice" in the sense that we find it to make the analysis conceptually cleaner; in the approaches taken in [28, 50, 51, 106] this is not done.

and is proportional to that scaling weight, which means that  $\pi_{\phi(0)} = 0$ , and the eigenmode expansion consequently begins at scaling weight one. Similarly, for the gauge field momentum, we find that

$$\pi^{i} = -\frac{1}{2\kappa^{2}} F_{rj} \gamma^{ji} = -\frac{1}{2\kappa^{2}} \dot{A}_{j} \gamma^{ji} \stackrel{\partial_{r} \simeq \delta_{D}}{\simeq} 0, \qquad (3.2.44)$$

i.e. the expansion for  $\pi^i$  starts at scaling weight one, implying that the expansion for  $\pi^a$  starts at scaling weight zero. Combining our results, we find that the expansions (3.2.30), (3.2.40) and (3.2.41) take the form

$$\pi_{\phi} = \pi_{\phi(1)} + \cdots \widetilde{\pi}_{\phi(d)} \log e^{-2r} + \pi_{\phi(d)} + \cdots, \qquad (3.2.45)$$

$$\pi^{ab} = \pi^{ab}_{(0)} + \pi^{ab}_{(1)} + \pi^{ab}_{(2)} + \dots + \widetilde{\pi}^{ab}_{(d)} \log e^{-2r} + \pi^{ab}_{(d)} + \dots , \qquad (3.2.46)$$

$$\pi^{a} = \pi^{a}_{(0)} + \cdots \widetilde{\pi}^{a}_{(d-1)} \log e^{-2r} + \pi^{a}_{(d-1)} + \cdots, \qquad (3.2.47)$$

where, explicitly (compare (3.2.43))

$$\pi^{ab}_{(0)} = \frac{d-1}{2\kappa^2} \eta^{ab},\tag{3.2.48}$$

whereas  $\pi_{-1}^{a}$  and  $\pi_{\phi(0)}$  were determined to be zero by the same analysis. Since the trace of  $\pi_{(w)}^{ab}$  is related to the principal function density  $\mathscr{L}_{(w)}$  via (3.2.33), this in turn means that the expansion for  $\mathscr{L}$  (3.2.24) starts at scaling weight zero: using (3.2.48) the expression for  $\mathscr{L}_{(0)}$  can be read off immediately,

$$d\mathscr{L}_{(0)} = 2\eta_{ab}\pi^{ab}_{(0)} = \frac{d(d-1)}{\kappa^2} \Rightarrow \mathscr{L}_{(0)} = \frac{d-1}{\kappa^2}.$$
(3.2.49)

In general, we find the  $\mathscr{L}_{(w)}$ 's by relating the expressions for these in terms of the momenta traces  $\pi_{(w)}$  (3.2.33) to the constraints  $\mathcal{H}_{(w)} = 0$ . This is achieved by noting that the Hamiltonian constraint (3.2.39) can be recast in the form

$$\mathcal{H}_{(w)} = \left[ K_{ab} \pi^{ab} - 2\kappa^2 \left( \frac{1}{2} \pi_{\phi}^2 + \frac{1}{2} \pi^a \pi_a \right) + \frac{1}{2\kappa^2} \left( -R^{(\gamma)} + \frac{1}{2} \partial_i \phi \partial^i \phi + \frac{1}{4} F_{ab} F^{ab} - d(d-1) \right) \right]_{(w)} = 0,$$
(3.2.50)

for all w, which, using the flat index momenta (3.2.45)–(3.2.47), means that<sup>11</sup>

$$\mathcal{H}_{(0)} = K_{(0)ab} \pi^{ab}_{(0)} - \kappa^2 \pi^a_{(0)} \pi_{(0)a} - \frac{d(d-1)}{2\kappa^2},$$
(3.2.51)

$$\mathcal{H}_{(1)} = 2K_{(0)ab}\pi^{ab}_{(1)} - 2\kappa^2 \pi^a_{(1)}\pi_{(0)a}$$
(3.2.52)

$$\mathcal{H}_{(2)} = 2K_{(0)ab}\pi^{ab}_{(2)} + K_{(1)ab}\pi^{ab}_{(1)} - \kappa^2 \pi^2_{\phi(1)} - \kappa^2 \pi^a_{(1)}\pi_{(1)a} - 2\kappa^2 \pi^a_{(2)}\pi_{(0)a} + \frac{1}{2\kappa^2} \left(\frac{1}{2}\partial_i \phi \partial^i \phi - R^{(\gamma)}\right),$$
(3.2.53)

$$\mathcal{H}_{(3)} = \sum_{n=0}^{3} K_{(n)ab} \pi^{ab}_{(3-n)} - 2\kappa^2 \pi_{\phi(1)} \pi_{\phi(2)} - 2\kappa^2 \pi^a_{(2)} \pi_{(1)a}, \qquad (3.2.54)$$

$$\mathcal{H}_{(4)} = \sum_{n=0}^{3} K_{(n)ab} \pi^{ab}_{(4-n)} - 2\kappa^2 \left( \pi_{\phi(1)} \pi_{\phi(3)} + \frac{1}{2} \pi^2_{\phi(2)} + \frac{1}{2} \pi^a_{(2)} \pi_{(2)a} + \pi^a_{(1)} \pi_{(3)a} + \pi^a_{(0)} \pi_{(4)a} \right) + \frac{1}{8\kappa^2} F_{ab} F^{ab}_{(3.2.55)}$$

$$(3.2.55)$$

$$\mathcal{H}_{(w)} = \sum_{n=0}^{w} K_{(n)ab} \pi^{ab}_{(w-n)} - 2\kappa^2 \left[ \frac{1}{2} \pi^2_{\phi} + \frac{1}{2} \pi^a \pi_a \right]_{(w)}, \quad 4 < w \le d.$$
(3.2.56)

Note that  $\pi^a_{(w)}$  is the flat version of  $\pi^i_{(w+1)}$ , implying that  $\pi^a_{(0)}$ , for example, comes from  $\mathcal{S}_{(1)}$ . We also remark that due to the fact that both  $\phi$  and  $A_i$  have scaling weight zero (cf. the asymptotic behaviour (3.2.20)), the scaling weight of various quantities such as  $\mathcal{H}_{(w)}$  will vanish for odd values of w, i.e.  $\mathcal{H}_{(2n+1)} = 0$  for  $n \in \mathbb{Z}$ , something we will also find below.

<sup>11</sup> Note that  $R^{(\gamma)}$  has scaling weight two, since the variation with respect to the metric  $\gamma_{ij}$  reads  $\delta R^{(\gamma)} = -R^{(\gamma)ij}\delta\gamma_{ij}$ , implying that  $\delta_D R^{(\gamma)} = -2R^{(\gamma)}$ , i.e.  $R^{(\gamma)}$  has scaling weight two.
In order to simplify the constraints (3.2.51)–(3.2.56), we have made use of the fact that  $K_{(n)ab}\pi^{ab}_{(m)} = K_{(m)ab}\pi^{ab}_{(n)}$ , which follows from the expression for  $\pi^{ij}$  in (3.2.8), since

$$K_{(n)ab}\pi^{ab}_{(m)} = \frac{1}{2\kappa^2} k_{(n)ab} \left( K_{(m)}\eta^{ab} - K^{ab}_{(m)} \right)$$
(3.2.57)

$$= \frac{1}{2\kappa^2} \left( K_{(n)} K_{(m)} - K_{(n)ab} K_{(m)}^{ab} \right), \qquad (3.2.58)$$

which is clearly symmetric under the interchange  $n \leftrightarrow m$ . We then observe that the leading behaviour  $K_{ab}$  is determined by the asymptotic identification  $\partial_r \simeq \delta_D$ , which gives us

$$K_{ij} \simeq \gamma_{ij} \Rightarrow K_{(0)ab} = \eta_{ab}. \tag{3.2.59}$$

This allows us to express the relation (3.2.33) in the following manner

$$2K_{(0)ab}\pi^{ab}_{(w)} = (d-w)\mathscr{L}_{(w)}, \tag{3.2.60}$$

$$2K_{(0)ab}\pi^{ab}_{(d)} = -2\hat{\mathcal{I}}_{(d)}.$$
(3.2.61)

We note that the combination  $2K_{(0)ab}\pi_{(w)}^{ab}$  is ubiquitous in the constraints (3.2.51)–(3.2.56). The way we obtain  $\mathscr{L}_{(w)}$  is then by imposing  $\mathcal{H}_{(w)} = 0$  and recognizing the expression for  $\mathscr{L}_{(w)}$  as given in (3.2.60); i.e. the constraint  $\mathcal{H}_{(0)}$  translates into the following expression by use of (3.2.60):

$$0 = \mathcal{H}_{(0)} = \frac{d}{2}\mathcal{L}_{(0)} - \kappa^2 \pi^a_{(0)} \pi_{(0)a} - \frac{d(d-1)}{2\kappa^2} \Rightarrow \mathcal{L}_{(0)} = \frac{d-1}{\kappa^2} + \frac{2}{d}\kappa^2 \pi^a_{(0)} \pi_{(0)a}.$$
 (3.2.62)

However, since  $\mathscr{L}_{(0)}$  is given by (3.2.49), it follows that

$$\pi^a_{(0)} = 0, \tag{3.2.63}$$

i.e.  $\mathscr{L}_{(1)}$  does not depend on  $A_i$ . The level one constraint  $\mathcal{H}_{(1)} = 0$  simplifies by use of (3.2.63) to  $2\pi_{(1)} = 0$ , that is, by the relation (3.2.33),  $(d-1)\mathscr{L}_{(1)} = 0$ , so that

$$\mathscr{L}_{(1)} = 0 \therefore \pi_{\phi(1)} = 0 = \pi_{(1)}^{ab}. \tag{3.2.64}$$

Combining the results (3.2.63) and (3.2.64) implies that the level two constraint  $\mathcal{H}_{(2)} = 0$  now reads

$$0 = \mathcal{H}_{(2)} = 2\pi_{(2)} - \kappa^2 \pi^a_{(1)} \pi_{(1)a} + \frac{1}{2\kappa^2} \left( \frac{1}{2} \partial_i \phi \partial^i \phi - R^{(\gamma)} \right)$$
(3.2.65)

$$= (d-2)\mathscr{L}_{(2)} - \kappa^2 \pi^a_{(1)} \pi_{(1)a} + \frac{1}{2\kappa^2} \left(\frac{1}{2} \partial_i \phi \partial^i \phi - R^{(\gamma)}\right).$$
(3.2.66)

Now, since  $\pi_{(1)}^a = e_i^a \pi_{(2)}^i = \frac{1}{\sqrt{-\gamma}} e_i^a \frac{\delta \delta_{(2)}}{\delta A_i}$ , the above turns, in principle, into a functional differential equation for  $\mathscr{L}_{(2)}$ . However, we can use the U(1) constraint  $\mathcal{F} = 0$  (cf. (3.2.18)) to determine  $\pi_{(2)}^i$ . This constraint can be recast in the form

$$D_i \pi^i_{(w)} = D_i \left[ \frac{\delta \mathcal{S}_{(w)}}{\delta A_i} \right] = 0, \qquad \forall w.$$
(3.2.67)

Now, the most general expression involving *A* that can appear in the action at scaling weight two has the form

$$\mathcal{S}_{(2)} \supset \sum_{n_1} \phi^{n_1} c_{n_1}^{(1)} \partial^i A_i + \sum_{n_2} \phi^{n_1} c_{n_2}^{(2)} A^i A_i + \sum_{n_3} \phi^{n_3} c_{n_3}^{(3)} A^i \partial_i \phi + \sum_{n_4} \phi^{n_4} c_{n_4}^{(4)} D_i A^i,$$
(3.2.68)

but, crucially, none of these terms—by themselves or in any combination—are in compliance with the requirement (3.2.67), i.e. all the  $c^{(m)}$ 's above vanish. Equivalently, we note that the gauge choice<sup>12</sup>  $A_r = 0$  does not completely fix the gauge, as we now show: under a general gauge transformation, the gauge field transforms as

$$A_{\mu} \to A'_{\mu} = A_{\mu} + \partial_{\mu}\lambda, \qquad (3.2.69)$$

<sup>12</sup> This is an example of an axial gauge, i.e. a gauge of the form  $n^{\mu}A_{\mu} = 0$  for some normal vector  $n^{\mu}$ ; for details see e.g. [69].

for some function  $\lambda$ . The condition  $A_r = 0$  is easily realised by choosing  $\lambda$  to be of the form

$$\lambda = -\int \mathrm{d}r \; A_r + f(x, t, y), \tag{3.2.70}$$

where *f* represents *the residual gauge symmetry*. Requiring our on-shell action to be invariant under residual gauge transformations, we recover the same result as from the  $\mathcal{F} = 0$  constraint that  $\mathcal{S}_{(2)}$  cannot involve  $A_i$ , that is

$$\pi^{i}_{(2)} = 0,$$
 or in flat indices  $\pi^{a}_{(1)} = 0,$  (3.2.71)

leading us to conclude that

$$\mathscr{L}_{(2)} = \frac{1}{2(d-2)\kappa^2} \left( R^{(\gamma)} - \frac{1}{2} \partial_i \phi \partial^i \phi \right).$$
(3.2.72)

The level three constraint vanishes—using  $\pi_{(1)}^{ab} = 0 = \pi_{(1)}^{a}$ , cf. (3.2.64), (3.2.71), as well as the useful relation for the  $\mathscr{L}_{(w)}$ 's (3.2.33)—identically

$$\mathcal{L}_{(3)} = 0.$$
 (3.2.73)

We now specialize to d = 4. We have already computed the first two  $\mathcal{L}$ 's; in d = 4, they take the form

$$\mathscr{L}_{(0)} = \frac{3}{\kappa^2},\tag{3.2.74}$$

$$\mathscr{L}_{(2)} = \frac{1}{4\kappa^2} \left( R^{(\gamma)} - \frac{1}{2} \partial_i \phi \partial^i \phi \right).$$
(3.2.75)

At level four, the constraint reads—after taking into account all our findings:

$$0 = \mathcal{H}_{(4)} = 2K_{(0)ab}\pi^{ab}_{(4)} + K_{(2)ab}\pi^{ab}_{(2)} - \kappa^2\pi^2_{\phi(2)} + \frac{1}{8\kappa^2}F_{ab}F^{ab}$$
(3.2.76)

$$= -2\widetilde{\mathscr{L}}_{(4)} + K_{(2)ab}\pi^{ab}_{(2)} - \kappa^2\pi^2_{\phi(2)} + \frac{1}{8\kappa^2}F_{ab}F^{ab}, \qquad (3.2.77)$$

implying that

$$\widetilde{\mathscr{L}}_{(4)} = \frac{1}{2} K_{(2)ab} \pi^{ab}_{(2)} - \frac{\kappa^2}{2} \pi^2_{\phi(2)} + \frac{1}{8\kappa^2} F_{ab} F^{ab}.$$
(3.2.78)

This involves only quantities that can be determined from  $\mathscr{L}_{(2)}$  (cf. (3.2.72)). First note that the level two constraint  $\mathcal{H}_{(2)} = 0$  implies that for d = 4, we get  $\pi^a_{(2)a} = \mathscr{L}_{(2)}$ , and so

$$\pi^{a}_{(2)a} = \pi_{(2)} = \frac{1}{4\kappa^{2}} \left( R^{(\gamma)} - \frac{1}{2} \partial_{i} \phi \partial^{i} \phi \right).$$
(3.2.79)

Then, observe that

$$\pi^{ab} = \frac{1}{2\kappa^2} \left( K\eta^{ab} - K^{ab} \right) \therefore K_{(2)} = \frac{2\kappa^2}{3} \pi_{(2)}, \tag{3.2.80}$$

implying that  $K_{(2)}$  becomes

$$K_{(2)} = \frac{R^{(\gamma)}}{6} - \frac{1}{12} \partial_i \phi \partial^i \phi.$$
(3.2.81)

Now we can determine  $\pi_{(2)ab}$  from our knowledge of  $\mathscr{L}_{(2)}$ ; since  $\mathscr{S}_{(2)} = \int_{\Sigma_r} d^4x \sqrt{-\gamma} \mathscr{L}_{(2)}$ , we see that

$$\pi_{(2)ab} = e_a^i e_b^j \frac{1}{\sqrt{-\gamma}} \frac{\delta \mathcal{S}_{(2)}}{\delta \gamma^{ij}}$$
(3.2.82)

$$= e_a^i e_b^j \frac{1}{\sqrt{-\gamma}} \frac{\delta}{\delta \gamma^{ij}} \frac{1}{2\kappa^2} \int_{\Sigma_r} \mathrm{d}^4 x \, \sqrt{-\gamma} \left( \frac{R^{(\gamma)}}{2} - \frac{1}{4} \partial_i \phi \partial^i \phi \right) \tag{3.2.83}$$

$$=\frac{1}{2\kappa^2}\left(\frac{1}{8}\eta_{ab}\partial_i\phi\partial^i\phi - \frac{1}{4}R^{(\gamma)}\eta_{ab} - \frac{1}{4}\partial_a\phi\partial_b\phi + \frac{1}{2}R^{(\gamma)}_{ab}\right).$$
(3.2.84)

Our expression for the momentum above allows us to find the scaling weight two extrinsic curvature,

$$\pi_{(2)ab} = \frac{1}{2\kappa^2} \left( K_{(2)}\eta_{ab} - K_{(2)ab} \right), \qquad (3.2.85)$$

29

and so, using the trace of  $K_{(2)}$  (3.2.81) and our expression for  $\pi_{(2)ab}$  (3.2.84), we get

$$K_{(2)ab} = \frac{1}{24} \eta_{ab} \partial_i \phi \partial^i \phi - \frac{1}{12} \eta_{ab} R^{(\gamma)} - \frac{1}{4} \partial_a \phi \partial_b \phi + \frac{1}{2} R^{(\gamma)}_{ab}.$$
 (3.2.86)

We are now in a position to work out the combination  $K_{(2)ab}\pi^{ab}_{(2)}$  appearing in the anomaly (3.2.78):

$$K_{(2)ab}\pi^{ab}_{(2)} = -\frac{1}{2\kappa^2} \left( \frac{1}{24} \left( \partial_i \phi \partial^i \phi \right)^2 + \frac{1}{12} R^{(\gamma)} \partial_i \phi \partial^i \phi - \frac{1}{12} R^{(\gamma)^2} - \frac{1}{4} R^{(\gamma)}_{ij} \partial^i \phi \partial^j \phi + \frac{1}{4} R^{(\gamma)}_{ij} R^{(\gamma)ij} \right),$$
(3.2.87)

The final piece is a Laplacian term, coming from

$$\pi_{\psi(2)} = \frac{1}{\sqrt{-\gamma}} \frac{\delta S_{(2)}}{\delta \phi} = -\frac{1}{4\kappa^2} \Box^{(\gamma)} \phi, \qquad (3.2.88)$$

where we have used that  $\Box^{(\gamma)}\phi = \frac{1}{\sqrt{-\gamma}}\partial_i (\sqrt{-\gamma}\partial^i \phi)$ . Combining everything, the anomaly (3.2.78) takes the form

$$\widetilde{\mathscr{L}}_{(4)} = \frac{1}{8\kappa^2} F_{ij} F^{ij}$$

$$- \frac{1}{4\kappa^2} \left( \frac{1}{24} \left( \partial_i \phi \partial^i \phi \right)^2 + \frac{1}{12} R^{(\gamma)} \partial_i \phi \partial^i \phi - \frac{1}{12} R^{(\gamma)^2} - \frac{1}{4} R^{(\gamma)}_{ij} \partial^i \phi \partial^j \phi + \frac{1}{4} R^{(\gamma)}_{ij} R^{(\gamma)ij} \right)$$
(3.2.90)
$$(3.2.90)$$

$$-\frac{1}{32\kappa^2} \left(\Box^{(\gamma)}\phi\right)^2. \tag{3.2.91}$$

Note that we have  $\log e^{-2r}$ , so when rewriting to Poincaré coordinates (for details, see appendix A), there will be an additional factor of two. Combining all our findings, the counterterm becomes (including a minus, since we need to subtract the divergences)

$$S_{\rm ct} = \frac{1}{\kappa^2} \int_{\Sigma_r} \mathrm{d}^4 x \,\sqrt{-\gamma} \left( -\frac{1}{4} \left( R^{(\gamma)} + 12 - \frac{1}{2} \partial_i \phi \partial^i \phi \right) - \frac{1}{4} \mathcal{A} \log e^{-2r} \right), \tag{3.2.92}$$

where the anomaly can be rewritten as

$$\mathcal{A} = -\frac{1}{4} \left( Q_{ij} Q^{ij} - \frac{1}{3} Q^2 - 2F_{ij} F^{ij} + \frac{1}{2} \left( \Box^{(\gamma)} \phi \right)^2 \right), \qquad (3.2.93)$$

where

$$Q_{ij} = R_{ij}^{(\gamma)} - \frac{1}{2} \partial_i \phi \partial_j \phi.$$
(3.2.94)

In order to obtain the actually renormalized action on the boundary, we take the limit  $r \to \infty$ , i.e.

$$\hat{S}_{\text{ren}} := \lim_{r \to \infty} S_{\text{ren}} = \lim_{r \to \infty} \int_{\Sigma_r} d^d x \, \sqrt{-\gamma} \mathscr{L}_{(d)} \tag{3.2.95}$$

$$= \frac{1}{2\kappa^2} \int_{\mathcal{M}} d^5 x \,\sqrt{-g} \left( R^{(g)} + 12 - \frac{1}{2} \partial_{\mu} \phi \partial^{\mu} \phi - \frac{1}{4} F_{\mu\nu} F^{\mu\nu} \right) + \hat{S}_{\rm GH} + \hat{S}_{\rm ct}, \tag{3.2.96}$$

where we have used that  $\mathscr{L}_{(d)}$  is the first term that does not produce a divergence, while all higher order terms will simply vanish, and where quantities with hats have had the limit  $r \to \infty$  applied to them.

Removing the field strength from the counterterm (3.2.92), our result agrees with previous results for Einstein-Dilaton theory, which was considered in [51, 57, 107]. While at a first glance it may seem surprising that for d = 4, the only modification to the counterterm action due to the addition of a Maxwell field occurs in the anomaly, this is in hindsight obvious: there is simply no gauge invariant combination with scaling weight 2. If we take d = 6, the situation changes, and, roughly speaking, what was  $\widetilde{\mathcal{L}}_{(4)}$  in d = 4 becomes  $\mathscr{L}_{(4)}$  in d = 6 (with some coefficients changed).

#### 30 HOLOGRAPHIC RENORMALIZATION

### 3.2.2 Renormalized Ward Identities & One-Point Functions

In this section, we derive the Ward identities of the EMD model of (3.2.1), which are essentially the boundary versions of the first-class constraints (3.2.18). Varying the renormalized on-shell action on  $\mathcal{M}$  with  $\partial \mathcal{M} = \Sigma_r$ , we get

$$\delta S_{\rm ren} = \frac{1}{2\kappa^2} \int_{\mathcal{M}} d^5 x \, \sqrt{-g} \left( \mathcal{E}_{\mu\nu} \delta g^{\mu\nu} + \mathcal{E}_{\phi} \delta \phi + \mathcal{E}_{\mu} \delta A^{\mu} \right) \tag{3.2.97}$$

$$+\frac{1}{2\kappa^2}\int_{\Sigma_r} \mathrm{d}^4x \,\sqrt{-\gamma} \left(\frac{1}{2}\left\langle T^{ij}\right\rangle_{\mathrm{ren}} \delta\gamma^{ij} + \left\langle \mathcal{O}_{\phi}\right\rangle_{\mathrm{ren}} \delta\phi + \left\langle J^i\right\rangle_{\mathrm{ren}} \delta A_i + \mathcal{A}\delta r\right),\qquad(3.2.98)$$

where  $\mathcal{E}_{\mu\nu}$ ,  $\mathcal{E}_{\phi}$  and  $\mathcal{E}_{\mu}$  are the equations of motion for the respective fields (3.2.2)–(3.2.3), and where the renormalized responses are given in terms of dilatation operator expansion eigenmodes via the form of the on-shell renormalized action (3.2.95)

$$\langle T^{ij} \rangle_{\text{ren}} = \frac{4\kappa^2}{\sqrt{-\gamma}} \frac{\delta S_{\text{ren}}}{\delta \gamma_{ij}} = 4\kappa^2 \pi^{ij}_{(6)}, \qquad (3.2.99)$$

$$\left\langle \mathcal{O}_{\phi} \right\rangle_{\text{ren}} = \frac{2\kappa^2}{\sqrt{-\gamma}} \frac{\delta S_{\text{ren}}}{\delta \phi} = 2\kappa^2 \pi_{\phi(4)},$$
 (3.2.100)

$$\langle J^i \rangle_{\rm ren} = \frac{2\kappa^2}{\sqrt{-\gamma}} \frac{\delta S_{\rm ren}}{\delta A_i} = 2\kappa^2 \pi^i_{(4)}.$$
(3.2.101)

These expressions are all evaluated at the regularized hypersurface  $\Sigma_r$ . Using the asymptotic behaviours

$$\gamma_{ij} \simeq e^{2r} \gamma_{(0)ij}, \quad \phi \simeq \phi_{(0)}, \quad A_i \simeq A_{(0)i}, \quad \sqrt{-\gamma} \simeq e^{4r} \sqrt{-\gamma_{(0)}},$$
 (3.2.102)

it follows that, since  $S_{\text{ren}}$  approaches the finite quantity  $\hat{S}_{\text{ren}}$  as  $r \to \infty$ , we must multiply the responses with a suitably chosen factor in order to obtain finite values—i.e. the actual VEVs—in the limit  $r \to \infty$ , i.e.

$$\langle \hat{T}^{ij} \rangle_{\text{ren}} := \lim_{r \to \infty} e^{6r} \langle T^{ij} \rangle_{\text{ren}} = \frac{4\kappa^2}{\sqrt{-\gamma_{(0)}}} \frac{\delta S_{\text{ren}}}{\delta \gamma_{(0)ij}} = 4\kappa^2 \hat{\pi}^{ij}_{(6)}$$
(3.2.103)

$$\left\langle \hat{\mathcal{O}}_{\phi} \right\rangle_{\text{ren}} := \lim_{r \to \infty} e^{4r} \left\langle \mathcal{O}_{\phi} \right\rangle_{\text{ren}} = \frac{2\kappa^2}{\sqrt{-\gamma_{(0)}}} \frac{\delta S_{\text{ren}}}{\delta \phi_{(0)}} = 2\kappa^2 \hat{\pi}_{\phi(4)}, \tag{3.2.104}$$

$$\langle \hat{J}^i \rangle_{\text{ren}} := \lim_{r \to \infty} e^{4r} \langle J^i \rangle_{\text{ren}} = \frac{2\kappa^2}{\sqrt{-\gamma_{(0)}}} \frac{\delta S_{\text{ren}}}{\delta A_{(0)i}} = 2\kappa^2 \hat{\pi}^i_{(4)}. \tag{3.2.105}$$

Now, consider the momentum constraint  $\mathcal{H}^i = 0$ ,

$$-2D_j\pi_i^j + F_{ij}\pi^j + \pi_{\phi}\partial_i\phi = 0, \qquad (3.2.106)$$

the weight four part of which reads<sup>13</sup>

$$-2D_j\pi^j_{(4)i} + F_{ij}\pi^j_{(4)} + \pi_{\phi(4)}\partial_i\phi = 0, \qquad (3.2.107)$$

which, using the explicit VEVs (3.2.99)-(3.2.101), becomes

$$-D_{j} \langle T^{j}_{i} \rangle_{\text{ren}} + F_{ij} \langle J^{j} \rangle_{\text{ren}} + \langle \mathcal{O}_{\phi} \rangle_{\text{ren}} \partial_{i} \phi = 0, \qquad (3.2.108)$$

which we can multiply with  $e^{4r}$  and take the limit  $r \to \infty$  to obtain the diffeomorphism Ward identity,

$$-D_{(0)j} \left\langle \hat{T}_{i}^{j} \right\rangle_{\text{ren}} + F_{(0)ij} \left\langle \hat{J}^{j} \right\rangle_{\text{ren}} + \left\langle \hat{\mathcal{O}}_{\phi} \right\rangle_{\text{ren}} \partial_{i} \phi_{(0)} = 0.$$
(3.2.109)

We now proceed to derive the trace Ward identity. Using the relation (3.2.61), which states that

$$2\pi^{i}_{(4)i} = -2\widetilde{\mathscr{L}}_{(4)},$$
 (3.2.110)

<sup>13</sup> Note that when lowering the *i*-index on  $\pi_{(4)'}^{ij}$  we get  $\pi_{(4)i}^{j}$  due to the fact that the metric has weight -2; also note that  $F_{ij}$  has weight zero.

we immediately obtain the trace Ward identity—again after multiplying by  $e^{4r}$  and taking the limit  $r \rightarrow \infty$ :

$$\langle \hat{T}^i_i \rangle_{\rm ren} = \hat{\mathcal{A}},$$
 (3.2.111)

where

$$\hat{\mathcal{A}} = -2 \lim_{r \to \infty} e^{4r} \widetilde{\mathscr{L}}_{(4)}, \qquad (3.2.112)$$

is the conformal anomaly on the boundary. Finally, the constraint  $\mathcal F$  with

$$\mathcal{F} = -D_i \pi^i, \tag{3.2.113}$$

turns into the U(1) gauge transformation Ward identity, since at weight 4, it reads

$$0 = D_i \pi^i_{(4)} = D_i \langle \hat{f}^i \rangle_{\text{ren}}, \qquad (3.2.114)$$

which, when multiplied with  $e^{dr}$  and the limit  $r \to \infty$  is taken produces the result

$$D_{(0)i} \left< \hat{f}^{i} \right>_{\rm ren} = 0, \tag{3.2.115}$$

which is the U(1) Ward identity.

## 3.2.3 Generating the d = 4 Fefferman-Graham Expansions

In order to construct the FG expansions of the fields, we are going to apply the gauge-fixed first-order flow equations (3.2.11)-(3.2.13), which for our purposes read

$$\dot{\phi} = -2\kappa^2 \left( \pi_{\phi(2)} + \tilde{\pi}_{\phi(4)} \log e^{-2r} + \pi_{\phi(4)} + \cdots \right),$$
(3.2.118)

$$\dot{A}_{i} = -2\kappa^{2} \left( \tilde{\pi}_{(2)i} \log e^{-2r} + \pi_{(2)i} + \cdots \right).$$
(3.2.119)

In section 3.2.1, we found the following expressions

$$\pi_{(2)}^{ij} = \frac{3}{2\kappa^2} \gamma^{ij}, \tag{3.2.120}$$

$$\pi^{ij}_{(4)} = \frac{1}{2\kappa^2} \left( \frac{1}{8} \partial_k \phi \partial^k \phi \gamma^{ij} - \frac{1}{4} R^{(\gamma)} \gamma^{ij} - \frac{1}{4} \partial^i \phi \partial^j \phi + \frac{1}{2} R^{(\gamma)ij} \right), \tag{3.2.121}$$

$$\tilde{\pi}_{(6)}^{ij} = \frac{1}{\sqrt{-\gamma}} \frac{\delta}{\delta \gamma_{ij}} \int \sqrt{-\gamma} \widetilde{\mathscr{L}}_{(4)}, \qquad (3.2.122)$$

$$\pi_{\phi(2)} = -\frac{1}{4\kappa^2} \Box^{(\gamma)} \phi, \tag{3.2.123}$$

$$\tilde{\pi}_{\phi(4)} = \frac{1}{\sqrt{-\gamma}} \frac{\delta}{\delta\phi} \int \sqrt{-\gamma} \widetilde{\mathscr{F}}_{(4)}, \qquad (3.2.124)$$

$$\tilde{\pi}_{(2)i} = \frac{1}{2\kappa^2} D_j F^j{}_i. \tag{3.2.125}$$

We now introduce very general formal expansions for the fields involved,

$$\gamma_{ij} = \gamma_{ij}^{(0)} + \gamma_{ij}^{(1)} + \cdots, \qquad \phi = \phi^{(0)} + \phi^{(1)} + \cdots, \qquad A_i = A_i^{(0)} + A_i^{(1)} + \cdots, \quad (3.2.126)$$

where each order is assumed to be subleading in *r* to the previous, i.e. *these raised labels are not scaling weights*. When inserting these into the flow equations above, we obtain a series of differential equations which can be solved order-by-order. However, we can do better: making explicit FG ansätze

for (3.2.126) (in these expressions, the labels in the subscripts are not scaling weights, rather they label the respective terms in the FG expansions just as in (F.1.12)),

$$\gamma_{ij} = e^{2r} \left( h_{(0)ij} + e^{-2r} h_{(2)ij} + e^{-4r} \left( \log e^{-2r} h_{(4,1)ij} + h_{(4)ij} \right) + \cdots \right), \qquad (3.2.127)$$

$$\phi = \phi_{(0)} + e^{-2r}\phi_{(2)} + e^{-4r} \left( \log e^{-2r}\phi_{(4,1)} + \phi_{(4)} \right) + \cdots,$$
(3.2.128)

$$A_{i} = A_{(0)i} + e^{-2r} \left( \log e^{-2r} A_{(2,1)i} + A_{(2)i} \right) + \cdots,$$
(3.2.129)

the differential equations become algebraic. Using the momentum (3.2.120), we find the differential equation for the lowest order of the metric  $\dot{\gamma}_{ij}^{(0)} = 2\gamma_{ij}^{(0)}$ , which is solved by  $\gamma^{(0)} = e^{2r}h_{(0)ij}$ , where  $h_{(0)ij}$  plays the rôle of an integration constant, which is in agreement with the ansatz (3.2.127). The same happens for the scalar and the Maxwell field, i.e. we get  $\phi^{(0)} = \phi_{(0)}$  and  $A_i^{(0)} = A_{(0)i}$ . At second order, we get the differential equation for the metric

$$\dot{\gamma}_{ij}^{(1)} = 2\gamma_{ij}^{(1)} + R_{ij}^{(h_{(0)})} - \frac{1}{2}\partial_i\phi_{(0)}\partial_j\phi_{(0)} - \frac{1}{6}R^{(h_{(0)})}h_{(0)ij} + \frac{1}{12}h_{(0)ij}\left(\partial\phi_{(0)}\right)^2, \qquad (3.2.130)$$

where we have kept only the lowest order terms in the expansion. Using the FG ansatz (3.2.127), the left-hand side above vanishes and provides us with the relation

$$h_{(2)ij} = -\frac{1}{2} \left( R_{ij}^{(h_{(0)})} - \frac{1}{2} \partial_i \phi_{(0)} \partial_j \phi_{(0)} - \frac{1}{6} R^{(h_{(0)})} h_{(0)ij} + \frac{1}{12} h_{(0)ij} \left( \partial \phi_{(0)} \right)^2 \right).$$
(3.2.131)

The second order equation for the scalar reads

$$\dot{\phi}^{(1)} = \frac{1}{2} \Box_{(\gamma)} \phi = \frac{1}{2} \frac{1}{\sqrt{-\gamma}} \partial_i \left( \sqrt{-\gamma} \gamma^{ij} \partial_j \phi \right), \qquad (3.2.132)$$

which, when using the FG ansatz becomes

$$2e^{2r}\phi_{(2)} = \frac{e^{2r}}{2} \frac{1}{\sqrt{-h_{(0)}}} \partial_i \left(\sqrt{-h_{(0)}} h_{(0)}^{ij} \partial_j \phi_{(0)}\right) + \mathcal{O}(e^{4r}),$$
(3.2.133)

so that

$$\phi_{(2)} = \frac{1}{4} \Box_{(0)} \phi_{(0)}. \tag{3.2.134}$$

Moving to the next order, we find it useful to consider the the flow equation (3.2.117) in the form

$$\dot{\gamma}_{ij} = -4\kappa^2 \left( \overbrace{\pi_{(-2)ij} - \frac{1}{3}\pi_{(0)}\gamma_{ij}}^{(*)} + \overbrace{\pi_{(0)ij} - \frac{1}{3}\pi_{(2)}\gamma_{ij}}^{(\dagger)} + \log e^{-2r} \underbrace{\left( \overbrace{\pi_{(2)ij} - \frac{1}{3}\pi_{(4)}\gamma_{ij}}^{\ddagger} \right)}_{(3.2.135)} + \overbrace{\pi_{(2)ij} - \frac{1}{3}\pi_{(4)}\gamma_{ij}}^{(**)} + \cdots \right)$$

The differential equation at the next order consequently takes the form

$$\dot{\gamma}_{ij}^{(2)} = \overbrace{2\gamma_{ij}^{(2)}}^{\text{from }(*)} + e^{-2r} \log e^{-2r} [\text{lowest order terms from }(\ddagger)] + e^{-2r} [\text{terms from expansion of }(\dagger)]$$

$$(3.2.136)$$

$$+ e^{-2r} [\text{lowest order terms from }(**)].$$

$$(3.2.137)$$

Now, the Fefferman-Graham ansatz implies that  $\gamma_{ij}^{(2)} = e^{-2r} \left( \log e^{-2r} h_{(4,1)ij} + h_{(4)ij} \right)$ , which means that after subtracting  $2\gamma_{ij}^{(2)}$  (in red) on both sides, we obtain

$$-4e^{-2r}\log e^{-2r}h_{(4,1)ij} - e^{-2r}\left(4h_{(4)ij} + 2h_{(4,1)ij}\right) = \text{appropriate terms from } (*), (\dagger), (\ddagger), (\ast*),$$
(3.2.138)

where "appropriate terms" means terms that have the correct radial dependence when all fields are FG expanded. In particular, we can read off the value of  $h_{(4,1)ij}$  by computing the term involving the logarithm on the right-hand side; this gives<sup>14</sup>

$$h_{(4,1)ij} = \frac{1}{8} D_{(0)}^k \left( D_{(0)i} h_{(2)jk} + D_{(0)j} h_{(2)ik} - D_{(0)k} h_{(2)ij} \right) - \frac{1}{8} D_{(0)i} D_{(0)j} h_{(2)k}^k$$
(3.2.139)

$$+\frac{1}{2}h_{(2)ik}h_{(2)j}^{k}-\frac{1}{4}\partial_{(i}\phi_{(0)}D_{(0)j)}\phi_{(2)}-h_{(0)ij}\left(\frac{1}{8}h_{(2)}^{kl}h_{(2)kl}+\frac{1}{4}\phi_{(2)}^{2}\right)$$
(3.2.140)

$$+\frac{1}{6}F_{(0)ik}F_{(0)j}^{\ \ k}-\frac{1}{24}h_{(0)ij}F_{(0)kl}F_{(0)}^{kl}, \qquad (3.2.141)$$

where  $F_{(0)ij} = 2\partial_{(i}A_{(0)j)}$  and where we have used that  $\gamma^{ij} = e^{-2r}h_{(0)}^{ij} + \mathcal{O}(e^{-4r}\log e^{-2r})$  (see appendix **F**). Now, note that the terms contributing from (\*\*) are

$$(**) = e^{-2r} \left[ \langle \hat{T}_{ij} \rangle_{\text{ren}} - \frac{1}{3} h_{(0)ij} \langle \hat{T}^i_{i} \rangle_{\text{ren}} \right] + \text{higher orders,}$$
(3.2.142)

where we have used the relation for the VEV (3.2.103). Combining this with with our result for  $h_{(4,1)ij}$ , we now find [86]

$$h_{(4)ij} = X_{ij} - \frac{1}{4} \left\langle \hat{T}_{ij} \right\rangle_{\text{ren}}, \qquad (3.2.143)$$

where

$$X_{ij} = \frac{1}{2}h_{(2)ik}h_{(2)j}^k - \frac{1}{4}h_{(2)k}^kh_{(2)ij} + \frac{1}{8}h_{(0)ij}\hat{\mathcal{A}} - \frac{3}{2}h_{(4,1)ij}, \qquad (3.2.144)$$

with the boundary anomaly  $\hat{A}$  defined in (3.2.112). We could now repeat the analysis above for the scalar, but nothing is changed from the case with no Maxwell field, so we refer to the standard treatments [51, 86], where the full FG expansion for the scalar can be found. The result is:

$$\phi_{(4,1)} = -\frac{1}{8} \left[ \Box_{(0)}\phi_{(2)} + 2\phi_{(2)}h^{i}_{(2)i} + \frac{1}{2}\partial^{i}\phi_{(0)}D_{(0)i}h^{j}_{(2)j} - h^{ij}_{(2)}D_{(0)i}\partial_{j}\phi_{(0)} - \partial^{i}\phi_{(0)}D^{j}_{(0)}h_{(2)ij} \right],$$
(3.2.145)

$$\langle \tilde{\mathcal{O}}_{\phi} \rangle_{\text{ren}} = 4\phi_{(4)} + \phi_{(2)}h_{(2)i}^{i} + 6\phi_{(4,1)}.$$
 (3.2.146)

Now, we turn our attention to the Maxwell field. This expansion has not appeared in the literature before to our knowledge. Looking at the FG expansion for the Maxwell field, we see that  $A_i^{(2)} = e^{-2r} \left( \log e^{-2r} A_{(2,1)i} + A_{(2)i} \right)$ , so the flow equation (3.2.119) takes the form

$$\dot{A}_{i}^{(2)} = -2e^{-2r} \left( A_{(2,1)i} + A_{(2)i} \right) - 2e^{-2r} \log e^{-2r} A_{(2,1)i}$$
(3.2.147)

$$= -D_{(0)j}F_{(0)ki}h_{(0)}^{kj}e^{-2r}\log e^{-2r} - e^{-2r}\langle \hat{J}_i \rangle_{\text{ren}} + \text{higher order terms}, \qquad (3.2.148)$$

from which we may immediately read off the component  $A_{(2,1)i}$ :

$$A_{(2,1)i} = \frac{1}{2} D_{(0)j} F^{j}_{(0)i'}$$
(3.2.149)

and

$$A_{(2)i} = \frac{1}{2} \left\langle \hat{J}_i \right\rangle_{\text{ren}} - A_{(2,1)i}.$$
(3.2.150)

# 3.3 Holographic renormalization of p-forms & the scalar-vector duality

#### 3.3.1 From the Maxwell One-Form to p-Forms: A Conjecture

Note that the FG expansion for the Maxwell field (3.2.129) for d = 4 has its logarithmic term appear at order  $O(e^{-2r})$ , which follows from the flow equation (3.2.119). This is a consequence of the fact

<sup>14</sup> We have employed *xAct* [88] and *xTras* [108] to obtain these results as well as used known results for the case without the Maxwell field, see e.g. [51, 57, 86]. Note also the extra factor of two that comes from our convention of writing  $\log e^{-2r}$ , which in Poincaré coordinates reads  $2 \log r_{\text{Poin.}}$ ; when we'll use these results in chapter 6, we will multiply all logarithmic terms by two, since we will change conventions. We hope this does not cause any confusion.

that the Maxwell field is a one-form, since in a general (even) dimension, the associated momentum  $\pi^i$  admits a  $\delta_D$ -expansion of the form<sup>15</sup>

$$\pi^{i} = \dots + \log e^{-2r} \tilde{\pi}^{i}_{(d)} + \pi^{i}_{(d)} + \dots, \qquad (3.3.1)$$

which means that  $\pi_i = \gamma_{ij}\pi^j = \cdots + \log e^{-2r} \tilde{\pi}_{(d-2)i} + \pi_{(d-2)i}$ . Now,  $A_i$  is roughly related to  $\pi_i$  via the flow equations, and so we see that something peculiar happens when d = 2: the VEV part of the expansion occurs at scaling weight zero. More explicitly, the equation of motion for  $A_{\mu}$  (3.2.3) on a AdS<sub>3</sub> background in domain wall coordinates with the ansatz  $A_{\mu} \simeq e^{cr}A_{(0)\mu}$  for, say, the time component reads

$$0 = e^{-2r}\partial_r \left(e^{2r}e^{-2r}F_{rt}\right) = e^{(c-2)r}c^2 A_{(0)},$$
(3.3.2)

implying a double root at c = 0, in agreement with our somewhat heuristic argument above. So, both the source and the VEV carry the same scaling weight, which is at first sight a little puzzling. It turns out, however, that the case d = 2 can be dealt with using Hodge dualization, which we will consider in section 3.3.2. More generally, for a Maxwell field in an  $AdS_{d+1}$  background, we see that the scaling weight of the VEV—that is to say the power of  $e^{-r}$  multiplying it—is given by

$$w_{\rm VEV} = d - 2.$$
 (3.3.3)

Repeating the calculation for d = 1, it is now no longer true that  $A_i$  has scaling weight zero, i.e. that  $A_i \simeq A_{(0)i}$  as in (3.2.20); instead we find that it has scaling weight 1 which changes the entire analysis of section 3.2.1.

We now conjecture a generalization this to *p*-forms: assuming that the equations of motion for a *p*-form  $C^{(p)}$  exhibit the same structure, it is clear that the scaling weight of the VEV gets modified to

$$w_{\rm VEV} = d - 2p, \tag{3.3.4}$$

which leads us to conjecture<sup>16</sup> that, for general p, holographic renormalization works differently depending on whether

$$(i): d-2p > 0$$
 or  $(ii): d-2p < 0$  or  $(iii): d-2p = 0.$  (3.3.5)

#### 3.3.2 The d = 2 Scalar-Vector Duality & Its Generalization

When d = 2—i.e. when we are in case (*iii*) of (3.3.5) for p = 1—we saw above that both the source and the VEV carry the same scaling weight. We now show how this situation can be taken care of by Hodge dualization: in three dimensions, the electromagnetic part of the Wick-rotated<sup>17</sup> action is

$$S_A = \frac{1}{4} \int_{\mathcal{M}} \star F \wedge F = \frac{1}{4} \int_{\mathcal{M}} d^3 x \sqrt{g} F_{\mu\nu} F^{\mu\nu}.$$
(3.3.6)

The field strength two-form *F* is related to a dual scalar  $\varphi$  via<sup>18</sup>

$$F^{\mu\nu} = \epsilon^{\mu\nu\rho} \partial_{\rho} \varphi / \sqrt{g}, \qquad (3.3.7)$$

so that

$$S_A = \frac{1}{4} \int_{\mathcal{M}} \mathrm{d}^3 x \, \sqrt{g} \epsilon_{\mu\nu\sigma} \epsilon^{\mu\nu\rho} \partial_\rho \varphi \partial^\sigma \varphi \tag{3.3.8}$$

$$= \frac{1}{2} \int_{\mathcal{M}} \mathrm{d}^3 x \, \sqrt{g} \partial_\rho \varphi \partial^\rho \varphi, \qquad (3.3.9)$$

where we have used the identity  $\epsilon_{\mu\nu\sigma}\epsilon^{\mu\nu\rho} = 2\delta^{\rho}_{\sigma}$ . In two spatial dimensions renormalizing our EMD model is thus equivalent to renormalizing gravity coupled to two massless scalars. The action describing such a theory is given by

$$S = \frac{1}{2\kappa^2} \int d^3x \,\sqrt{-g} \left( R^{(g)} + 2 - \frac{1}{2} \partial_\mu \phi \partial^\mu \phi - \frac{1}{2} \partial_\mu \phi \partial^\mu \phi \right). \tag{3.3.10}$$

18 With the convention  $\epsilon^{01r} = 1$ .

<sup>15</sup> The first term is  $\pi^i_{(4)}$  in dimensions above two.

<sup>16</sup> I thank Kostas Skenderis for telling me about this condition. Whether the mechanism in the case of *p*-forms is really as simple as the one I have described remains to be seen. The general analysis of holographic renormalization of *p*-forms will be the topic of an upcoming paper by Skenderis and Papadimitriou [52].

<sup>17</sup> We Wick rotate so that the Hodge dual is simpler.

The ADM parametrized momenta are given by

$$\pi^{ij} = \frac{1}{2\kappa^2} \left( K\gamma^{ij} - K^{ij} \right), \quad \pi_{\phi} = \frac{1}{2\kappa^2} N^{-1} \left( \phi - N^i \partial_i \phi \right), \quad \pi_{\varphi} = \frac{1}{2\kappa^2} N^{-1} \left( \varphi - N^i \partial_i \varphi \right). \tag{3.3.11}$$

The Hamiltonian is given by

$$H = \int_{\Sigma_r} \mathrm{d}^2 x \, \sqrt{-\gamma} \left( N \mathcal{H} + N_i \mathcal{H}^i \right), \qquad (3.3.12)$$

where

$$\mathcal{H} = -2\kappa^2 \left( \pi^{ij}\pi_{ij} - \pi^2 + \frac{1}{2}\pi_{\phi}^2 + \frac{1}{2}\pi_{\varphi}^2 \right) + \frac{1}{2\kappa^2} \left( -R^{(\gamma)} + \frac{1}{2}\partial_i\phi\partial^i\phi + \frac{1}{2}\partial_i\varphi\partial^i\varphi - 2 \right), \quad (3.3.13)$$

$$\mathcal{H}^{i} = -2D_{j} + \pi_{\phi}\partial^{i}\phi + \pi_{\phi}\partial^{i}\phi.$$
(3.3.14)

As usual, the equations of motion for the Lagrange multiplier fields impose the first class constraints,

$$\mathcal{H} = 0, \quad \mathcal{H}^i = 0. \tag{3.3.15}$$

Gauge-fixing to N = 1 and  $N_i = 0$ , we impose AlAdS boundary conditions, which read

$$\gamma_{ij} \simeq e^{2r} g_{(0)ij}(x), \quad \phi \simeq \phi_{(0)}(x) \quad \varphi \simeq \varphi_{(0)}(x),$$
(3.3.16)

where *x* represents the coordinates on  $\Sigma_r$ . Carrying out the same procedure as in section 3.2, we find that

$$\mathscr{L}_{(0)} = \frac{1}{2\kappa^2},\tag{3.3.17}$$

$$\widetilde{\mathscr{L}}_{(2)} = \frac{1}{\kappa^2} R^{(\gamma)} - \frac{1}{2\kappa^2} \left( \partial_i \phi \partial^i \phi + \partial_i \varphi \partial^i \varphi \right).$$
(3.3.18)

Hodge dualizing back to the vector theory, we obtain

$$\partial_i \varphi = \frac{1}{2} \epsilon_{\mu\nu i} F^{\mu\nu} \sqrt{g}, \tag{3.3.19}$$

$$\therefore \frac{1}{2\kappa^2} \partial_i \varphi \partial^i \varphi = \frac{1}{8\kappa^2} \epsilon_{\mu\nu i} \epsilon^{\rho\lambda i} F^{\mu\nu} F_{\rho\lambda} = \frac{1}{8\kappa^2} \left[ 2\epsilon_{rji} \epsilon^{rki} F^{rj} F_{rk} \right] = \frac{1}{8\kappa^2} \left[ 4F^{ri} F_{ri} \right] = \frac{1}{2\kappa^2} F^{ri} F_{ri}, \quad (3.3.20)$$

implying that the anomaly takes the form

$$\widetilde{\mathscr{L}}_{(2)} = \frac{1}{\kappa^2} R^{(\gamma)} - \frac{1}{2\kappa^2} \partial_i \phi \partial^i \phi - \frac{1}{2\kappa^2} F^{ri} F_{ri}.$$
(3.3.21)

To generalize this to *p*-forms, we note that the field strength (p + 1)-form  $F^{(p+1)} = dC^{(p)}$  of the *p*-form  $C^{(p)}$  in d = 2p, corresponding to case (*iii*) of (3.3.5), is dual to a *p*-form, which is not immediately useful. A more fruitful observation is that in d = p + 1, the field strength  $F^{(p+1)}$  is again dual to a scalar in the sense of (3.3.7), which means that the methods of this section generalize to a way of easily performing holographic renormalization of free massless *p*-form fields in  $AdS_{p+2}$ , since in this case it is equivalent to renormalizing a free massless scalar.

## 3.4 OUTLOOK

The results of this chapter will be important when we consider charged Lifshitz holography in chapter 6, since the renormalized electromagnetic uplift can be Scherk-Schwarz reduced to an Einstein-Procadilaton-Maxwell-scalar model, which admits Lifshitz solutions that correspond to z = 0 Schrödinger spactimes<sup>19</sup> in the higher dimensional theory.

More generally, there are many unsolved problems in holographic renormalization. As we demonstrated, even in the case of AlAdS spacetimes, there are subtleties regarding the renormalization of *p*-form fields that are not widely known. Holographic renormalization has also been undertaken for other geometries of holographic interest: Schrödinger space-times were considered in e.g. [109–111], while Lifshitz spacetimes were dealt with in e.g. [25, 27, 28, 106], but there are still several outstanding

<sup>19</sup> And these are AlAdS.

problems. In Appendix G we provide a review of the renormalization of Lifshitz geometries using the HJ approach developed in this chapter.

An interesting generalization of the procedure described in this chapter was introduced in [28, 106] in the context of renormalization of EPD models with Lifshitz boundary conditions, and involves the identification of yet another operator  $\delta'$  that *commutes* with, say,  $\delta_D$  (generally, whatever we choose for  $\delta$ ). One then performs a dual expansion in (common) eigenmodes of the two operators, which simplifies the analysis. It would be interesting to explore if such methods are also advantageous in the context of the models discussed in this chapter.

Newton-Cartan (NC) geometry was originally developed by Cartan to provide a geometric framework for Newtonian gravity [30, 31], and in the context of non-relativistic holography, NC geometry plays a prominent rôle as the boundary geometry.

Specifically, it was shown in [23, 24] that the boundary geometry of asymptotically locally z = 2 Lifshitz space-times is described by an extension of NC geometry known as twistless torsional Newton-Cartan geometry (TTNC), while for for generic values of the dynamical exponent,  $1 < z \le 2$ , the boundary geometry is described by torsional Newton-Cartan geometry (TNC) [29, 32]—this we explore in chapters 5 and 6.

Since TNC geometry plays a vital rôle in Lifshitz holography, we devote this chapter to its description. However, TNC geometry is both interesting and useful to study in its own right—independent of holography, it has, for example, recently been shown that TNC geometry provides an effective description of the quantum Hall effect [35, 36]. Interestingly, it was shown in [53] that making TNC dynamical corresponds to Hořava-Lifshitz (HL) gravity.

We begin the chapter in section (4.1) with a survey of how Riemannian geometry emerges from gauging the Poincaré algebra, which will be useful when replicating the procedure in order to obtain TNC geometry. This section is a significantly expanded analysis of appendix A of [53] and [54].

We then turn to a general description of *non-relativistic space-times* in section 4.2, which are the non-relativistic analogues of the familiar Lorentzian spacetimes from general relativity. Our exposition is based on [24, 33, 53] and involves elements of [112], although we do not use their terminology. Geometrically, non-relativistic space-times are described with TNC geometry, and we provide a classification of these non-relativistic structures depending on the properties of the so-called clock form,  $\tau$ , which, as we will see in section 4.2.4, is intimately connected with the torsion.

In section 4.2.2, we begin with a discussion of how to obtain the Bargmann and Galilei algebras via Inönü-Wigner contraction of a centrally extended Poincaré algebra. We then proceed to gauge both the Galilei and the Bargmann algebras in sections 4.2.3 and 4.2.4, which will give us TNC geometry in the same way as Riemannian geometry appears from the gauging of the Poincaré algebra. This analysis is based mainly on [34, 53, 54, 56].

We then turn our attention to how TNC geometry can be obtained via *null reduction* of Lorentzian spacetimes in section 4.3, which will be of great use in chapter 6.

Finally, in section 4.4, we explore how field theories couple to Newton-Cartan geometries from the perspective of null reduction, which again will be a great aid to our considerations in chapter 6. These last two sections both follow [57].

#### 4.1 WARM-UP: GAUGING THE POINCARÉ ALGEBRA TO OBTAIN GENERAL RELATIVITY

Following [54, 55] and Appendix A of [53], we proceed to derive the framework of general relativity from properties of the Poincaré algebra (see e.g. [113] for a nice review). The *d*-dimensional Poincaré algebra is $\mathfrak{so}(d-1,1)$  is generated by  $P_a$  and  $J_{ab}$  with the following non-zero commutators,

$$[J_{ab}, P_c] = \eta_{ac} P_b - \eta_{bc} P_a, \tag{4.1.1}$$

$$[J_{ab}, J_{cd}] = \eta_{ac} J_{bd} - \eta_{ad} J_{bc} - \eta_{bc} J_{ad} + \eta_{bd} J_{ac}.$$
(4.1.2)

Note that the tangent space indices a, b, c, ... in this section *include time*, in contradistinction to the rest of this chapter. The first step of the gauging procedure is to introduce the Lie algebra valued connection by

$$\mathcal{A}_{\mu} = P_{a}e_{\mu}^{a} + \frac{1}{2}M_{ab}\omega_{\mu}{}^{ab}, \qquad (4.1.3)$$

where antisymmetry of *M* means that we take  $\omega_{\mu}{}^{ab} = -\omega_{\mu}{}^{ba}$ . Mimicking the approach taken for Yang-Mills theory<sup>1</sup>, we fist observe that the connection behaves under *local ISO*(*d* - 1, 1) transformations as

$$\mathcal{A}_{\mu} \to L(x)\mathcal{A}_{\mu}L^{-1}(x) - L(x)\partial_{\mu}L^{-1}(x), \qquad L(x) \in ISO(d-1,1).$$
(4.1.4)

<sup>1</sup> For a nice review, see e.g. the book [114].

Infinitesimally, we may write  $L(x) = 1 + \Lambda(x)$  for  $\Lambda(x) \in \mathfrak{iso}(d-1, d)$ , and so, under infinitesimal transformations<sup>2</sup>, the connection transforms as

$$\delta \mathcal{A}_{\mu} = \partial_{\mu} \Lambda + [\mathcal{A}_{\mu}, \Lambda], \tag{4.1.5}$$

where, since  $\Lambda \in \mathfrak{iso}(d-1,1)$ , we may write

$$\Lambda = P_a \zeta^a(x) + \frac{1}{2} J_{ab} \sigma^{ab}(x). \tag{4.1.6}$$

Since

$$\delta \mathcal{A}_{\mu} = P_a \delta e^a_{\mu} + \frac{1}{2} J_{ab} \delta \omega_{\mu}{}^{ab}, \qquad (4.1.7)$$

the transformation (4.1.5) with  $\Lambda$  given by (4.1.6) implies that

$$\delta e^a_\mu = \partial_\mu \zeta^a - \omega_\mu{}^{ab} \zeta^b + \overbrace{\sigma^a{}_b e^b_\mu}^{(*)}, \tag{4.1.8}$$

$$\delta\omega_{\mu}{}^{ab} = \partial_{\mu}\sigma^{ab} + \sigma_{c}{}^{a}\omega_{\mu}{}^{bc} - \sigma_{c}{}^{b}\omega_{\mu}{}^{ac}.$$
(4.1.9)

To get a feel for how these are obtained, we show how to find the term (\*). This term comes from the commutator  $[A_{\mu}, \Lambda] = \frac{e_{\mu}^{a} \sigma^{bc}}{2} [P_{a}, J_{bc}] + \dots$ , where we can use the commutator relation (4.1.1) to rewrite

$$\frac{e_{\mu}^{a}\sigma^{bc}}{2}[P_{a},J_{bc}] = \frac{e_{\mu}^{a}\sigma^{bc}}{2}\eta_{a[c}P_{b]} = e_{\mu}^{a}\sigma^{b}{}_{a}P_{b},$$
(4.1.10)

where we have used antisymmetry of  $\sigma^{ab}$ . In order to make connection to local space-time diffeomorphisms, we introduce a new set of local (infinitesimal) transformations, denoted  $\bar{\delta}$ , where we replace the local translation parameter  $\zeta^a$  in  $\Lambda$  with a space-time vector  $\xi^{\mu}$  via  $\zeta^a = \xi^{\mu} e^a_{\mu}$ , which makes the emergence of diffeomorphisms manifest<sup>3</sup>, as we demonstrate. Writing

$$\Lambda = \xi^{\mu} \mathcal{A}_{\mu} + \Sigma, \text{ where } \Sigma = \frac{1}{2} \underbrace{\left(\sigma^{ab} - \xi^{\mu} \omega_{\mu}{}^{ab}\right)}_{=:\lambda^{ab}} J_{ab},$$
(4.1.11)

we are now ready to define  $\bar{\delta}A_{\mu}$ :

$$\bar{\delta}\mathcal{A}_{\mu} = \delta\mathcal{A}_{\mu} - \xi^{\nu}\mathcal{F}_{\mu\nu}, \qquad (4.1.12)$$

where  $\mathcal{F}_{\mu\nu}$  is the curvature of  $\mathcal{A}_{\mu}$ , defined in the usual manner,

$$\mathcal{F}_{\mu\nu} = \partial_{\mu}\mathcal{A}_{\nu} - \partial_{\nu}\mathcal{A}_{\mu} + [\mathcal{A}_{\mu}, \mathcal{A}_{\nu}]. \tag{4.1.13}$$

Writing out  $\delta$ , we find

$$\begin{split} \delta\mathcal{A}_{\mu} &= \partial_{\mu}\Lambda + [\mathcal{A}_{\mu},\Lambda] - \xi^{\nu}\mathcal{F}_{\mu\nu} \tag{4.1.14} \\ &= \mathcal{A}_{\nu}\partial_{\mu}\xi^{\nu} + \xi^{\nu}\partial_{\mu}\mathcal{A}_{\nu} + \partial_{\mu}\Sigma + \xi^{\nu}[\mathcal{A}_{\mu},\mathcal{A}_{\nu}] + [\mathcal{A}_{\mu},\Sigma] - \xi^{\nu}\partial_{\mu}\mathcal{A}_{\nu} + \xi^{\nu}\partial_{\nu}\mathcal{A}_{\mu} - \xi^{\nu}[\mathcal{A}_{\mu},\mathcal{A}_{\nu}] \tag{4.1.15}$$

$$= \pounds_{\xi} \mathcal{A}_{\mu} + \partial_{\mu} \Sigma + [\mathcal{A}_{\mu}, \Sigma], \tag{4.1.16}$$

where, as claimed, the transformation of  $A_{\mu}$  under diffeomorphisms generated by  $\xi$ —that is to say, the term  $\mathcal{L}_{\xi}A_{\mu}$ —appears explicitly. We also note that the parameters  $\lambda^{ab}$  now correspond to infinitesimal local Lorentz transformations generated by  $J_{ab}$ .

Now, going back to the curvature  $\mathcal{F}_{\mu\nu}$ , we may conveniently express it in terms of curvatures pertaining to the generators *P* and *J*,

$$\mathcal{F}_{\mu\nu} = R_{\mu\nu}{}^{a}(P)P_{a} + \frac{1}{2}R_{\mu\nu}{}^{ab}(J)J_{ab}.$$
(4.1.17)

<sup>2</sup> That is, ignoring terms of order  $\mathcal{O}(\Lambda^2, \Lambda \partial \Lambda)$ .

<sup>3</sup> This approach first appeared in [53] and was not used in e.g. [34, 54]. We shall refer to this trick as the  $\bar{\delta}$ -method.

The explicit expressions of the curvatures are determined by plugging the connection (4.1.3) into the curvature (4.1.13) and using the Poincaré algebra:

$$\mathcal{F}_{\mu\nu} = \partial_{[\mu}\mathcal{A}_{\nu]} + [\mathcal{A}_{\mu}, \mathcal{A}_{\nu}]$$

$$= 2P_{a}\partial_{[\mu}e^{a}_{\nu]} + J_{ab}\partial_{[\mu}\omega_{\nu]}{}^{ab} + \frac{1}{2}e^{a}_{\mu}\omega_{\nu}{}^{bc}\overbrace{[P_{a}, J_{bc}]}^{=-2\eta_{a[b}P_{c]}} + \frac{1}{4}\omega_{\mu}{}^{ab}\omega_{\nu}{}^{cd}\overbrace{[J_{ab}, J_{cd}]}^{=-4\eta_{[a[c}J_{d]b]}} + \frac{1}{2}e^{a}_{\nu}\omega_{\mu}{}^{bc}\overbrace{[J_{bc}, P_{a}]}^{2\eta_{a[b}P_{c]}}$$

$$(4.1.18)$$

$$(4.1.19)$$

$$(4.1.19)$$

$$=2P_{a}\left(\partial_{[\mu}e_{\nu]}^{a}-\omega_{[\mu}{}^{ab}e_{\nu]b}\right)+J_{ab}\left(\partial_{[\mu}\omega_{\nu]}{}^{ab}-\omega_{[\mu}{}^{ca}\omega_{\nu]}{}^{b}{}_{c}\right),$$
(4.1.20)

leading us to conclude that

$$R_{\mu\nu}{}^{a}(P) = 2\partial_{[\mu}e_{\nu]}^{a} - 2\omega_{[\mu}{}^{ab}e_{\nu]b}, \quad R_{\mu\nu}{}^{ab}(J) = 2\partial_{[\mu}\omega_{\nu]}{}^{ab} - 2\omega_{[\mu}{}^{ca}\omega_{\nu]}{}^{b}{}_{c}, \tag{4.1.21}$$

where the factor of two in the expression for  $R_{\mu\nu}{}^{ab}(J)$  comes from its definition in (4.1.17). Under  $\bar{\delta}$ -transformations, the quantities  $e^a_{\mu}$  and  $\omega_{\mu}{}^{ab}$  transform respectively as a vielbein and a spin connection: writing out (4.1.16), we find

$$\bar{\delta}\mathcal{A}_{\mu} = P_{a}\mathcal{L}_{\xi}e^{a}_{\mu} + \frac{1}{2}J_{ab}\mathcal{L}_{\xi}\omega_{\mu}{}^{ab} + \frac{1}{2}J_{ab}\partial_{\mu}\lambda^{ab} + \frac{1}{2}\lambda^{cd}\left(e^{a}_{\mu}[P_{a}, J_{cd}] + \frac{1}{2}\omega_{\mu}{}^{ab}[J_{ab}, J_{cd}]\right)$$
(4.1.22)

$$= P_{a}\pounds_{\xi}e_{\mu}^{a} + \frac{1}{2}J_{ab}\pounds_{\xi}\omega_{\mu}{}^{ab} + \frac{1}{2}J_{ab}\partial_{\mu}\lambda^{ab} + P_{a}\lambda^{a}{}_{b}e_{\mu}^{b} + \lambda_{c}{}^{b}\omega_{\mu}{}^{ca}J_{ab},$$
(4.1.23)

and so, using  $\bar{\delta}A_{\mu} = P_a \bar{\delta}e^a_{\mu} + \frac{1}{2}J_{ab}\bar{\delta}\omega_{\mu}{}^{ab}$ , we immediately get:

$$\bar{\delta}e^a_\mu = \pounds_{\tilde{\zeta}}e^a_\mu + \lambda^a{}_be^b_\mu, \tag{4.1.24}$$

$$\bar{\delta}\omega_{\mu}{}^{ab} = \pounds_{\bar{\zeta}}\omega_{\mu}{}^{ab} + \partial_{\mu}\lambda^{ab} + 2\lambda^{[a}{}_{c}\omega_{\mu}{}^{|c|b]}.$$
(4.1.25)

Thus,  $e^a_{\mu}$  does indeed transforms as a vielbein under  $\bar{\delta}$ -transformations, with  $\lambda^{ab}$  an infinitesimal local Lorentz transformation. Next, we introduce a space-time covariant derivative via

$$\mathcal{D}_{\mu}e_{\nu}^{a} = \partial_{\mu}e_{\nu}^{a} - \Gamma_{\mu\nu}^{\rho}e_{\rho}^{a} - \omega_{\mu}^{\ a}{}_{b}e_{\nu}^{b}, \qquad (4.1.26)$$

which transforms (by requirement) covariantly under  $\bar{\delta}$ -transformations. By this, we mean that

$$\bar{\delta}\left(\mathcal{D}_{\mu}e_{\nu}^{a}\right) = \pounds_{\xi}\left(\left(\mathcal{D}_{\mu}e_{\nu}^{a}\right)\right) + \lambda^{a}{}_{b}\left(\mathcal{D}_{\mu}e_{\nu}^{b}\right). \tag{4.1.27}$$

This allows us determine the  $\bar{\delta}$ -transformation of the affine connection  $\Gamma^{\rho}_{\mu\nu}$ , since

$$\bar{\delta}\left(\mathcal{D}_{\mu}e_{\nu}^{a}\right) = \overbrace{\partial_{\mu}\left(\bar{\delta}e_{\nu}^{a}\right)}^{=(\ast)} - \overbrace{\bar{\delta}\left(\Gamma_{\mu\nu}^{\rho}e_{\rho}^{a}\right)}^{=(\dagger)} - \overbrace{\bar{\delta}\left(\omega_{\mu}^{a}{}_{b}e_{\nu}^{b}\right)}^{=(\ddagger)}.$$
(4.1.28)

Considering each term in isolation and applying the transformation properties of the vielbein and spin connection, (4.1.24) and (4.1.25), we find

$$(*) = (\partial_{\mu}\xi^{\rho})(\partial_{\rho}e^{a}_{\nu}) + \xi^{\rho}(\partial_{\mu}\partial_{\rho}e^{a}_{\nu}) + e^{a}_{\rho}\partial_{\nu}\partial_{\mu}\xi^{\rho} + e^{b}_{\nu}\partial_{\mu}\lambda^{a}_{\ b} + \lambda^{a}_{\ b}\partial_{\mu}e^{a}_{\nu}, \tag{4.1.29}$$

whereas

$$(\dagger) = \Gamma^{\rho}_{\mu\nu} \left( \xi^{\sigma} \partial_{\sigma} e^{a}_{\rho} + e^{a}_{\sigma} \partial_{\nu} \xi^{\sigma} + \lambda^{a}{}_{b} e^{b}_{\rho} \right) + \bar{\delta} \Gamma^{\rho}_{\mu\nu} e^{a}_{\rho}, \qquad (4.1.30)$$

and, finally,

$$(\ddagger) = e_{\nu b} \left( \xi^{\rho} \partial_{\rho} \omega_{\mu}{}^{ab} + \omega_{\rho}{}^{ab} \partial_{\mu} \xi^{\rho} + \frac{\partial_{\mu} \lambda^{ab}}{\partial_{\mu} \partial_{\nu} \partial_{\nu} \partial_{\nu} \partial_{\nu} \partial_{\nu} \partial_{\mu} \partial_{\nu} \partial_{\mu} \partial_{\nu} \partial_{\mu} \partial_{\nu} \partial_{\nu} \partial_{\mu} \partial_{\nu} \partial_{\nu} \partial_{\nu} \partial_{\nu} \partial_{\nu} \partial_{\mu} \partial_{\nu} \partial_{\nu} \partial_{\mu} \partial_{\nu} \partial_{\mu} \partial_{\mu} \partial_{\mu} \partial_{\mu} \partial_{\nu} \partial_{\mu} \partial_{\mu} \partial_{\nu} \partial_{\mu} \partial$$

Colour-coded terms are equal, and will cancel out when taking the combination  $(*) - (\dagger) - (\ddagger)$ , leaving us with

$$\bar{\delta} \left( \mathcal{D}_{\mu} e^{a}_{\nu} \right) = \left( \partial_{\mu} \xi^{\rho} \right) \left( \partial_{\rho} e^{a}_{\nu} \right) + \xi^{\rho} \left( \partial_{\mu} \partial_{\rho} e^{a}_{\nu} \right) + e^{a}_{\rho} \left( \partial_{\nu} \partial_{\mu} \xi^{\rho} \right) + \lambda^{a}_{\ b} \partial_{\mu} e^{a}_{\nu} - \Gamma^{\rho}_{\mu\nu} \left( \xi^{\sigma} \partial_{\sigma} e^{a}_{\rho} + e^{a}_{\sigma} \partial_{\rho} \xi^{\sigma} + \lambda^{a}_{\ b} e^{b}_{\rho} \right) - \bar{\delta} \Gamma^{\rho}_{\mu\nu} e^{a}_{\rho}$$

$$(4.1.32)$$

$$- e_{\nu b} \left( \xi^{\rho} \partial_{\rho} \omega_{\mu}^{\ ab} + \omega_{\rho}^{\ ab} \partial_{\mu} \xi^{\rho} + \lambda^{a}_{\ c} \omega_{\mu}^{\ cb} \right) - \omega_{\mu}^{\ a}_{\ b} \left( \xi^{\rho} \partial_{\rho} e^{b}_{\nu} + e^{b}_{\rho} \partial_{\nu} \xi^{\rho} \right).$$

$$(4.1.33)$$

Now, compare this with the expression for  $\bar{\delta} (\mathcal{D}_{\mu} e_{\nu}^{a})$  of (4.1.27) written out in all its glory,

$$\bar{\delta}\left(\mathcal{D}_{\mu}e_{\nu}^{a}\right) = \pounds_{\xi}\left(\left(\mathcal{D}_{\mu}e_{\nu}^{a}\right)\right) + \lambda^{a}{}_{b}\left(\mathcal{D}_{\mu}e_{\nu}^{b}\right) \tag{4.1.34}$$

$$=\xi^{\sigma}(\partial_{\sigma}\partial_{\mu}e^{a}_{\nu})-\xi^{\sigma}e^{a}_{\rho}\partial_{\sigma}\Gamma^{\rho}_{\mu\nu}-\xi^{\sigma}\Gamma^{\rho}_{\mu\nu}\partial_{\sigma}e^{a}_{\rho}-\xi^{\sigma}\omega_{\mu}{}^{a}{}_{b}\partial_{\sigma}e^{b}_{\nu}-\xi^{\sigma}e^{b}_{\nu}\partial_{\sigma}\omega_{\mu}{}^{a}{}_{b}$$
(4.1.35)

$$+ \partial_{\mu}\xi^{\sigma} \left[ \partial_{\sigma}e^{a}_{\nu} - \Gamma^{\rho}_{\sigma\nu}e^{a}_{\rho} - \omega^{a}_{\sigma}{}^{b}e^{b}_{\nu} \right] + \partial_{\nu}\xi^{\sigma} \left[ \partial_{\mu}e^{a}_{\sigma} - \Gamma^{\rho}_{\mu\sigma}e^{a}_{\rho} - \omega^{a}_{\mu}{}^{b}b^{c}_{\sigma} \right]$$
(4.1.36)

$$+\lambda^{a}_{\ b}\left(\partial_{\mu}e^{b}_{\nu}-\Gamma^{\rho}_{\mu\nu}e^{b}_{\rho}-\omega_{\mu}{}^{b}_{\ c}e^{c}_{\nu}\right).$$
(4.1.37)

Demanding that these be equal, we see that, first off, everything to do with the spin connection already matches, whereas a single term from the Lie derivative is amiss. The constraint that the two  $\delta$ -variations be equal then reduces to

$$e^{a}_{\rho}(\partial_{\nu}\partial_{\mu}\xi^{\rho}) - \Gamma^{\rho}_{\mu\nu} \left(\xi^{\sigma}\partial_{\sigma}e^{a}_{\rho} + e^{a}_{\sigma}\partial_{\rho}\xi^{\sigma} + \lambda^{a}_{\ b}e^{b}_{\rho}\right) - \bar{\delta}\Gamma^{\rho}_{\mu\nu}e^{a}_{\rho} \stackrel{(!)}{=}$$
(4.1.38)

$$-\xi^{\sigma}e^{a}_{\rho}\partial_{\sigma}\Gamma^{\rho}_{\mu\nu} - \xi^{\sigma}\Gamma^{\rho}_{\mu\nu}\partial_{\sigma}e^{a}_{\rho} - \Gamma^{\rho}_{\sigma\nu}e^{a}_{\rho}\partial_{\mu}\xi^{\sigma} - \Gamma^{\rho}_{\mu\sigma}e^{a}_{\rho}\partial_{\nu}\xi^{\sigma} - \underline{\lambda^{a}}_{b}\Gamma^{\rho}_{\mu\nu}e^{b}_{\rho}, \qquad (4.1.39)$$

leading us to conclude that

$$\bar{\delta}\Gamma^{\rho}_{\mu\nu} = \partial_{\mu}\partial_{\nu}\xi^{\rho} + \xi^{\sigma}\partial_{\sigma}\Gamma^{\rho}_{\mu\nu} + \Gamma^{\rho}_{\sigma\nu}\partial_{\mu}\xi^{\sigma} + \Gamma^{\rho}_{\mu\sigma}\partial_{\nu}\xi^{\sigma} - \Gamma^{\sigma}_{\mu\nu}\partial_{\sigma}\xi^{\rho}.$$
(4.1.40)

The next step is to relate the curvatures to the affine connection, which we do by imposing the following vielbein postulate

$$\mathcal{D}_{\mu}e_{\nu}^{a}=0,$$
 (4.1.41)

which allows for an identification of the affine connection  $\Gamma^{\rho}_{\mu\nu}$  in terms of the spin connection  $\omega_{\mu}{}^{ab}$ . Taking the antisymmetric part of the vielbein postulate, we find

$$0 = \partial_{[\mu} e^{a}_{\nu]} - \Gamma^{\rho}_{[\mu\nu]} e^{a}_{\rho} - \omega_{[\mu}{}^{a}{}^{b}_{b} e^{b}_{\nu]}$$
(4.1.42)

$$\therefore R_{\mu\nu}{}^{a}(P) = 2\partial_{[\mu}e^{a}_{\nu]} - 2\omega_{[\mu}{}^{a}{}_{b}e^{b}_{\nu]} = 2\Gamma^{\rho}_{[\mu\nu]}e^{a}_{\rho}, \qquad (4.1.43)$$

where we used the curvatures (4.1.21). Thus, we see that  $R_{\mu\nu}{}^{a}(P)$  is really the torsion. The unique Lorentz invariant tensor we can build out of the vielbeine is the metric,  $g_{\mu\nu} = \eta_{ab}e^{a}_{\mu}e^{b}_{\nu}$ , implying that the connection may be written as

$$\Gamma^{\rho}_{\mu\nu} = e^{\rho}_a \partial_{\mu} e^a_{\nu} - e^{\rho}_a \omega_{\mu}{}^a{}_b e^b_{\nu}. \tag{4.1.44}$$

To see what the other curvature represents, we define the "usual" covariant derivative  $\nabla_{\mu} X_{\mu} := \partial_{\mu} X_{\nu} - \Gamma^{\rho}_{\mu\nu} X_{\rho}$ , the commutator of which is related to the Riemann tensor (ix)

$$[\nabla_{\mu}, \nabla_{\nu}] X_{\rho} = R_{\mu\nu\rho}{}^{\sigma} X_{\sigma} - \Gamma^{\sigma}_{[\mu\nu]} \nabla_{\sigma} X_{\rho}, \qquad (4.1.45)$$

where, explicitly,

$$R_{\mu\nu\sigma}^{\ \rho} = 2\partial_{[\nu}\Gamma^{\rho}_{\mu]\sigma} + 2\Gamma^{\rho}_{[\rho|\lambda|}\Gamma^{\lambda}_{\mu]\sigma'} \tag{4.1.46}$$

which, using the curvatures (4.1.21) and imposing the vielbein postulate (4.1.41), can be rewritten as

$$R_{\mu\nu\rho}^{\ \ \sigma} = -e_{\rho a} e_b^{\sigma} R_{\mu\nu}^{\ \ ab}(J), \tag{4.1.47}$$

implying that  $R_{\mu\nu}{}^{ab}(J)$  is the Riemann curvature two-form. Further, the vielbein postulate implies due to antisymmetry of the spin connection in the flat indices—that the metric is covariantly conserved with respect to  $\nabla_{\mu}$ , i.e.

$$\nabla_{\rho}g_{\mu\nu} = \partial_{\rho}g_{\mu\nu} - \Gamma^{\lambda}_{\rho\mu}g_{\lambda\nu} - \Gamma^{\lambda}_{\rho\nu}g_{\lambda\mu} = 2\omega_{\rho ab}e^{a}_{(\mu}e^{b}_{\nu)} = 0, \qquad (4.1.48)$$

where we have used the affine connection (4.1.44). This fixes the symmetric part of the connection and makes it equal to the Levi-Civita connection, while leaving the torsion unfixed. However, imposing the curvature constraint  $R_{\mu\nu}{}^{a}(P) = 0$  then corresponds to setting the torsion equal to zero, and, in turn, makes the spin connection entirely determined by the vielbeine in the usual manner. By imposing Einstein's equations, the theory can be put on-shell.

#### 4.2 NON-RELATIVISTIC SPACE-TIMES & NEWTON-CARTAN GEOMETRY

In this section, we introduce the appropriate geometric framework for the description of non-relativistic physics. Our exposition begins with a discussion of non-relativistic space-times and their geometric descriptions—see also [112, 115, 116] for similar work, where a slightly different terminology is used.

We then turn to a more detailed discussion of Newton-Cartan geometry as obtained from gauging appropriate algebras in the spirit of section 4.1. For a complementary analysis of Newton-Cartan geometry using frame bundles, we refer the reader to [117].

#### 4.2.1 Non-Relativistic Space-Times

A (d+1)-dimensional non-relativistic space-time consists of a triad  $(\mathcal{M}, \tau, h^{-1})$ , where  $\mathcal{M}$  is a (d+1)-dimensional manifold,  $\tau$  is a nowhere-vanishing one-form on  $\mathcal{M}$  known as the *clock form*, and  $h^{-1}$  is a rank-two contravariant symmetric tensor whose kernel is spanned by  $\tau$ , i.e.  $h^{-1}(\tau, \cdot) = 0$ , or, in components<sup>4</sup>,

$$h^{\mu\nu}\tau_{\nu} = 0. \tag{4.2.1}$$

The doublet  $(\tau, h^{-1})$  may be thought of as a degenerate metric structure, and thus is not invertible. We may, however, define the *projective inverse dyad* (v, h) satisfying

$$v^{\mu}\tau_{\mu} = -1, \quad h_{\mu\rho}h^{\rho\nu} = \delta^{\nu}_{\mu} + \tau_{\mu}v^{\nu}.$$
 (4.2.2)

These are not unique [117] and can be boosted to take a different form. The fields  $(\tau, v, h^{-1}, h)$  (with a proper choice of connection) realize a torsional Newton-Cartan (TNC) geometry, and it is often advantageous to trade the *h*'s for vielbeine, defined through  $h_{\mu\nu} = \delta_{ab} e^a_{\mu} e^b_{\nu}$  and  $h^{\mu\nu} = \delta^{ab} e^{\mu}_{a} e^{\nu}_{b}$ —something we will make extensive use of.

Now, consider two points  $A, B \in M$  connected by a path  $\gamma$ . Parametrizing the path  $\gamma$  by some parameter  $\lambda$ , we interpret the integral

$$T_{\gamma} = \int_{\gamma} \tau = \int_{\gamma} \tau_{\mu} \frac{\mathrm{d}x^{\mu}}{\mathrm{d}\lambda} \mathrm{d}\lambda, \qquad (4.2.3)$$

as the proper time that passes when going from *A* to *B* along  $\gamma$ .

Since there is no restriction on the clock form  $\tau$ , we will refer to this type of space-time as a TNC space-time. It is so far purely "metric" in nature (i.e. we have not introduced any connections, so no notion of parallelism is involved). Depending on the properties of the clock form  $\tau$ , we now proceed to introduce two special cases of the TNC space-time, which, when introducing a connection, will be in a one-to-one correspondence with the various forms of TNC geometry classified by the torsion of the connection (see section 4.2.3), starting with

**Definition:** A *TTNC space-time* is a non-relativistic space-time, whose clock form satisfies the Frobenius integrability condition,  $\tau \wedge d\tau = 0$ , which in component form reads  $\tau_{[\rho}\partial_{\mu}\tau_{\nu]} = 0$ . This is equivalent to the statement that  $\tau$  be hypersurface orthogonal (HSO).

By Frobenius' theorem [118], the one-form thus defines a foliation of  $\mathcal{M}$  by a family of codimension one hypersurfaces, or *leaves*. Each leaf corresponds to an absolute space, in the following sense: by Frobenius' theorem, we may *locally* write the clock form as  $\tau = f dt$  for some  $f, t \in C^{\infty}(\mathcal{M})$ , where f is the time unit and t is the *absolute time*. This absolute time is fixed on each leaf, and it is in this sense that they are absolute spaces. Furthermore, the metric h (or rather, the pullback of h to the appropriate leaf) is non-degenerate on these hypersurfaces since  $\tau = dt = 0$ , so the completeness relation (4.2.2) implies that  $h_{\mu\nu}$  defines a (torsion free) Riemannian geometry on each absolute space. Finally, the most restrictive case is characterized by a *closed* clock form:

**Definition:** An NC space-time is a non-relativistic space-time, whose clock form is exact, i.e.  $\tau = dt$ , where t is the absolute time.

This means—by Stokes theorem—that the proper time (4.2.3) between two points  $A, B \in \mathcal{M}$  is independent of the path one follows, which leads to a concept of absolute time not present in the more

<sup>4</sup> Note that  $h^{-1} = h^{\mu\nu}\partial_{\mu} \otimes \partial_{\nu}$ , so  $h^{\mu\nu}$  are the components of  $h^{-1}$ .



Figure 4.1: Foliation of a TTNC space-time by absolute spaces.

general TNC and TTNC space-times. In order to fully realize a NC geometry, we require a connection. A convenient and illuminating way of obtaining NC geometry in its various incarnations is to gauge an appropriate algebra, which we describe in the next section.

#### 4.2.2 Newton-Cartan Geometry from Local Algebras

The most general version of Newton-Cartan geometry, known as *torsional Newton-Cartan geometry* (TNC), can be obtained in several ways. The route we take involves adding torsion to the analysis of [54], which concerns the gauging of the Bargmann algebra. TNC can also be obtained by gauging the Schrödinger algebra which is locally scale invariant; an endeavour undertaken in [34] (see also chapter 6, where we gauge the Schrödinger algebra for z = 2). The latter approach is more computationally involved, but reveals a richer structure which is the same as the one encounters in general-*z* Lifshitz holography.

#### 4.2.2.1 The Bargmann Algebra from Ínönü-Wigner Contraction

It is possible [54] to obtain the Bargmann algebra by extending the (d + 1)-dimensional Poincaré algebra iso(d, 1) of (4.1.1) via a direct sum with a commutative subalgebra  $\mathfrak{g}_N$  generated by N (occasionally referred as the mass for reasons we will explain shortly),

$$iso(d,1) \to iso(d,1) \oplus \mathfrak{g}_N.$$
 (4.2.4)

To get to the Bargmann algebra, we perform the following Ínönü-Wigner contraction (see [119])

$$\overbrace{P_0 \to \frac{1}{\varepsilon^2} N + H, \quad P_a \to \frac{1}{\varepsilon} P_a, \quad J_{0a} \to \frac{1}{\varepsilon} G_a,}^{\varepsilon \to 0}$$
(4.2.5)

where the contraction parameter  $\varepsilon$  corresponds to the reciprocal speed of light. This choice is inspired by the non-relativistic approximation of  $P_0$  for a free particle of mass<sup>5</sup> *m*:

$$P_0 = \sqrt{c^2 P_a P^a + m^2 c^4} \approx mc^2 + \frac{P_a P^a}{2m}.$$
(4.2.6)

The algebra thus obtained is the Bargmann algebra barg(d, 1), and is generated by H (time translations),  $P_a$  (spatial rotations),  $G_a$  (Galilean boosts),  $J_{ab}$  (spatial rotations) and, finally, N (mass/particle number). The contraction produces the following non-vanishing commutators

$$[H, G_a] = P_a, \quad [J_{ab}, G_c] = 2\delta_{c[a}G_{b]}, \quad [J_{ab}, P_c] = 2\delta_{c[a}P_{b]}, \quad [J_{ab}, J_{cd}] = 4\delta_{[a[d}J_{c]b]}, \quad [P_a, G_b] = N\delta_{ab}.$$
(4.2.7)

These are obtained from the commutation relations of the Poincaré algebra (4.1.1); for example, we have that

$$[P_a, G_b] = [\varepsilon P_a, \varepsilon J_{0b}] = \varepsilon^2 \left( \overbrace{\eta_{0a}}^{=0} P_b - \delta_{ab} \left( \frac{1}{\varepsilon^2} N + H \right) \right) \to \delta_{ab} N.$$
(4.2.8)

<sup>5</sup> Corresponding to *N*.

Setting N = 0, we obtain the (d + 1)-dimensional Galilean algebra,  $\mathfrak{gal}(d, 1)$ . It is interesting to note that N is required to obtain massive representations of the Galilean algebra, for which there is a nice argument: following [54], we can understand this by considering the action for a non-relativistic free particle of mass m,

$$S = \int dt \ L = \frac{1}{2} \int dt \ m \dot{x}^2, \tag{4.2.9}$$

which is invariant under Gal(d, 1). However, the Lagrangian L is *not* invariant; rather, it transforms as a total derivative under infinitesimal Galilean boosts,  $\delta x^a = v^a t$ , since

$$\delta L = m\dot{x}_a \delta \dot{x}^a = \frac{\mathrm{d}}{\mathrm{d}t} \left( m x_a v^a \right), \qquad (4.2.10)$$

which implies that the naïve Nöther charge  $Q_{naïve} = p_a \delta x^a = m \dot{x}_a v^a t$  gets an additional boundary term, so that the correct boost Nöther charge takes the form

$$\mathcal{Q}_G = p_a v^a t - m x_a v^a. \tag{4.2.11}$$

The translation Noether charge is straightforward, i.e.  $Q_P = p_a \xi^a$  for infinitesimal translations  $\delta x^a = \xi^a$ , which means that their Poisson bracket reads

$$\{\mathcal{Q}_P, \mathcal{Q}_G\}_{PB} = \frac{\partial \mathcal{Q}_P}{\partial x^a} \frac{\partial \mathcal{Q}_G}{\partial p_a} - \frac{\partial \mathcal{Q}_P}{\partial p_a} \frac{\partial \mathcal{Q}_G}{\partial x^a} = mv^a \xi_a, \qquad (4.2.12)$$

which is in accord with the commutator (4.2.8).

## 4.2.3 Gauging Galilei

Mimicking the approach in section 4.1, we start by defining the following Lie algebra valued connection,

$$\mathcal{A}_{\mu} = H\tau_{\mu} + P_{a}e_{\mu}^{a} + G_{a}\omega_{\mu}{}^{a} + \frac{1}{2}J_{ab}\omega_{\mu}{}^{ab}.$$
(4.2.13)

For  $L(x) \in \mathfrak{gal}(d, 1)$ , the connection transforms as  $\mathcal{A}_{\mu} \to L(x)\mathcal{A}_{\mu}L^{-1}(x) - L(x)\partial_{\mu}L^{-1}(x)$ , and so, taking L(x) to be infinitesimal,  $L(x) = \mathbb{1} + \Lambda(x)$  for  $\Lambda(x) \in \mathfrak{gal}_2(d, 1)$ , we get

$$\delta \mathcal{A}_{\mu} = \partial_{\mu} \Lambda + [\mathcal{A}_{\mu}, \Lambda], \qquad (4.2.14)$$

i.e.  $\mathcal{A}_{\mu}$  transforms in the adjoint. Now, since  $\Lambda(x) \in \mathfrak{gal}(d, 1)$ , we may write

$$\Lambda(x) = H\zeta(x) + P_a\zeta^a(x) + G_a\tilde{\lambda}^a(x) + \frac{1}{2}J_{ab}\tilde{\lambda}^{ab}(x).$$
(4.2.15)

Next, we make contact with local space-time diffeomorphisms via the introduction of the  $\bar{\delta}$ -transformation, where we replace the local translation parameters  $\zeta^a$  in  $\Lambda$  with a space-time vector  $\xi^{\mu}$  defined via  $\zeta^a = \xi^{\mu} e^a_{\mu}$ , allowing us to write

$$\Lambda = \xi^{\mu} \mathcal{A}_{\mu} + \Sigma, \qquad (4.2.16)$$

which implies that

$$\Sigma = G_a \lambda^a + \frac{1}{2} J_{ab} \lambda^{ab}, \qquad (4.2.17)$$

where the un-tilded parameters of (4.2.17) are related to their tilded cousins of (4.2.15) in a manner similar to the way local Lorentz transformations emerged in the case of the Poincaré algebra in (4.1.11). The  $\bar{\delta}$ -transformation—compare eqs. (4.1.12) and (4.1.16)—then becomes

$$\bar{\delta}\mathcal{A}_{\mu} = \delta\mathcal{A}_{\mu} - \xi^{\nu}\mathcal{F}_{\mu\nu} = \pounds_{\xi}\mathcal{A}_{\mu} + \partial_{\mu}\Sigma + [\mathcal{A}_{\mu}, \Sigma], \qquad (4.2.18)$$

where  $\mathcal{F}_{\mu\nu}$  is the Yang-Mills curvature,

$$\mathcal{F}_{\mu\nu} = 2\partial_{[\mu}\mathcal{A}_{\nu]} + [\mathcal{A}_{\mu}, \mathcal{A}_{\nu}] \tag{4.2.19}$$

$$= HR_{\mu\nu}(H) + P_a R_{\mu\nu}{}^a(P) + G_a R_{\mu\nu}{}^a(G) + \frac{1}{2} J_{ab} R_{\mu\nu}{}^{ab}(J).$$
(4.2.20)

#### NEWTON-CARTAN GEOMETRY 44

We summarise this construction in the table below:

			-	
Symmetry	Generators	Gauge Field	Parameters	Curvatures
Time translations	Н	$\tau_{\mu}$	$\zeta(x)$	$R_{\mu u}(H)$
Spatial translations	$P_a$	$e^a_\mu$	$\zeta^a(x)$	$R_{\mu\nu}^{a}(P)$
Boosts	Ga	$\omega_{\mu}{}^{a}$	$\lambda^a(x)$	$R_{\mu\nu}^{a}(G)$
Spatial rotations	J <sub>ab</sub>	$\omega_{\mu}^{ab}$	$\lambda^{ab}(x)$	$R_{\mu\nu}^{ab}(J)$

Table 4.1: Generators of gal(d, 1) with their associated gauge fields, local parameters and covariant curvatures.

Writing out the expression for  $\bar{\delta}A_{\mu}$  in (4.2.19) with the help of our expressions for  $A_{\mu}$  and  $\Sigma$ —as we did for Poincaré in (4.1.18)–(4.1.20)—we find the following curvatures:

$$R_{\mu\nu}(H) = 2\partial_{[\mu}\tau_{\nu]}, \tag{4.2.21}$$

$$R_{\mu\nu}{}^{a}(P) = 2\partial_{[\mu}e_{\nu]}^{a} - 2\omega_{[\mu}{}^{ab}e_{\nu]b} - 2\omega_{[\mu}{}^{a}\tau_{\nu]}, \qquad (4.2.22)$$

$$R_{\mu\nu}^{\ \ ab}(J) = 2\partial_{[\mu}\omega_{\nu]}^{\ \ ab} - 2\omega_{[\mu}^{\ \ c[a}\omega_{\nu]}^{\ \ b]}{}_{c}, \qquad (4.2.23)$$

$$R_{\mu\nu}{}^{a}(G) = 2\partial_{[\mu}\omega_{\nu]}{}^{a} + 2\omega_{[\mu}{}^{b}\omega_{\nu]}{}^{a}{}_{b}.$$
(4.2.24)

Similarly, by writing out  $A_{\mu}$  and  $\Sigma$  in (4.2.18) and identifying coefficients in front of the generators, we find the following transformations of the gauge fields:

$$\bar{\delta}\tau_{\mu} = \pounds_{\xi}\tau_{\mu},\tag{4.2.25}$$

$$\bar{\delta}e^a_\mu = \pounds_{\xi}e^a_\mu + \lambda^a\tau_\mu + \lambda^a{}_be^b_\mu, \qquad (4.2.26)$$

$$\bar{\delta}e^{a}_{\mu} = \pounds_{\xi}e^{a}_{\mu} + \lambda^{a}\tau_{\mu} + \lambda^{a}{}_{b}e^{b}_{\mu}, \qquad (4.2.26)$$

$$\bar{\delta}\omega_{\mu}{}^{ab} = \pounds_{\xi}\omega_{\mu}{}^{ab} + \partial_{\mu}\lambda^{ab} + 2\lambda^{c[a}\omega_{\mu}{}^{b]}{}_{c}, \qquad (4.2.27)$$

$$\bar{\delta}\omega_{\mu}{}^{a} = \pounds_{\xi}\omega_{\mu}{}^{a} + \partial_{\mu}\lambda^{a} - \lambda^{b}\omega_{\mu}{}^{a}{}_{b} + \lambda^{a}{}_{b}\omega_{\mu}{}^{b}.$$
(4.2.28)

The gauge fields  $\tau_{\mu}$  and  $e^{a}_{\mu}$  transform under spatial rotations and Galilean boosts as the Newton-Cartan clock form and vielbeine, respectively [54], and we identify them as such. Since they are of rank 1 and rank *d*, respectively, they are not invertible in a (d + 1)-dimensional spacetime, but we can define projective inverses  $v^{\mu}$  and  $e_a^{\mu}$ , satisfying the relations

$$v^{\mu}\tau_{\mu} = -1, \quad v^{\mu}e^{a}_{\mu} = 0, \quad \tau_{\mu}e^{\mu}_{a} = 0, \quad e^{a}_{\mu}e^{\mu}_{b} = \delta^{a}_{b}, \quad e^{\mu}_{a}e^{a}_{\nu} = \delta^{\mu}_{\nu} + v^{\mu}\tau_{\nu}.$$
 (4.2.29)

The projective inverses transform under  $\bar{\delta}$ -transformations in the following manner,

$$\bar{\delta}v^{\mu} = \pounds_{\tilde{c}}v^{\mu} + \lambda^{a}e^{\mu}_{a}, \qquad (4.2.30)$$

$$\bar{\delta}e_a^\mu = \pounds_{\xi}e_a^\mu + \lambda_a{}^b e_b^\mu, \qquad (4.2.31)$$

which are derived by considering<sup>6</sup> the relations  $0 = \bar{\delta} \left( v^{\mu} \tau_{\mu} \right)$  and  $\bar{\delta} \left( v^{\mu} e^{a}_{\mu} \right)$  and using the identities (4.2.29). The curvatures (4.2.21)–(4.2.24) are subject to the usual Bianchi identity  $d_D \mathcal{F} = 0$  for  $d_D$  the exterior covariant derivative, which in local coordinates reads

$$0 = D_{[\mu} \mathcal{F}_{\nu \rho]} = \partial_{[\mu} \mathcal{F}_{\nu \rho]} + \left[ \mathcal{A}_{[\mu}, \mathcal{F}_{\nu \rho]} \right], \qquad (4.2.32)$$

where D is the gauge covariant derivative in the adjoint representation. In order to identify the structure of the covariant derivative acting on the gauge fields, we use the standard relation  $D_{\mu}A_{\nu}$  =  $\partial_{\mu}\mathcal{A}_{\nu} + [\mathcal{A}_{\mu}, \mathcal{A}_{\nu}]$ , which allows us to identify additional terms in the covariant derivatives that ensure that they transform covariantly under  $\bar{\delta}$ -transformations, (4.2.25)–(4.2.28). By taking into account all the commutators of gal(d, 1) that are proportional to *H* and *P*<sub>a</sub>, we find the following expressions for the covariant derivatives,

$$\mathcal{D}_{\mu}\tau_{\nu} = \partial_{\mu}\tau_{\nu} - \Gamma^{\rho}_{\mu\nu}\tau_{\rho}, \qquad (4.2.33)$$

$$\mathcal{D}_{\mu}e_{\nu}^{a} = \partial_{\mu}e_{\nu}^{a} - \Gamma_{\mu\nu}^{\rho}e_{\rho}^{a} - \omega_{\mu}^{\ a}{}_{b}e_{\nu}^{b} - \omega_{\mu}^{\ a}\tau_{\nu}, \qquad (4.2.34)$$

$$\mathcal{D}_{\mu}v^{\nu} = \partial_{\mu}v^{\nu} + \Gamma^{\nu}_{\mu\lambda}v^{\lambda} - \omega_{\mu}{}^{a}e^{\nu}_{a}, \qquad (4.2.35)$$

$$\mathcal{D}_{\mu}e_{a}^{\nu} = \partial_{\mu}e_{a}^{\nu} + \Gamma_{\mu\lambda}^{\nu}e_{a}^{\lambda} + \omega_{\mu}{}^{b}{}_{a}e_{b}^{\nu}.$$

$$(4.2.36)$$

<sup>6</sup> And, for simplicity, ignoring diffeomorphisms.

We can explicitly check that they transform covariantly; for example, as we had for the Poincaré algebra,  $D_{\mu}\tau_{\nu}$  should transform only under diffeomorphisms, i.e. we require that

$$\bar{\delta}\left(\mathcal{D}_{\mu}\tau_{\nu}\right) = \mathcal{L}_{\xi}\left(\mathcal{D}_{\mu}\tau_{\nu}\right) \tag{4.2.37}$$

$$=\xi^{\rho}\partial_{\rho}\left(\mathcal{D}_{\mu}\tau_{\nu}\right)+\partial_{\mu}\xi^{\rho}\left(\mathcal{D}_{\rho}\tau_{\nu}\right)+\partial_{\nu}\xi^{\rho}\left(\mathcal{D}_{\mu}\tau_{\rho}\right)$$
(4.2.38)

$$=\xi^{\rho}\left(\partial_{\rho}\partial_{\mu}\tau_{\nu}\right)-\xi^{\rho}\left(\partial_{\rho}\Gamma^{\lambda}_{\mu\nu}\right)\tau_{\lambda}-\xi^{\rho}\Gamma^{\lambda}_{\mu\nu}\partial_{\rho}\tau_{\lambda}$$
(4.2.39)

$$+ \partial_{\mu}\xi^{\rho} \left(\partial_{\rho}\tau_{\nu} - \Gamma^{\lambda}_{\rho\nu}\tau_{\lambda}\right) + \partial_{\nu}\xi^{\rho} \left(\partial_{\mu}\tau_{\rho} - \Gamma^{\lambda}_{\mu\rho}\tau_{\lambda}\right).$$
(4.2.40)

By using (4.2.25)–(4.2.28), the  $\bar{\delta}$ -variation of the covariant derivative may equivalently be expressed in the following manner,

$$\bar{\delta}\left(\mathcal{D}_{\mu}\tau_{\nu}\right) = \partial_{\mu}\bar{\delta}\tau_{\nu} - \bar{\delta}\Gamma^{\rho}_{\mu\nu}\tau_{\rho} - \Gamma^{\rho}_{\mu\nu}\bar{\delta}\tau_{\rho}, \qquad (4.2.41)$$

which, when equated with (4.2.40), produces the same transformation property (4.1.40) for  $\Gamma^{\nu}_{\mu\lambda}$  that we found in section 4.1. We now impose the vielbein postulates,

$$\mathcal{D}_{\mu}\tau_{\nu}=0, \qquad (4.2.42)$$

$$\mathcal{D}_{\mu}e_{\nu}^{a}=0.$$
 (4.2.43)

These postulates allow us to express the affine connection in terms of  $\omega_{\mu}{}^{a}$  and  $\omega_{\mu}{}^{ab}$  by using the identities (4.2.29). In particular, multiplying the first postulate (4.2.42) with  $v^{\lambda}$  produces the relation

 $v^{\lambda}\partial_{\mu}\tau_{\nu} = -\Gamma^{\lambda}_{\mu\nu} + e^{\lambda}_{a} \widetilde{\Gamma^{\rho}_{\mu\nu}} e^{a}_{\rho}$ , while the second postulate (4.2.43) can be rearranged to give an expression for (\*). This produces the result

$$\Gamma^{\lambda}_{\mu\nu} = -v^{\lambda}\partial_{\mu}\tau_{\nu} + e^{\lambda}_{a}\left(\partial_{\mu}e^{a}_{\nu} - \omega_{\mu}{}^{a}{}_{b}e^{b}_{\nu} - \omega_{\mu}{}^{a}\tau_{\nu}\right).$$
(4.2.44)

The vielbein postulates also allow for a identification of the curvatures (4.2.21) and (4.2.22) in terms of the affine connection: the postulates (4.2.42) and (4.2.43) imply that

$$R_{\mu\nu}(H) = 2\Gamma^{\rho}_{[\mu\nu]}\tau_{\rho}, \qquad (4.2.45)$$

$$R_{\mu\nu}^{\ \ a}(P) = 2\Gamma^{\rho}_{[\mu\nu]}e^{a}_{\rho'}$$
(4.2.46)

so they are related to the torsion tensor in the following manner; contracting the curvature (6.4.63) with  $e_a^{\lambda}$  and using the geometric identities (4.2.29), we get

$$R_{\mu\nu}^{\ \ a}e_a^{\lambda} = 2\Gamma^{\rho}_{[\mu\nu]}e_{\rho}^{a}e_a^{\lambda} \tag{4.2.47}$$

$$=2\Gamma^{\rho}_{[\mu\nu]}\left(\delta^{\lambda}_{\rho}+v^{\lambda}\tau_{\rho}\right) \tag{4.2.48}$$

$$=2\Gamma^{\lambda}_{[\mu\nu]}+v^{\lambda}\underbrace{2\Gamma^{\rho}_{[\mu\nu]}\tau_{\rho}}_{=R_{\mu\nu}(H)}$$
(4.2.49)

$$\therefore 2\Gamma^{\lambda}_{[\mu\nu]} = -v^{\lambda}R_{\mu\nu}(H) + e^{\lambda}_{a}R_{\mu\nu}{}^{a}(P).$$
(4.2.50)

We now introduce a new covariant derivative  $\nabla_{\mu}$  involving only the affine connection  $\Gamma^{\rho}_{\mu\nu}$ , which means that writing out the first vielbein postulate (4.2.42) using the definition (4.2.33) gives us the relation

$$\nabla_{\mu}\tau_{\nu} = 0, \qquad (4.2.51)$$

while antisymmetry of  $\omega_{\mu}{}^{ab}$  means that, by the second vielbein postulate (4.2.43), the object  $h^{\mu\nu} = \delta^{ab} e^{\mu}_{a} e^{\nu}_{b}$  satisfies

$$abla_{\mu}h^{
u
ho} = 0,,$$
(4.2.52)

where eqs. (4.2.51) and (4.2.52) are the TNC version of metric compatibility. With this new derivative at hand, we now turn to the identification of the other curvatures. To that end, we begin by computing the Riemann tensor,

$$[\nabla_{\mu}, \nabla_{\nu}]X_{\rho} = R_{\mu\nu\sigma}{}^{\rho}X_{\rho} - 2\Gamma^{\rho}_{[\mu\nu]}\nabla_{\rho}X_{\sigma}, \qquad (4.2.53)$$

for an arbitrary one-form  $X_{\rho}$ , where

$$R_{\mu\nu\sigma}{}^{\rho}X_{\rho} = -\partial_{\mu}\Gamma^{\rho}_{\nu\sigma} + \partial_{\nu}\Gamma^{\rho}_{\mu\sigma} - \Gamma^{\rho}_{\mu\lambda}\Gamma^{\lambda}_{\nu\sigma} + \Gamma^{\rho}_{\nu\lambda}\Gamma^{\lambda}_{\mu\sigma}, \qquad (4.2.54)$$

which, upon using our expression for  $\Gamma^{\rho}_{\mu\nu}$  (4.2.44) as well as the curvatures (4.2.23)–(4.2.24), produces the result

$$R_{\mu\nu\sigma}{}^{\rho} = e^{\rho}_{a} \tau_{\sigma} R_{\mu\nu}{}^{a}(G) - e_{\sigma a} e^{\rho}_{b} R_{\mu\nu}{}^{ab}(J).$$
(4.2.55)

#### 4.2.4 Extension to Bargmann and the Affine Connection

We now extend the analysis above to the Bargmann algebra (4.2.7) and include the central element *N*. We denote the associated gauge connection by  $m_{\mu}$ , so that

$$\mathcal{A}_{\mu} = H\tau_{\mu} + P_{a}e_{\mu}^{a} + G_{a}\omega_{\mu}{}^{a} + \frac{1}{2}J_{ab}\omega_{\mu}{}^{ab} + Nm_{\mu}, \qquad (4.2.56)$$

$$\Sigma = G_a \lambda^a + \frac{1}{2} J_{ab} \lambda^{ab} + N\sigma, \qquad (4.2.57)$$

$$\mathcal{F}_{\mu\nu} = 2\partial_{[\mu}\mathcal{A}_{\nu]} + [\mathcal{A}_{\mu}, \mathcal{A}_{\nu}] \tag{4.2.58}$$

$$= HR_{\mu\nu}(H) + P_a R_{\mu\nu}{}^a(P) + G_a R_{\mu\nu}{}^a(G) + \frac{1}{2} J_{ab} R_{\mu\nu}{}^{ab}(J) + NR_{\mu\nu}(N).$$
(4.2.59)

where  $\sigma$  is the parameter associated with the *N* transformation. As *N* is central, all the results of the previous section are unchanged; in particular the transformations (4.2.25)–(4.2.28) are the same with the augmentation

$$\bar{\delta}m_{\mu} = \pounds_{\xi}m_{\mu} + \partial_{\mu}\sigma + e^{a}_{\mu}\lambda_{a}.$$
(4.2.60)

To make contact with TNC geometry arising in Lifshitz holography as it appears in [23, 24, 29, 33], it is convenient to introduce a background Stückelberg field  $\chi$  transforming as

$$\bar{\delta}\chi = \pounds_{\xi}\chi + \sigma, \tag{4.2.61}$$

and to define

$$M_{\mu} = m_{\mu} - \partial_{\mu}\chi, \qquad (4.2.62)$$

which is invariant under local *N* transformations. It was shown in [34] that this has the effect of replacing  $m_{\mu}$  with  $M_{\mu}$  everywhere. It is convenient to define a series of Galilean boost invariant objects,

$$\hat{v}^{\mu} = v^{\mu} - h^{\mu\nu}M_{\nu}, \quad \hat{e}^{a}_{\mu} = e^{a}_{\mu} - M_{\nu}e^{\nu a}\tau_{\mu}, \quad \tilde{\Phi} = -v^{\mu}M_{\mu} + \frac{1}{2}h^{\mu\nu}M_{\mu}M_{\nu}, \quad (4.2.63)$$

$$h^{\mu\nu} = \delta^{ab} e^{\mu}_{a} e^{\nu}_{b}, \quad \bar{h}_{\mu\nu} = \delta_{ab} e^{a}_{\mu} e^{b}_{\nu} - \tau_{\mu} M_{\nu} - \tau_{\nu} M_{\mu}. \tag{4.2.64}$$

These objects are invariant under local Galilean boosts (we will discuss their properties in a holographic context in section 5.2.4). The quantity  $\tilde{\Phi}$  is closely related to the Newtonian potential when the space-time is flat [33, 34]; for this reason we shall refer to it as the Newtonian potential. We note that it represents the component of  $M_{\mu}$  that cannot be boosted away. Now, the next crucial observation is that the objects ( $\hat{e}^{a}_{\mu}, \hat{v}^{\mu}, \tau_{\mu}, e^{\mu}_{a}$ ) form an orthonormal set, i.e. they satisfy the same orthogonality relations as in (4.2.29); for example we have that

$$\hat{v}^{\mu}\tau_{\mu} = v^{\mu}\tau_{\mu} - \delta^{ab} \underbrace{e^{\mu}_{a}\tau_{\mu}}_{e^{\nu}_{b}} e^{\nu}_{b} = v^{\mu}\tau_{\mu} = -1.$$
(4.2.65)

The invariant objects further satisfy the following useful relations:

$$h^{\nu\rho}\bar{h}_{\rho\mu} = \delta^{\nu}_{\mu} + \vartheta^{\nu}\tau_{\mu}, \quad \vartheta^{\mu}\bar{h}_{\mu\nu} = 2\tau_{\nu}\tilde{\Phi}, \quad \hat{e}^{a}_{\mu}\hat{e}_{\nu a} = \bar{h}_{\mu\nu} + 2\tilde{\Phi}\tau_{\mu}\tau_{\nu}, \quad -\vartheta^{\nu}\tau_{\mu} + \hat{e}^{a}_{\mu}e^{\nu}_{a} = \delta^{\nu}_{\mu}. \quad (4.2.66)$$

\_0

The first of these, for example, is derived in the following manner

$$h^{\nu\rho}\bar{h}_{\rho\mu} = e^{\nu}_{a}e^{a}_{\mu} - e^{a\nu} \, \overline{e^{\rho}_{a}\tau_{\rho}} \, M_{\mu} - \tau_{\mu}M_{\rho}e^{\rho a}e^{\nu}_{a}$$
(4.2.67)

$$=\delta^{\nu}_{\mu} + (v^{\nu} - M_{\rho}h^{\rho\nu})\tau_{\mu} \tag{4.2.68}$$

$$=\delta^{\nu}_{\mu} + \hat{v}^{\nu}\tau_{\mu}, \tag{4.2.69}$$

where we have used (4.2.29). From the invariant objects we have considered so far, the most general affine and metric compatible (in the sense of (4.2.51) and (4.2.52)) connection was constructed in [33, 53] and has the form

$$\Gamma^{\rho}_{\mu\nu} = -v^{\rho}\partial_{\mu}\tau_{\nu} + \frac{1}{2}h^{\rho\lambda}\left(\partial_{\mu}h_{\nu\lambda} + \partial_{\nu}h_{\mu\lambda} - \partial_{\lambda}h_{\mu\nu}\right) + W^{\rho}_{\mu\nu}, \qquad (4.2.70)$$

$$W^{\rho}_{\mu\nu} = \frac{1}{2}h^{\rho\lambda} \left(\tau_{\mu}K_{\lambda\nu} + \tau_{\nu}K_{\lambda\mu} + L_{\lambda\mu\nu}\right), \qquad (4.2.71)$$

$$K_{\mu\nu} = -K_{\nu\mu}, \quad L_{\lambda\mu\nu} = -L_{\nu\mu\lambda}, \tag{4.2.72}$$

where  $K_{\mu\nu}$  and  $L_{\lambda\mu\nu}$  transform as tensors under diffeomorphisms. The object  $W^{\rho}_{\mu\nu}$  is known as the *pseudo-contortion* tensor in analogy with the object that arises in Riemannian geometry [56]. We now see that TNC connections (4.2.70) in general have non-vanishing torsion, since for any choice of pseudo-contortion, we have

$$2\Gamma^{\rho}_{[\mu\nu]}\tau_{\lambda} = \partial_{\mu}\tau_{\nu} - \partial_{\nu}\tau_{\mu}, \qquad (4.2.73)$$

which makes the connection between the clock form and torsion manifest. In summary: for Newton-Cartan (NC) geometry, the torsion vanishes since  $\tau$  is closed,  $d\tau = 0$ , while in twistless torsional Newton-Cartan (TTNC) geometry,  $\tau$  obeys the Frobenius condition  $\tau \wedge d\tau = 0$  implying that the *twist* vanishes (see also chapter 5),  $h^{\mu\rho}h^{\nu\sigma}(\partial_{\mu}\tau_{\nu} - \partial_{\nu}\tau_{\mu}) = 0$ , and, finally, torsional Newton-Cartan (TNC) geometry, where no restrictions are imposed on  $\tau$ . This result is worthy of tabulation:

Table 4.2: The three Newton-Cartan geometries.

Geometry	Constr. on $ au$	Abs. time	Abs. space	Torsion
TNC	None	NO	NO	YES
TTNC	$ au \wedge \mathrm{d} au = 0$	NO	YES	YES
NC	d au = 0	YES	YES	NO

There exists a unique TNC connection linear in  $M_{\mu}$  given by [34, 120]

$$\overline{\Gamma}^{\rho}_{\mu\nu} = -\vartheta^{\rho}\partial_{\mu}\tau_{\nu} + \frac{1}{2}h^{\rho\sigma}\left(\partial_{\mu}\bar{h}_{\nu\sigma} + \partial_{\nu}\bar{h}_{\mu\sigma} - \partial_{\sigma}\bar{h}_{\mu\nu}\right), \qquad (4.2.74)$$

which, from the perspective of the Noether procedure, is the minimal TNC connection [56]. This connection is obtained by setting

$$K_{\sigma\rho} = 2\partial_{[\sigma}M_{\rho]}, \quad L_{\sigma\mu\nu} = 2M_{\sigma}\partial_{[\mu}\tau_{\nu]} - 2M_{\mu}\partial_{[\nu}\tau_{\sigma]} + 2M_{\nu}\partial_{[\sigma}\tau_{\mu]}. \tag{4.2.75}$$

#### 4.3 TNC GEOMETRY FROM NULL REDUCTION

An alternative route to TNC geometry is through null reduction of a Lorentzian (d + 1)-dimensional manifold [23, 24, 57, 121, 122]. Consider to this end the following null reduction ansatz for the metric

$$ds^{2} = \gamma_{AB} dx^{A} dx^{B} = 2\tau_{\mu} dx^{\mu} \left( du - m_{\nu} dx^{\nu} \right) + h_{\mu\nu} dx \mu dx^{\nu}$$
(4.3.1)

$$= 2\tau_{\mu}dx^{\mu}du + \bar{h}_{\mu\nu}dx^{\mu}dx^{\nu}, \qquad (4.3.2)$$

where  $A = (u, \mu)$  and

$$h_{\mu\nu} = \delta_{ab} e^a_{\mu} e^b_{\nu}, \quad \bar{h}_{\mu\nu} = h_{\mu\nu} - 2\tau_{(\mu} m_{\nu)}, \tag{4.3.3}$$

where  $a = 1, \dots, d$ . The reduction ansatz (4.3.1) is the most general metric with  $\gamma_{uu} = 0$ , and we take  $\partial_u$  to be a Killing vector for this metric, which thus becomes a null Killing vector of  $\gamma_{AB}$ . The fields are  $\tau_{\mu}$  and  $e_{\mu}^{a}$  are the vielbeine of the *d*-dimensional TNC geometry. The inverse metric is taken to be

$$\gamma^{\mu\nu} = 2\widetilde{\Phi}, \quad \gamma^{\mu\mu} = -\widehat{v}^{\mu}, \quad \gamma^{\mu\nu} = h^{\mu\nu}, \tag{4.3.4}$$

where

$$\widetilde{\Phi} = -v^{\rho}m_{\rho} + \frac{1}{2}h^{\rho\sigma}m_{\rho}m_{\sigma}, \qquad (4.3.5)$$

$$\hat{v}^{\mu} = v^{\mu} - h^{\mu\nu} m_{\nu}, \qquad (4.3.6)$$

$$h^{\mu\nu} = \delta^{ab} e^{\mu}_{a} e^{\nu}_{b}, \tag{4.3.7}$$

where the inverse vielbeine  $v^{\mu}$  and  $e^{\mu}_{a}$  are defined via the usual relations (4.2.29). The following diffeomorphisms preserve the form of the null Killing vector,

$$u' = u - \sigma(x), \tag{4.3.8}$$

$$x'^{\mu} = x'^{\mu}(x). \tag{4.3.9}$$

Under these, the metric transforms as

$$\gamma'_{AB} = \frac{\partial x'^C}{\partial x^A} \frac{\partial x'^D}{\partial x^B} \gamma_{CD}, \qquad (4.3.10)$$

so requiring

$$\gamma'_{\mu\nu} = h'_{\mu\nu} - 2\tau'_{(\mu}m'_{\nu)}, \tag{4.3.11}$$

the transformation (4.3.10) implies that under the diffeomorphisms (4.3.8)-(4.3.9)

$$\gamma'_{\mu\nu} = \partial_{\mu} x'^{\rho} \partial_{\nu} x'^{\sigma} \gamma_{\rho\sigma} + 2 \partial_{(\mu} u' \partial_{\nu)} x'^{\sigma} \gamma_{u\sigma}$$
(4.3.12)

$$=h'_{\mu\nu}-2\partial_{\mu}x'^{\rho}\partial_{\nu}x'^{\sigma}\tau_{(\rho}m_{\sigma)}+2\partial_{(\mu}u'\partial_{\nu)}x'^{\sigma}\tau_{\sigma}, \qquad (4.3.13)$$

which, by (4.3.11), implies the transformation properties

$$\tau'_{\mu} = \frac{\partial x'^{\nu}}{\partial x^{\mu}} \tau_{\nu}, \quad m'_{\mu} = \frac{\partial x'^{\nu}}{\partial x^{\mu}} m_{\nu} - \partial_{\mu} \sigma.$$
(4.3.14)

In this sense, the  $m_{\mu}$  vector transforms as a U(1) connection under shifts of u. As we have seen, the ansatz (4.3.1) preserves the following (infinitesimal) local tangent space transformations for the vielbeine

$$\delta \tau_{\mu} = 0, \tag{4.3.15}$$

$$\delta e^a_\mu = \tau_\mu \lambda^a + \lambda^a{}_b e^b_\mu, \tag{4.3.16}$$

$$\delta m_{\mu} = \partial_{\mu} \sigma + \lambda_a e^a_{\mu}. \tag{4.3.17}$$

Similarly, the inverse vielbeine transform according to

$$\delta v^{\mu} = \lambda^a e^{\mu}_a, \qquad (4.3.18)$$

$$\delta e_a^\mu = \lambda_a^{\ b} e_b^\mu \tag{4.3.19}$$

It is occasionally useful to work with the boost invariant vielbeine (see also table 5.2)  $\tau_{\mu}$ ,  $\hat{e}^{a}_{\mu}$  and their inverses  $\hat{v}^{\mu}$ ,  $e^{\mu}_{a}$ , where

$$\hat{e}^{a}_{\mu} = e^{a}_{\mu} - m_{\nu} e^{\nu a} \tau_{\mu}, \qquad (4.3.20)$$

which does not change the orthogonality relations, i.e.

$$\hat{v}^{\mu}\hat{e}^{a}_{\mu}=0, \qquad \hat{v}^{\mu}\tau_{\mu}=-1, \qquad e^{\mu}_{a}\tau_{\mu}=0, \qquad e^{\mu}_{a}\hat{e}^{b}_{\mu}=\delta^{b}_{a}.$$
 (4.3.21)

Further, the following spatial metric will prove useful

$$\hat{h}_{\mu\nu} = \delta_{ab} \hat{e}^{a}_{\mu} \hat{e}^{b}_{\nu} = \bar{h}_{\mu\nu} + 2 \widetilde{\Phi} \tau_{\mu} \tau_{\nu}.$$
(4.3.22)

Note in particular that this analysis does not impose any constraints on  $\tau_{\mu}$ . In order to complete this realization of TNC geometry, we need only introduce the (minimal) connection  $\Gamma^{\rho}_{\mu\nu}$  of (4.2.74). We will encounter null reduction again in chapter 6 in the context of charged Lifshitz holography.

#### 4.4 THE ENERGY-MOMENTUM TENSOR AND FIELD THEORIES ON TNC BACKGROUNDS

In this section, we consider how non-relativistic field theories couple to TNC geometry. There are multiple viable approaches, see e.g. [33, 56]; in what follows we will summarize these results and derive them from null reduction [24, 57]. In this section, we keep the spatial dimension d of the TNC background general. Given some action S describing a field theory coupled to TNC geometry, the

energy-momentum tensor is defined as the response to variations with respect to the TNC fields of (4.2.63)-(4.2.64) and (4.3.22)

$$\delta_{\rm bg}S = \int \mathrm{d}^{d+1}x \ e \left[ -\tau_{\nu}T^{\nu}{}_{\mu}\delta\vartheta^{\mu} + \left(\hat{h}_{\sigma\nu}\vartheta^{\mu}T^{\nu}{}_{\mu}\right)\tau_{\rho}\delta h^{\rho\sigma} + \frac{1}{2}\left(\hat{h}_{\rho\nu}\hat{h}_{\sigma\lambda}h^{\lambda\mu}T^{\nu}{}_{\mu}\right)\delta h^{\rho\sigma} + \tau_{\mu}T^{\mu}\delta\tilde{\Phi} \right].$$
(4.4.1)

Alternatively, the variation with respect to the unhatted TNC complex can be expressed as

$$\delta_{\text{bg}}S = \int \mathrm{d}^{d+1}x \ e \left[ -\mathcal{T}_{\mu}\delta v^{\mu} + \frac{1}{2}\mathcal{T}_{\mu\nu}\delta h^{\mu\nu} + T^{\mu}\delta m_{\mu} \right].$$
(4.4.2)

Using the definitions of the TNC fields (4.2.63)–(4.2.64), the relation between the two sets of responses reads

$$h^{\nu\rho}\mathcal{T}_{\rho\mu} - v^{\nu}\mathcal{T}_{\mu} = T^{\nu}{}_{\mu} + T^{\nu}m_{\mu}.$$
(4.4.3)

It can be shown [24, 57] (see also chapter 6) that the null reduction relates the energy-momentum tensor  $T^{\nu}{}_{\mu}$  and the mass current  $T_{\mu}$  to the higher dimensional energy-momentum tensor  $t^{A}{}_{B}$  in the following manner

$$t^{\mu\nu} = 2\tilde{\Phi}T^{\mu} - \hat{v}^{\sigma}T^{\mu}{}_{\sigma}, \quad t^{\mu\nu} = -\hat{v}^{\mu}T^{\nu} + h^{\mu\rho}T^{\nu}{}_{\rho}.$$
(4.4.4)

Note that  $t^{\mu u}$ —i.e. the response to varying  $\gamma_{uu}$ —is arbitrary since  $\gamma_{uu} = 0$ , so we need not worry about it. We now proceed to exploit the properties of the higher-dimensional energy-momentum tensor in order to extract information about the reduced objects. First off, symmetry of  $t^{\mu\nu}$  implies that  $t^{[\mu\nu]} = 0$ , or

$$-\hat{v}^{\mu}T^{\nu} + h^{\mu\rho}T^{\nu}{}_{\rho} + \hat{v}^{\nu}T^{\mu} - h^{\nu\rho}T^{\mu}{}_{\rho} = 0.$$
(4.4.5)

From this, we obtain what turns out to be the boost and rotation Ward identities,

$$0 = -\hat{h}_{\mu\nu}T^{\mu} + \tau_{\mu}h^{\rho\sigma}\hat{h}_{\nu\sigma}T^{\mu}{}_{\rho}, \quad 0 = \hat{h}_{\mu\rho}\hat{h}_{\nu\lambda}h^{\lambda\sigma}T^{\rho}{}_{\sigma} - (\mu\leftrightarrow\nu), \tag{4.4.6}$$

where the first identity is obtained by contracting (4.4.5) with  $\tau_{\nu}\hat{h}_{\sigma\mu}$  and renaming  $\sigma \leftrightarrow \nu$ , while the second is obtained by contracting (4.4.5) with  $\hat{h}_{\sigma\nu}\hat{h}_{\lambda\mu}$  and renaming  $(\sigma, \lambda) \leftrightarrow (\mu, \nu)$  as well as dummy indices in the resulting expression. Continuing in this manner, we now consider the implications of diffeomorphism invariance and tracelessness, i.e. the relations  $\nabla_A T^A{}_B = 0 = T^A{}_A$ . Lowering the indices of the energy momentum tensor of (4.4.4), one finds the expressions [57]

$$t^{\mu}{}_{\nu} = 2\tilde{\Phi}\tau_{\mu}T^{\mu} - \hat{v}^{\nu}\tau_{\mu}T^{\mu}{}_{\nu}, \quad t^{\mu}{}_{\nu} = 2\tilde{\Phi}\tau_{\mu}T^{\mu}{}_{\nu} - \hat{v}^{\sigma}\hat{h}_{\nu\rho}T^{\rho}{}_{\sigma} + \tau_{\nu}\hat{v}^{\rho}\hat{v}^{\sigma}t_{\rho\sigma}, \quad t^{\mu}{}_{u} = T^{\mu}{}_{\nu}, \quad (4.4.7)$$

as well as,

$$\nabla_A t^A{}_u = \partial_\mu \left( eT^\mu \right), \tag{4.4.8}$$

$$\nabla_A t^A{}_{\mu} = e^{-1} \partial_{\nu} \left( eT^{\nu}{}_{\mu} \right) + T^{\rho}{}_{\nu} \left( \hat{v}^{\nu} \partial_{\mu} \tau_{\rho} - e^{\nu}_a \partial_{\mu} \hat{e}^a_{\rho} \right) + \tau_{\nu} T^{\nu} \partial_{\mu} \tilde{\Phi}, \tag{4.4.9}$$

$$t^{A}{}_{A} = -2\hat{v}^{\nu}\tau_{\mu}T^{\mu}{}_{\nu} + \hat{e}^{a}_{\mu}e^{\nu}_{a}T^{\mu}{}_{\nu} + 2\tilde{\Phi}\tau_{\mu}T^{\mu}.$$
(4.4.10)

These become the U(1), diffeomorphism and z = 2 dilatation Ward identities respectively, something we will be explicit about in chapter 6, where we will put this analysis to good use in the context of charged Lifshitz holography.

We briefly remark that it is possible to write the diffeomorphism Ward identity (4.4.9) in a more TNC covariant form by introducing the *Riemann-Cartan* connection instead of the minimal connection (4.2.74) [57, 121]. This connection is given by

$$\check{\Gamma}^{\lambda}_{\mu\rho} = -\hat{v}^{\lambda}\partial_{\mu}\tau_{\rho} + \frac{1}{2}h^{\nu\lambda}\left(\partial_{\mu}\hat{h}_{\rho\nu} + \partial_{\rho}\hat{h}_{\mu\nu} - \partial_{\nu}\hat{h}_{\mu\rho}\right) + h^{\nu\lambda}\tau_{\rho}K_{\mu\nu},\tag{4.4.11}$$

where  $K_{\mu\nu} = \frac{1}{2} \mathcal{L}_{\hat{\sigma}} \hat{h}_{\mu\nu}$  is the extrinsic curvature. The Riemann-Cartan connection obeys a variant of metric compatibility, namely

$$\check{
abla}_{\mu} au_{
u} = 0, \qquad \check{
abla}_{\mu}\hat{h}_{
u
ho} = 0, \qquad \check{
abla}_{\mu}\hat{\vartheta}^{
u} = 0, \qquad \check{
abla}_{\mu}h^{
u
ho} = 0, \tag{4.4.12}$$

and thus the diffeomorphism Ward identity of (4.4.9) takes the form

$$\nabla_{A}t^{A}{}_{\mu} = \check{\nabla}_{\mu}T^{\mu}{}_{\nu} + 2\check{\Gamma}^{\rho}{}_{[\mu\rho]}T^{\mu}{}_{\nu} - 2\check{\Gamma}^{\mu}{}_{[\nu\rho]}T^{\rho}{}_{\mu} + \tau_{\mu}T^{\mu}\partial_{\nu}\tilde{\Phi}.$$
(4.4.13)

## 4.5 OUTLOOK

In this chapter, we have described TNC geometry and observed how it arises by gauging local algebras. There are many possible generalizations of this, and many of these have been pursued.

One possible extension involves adding supersymmetry: indeed, in [123] the  $\mathcal{N} = 2$  super-Poincaré algebra is contracted to obtain the three-dimensional  $\mathcal{N} = 2$  super-Bargmann algebra which is then subsequently gauged to construct supersymmetric Newton-Cartan geometry, while in [124], this analysis was extended to the super-Schrödinger algebra.

Another interesting development would be to gauge the Schrödinger-Virasoro algebra (see [125]) and see what kind of structure would emerge. There has also been some work on non-relativistic twistor theory using TNC geometry [126–128], and it would be interesting to make a connection to that.

Recently, TNC gravity was discussed in [129] from the perspective of Schrödinger field theories, and it would be interesting to work out the relation to HL gravity, since, as mentioned in the introduction, HL gravity and dynamical TNC geometry are equivalent.

In another direction, TNC geometry provides a natural geometric arena for the study of non-relativistic phenomena in general: as mentioned in the introduction, (T)TNC geometry has been used to model the (fractional) quantum Hall effect in [35–37].

In this chapter, we provide a review of Lifshitz holography for d = 3 as developed in [29, 32].

To whet the reader's appetite, we begin in section 5.1 with a brief exposition of Lifshitz field theory, which is of relevance to condensed matter systems. The analysis follows [130].

The rest of the chapter mainly provides a detailed exposition of the results of [29], incorporating elements from [32, 33, 57]: In section 5.2, we then proceed to develop Lifshitz holography for values of the dynamical exponent  $1 < z \le 2$ . The bulk theory consists of a class of Einstein-Proca-Dilaton (EPD) models, described in section 5.2.1, which admit pure Lifshitz vacuum solutions, around which we then consider perturbations and develop the notion of asymptotically locally Lifshitz (AlLif) geometries in section 5.2.2.

This allows us to identify the sources as the leading components of the bulk fields in a nearboundary expansion. There is a caveat to this identification; in particular, it turns out that the the boundary gauge field  $M_{\mu}$ , which, roughly speaking, is the source from the Proca field *B*, appears as a subleading term in the near-boundary expansion of the time-like bulk vielbein  $E^0$ . We also demonstrate how the Lorentz group contracts to the Galilei group under imposition of the AlLif boundary conditions and show how the sources transform under the contracted Lorentz group.

Having identified the sources of Lifshitz holography, we further investigate their transformation properties in section 5.2.3 under Stückelberg gauge transformations and diffeomorphisms preserving the radial gauge choice of section 5.2.2 (the so called Penrose-Brown-Henneaux (PBH) transformations), which will give rise to boundary diffeomorphisms and dilatations.

In section 5.2.4, we show how TNC geometry as described in chapter 4 emerges in the context of Lifshitz holography, and discuss the relation to the Schrödinger algebra.

We then consider the VEVs and Ward identities of Lifshitz holography in section 5.3, and see how explicit renormalization of the EPD model can be circumvented by assuming the existence of a counterterm. This allows us to determine general properties of the VEVs and is the topic of section 5.3.1.

We remark that for z = 2, it is possible to obtain Lifshitz geometries by means of a Scherk-Schwarz reduction of a five-dimensional z = 0 Schrödinger spacetime (which is an example of an AlAdS spacetime), which makes it possible to study Lifshitz holography by using properties of the more familiar AdS/CFT correspondence. This analysis is performed in [23, 24] (building on previous work [107]), and we will have more to say about this approach in chapter 6. In particular, via the reduction procedure, it is possible to obtain an explicit counterterm action from knowledge of the appropriate AdS counterterm, which is hugely advantageous.

More generally, renormalization of the EPD model AlLif boundary conditions was considered in [28]. Note that it is possible to consider Lifshitz holography starting from an Einstein-Proca model in the bulk [27, 131], and we consider the explicit renormalization of such a model in appendix G following [27, 96].

Note also that not much is known about the boundary theory; even in the case z = 2, where the boundary theory is a null reduction of  $\mathcal{N} = 4$  SYM with a  $\theta$ -term [24], explicit results are lacking.

In section 5.3.2, we define the Hollands-Ishibashi-Marolf (HIM) boundary stress tensor for our holographic theory. In particular, since energy and mass are no longer equivalent, the analogue of the familiar energy-momentum tensor in general relativity now consists of two distinct objects: the HIM stress tensor and the mass current.

Finally, in section 5.3.3, we derive the Ward identities for Lifshitz holography and covariantize them.

### 5.1 THE QUANTUM LIFSHITZ MODEL

Introduced in [130], the simplest Lifshitz invariant scalar field theory in 2 + 1 dimensions is known as the *quantum Lifshitz model* and is governed by the action

$$S = \frac{1}{2} \int dt d^2 x \left( (\partial_t \varphi)^2 - \kappa^2 (\partial_i \partial^i \varphi)^2 \right), \qquad (5.1.1)$$

which is invariant under global Lifshitz rescalings with z = 2. The corresponding Hamiltonian is given by

$$H = \frac{1}{2} \int d^2 x \ \overbrace{\left[\Pi^2 + \kappa^2\right]}^{=:\mathcal{H}},\tag{5.1.2}$$

where  $\Pi = \dot{\phi}$ . The Hamiltonian (5.1.2) furnishes a field theoretic description of *Lifshitz points*, arising in e.g. the description smectic liquid crystals [132]. This model has an intriguing relation to regular two-dimensional conformal field theory due to *detailed balance* [130, 133]. This is essentially due to the fact that the Hamiltonian (5.1.2) can be written in the form

$$\mathcal{H} = Q^{\dagger}Q, \qquad Q = \frac{1}{\sqrt{2}} \left(\kappa \partial^{i}\partial_{i}\varphi - i\Pi\right),$$
(5.1.3)

where  $\Pi$  is the momentum operator conjugate to  $\varphi$ , which in the Schrödinger picture is the functional derivative  $\Pi = -i\frac{\delta}{\delta\varphi}$ . Hamiltonians<sup>1</sup> satisfying detailed balance (i.e. a condition of the form (5.1.3)) have eigenvalues satisfying  $E \ge 0$ , implying that if we can find a state  $|0\rangle$  annihilated by Q(x) for all x, then it is necessarily a ground state; that is to say, the corresponding ground state wave functional  $\Psi_0[\varphi] = \langle [\varphi] | 0 \rangle$  satisfies  $Q\Psi_0 = 0$  for all x, which is nothing but a first-order functional differential equation,  $\left(\frac{\delta}{\delta\varphi} + \kappa \partial_i \partial^i \varphi\right) \Psi_0[\varphi] = 0$ , which is solved by

$$\Psi_{0}[\varphi] = \frac{1}{\sqrt{Z}} \exp\left[-\frac{\kappa}{2} \int d^{2}x \,\partial_{i}\varphi \partial^{i}\varphi\right], \qquad \mathcal{Z} = \int \mathfrak{D}[\varphi] \exp\left[-\kappa \int d^{2}x \,\partial_{i}\varphi \partial^{i}\varphi\right]. \tag{5.1.4}$$

The probability of finding the ground state in the field configuration  $|[\varphi]\rangle$  is given by the absolute square of the overlap,

$$|\Psi_0[\varphi]|^2 = \frac{1}{\mathcal{Z}} \exp\left[-\kappa \int d^2 x \ \partial_i \varphi \partial^i \varphi\right], \qquad (5.1.5)$$

from which it follows that VEVs of local operators  $\mathcal{O}[\varphi(x)]$  are given by

$$\langle 0|\mathcal{O}[\varphi(x_1)]\dots\mathcal{O}[\varphi(x_n)]|0\rangle = \frac{1}{\mathcal{Z}}\int \mathfrak{D}[\varphi]\mathcal{O}[\varphi(x_1)]\dots\mathcal{O}[\varphi(x_n)]\exp\left[-\kappa\int d^2x\;\partial_i\varphi\partial^i\varphi\right],$$
(5.1.6)

and so we infer that the quantum Lifshitz model is in fact equivalent to a two-dimensional theory of a free scalar, which is a CFT. This intriguing connection has been used in [135] to compute the ground state entanglement entropy of the quantum Lifshitz model (see also [136] for more recent results in that direction). Interestingly, supersymmetric Lifshitz field theories have also been studied, see [137].

## 5.2 LIFSHITZ HOLOGRAPHY

In this section, we describe general-z Lifshitz holography as developed in [29, 32, 33, 57].

#### 5.2.1 The EPD Model

The bulk theory of our holographic setup is comprised by the Einstein-Proca-dilaton (EPD) model<sup>2</sup>, which is a fully relativistic theory. The class of four-dimensional EPD theories are described by the following family of actions,

$$S = \int d^4x \,\sqrt{-g} \left( R - \frac{1}{4} Z(\Phi) F^2 - \frac{1}{2} W(\Phi) B^2 - \frac{k}{2} (\partial \Phi)^2 - V(\Phi) \right), \tag{5.2.1}$$

where *k* is a convenient parameter (it has value three for the model we consider in chapter 6, for example), and F = dB for the Proca field  $B_M$ , where capital Roman indices  $M = (r, \mu)$  are used for four-dimensional bulk, and Greek indices  $\mu$  for the boundary. The functions  $Z(\Phi)$  and  $W(\Phi)$  are

<sup>1</sup> An example of another Hamiltonian of this form is the Rokshar-Kivelson Hamiltonian for the quantum dimer model, see [134].

<sup>2</sup> For the Einstein-Proca models that we consider in appendix G, black brane solutions are not known analytically—these were instead studied numerically in [138, 139].

arbitrary positive functions, while the dilatonic potential is negative close to a Lifshitz solution. The equations of motion take the form:

$$G_{MN} = \frac{k}{2} \left( \partial_M \Phi \partial_N \Phi - \frac{1}{2} (\partial \Phi)^2 g_{MN} \right) - \frac{1}{2} V(\Phi) g_{MN}$$
(5.2.2)  
+  $\frac{1}{2} Z(\Phi) \left( F_{MP} F_N^{\ P} - \frac{1}{4} F^2 g_{MN} \right) + \frac{1}{2} W(\Phi) \left( B_M B_N - \frac{1}{2} B^2 g_{MN} \right),$ (5.2.3)

$$\frac{k}{\sqrt{-g}}\partial_M\left(\sqrt{-g}\partial^M\Phi\right) = \frac{1}{4}\frac{\mathrm{d}Z}{\mathrm{d}\Phi}F^2 + \frac{1}{2}\frac{\mathrm{d}W}{\mathrm{d}\Phi}B^2 + \frac{\mathrm{d}V}{\mathrm{d}\Phi},\tag{5.2.4}$$

$$\frac{k}{\sqrt{-g}}\partial_M\left(\sqrt{-g}Z(\Phi)F^{MN}\right) = W(\Phi)B^N.$$
(5.2.5)

To determine the first equation of motion, it is most convenient to vary the action using  $\delta\sqrt{-g} = -\frac{1}{2}\sqrt{-g}g_{MN}\delta g^{MN}$  and  $\delta(\sqrt{-g}R) = G_{MN}\delta g^{MN}$ , while the other two equations of motion are conveniently obtained by using the Euler-Lagrange equations, e.g. for the scalar,

$$\frac{\partial \mathcal{L}}{\partial \Phi} = \sqrt{-g} \left( \frac{1}{4} \frac{\mathrm{d}Z}{\mathrm{d}\Phi} F^2 + \frac{1}{2} \frac{\mathrm{d}W}{\mathrm{d}\Phi} B^2 + \frac{\mathrm{d}V}{\mathrm{d}\Phi} \right) = \partial_M \frac{\partial \mathcal{L}}{\partial (\partial_M \Phi)} = k \partial_M \left( \sqrt{-g} \partial^M \Phi \right), \qquad (5.2.6)$$

which, upon division by  $\sqrt{-g}$  gives the postulated equation of motion. The model (5.2.1) has a broken U(1) symmetry due to the mass term—this, however, can be remedied by making the following Stückelberg decomposition (see [140] for a review of the formalism) of the massive vector,

$$B_M = A_M - \partial_M \Xi. \tag{5.2.7}$$

Note that  $\Xi$  has dimensions of length, while all other fields are dimensionless. The EPD model admits Lifshitz solutions for z > 1; these take the form

$$ds^{2} = -\frac{1}{r^{2z}}dt^{2} + \frac{1}{r^{2}}\left(dr^{2} + dx^{2} + dy^{2}\right),$$
(5.2.8)

$$B = \frac{A_0}{r^2} \mathrm{d}t,\tag{5.2.9}$$

$$\Phi = \Phi_{\star} \qquad (\Phi_{\star} \text{ const.}), \tag{5.2.10}$$

as long as

$$A_0^2 = \frac{2(z-1)}{zZ_0}, \quad \frac{W_0}{Z_0} = 2z, \quad V_0 = -(z^2 + z + 4), \quad V_1 = (za + 2b)(z-1), \tag{5.2.11}$$

where

$$a = \frac{Z_1}{Z_0}, \ b = \frac{W_1}{W_0},$$
 (5.2.12)

where we have defined

$$V_0 := V(\Phi_\star), \ V_1 := V'(\Phi_\star), \ V_2 := V''(\Phi_\star).$$
 (5.2.13)

Note that the last equation in (5.2.11) is a condition on the potential making sure that Lifshitz is a solution to the family of actions (5.2.1). For the details of this analysis, we refer to [32].

#### 5.2.2 AlLif Boundary Conditions & the Sources

The spatial anisotropy makes the use of vielbeine convenient [27], so we define asymptotically locally Lifshitz (AlLif) space-times by specifying boundary conditions for the vielbeine. Defining a holographic radial coordinate r by requiring the metric to be asymptotically (conformally) radial, i.e.

$$ds^{2} = \frac{\mathrm{d}r^{2}}{Rr^{2}} - E^{0}E^{0} + \delta_{ab}E^{a}E^{b}, \qquad (5.2.14)$$

where  $E_r^0 = 0 = E_r^a$ , and where the function *R* is analogous to a radial lapse, which is sometimes [27] set to one—this is known as radial gauge. The benefit of leaving *R* unfixed was realized in [23]. The

boundary is located at r = 0. The vielbeine are related to the induced metric  $\gamma_{\mu\nu}$  on slices of constant r via

$$\gamma_{\mu\nu} dx^{\mu} dx^{\nu} = -E^0 E^0 + \delta_{ab} E^a E^b, \qquad (5.2.15)$$

and the boundary conditions for AlLif read

$$R = \mathcal{O}(1), \quad E^{0}_{\mu} = \mathcal{O}(r^{-z}), \quad E^{a}_{\mu} = \mathcal{O}(r^{-1}), \quad (z > 1).$$
(5.2.16)

These boundary conditions imply that

$$\sqrt{-g} = \mathcal{O}(r^{-z-3}).$$
 (5.2.17)

For the Proca field,  $B = B_r dr + B_\mu dx^\mu$ , we have that (as was the case for the pure Lifshitz solution)

$$B_{\mu} = \mathcal{O}(r^{-z}),$$
 (5.2.18)

which can be obtained from an asymptotic analysis of the equations of motion. Now, due to antisymmetry of the field strength tensor, the equation of motion for the Proca field (5.2.5) implies that

$$0 = \partial_N \left( \sqrt{-g} W(\Phi) B^N \right) = \partial_r \left( \sqrt{-g} W(\Phi) B^r \right) + \partial_\mu \left( \sqrt{-g} W(\Phi) B^\mu \right), \tag{5.2.19}$$

which is a differential equation for  $B_r$  in terms of  $B_\mu$ —using  $B_r = g^{rr}B_r = Rr^2B_r$ , it takes the form

$$0 = \partial_r \left( \sqrt{-g} W(\Phi) R r^2 B_r \right) + \partial_\mu \left( \sqrt{-g} W(\Phi) B^\mu \right), \qquad (5.2.20)$$

which we can readily integrate to obtain (remembering our freedom to add an *r*-independent function -f(x)),

$$0 = \sqrt{-g}W(\Phi)Rr^{2}B_{r} - f(x) + \int_{0}^{r} dr'\partial_{\mu} \left(\sqrt{-g}W(\Phi)B^{\mu}\right)$$
(5.2.21)

$$\therefore B_r = \frac{f(x)}{\sqrt{-gW(\Phi)Rr^2}} - \frac{1}{\sqrt{-gW(\Phi)Rr^2}} \int_0^r \mathrm{d}r' \partial_\mu \left(\sqrt{-g}W(\Phi)B^\mu\right).$$
(5.2.22)

In particular, the extra term becomes important in a radial expansion at order  $r^{z+1}$ , and thus it does not affect the leading behaviour. We may express the boundary conditions for  $B_{\mu}$  and  $E_{\mu}^{0}$ —eqs. (5.2.18) and (5.2.16), respectively—by inferring the existence of a function  $\alpha(x, r)$  such that,

$$B_{\mu} - \alpha E_{\mu}^{0} = o(r^{-z}), \qquad (5.2.23)$$

with  $\alpha = O(1)$  near r = 0, and where the Landau symbol  $o(r^{-z})$  means<sup>3</sup> that whatever is on the LHS grows strictly slower than  $r^{-z}$ . The rôle of the function  $\alpha$  will be made clear later. The boundary condition on the dilaton reads

$$\Phi \simeq r^{\Delta} \phi, \qquad \Delta \ge 0,$$
 (5.2.24)

where  $\simeq$  refers to the leading order term in the near-boundary expansion, and  $\phi(x)$  is the boundary value of the dilaton. In particular, in order to find  $\Delta$ , we need to solve the equations of motion (specifically, by considering radial perturbations around the Lifshitz solution (5.2.8)), which requires us to expand the dilation around  $\Phi = \Phi_{\star}$ , and this depends on whether  $\Delta > 0$  or  $\Delta = 0$ . Without loss of generality, we can shift  $\Phi$  and set  $\Phi_{\star} = 0$ , which we will do in what follows. In terms of the quantities introduced above, we impose the following boundary conditions for the vielbeine,

$$E^{0}_{\mu} \simeq r^{-z} \alpha^{1/3}_{(0)} \tau_{\mu}, \quad E^{a}_{\mu} \simeq r^{-1} \alpha^{-1/3}_{(0)} e^{a}_{\mu}, \quad R \simeq R_{(0)},$$
(5.2.25)

where  $\alpha_{(0)}$  and  $R_{(0)}$  are the leading terms of  $\alpha$  and R, respectively, which are fixed by the equations of motion *as long as we are in the range*  $1 < z \leq 2$  (see [32] for details), so from now on we specialize to this case. Note also that we do not demand (by hand) that  $\tau_{\mu}$  be hypersurface orthogonal (HSO)<sup>4</sup>,

<sup>3</sup> More precisely, f(x) = o(g(x)) means that for all  $\epsilon > 0$  there exists an  $x_0 \in \mathbb{R}$  such that  $f(x) < \epsilon g(x)$  for all  $x > x_0$ . Similarly, if  $f(x) = \mathcal{O}(g(x))$  means that there exists a  $\epsilon > 0$  for which you can find an  $x_0 \in \mathbb{R}$  such that  $f(x) \le g(x)$  for all  $x > x_0$ . Since both o(g(x)) and  $\mathcal{O}(g(x))$  are more appropriately viewed as sets, we should strictly speaking write e.g.  $f(x) \in o(g(x))$ , but we'll stick with the notation f(x) = o(g(x)).

<sup>4</sup> That is, orthogonal to leaves of constant time.

 $\tau \wedge d\tau = 0$ ; rather we let the equations of motion determine it. It turns out that for z > 2,  $\tau_{\mu}$  is always HSO [29]. In particular, this the HSO condition be rewritten as the vanishing of the *twist* of  $\tau$ ,  $\omega^2 = 0$ , where

$$\omega^2 = \frac{1}{2} \left( \varepsilon^{\mu\nu\rho} \tau_{\mu} \partial_{\nu} \tau_{\rho} \right)^2.$$
 (5.2.26)

In general, one finds that the solution splits up into four branches (details will be given in [32]), namely

$$(i): 1 < z < 2 \text{ and } \Delta > 0, \quad (ii): 1 < z < 2 \text{ and } \Delta = 0,$$
 (5.2.27)

$$(iii): z = 2 \text{ and } \Delta > 0, \quad (iv): z = 2 \text{ and } \Delta = 0.$$
 (5.2.28)

It turns out that case (*iv*) splits up into two additional branches: either  $\tau$  is HSO, in which case  $W = 4Z^{2/3}$ . If this is not satisfied, the twist (5.2.26) must instead be given by  $\omega^2 = -2(Z(\phi))^{2/3} + \frac{1}{2}W(\phi)$ , which becomes a constraint on the source  $\phi$ .

We will take  $\alpha$  and *R* to depend on  $\Phi$ , so in particular, the leading terms will depend on  $\Delta$ . Inverting (5.2.25), the inverse vielbeine are subject to the boundary conditions,

$$E_0^{\mu} \simeq -r^z \alpha_{(0)}^{-1/3} v^{\mu}, \quad E_a^{\mu} \simeq r \alpha_{(0)}^{1/3} e_a^{\mu},$$
 (5.2.29)

but since

$$1 = \delta_0^0 = E_0^{\mu} E_{\mu}^0 = -v^{\mu} \tau_{\mu}, \quad 0 = \delta_0^a = E_0^{\mu} E_{\mu}^a = -r^{z-1} \alpha_{(0)}^{-2/3} v^{\mu} e_{\mu}^a, \tag{5.2.30}$$

$$0 = \delta_a^0 = E_a^{\mu} E_{\mu}^0 = r^{1-z} \alpha_{(0)}^{2/3} e_a^{\mu} \tau_{\mu}, \quad \delta_b^a = E_b^{\mu} E_{\mu}^a = e_b^{\mu} e_{\mu}^a, \tag{5.2.31}$$

we obtain the orthogonality relations,

$$v^{\mu}\tau_{\mu} = -1, \quad v^{\mu}e^{a}_{\mu} = 0, \quad e^{\mu}_{a}\tau_{\mu} = 0, \quad e^{\mu}_{b}e^{a}_{\mu} = \delta^{a}_{b}.$$
 (5.2.32)

Further, the relation

$$\delta^{\mu}_{\nu} = E^{0}_{\nu}E^{\mu}_{0} + E^{a}_{\nu}E^{\mu}_{a} = -v^{\mu}\tau_{\nu} + e^{\mu}_{a}e^{a}_{\nu}, \qquad (5.2.33)$$

implies the important completeness relation

$$e_a^{\mu}e_{\nu}^{a} = \delta_{\nu}^{\mu} + v^{\mu}\tau_{\nu}. \tag{5.2.34}$$

Following the arguments given in [24], we now scrutinize the behaviour of the boundary vielbeine under local Lorentz transformations (LLTs). Since z > 1, the boundary conditions (5.2.16) imply that the time-like vielbein diverges faster than the space-like vielbeine, which means that the local light cones flatten out, and so the Lorentz group contracts to the Galilei group; in other words:  $r \rightarrow \infty$ corresponds to sending the speed of light to infinity, which, as we have seen, contracts the Lorentz group to the Galilean group. To this end, we start by considering the transformation properties of the bulk vielbeine under LLTs; specifically we consider SO(2, 1) transformations leaving the radial direction invariant. For such transformations, we may write

$$E^{0}_{\mu} = \Lambda^{0}{}_{0'}E^{0'}_{\mu} + \Lambda^{0}{}_{a'}E^{a'}_{\mu}, \qquad (5.2.35)$$

$$E^{a}_{\mu} = \Lambda^{a}_{\ 0'} E^{0'}_{\mu} + \Lambda^{a}_{\ a'} E^{a'}_{\mu}.$$
(5.2.36)

Now, for generic frame indices  $(\underline{a}, \underline{b}, \underline{c}, \underline{d})$ —which includes time—the defining property for Lorentz transformations,  $\eta_{\underline{a}\underline{b}} \Lambda^{\underline{a}}{}_{\underline{c}} \Lambda^{\underline{b}}{}_{\underline{d}} = \eta_{\underline{c}\underline{d}}$ , leads to

$$-\Lambda^{0}{}_{0'}\Lambda^{0}{}_{0'} + \delta_{ab}\Lambda^{a}{}_{0'}\Lambda^{b}{}_{0'} = -1, \qquad (5.2.37)$$

$$-\Lambda^{0}{}_{0'}\Lambda^{0}{}_{a'} + \delta_{ab}\Lambda^{a}{}_{0'}\Lambda^{b}{}_{a'} = 0, (5.2.38)$$

$$-\Lambda^{0}{}_{a'}\Lambda^{0}{}_{b'} + \delta_{ab}\Lambda^{a}{}_{a'}\Lambda^{b}{}_{b'} = \delta_{a'b'}.$$
(5.2.39)

The Lorentz transformed boundary conditions take the form,

$$E_{\mu}^{0'} \simeq r^{-z} \alpha_{(0)}^{1/3} \tau_{\mu'}', \qquad E_{\mu}^{a'} \simeq r^{-1} \alpha_{(0)}^{-1/3} e_{\mu}^{a'},$$
 (5.2.40)

since neither *r* nor  $\alpha_{(0)}$  transforms. By comparing powers of *r*, the above together with the transformations (5.2.35)–(5.2.36) imply that

$$\Lambda^{0}{}_{0'} \simeq \Lambda^{0}_{(0)0'}, \quad \Lambda^{0}{}_{a'} \simeq r^{1-z} \alpha^{2/3}_{(0)} \Lambda^{0}_{(0)a'}, \quad \Lambda^{a}{}_{0'} \simeq r^{z-1} \alpha^{-2/3}_{(0)} \Lambda^{a}_{(0)0'}, \quad \Lambda^{a}{}_{a'} \simeq \Lambda^{a}_{(0)a'}, \quad (5.2.41)$$

but the constraints (5.2.37)–(5.2.39) imply that, for example

$$-\Lambda^{0}_{(0)0'}\Lambda^{0}_{(0)0'} + \delta_{ab}r^{2(z-1)}\Lambda^{a}_{(0)0'}\Lambda^{b}_{(0)0'} = -1.$$
(5.2.42)

Since z > 1, we can discard the second term altogether, which makes  $\Lambda^a_{(0)0'}$  a set of two free parameters—implying that

$$\Lambda^{0}_{(0)0'}\Lambda^{0}_{(0)0'} = 1, \tag{5.2.43}$$

prompting us to choose

$$\Lambda^0_{(0)0'} = 1, \tag{5.2.44}$$

so that we may recover the identity. Similar considerations lead us to conclude that

$$\Lambda^0_{(0)a'} = 0, \tag{5.2.45}$$

$$\delta_{ab}\Lambda^{a}_{(0)a'}\Lambda^{b}_{(0)b'} = \delta_{a'b'}.$$
(5.2.46)

These findings agree with those obtained in [24] when setting z = 2. This, in conjunction with the transformation of the time-like bulk vielbein (5.2.35), means that

$$\tau_{\mu} = \tau_{\mu}^{\prime}, \tag{5.2.47}$$

while the transformation of the spatial bulk vielbein (5.2.36) implies that

$$e^{a}_{\mu} = \Lambda^{a}_{(0)0'} e^{0'}_{\mu} + \Lambda^{a}_{(0)a'} e^{a'}_{\mu}.$$
(5.2.48)

The transformations  $\Lambda^a_{(0)0'}$  and  $\Lambda^a_{(0)b}$  correspond to local Galilean boosts and local rotations, respectively (and will collectively be denoted as local Galilean transformations: LGTs). In order to obtain the infinitesimal versions of these local tangent space transformations, we define

$$\Lambda^{a}_{(0)0'} = \varepsilon \lambda^{a} + \mathcal{O}(\varepsilon^{2}), \qquad (5.2.49)$$

$$\Lambda^a_{(0)b} = \delta^a_b + \varepsilon \lambda^a{}_b + \mathcal{O}(\varepsilon^2).$$
(5.2.50)

With these, we can summarize our findings above as

$$\delta \tau_{\mu} = 0, \tag{5.2.51}$$

$$\delta e^a_\mu = \lambda^a \tau_\mu + \lambda^a_{\ b} e^b_\mu. \tag{5.2.52}$$

where we have used  $\tau_{\mu} = \tau'_{\mu}$  in the second line. To determine the transformation properties of the inverse vielbeine, we use the orthogonality relations (5.2.32), i.e.

$$0 = \delta(v^{\mu}e^{a}_{\mu}) = e^{a}_{\mu}\delta v^{\mu} + v^{\mu}\delta e^{a}_{\mu} = e^{a}_{\mu}\delta v^{\mu} + \lambda^{a} \underbrace{v^{\mu}\tau_{\mu}}^{=-1} + \lambda^{a} \underbrace{v^{\mu}e^{b}_{\mu}}^{=0},$$
(5.2.53)

i.e.<sup>5</sup>  $\delta v^{\mu} = \lambda^{a} e^{\mu}_{a}$ , since  $e^{\mu}_{b} e^{a}_{\mu} = \delta^{a}_{b}$ —which we will now vary to get

$$0 = \delta(e_b^{\mu}e_{\mu}^{a}) = e_{\mu}^{a}\delta e_b^{\mu} + e_b^{\mu}\delta e_{\mu}^{a} = e_{\mu}^{a}\delta e_b^{\mu} + \lambda^{a} \underbrace{e_b^{\mu}\tau_{\mu}}_{a} + \underbrace{e_b^{\mu}e_{\mu}^{a}\lambda^{a}}_{b},$$
(5.2.54)

which, by the same reasoning, means that  $\delta e_a^{\mu} = -\lambda_b^b a e_b^{\mu} = \lambda_b^a e_b^{\mu}$ , where we used antisymmetry of the infinitesimal *SO*(2) rotations. To sum up our findings:

$$\delta v^{\mu} = \lambda^a e^{\mu}_a, \qquad (5.2.55)$$

$$\delta e_a^\mu = \lambda_a^{\ b} e_b^\mu. \tag{5.2.56}$$

<sup>5</sup> Note that this is compatible with  $0 = \delta(v^{\mu}\tau_{\mu}) = \tau_{\mu}\delta v^{\mu}$  since  $\tau_{\mu}e_{a}^{\mu} = 0$ .

Now, all terms in the near-boundary expansion of the metric (5.2.14) should be invariant under Galilean transformations when expressed in terms of the boundary vielbeine; in particular, at order  $r^{-2}$ , we get

$$\mathcal{O}(r^{-2}): \alpha_{(0)}^{-2/3} \delta_{ab} e^a_{\nu} e^b_{\mu} + \cdots, \qquad (5.2.57)$$

with the dots denoting the contributions from the expansion of  $E^0_{\mu}E^0_{\nu}$ . Galilean invariance of the whole expression implies Galilean invariance of each order in *r*, but the first term of (5.2.57) is not invariant, since by the transformation (5.2.52)

$$\delta\left(e^{a}_{\mu}e^{b}_{\nu}\right) = \lambda_{a}\tau_{\mu}e^{a}_{\nu} + \lambda_{a}\tau_{\nu}e^{a}_{\mu}.$$
(5.2.58)

From this we infer the existence of a term coming from  $E^0_{\mu}E^0_{\nu}$ , the rôle of which is to make the whole expression invariant—we can engineer such a term by demanding

$$E^{0}_{\mu} = r^{-z} \alpha^{1/3}_{(0)} + \dots + r^{z-2} \alpha^{-1}_{(0)} X_{\mu}, \qquad (5.2.59)$$

and so the complete term at order  $r^{-2}$  is

$$\mathcal{O}(r^{-2}): \alpha_{(0)}^{-2/3} \left( \delta_{ab} e^a_\mu e^b_\mu - \tau_\mu X_\nu - \tau_\nu X_\mu \right),$$
(5.2.60)

where invariance implies that under local Galilean transformations,  $X_{\mu}$  transforms as

$$\delta X_{\mu} = e^a_{\mu} \lambda_a. \tag{5.2.61}$$

Now, since such an object<sup>6</sup> cannot be constructed from the boundary vielbein sources  $\tau_{\mu}$  and  $e_{\mu}^{a}$ , there must exist a boundary vector field  $M_{\mu}$  such that

$$X_{\mu} = M_{\mu} + I_{\mu}, \tag{5.2.62}$$

where  $I_{\mu}$  is invariant under LGTs, and we assume without loss of generality<sup>7</sup> that  $I_{\mu} = I\tau_{\mu}$  for I a scalar. For the time-like bulk vielbein we may consequently write

$$E^{0}_{\mu} = r^{-z} \alpha^{1/3}_{(0)} \tau_{\mu} + \dots + r^{z-2} \alpha^{-1}_{(0)} \left( M_{\mu} + I \tau_{\mu} \right).$$
(5.2.63)

For the massive vector, which is Galilean boost invariant at each order of r, we may write (the choice of coefficients harks back to the relation for the Proca field (5.2.23) and will be justified shortly)

$$B_{\mu} = r^{-z} \alpha_{(0)}^{4/3} \tau_{\mu} + \dots + r^{z-2} \tilde{I} \tau_{\mu} + \dots, \qquad (5.2.64)$$

where I is another scalar. We are now ready to make more explicit the relation (5.2.23): for a function  $\alpha$  of the form

$$\alpha = \alpha_{(0)} + r^{2z-2} \alpha_{(0)} \left( \tilde{I} - I \right) + \dots,$$
(5.2.65)

we get an explicit version of (5.2.23)

$$B_{\mu} - \alpha(\Phi) E_{\mu}^{0} \simeq -r^{z-2} M_{\mu}.$$
 (5.2.66)

Contracting with the bulk vielbeine, we find that the following leading behaviour of the frame components of the Proca field,

$$B_0 = E_0^{\mu} B_{\mu} = \mathcal{O}(1), \qquad B_a = E^{\mu} B_{\mu} = \mathcal{O}(r^{z-1}), \tag{5.2.67}$$

where we have used (5.2.66) as well as the boundary conditions (5.2.16). Combining these results, we find for the Proca field

$$B^{\mu} = E_0^{\mu} B^0 + E_a^{\mu} B^a = \mathcal{O}(r^z), \qquad (5.2.68)$$

<sup>6</sup> That is, an object transforming as in (5.2.61).

<sup>7</sup> That is, without affecting the properties of the new source  $M_{\mu}$ .

and thus, by the relation for  $B_r$  (5.2.22), since  $\frac{1}{r^2\sqrt{-g}} = \mathcal{O}(r^{z+1})$  and  $\int_0^r dr' \partial_\mu (\sqrt{-g}W(\Phi)B^\mu) = \mathcal{O}(r^{-2})$ , we get

$$B_r = \mathcal{O}(r^{z-1}).$$
 (5.2.69)

Now, returning the Stückelberg decomposition of  $B_{\mu}$ , (5.2.7), the expression (5.2.66) allows us to do the same for  $M_{\mu}$ ,

$$M_{\mu} = \tilde{m}_{\mu} - \partial_{\mu}\chi, \qquad (5.2.70)$$

provided that

$$\partial_{\mu}\Xi \simeq -r^{z-2}\partial_{\mu}\chi$$
, or  $\Xi \simeq -r^{z-2}\chi$ . (5.2.71)

This can always be done since the Stückelberg scalar is unphysical and thus we can impose any boundary condition we like. For the radial component of the Proca field, (5.2.69) implies

$$B_r = A_r - \partial_r \Xi = \mathcal{O}(r^{z-1}), \tag{5.2.72}$$

but the scaling of the Stückelberg scalar  $\Xi$  (5.2.71) implies that  $\partial_r \Xi \simeq -(z-2)r^{z-3}\chi$ , so the only way the leading behaviour of  $B_r$  can be  $\mathcal{O}(r^{z-1})$  is if

$$A_r \simeq -(z-2)r^{z-3}\chi, \tag{5.2.73}$$

such that it cancels in (5.2.71). For the other components of *A*, the definition of the source  $M_{\mu}$  (5.2.66) gives us

$$A_{\mu} - \partial_{\mu} \Xi - \alpha(\Phi) E^{0}_{\mu} \simeq -r^{z-2} \left( \tilde{m}_{\mu} - \partial_{\mu} \chi \right), \qquad (5.2.74)$$

$$\therefore A_{\mu} - \alpha(\Phi) E_{\mu}^{0} \simeq -r^{z-2} \tilde{m}_{\mu}, \qquad (5.2.75)$$

where we have used the scaling of the Stückelberg field (5.2.71).

## 5.2.3 More Local Transformations of the Sources

So far, we have seen how the sources behave under local Galilean boosts and rotations. Another local transformation arises as a consequence of working with the Stückelberg field  $\Xi$ : this results in a local U(1) symmetry, which acts on  $A_M$  and  $\Xi$  as

$$\delta A_M = \partial_M \Lambda, \quad \delta \Xi = \Lambda,$$
 (5.2.76)

but in order to preserve the boundary conditions (5.2.73) and (5.2.71), we take

$$\Lambda \simeq -r^{z-2}\sigma. \tag{5.2.77}$$

In turn, by the definitions of  $\chi$  and  $\tilde{m}_{\mu}$  in (5.2.71) and (5.2.75), respectively, this implies that under local U(1) transformations, the sources  $\tilde{m}_{\mu}$  and  $\chi$  transform as

$$\delta \tilde{m}_{\mu} = \partial_{\mu} \sigma, \quad \delta \chi = \sigma. \tag{5.2.78}$$

There is another class of local symmetries, namely bulk diffeomorphisms preserving the conformally radial gauge choice of (5.2.14). These are the PBH transformations, which we now describe. Requiring that  $Rg_{MN}$  remains in radial gauge after acting on with a diffeomorphism amounts to the statement

$$\delta\left(Rg_{rr}\right) = 0,\tag{5.2.79}$$

where

$$\delta\left(Rg_{MN}\right) = \pounds_{\zeta}\left(Rg_{MN}\right),\tag{5.2.80}$$

where  $\zeta$  is the bulk vector generating PBH transformations. The condition (5.2.79) implies that

$$\pounds_{\zeta} \left( Rg_{MN} \right) = \zeta^{M} \partial_{M} g_{rr} + 2 \partial_{r} \zeta^{M} g_{Mr}$$
(5.2.81)

$$= -2\zeta^{r}\frac{1}{r^{3}} + \frac{2}{r^{2}}\partial_{r}\zeta^{r} = 0, \qquad (5.2.82)$$

so solving the resulting differential equation  $\zeta^r = r\partial_r \zeta^r$  gives us  $\zeta^r = -r\Lambda_D$  for some arbitrary function<sup>8</sup>  $\Lambda_D(x)$  of the boundary coordinates<sup>9</sup>. It can be shown [33] that the general behaviour of the generator of PBH transformations  $\zeta^{\mu}$  is  $\zeta^{\mu} = \zeta^{\mu} + O(r^2)$  by considering non-constant  $\Lambda_D$ . Summarizing our results, we have found that

$$\zeta^r = -r\Lambda_D, \quad \zeta^\mu = \xi^\mu + \mathcal{O}(r^2). \tag{5.2.83}$$

The PBH diffeomorphisms generated by  $\zeta^M$  act on all bulk fields, and by applying the result (5.2.83), we are able to obtain the transformation properties of the sources. Acting on  $E^0_{\mu}$ , for example, with PBH diffeomorphisms gives us

$$\pounds_{\zeta} E^0_{\mu} = \zeta^M \partial_M E^0_{\mu} + E^0_{\nu} \partial_{\mu} \zeta^{\nu} = \zeta^r \partial_r E^0_{\mu} + \pounds_{\xi} E^0_{\mu}$$
(5.2.84)

$$= -r\Lambda_D \partial_r E^0_\mu + \pounds_{\xi} E^0_\mu \tag{5.2.85}$$

$$= z\Lambda_D E^0_{\mu} + \pounds_{\xi} E^0_{\mu}, \tag{5.2.86}$$

where we note how the sign and the factor of r works out nicely in the last equality and makes explicit how  $\Lambda_D$  corresponds to dilatations. The complete behaviour of  $\tau_{\mu}$  under local transformations consequently works out to be:

$$\delta \tau_{\mu} = \pounds_{\xi} \tau_{\mu} + z \Lambda_D \tau_{\mu}. \tag{5.2.87}$$

Repeating this exercise for all the other sources, we obtain

$$\delta e^a_\mu = \pounds_{\xi} e^a_\mu + \lambda^a \tau_\mu + \lambda^a{}_b e^b_\mu + \Lambda_D e^a_\mu.$$
(5.2.88)

$$\delta M_{\mu} = \pounds_{\xi} M_{\mu} + e^a_{\mu} \lambda_a + (2 - z) \Lambda_D M_{\mu}, \qquad (5.2.89)$$

$$\delta\chi = \pounds_{\xi}\chi + \sigma + (2 - z)\Lambda_D\chi, \tag{5.2.90}$$

$$\delta v^{\mu} = \pounds_{\xi} v^{\mu} + \lambda^a e^{\mu}_a - z \Lambda_D v^{\mu}, \qquad (5.2.91)$$

$$\delta e_a^{\mu} = \pounds_{\xi} e_a^{\mu} + \lambda_a^{\ b} e_b^{\mu} - \Lambda_D, \tag{5.2.92}$$

$$\delta M_a = \pounds_{\xi} M_a + \lambda_a^{\ b} M_b + \lambda_a + (1-z)\Lambda_D M_a, \tag{5.2.93}$$

$$\delta\phi = \pounds_{\xi}\phi - \Delta\Lambda_D\phi, \tag{5.2.94}$$

where  $M_a = e_a^{\mu} M_{\mu}$ . In the above,  $\lambda^a$  corresponds to Galilean boosts (occasionally known as Milne boosts, see e.g. [141]) (*G*),  $\lambda_a^{\ b}$  to spatial rotations (*J*),  $\Lambda_D$  to dilatations (*D*) and  $\sigma$ , at last, corresponds to Stückelberg gauge transformations (*N*). The transformations (6.3.29)–(6.3.34) are similar to (4.2.25)–(4.2.26) and (4.2.60), (4.2.61), except they have some additional structure. This is due to the fact that *G*, *J*, *D*, *N* realize a Schrödinger algebra (see [34]), which is a conformal extension the Bargmann algebra that we discussed in chapter 4. The transformation of  $\tilde{m}_{\mu}$  will also be required and is found by using the definition  $\tilde{m}_{\mu} = M_{\mu} + \partial_{\mu}\chi$ :

$$\delta \tilde{m}_{\mu} = \delta M_{\mu} + \partial_{\mu} \delta \chi$$

$$= \pounds_{\xi} (\tilde{m}_{\mu} - \partial_{\mu} \chi) + e^{a}_{\mu} \lambda_{a} + (2 - z) \Lambda_{D} (\tilde{m}_{\mu} - \partial_{\mu} \chi) + \partial_{\mu} (\pounds_{\xi} \chi) + \partial_{\mu} \sigma + (2 - z) \left[ \Lambda_{D} \partial_{\mu} \chi + \chi \partial_{\mu} \Lambda_{D} \right]$$
(5.2.95)
(5.2.96)

$$= \pounds_{\xi} \tilde{m}_{\mu} + e^a_{\mu} \lambda_a + (z-2)\Lambda_D \tilde{m}_{\mu} + \partial_{\mu} \sigma + (2-z)\chi \partial_{\mu} \Lambda_D.$$
(5.2.97)

We can collect our findings regarding sources and their scaling dimensions (dilatation weights w, which—due to the sign in (5.2.83)—is defined via  $\Lambda_D X = -wX$ ) in the table below:

Table 5.1: Sources and their scaling dimensions. Note that the sets  $(\tau_{\mu}, e^{a}_{\mu})$  and  $(v^{\mu}, e^{\mu}_{a})$  are not independent, one must choose to work with one of them.

source	φ	$ au_{\mu}$	$e^a_\mu$	$v^{\mu}$	$e^{\mu}_{a}$	$\tilde{m}_0$	т <sub>а</sub>	χ
scaling dim.	Δ	-z	-1	z	1	2z - 2	z-1	z-2

When counting the number of components of the sources, we must choose to work with either  $(\tau_{\mu}, e_{\mu}^{a})$  or  $(v^{\mu}, e_{a}^{\mu})$ . Choosing one of these, we count a total of 14 components<sup>10</sup>. The local transformations of the sources provide additional constraints reducing the number of free sources. The total

<sup>8</sup> Which we will later identify to correspond to dilatations.

<sup>9</sup> Note that the sign in the identification  $\zeta^r = -r\Lambda_D$  is arbitrary. Refs. [29, 33] use—as we do—the convention of (5.2.83), while ref. [32] uses a plus. It does of course not change anything.

<sup>10</sup> The specific counting is  $\phi : 1, \tau_{\mu} : 3, e_{\mu}^{a} : 6, \tilde{m}_{0} : 1, \tilde{m}_{a} : 2, \chi : 1$ , which adds up to 14.

number of symmetry parameters is<sup>11</sup> 8, resulting in a total of 14 - 8 = 6 free sources; at least as long as we're in the range 1 < z < 2. For z = 2, there is an additional constraint (as described just below (5.2.28)), which consequently reduces the number of free sources to 5. Before turning to the identification of the VEVs corresponding to the sources of table 5.1, we discuss the boundary geometry of our holographic setup.

# 5.2.4 Emergence of TNC Geometry & Schrödinger Symmetry

In this section, we demonstrate the emergence of TNC geometry of chapter 4 as the boundary geometry, following [29, 33, 34].

From the sources we derived in section 5.2.2, we can construct a series of boost invariant objects<sup>12</sup>,

$$\hat{v}^{\mu} = v^{\mu} - h^{\mu\nu}M_{\nu}, \quad \hat{e}^{a}_{\mu} = e^{a}_{\mu} - M_{\nu}e^{\nu a}\tau_{\mu}, \quad \tilde{\Phi} = -v^{\mu}M_{\mu} + \frac{1}{2}h^{\mu\nu}M_{\mu}M_{\nu}, \quad (5.2.98)$$

$$h^{\mu\nu} = \delta^{ab} e^{\mu}_{a} e^{\nu}_{b}, \quad \bar{h}_{\mu\nu} = \delta_{ab} e^{a}_{\mu} e^{b}_{\nu} - \tau_{\mu} M_{\nu} - \tau_{\nu} M_{\mu}.$$
(5.2.99)

These are entirely equivalent to (4.2.63)–(4.2.64). These objects are all invariant under (*G*, *N*) and nearly all of them are also invariant under *J*; for example, the (*G*, *J*, *N*)-transformation<sup>13</sup> of  $\vartheta^{\mu}$  reads,

$$\delta \hat{v}^{\mu} = \delta v^{\mu} - \delta^{ab} \left( M_b \delta e^{\mu}_a + e^{\mu}_a \delta M_b \right)$$
(5.2.101)

$$=\lambda^{a}e^{\mu}_{a}-\delta^{ab}\left(M_{b}\lambda_{a}{}^{c}e^{\mu}_{c}+e^{\mu}_{a}\left(\lambda_{b}{}^{c}M_{c}+\lambda_{b}\right)\right)$$
(5.2.102)

$$= M^a \lambda_a^{\ c} e^\mu_c + e^\mu_a \lambda^{ac} M_c \tag{5.2.103}$$

$$= M_a \lambda^{ac} e_c^{\mu} + e_a^{\mu} \lambda^{ac} M_c \tag{5.2.104}$$

$$=0,$$
 (5.2.105)

where we have used that infinitesimal rotations are anti-symmetric<sup>14</sup>,

$$\lambda^{ab} = -\lambda^{ba}.\tag{5.2.106}$$

By these invariant objects, we mean quantities of specific dilatation weight invariant under certain symmetries.

Table 5.2: Newton-Cartan complexes and associated invariance. If a quantity is invariant under e.g. *G* transformations, this will be marked with a " $\checkmark$ " in the corresponding column, while a quantity not invariant under a given transformation will have a "X" in the corresponding column.

Inv. Obj.	G	J	Ν	Scaling Dim.	
$\hat{v}^{\mu}$	$\checkmark$	$\checkmark$	$\checkmark$	Z	
$\hat{e}^a_\mu$	$\checkmark$	Χ	$\checkmark$	-1	
$ ilde{\Phi}$	$\checkmark$	$\checkmark$	$\checkmark$	2z - 2	
$ au_{\mu}$	$\checkmark$	$\checkmark$	$\checkmark$	-z	
$h^{\mu u}$	$\checkmark$	$\checkmark$	$\checkmark$	2	
$h^{\mu u}$	$\checkmark$	$\checkmark$	$\checkmark$	2	
$\overline{h}_{\mu u}$	$\checkmark$	$\checkmark$	$\checkmark$	-2	

$$\delta v^{\mu} = \lambda^{a} e^{\mu}_{a}, \quad \delta M_{a} = \lambda_{a}^{\ b} M_{b} + \lambda_{a}, \quad \delta e^{\mu}_{a} = \lambda_{a}^{\ b} e^{\mu}_{b}. \tag{5.2.100}$$

14 This follows from the defining property,  $R^T R = 1$ . An infinitesimal rotation  $R = 1 + \omega$  then satisfies  $1 + \omega^T + \omega = 1$ , implying, as claimed, that  $\omega^T = -\omega$ .

<sup>11</sup> This time, the explicit counting is  $\xi^{\mu} : 3, \lambda^{a} : 2, \lambda^{ab} : 1, \Lambda_{D} : 1, \sigma : 1$ , which adds up to 8. Note that antisymmetry of rotations implies that it only has a single free parameter.

<sup>12</sup> Note that in contradistinction to chapter 6, the *N*-invariant objects in (5.2.98)–(5.2.99) will not have a subscript ( $\chi$ ) indicating invariance in this chapter. We hope that this inter-chapter change of notation will not confuse the reader.

<sup>13</sup> In our calculations, we use that the (G, J, N)-transformations of the fields involved are as follows:

We now explicitly verify the statements made in table 5.2.

$$\delta_{GJN}\hat{e}^a_\mu = \delta_{GJN}e^a_\mu - \overleftarrow{\delta_{GJN}\tau_\mu}M^a - \tau_\mu\delta_{GJN}M^a$$
(5.2.107)

0

$$= \lambda^{a} \tau_{\mu} + \lambda^{a} {}_{b} e^{b}_{\mu} - \tau_{\mu} \left( \lambda^{ab} M_{b} + \lambda^{a} \right).$$
(5.2.108)

Similarly,

$$\delta_{GJN}\tilde{\Phi} = -M_{\mu}\delta_{GJN}v^{\mu} - v^{\mu}\delta_{GJN}M_{\mu} - M^{a}\delta_{GJN}M_{a}$$
(5.2.109)  
=0

$$= -\lambda^{a} M_{a} - \widetilde{v^{\mu} e^{a}_{\mu}} \lambda_{a} + M^{a} \left( \lambda_{a}^{b} M_{b} + \lambda_{a} \right)$$
(5.2.110)

$$=\lambda^{ab}M_aM_b \stackrel{\text{eq.}(5.2.106)}{=} 0. \tag{5.2.111}$$

And

$$\delta_{GJN}h^{\mu\nu} = e_b^{\nu}\lambda^{bc}e_c^{\mu} + e_a^{\mu}\lambda^{ac}e_c^{\nu} = 0, \qquad (5.2.112)$$

where we have used the property (5.2.106). Finally, we have

$$\delta_{GJN}\bar{h}_{\mu\nu} = \delta_{ab} \left( e^a_\mu \delta_{GJN} e^b_\nu + e^b_\nu \delta_{GJN} e^a_\mu \right) - \tau_\mu \delta_{GJN} M_\nu - \tau_\nu \delta_{GJN} M_\mu \tag{5.2.113}$$

$$= \tau_{\nu} e^{a}_{\mu} \lambda_{a} + e^{a}_{\mu} \lambda_{ac} e^{c}_{\nu} + \tau_{\mu} e^{b}_{\nu} \lambda_{b} + e^{b}_{\nu} \lambda_{bc} e^{c}_{\mu} - \tau_{\nu} e^{b}_{\mu} \lambda_{a} - \tau_{\mu} e^{a}_{\nu} \lambda_{a} = 0.$$
(5.2.114)

Further, we have already shown in the geometric relations (5.2.32) and (5.2.34) that these objects share the properties of the TNC fields (cf. (4.2.29)). From the sources, we can again build the (minimal) affine connection of (4.2.74), providing a full realization of TNC geometry on the boundary.

Note that for z > 2, and for many cases with z = 2 (case (*iv*) of (5.2.28) with  $\tau_{\mu}$  HSO), the boundary is described by TTNC geometry. In this case, we can apply a local dilatation to make  $\tau$  closed, turning TTNC into a local NC geometry. The TTNC torsion can thus be ascribed to the dilatation invariance.

Now, the sources of table 5.1 transform under  $\mathfrak{sch}_z(2,1)$ , i.e. the (2 + 1)-dimensional Schrödinger algebra with critical exponent z, as we already remarked in section 5.2.3. This is entirely analogous to the Galilean and Bargmannn structures considered in section 4.2.2; in particular, mimicking the approach taken when gauging the Galilean group in section 4.2.3, we can write the transformations of the sources (6.3.29)–(6.3.34) under H, P, G, J, N, D as

$$\bar{\delta}\mathcal{A}_{\mu} = \pounds_{\xi}\mathcal{A}_{\mu} + \partial_{\mu}\Sigma + [\mathcal{A}_{\mu}, \Sigma], \qquad (5.2.115)$$

where  $A_{\mu}, \Sigma \in \mathfrak{sch}_{z}(2, 1)$  with the expressions

$$\mathcal{A}_{\mu} = H\tau_{\mu} + P_{a}e^{a}_{\mu} + G_{a}\omega_{\mu}{}^{a} + \frac{1}{2}J_{ab}\omega_{\mu}{}^{ab} + Nm_{\mu} + Db_{\mu}, \qquad (5.2.116)$$

$$\Sigma = G_a \lambda^a + \frac{1}{2} J_{ab} \lambda^{ab} + N\sigma + D\Lambda_D, \qquad (5.2.117)$$

which differs from the Bargmann algebra only in the addition of dilatations *D* with corresponding connection  $b_{\mu}$ . The commutation relations of  $\mathfrak{sch}_{z}(2,1)$  read

$$[D,H] = -zH, \quad [D,P_a] = -P_a, \quad [D,G_a] = (z-1)G_a, \quad [D,N] = (z-2)N, \quad [H,G_a] = P_a,$$
(5.2.118)

$$[P_a, G_b] = \delta_{ab}N, \quad [J_{ab}, G_c] = 2\delta_{c[a}G_{b]}, \quad [J_{ab}, P_c] = 2\delta_{c[a}P_{b]}, \quad [J_{ab}, J_{cd}] = 4\delta_{[a[d}J_{c]b]}. \tag{5.2.119}$$

Note that  $\tilde{m}_{\mu} = m_{\mu} - (z - 2)\chi b_{\mu}$ , where  $\chi$  is the Stückelberg field. Thus, the boundary geometry for arbitrary z can be obtained by gauging  $\mathfrak{sch}_z(2,1)$  (with the inclusion of the Stückelberg field  $\chi$ , just like in section 4.2.4). This is done by following the gauging procedure that we described in chapter 4, and was considered in<sup>15</sup> [34]. Finally, we remark that when z = 2, and the boundary geometry is TTNC, there is an additional special conformal transformation K contained in  $\mathfrak{sch}_2(2,1)$  that we need to take into account when writing down the gauge connection (5.2.116) [34]—see also chapter 6, where we carry out this procedure explicitly.

<sup>15</sup> The approach of [34] differs slightly from the approach we follow in this work. In [34], the behaviour under *H*, *P* transformations is inferred via the imposition of curvature constraints rather than by introducing the  $\bar{\delta}$ -transformation.

## 5.2.5 Bulk Metric AlLif Boundary Conditions Revisited

The analyses of the preceding sections allow us to recast the boundary conditions of section 5.2.2 in terms of the metric and the Proca field. Using (5.2.14) with (5.2.25) and (5.2.63), we find that

$$= \frac{\mathrm{d}r}{Rr^2} - \alpha_{(0)}^{2/3} r^{-2z} \tau_{\mu} \tau_{\nu} \mathrm{d}x^{\mu} \mathrm{d}x^{\nu} + \dots + \alpha_{(0)}^{-2/3} r^{-2} \left( \bar{h}_{\mu} \nu + I \tau_{\mu} \tau_{\nu} \right) \mathrm{d}x^{\mu} \mathrm{d}x^{\nu} + \dots,$$
(5.2.121)

while for the Proca field, a similar computation reveals that

$$B = B_r dr + \alpha_{(0)}^{4/3} r^{-z} \tau_\mu dx^\mu + \dots + r^{z-2} \tilde{I} \tau_\mu dx^\mu + \dots$$
 (5.2.122)

### 5.3 VEVS AND COVARIANT WARD IDENTITIES IN LIFSHITZ HOLOGRAPHY

In this section, we continue our investigation of Lifshitz holography focussing on the VEVs and their Ward identities. The analysis is based on [29], but contains more details.

#### 5.3.1 VEVs & Circumvention of Renormalization

We now want to consider the vevs associated with our sources. In principle, we really need the renormalized on-shell action, which would require us to holographically renormalize the EPD model with the Lifshitz solution. However, we can apply general arguments to find the leading behaviour of the VEVs. In particular, as was our modus operandi when doing holographic renormalization in chapter 2, we know that the on-shell action on a radial hypersurface can be written as

$$S = \int_{\Sigma_r} \mathrm{d}^3 x \, \sqrt{-\gamma} \mathcal{L}, \tag{5.3.1}$$

where

$$\sqrt{-\gamma} \simeq \alpha_{(0)}^{-1/3} r^{-z-2} e.$$
 (5.3.2)

and  $\mathcal{L}$  is a function of the bulk data, which will become the sources on the boundary. This bulk data (cf. section 5.2.3) we take to be

$$\chi = \{E_0^{\mu}, E_a^{\mu}, \Phi, A_a, \Xi, \varphi\},$$
(5.3.3)

where we have defined  $\varphi = E_0^{\mu} \left( A_{\mu} - \alpha(\Phi) E_{\mu}^0 \right)$ , which is intimately connected with the new source  $\tilde{m}_{\mu}$ . We note that

$$\varphi = \mathcal{O}(r^{2z-2}). \tag{5.3.4}$$

Now, write

$$\sqrt{-\gamma}\mathcal{L} = \sum_{w} e \mathcal{L}_{(w)} r^{w-z-2}, \qquad (5.3.5)$$

where  $e = \det(\tau_{\mu}, e_{\mu}^{a}) = \sqrt{-\gamma}$ . The holographic renormalization procedure, as we have seen, removes all divergent terms<sup>16</sup>, that is, terms ~  $r^{\#}$  for # < 0. On the other hand, terms with  $r^{\varrho}$  where  $\varrho > 0$ can be ignored, since they will play no rôle on the boundary. Thus, the term with scaling weight w = z + 2 is precisely what determines the renormalized on-shell action. By the same token, varying the on-shell action, we get

$$\delta S_{\rm ren} = \int_{\Sigma_r} d^3 x \ e \left[ \mathcal{V} \delta \chi - \mathcal{A} \frac{\delta r}{r} \right], \qquad (5.3.6)$$

<sup>16</sup> As  $r \rightarrow 0$ .
where we have included the  $\alpha_{(0)}^{-1/3}r^{-z-2}$ -dependence (coming from  $\sqrt{-\gamma}$ , cf. (5.3.2)) in the "VEV complex"  $\mathcal{V}$ ; we will have to take this into account later when determining the near-boundary behaviour of the VEVs. The VEV complex consists of the responses

$$\mathcal{V} = \{\mathcal{S}^0_{\mu}, \mathcal{S}^a_{\mu}, \mathcal{T}_{\Phi}, \mathcal{T}^a, \mathcal{T}_{\Xi}, \mathcal{T}_{\varphi}\},\tag{5.3.7}$$

which arise when varying with respect to the bulk data (5.3.3), e.g.  $S^0_{\mu} = \frac{1}{e} \frac{\delta S_{\text{Ren}}}{\delta E^0_{\mu}}$ . Since the variation of the action (5.3.6) is finite by construction, we must demand that, for example, the combination

$$S^0_{\mu}\delta E^{\mu}_0 \tag{5.3.8}$$

does not depend on r, so including the  $\alpha_{(0)}^{-1/3}r^{-z-2}$ -dependence from the metric determinant in the response and using the relation for the near-boundary behaviour of the bulk fields (5.2.29), we find that the combination

$$\alpha_{(0)}^{-2/3} r^{-z-2+z} \mathcal{S}^0_{\mu} \tag{5.3.9}$$

should not depend on *r* and  $\alpha_{(0)}$ ; that is,

$$S^0_{\mu} \simeq r^2 \alpha^{2/3}_{(0)} S^0_{\mu}, \tag{5.3.10}$$

where  $S^0_{\mu}$  is the VEV corresponding to  $e^{\mu}_0$ . Repeating this analysis for  $E^{\mu}_a$  and again using (5.2.29), we find that

$$r^{-z-2+1}\mathcal{S}_{u}^{0} \tag{5.3.11}$$

should be independent of r, and so

$$S^a_{\mu} \simeq r^{z+1} S^a_{\mu},$$
 (5.3.12)

where  $r^{z+1}S^a_{\mu}$  is the vev corresponding to  $e^{\mu}_a$ . Continuing this process, we reproduce the results of [29]:

$$\mathcal{T}_{\varphi} \simeq r^{4-z} \alpha_{(0)}^{2/3} T^{0}, \quad \mathcal{T}^{a} \simeq r^{3} T^{a}, \quad \mathcal{T}_{\Xi} \simeq r^{4} \alpha_{(0)}^{1/3} \left\langle O_{\chi} \right\rangle, \quad \mathcal{T}_{\Phi} \simeq r^{z+2-\Delta} \alpha_{(0)}^{1/3} \left\langle O_{\phi} \right\rangle, \quad \mathcal{A} \simeq r^{z+2} \alpha_{(0)}^{1/3} \mathcal{A}_{(0)}. \tag{5.3.13}$$

This means that we can write the variation of the on-shell action as (on the boundary, i.e. on  $\Sigma_0 = \partial \mathcal{M}$ )

$$\delta S_{\rm ren} = \int_{\partial \mathcal{M}} \mathrm{d}^3 x \ e \left( -S^0_\mu \delta v^\mu + S^a_\mu \delta e^\mu_a + T^0 \delta \tilde{m}_0 + T^a \delta \tilde{m}_a + \langle O_\chi \rangle \ \delta \chi + \langle \tilde{O}_\phi \rangle \ \delta \phi - \mathcal{A}_{(0)} \frac{\delta r}{r} \right).$$
(5.3.14)

Since the responses were explicitly constructed such that all factors of r and  $\alpha_{(0)}$  drop out, the expression for  $\delta S_{\text{ren}}$  in (5.3.14) follows immediately—except for the terms involving  $\tilde{m}$ . We now show how they come about: by (5.2.75), we have

$$\mathcal{T}_{\varphi}\delta\varphi \simeq T^0 e_0^{\mu}\delta\tilde{m}_{\mu} = T^0\delta\tilde{m}_0. \tag{5.3.15}$$

Also by (5.2.75),  $E^0_{\mu}E^{\mu}_a = 0$  implies  $A_a \simeq -r^{z-2}\tilde{m}_a$ , and so

$$\mathcal{T}^a \delta A_a \simeq T^a \delta \tilde{m}_a.$$
 (5.3.16)

Finally, the vev corresponding to  $\phi$  requires additional terms when  $\Delta = 0$  (we indicate this by a tilde on the VEV)

$$\langle \tilde{O}_{\phi} \rangle = \langle O_{\phi} \rangle + \delta_{\Delta,0} \left[ \frac{1}{3} v^{\mu} \left( S^{0}_{\mu} + T^{0} \tilde{m}_{\mu} \right) + \frac{1}{3} e^{\mu}_{a} \left( S^{a}_{\mu} + T^{a} \tilde{m}_{\mu} \right) \right] \frac{d \log \alpha_{(0)}}{d\phi}.$$
(5.3.17)

where the extra contributions for  $\Delta = 0$  come from factors of  $\alpha_{(0)}(\phi)$  in the leading behaviour of the sources. Also note that for  $\tau_{\mu}$  HSO (i.e. for TTNC), it can be shown that  $\langle \tilde{O}_{\phi} \rangle = 0$ ; this is due to the presence of an extra constraint, and in the upliftable case (see [23, 24] and chapter 6) can be traced back to the FG expansion in the higher-dimensional theory.

We now turn our attention to the behaviour of the VEVs under local transformations: being tensors defined on the boundary, we infer that they transform under the PBH transformations that we considered in section 5.2.3. The radial PBH transformation—corresponding to dilatations—implies a dilatation weight of the vevs corresponding to the power of *r* in the near-horizon expansions of the corresponding responses of (5.3.13). All in all, one finds<sup>17</sup> [29, 32]

$$\delta S^{0}_{\mu} = -T^{0} \partial_{\mu} \sigma - 2\Lambda_{D} S^{0}_{\mu} + \dots, \quad \delta S^{a}_{\mu} = \lambda^{a} S^{0}_{\mu} + \lambda^{a}_{\ b} S^{b}_{\mu} - T^{a} \partial_{\mu} \sigma - (z+1)\Lambda_{D} S^{a}_{\mu} + \dots,$$
(5.3.18)

$$\delta T^{0} = (z-4)\Lambda_{D}T^{0} + \dots, \quad \delta T^{a} = \lambda^{a}T^{0} + \lambda^{a}{}_{b}T^{b} - 3\Lambda_{D}T^{a} + \dots,$$
(5.3.19)

$$\delta \langle O_{\chi} \rangle = -4\Lambda_D \langle O_{\chi} \rangle + \dots, \quad \delta \langle O_{\phi} \rangle = \delta_{\Delta,0} \frac{\mathrm{d} \log \alpha_{(0)}}{\mathrm{d}\phi} \lambda_a T^a - (z+2-\Delta)\Lambda_D \langle O_{\phi} \rangle + \dots,$$
(5.3.20)

where dots denote Lie derivatives along  $\xi^{\mu}$  and possibly derivatives of  $\Lambda_D$ . We note that the VEVs also transform under the Schrödinger group, just like the sources.

## 5.3.2 Boundary Energy-Momentum Tensor and Mass Current

The HIM stress-tensor, which was introduced in [142], is relevant when additional non-scalar fields are present in the bulk model. Following [142], we proceed to define

$$e\tilde{T}^{\mu}_{\nu} = e^{\mu}_{a}\frac{\delta S_{\text{Ren}}}{\delta e^{\nu}_{a}} + v^{\mu}\frac{\delta S_{\text{Ren}}}{\delta v^{\nu}_{0}}$$
(5.3.21)

$$= e \left( S_{\nu}^{a} e_{a}^{\mu} - S_{\nu}^{0} v^{\mu} \right).$$
 (5.3.22)

However, under local Stückelberg gauge transformations (N), this is not invariant, since

$$\delta_N \tilde{T}^{\mu}_{\ \nu} = e^{\mu}_a \delta_N S^a_{\nu} + S^a_{\nu} \overleftarrow{\delta_N e^{\mu}_a} - v^{\mu} \delta_N S^0_{\nu} - S^0_{\nu} \overleftarrow{\delta_N v^{\mu}}$$
(5.3.23)

$$= T^a \partial_\nu \sigma e^\mu_a - v^\mu T^0 \partial_\mu \sigma. \tag{5.3.24}$$

However, we can define an "improved" HIM tensor which is invariant under the full Schrödinger group by noting that  $\delta_N \chi = \sigma$ , implying that the gauge invariant extension of (5.3.24) is

$$T^{\mu}_{\ \nu} = (S^{a}_{\nu} + T^{a}\partial_{\nu}\chi) e^{\mu}_{a} - \left(S^{0}_{\nu} + T^{0}\partial_{\nu}\chi\right) v^{\mu}.$$
(5.3.25)

Being built of e.g.  $S_{\nu}^{a}$  and  $\nu^{\mu}$ , we infer that  $T_{\nu}^{\mu}$  has scaling dimension z + 2, making it marginal for two spatial dimensions. The vielbein projections of  $T_{\nu}^{\mu}$  become the energy density  $(T_{\nu}^{\mu}\tau_{\mu}v^{\nu})$ , momentum flux  $(T_{\nu}^{\mu}\tau_{\mu}e_{a}^{\nu})$ , energy flux  $T_{\nu}^{\mu}e_{\mu}^{a}v^{\nu}$  as well as stress  $(T_{\nu}^{\mu}e_{\mu}^{a}e_{b}^{\nu})$ ; the mass density is given by  $T^{0} = T^{\mu}\tau_{\mu}$ , while the mass flux is  $T^{a} = T^{\mu}e_{\mu}^{a}$ . These quantities appear also in e.g. [26, 27]. We summarize our findings in table 5.3 below.

Table 5.3: Tangent space projections of  $T^{\mu}{}_{\nu}$  and  $T^{\mu}$  with associated scaling weights. Quantities in parentheses denote the corresponding symbol in [26].

VEV	$T^{\mu}{}_{ u} au_{\mu}v^{ u}\left(\mathcal{E} ight)$	$T^{\mu}{}_{\nu}\tau_{\mu}e^{\nu}_{a}\left(\mathcal{P}_{a}\right)$	$T^{\mu}{}_{\nu}e^{a}_{\mu}v^{\nu}$ $(\mathcal{E}^{a})$	$T^{\mu}{}_{\nu}e^a_{\mu}e^{\nu}_b\ (\Pi^a{}_b)$	$T^{\mu} au_{\mu}( ho)$	$T^{\mu}e^{a}_{\mu}~( ho^{a})$
scaling dim.	z+2	3	2z + 1	z+2	4-z	3

## 5.3.3 Ward Identities

The Ward identities are obtained by demanding invariance of (5.3.14) under the transformations (6.3.29)-(6.3.34). The derivation of the Ward identities will, it turns out, depend on which of the four cases of (5.2.28) we consider, but the final result will not change [29, 32].

The boost Ward identity, for example, is obtained by replacing the variations in (5.3.14) by their corresponding transformations under Galilean boosts. The only sources transforming under boosts are  $v^{\mu}$ ,  $\tilde{m}_{a}$  and  $\tilde{m}_{a}$ , where the boost transformation properties of the two last follow from (5.2.97), i.e.

$$\delta_G \tilde{m}_a = \delta_G (\tilde{m}_\mu e_a^\mu) = \tilde{m}_\mu \overleftarrow{\delta_G e_a^\mu} + e_a^\mu \overleftarrow{\delta_G \tilde{m}_\mu} = \lambda_a, \qquad (5.3.26)$$

<sup>17</sup> Note the typo in [29] where all  $\delta_N$  transformations, i.e. those involving  $\sigma$ , of the VEVs have the wrong sign.

as well as

$$\delta_G \tilde{m}_0 = \delta_G (-v^\mu \tilde{m}_\mu) = -\tilde{m}_\mu \underbrace{\delta_G v^\mu}_{\delta_G v^\mu} - v^\mu \underbrace{\lambda^b e_{\mu b}}_{\lambda^b e_{\mu b}}^{=0, \text{ since } v^\mu e_{\mu b} = 0} = -\lambda^a \tilde{m}_a, \tag{5.3.27}$$

which means that

$$0 = \delta_G S_{\rm ren} = \int d^3 x \ e \left[ -S^0_{\mu} e^{\mu}_a - T^0 \tilde{m}_a + \delta_{ab} T^b \right] \lambda^a, \tag{5.3.28}$$

that is to say, the boost Ward identity is

$$-S^{0}_{\mu}e^{\mu}_{a} - T^{0}\tilde{m}_{a} + \delta_{ab}T^{b} = 0.$$
(5.3.29)

Now, this result can be *covariantized*, by which we mean a rewriting of the result above involving the boost invariant quantities of (5.2.98)–(5.2.99). First, observe that

$$\hat{e}^{a}_{\mu}T^{\mu} = \left(e^{a}_{\mu} - (\tilde{m}_{\nu} - \partial_{\nu}\chi)e^{\nu a}\tau_{\mu}\right)\left(-T^{0}v^{\mu} + T^{b}e^{\mu}_{b}\right)$$
(5.3.30)

$$= -T^0(\tilde{m}_\nu - \partial_\nu \chi)e^{\nu a} + T^a, \qquad (5.3.31)$$

where we have used the orthogonality property  $\tau_{\mu}e_{a}^{\mu}$ . Next, note that

$$-\tau_{\nu}e^{\mu a}T^{\nu}{}_{\mu} = -\tau_{\nu}e^{\mu a}\left[-(S^{0}_{\mu} + T^{0}\partial_{\mu}\chi)v^{\nu} + (S^{a}_{\mu} + T^{a}\partial_{\mu}\chi)e^{\nu}_{a}\right]$$
(5.3.32)

$$= -S^0_{\mu} - T^0 \partial_{\mu} \chi, \tag{5.3.33}$$

and thus the boost Ward identity (5.3.29) can be expressed covariantly in the form

$$\hat{e}^a_\mu T^\mu = \tau_\mu e^{\nu a} T^\mu_{\ \nu}.$$
 (5.3.34)

Repeating the procedure, we see that that the rotation Ward identity—appropriately antisymmetrized (due to antisymmetry of  $\lambda_{ab}$ )—reads

$$S^a_\mu e^{\mu b} + T^a \tilde{m}^b - (a \leftrightarrow b) = 0.$$
(5.3.35)

Multiplying with  $\tilde{m}^b$  and antisymmetrizing, we get the relation

$$S^{0}_{\mu}e^{\mu[a}\tilde{m}^{b]} = T^{[a}\tilde{m}^{b]}, \qquad (5.3.36)$$

which means that the naïve rotation Ward identity (5.3.35) can be recast in the form

$$S^{[a}_{\mu}e^{\mu b]} + S^{0}_{\mu}e^{\mu[a}\tilde{m}^{b]} = 0.$$
(5.3.37)

Now, consider the combination

$$\hat{e}_{\nu}^{a}e^{b\mu}T^{\nu}{}_{\mu} = \left(e_{\nu}^{a} - M_{\rho}e^{\rho a}\tau_{\nu}\right)e^{b\mu}\left(-(S_{\mu}^{0} + T^{0}\partial_{\mu}\chi)v^{\nu} + (S_{\mu}^{c} + T^{c}\partial_{\mu}\chi)e_{c}^{\nu}\right)$$

$$= e^{\mu b}S_{\mu}^{a} + e^{\mu b}T^{a}\partial_{\mu}\chi - \tilde{m}_{\rho}e^{\rho a}e^{\mu b}S_{\mu}^{0} - \tilde{m}_{\rho}e^{\rho a}e^{\mu b}T^{0}\partial_{\mu}\chi - \partial_{\rho}\chi e^{\rho a}e^{\mu b}T^{0}\partial_{\mu}\chi + \partial_{\rho}\chi e^{\rho a}e^{\mu b}S_{\mu}^{0},$$
(5.3.38)
(5.3.39)

where we immediately note that the next-to-last term (in red) is symmetric and so vanishes upon antisymmetrization, and hence

$$\hat{e}_{\nu}^{[a}e^{b]\mu}T^{\nu}{}_{\mu} = e^{\mu[b}S^{a]}{}_{\mu} - \tilde{m}^{[a}e^{\mu b]}S^{0}_{\mu} + \partial_{\mu}\chi \left(e^{\mu b}T^{a} - \tilde{m}^{a}e^{\mu b}T^{0} + e^{\mu a}e^{\rho b}S^{0}_{\rho}\right)$$
(5.3.40)

$$-\partial_{\mu}\chi\left(e^{\mu a}T^{b}-\tilde{m}^{b}e^{\mu a}T^{0}+e^{\mu b}e^{\rho a}S^{0}_{\rho}\right)$$
(5.3.41)

$$= e^{\mu[b}S^{a]}_{\mu} - \tilde{m}^{[a}e^{\mu b]}S^{0}_{\mu} + \partial_{\mu}\chi \left(e^{\mu b}T^{a} - \tilde{m}^{a}e^{\mu b}T^{0} - e^{\mu b}e^{\rho a}S^{0}_{\rho}\right)$$
(5.3.42)

$$-\partial_{\mu}\chi\left(e^{\mu a}T^{b}-\tilde{m}^{b}e^{\mu a}T^{0}-e^{\mu a}e^{\rho b}S^{0}_{\rho}\right)$$
(5.3.43)

$$= e^{\mu[b}S^{a]}_{\mu} - \tilde{m}^{[a}e^{\mu b]}S^{0}_{\mu} + \partial_{\mu}\chi e^{\mu b} \left(T^{a} - \tilde{m}^{a}T^{0} - e^{\rho a}S^{0}_{\rho}\right)$$
(5.3.44)

$$-\partial_{\mu}\chi e^{\mu a} \left(T^{b} - \tilde{m}^{b}T^{0} - e^{\rho b}S^{0}_{\rho}\right)$$
(5.3.45)

$$= e^{\mu[b}S^{a]}_{\mu} - \tilde{m}^{[a}e^{\mu b]}S^{0}_{\mu}$$
(5.3.46)  
$$= e^{\mu[b}S^{a]}_{\mu} + \tilde{m}^{[b}e^{\mu a]}S^{0}_{\nu}.$$
(5.3.47)

$$=e^{\mu[b}S^{a]}_{\mu} + \tilde{m}^{[b}e^{\mu a]}S^{0}_{\mu}, \tag{5.3.47}$$

where we have used the boost Ward identity in the form (5.3.29). Combining our findings, we conclude that the rotation Ward identity is covariantly expressed as

$$0 = \hat{e}_{\nu}^{[a} e^{b]\mu} T^{\nu}{}_{\mu}. \tag{5.3.48}$$

Repeating this procedure, we find that the rest of the Ward identities become [29]

$$\langle O_{\chi} \rangle = e^{-1} \partial_{\mu} (eT^{\mu})$$
 (gauge transformations), (5.3.49)

$$\mathcal{A}_{(0)} = -z\hat{v}^{\nu}\tau_{\mu}T^{\mu}{}_{\nu} + \hat{e}^{a}_{\nu}e^{\nu}_{a}T^{\mu}{}_{\nu} + 2(z-1)\tilde{\Phi}\tau_{\mu}T^{\mu} \quad \text{(dilatations),}$$
(5.3.50)

$$0 = \nabla_{\mu} T^{\mu}{}_{\nu} + 2\Gamma^{\rho}_{[\mu\rho]} T^{\mu}{}_{\nu} - 2\Gamma^{\rho}_{[\nu\rho]} T^{\rho}{}_{\nu} - T^{\mu} \hat{e}^{a}_{\mu} D_{\nu} M_{a} + \tau_{\mu} T^{\mu} \partial_{\nu} \tilde{\Phi} \quad \text{(diffeomorphisms),}$$

$$(5.3.51)$$

where  $D_{\mu}M^{a} = \partial_{\mu}M^{a} - \omega_{\mu}{}^{a}{}_{b}M^{b}$ . Note the similarity to the Ward identities derived from null reduction in (4.4.8)–(4.4.10) when setting z = 2.

# 5.4 OUTLOOK

Many interesting avenues for future research into Lifshitz holography exist. Studying holographic entanglement entropy in the context of Lifshitz spacetimes is an interesting unsolved problem, considered in [143, 144] (also considered recently from a field theory perspective in [145, 146]). There are also potentially very interesting ties between Lifshitz holography and Hoîava-Lifshitz (HL) gravity: it was shown in [53] that dynamical NC geometry is equivalent to HL gravity, and since NC geometry plays a prominent role in Lifshitz holography, it would be exciting to explore the rôle of HL gravity in Lifshitz holography.

In a similar spirit, having a dynamical NC bulk (i.e. a HL bulk) would an extremely interesting generalization; something that was suggested from the perspective of HL gravity in [147]. It would be extremely interesting to develop upon these results given new knowledge that HL gravity is the same as dynamical TNC geometry and explore the general rôle TNC geometry plays in such holographic setups. In many ways, starting from HL gravity is more natural, and we hope to report on such an analysis in the future.

Furthermore, studying holographic Lifshitz hydrodynamics is also worthwhile; this was done in [148, 149] as well as in [57]. More generally, studying non-relativistic fluid dynamics independently from holography using TNC geometry is the topic of an upcoming paper [150].

In this chapter, we develop charged Lifshitz holography for z = 2, building on the insights from the last chapter. The bulk action consists of a generalization of the EPD model considered in chapter 5: an EPD-Maxwell-scalar model, which is related to the EMD model (3.2.1) we renormalized in chapter 3 via Scherk-Schwarz reduction; consequently we will refer to this EMD model as the *electromagnetic uplift*. As we will show, the new sources transform, perhaps not entirely unexpectedly, as the fields of Galilean Electrodynamics (GED), and while there is some freedom in defining the sources, we find that there is a natural choice respecting the symmetries. We also show that with this choice, the (integrated) Weyl anomaly becomes Hořava-Lifshitz (HL) gravity coupled to GED. This generalizes the observation made in [24, 107] that the Weyl anomaly in pure z = 2 Lifshitz holography is related to HL gravity.

The analysis is a novel extension of the approach taken in [23, 24], and the findings of this chapter, along with those of chapter 3, will be the subject of [41], to appear.

We begin with an introduction to GED in section 6.1, where we discuss the relation to previous work and show in section 6.1.2 it may be obtained as a null reduction of "ordinary" electromagnetism. In this section, we also provide dictionary between the notation used in this work and that used in previous literature. In section 6.1.3, we provide a new dimensional analysis of GED on anisotropic backgrounds.

In section 6.2, we discuss the Scherk-Schwarz reduction of the five-dimensional electromagnetic uplift. We begin our exposition with a recount of the uplift as used in previous work [23, 24, 57] and show how this leads to a specific EPD model. A detailed derivation of this is presented in appendix H. In section 6.2.2 we discuss the additional subtleties involved with the extra Maxwell field and present the full result of the Scherk-Schwarz reduction of the electromagnetic uplift. We then vary the resulting new action in section 6.2.3 and determine all equations of motion as well as the responses. Finally, in section 6.2.4, we demonstrate how the z = 2 Lifshitz solution in the dimensionally reduced model is related to a z = 0 Schrödinger geometry, which is asymptotically AdS.

We then turn our attention to the sources in section 6.3. We define all sources in section 6.3.1 and discuss the properties of the new sources; in particular we find that the new sources appear as subleading terms in the near-boundary expansion of the bulk U(1) gauge field. This is entirely analogous to the source  $m_{\mu}$ , which also appears as a subleading term in the expansion of the time-like bulk vielbein. In section 6.3.2, we first recall how the known sources transform before we proceed to derive the novel transformation properties of the new sources under all local transformations and demonstrate that they transform as GED fields, which were discussed in section 6.1.2.

In section 6.4, we discuss the boundary geometry of our model. In particular, we show that the Scherk-Schwarz reduction employed in section 6.2 becomes null on the boundary, which, as demonstrated in section 4.3, directly leads to TNC geometry. Building on our considerations in chapter 4, we discuss the relation to the gauging of the Schrödinger algebra for z = 2,  $\mathfrak{sch}_2(2,1)$ , which we show leads to TTNC geometry, since the imposition of certain curvature constraints forces  $\tau_{\mu}$  to be HSO. The discussion is based [34, 53]. In order to obtain TNC, we must add torsion by hand. We also discuss the Stückelberg symmetry of the central charge.

As advertised, the reduced anomaly takes the form of a Lagrangian describing Hořava-Lifshitz gravity coupled to GED, as we demonstrate in section 6.5. In order to obtain this result, we—following [24]—make a few simplifying assumptions, namely we consider the less general case of NC boundary geometry, where the torsion vanishes, and set the Stückelberg field to zero.

In order to make the connection between the electromagnetic uplift and the reduced model manifest, we determine the near-boundary expansions of the four-dimensional bulk fields in section 6.6. Based on the novel FG expansions of the five-dimensional fields that we determined in chapter 3, we explicitly see the emergence of structure that leads to the definition of the sources in section 6.3. In this section, we also derive a constraint relating the boundary value of the dilaton to the twist, which has the important consequence that  $\phi$  is not an independent field.

In section 6.7, we determine the VEVs corresponding to the sources introduced in section 6.3. In particular, we relate the VEVs to the responses worked out in section 6.2.3 and work out their near-boundary behaviour in section 6.7.1. We then derive the relation between the four-dimensional VEVs

and the five-dimensional VEVs of the electromagnetic uplift in section 6.7.2, which allows us to determine all local transformations of the VEVs in section 6.7.3.

Having determined all the VEVs, we derive novel Ward identities for charged Lifshitz holography in section 6.8 by demanding invariance of the variation of the section under the local symmetries. We begin in section 6.8.1 with a definition of the HIM stress-energy tensor and define several GED-augmented versions of this object that will appear in the Ward identities. We also provide the map between these objects and the five-dimensional stress-energy tensor based on our results in section 6.7.2. In section 6.8.2, we then derive all Ward identities and consider their TNC covariant forms. Finally, in section 6.8.3, we consider the Ward identities from a reduction perspective and show how the Ward identities obtained in section 6.8.2 are obtained from the five-dimensional Ward identities derived in chapter 3.

Finally, in section 6.9, we present a conjecture for the extension of these results to general values of z based on our results in section 6.1.3.

### 6.1 GALILEAN ELECTRODYNAMICS

Galilean electrodynamics (GED) was recently explored in the context of Newton-Cartan geometry in [58] following earlier work in [151] (see also [152] for a historical review of Galilean electromagnetism). In this section, we summarize the construction of [58] with a particular emphasis on the coupling of GED to TNC geometries as obtained from null reduction; this will serve to introduce the notation we will employ in a holographic context later in the chapter.

### 6.1.1 Non-Relativistic Electrodynamics & the Unification of Two Limits

Following [151], we identify two limits of Maxwellian electromagnetism that produce Galilean invariant theories, known as the *electric* and *magnetic* limits, respectively. We will initially work at the level of equations of motion and eventually identify a larger theory admitting an off-shell description into which both these limits can be embedded [153]; this model will be referred to as Galilean Electrodynamics (GED) [58]. In order to discuss relativistic effects, we will keep factors of *c* explicit in this section.

Denote by  $\hat{a}_{\mu}$  a U(1) gauge field on  $\mathbb{R}^{d,1}$  with coordinates  $(t, x^i)$ . The associated field strength is defined as  $\hat{f}_{\mu\nu} = 2\partial_{[\mu}\hat{a}_{\nu]}$ , implying that the equations of motion in the absence of sources,  $\partial_{\mu}\hat{f}^{\mu\nu} = 0$ , can be written as

$$\partial_i \left( \partial_i \hat{a}_t - \frac{1}{c} \partial_t \hat{a}_i \right) = 0, \qquad \frac{1}{c} \partial_t \left( \partial_i \hat{a}_t - \frac{1}{c} \partial_t \hat{a}_i \right) + \partial_j \hat{f}_{ji} = 0.$$
(6.1.1)

The gauge field transforms under gauge transformations as

$$\hat{a}'_t = \hat{a}_t + \frac{1}{c} \partial_t \Gamma, \qquad \hat{a}'_i = \hat{a}_i + \partial_i \Gamma.$$
(6.1.2)

Depending on whether  $\hat{a}_{\mu}$  is time-like or space-like, we can take either the electric or magnetic limit: these are summarized in the table below.

Name	Limit structure	EoM			
	$\hat{a}_t = -\vartheta, \hat{a}_i = \frac{1}{c}a_i, \Gamma = \frac{1}{c}\Gamma_{(0)}$	$\partial_i \partial_i \vartheta = 0 = \partial_t \partial_i \vartheta + \partial_j f_{ji}$			
Electric	$c \to \infty$ with $\vartheta, a_i, \Gamma_{(0)}$ fixed	where $f_{ij} = 2\partial_{[i}a_{j]}$			
	$\hat{a}_t = -\tilde{\varphi}, \hat{a}_i = ca_i, \Gamma = c\Gamma_{(0)}$	$\partial_i \tilde{E}_i = 0 = \partial_j f_{ji}$			
Magnetic	$c \to \infty$ with $\tilde{\varphi}, a_i, \Gamma_{(0)}$ fixed	where $\tilde{E}_i = -\partial_i \tilde{\varphi} - \partial_t a_i$			

Table 6.1: The electric and magnetic limits and the resulting equations of motion (EoM).

In order to obtain GED, we start with a Maxwell action with an additional free scalar  $\hat{\phi}$ , the Lagrangian density of which reads

$$\mathcal{L} = \frac{1}{2c^2} \left( \partial_t \hat{a}_i - c \partial_i \hat{a}_t \right)^2 - \hat{f}_{ij} \hat{f}_{ij} + \frac{1}{2c^2} \left( \partial_t \hat{\phi} \right)^2 - \frac{1}{2} \left( \partial_i \hat{\phi} \right)^2, \tag{6.1.3}$$

and take the GED limit, defined by

$$\hat{\phi} = c\vartheta, \qquad \hat{a}_t = -c\vartheta - \frac{1}{c}\tilde{\varphi}, \qquad \hat{a}_i = a_i, \qquad c \to \infty \text{ with } \vartheta, \tilde{\varphi}, a_i \text{ fixed},$$
(6.1.4)

giving rise to the GED action,

$$S = \int \mathrm{d}^{d+1}x \, \left( -\frac{1}{2} f_{ij} f_{ij} - \tilde{E}_i \partial_i \vartheta + \frac{1}{2} \left( \partial_t \vartheta \right)^2 \right). \tag{6.1.5}$$

The equations of motion for this action are identical to those of the electric limit (cf. table 6.1) along with the additional equation of motion for  $\vartheta$ , which takes the form  $\partial_t^2 \vartheta - \partial_i \tilde{E}_i = 0$ . In this sense, the electric limit sits inside the GED model. To obtain the magnetic limit, we may simply (and consistently) set  $\vartheta = 0$  in the equations of motion for the GED model (6.1.5). For more details, including transformation properties of the fields in the various limits, we refer the reader to [58].

## 6.1.2 Null Reduction of Maxwell Electrodynamics on a Curved Background

A convenient way to obtain GED coupled to TNC geometry is the null reduction (cf. section 4.3) of the Maxwell action on a (d + 2)-dimensional Lorentzian background described by a metric  $\gamma_{AB}$ ,

$$S = -\frac{1}{4} \int \mathrm{d}^{d+2} x \,\sqrt{-\gamma} \mathcal{F}_{AB} \mathcal{F}^{AB},\tag{6.1.6}$$

where  $\mathcal{F} = d\mathcal{A}$ . The null reduction ansatz has the form

$$ds^{2} = \gamma_{AB} dx^{A} dx^{B} = 2\tau_{\mu} dx^{\mu} \left( du - m_{\nu} dx^{\nu} \right) + h_{\mu\nu} dx^{\mu} dx^{\nu}, \qquad (6.1.7)$$

$$\sqrt{-\gamma} = e, \qquad \gamma^{\mu\nu} = 2\tilde{\Phi}, \qquad \gamma^{\mu\mu} = -\hat{v}^{\mu}, \qquad \gamma^{\mu\nu} = h^{\mu\nu}, \tag{6.1.8}$$

$$\mathcal{A}_{\mu} = a_{\mu} - \vartheta m_{\mu} - \tilde{\varphi} \tau_{\mu} =: \hat{a}_{\mu}, \qquad \mathcal{A}_{u} = \vartheta, \tag{6.1.9}$$

where none of the fields depends on *u*. In the above, *A*, *B*, *C*, ... represent (d + 2)-dimensional space-time indices, while  $\mu$ ,  $\nu$ , ... represent (d + 1)-dimensional space-time indices and exclude the compact direction *u*,

For easy reference, we summarize some properties of the Newton-Cartan fields that we will use repeatedly in this chapter (see also chapter 4):

$$v^{\mu}\tau_{\mu} = -1, \qquad v^{\mu}\tau_{\nu} + \delta^{\mu}_{\nu} = e^{\mu}_{a}e^{a}_{\nu}, \qquad e^{a}_{\mu}v^{\mu} = 0, \qquad e^{\mu}_{a}\tau_{\mu} = 0, \qquad e^{\mu}_{a}e^{b}_{\mu} = \delta^{b}_{a},$$
(6.1.10)

and

$$\hat{v}^{\mu} = v^{\mu} - e^{\mu}_{a} e^{\nu a} m_{\nu}, \qquad \hat{e}^{a}_{\mu} = e^{a}_{\mu} - \tau_{\mu} m_{\nu} e^{\nu a}, \qquad \bar{h}_{\mu\nu} = h_{\mu\nu} - 2m_{(\mu}\tau_{\nu)}, \qquad \tilde{\Phi} = -v^{\mu} m_{\mu} + \frac{1}{2} h^{\mu\nu} m_{\mu} m_{\nu}.$$
(6.1.11)

For the curvatures, we therefore obtain

$$\mathcal{F}_{\mu\nu} = 2\partial_{[\mu}\hat{a}_{\nu]} =: \hat{f}_{\mu\nu}, \qquad \mathcal{F}_{\mu\mu} = \partial_{\mu}\vartheta, \qquad (6.1.12)$$

Thus the term  $\mathcal{F}_{AB}\mathcal{F}^{AB}$  in the action (6.1.6) becomes

$$\mathcal{F}_{AB}\mathcal{F}^{AB} = \gamma^{AC}\gamma^{BD}\mathcal{F}_{AB}\mathcal{F}_{CD} = \gamma^{\mu\nu}\gamma^{\rho\sigma}\mathcal{F}_{\mu\rho}\mathcal{F}_{\nu\sigma} + 2\gamma^{\mu\nu}\gamma^{\mu\nu}\mathcal{F}_{u\mu}\mathcal{F}_{u\nu} + 4\gamma^{\mu\mu}\gamma^{\nu\rho}\mathcal{F}_{u\nu}\mathcal{F}_{\mu\rho} + 2\gamma^{\mu\mu}\gamma^{\nu\nu}\mathcal{F}_{u\nu}\mathcal{F}_{\mu\nu}$$

$$(6.1.13)$$

$$= h^{\mu\nu}h^{\rho\sigma}\hat{f}_{\mu\rho}\hat{f}_{\nu\sigma} + 4\tilde{\Phi}h^{\mu\nu}\partial_{\mu}\partial_{\nu}\partial + 4\hat{v}^{\mu}h^{\nu\rho}\partial_{\nu}\partial\hat{f}_{\mu\rho} - 2\hat{v}^{\mu}\hat{v}^{\nu}\partial_{\mu}\partial_{\nu}\partial,$$

$$(6.1.14)$$

which produces the GED action

$$S_{\text{GED}} = -\int \mathrm{d}^{d+1}x \; e \left[\frac{1}{4}h^{\mu\nu}h^{\rho\sigma}\hat{f}_{\mu\rho}\hat{f}_{\nu\sigma} + \tilde{\Phi}h^{\mu\nu}\partial_{\mu}\vartheta\partial_{\nu}\vartheta + \hat{v}^{\mu}h^{\nu\rho}\partial_{\nu}\vartheta\hat{f}_{\mu\rho} - \frac{1}{2}\hat{v}^{\mu}\hat{v}^{\nu}\partial_{\mu}\vartheta\partial_{\nu}\vartheta\right]. \tag{6.1.15}$$

The GED fields transform under  $Barg(d, 1) \times U_{\Gamma}(1)$  as<sup>1</sup> [58]

$$\delta a_{\mu} = \vartheta e^{a}_{\mu} \lambda_{a} + \tau_{\mu} a_{\nu} e^{\nu}_{a} \lambda^{a} + \tau_{\mu} v^{\nu} \partial_{\nu} \Gamma + \partial_{\mu} \Gamma, \qquad \delta \tilde{\varphi} = a_{\nu} e^{\nu}_{a} \lambda^{a} + v^{\nu} \partial_{\nu} \Gamma, \qquad \delta \vartheta = 0, \tag{6.1.16}$$

where  $\lambda^a$  parametrizes Galilean boosts, while  $\Gamma$  represents a  $U_{\Gamma}(1)$  gauge transformation.

In order to facilitate a comparison with the work on GED in [58], we construct a table below providing a dictionary between different notations.

<sup>1</sup> Note the typo in the transformation property of  $a_{\mu}$  under gauge transformations in eq. (6.6) in [58].

.

Festuccia et al. [58]	Notation in present work	Description
$A_{\mu}$	${\cal A}_{(0)\mu}$ , $\hat{a}_{\mu}$	b'dary value of 3-dimensional part of 5-dim. gauge field
$a_{\mu}$	$a_{\mu}$	Spatial part of "GED gauge field"; satisfies $v^{\mu}a_{\mu}=0$
$ ilde{arphi}$	$ ilde{arphi}$	Temporal part of GED gauge field
$\overline{A}_{\mu}$ (Note the bar!)	$a_\mu - artheta  au_\mu$	The GED gauge field
φ	θ	The GED scalar; comes from $\mathcal{A}_u$
$M_{\mu}$	$m_{\mu}$	Newton-Cartan gauge field
$\tau_{\mu}$ $\tau_{\mu}$		Newton-Cartan clock form
$v^{\mu}$ $v^{\mu}$		Inverse NC clock form, $v^{\mu}\tau_{\mu} = -1$
$e^a_\mu$	$e^a_\mu$	NC vielbein
$e_a^{\mu}$ $e_a^{\mu}$		Inverse NC vielbein, $e^a_\mu e^\mu_b = \delta^a_b$

Table 6.2: Dictionary between [58] and this work.

## 6.1.3 Dimensional Analysis & Scaling Weights for Anisotropic Backgrounds

The GED action (6.1.5) enables us to determine the dimensionality of the GED fields and their associated sources when the background is anisotropic, characterized by time scaling with z > 1 and spatial directions scaling with weight one. Restricting to d = 2, the measure acquires scaling weight -(z + 2). Introducing the scaling weight operator  $\Delta[\cdot]$ , we observe that for  $a_i$ , the relevant piece of the action reads  $f_{ij}f_{ij}$ . Now, since  $\Delta[f_{ij}] = 1 + \Delta[a_i]$  due to the *spatial* derivative, we obtain the relation

$$2 + z = 2 + 2\Delta[a_i] \Rightarrow \Delta[a_i] = \frac{z}{2}.$$
(6.1.17)

For  $\vartheta$ , the piece  $(\partial_t \vartheta)^2$  in the action yields

$$2 + z = 2(2 + \Delta[\vartheta]) \Rightarrow \Delta[\vartheta] = \frac{z - 2}{2}, \qquad (6.1.18)$$

while the term  $\partial_i \tilde{\varphi} \partial_i \vartheta$  gives us

$$2 + z = 2 + \Delta[\tilde{\varphi}] + \Delta[\vartheta] = \frac{2 + z}{2} + \Delta[\tilde{\varphi}] \Rightarrow \Delta[\tilde{\varphi}] = \frac{2 + z}{2}.$$
(6.1.19)

We note that this is consistent since the last piece  $\partial_t a_i \partial_i \vartheta$  has scaling weight 2 + z/2 + 1 + (z-2)/2 = 2 + z. To summarize: on anisotropic backgrounds, the fields of GED scale as in table 6.3.

Field <i>a<sub>i</sub></i>		$ ilde{arphi}$	v		
Weight	z/2	z/2 + 1	z/2 - 1		

Table 6.3: Scaling Weights of the GED fields.

The GED fields generally couple to "GED charges", which we will simply refer to as VEVs. When considering such couplings, we must add a Lagrangian of the form

$$\mathcal{L}_{\text{VEV}} = j^i a_i + j^0 \tilde{\varphi} + \langle O_\vartheta \rangle \,\vartheta. \tag{6.1.20}$$

Performing the same analysis as above, we find that the VEVs have scaling weights as in table 6.4.

VEV	j <sup>i</sup>	$j^0$	$\langle O_{\vartheta} \rangle$		
Weight	z/2 + 2	z/2 + 1	z/2 + 3		

Table 6.4: Scaling Weights of the GED VEVs.

#### 6.2 THE ELECTROMAGNETIC UPLIFT

### 6.2.1 The Uplift & z = 2 Pure Lifshitz Holography

In this section, we review the results of [23, 24]. This is also the approach of [57], which deals with hydrodynamics in the context of Lifshitz holography. As we will discuss below, when making the choices

$$Z = e^{3\Phi}, \quad W = 4, \quad V(\Phi) = 2e^{-3\Phi} - 12e^{-\Phi}, \quad k = 3,$$
 (6.2.1)

in the EPD model (5.2.1), i.e. when

$$S = \int d^4x \,\sqrt{-g} \left( R - \frac{1}{4}e^{3\Phi}F^2 - 2B^2 - \frac{3}{2}\left(\partial\Phi\right)^2 - 2e^{-3\Phi} - 12e^{-\Phi} \right),\tag{6.2.2}$$

the action (6.2.2) can be uplifted to the following five-dimensional action,

$$S_{5d} = \frac{1}{2\kappa_5^2} \int d^5x \sqrt{-\mathcal{G}} \left( R^{(\mathcal{G})} + 12 - \frac{1}{2} \partial_{\mathcal{M}} \psi \partial^{\mathcal{M}} \psi \right), \qquad (6.2.3)$$

with  $\kappa_5^2 = 8\pi G_5$  and  $\mathcal{M} = (u, M)$ . The reduced model admits z = 2 Lifshitz solutions.

The action (6.2.2) is the  $S^1$  Scherk-Schwarz reduction of the uplift (H.2.9) (see appendix H): here, the global symmetry is the shift symmetry of  $\psi$ ; i.e. the action (H.2.9) is invariant under  $\psi \rightarrow \psi + \Lambda$  for  $\Lambda$  a constant. When going around the circle once, the reduction scheme tells us that  $\psi$  should come back to itself plus a local shift.

Explicitly, the reduction ansatz has the form

$$ds_{5}^{2} = e^{-\Phi}g_{MN}dx^{M}dx^{N} + e^{2\Phi}\left(du + A_{M}dx^{M}\right)^{2},$$
(6.2.4)

$$\psi = 2u + 2\Xi,\tag{6.2.5}$$

where  $\mathcal{M}, \mathcal{N}, \ldots$  are five-dimensional space-time indices, and  $\mathcal{M}, \mathcal{N}, \ldots$  are four-dimensional space-time indices that exclude the compact direction u. The four-dimensional fields  $g_{MN}, A_M, \Xi$  and  $\Phi$  do not depend on the compactified u-direction, which—since we're reducing on a circle of radius L—is periodically identified,  $u \sim u + 2\pi L$ . Note also that since our normalization is such that  $\frac{1}{16\pi G_4} = 1$ , the five-dimensional Newton constant satisfies  $\frac{2\pi L}{16\pi G_5} = 1$ .

The renormalized on-shell four-dimensional  $\widetilde{EPD}z = 2$  action has the schematic form

$$S_{\rm ren} = S + S_{\rm gh} + S_{\rm ct},$$
 (6.2.6)

where  $S_{\rm gh} = 2 \int d^3x \sqrt{-h}K$  is the Gibbons-Hawking boundary term.

Performing this reduction gives the action [57] (we present a detailed derivation of this result in appendix H)

$$S = \int d^4x \,\sqrt{-g} \left( R - \frac{1}{4} e^{3\Phi} F_{MN} F^{MN} - 2B_M B^M - \frac{3}{2} \partial_M \Phi \partial^M \Phi - V \right) + 2 \int d^3x \,\sqrt{-h} K + S_{\rm ct}, \tag{6.2.7}$$

where  $B_{\mu} = A_{\mu} - \partial_{\mu} \Xi$ .

### 6.2.2 Scherk-Schwarz Reduction of the Electromagnetic Uplift

We now add a free abelian gauge field to the five-dimensional model considered above, resulting in what we shall call the *electromagnetic uplift*,

$$S = \frac{1}{2\kappa_5^2} \int d^5 x \,\sqrt{-\mathcal{G}} \left( R + 12 - \frac{1}{2} \partial_{\mathcal{M}} \psi \partial^{\mathcal{M}} \psi - \frac{1}{4} \mathcal{F}_{\mathcal{MN}} \mathcal{F}^{\mathcal{MN}} \right), \tag{6.2.8}$$

where  $\mathcal{F}_{MN}$  is the field strength of the U(1) gauge field  $\mathcal{A}_{M}$ , i.e.

$$\mathcal{F}_{\mathcal{MN}} = 2\partial_{[\mathcal{M}}\mathcal{A}_{\mathcal{N}]}.\tag{6.2.9}$$

The equations of motion are

$$G_{\mathcal{M}\mathcal{N}} = \frac{1}{2} \left( \partial_{\mathcal{M}} \psi \partial_{\mathcal{N}} \psi - \frac{1}{2} (\partial \psi)^2 \mathcal{G}_{\mathcal{M}\mathcal{N}} \right) + 6 \mathcal{G}_{\mathcal{M}\mathcal{N}} + \frac{1}{2} \left( \mathcal{F}_{\mathcal{M}\mathcal{P}} \mathcal{F}_{\mathcal{N}}^{\mathcal{P}} - \frac{1}{4} \mathcal{F}^2 \mathcal{G}_{\mathcal{M}\mathcal{N}} \right), \quad (6.2.10)$$

$$\frac{1}{\sqrt{-\mathcal{G}}}\partial_{\mathcal{M}}\left(\sqrt{-\mathcal{G}}\partial^{\mathcal{M}}\psi\right) = 0, \quad \frac{1}{\sqrt{-\mathcal{G}}}\partial_{\mathcal{M}}\left(\sqrt{-\mathcal{G}}\mathcal{F}^{\mathcal{M}\mathcal{N}}\right) = 0, \tag{6.2.11}$$

where  $G_{MN}$  is the five-dimensional Einstein tensor of  $\mathcal{G}_{MN}$ . Upon reduction, the action (6.2.8) will generically produce an extra gauge field and a scalar<sup>2</sup> compared to the pure uplift considered in section 6.2.1. The massless scalar  $\Upsilon = \mathcal{A}_u$ , with its origin in the five-dimensional gauge field, is associated with spontaneous breaking of the five-dimensional  $U_{\hat{\Gamma}}(1)$  gauge transformations,  $\delta_{\hat{\Gamma}}\mathcal{A}_{\mathcal{M}} = \partial_{\mathcal{M}}\hat{\Gamma}$ , that depend on the compactification coordinate *u*. In the reduced theory, we thus have diffeomorphisms and gauge transformations of the form

$$\hat{\zeta}^N = \zeta^N(x), \qquad \hat{\zeta}^u = \Lambda(x), \qquad \hat{\Gamma} = \Gamma(x).$$
(6.2.12)

Now, we want to identify the Maxwell field of the reduced theory. The naïve choice  $\mathcal{A}_M$  does not transfrom correctly under KK  $U_{\Lambda}(1)$  gauge transformations—that is, gauge transformations of A, which is a remnant of diffeomorphism invariance in the compact direction. This is an ubiquitous feature of KK reductions when *p*-form fields are involved, and also happens e.g. when reducing elevendimensional supergravity to ten-dimensional type IIB supergravity. The higer-dimensional gauge field  $\mathcal{A}_M$  transforms under diffeomorphisms and  $U_{\hat{\Gamma}}(1)$  gauge transformations, we may identity the correct lower dimensional gauge field,

$$\delta \mathcal{A}_{\mathcal{M}} = \hat{\zeta}^{\mathcal{N}} \partial_{\mathcal{N}} \mathcal{A}_{\mathcal{M}} + \mathcal{A}_{\mathcal{N}} \partial_{\mathcal{M}} \hat{\zeta}^{\mathcal{N}} + \partial_{\mathcal{M}} \hat{\Gamma}.$$
(6.2.13)

We are thus lead to conclude that the naïve reduction  $A_M$  transforms according to

$$\delta \mathcal{A}_{M} = \zeta^{N} \partial_{N} \mathcal{A}_{M} + \mathcal{A}_{N} \partial_{M} \zeta^{N} + \overbrace{\mathcal{A}_{u}}^{=\Upsilon} \partial_{M} \Lambda + \partial_{M} \Gamma, \qquad (6.2.14)$$

which transforms as a lower-dimensional vector field but also under  $U_{\Lambda}(1)$  gauge transformations. The combination

$$C_M = \mathcal{A}_M - \Upsilon A_M, \tag{6.2.15}$$

transforms as desired, so we identify it as the lower-dimensional gauge field. Under *u*-independent gauge transformations  $\hat{\Gamma}(x, u) = \Gamma(x)$ —which we will denote  $U_{\Gamma}(1)$ —only  $C_M$  transforms:  $\delta_{\Gamma}C_M = \partial_M \Gamma$ .

We now identify the four-dimensional field strengths, which is closely related to  $\mathcal{F}_{MN}$ , which is manifestly invariant under  $U_{\hat{\Gamma}}(1)$  and  $U_{\Lambda}(1)$  transformations. The frame version of the field strength reads

$$\mathcal{F}_{\underline{a}\underline{b}} = e_{\underline{a}}^{M} e_{\underline{b}}^{N} \left( 2\partial_{[M} C_{N]} + 2Y \partial_{[M} A_{N]} \right), \tag{6.2.16}$$

where  $H_{MN}$  is  $U_{\Lambda}(1)$  and  $U_{\Gamma}(1)$  invariant. For the scalar, this produces the term

$$\mathcal{F}_{\underline{a}u} = e^{\Phi} \partial_{\underline{a}} \Upsilon, \tag{6.2.17}$$

implying that

$$\mathcal{F}^2 = \mathcal{F}_{\underline{a}\underline{b}}\mathcal{F}^{\underline{a}\underline{b}} + 2\mathcal{F}_{\underline{a}\underline{u}}\mathcal{F}^{\underline{a}}_{\underline{u}} = H^2 + \Upsilon^2 F^2 + 2\Upsilon H_{MN}F^{MN} + 2e^{2\Phi}\partial_M \Upsilon \partial^M \Upsilon, \tag{6.2.18}$$

where *H* is the field strength for *C* and *F* is the field strength for *A*. Using  $\frac{\pi L}{\kappa_5^2} = 1$ , we can summarize our findings as

$$\frac{1}{2\kappa_5^2}\sqrt{-\mathcal{G}}\frac{1}{4}\mathcal{F}_{\mathcal{MN}}\mathcal{F}^{\mathcal{MN}} \xrightarrow{\text{reduces to}} \frac{\sqrt{-g}e^{-\Phi}}{4} \left(H^2 + \Upsilon^2 F^2 + 2\Upsilon H_{MN}F^{MN} + 2e^{2\Phi}\partial_M \Upsilon \partial^M \Upsilon\right).$$
(6.2.19)

<sup>2</sup> More specifically, we get the usual KK tower of massive (i.e. Stückelberged) gauge fields. We keep only the massless vector and the massless scalar.

Since nothing else is changed from uplift discussed in section 6.2.1, we infer that the reduced theory has the form of an EPD-Maxwell-scalar model,

$$S = \int d^4x \,\sqrt{-g} \left( R - \frac{1}{4} \left( e^{3\Phi} + e^{-\Phi} Y^2 \right) F^2 - 2B^2 - \frac{3}{2} \left( \partial \Phi \right)^2 - \frac{1}{4} e^{-\Phi} H^2 \right)$$
(6.2.20)

$$-\frac{1}{2}Ye^{-\Phi}H_{MN}F^{MN} - \frac{1}{2}e^{\Phi}(\partial Y)^2 - 2e^{-3\Phi} - 12e^{-\Phi}\Big), \qquad (6.2.21)$$

Note that the counterterm is unchanged under addition of the additional Maxwell field, as we demonstrated in section 3.2.1. This is a very non-trivial result, since it implies that the extra massless field in the lower-dimensional theory do not give rise to divergences in the case of Lifshitz boundary conditions. Explicitly, it reads

$$S_{\rm ct} = 2 \int_{\partial \mathcal{M}} d^3 x \, \sqrt{-\tilde{h}} \left( -\frac{1}{4} e^{\Phi/2} \left[ R^{(\tilde{h})} - \frac{3}{2} \partial_{\mu} \Phi \partial^{\mu} \Phi - \frac{1}{4} e^{3\Phi} F_{\mu\nu} F^{\mu\nu} - 2B_{\mu} B^{\mu} + 10 e^{-\Phi} \right] \right)$$
(6.2.22)  
$$- \log r \int_{\partial \mathcal{M}} d^3 x \, \sqrt{-\tilde{h}} e^{-\Phi/2} \mathcal{A}_{\rm red},$$
(6.2.23)

where  $\tilde{h}$  is the boundary metric. Here, *r* is the radial coordinate (which is inherited from the z = 0 Schrödinger space-time), and  $A_{red}$  is the Scherk-Schwarz reduced Weyl anomaly—we will discuss this later in section 6.5.

## 6.2.3 Variation of the Action

After the reduction, the renormalized action reads

$$S_{\rm ren} = S + 2 \int_{\partial \mathcal{M}} d^3x \sqrt{-\tilde{h}} K + S_{\rm ct}, \qquad (6.2.24)$$

where *S* is given in (6.2.21), and  $S_{ct}$  in (6.2.23). The total variation of the renormalized action consequently takes the form

$$\delta S_{\rm ren} = \int_{\mathcal{M}} d^4 x \, \sqrt{-g} \left( \mathcal{E}_{MN} \delta g^{MN} + \mathcal{E}^N_{(B)} \delta B_N + \mathcal{E}_{\Phi} \delta \Phi + \mathcal{E}_Y \delta Y + \mathcal{E}^N_{(C)} \delta C_N \right)$$

$$+ \int_{\partial \mathcal{M}} d^3 x \, \sqrt{-\tilde{h}} \left( \frac{1}{2} T_{\mu\nu} \delta h^{\mu\nu} + \mathcal{T}^{\nu} \delta B_{\nu} + T_{\Phi} \delta \Phi + \mathcal{T}^{\nu} \delta C_{\nu} + \mathcal{T}_Y \delta Y - \frac{\delta r}{r} e^{-\Phi/2} \mathcal{A}_{\rm red} \right),$$
(6.2.26)

where

$$\mathcal{E}_{MN} = G_{MN} + \frac{1}{8} \left( e^{3\Phi} + e^{-\Phi} Y^2 \right) g_{MN} F_{PQ} F^{PQ} - \frac{1}{2} \left( e^{3\Phi} + e^{-\Phi} Y^2 \right) F_{MP} F_N^{\ P} + g_{MN} B_P B^P - 2B_M B_N$$

$$(6.2.27)$$

$$+ \frac{3}{4} g_{MN} \partial_P \Phi \partial^P \Phi - \frac{3}{2} \partial_M \Phi \partial_N \Phi + g_{MN} e^{-\Phi} \left( -6 + e^{-2\Phi} \right) + \frac{1}{8} e^{-\Phi} H_{PQ} H^{PQ} g_{MN} - \frac{1}{2} e^{-\Phi} H_{MP} H_N^{\ P}$$

$$(6.2.28)$$

$$+\frac{1}{4}e^{-\Phi}YH_{PQ}F^{PQ}g_{MN} - e^{-\Phi}YH_{MP}F_N^P - \frac{1}{2}e^{\Phi}\partial_MY\partial_NY + \frac{1}{4}e^{\Phi}g_{MN}\partial_PY\partial^PY, \qquad (6.2.29)$$

$$\mathcal{E}_{\Phi} = 3\Box_{(g)}\Phi - \frac{1}{4} \left( 3e^{3\Phi} - e^{-\Phi}Y^2 \right) F_{MN}F^{MN} + 6e^{-3\Phi} - 12e^{-\Phi}$$
(6.2.30)

$$+\frac{1}{4}e^{-\Phi}H_{MN}H^{MN} + \frac{1}{2}e^{-\Phi}YH_{MN}F^{MN} + \frac{1}{2}e^{\Phi}\partial_MY\partial^MY, \qquad (6.2.31)$$

$$\mathcal{E}_{(B)}^{N} = \nabla_{M} \left( \left[ e^{3\Phi} + e^{-\Phi} Y^{2} \right] F^{MN} + e^{-\Phi} H^{MN} \right) - 4B^{N},$$
(6.2.32)

$$\mathcal{E}_{Y} = \frac{1}{\sqrt{-g}} \partial_{M} \left( \sqrt{-g} e^{\Phi} \partial^{M} Y \right) - \frac{1}{2} e^{-\Phi} Y F^{2} - \frac{1}{2} e^{-\Phi} H^{MN} F_{MN}$$

$$\mathcal{E}_{(C)}^{N} = \nabla_{M} \left( e^{-\Phi} H^{MN} + Y e^{-\Phi} F^{MN} \right),$$
(6.2.34)

as well as

$$T_{\mu\nu} = -2Kh_{\mu\nu} + 2K_{\mu\nu} - e^{\Phi/2}G^{(h)}_{\mu\nu} + 5e^{-\phi/2}\tilde{h}_{\mu\nu}$$
(6.2.35)

$$+\frac{1}{2}e^{7\Phi/2}F_{\mu\rho}F_{\nu}^{\ \rho}-\frac{1}{8}e^{7\Phi/2}h_{\mu\nu}F_{\rho\sigma}F^{\rho\sigma}-e^{\Phi/2}h_{\mu\nu}B_{\rho}B^{\rho}+2e^{\Phi/2}B_{\mu}B_{\nu}$$
(6.2.36)

$$+\frac{1}{2}e^{\Phi/2}\left(\nabla_{(\tilde{h})\mu}\partial_{\nu}\Phi - \tilde{h}_{\mu\nu}\Box_{(\tilde{h})}\Phi\right) + \frac{7}{4}e^{\Phi/2}\partial_{\mu}\Phi\partial_{\nu}\Phi - e^{\Phi/2}\tilde{h}_{\mu\nu}\partial_{\rho}\Phi\partial^{\rho}\Phi, \qquad (6.2.37)$$

$$T_{\Phi} = -3n^{M}\partial_{M}\Phi - \frac{1}{4}e^{\Phi/2}R^{(\tilde{h})} - \frac{3}{8}e^{\Phi/2}\partial_{\mu}\Phi\partial^{\mu}\Phi - \frac{3}{2}e^{\Phi/2}\Box_{(\tilde{h})}\Phi$$
(6.2.38)

$$+\frac{7}{16}e^{7\Phi/2}F_{\mu\nu}F^{\mu\nu}+\frac{1}{2}e^{\Phi/2}B_{\mu}B^{\mu}+\frac{5}{2}e^{-\Phi/2},$$
(6.2.39)

$$\mathcal{T}^{\nu} = -\left(\left[e^{3\Phi} + e^{-\Phi}Y^2\right]F^{M\nu} + e^{-\Phi}H^{M\nu}\right)n_M - \frac{1}{2}\nabla_{(\tilde{h})\mu}\left(e^{7\Phi/2}F^{\mu\nu}\right) + 2e^{\Phi/2}B^{\nu}, \quad (6.2.40)$$

$$\widetilde{\mathcal{T}}^{\nu} = -\left(e^{-\Phi}H^{M\nu} + 2\Upsilon e^{-\Phi}F^{M\nu}\right)n_M,\tag{6.2.41}$$

$$\mathcal{T}_Y = -e^{\Phi} n^M \partial_M Y. \tag{6.2.42}$$

Note that we have not varied counterterm in the expressions above, so the expression are correct up to log *r*-terms. The reduced bulk is described by the metric

$$ds^{2} = e^{\Phi} \frac{\mathrm{d}r^{2}}{r^{2}} + \tilde{h}_{\mu\nu} \mathrm{d}x^{\mu} \mathrm{d}x^{\nu}, \qquad (6.2.43)$$

so the vector normal to hyperplanes of constant r is given by

$$n^M = -\mathcal{N}\delta_r^M,\tag{6.2.44}$$

where  $\mathcal{N}$  is a normalization constant and the minus is due to the fact that the boundary is located at r = 0. The normalization constant is determined by the condition  $g_{MN}n^Mn^N = 1$ , which translates into the requirement  $1 = g_{rr}\mathcal{N}^2 = \mathcal{N}^2 e^{\Phi}/r$ , which in turn implies that  $\mathcal{N} = re^{-\Phi/2}$ . This means that we have the following relation for the extrinsic curvature

$$K = \tilde{h}^{\mu\nu} K_{\mu\nu}, \qquad K_{\mu\nu} = -\frac{1}{2} \pounds_n \tilde{h}_{\mu\nu}, \qquad n^M = -r e^{-\Phi/2} \delta_r^M \tag{6.2.45}$$

When appropriately truncated, these results agree with those of [57].

### 6.2.4 Oxidation to a z = 0 Schrödinger Geometry

Starting from the z = 0 Schrödinger geometry (see also previous work in [24, 57, 154, 155])

$$ds_5^2 = \frac{\mathrm{d}r^2}{r^2} + \frac{1}{r^2} \left( 2\mathrm{d}t\mathrm{d}u + \mathrm{d}x^2 + \mathrm{d}y^2 \right) + \mathrm{d}u^2, \tag{6.2.46}$$

$$\psi = 2u + 2\Xi_{(0)},\tag{6.2.47}$$

$$\mathcal{A} = \mathcal{A}_{(0)\mu} \mathrm{d}x^{\mu}, \tag{6.2.48}$$

where  $\Xi_{(0)}$  and  $A_{(0)}$  are constants, the Scherk-Schwarz reduction ansatz (H.2.10) gives rise to the Lifshitz solution in the reduced theory:

$$ds^{2} = e^{\Phi_{(0)}} \left( \frac{\mathrm{d}r^{2}}{r^{2}} - e^{-2\Phi_{(0)}} \frac{\mathrm{d}t^{2}}{r^{4}} + \frac{1}{r^{2}} \left( \mathrm{d}x^{2} + \mathrm{d}y^{2} \right) \right), \tag{6.2.49}$$

$$B = e^{-2\Phi_{(0)}} r^{-2} dt \qquad \text{(Constant Stückelberg field)}, \tag{6.2.50}$$

$$\Phi = \Phi_{(0)}, \tag{6.2.51}$$

$$Y = Y_{(0)},$$
 (6.2.52)

$$C = \mathcal{A}_{(0)\mu} dx^{\mu} - Y_{(0)} e^{-2\Phi_{(0)}} r^{-2} dt.$$
(6.2.53)

This solution can be verified by plugging it into the equations of motion we found above.

The z = 0 Schrödinger metric (6.2.46) satisfies the AlAdS requirement (A.2.5), since e.g.

$$R_{trru} = \frac{1}{r^4} = \mathcal{G}_{tr}\mathcal{G}_{ru} - \mathcal{G}_{tu}\mathcal{G}_{rr} + \mathcal{O}(r^{-3}), \qquad (6.2.54)$$

while a similar analysis for the other seven non-zero components of the Riemann tensor shows that, indeed, the z = 0 Schrödinger metric belongs to the class of AlAdS metrics, which is what permits the use of the results obtained in chapter 3.

### 6.3 SOURCES & THEIR TRANSFORMATIONS

In this section, we find the sources of charged Lifshitz Holography. We will be guided by the fact that we expect to see GED (see [58]) on the boundary, so we can define our sources in such a way that they transform appropriately. Guided by insights from [23, 24] involving the reduced conformal anomaly—which eventually led to the discovery that dynamical Newton-Cartan geometry corresponds to (non-relativistic) Hořava-Lifshitz (HL) gravity [53]—we expect that the anomaly becomes a Lagrangian describing GED coupled to HL gravity.

## 6.3.1 Defining the Sources

We begin our discussion with a recap of the known sources, which were found in [23, 24, 57]. Writing the metric (6.2.43) in terms of vielbeine,

$$ds^{2} = e^{\Phi} \frac{\mathrm{d}r^{2}}{r^{2}} - E^{0}E^{0} + \delta_{ab}E^{a}E^{b}, \qquad (6.3.1)$$

and performing a Stückelberg decomposition of the massive vector,  $B_{\nu} = A_{\nu} - \partial_{\nu}\Xi$ , we find the sources  $v^{\mu}$ ,  $e^{\mu}_{a}$ ,  $m_{\mu}$ ,  $\phi$ ,  $\chi$ , defined as

$$E_0^{\mu} \simeq -r^2 \alpha_{(0)}^{-1/3} v^{\mu}, \tag{6.3.2}$$

$$E_a^{\mu} \simeq r \alpha_{(0)}^{1/3} e_a^{\mu}, \tag{6.3.3}$$

$$A_{\mu} - \alpha(\Phi) E_{\mu}^{0} \simeq -m_{\mu}, \qquad (6.3.4)$$

$$\Phi \simeq \phi, \tag{6.3.5}$$

$$E \simeq -\chi,$$
 (6.3.6)

$$\varphi \simeq r^2 \alpha_{(0)}^{-1/3} v^{\mu} m_{\mu}, \tag{6.3.7}$$

$$A_a \simeq -r\alpha_{(0)}^{1/3} e_a^{\mu} m_{\mu}, \tag{6.3.8}$$

where (in the notation of chapter 5)

$$\alpha(\Phi) = e^{-3\Phi/2}, \qquad \alpha = \alpha_{(0)} + \dots + r^2 \alpha_{(0)}^{-1/3} \left(\tilde{I} - I\right) + \dots,$$
(6.3.9)

where the dots hide a logarithmic term, and

$$m_{\mu} = M_{\mu} + \partial_{\mu}\chi. \tag{6.3.10}$$

The source  $M_{\mu}$  appears in the expansion of the vielbein  $E^0_{\mu}$  at order  $\mathcal{O}(r^0)$  (see e.g. [33] and our discussion in chapter 5)

$$E^{0}_{\mu} = r^{-2} \alpha^{1/3}_{(0)} \tau_{\mu} + \dots + \alpha^{-1}_{(0)} M_{\mu} + \dots, \qquad (6.3.11)$$

where we have omitted logarithmic terms. Note that the dots hide additional *N*, *G*, *J*-invariant terms at order  $\mathcal{O}(r^0)$ , see section 6.6. Note further that combining the expansion (6.3.11) with the definition of  $m_{\mu}$  (6.3.4) and the expansion for  $\alpha$  in (6.3.9) gives us the relation

$$A_{\mu} = \alpha_{(0)}^{4/3} r^{-2} \tau_{\mu} + \dots - \partial_{\mu} \chi + \dots, \qquad (6.3.12)$$

where, again, the dots hide invariant terms of order  $O(r^0)$ . Now, from a reduction perspective, the Maxwell field consists of two pieces,

$$C_{\mu} = \mathcal{A}_{\mu} - \Upsilon A_{\mu}, \tag{6.3.13}$$

which we see reflected at the solution level. Guided by the expected behaviour of the anomaly discussed in section 6.5—i.e. that the anomaly should become HL gravity coupled to GED—we define a new source  $(a_{\mu}, \tilde{\varphi})$ 

$$C_{\mu} + \Upsilon \alpha(\Phi) E_{\mu}^{0} \simeq a_{\mu} - \tilde{\varphi} \tau_{\mu}, \qquad (6.3.14)$$

where we have performed a decomposition into spatial and temporal directions; that is, in the above we have that  $a_{\mu}v^{\mu} = 0$ . Note that this is completely analogous to the way the source  $m_{\mu}$  appears in (6.3.4).

We can isolate the time component of this source by contracting it with  $E_0^{\mu}$ , giving

$$\Omega := E_0^{\mu} C_{\mu} + \Upsilon \alpha \simeq -r^2 \alpha_{(0)}^{-1/3} \widetilde{\varphi}, \qquad (6.3.15)$$

which is the analogue of (6.3.7). The relation  $E^0_{\mu}E^{\mu}_a = \delta^0_a = 0$  further gives us (when applying  $E^{\mu}_a$  to (6.3.14))

$$C_a \simeq r \alpha_{(0)}^{1/3} a_a.$$
 (6.3.16)

This allows us to identify the fall-off of  $C_{\mu}$  using the behaviour of  $A_{\mu}$  in (6.3.12)

$$C_{\mu} = -\vartheta \alpha_{(0)}^{4/3} \tau_{\mu} r^{-2} + \dots + a_{\mu} - \vartheta \left( m_{\mu} - \partial_{\mu} \chi \right) - \tilde{\varphi} \tau_{\mu} + \dots$$
(6.3.17)

$$= -\vartheta \alpha_{(0)}^{4/3} \tau_{\mu} r^{-2} + \dots + a_{\mu} - \vartheta M_{\mu} - \tilde{\varphi} \tau_{\mu} + \dots$$
(6.3.18)

In summary, our holographic setup involves the following sources

$$E_0^{\mu} \simeq -r^2 \alpha_{(0)}^{-1/3} v^{\mu}, \tag{6.3.19}$$

$$E_a^{\mu} \simeq r \alpha_{(0)}^{1/3} e_a^{\mu}, \tag{6.3.20}$$

$$A_{\mu} - \alpha(\Phi) E^{0}_{\mu} \simeq -m_{\mu},$$
 (6.3.21)

$$\Phi \simeq \phi, \tag{6.3.22}$$

$$\Xi \simeq -\chi, \tag{6.3.23}$$

$$\varphi \simeq r^2 \alpha_{(0)}^{-1/5} v^{\mu} m_{\mu}, \tag{6.3.24}$$

$$Y \simeq \vartheta$$
 (6.3.25)

$$C_{\mu} + \Upsilon \alpha(\Phi) E_{\mu}^{0} \simeq a_{\mu} - \tilde{\varphi} \tau_{\mu}, \qquad (6.3.26)$$

$$E_0^{\nu}C_{\nu} + \Upsilon\alpha(\Phi) \simeq -r^2 \alpha_{(0)}^{-1/3} \widetilde{\varphi}, \qquad (6.3.27)$$

$$C_a \simeq r \alpha_{(0)}^{1/3} a_a,$$
 (6.3.28)

where  $a_{\mu}v^{\mu} = 0$ .

### 6.3.2 Transformation Properties

The known sources (6.7.12)–(6.7.20) transform as follows [24, 57] (see also chapter 5),

$$\delta e^a_\mu = \pounds_{\xi} e^a_\mu + \lambda^a \tau_\mu + \lambda^a{}_b e^b_\mu + \Lambda_D e^a_\mu. \tag{6.3.29}$$

$$\delta\chi = \pounds_{\xi}\chi + \sigma, \tag{6.3.30}$$

$$\delta v^{\mu} = \pounds_{\xi} v^{\mu} + \lambda^a e^{\mu}_a - 2\Lambda_D v^{\mu}, \qquad (6.3.31)$$

$$\delta e_a^{\mu} = \pounds_{\tilde{c}} e_a^{\mu} + \lambda_a^{\ b} e_b^{\mu} - \Lambda_D, \tag{6.3.32}$$

$$\delta \phi = 0, \tag{6.3.33}$$

$$\delta m_{\mu} = \pounds_{\xi} m_{\mu} + e^a_{\mu} \lambda_a + \partial_{\mu} \sigma, \qquad (6.3.34)$$

The rest of this section is devoted to a detailed analysis of the transformation properties of the new sources  $a_{\mu}$ ,  $\tilde{\varphi}$ ,  $\vartheta$ .

### 6.3.2.1 Galilean Boosts and Rotations

 $C_{\mu}$  is invariant under bulk local Lorentz transformations—in particular boosts. These contract to Galilean boosts near the boundary, which means that in a near-boundary expansion, invariance under local Lorentz boosts is translated into invariance under Galilean boosts at each order of *r*. Using the near-boundary expansion of  $C_{\mu}$  (6.3.18), this gives us at order<sup>3</sup>  $\mathcal{O}(r^0)$ 

$$0 = \delta_G \left( a_\mu - \vartheta M_\mu - \tilde{\varphi} \tau_\mu \right) = \delta_G a_\mu - \vartheta \overleftarrow{\delta_G M_\mu} - \tau_\mu \delta_G \tilde{\varphi}.$$
(6.3.35)

<sup>3</sup> Since  $\vartheta$  is a scalar, it transforms trivially,  $\delta_G \vartheta = 0$ .

Thus, see immediately that we must have

$$\delta_{G}a_{\mu} = \vartheta \lambda_{a}e_{\mu}^{a} + \tau_{\mu}\left(\cdot\right), \qquad (6.3.36)$$

where  $(\cdot)$  is so far undetermined. However, the condition  $v^{\mu}a_{\mu} = 0$  provides us with the additional constraint

$$0 = \delta_G(v^{\mu}a_{\mu}) = a_{\nu}\delta_G v^{\nu} + v^{\mu}\delta a_{\mu} = a_{\nu}\lambda^a e_a^{\nu} - (\cdot), \qquad (6.3.37)$$

where we have used the (6.3.36) as well as the orthogonality property  $v^{\mu}e^{a}_{\mu} = 0$ . Thus the transformation of  $a_{\mu}$  becomes

$$\delta_G a_\mu = \vartheta \lambda_a e^a_\mu + \tau_\mu a_\nu e^\nu_a \lambda^a, \qquad (6.3.38)$$

and hence (6.3.35) gives us that

$$\delta_G \tilde{\varphi} = a_\nu e_a^\nu \lambda^a. \tag{6.3.39}$$

For rotations, we do the same as above. We use invariance of the bulk field  $C_{\mu}$  under rotations, i.e.  $\delta_J C_{\mu} = 0$ , which implies that at order  $\mathcal{O}(r^0)$  we have

$$0 = \delta_J \left( a_\mu - \vartheta M_\mu - \tilde{\varphi} \tau_\mu \right) = \delta_J a_\mu - \tau_\mu \delta_G \tilde{\varphi}.$$
(6.3.40)

As with boosts, rotation invariance at order  $O(r^{-2})$  is manifest. Hence, as before,  $\delta_J a_\mu$  will be proportional to  $\tau$ . Using again the orthogonality property  $a_\mu v^\mu = 0$ , we find that  $\delta_J a_\mu = 0$  since  $\delta_J v^\mu = 0$ , leading us to conclude that

$$\delta_J a_\mu = 0 = \delta_J \tilde{\varphi}. \tag{6.3.41}$$

### 6.3.2.2 Gauge Transformations

There are two types of gauge transformations: Stückelberg gauge transformations,  $U_{\Lambda}(1)$ , and gauge transformations of  $C_{\mu} \rightarrow C_{\mu} + \partial_{\mu}\Gamma$ , which we will call  $U_{\Gamma}(1)$ . Starting with  $U_{\Gamma}(1)$ , we recall that  $C_{\mu}$  transforms—by construction—according to

$$\delta_{\Gamma} C_{\mu} = \partial_{\mu} \Gamma, \tag{6.3.42}$$

where  $\Gamma$  has an expansion of the form (reflecting the structure of  $C_{\mu}$  in (6.3.18))

$$\Gamma = \Gamma_{(0)} + \Gamma_{(1,0)} \log r + \Gamma_{(2)} r^2 + \cdots .$$
(6.3.43)

At order  $\mathcal{O}(r^0)$ , the transformation (6.3.42) takes the form

$$\partial_{\mu}\Gamma = \delta_{\Gamma}a_{\mu} - \tau_{\mu}\delta_{\Gamma}\tilde{\varphi}, \qquad (6.3.44)$$

since neither  $\vartheta$  nor  $M_{\mu}$  transforms under  $U_{\Gamma}(1)$ . Combining this with the orthogonality relation,

$$0 = \delta_{\Gamma}(v^{\mu}a_{\mu}) = v^{\mu}\delta_{\Gamma}a_{\mu}, \tag{6.3.45}$$

we see that (6.3.44) implies that  $\delta_{\Gamma}a_{\mu}$  also involves a term proportional to  $\tau$ , which we can determine using (6.3.45), giving

$$\delta_{\Gamma}a_{\mu} = \partial_{\mu}\Gamma_{(0)} + \tau_{\mu}v^{\nu}\partial_{\nu}\Gamma_{(0)}, \qquad \delta_{\Gamma}\tilde{\varphi} = v^{\mu}\partial_{\mu}\Gamma_{(0)}, \tag{6.3.46}$$

On the boundary  $r \to 0$ , the gauge group is truncated, i.e.  $U_{\Gamma}(1) \to U_{\Gamma_{(0)}}(1)$ , and, similarly, for Stückelberg gauge transformations, we obtain the group  $U_{\Lambda}(1) \to U_{\sigma}(1)$ , where  $\sigma$  is (minus) the boundary value of  $\Lambda$ .

Repeating the calculation for  $\delta_{\Lambda}$  transformations, we obtain immediately that on the boundary (where  $\delta_{\Lambda} \rightarrow \delta_{\sigma}$ ),

$$\delta_{\sigma}a_{\mu} = 0, \qquad \delta_{\sigma}\tilde{\varphi} = 0, \qquad \delta_{\sigma}\vartheta = 0.$$
 (6.3.47)

### 6.3.2.3 PBH Transformations: Diffeomorphisms, Dilatations & Stückelberg Gauge Transformations

The PBH transformations in the electromagnetic uplift translate into boundary diffeomorphisms, Weyl transformations, and Stückelberg gauge transformations. As we saw in section 5.2.3 (see also appendix A of [24]), the PBH transformations are the diffeomorphisms preserving the radial gauge choice of the FG expansion, i.e. diffeomorphisms acting infinitesimally on the five-dimensional fields in the manner

$$\delta_{\hat{\ell}}\mathcal{G}_{\mathcal{MN}} = \pounds_{\hat{\ell}}\mathcal{G}_{\mathcal{MN}},\tag{6.3.48}$$

$$\delta_{\hat{\zeta}}\mathcal{A}_{\mathcal{M}} = \pounds_{\hat{\zeta}}\mathcal{A}_{\mathcal{M}},\tag{6.3.49}$$

$$\delta_{\hat{\ell}}\psi = \pounds_{\hat{\ell}}\psi, \tag{6.3.50}$$

and satisfying  $\pounds_{\hat{c}} \mathcal{G}_{rr} = 0 = \pounds_{\hat{c}} \mathcal{G}_{\mathcal{M}r}$ . Solving these equations gives

$$\hat{\zeta}^r \simeq -r\Lambda_D, \qquad \hat{\zeta}^A \simeq \hat{\zeta}^A, \tag{6.3.51}$$

where  $\hat{\xi}$  and  $\Lambda_D$  are independent of *r*. This leads to the transformation properties

$$\delta_{\hat{\xi}}\gamma_{(0)AB} = \hat{\xi}^C \partial_C \gamma_{(0)AB} + \gamma_{(0)CB} \partial_A \hat{\xi}^C + \gamma_{(0)AC} \partial_B \hat{\xi}^C + 2\Lambda_D \gamma_{(0)AB}, \qquad (6.3.52)$$

$$\delta_{\hat{\xi}}\psi_{(0)} = \hat{\xi}^A \partial_A \psi_{(0)}, \tag{6.3.53}$$

$$\delta_{\hat{\zeta}}\mathcal{A}_{(0)A} = \hat{\zeta}^B \partial_B \mathcal{A}_{(0)A} + \mathcal{A}_{(0)B} \partial_A \hat{\zeta}^B.$$
(6.3.54)

In the reduced theory, we take the diffeomorphisms to be independent of the compact direction. Writing  $\hat{\xi}^{\mu} = \xi^{\mu}$  and  $-\sigma$  for the *u*-component of  $\xi$ , we get diffeomorphisms, Weyl transformations and Stückelberg gauge transformations in exactly the same fashion as in section 5.2.3.

## 6.3.2.4 Residual Transformations

In addition to the diffeomorphisms and gauge transformations (6.2.12) in the reduced theory, the reduction is also preserved under the *u*-dependent transformations  $\hat{\zeta}^u = cu$  and  $\hat{\Gamma} = \ell u$ , where  $c, \ell$  are constants. These are remnants of full symmetry in the electromagnetic uplift and are only preserved to leading order. We will refer to them as  $\delta_c$  and  $\delta_\ell$ , respectively. In [24],  $\delta_c$  was called *local dilatations*, and the relevant field transformation reads

$$\delta_c \phi = c. \tag{6.3.55}$$

The GED scalar transforms undre  $\delta_\ell$  transformations according to

$$\delta_{\ell}\vartheta = \ell. \tag{6.3.56}$$

Note that when solving the equations of motion in the reduced theory, as we discussed in section 6.2.4, the  $\delta_{\ell}$  symmetry manifests itself by leaving  $\Upsilon_{(0)}$ —which is assumed to be constant—undetermined by the equations of motion.

### 6.3.2.5 Summary of Local Transformations

Below, we present our results for the transformations of all our sources

$$\delta e^a_\mu = \pounds_{\xi} e^a_\mu + \lambda^a \tau_\mu + \lambda^a{}_b e^b_\mu + \Lambda_D e^a_\mu. \tag{6.3.57}$$

$$\chi = \pounds_{\xi} \chi + \sigma, \tag{6.3.58}$$

$$\delta v^{\mu} = \pounds_{\tilde{\zeta}} v^{\mu} + \lambda^a e^{\mu}_a - 2\Lambda_D v^{\mu}, \qquad (6.3.59)$$

$$\delta e_a^{\mu} = \pounds_{\xi} e_a^{\mu} + \lambda_a^{\ b} e_b^{\mu} - \Lambda_D e_a^{\mu}, \qquad (6.3.60)$$

$$\delta \phi = \pounds_{\xi} \phi, \tag{6.3.61}$$

$$\delta m_{\mu} = \pounds_{\xi} m_{\mu} + e^a_{\mu} \lambda_a + \partial_{\mu} \sigma, \qquad (6.3.62)$$

$$\delta \vartheta = f_{\xi} \vartheta, \qquad (6.3.63)$$

$$\delta \tilde{\vartheta} = f_{z} \tilde{\vartheta} + g_{z} \lambda^{a} + \eta^{\mu} \partial_{z} \Gamma_{(z)} = 2 \Lambda_{D} \tilde{\vartheta} \qquad (6.3.64)$$

$$\delta\varphi = \pounds_{\xi}\varphi + a_a\lambda^a + v^{\mu}\partial_{\mu}\Gamma_{(0)} - 2\Lambda_D\varphi, \qquad (6.3.64)$$

$$\delta a_{\mu} = f_{\pi}a_{\mu} + \vartheta\lambda_{\nu}e^a + \tau_{\nu}a_{\nu}e^{\nu}\lambda^a + \partial_{\nu}\Gamma_{(0)} + \tau_{\nu}v^{\nu}\partial_{\nu}\Gamma_{(0)} \qquad (6.2.65)$$

$$\delta a_{\mu} = \mathcal{L}_{\xi} a_{\mu} + \vartheta \Lambda_{a} e_{\mu}^{*} + \tau_{\mu} a_{\nu} e_{a}^{*} \Lambda^{*} + \partial_{\mu} \mathbf{1}_{(0)} + \tau_{\mu} \vartheta^{*} \partial_{\nu} \mathbf{1}_{(0)}, \qquad (6.3.65)$$

$$\delta a_a = \pounds_{\xi} a_a + \vartheta \lambda_a + \lambda_a{}^{\nu} a_b - \Lambda_D a_a + e_a^{\mu} \partial_{\mu} \Gamma_{(0)}.$$
(6.3.66)

Note that the sources  $a_{\mu}$ ,  $\vartheta$ ,  $\tilde{\varphi}$  precisely transform as GED fields (6.1.16). We also see that the symmetry group is the direct product group

$$Diff(\mathcal{M}) \times Sch_2(2,1) \times U(1). \tag{6.3.67}$$

Under dilatations, we have

Source	φ	$ au_{\mu}$	$e^a_\mu$	$v^{\mu}$	$e^{\mu}_{a}$	$m_0$	m <sub>a</sub>	χ	a <sub>a</sub>	$\tilde{\varphi}$	θ
scaling dim.	0	-2	-1	2	1	2	1	0	1	2	0

Table 6.5: The sources and their scaling dimensions

We note that for the new sources, this behaviour precisely agrees with the scaling weights determined in the field theory in table 6.3 when putting z = 2.

## 6.4 BOUNDARY GEOMETRY & $\mathfrak{sch}_2(2,1)$

### 6.4.1 Null Reduction on the Boundary

In this section, we show how the Scherk-Schwarz reduction employed in section (6.2) becomes a null reduction—as considered in section 4.3—on the boundary. Recall that

$$A_{\mu} - \alpha(\Phi) E^{0}_{\mu} \simeq -m_{\mu}, \qquad B_{\mu} - \alpha(\Phi) E^{0}_{\mu} \simeq -M_{\mu},$$
 (6.4.1)

where  $B_{\mu} = A_{\mu} - \partial_{\mu} \Xi$ , which implies that

$$(B_{\mu} + \partial_{\mu}\Xi) - \alpha(\Phi)E^{0}_{\mu} \simeq -M_{\mu} - \partial_{\mu}\chi \iff A_{\mu} - \alpha(\Phi)E^{0}_{\mu} \simeq -(M_{\mu} + \partial_{\mu}\chi), \qquad (6.4.2)$$

where, explicitly,

$$m_{\mu} = M_{\mu} + \partial_{\mu}\chi, \quad \text{or} \quad M_{\mu} = m_{\mu} - \partial_{\mu}\chi.$$
 (6.4.3)

The reduction ansatz can be written in the form

$$ds_5^2 = \frac{dr^2}{r^2} + \gamma_{AB} dx^A dx^B = e^{-\Phi} \left[ e^{\Phi} \frac{dr^2}{r^2} + \tilde{h}_{\mu\nu} dx^{\mu} dx^{\nu} \right] + e^{2\Phi} (du + A_{\mu} dx^{\mu})^2.$$
(6.4.4)

where the metric  $\gamma_{AB}$  on leaves of constant *r* can be written in Fefferman-Graham form (see appendices A and ??); in particular, the boundary is described by the leading term  $\gamma_{(0)AB}$  which has dilatation weight 2, i.e.

$$\gamma_{AB} \simeq r^{-2} \gamma_{(0)AB}.$$
 (6.4.5)

The four-dimensional part in (C.1.20)—i.e. the quantity in square brackets—can be written in terms of vielbeine in the manner depicted in (6.3.1). We now dissect the ansatz (C.1.20) and write it piece by piece in terms of sources via the near-boundary expansions of the bulk fields worked out in section 6.3. Using the Stückelberg decomposition of the mass gauge field (6.4.3), the first piece of (C.1.20) becomes

$$-e^{-\Phi}E^{0}_{\mu}E^{0}_{\nu}dx^{\mu}dx^{\nu} = -\alpha^{2/3}_{(0)}\left(r^{-2}\alpha^{1/3}_{(0)}\tau_{\mu} + \alpha^{-1}_{(0)}M_{\mu} + \cdots\right)\left(r^{-2}\alpha^{1/3}_{(0)}\tau_{\nu} + \alpha^{-1}_{(0)}M_{\nu} + \cdots\right)dx^{\mu}dx^{\nu}$$
(6.4.6)

$$= \left(-\alpha_{(0)}^{4/3}r^{-4}\tau_{\mu}\tau_{\nu}dx - 2r^{-2}\tau_{(\mu}M_{\nu)} - \alpha_{(0)}^{-1/3}M_{\mu}M_{\nu}\right)dx^{\mu}dx^{\nu} + \cdots, \qquad (6.4.7)$$

as well

$$e^{-\Phi}\delta_{ab}E^{a}_{\mu}E^{b}_{\nu}\mathrm{d}x^{\mu}\mathrm{d}x^{\nu} = \alpha^{2/3}_{(0)}\delta_{ab}r^{-2}\alpha^{-2/3}_{(0)}e^{a}_{\mu}e^{b}_{\nu}\mathrm{d}x^{\mu}\mathrm{d}x^{\nu} = r^{-2}e_{a\mu}a^{a}_{\nu} =: r^{-2}h_{\mu\nu}, \tag{6.4.8}$$

whereas

$$e^{2\Phi}A_{\mu}A_{\nu}\mathrm{d}x^{\mu}\mathrm{d}x^{\nu} = e^{2\Phi}\left(B_{\mu} + \partial_{\mu}\Xi\right)\left(B_{\nu} + \partial_{\mu}\Xi\right)\mathrm{d}x^{\mu}\mathrm{d}x^{\nu}$$

$$(6.4.9)$$

$$= \alpha_{(0)}^{-4/3} \left( \alpha_{(0)}^{4/3} r^{-2} \tau_{\mu} - \partial_{\mu} \chi \right) \left( \alpha_{(0)}^{4/3} r^{-2} \tau_{\nu} - \partial_{\nu} \chi \right) dx^{\mu} dx^{\nu} + \cdots$$
(6.4.10)

$$= \alpha_{(0)}^{4/3} r^{-4} \tau_{\mu} \tau_{\nu} \mathrm{d}x^{\mu} \mathrm{d}x^{\nu} - 2r^{-2} \partial_{\mu} \chi \tau_{\nu} \mathrm{d}x^{\mu} \mathrm{d}x^{\nu} + \alpha_{(0)}^{-4/3} \partial_{\mu} \chi \partial_{\nu} \chi \mathrm{d}x^{\mu} \mathrm{d}x^{\nu} + \cdots$$
(6.4.11)

The cross term reduces to

$$2e^{2\Phi}A_{\mu}dudx^{\mu} = 2e^{2\Phi}\left(B_{\mu} - \partial_{\mu}\Xi\right)dudx^{\mu}$$
(6.4.12)

$$= 2\alpha_{(0)}^{-4/3} \left( \alpha_{(0)}^{4/3} r^{-2} \tau_{\mu} + \partial_{\mu} \chi \right) du dx^{\mu} + \cdots$$
(6.4.13)

$$=2r^{-2}\tau_{\mu}dudx^{\mu}+2\alpha_{(0)}^{-4/3}\partial_{\mu}\chi dudx^{\mu}+\cdots, \qquad (6.4.14)$$

and, finally,

$$e^{2\Phi} \mathrm{d}u^2 = \alpha_{(0)}^{-4/3} \mathrm{d}u^2 + \cdots,$$
 (6.4.15)

which does not contribute at order  $r^{-2}$ . Now, collecting all terms of order  $r^{-4}$ , we see that they cancel (cf. equations (6.4.7) and (6.4.11)). The boundary metric  $\gamma_{(0)AB}$  is consequently given by

$$ds^{2} = \gamma_{(0)AB} dx^{A} dx^{B} = -2\tau_{\mu} M_{\nu} dx^{\mu} dx^{\nu} - 2\partial_{\mu} \chi \tau_{\nu} dx^{\mu} dx^{\nu} + 2\tau_{\mu} du dx^{\mu} + h_{\mu\nu} dx^{\mu} dx^{\nu} \quad (6.4.16)$$

$$= 2\tau_{\mu} dx^{\mu} \left( du - (M_{\nu} + \partial_{\nu} \chi) dx^{\nu} \right) + h_{\mu\nu} dx^{\mu} dx^{\nu}$$
(6.4.17)

$$= 2\tau_{\mu} dx^{\mu} (du - m_{\nu} dx^{\nu}) + h_{\mu\nu} dx^{\mu} dx^{\nu}, \qquad (6.4.18)$$

which is identical to the null reduction ansatz employed in section 4.3, which was shown to give rise to TNC geometry: thus we see directly that the boundary geometry is TNC.

## 6.4.1.1 Null Reduction of the Connection

We will now consider explicitly the consequences of the null reduction for the affine connection  $\mathring{\Gamma}^{C}_{AB}$ , where we use a 'o' to denote that this is the affine connection of the metric  $\gamma_{(0)AB}$ . Taking the three legs in non-compact directions, we find that

$$\mathring{\Gamma}^{\lambda}_{\mu\nu} = \frac{1}{2} h^{\lambda\rho} \left( \partial_{\mu} \bar{h}_{\nu\lambda} + \partial_{\nu} \bar{h}_{\mu\lambda} - \partial_{\lambda} \bar{h}_{\mu\nu} \right) - \vartheta^{\lambda} \partial_{(\mu} \tau_{\nu)}$$
(6.4.19)

$$=\overline{\Gamma}^{\lambda}_{\mu\nu} + \hat{v}^{\lambda}\partial_{[\mu}\tau_{\nu]} \tag{6.4.20}$$

$$=\overline{\Gamma}^{\lambda}_{\mu\nu}-\overline{\Gamma}^{\lambda}_{[\mu\nu]},\tag{6.4.21}$$

where we define

$$\overline{\Gamma}^{\lambda}_{\mu\nu} = -\hat{v}^{\lambda}\partial_{\mu}\tau_{\nu} + \frac{1}{2}h^{\lambda\rho}\left(\partial_{\mu}\bar{h}_{\nu\lambda} + \partial_{\nu}\bar{h}_{\mu\lambda} - \partial_{\lambda}\bar{h}_{\mu\nu}\right), \qquad (6.4.22)$$

which is the unique TNC metric compatible connection linear in  $m_{\mu}$  [33] that we also discussed in chapter 4. It is (*G*, *J*) invariant and satisfies metric compatibility in the usual sense,

$$\nabla_{\mu}\tau_{\nu} = 0, \qquad \nabla_{\mu}h^{\nu\lambda} = 0. \tag{6.4.23}$$

It can be made N invariant by replacing  $m_{\mu}$  with  $M_{\mu}$ . The other components of  $\mathring{\Gamma}^{A}_{BC}$  read

$$\mathring{\Gamma}^{u}_{\mu\nu} = -\frac{1}{2} \hat{v}^{\lambda} \left( \partial_{\mu} \bar{h}_{\lambda\nu} + \partial_{\nu} \bar{h}_{\lambda\mu} - \partial_{\lambda} \bar{h}_{\mu\nu} \right) + 2 \tilde{\Phi} \partial_{(\mu} \tau_{\nu)}, \qquad (6.4.24)$$

$$\mathring{\Gamma}^{\lambda}_{\mu\mu} = h^{\lambda\rho} \partial_{(\mu} \tau_{\rho)}, \tag{6.4.25}$$

$$\mathring{\Gamma}^{u}_{\mu\mu} = -\hat{v}^{\lambda}\partial_{[\mu}\tau_{\lambda]}, \qquad (6.4.26)$$

$$\mathring{\Gamma}^{u}_{uu} = 0 = \mathring{\Gamma}^{\mu}_{uu}. \tag{6.4.27}$$

6.4.2 Gauging  $\mathfrak{sch}_2(2,1)$ 

Complementary to our discussion of TNC geometry in chapter 4 and the emergence of TNC geometry in section 5.2.4 for generic values of  $z \in (1, 2]$ , we discuss in this section how TTNC explicitly occurs when gauging  $\mathfrak{sch}_2(d, 1)$ , which was first shown in [34]. In order to realize the more general TNC geometry, torsion has to be added by hand in this approach. For an overview of Schrödinger algebras and their rôle as Newton-Cartan space-time symmetry groups, see e.g. [115, 125].

For z = 2, it is possible to add an additional special conformal transformation *K*, which cannot be done in the general-*z* case. To keep the analysis as general as possible, we do that in what follows. As such,  $\mathfrak{sch}_2(2,1)$  is generated by: *D* (dilatations) and *K* (special conformal transformations), in

addition to those of the Bargmann algebra, bar(2, 1), which is generated by H (time translations), P (spatial translations), G (Galilean boosts),  $J_{ab}$  (spatial rotations) and N (central charge). The non-zero commutation relations of  $\mathfrak{sch}_2(2, 1)$  read

$$[D,H] = -2H, \quad [H,K] = D, \quad [D,K] = 2K, \quad [H,G_a] = P_a$$
(6.4.28)

$$[D, P_a] = -P_a, \quad [D, G_a] = G_a, \quad [K, P_a] = -G_a, \quad [P_a, G_b] = \delta_{ab}N, \tag{6.4.29}$$

$$[J_{ab}, P_c] = 2\delta_{c[a}P_{b]}, \quad [J_{ab}, G_c] = 2\delta_{c[a}G_{b]}, \quad [J_{ab}, J_{cd}] = 4\delta_{[a[d}J_{c]b]}.$$
(6.4.30)

For z = 2, the Bargmann central charge *N* is still a central element. Further, we note in passing that it can be shown [115] that  $\mathfrak{sch}_2(2,1)$  admits the following Levi decomposition,

$$\mathfrak{sch}_2(2,1) \cong (\mathfrak{so}(3) \times \mathfrak{sl}(2,\mathbb{R})) \ltimes (\mathbb{R}^3 \times \mathbb{R}^3).$$
 (6.4.31)

We will use the  $\bar{\delta}$ -transformation approach discussed in chapter 4, where contact with diffeomorphisms is manifest; this was not done in [34]. As usual, we start by defining the Lie algebra valued connection

$$\mathfrak{A}_{\mu} = H\tau_{\mu} + P_{a}e_{\mu}^{a} + G_{a}\omega_{\mu}^{\ a} + \frac{1}{2}J_{ab}\omega_{\mu}^{\ ab} + Nm_{\mu} + Db_{\mu} + Kf_{\mu}.$$
(6.4.32)

For  $L(x) \in Sch_2(d, 1)$ , the connection transforms in the adjoint,  $\mathfrak{A}_{\mu} \to L(x)\mathfrak{A}_{\mu}L^{-1}(x) - L(x)\partial_{\mu}L^{-1}(x)$ , so taking L(x) to be infinitesimal,  $L(x) = \mathbb{1} + \Lambda(x)$  with  $\Lambda(x) \in \mathfrak{sch}_2(d, 1)$ , we get

$$\delta\mathfrak{A}_{\mu} = \partial_{\mu}\Lambda + [\mathfrak{A}_{\mu}, \Lambda], \tag{6.4.33}$$

but, since  $\Lambda(x) \in \mathfrak{sch}_2(d, 1)$ , we may write

$$\Lambda(x) = H\zeta(x) + P_a\zeta^a(x) + G_a\tilde{\lambda}^a(x) + \frac{1}{2}J_{ab}\tilde{\lambda}^{ab}(x) + N\tilde{\sigma}(x) + D\tilde{\Lambda}_D(x) + K\tilde{\Lambda}_K(x).$$
(6.4.34)

For the  $\bar{\delta}$ -transformation, we replace the local translation parameters  $\zeta^a$  in  $\Lambda$  with a space-time vector  $\xi^{\mu}$  defined via  $\zeta^a = \xi^{\mu} e^a_{\mu}$ , which allows us to write

$$\Lambda = \xi^{\mu} \mathfrak{A}_{\mu} + \Sigma, \tag{6.4.35}$$

which implies that, as we have seen

$$\Sigma = G_a \lambda^a + \frac{1}{2} J_{ab} \lambda^{ab} + N\sigma + D\Lambda_D + K\Lambda_K, \qquad (6.4.36)$$

where the tilde-less parameters of (6.4.36) are related to their tilded counterparts in (6.4.34) in the same as in (4.1.11). The  $\bar{\delta}$ -transformation then becomes

$$\bar{\delta}\mathfrak{A}_{\mu} = \delta\mathfrak{A}_{\mu} - \xi^{\nu}\mathcal{F}_{\mu\nu} = \pounds_{\xi}\mathfrak{A}_{\mu} + \partial_{\mu}\Sigma + [\mathfrak{A}_{\mu}, \Sigma], \qquad (6.4.37)$$

where  $\mathcal{F}_{\mu\nu}$  is the Yang-Mills curvature,

$$\mathcal{F}_{\mu\nu} = 2\partial_{[\mu}\mathcal{A}_{\nu]} + [\mathfrak{A}_{\mu}, \mathfrak{A}_{\nu}]$$

$$= HR_{\mu\nu}(H) + P_{a}R_{\mu\nu}{}^{a}(P) + G_{a}R_{\mu\nu}{}^{a}(G) + \frac{1}{2}J_{ab}R_{\mu\nu}{}^{ab}(J) + NR_{\mu\nu}(N) + DR_{\mu\nu}(D) + KR_{\mu\nu}(K).$$
(6.4.39)
(6.4.39)

We summarise this construction in the table below:

Table 6.6: Generators of  $\mathfrak{sch}_2(2,1)$  with their associated gauge fields, local parameters and covariant curvatures.

Symmetry	Generators	Gauge Field	Parameters	Curvatures
Time translations	Н	$ au_{\mu}$	$\zeta(x)$	$R_{\mu u}(H)$
Spatial translations	$P_a$	$e^a_\mu$	$\zeta^a(x)$	$R_{\mu\nu}^{a}(P)$
Boosts	Ga	$\omega_{\mu}^{a}$	$\lambda^a(x)$	$R_{\mu\nu}^{a}(G)$
Spatial rotations	J <sub>ab</sub>	$\omega_{\mu}{}^{ab}$	$\lambda^{ab}(x)$	$R_{\mu\nu}^{ab}(J)$
Central ch. traf'os	Ν	$m_{\mu}$	$\sigma(x)$	$R_{\mu\nu}(N)$
Dilatations	D	$b_{\mu}$	$\Lambda_D(x)$	$R_{\mu\nu}(D)$
Spec. conf. traf'os	K	$f_{\mu}$	$\Lambda_K(x)$	$R_{\mu\nu}(K)$

Writing out the expression of (6.4.38) for  $\bar{\delta}\mathfrak{A}_{\mu}$  using our expressions for  $\mathfrak{A}_{\mu}$  and  $\Sigma$ , we find the following expressions for the curvatures:

$$R_{\mu\nu}(H) = 2\partial_{[\mu}\tau_{\nu]} - 4b_{[\mu}\tau_{\nu]}, \qquad (6.4.40)$$

$$R_{\mu\nu}{}^{a}(P) = 2\partial_{[\mu}e^{a}_{\nu]} - 2\omega_{[\mu}{}^{ab}e_{\nu]b} - 2\omega_{[\mu}{}^{a}\tau_{\nu]} - 2b_{[\mu}e^{a}_{\nu]}, \qquad (6.4.41)$$

$$R_{\mu\nu}{}^{ab}(J) = 2\partial_{[\mu}\omega_{\nu]}{}^{ab} - 2\omega_{[\mu}{}^{c[a}\omega_{\nu]}{}^{b]}{}^{c}, \qquad (6.4.42)$$

$$R_{\mu\nu}{}^{a}(G) = 2\partial_{[\mu}\omega_{\nu]}{}^{a} + 2\omega_{[\mu}{}^{b}\omega_{\nu]}{}^{a}{}_{b} - 2\omega_{[\mu}{}^{a}b_{\nu]} - 2f_{[\mu}e_{\nu]}^{a}, \qquad (6.4.43)$$

$$R_{\mu\nu}(D) = 2\partial_{[\mu}b_{\nu]} - 2f_{[\mu}\tau_{\nu]}, \qquad (6.4.44)$$

$$R_{\mu\nu}(K) = 2\partial_{[\mu}f_{\nu]} + 4b_{[\mu}f_{\nu]}, \qquad (6.4.45)$$

$$R_{\mu\nu}(N) = 2\partial_{[\mu}m_{\nu]} - 2\omega_{[\mu}{}^{a}e_{\nu]a}.$$
(6.4.46)

Similarly, by writing out  $\mathfrak{A}_{\mu}$  and  $\Sigma$  as they appear in (6.4.37) and identifying coefficients in front of the generators, we find the following transformations of the gauge fields:

$$\delta\tau_{\mu} = \pounds_{\xi}\tau_{\mu} + 2\Lambda_D\tau_{\mu},\tag{6.4.47}$$

$$\bar{\delta}e^a_\mu = \pounds_{\xi}e^a_\mu + \lambda^a \tau_\mu + \Lambda_D e^a_\mu, \tag{6.4.48}$$

$$\bar{\delta}\omega_{\mu}{}^{ab} = \pounds_{\tilde{\zeta}}\omega_{\mu}{}^{ab} + \partial_{\mu}\lambda^{ab} + 2\lambda^{c[a}\omega_{\mu}{}^{b]}{}^{c]}{}^{c}, \qquad (6.4.49)$$

$$\bar{\delta}\omega_{\mu}{}^{a} = \pounds_{\xi}\omega_{\mu}{}^{a} + \partial_{\mu}\lambda^{a} - \lambda^{b}\omega_{\mu}{}^{a}{}_{b} + \lambda^{a}{}_{b}\omega_{\mu}{}^{b} + \lambda^{a}b_{\mu} - \Lambda_{D}\omega_{\mu}{}^{a} + \Lambda_{K}e^{a}_{\mu}, \qquad (6.4.50)$$

$$\delta b_{\mu} = \pounds_{\zeta} b_{\mu} + \partial_{\mu} \Lambda_D + \Lambda_K \tau_{\mu}, \tag{6.4.51}$$

$$\bar{\delta}f_{\mu} = \pounds_{\xi}f_{\mu} + \partial_{\mu}\Lambda_{K} + 2\Lambda_{K}b_{\mu} - 2\Lambda_{D}f_{\mu}, \qquad (6.4.52)$$

$$\bar{\delta}m_{\mu} = \pounds_{\xi}m_{\mu} + \partial_{\mu}\sigma + \lambda^{a}e_{\mu a}. \tag{6.4.53}$$

The gauge fields  $\tau_{\mu}$  and  $e^{a}_{\mu}$  transform under spatial rotations and Galilean boosts as Newton-Cartan temporal and spatial vielbeine, respectively [54], and we identify them as such. Since they are of rank 1 and rank *d*, respectively, in a (d + 1)-dimensional spacetime, they are not invertible, but we can define projective inverses  $v^{\mu}$  and  $e^{\mu}_{a}$ , satisfying the usual set of relations

$$v^{\mu}\tau_{\mu} = -1, \quad v^{\mu}e^{a}_{\mu} = 0, \quad \tau_{\mu}e^{\mu}_{a} = 0, \quad e^{a}_{\mu}e^{\mu}_{b} = \delta^{a}_{b}, \quad e^{\mu}_{a}e^{a}_{\nu} = \delta^{\mu}_{\nu} + v^{\mu}\tau_{\nu}.$$
 (6.4.54)

The projective inverses transform under  $\bar{\delta}$ -transformations in the following manner,

$$\bar{\delta}v^{\mu} = \pounds_{\xi}v^{\mu} + \lambda^{a}e^{\mu}_{a} - 2\Lambda_{D}v^{\mu}, \qquad (6.4.55)$$

$$\bar{\delta}e^{\mu}_{a} = \pounds_{\xi}e^{\mu}_{a} + \lambda_{a}^{\ b}e^{\mu}_{b} - \Lambda_{D}e^{\mu}_{a}, \qquad (6.4.56)$$

which are derived by considering the relations  $0 = \bar{\delta} (v^{\mu} \tau_{\mu})$  and  $\bar{\delta} (v^{\mu} e^{a}_{\mu})$  and using the identities (6.4.54). So far, all the gauge fields are independent. This can be remedied via the imposition of certain curvature constraints—as detailed in [34]—which allows us to make contact with TTNC. These constraints read

$$0 = R_{\mu\nu}(H) = R_{\mu\nu}{}^{a}(P) = R_{\mu\nu}(N) = R_{\mu\nu}(D) = R_{0a}{}^{a}(G) + 2m^{b}R_{0a}{}^{a}{}_{b}(J) + m^{b}m^{c}R_{ba}{}^{a}{}_{c}(J).$$
(6.4.57)

These constraints—along with some additional constraints coming from imposing the curvature constraints in the Bianchi identities (see [34] for details)—leave us with the independent fields  $\tau_{\mu}$ ,  $e^a_{\mu}$  and  $v^{\mu}b_{\mu}$ , which we recognize as the usual fields of TNC geometry except for the odd one out:  $v^{\mu}b_{\mu}$ . However, note that under local transformations, this object transforms in the following manner

$$\bar{\delta}(v^{\mu}b_{\mu}) = \pounds_{\xi}(v^{\mu}b_{\mu}) + v^{\mu}\partial_{\mu}\Lambda_D - 2\Lambda_D v^{\mu}b_{\mu} + \lambda^a e^{\mu}_a b_{\mu} - \Lambda_K, \qquad (6.4.58)$$

leading us to identify the combination  $v^{\mu}b_{\mu}$  as a Stückelberg field for *K* transformations. This prompts us to gauge fix it to  $v^{\mu}b_{\mu} = 0$ , thereby introducing the following *compensating special conformal transformation* 

$$\Lambda_K = v^\mu \partial_\mu \Lambda_D + \lambda^a e^\mu_a b_\mu. \tag{6.4.59}$$

The next step in the gauging procedure is to write down covariant derivatives, which are covariant with respect to the transformations (6.4.47)–(6.4.53). By taking into account all the commutators of

 $\mathfrak{sch}_2(d, 1)$  that are proportional to *H* and *P*<sub>a</sub>, we find the following expressions for the covariant derivatives:

$$\mathcal{D}_{\mu}\tau_{\nu} = \partial_{\mu}\tau_{\nu} - \tilde{\Gamma}^{\rho}_{\mu\nu}\tau_{\rho} - 2b_{\mu}\tau_{\nu}, \qquad (6.4.60)$$

$$\mathcal{D}_{\mu}e_{\nu}^{a} = \partial_{\mu}e_{\nu}^{a} - \tilde{\Gamma}_{\mu\nu}^{\rho}e_{\rho}^{a} - \omega_{\mu}{}^{a}{}_{b}e_{\nu}^{b} - \omega_{\mu}{}^{a}\tau_{\nu} - b_{\mu}e_{\nu}^{a}.$$
(6.4.61)

Imposing the vielbein postulates,  $\mathcal{D}_{\mu}\tau_{\nu} = 0 = \mathcal{D}_{\mu}e_{\nu}^{a}$  allows for a identification of the curvatures (6.4.40) and (6.4.41) in terms of the affine connection. Since the postulates imply that

$$\tilde{\Gamma}^{\rho}_{\mu\nu}\tau_{\rho} = \partial_{\mu}\tau_{\nu} - 2b_{\mu}\tau_{\nu}, \qquad \tilde{\Gamma}^{\rho}_{\mu\nu}e^{a}_{\rho} = \partial_{\mu}e^{a}_{\nu} - \omega_{\mu}{}^{a}{}_{b}e^{a}_{\nu} - \omega_{\mu}{}^{a}\tau_{\nu} - b_{\mu}e^{a}_{\nu}, \qquad (6.4.62)$$

we immediately obtain

$$R_{\mu\nu}(H) = 2\Gamma^{\rho}_{[\mu\nu]}\tau_{\rho}, \qquad R^{a}_{\mu\nu}(P) = 2\Gamma^{\rho}_{[\mu\nu]}e^{a}_{\rho}, \qquad (6.4.63)$$

implying, via the curvature constraints (6.4.57), that  $\tilde{\Gamma}^{\rho}_{\mu\nu}$  is symmetric. It can be shown that it is uniquely determined by [34]

$$\tilde{\Gamma}^{\rho}_{\mu\nu} = -\hat{v}^{\rho} D^{(b)}_{\mu} \tau_{\nu} + \frac{1}{2} h^{\rho\sigma} \left( D^{(b)}_{\mu} \bar{h}_{\nu\sigma} + D^{(b)}_{\nu} \bar{h}_{\mu\sigma} - D^{(b)}_{\sigma} \bar{h}_{\mu\nu} \right),$$
(6.4.64)

where  $D_{\mu}^{(b)} = \partial_{\mu} + 2b_{\mu}$  is the dilatation covariant derivative. Dropping all terms involving  $b_{\mu}$ , we recover the minimal TNC connection,

$$\overline{\Gamma}^{\rho}_{\mu\nu} = -\hat{v}^{\rho}\partial_{\mu}\tau_{\nu} + \frac{1}{2}h^{\rho\sigma}\left(\partial_{\mu}\bar{h}_{\nu\rho} + \partial_{\nu}\bar{h}_{\mu\rho} - \partial_{\rho}\bar{h}_{\mu\nu}\right), \qquad (6.4.65)$$

with torsion

$$\overline{\Gamma}^{\rho}_{[\mu\nu]} = -\vartheta^{\rho}\partial_{[\mu}\tau_{\nu]}. \tag{6.4.66}$$

This intimate relationship between torsion and dilatation covariance leads us—following [34]—to attribute torsion to the gauge field  $b_{\mu}$  in the sense above.

The curvature constraint  $R_{\mu\nu}(H)$  implies that

$$\partial_{[\mu}\tau_{\nu]} = 2b_{[\mu}\tau_{\nu]},$$
 (6.4.67)

which means that

$$\tau \wedge d\tau = \tau_{[\rho} \partial_{\rho} \tau_{\nu]} = \frac{1}{3} \left( \tau_{\mu} \partial_{[\rho} \tau_{\nu]} + \tau_{\rho} \partial_{[\nu} \tau_{\mu]} + \tau_{\nu} \partial_{[\mu} \tau_{\rho]} \right) = \frac{2}{3} \left( b_{\rho} \tau_{[\mu} \tau_{\nu]} + b_{\mu} \tau_{[\rho} \tau_{\nu]} + b_{\nu} \tau_{[\mu} \tau_{\rho]} \right) = 0,$$
(6.4.68)

implying, by Frobenius' theorem, that  $\tau_{\mu}$  is hypersurface orthogonal, as we discussed in chapter 4. This leads to a preferred foliation in terms of (Riemannian) slices of absolute simultaneity. Equivalently, the condition (6.4.67) may be expressed as the vanishing of the twist tensor,

$$\omega^{\mu\nu} = 2h^{\mu\rho}h^{\nu\sigma}\partial_{[\rho}\tau_{\sigma]} = 0. \tag{6.4.69}$$

In this sense, as also emphasized in [32], the more general TNC geometry can be achieved by first gauging the Schrödinger algebra and then adding torsion by hand.

### 6.4.2.1 Introducing Stückelberg Symmetry of the Central Charge

As we have seen, in Lifshitz holography the field  $m_{\mu}$  is accompanied by a Stückelberg scalar  $\chi$ . The purpose of this subsection is to make manifest the connection between TTNC geometry as obtained from gauging  $\mathfrak{sch}_2(2,1)$  in section 6.4.2 and the sources found in section 6.3. Since we already discussed this mechanism in section 4.2.4, we will be brief.

By imposing the curvature constraints (6.4.57), we saw how we were left with three independent fields  $\tau_{\mu}$ ,  $e_{\mu}^{a}$  and  $m_{\mu}$ . By introducing an additional scalar field  $\chi$  transforming as

$$\bar{\delta}\chi = \pounds_{\xi}\chi + \sigma, \tag{6.4.70}$$

we promote *N* to a Stückelberg symmetry, the effect of which will be the replacement of  $m_{\mu}$  by  $M_{\mu}$  given by the usual expression

$$M_{\mu} = m_{\mu} - \partial_{\mu}\chi \tag{6.4.71}$$

in all geometric quantities. We stress that it is not necessary to do so, just as when considering a Stückelberged Proca field, one can either choose to work directly with the Proca field or a Maxwell field and the Stückelberg field. We will use a mixture between the two approaches.

## 6.5 The anomaly $\mathscr{E}$ hořava-lifshitz gravity

### 6.5.1 *Reduction of the Anomaly*

Holographic renormalization of the electromagnetic uplift produces the boundary anomaly

$$\mathcal{A} = -\frac{1}{4} \left( Q_{(0)AB} Q_{(0)}^{AB} - \frac{1}{3} Q_{(0)}^2 - 2\mathcal{F}_{(0)AB} \mathcal{F}_{(0)}^{AB} + \frac{1}{2} \left( \Box_{(\gamma_{(0)})} \psi_{(0)} \right)^2 \right), \tag{6.5.1}$$

where

$$Q_{(0)AB} = R_{AB}^{(\gamma_{(0)})} - \frac{1}{2} \partial_A \psi_{(0)} \partial_B \psi_{(0)}.$$
(6.5.2)

The anomaly is related to HL gravity, as we now briefly discuss—this was observed in [107] and formulated in a TNC covariant manner in [24]. In the case of ALif, where the torsion vanishes and where  $\chi$  (i.e. the boundary value of bulk Stückelberg field  $\Xi$ ) is a constant (for simplicity we will set  $\chi = 0$ ), the reduced anomaly becomes [24]

$$\mathcal{A} = -\left(\mathcal{K}^{\mu\nu}\mathcal{K}_{\mu\nu} - \frac{1}{2}\left(h^{\mu\nu}\mathcal{K}_{\mu\nu}\right)^{2}\right) - \frac{1}{24}\left(\mathcal{R} + e^{\mu}_{a}e^{a}_{\nu}\nabla_{\mu}\left(h^{\nu\rho}\mathcal{L}_{v}\tau_{\rho}\right)\right)^{2} + e^{-1}\partial_{\mu}\left(eJ^{\mu}\right) - 2\mathcal{L}_{\text{GED}}$$

$$(6.5.3)$$

where  $\mathcal{K}_{\mu\nu} = \frac{1}{2} \mathcal{L}_{\nu} h_{\mu\nu}$  is the TNC extrinsic curvature, and  $\mathcal{R}$  is the Ricci scalar of  $\overline{\Gamma}$ .  $J^{\mu}$  is some complicated current (cf. [24]), while  $\mathcal{L}_{\text{GED}}$  is the Lagrangian for GED which can be read off from (6.1.15). Combining our findings, we may thus write

$$-\frac{1}{2}\int d^{3}x \ e \ \mathcal{A} = S_{\rm PHL} + S_{\rm GED}, \tag{6.5.4}$$

where PHL stands for projectable Hořava-Lifshitz, see [53] for details.

#### 6.6 HOLOGRAPHIC RECONSTRUCTION OR MOCK FG EXPANSIONS

From the FG expansions of the higher-dimensional fields that we determined in chapter 3, we can work out the corresponding expansions of the four dimensional bulk fields. In particular, having shown in section 6.4.1 that the reduction becomes null on the boundary, we can use our explicit expression for  $\gamma_{(0)AB}$  as well as our knowledge of the sources from section 6.3 to determine the expansions. We start by translating our results from chapter 3 into the notation of this chapter.

From a five-dimensional perspective, we have the fields  $ds_5^2 = \frac{dr^2}{r^2} + \gamma_{AB} dx^A dx^B$ ,  $\psi$ ,  $A_A$ , admitting FG expansions of the form

$$\gamma_{AB} = \frac{1}{r^2} \left( \gamma_{(0)AB} + r^2 \gamma_{(2)AB} + r^4 \left( \log r \gamma_{(4,1)AB} - \gamma_{(4)AB} \right) \right) + \mathcal{O}(r^6 \log r), \tag{6.6.1}$$

$$\psi = \psi_{(0)} + r^2 \psi_{(2)} + r^4 \log r \psi_{(4,1)} + r^4 \psi_{(4)} + \mathcal{O}(r^6 \log r), \tag{6.6.2}$$

$$\mathcal{A}_{A} = \mathcal{A}_{(0)A} + r^{2} \log r \mathcal{A}_{(2,1)A} + r^{2} \mathcal{A}_{(2)A} + \mathcal{O}(r^{4} \log r),$$
(6.6.3)

where the coefficients are given by (see section 3.2.3 and also footnote 14)

$$\gamma_{(2)AB} = -\frac{1}{2} \left( R_{AB}^{(\gamma_{(0)})} - \frac{1}{2} \partial_A \psi_{(0)} \partial_B \psi_{(0)} \right) + \frac{1}{12} \gamma_{(0)AB} \left( R^{(\gamma_{(0)})} - \frac{1}{2} (\partial \psi_{(0)})^2 \right), \tag{6.6.4}$$

$$\psi_{(2)} = \frac{1}{4} \Box_{(\gamma_{(0)})} \psi_{(0)}, \tag{6.6.5}$$

$$\mathcal{A}_{(2,1)A} = \nabla_B^{(\gamma_{(0)})} \mathcal{F}^B_{(0)A'}$$
(6.6.6)

$$\mathcal{A}_{(2)A} = \frac{1}{2} \mathcal{J}_{(0)A} - \frac{1}{2} \mathcal{A}_{(2,1)A}, \tag{6.6.7}$$

$$\gamma_{(4,1)AB} = \frac{1}{4} \nabla^{(\gamma_{(0)})C} \left( \nabla^{(\gamma_{(0)})}_{A} \gamma_{(2)BC} + \nabla^{(\gamma_{(0)})}_{B} \gamma_{(2)AC} - \nabla^{(\gamma_{(0)})}_{C} \gamma_{(2)AB} \right) - \frac{1}{4} \nabla^{(\gamma_{(0)})}_{A} \nabla^{(\gamma_{(0)})}_{B} \gamma^{C}_{(2)C}$$
(6.6.8)

$$+\gamma_{(2)AC}\gamma^{C}_{(2)B} - \frac{1}{2}\partial_{(A}\psi_{(0)}\nabla^{(\gamma_{(0)})}_{B)}\psi_{(2)} - \gamma_{(0)AB}\left(\frac{1}{4}\gamma^{CD}_{(2)}\gamma_{(2)CD} + \frac{1}{2}\psi^{2}_{(2)}\right)$$
(6.6.9)

$$+\frac{1}{3}\mathcal{F}_{(0)AC}\mathcal{F}_{(0)B}^{\ C}-\frac{1}{12}\gamma_{(0)AB}\mathcal{F}_{(0)CD}\mathcal{F}_{(0)}^{CD},$$
(6.6.10)

$$\psi_{(4,1)} = -\frac{1}{4} \left( \Box_{(\gamma_{(0)})} \psi_{(2)} + 2\psi_{(2)} \gamma^{A}_{(2)A} + \frac{1}{2} \partial^{A} \psi_{(0)} \nabla^{(\gamma_{(0)})}_{A} \gamma^{B}_{(2)B} - \gamma^{AB}_{(2)} \nabla^{(\gamma_{(0)})}_{A} \partial_{B} \psi_{(0)} - \partial^{A} \psi_{(0)} \nabla^{(\gamma_{(0)})B}_{(0)AB} \right)$$

$$(6.6.11)$$

$$\psi_{(4)} = \frac{1}{4} \langle \mathcal{O}_{\psi} \rangle - \frac{1}{4} \psi_{(2)} \gamma^{A}_{(2)A} - \frac{3}{4} \psi_{(4,1)}.$$
(6.6.12)

The fields that we are interested live on three-dimensional surfaces of constant r,

$$h_{\mu\nu} = e^{\Phi} \left( \gamma_{\mu\nu} - e^{-2\Phi} \gamma_{\mu u} \gamma_{\nu u} \right), \qquad A_{\mu} = e^{-2\Phi} \gamma_{\mu u}, \qquad \Phi = \frac{1}{2} \log \gamma_{u u}, \qquad C_{\mu} = \mathcal{A}_{\mu} - \mathcal{A}_{u} e^{-2\Phi} \gamma_{\mu u}$$
(6.6.13)

First off, since  $\gamma_{(0)uu} = 0$ , the expansion for the dilaton becomes

$$\Phi = \frac{1}{2} \log \left( \gamma_{(2)uu} + r^2 \log r \gamma_{(4,1)uu} + r^2 \gamma_{(4)uu} + \cdots \right)$$
(6.6.14)

$$= \frac{1}{2} \log \left( \gamma_{(2)uu} \right) + \frac{1}{2} r^2 \log r \gamma_{(4,1)uu} \gamma_{(2)uu}^{-1} + \frac{1}{2} r^2 \gamma_{(4)uu} \gamma_{(2)uu}^{-1} + \cdots$$
(6.6.15)

$$= \phi + r^2 \log r \Phi_{(2,1)} + r^2 \Phi_{(2)} + \cdots, \qquad (6.6.16)$$

where

$$e^{2\phi} = \gamma_{(2)uu} = -\frac{1}{2}R_{uu}^{(\gamma_{(0)})} + 1, \qquad \Phi_{(2,1)} = \frac{1}{2}e^{-2\phi}\gamma_{(4,1)uu}, \qquad \Phi_{(2)} = \frac{1}{2}e^{-2\phi}\left(X_{uu} - \frac{1}{4}t_{uu}\right), \tag{6.6.17}$$

and we have used the FG coefficients (6.6.4)–(6.6.12). In particular, in the first relation we have used that  $\gamma_{(0)uu} = 0$  and  $\psi_{(0)} = 2u - 2\chi$ . Note that the Ricci tensor above can be rewritten in the form  $R_{uu}^{(\gamma_{(0)})} = 2\mathring{\Gamma}_{u[u,A]}^{A} + 2\mathring{\Gamma}_{B[A}^{A}\mathring{\Gamma}_{u]u}^{B} = \frac{1}{2} \left(\epsilon^{\mu\nu\rho}\tau_{\mu}\partial_{\nu}\tau_{\rho}\right)^{2}$ , where  $\epsilon^{\mu\nu\rho} = e^{-1}\epsilon^{\mu\nu}$  with  $\epsilon^{\mu\nu\rho}$  the Levi-Civita symbol. In particular, this implies that  $\phi$  is not an independent source, which has the important consequence that the corresponding VEV will vanish—we will explore the consequences of this in section 6.7, where we also provide a different reason why the VEV should vanish. This also allows us to identify the coefficients of the expansion of  $\alpha$  (cf. eq. (6.3.9)),

$$\alpha(\Phi) = e^{-3\Phi/2} = \overbrace{e^{-3\phi/2}}^{=\alpha_{(0)}(\phi)} -\frac{3}{2}r^2\log r\alpha_{(0)}\Phi_{(2,1)} - \frac{3}{2}r^2\alpha_{(0)}\Phi_{(2)} + \cdots, \qquad (6.6.18)$$

so that the combination  $\tilde{I} - I$  appearing in (6.3.9) is equal to  $\tilde{I} - I = -\frac{3}{2}\alpha_{(0)}^{4/3}\Phi_{(2)}$ .

Similarly, for the KK vector, we find the expansion

$$A_{\mu} = r^{-2}A_{(0)\mu} + \log rA_{(2,1)\mu} + A_{(2)\mu} + r^{2}\log^{2} rA_{(4,2)\mu} + r^{2}\log rA_{(4,1)\mu} + r^{2}A_{(4)\mu} + \cdots,$$
(6.6.19)

where

$$A_{(0)\mu} = \alpha_{(0)}^{4/3} \tau_{\mu}, \qquad A_{(2,1)\mu} = -2\alpha_{(0)}^{4/3} \Phi_{(2,1)} \tau_{\mu}, \qquad A_{(2)\mu} = \alpha_{(0)}^{4/3} \gamma_{(2)\mu\nu} - 2\Phi_{(2)} A_{(0)\mu}, \quad (6.6.20)$$

$$A_{(4,2)\mu} = \Phi_{(2,1)}^2 A_{(0)\mu}, \qquad A_{(4,1)\mu} = \alpha_{(0)}^{4/3} \gamma_{(4,1)\mu u} - 2\Phi_{(2,1)} A_{(2)\mu} - 2\Phi_{(4,1)} A_{(0)\mu}.$$
(6.6.21)

Note that  $A_{(2)\mu}$  transforms as  $\delta_{\sigma}A_{(2)\mu} = -\partial_{\mu}\sigma$  under Stückelberg gauge transformations (i.e. diffeomorphisms in the compact direction), which follows from the transformation property  $\delta_{\sigma}R_{\mu u}^{(\gamma_{(0)})} = -R_{uu}^{(\gamma_{(0)})}\partial_{\mu}\sigma$  and then using the first relation of (6.6.17). For the metric, we get

$$h_{\mu\nu} = r^{-4}h_{(0)\mu\nu} + r^{-2}\log rh_{(2,1)\mu\nu} + r^{-2}h_{(2)\mu\nu} + \mathcal{O}(\log^2 r),$$
(6.6.22)

where

$$h_{(0)\mu\nu} = -\alpha_{(0)}^{2/3} \tau_{\mu} \tau_{\nu}, \qquad h_{(2,1)\mu\nu} = -h_{(0)\mu\nu} \Phi_{(2,1)}, \qquad h_{(2)\mu\nu} = \alpha_{(0)}^{-2/3} \bar{h}_{\mu\nu} - 2\alpha_{(0)}^{2/3} \tau_{\mu} \gamma_{(2)\nu\mu} + h_{(2,1)\mu\nu}.$$
(6.6.23)

Writing

$$h_{\mu\nu} = -E^0_{\mu}E^0_{\nu} + \delta_{ab}E^a_{\mu}E^b_{\nu}, \qquad E^0_{\mu} \sim r^{-2}\tau_{\mu}, \ E^a_{\mu} \sim r^{-1}e^a_{\mu}.$$
(6.6.24)

we infer from the second term in (6.6.22) that

$$E^{0}_{\mu} = r^{-2} \alpha^{1/3}_{(0)} \tau_{\mu} + \log r Y_{\mu} - \alpha^{-1}_{(0)} \left( M_{\mu} - I \tau_{\mu} \right) + \cdots, \qquad (6.6.25)$$

where we find that

$$Y_{\mu} = \frac{1}{2} \alpha_{(0)}^{1/3} \Phi_{(2,1)} \tau_{\mu}, \tag{6.6.26}$$

whereas  $M_{\mu}$  emerges from  $h_{(2)\mu\nu}$  in (6.6.23) as the combination involving only a single factor of  $\tau_{\mu}$ . Since there is no term of order  $r^{-1} \log r$ , we obtain

$$E^a_{\mu} = r^{-1} \alpha^{-1/3}_{(0)} + \mathcal{O}(r^0). \tag{6.6.27}$$

(6.6.31)

The behaviour of the new fields follows straightforwardly from their FG expansions and the expansions of the four-dimensional fields that we have worked out above,

$$Y = \vartheta + r^2 \log r Y_{(2,1)} + r^2 Y_{(2)} + \cdots,$$
(6.6.28)

$$C_{\mu} = C_{(0)\mu}r^{-2} + \log rC_{(2,1)\mu} + C_{(2)\mu} + r^2 \log rC_{(4,1)\mu} + r^2C_{(4)\mu} + \cdots$$
(6.6.29)

where

$$Y_{(2,1)} = e^{-1}\partial_{\mu}(e\partial^{\mu}\vartheta), \qquad Y_{(2)} = \frac{1}{2}\left(\mathcal{J}_{(0)\mu} - e^{-1}\partial_{\mu}(e\partial^{\mu}\vartheta)\right), \qquad (6.6.30)$$

$$C_{(0)\mu} = -\vartheta\alpha_{(0)}^{4/3}\tau_{\mu}, \qquad C_{(2,1)} = -\alpha_{(0)}^{4/3}\left(Y_{(2,1)} - 2\vartheta\Phi_{(2,1)}\right)\tau_{\mu}, \qquad C_{(2)\mu} = a_{\mu} - \vartheta m_{\mu} - \tilde{\varphi}\tau_{\mu} - \vartheta A_{(2)\mu}.$$

### 6.7 THE VEVS

In this section, we consider the VEVs corresponding to the sources introduced in section 6.3.1. We consider both the relation to the four-dimensional responses determined in section 6.2.3 and the relation to the VEVs in the electromagnetic uplift.

### 6.7.1 Definition & Near-Boundary Behaviour

The goal of this section is to identify the VEVs corresponding to the sources in (6.3.19)–(6.3.28). These can be obtained in terms of the responses (6.2.35)–(6.2.42) via the following identity (which is understood to be sitting inside an integral),

$$\frac{1}{2}T_{\mu\nu}\delta h^{\mu\nu} + \mathcal{T}^{\nu}\delta B_{\nu} + T_{\Phi}\delta\Phi + \mathcal{T}^{\nu}\delta C_{\nu} + \mathcal{T}_{Y}\delta Y =$$
(6.7.1)

$$S^{0}_{\mu}\delta E^{0}_{\mu} + S^{a}_{\mu}\delta E^{\mu}_{a} + \mathcal{T}_{\varphi}\delta\varphi + \mathcal{T}^{a}\delta A_{a} + \mathcal{T}_{\Xi}\delta\Xi + \mathcal{T}_{\Phi}\delta\Phi + \mathcal{T}^{a}\delta C_{a} + \mathcal{T}_{\Omega}\delta\Omega + \mathcal{T}_{Y}\deltaY,$$
(6.7.2)

where we used that  $B_{\nu} = A_{\nu} - \partial_{\nu} \Xi$  (which, due to the dependence of the derivative, implies that the identity above holds up to a total derivative),  $A_a = E_a^{\mu} A_{\mu}$ . In the above, we have used in particular

$$\varphi = E_0^{\nu} A_{\nu} - \alpha(\Phi), \tag{6.7.3}$$

$$\Omega = E_0^{\nu} C_{\nu} + \Upsilon \alpha(\Phi), \tag{6.7.4}$$

In order to identify the VEVs appearing in (6.7.2), we need to relate the variations of the sources to the variations of  $(h_{\mu\nu}, B_{\nu}, \Phi)$ . For  $h_{\mu\nu}$  and  $\Phi$  this is straightforward, but for the massive vector and the Maxwell field a few subtleties arise, so we perform that calculation in detail. We start by noting that

$$\mathcal{T}^{\nu}\delta B_{\nu} = \mathcal{T}^{\nu}\delta A_{\nu} + e^{-1}\partial_{\nu}(e\mathcal{T}^{\nu}) + \text{total derivative,}$$
(6.7.5)

due to the Stückelberg decomposition of the massive vector. Further, we see that the flat version satisfies

$$\mathcal{T}^{0}\delta A_{0} + \mathcal{T}^{a}\delta A_{a} = E^{0}_{\mu}\mathcal{T}^{\mu}\delta\left(E^{\nu}_{0}A_{\nu}\right) + E^{a}_{\mu}\mathcal{T}^{\mu}\delta\left(E^{\nu}_{a}A_{\nu}\right) = \mathcal{T}^{\nu}_{\delta}A_{\nu}$$

$$(6.7.6)$$

$$= E^{0}_{\mu} \mathcal{T}^{\mu} A_{\nu} \delta E^{\nu}_{a} + E^{a}_{\mu} \mathcal{T}^{\mu} A_{\nu} \delta E^{\nu}_{a} + \left( \overline{E^{0}_{\mu} \mathcal{T}^{\mu} E^{\nu}_{0} + E^{a}_{\mu} \mathcal{T}^{\mu} E^{\nu}_{a}} \right) \delta A_{\nu}, \qquad (6.7.7)$$

so that

$$\mathcal{T}^{\nu}\delta A_{\nu} = \mathcal{T}^{0}\delta A_{0} + \mathcal{T}^{a}\delta A_{a} - E^{0}_{\mu}\mathcal{T}^{\mu}A_{\nu}\delta E^{\nu}_{a} - E^{a}_{\mu}\mathcal{T}^{\mu}A_{\nu}\delta E^{\nu}_{a}.$$
(6.7.8)

Observe further that since—by (6.7.3)— $A_0 = \varphi + \alpha(\Phi)$ ,

$$\mathcal{T}^{0}\delta A_{0} = E^{0}_{\mu}\mathcal{T}^{\mu}\left(\underbrace{\frac{\delta A_{0}}{\delta \varphi}}_{=1}\delta \varphi + \underbrace{\frac{\delta A_{0}}{\delta \Phi}}_{=\frac{\mathrm{d}\alpha(\Phi)}{\mathrm{d}\Phi}}\delta \Phi\right).$$
(6.7.9)

Repeating the analysis above for the Maxwell field, we find that

$$\mathcal{T}^{\nu}\delta C_{\nu} = \mathcal{T}^{0}\delta C_{0} + \mathcal{T}^{a}\delta C_{a} - E^{0}_{\mu}\mathcal{T}^{\mu}C_{\nu}\delta E^{\nu}_{a} - E^{a}_{\mu}\mathcal{T}^{\mu}C_{\nu}\delta E^{\nu}_{a}, \qquad (6.7.10)$$

where, since  $C_0 = \Omega - \Upsilon \alpha$ 

$$\mathcal{T}^{0}\delta C_{0} = E^{0}_{\mu}\mathcal{T}^{\mu}\left(\delta\Omega - Y\frac{\mathrm{d}\alpha}{\mathrm{d}\Phi}\delta\Phi - \alpha\delta Y\right).$$
(6.7.11)

Combining our findings, the responses can be written in terms of the quantities of (6.2.35)-(6.2.42)

$$S^{0}_{\mu} = -\left(T_{\mu\nu}E^{\nu}_{0} + \mathcal{T}^{\rho}E^{0}_{\rho}A_{\mu} + \mathcal{T}^{\rho}E^{0}_{\rho}C_{\mu}\right), \qquad (6.7.12)$$

$$\mathcal{S}^{a}_{\mu} = \left(T_{\mu\nu}E^{\nu a} - \mathcal{T}^{\rho}E^{a}_{\rho}A_{\mu} - \mathcal{T}^{\rho}E^{a}_{\rho}C_{\mu}\right),\tag{6.7.13}$$

$$\mathcal{T}_{\varphi} = \mathcal{T}^{\nu} E_{\nu}^{0}, \tag{6.7.14}$$

$$\mathcal{T}_{\Phi} = T_{\Phi} + \mathcal{T}^{\nu} E_{\nu}^{0} \frac{d\alpha(\Phi)}{d\Phi} + \Upsilon E_{\mu}^{0} \widetilde{\mathcal{T}}^{\mu} \frac{d\alpha}{d\Phi}, \qquad (6.7.15)$$

$$\mathcal{T}_{\mu}^{a} = \mathcal{T}^{\nu} \Gamma^{a}$$

$$\mathcal{T}^a = \mathcal{T}^\nu E^a_{\nu},\tag{6.7.16}$$

$$\mathcal{T}_{\Xi} = e^{-1} \partial_{\mu} \left( e \mathcal{T}^{\mu} \right), \tag{6.7.17}$$

$$\mathcal{T}^a = \mathcal{T}^\nu E^a_{\nu\nu} \tag{6.7.18}$$

$$\mathcal{T}_{\Omega} = \mathcal{T}^{\nu} E_{\nu}^{0}, \tag{6.7.19}$$

$$\mathcal{T}_Y = T_Y + \mathcal{T}^{\mu} E^0_{\mu} \alpha \tag{6.7.20}$$

Below we list the responses and their near-boundary behaviours and thus identify the VEVs. Note that we have to take into account the near-boundary behaviour of the measure  $\sqrt{-\gamma}$  as was the case in section 5.3.1, where we performed the same analysis for general values of *z*. Doing this, we obtain

$$S^0_{\mu} \simeq r^2 \alpha^{2/3}_{(0)} S^0_{\mu'} \tag{6.7.21}$$

$$\mathcal{S}^a_\mu \simeq r^3 \mathcal{S}^a_\mu,\tag{6.7.22}$$

$$\mathcal{T}_{\varphi} \simeq -r^{2} \alpha_{(0)}^{2/3} T^{0},$$
 (6.7.23)  
 $\mathcal{T}^{a} \simeq -r^{3} T^{a},$  (6.7.24)

$$\mathcal{T}^{*} \simeq -r^{2} T^{*}, \qquad (6.7.24)$$

$$\mathcal{T}_{\Phi} \simeq r^{4} \alpha^{1/3} \langle O_{\Phi} \rangle \qquad (6.7.25)$$

$$\mathcal{T}_{\phi} \simeq \mathcal{T}_{\alpha(0)} \langle 0_{\phi} \rangle, \qquad (6.7.25)$$

$$\mathcal{T}_{\alpha(0)} = \mathcal{T}_{\alpha(0)} \langle 0_{\phi} \rangle, \qquad (6.7.25)$$

$$\mathcal{T}_{\Xi} \simeq -r^{*} \alpha_{(0)}^{\prime \prime \prime \prime} \langle O_{\chi} \rangle , \qquad (6.7.26)$$

$$\mathcal{T}_{a}^{a} \sim r^{3} i^{a} \qquad (6.7.27)$$

$$\mathcal{D}^{*} \simeq r^{2} j^{2} j^{3} ,$$
 (6.7.27)  
 $\mathcal{T}_{-} \simeq r^{2} a^{2/3} i^{0}$  (6.7.28)

$$\eta_{\Omega} \simeq r^{-} \alpha_{(0)}^{-} f^{-},$$
 (6.7.28)

$$\mathcal{T}_{Y} \simeq r^{4} \alpha_{(0)}^{1/3} \left\langle O_{\vartheta} \right\rangle. \tag{6.7.29}$$

From these, we may immediately read off their scaling weights:

VEV	$S^0_\mu$	$S^a_\mu$	$T^0$	T <sup>a</sup>	$\langle O_{\phi} \rangle$	$\langle O_{\chi} \rangle$	j <sup>a</sup>	j <sup>0</sup>	$\langle O_{\vartheta} \rangle$
scaling dim.	2	3	2	3	4	4	3	2	4

Table 6.7: The VEVs and their scaling dimensions.

We note that, as was the case with the sources, the scaling dimensions of the new VEVs correspond to the results in the field theory (cf. table 6.4) when setting z = 2.

## 6.7.2 Relation to Five-Dimensional VEVs Via Null Reduction

We now proceed to relate the five-dimensional VEVs to the four-dimensional VEVs and perform a general analysis of these. In the uplift, we have that

$$\delta S_{\rm ren} = \lim_{r \to 0} \frac{1}{2\kappa_5^2} \int_{\Sigma_r} d^4 x \sqrt{-\gamma} \left( \frac{1}{2} T_{AB} \delta \gamma^{AB} + T_{\psi} \delta \psi + \mathcal{J}^A \delta \mathcal{A}_A \right)$$
(6.7.30)

$$= \int_{\partial \mathcal{M}} \mathrm{d}^{3}x \, e\left(\underbrace{\frac{1}{2} t_{AB} \delta \gamma_{(0)}^{AB}}_{=:(*)} + \underbrace{\langle O_{\psi} \rangle \, \delta \psi_{(0)}}_{=:(\dagger)} + \underbrace{\mathcal{J}_{(0)}^{A} \delta \mathcal{A}_{(0)A}}_{=:(\ddagger)}\right), \tag{6.7.31}$$

where we have performed the *u*-integral and used that  $\sqrt{-\gamma_{(0)}} =: e = \det(\tau_{\mu}, e_{\mu}^{a})$ , which follows from (??). In the four-dimensional theory, we then have

$$\delta S_{\text{ren}} = \lim_{r \to \infty} \int_{\partial \mathcal{M}} d^3 x \, \sqrt{-h} \left( \frac{1}{2} T_{\mu\nu} \delta h^{\mu\nu} + \mathcal{T}^{\nu} \delta B_{\nu} + T_{\Phi} \delta \Phi + \mathcal{T}^{\nu} \delta C_{\nu} + \mathcal{T}_{Y} \delta Y \right) \tag{6.7.32}$$

$$= \int_{\partial \mathcal{M}} \mathrm{d}^{3}x \ e \left( -S^{0}_{\mu} \delta v^{\mu} + S^{a}_{\mu} \delta e^{\mu}_{a} + T^{0} \delta m_{0} + T^{a} \delta m_{a} + \langle O_{\chi} \rangle \ \delta \chi + \langle \tilde{O}_{\phi} \rangle \ \delta \phi + j^{a} \delta a_{a} - j^{0} \delta \tilde{\varphi} + \langle O_{\vartheta} \rangle \ \delta \vartheta \right),$$

$$(6.7.33)$$

where  $m_0 = -v^{\mu}m_{\mu}$ ,  $m_a = e^{\mu}_a m_{\mu}$ , and where we have used

$$\psi_{(0)} = 2u - 2\chi, \quad \langle O_{\psi} \rangle = -\frac{1}{2} \langle O_{\chi} \rangle.$$
(6.7.34)

This implies that  $\delta \psi_{(0)} = -2\delta \chi$ . The VEV  $\langle \tilde{O}_{\phi} \rangle$  is given by

$$\langle \tilde{O}_{\phi} \rangle = \langle O_{\phi} \rangle - \frac{1}{2} \left[ v^{\mu} (S^{0}_{\mu} + T^{0} m_{\mu}) + e^{\mu}_{a} (S^{a}_{\mu} + T^{a} m_{\mu}) + j^{a} a_{a} - j^{0} \tilde{\varphi} \right] = 0, \qquad (6.7.35)$$

and vanishes, as we will explain shortly. The extra terms in  $\langle \tilde{O}_{\phi} \rangle$  come from the presence of  $\alpha_{(0)}(\phi)$ -terms in the responses (6.3.2)–(6.3.8) considered from the perspective of the right-hand side of the identity (6.7.2), e.g.

$$\sqrt{-\gamma}S^{0}_{\mu}\delta E^{\mu}_{0} \simeq -eS^{0}_{\mu}\delta v^{\mu} - e\alpha^{1/3}_{(0)}S^{0}_{\mu}v^{\mu}\frac{d\left(\alpha^{-1/3}_{(0)}\right)}{d\phi}\delta\phi = -eS^{0}_{\mu}\delta v^{\mu} - \frac{1}{2}eS^{0}_{\mu}v^{\mu}\delta\phi.$$
(6.7.36)

The new terms will contribute

$$\sqrt{-\gamma} \mathcal{T}^a \delta C_a \simeq e j^a \delta a_a + e \alpha_{(0)}^{-1/3} j^a a_a \frac{\mathrm{d}\alpha_{(0)}^{1/3}}{\mathrm{d}\phi} \delta \phi = e j^a \delta a_a - \frac{1}{2} e j^a a_a \delta \phi, \qquad (6.7.37)$$

as well as (using  $\Omega \simeq -r^2 \alpha_{(0)}^{-1/3} \tilde{\varphi}$ )

$$\sqrt{-\gamma}\mathcal{T}_{\Omega}\delta\Omega \simeq -ej^{0}\delta\tilde{\varphi} - \alpha_{(0)}^{1/3}ej^{0}\tilde{\varphi}\frac{\mathrm{d}\alpha_{(0)}^{-1/3}}{\mathrm{d}\phi}\delta\phi = -ej^{0}\delta\tilde{\varphi} - \frac{1}{2}ej^{0}\tilde{\varphi}\delta\phi.$$
(6.7.38)

Since none of the bulk fields have a near-boundary behaviour that involves the source  $\vartheta$ , the same *does not happen* to  $\langle O_{\vartheta} \rangle$ ,

$$\langle \tilde{O}_{\vartheta} \rangle = \langle O_{\vartheta} \rangle$$
, (6.7.39)

i.e. there will be no additional terms in front of  $\delta \vartheta$  as was the case for  $\delta \phi$ .

Now, recall from our analysis of the FG expansions in section 6.6 that

$$e^{2\phi} = -\frac{1}{4}\omega^2 + 1, \tag{6.7.40}$$

where  $\varpi$  is the twist. This implies that the source  $\phi$  is not independent and has the consequence that the variation of the renormalized on-shell action with respect to  $\phi$  gives zero. This can also be seen from the shift transformation (the  $\delta_c$  transformation, also known as local dilatations in [24]) we considered in section 6.3.2.4, which directly implies  $\langle \tilde{O}_{\phi} \rangle = 0$ . Similarly, from the gauge shift transformation  $\delta_{\ell}$ , we get the relation

$$\langle O_{\vartheta} \rangle = 0. \tag{6.7.41}$$

Now, as we demonstrated in section 6.4.1, the Scherk-Schwarz reduction employed becomes a null reduction on the boundary. In order to relate the VEVs appearing on the right-hand side in (6.7.21)–(6.7.29) to the VEVs in the electromagnetic uplift, we perform a null reduction of the expression in (6.7.31), which is then equated with the same expression from a lower-dimensional perspective (6.7.33). In components, the null reduction (6.4.18) can be written as

$$\gamma_{(0)}^{\mu\nu} = h^{\mu\nu} = e^{\mu a} e^{\nu}_{a}, \qquad \gamma_{(0)}^{\mu\mu} = -\hat{v}^{\mu} = e^{\mu a} m_{a} - v^{\mu}, \qquad \gamma_{(0)}^{\mu\nu} = 2\tilde{\Phi} = m^{a} m_{a} + 2m_{0}, \quad (6.7.42)$$

which means that  $\delta h^{\mu\nu} = 2e_a^{(\mu}\delta e^{\nu)a}$ . From the expansion of  $C_{\mu}$  in (6.3.18), we note that

$$\mathcal{A}_{(0)\mu} = a_{\mu} - \vartheta m_{\mu} - \tilde{\varphi} \tau_{\mu}, \qquad \mathcal{A}_{(0)\mu} = \vartheta, \qquad (6.7.43)$$

which in particular means that we can rewrite

$$\mathcal{A}_{(0)\mu} = e^a_\mu \left( a_a - \vartheta m_a \right) - \tau_\mu \left( \vartheta m_0 + \tilde{\varphi} \right). \tag{6.7.44}$$

Omitting the integral<sup>4</sup> and applying the null reduction of (6.7.42) to each term in (6.7.31), we obtain:

$$(*) = t_{\mu\nu}e_a^{\mu}\delta e^{\nu a} + t_{u\mu}\left(e^{\mu a}\delta m_a + m_a\delta e^{\mu a} - \delta v^{\mu}\right) + t_{uu}\left(m^a\delta m_a + \delta m_0\right),$$
(6.7.45)

$$(\dagger) = \langle O_{\chi} \rangle \, \delta\chi \qquad \text{(follows immediately from (6.7.34))}, \tag{6.7.46}$$

$$(\ddagger) = \mathcal{J}_{(0)}^{\mu} \delta \vartheta + \mathcal{J}_{(0)}^{\mu} \left[ e_{\mu}^{a} \left( \delta a_{a} - \vartheta \delta m_{a} - m_{a} \delta \vartheta \right) - \tau_{\mu} \left( \vartheta \delta m_{0} + m_{0} \delta \vartheta + \delta \tilde{\varphi} \right) \right] + \mathcal{J}_{(0)}^{\mu} (a_{a} - \vartheta m_{a}) \delta e_{\mu}^{a} - \mathcal{J}_{(0)}^{\mu} \left( \vartheta m_{0} + \tilde{\varphi} \right) \delta \tau_{\mu}.$$

$$(6.7.47)$$

<sup>4</sup> When omitting the integral, we must recall our ability to integrate by parts. Equalities therefore only hold up to total derivatives.

We observe that the expression above involves two additional variations,  $\delta e^a_\mu$  and  $\delta \tau_\mu$ , in the last two terms, and that these are not present in the expression for  $\delta S_{ren}$  in (6.7.33) which we'll eventually want to compare with. Therefore, we use the second identity in (6.1.10) to write, first off

$$\mathcal{J}^{\mu}_{(0)} = \left(e^{\mu}_{a}e^{a}_{\nu} - \tau_{\nu}v^{\mu}\right)\mathcal{J}^{\nu}_{(0)}.$$
(6.7.48)

Now, using the first, third, fourth and fifth identity of (6.1.10), we find that

$$\tau_{\mu}\delta v^{\mu} = -v^{\mu}\delta\tau_{\mu}, \qquad e^{a}_{\mu}\delta v^{\mu} = -v^{\mu}\delta e^{a}_{\mu}, \qquad \tau_{\mu}\delta e^{\mu}_{a} = -e^{\mu}_{a}\delta\tau_{\mu}, \qquad e^{a}_{\mu}\delta e^{\mu}_{b} = -e^{\mu}_{b}\delta e^{a}_{\mu}, \quad (6.7.49)$$

and so we can rewrite the last two terms in the expression for the term  $(\ddagger)$  (6.7.47) in the following manner:

$$\mathcal{J}^{\mu}_{(0)}(a_a - \vartheta m_a)\delta e^a_{\mu} = \mathcal{J}^{\nu}_{(0)}(a_a - \vartheta m_a)e^a_{\mu} \left[\tau_{\nu}\delta v^{\mu} - e^b_{\nu}\delta e^{\mu}_b\right], \qquad (6.7.50)$$

$$\mathcal{J}^{\mu}_{(0)}(m_0 + \tilde{\varphi})\delta\tau_{\mu} = \mathcal{J}^{\nu}_{(0)}(\vartheta m_0 + \tilde{\varphi})\tau_{\mu} \left[\tau_{\nu}\delta v^{\mu} - e^b_{\nu}\delta e^{\mu}_b\right].$$
(6.7.51)

Combining our findings, we obtain

$$\frac{1}{2}t_{AB}\delta\gamma^{AB}_{(0)} + \langle O_{\psi}\rangle\,\delta\psi_{(0)} + \mathcal{J}^{A}_{(0)}\delta\mathcal{A}_{(0)A} = -\left[t_{u\mu} - \mathcal{J}^{\nu}_{(0)}\tau_{\nu}\left\{(a_{a} - \vartheta m_{a})e^{a}_{\mu} - (\vartheta m_{0} + \tilde{\varphi})\tau_{\mu}\right\}\right]\delta v^{\mu} \quad (6.7.52)$$

$$+\left[t_{\nu\mu}e^{\nu a}+t_{u\mu}m^{a}-\mathcal{J}_{(0)}^{\nu}e_{\nu}^{a}\left\{(a_{b}-\vartheta m_{b})e_{\mu}^{b}-(\vartheta m_{0}+\tilde{\varphi})\tau_{\mu}\right\}\right]\delta e_{a}^{\mu}+\left[t_{uu}-\vartheta\mathcal{J}_{(0)}^{\mu}\tau_{\mu}\right]\delta m_{0}$$
(6.7.53)

$$+\left[t_{u\mu}e^{\mu a}+t_{uu}m^{a}-\mathcal{J}^{\mu}_{(0)}e^{a}_{\mu}\vartheta\right]\delta m_{a}+\left[-\mathcal{J}^{\mu}_{(0)}(\tau_{\mu}m_{0}+e^{a}_{\mu}m_{a})+\mathcal{J}^{\mu}_{(0)}\right]\delta\vartheta+\mathcal{J}^{\mu}_{(0)}e^{a}_{\mu}\delta a_{a}+\langle O_{\chi}\rangle\,\delta\chi-\mathcal{J}^{\mu}_{(0)}\tau_{\mu}\delta\tilde{\varphi}$$

$$(6.7.54)$$

giving us the result

$$S^{0}_{\mu} = t_{u\mu} - \mathcal{J}^{\nu}_{(0)} \tau_{\nu} \left\{ (a_{a} - \vartheta m_{a})e^{a}_{\mu} + (\vartheta m_{0} - \tilde{\varphi})\tau_{\mu} \right\} = t_{u\mu} - \mathcal{J}^{\nu}_{(0)} \tau_{\nu} \hat{a}_{\mu},$$

$$S^{a}_{\mu} = t_{\nu\mu}e^{\nu a} + t_{u\mu}m^{a} - \mathcal{J}^{\nu}_{(0)}e^{a}_{\nu} \left\{ (a_{b} - \vartheta m_{b})e^{b}_{\mu} - (\vartheta m_{0} + \tilde{\varphi})\tau_{\mu} \right\} = t_{\nu\mu}e^{\nu a} + t_{u\mu}m^{a} - \mathcal{J}^{\nu}_{(0)}e^{a}_{\nu} \hat{a}_{\mu},$$
(6.7.55)

$$T^{0} = t_{uu} - \mathcal{J}^{\mu}_{(0)} \vartheta \tau_{\mu}, \tag{6.7.57}$$

$$T^{a} = t_{u\mu}e^{\mu a} + t_{uu}m^{a} - \mathcal{J}^{\mu}_{(0)}e^{a}_{\mu}\vartheta, \qquad (6.7.58)$$

$$\langle O_{\theta} \rangle = -\mathcal{J}_{(0)}^{\mu}(\tau_{\mu}m_{0} + e_{\mu}^{a}m_{a}) + \mathcal{J}_{(0)}^{u} = \mathcal{J}_{(0)}^{u} - \mathcal{J}_{(0)}^{\mu}m_{\mu} = 0,$$
(6.7.59)

$$O_{\chi} \rangle = -2 \langle O_{\psi} \rangle, \tag{6.7.60}$$
$$i^a - \mathcal{T}^{\mu} c^a \tag{6.7.61}$$

$$j^{0} = \mathcal{J}_{(0)}^{\mu} \tau_{\mu}, \tag{6.7.62}$$

where we have introduced the notation

$$\hat{a}_{\mu} = a_{\mu} - \vartheta m_{\mu} - \tilde{\varphi} \tau_{\mu}, \tag{6.7.63}$$

(6.7.56)

which is the same GED field combination we considered in section 6.1. These allow is to determine the transformation properties of the VEVs, which we will consider in section 6.7.3. Now, we invert the relations above to find the expressions for the higher dimensional VEVs in terms of the lower dimensional ones, which will allow for a straightforward translation of the higher-dimensional Ward identities:

$$\mathcal{J}_{(0)}^{\mu} = j^{\mu}, \qquad t_{u\mu} = S_{\mu}^{0} + j^{0}\hat{a}_{\mu}, \qquad e^{\nu a}t_{\nu\mu} = S_{\mu}^{a} + j^{a}\hat{a}_{\mu} - (S_{\mu}^{0} + j^{0}\hat{a}_{\mu})m^{a}, \qquad t_{uu} = T^{0} + \vartheta j^{0}.$$
(6.7.64)

### 6.7.3 Local Transformations of the VEVs

Having obtained expressions for the four-dimensional VEVs in terms of the VEVs in the electromagnetic uplift in (6.7.55)-(6.7.62), we can use our knowledge of the transformation properties of  $t_{AB}$  to obtain the transformations of the VEVs, just like in section 6.3.2.3. Using

$$\pounds_{\hat{\xi}} t_{AB} = \hat{\xi}^C \partial_C t_{AB} + t_{CB} \partial_A \hat{\xi}^C + t_{AC} \partial_B \hat{\xi}^C + \delta_D t_{AB}, \qquad (6.7.65)$$

with

$$\hat{\xi}^u = -\sigma, \tag{6.7.66}$$

we find

$$\delta S^0_{\mu} = \pounds_{\xi} S^0_{\mu} - T^0 \partial_{\mu} \sigma + j^0 \vartheta \partial_{\mu} \sigma - j^0 \partial_{\mu} \Gamma_{(0)} - 2\Lambda_D S^0_{\mu}, \qquad (6.7.67)$$

$$\delta S^a_{\mu} = \pounds_{\xi} S^a_{\mu} + \lambda^a S^0_{\mu} + \lambda^a{}_b S^b_{\mu} - T^a \partial_{\mu} \sigma + j^a \vartheta \partial_{\mu} \sigma - j^a \partial_{\mu} \Gamma_{(0)} - 3\Lambda_D S^a_{\mu}, \tag{6.7.68}$$

$$\delta T^0 = \pounds_{\xi} T^0 - 2\Lambda_D T^0, \tag{6.7.69}$$

$$\delta T^a = \pounds_{\xi} T^a + \lambda^a T^0 + \lambda^a{}_b T^b - 3\Lambda_D T^a, \qquad (6.7.70)$$

$$\delta \left\langle O_{\chi} \right\rangle = \pounds_{\xi} \left\langle O_{\chi} \right\rangle - 4\Lambda_D \left\langle O_{\chi} \right\rangle, \tag{6.7.71}$$

$$\delta j^0 = \pounds_{\xi} j^0 - 2\Lambda_D j^0, \tag{6.7.72}$$

$$\delta j^a = \pounds_{\xi} j^a + \lambda^a j^0 + \lambda^a{}_b j^b - 3\Lambda_D j^a. \tag{6.7.73}$$

Thus, we see that-just like the sources-the VEVs transform under the symmetry group

$$Diff(\mathcal{M}) \times Sch_2(d, 1) \times U(1).$$
 (6.7.74)

## 6.8 the ward identities $\mathcal E$ boundary stress tensor

### 6.8.1 The HIM Boundary Energy-Momentum Tensor & Augmentations

The HIM stress tensor,

$$T^{\mu}{}_{\nu} = -v^{\mu}S^{0}_{\nu} + e^{\mu}_{a}S^{a}_{\nu}, \qquad (6.8.1)$$

transforms under gauge transformations ( $U_{\sigma}(1)$  and  $U_{\Gamma}(1)$ ) as

$$\delta_{\sigma,\Gamma} T^{\mu}{}_{\nu} = -\left(T^{\mu} + j^{\mu}\vartheta\right)\partial_{\nu}\sigma - j^{\mu}\partial_{\nu}\Gamma_{(0)}.$$
(6.8.2)

This leads us to define a totally gauge invariant augmentation,

$$T_{(\chi,\hat{a})}{}^{\mu}{}_{\nu} = T^{\mu}{}_{\nu} + T^{\mu}\partial_{\nu}\chi + j^{\mu}\hat{a}_{(\chi)\nu}, \qquad (6.8.3)$$

where

$$\hat{a}_{(\chi)\mu} = a_{\mu} - \vartheta M_{\mu} - \tilde{\varphi}\tau_{\mu}. \tag{6.8.4}$$

Note in particular that  $T_{(\chi,\hat{a})}^{\mu}{}_{\nu}$  is also invariant under  $\delta_{\ell}$ . The combination  $T^{\mu} + j^{\mu}\vartheta$  featuring prominently above is  $\delta_{\ell}$  invariant, and we write it as

$$T^{\mu}_{(\hat{a})} = T^{\mu} + j^{\mu} \vartheta.$$
 (6.8.5)

Note that this implies the relation

$$T_{(\chi,\hat{a})}^{\mu}{}_{\nu} = T^{\mu}_{(\hat{a})\nu} + T^{\mu}_{(\hat{a})}\partial_{\nu}\chi, \qquad (6.8.6)$$

where

$$T_{(\hat{a})}^{\mu}{}_{\nu} = T^{\mu}{}_{\nu} + j^{\mu}\hat{a}_{\nu} \tag{6.8.7}$$

is the  $U_{\Gamma}(1)$  gauge invariant HIM tensor. We will eventually want to write Ward identities in terms of quantities carrying space-time indices, so it is useful to note that

$$T^{\mu} = -T^{0}v^{\mu} + T^{a}e^{\mu}_{a}, \qquad t^{\mu} = -j^{0}v^{\mu} + j^{a}e^{\mu}_{a}.$$
(6.8.8)

We also have augmented Galilean boost invariant objects,

$$\tilde{\Phi}_{(\chi)} = -v^{\rho}M_{\rho} + \frac{1}{2}h^{\rho\sigma}M_{\rho}M_{\sigma}, \qquad (6.8.9)$$

$$\hat{v}^{\mu}_{(\chi)} = v^{\mu} - h^{\mu\nu} M_{\nu}, \tag{6.8.10}$$

$$\hat{e}^{a}_{(\chi)\mu} = e^{a}_{\mu} - M_{\nu}e^{\nu a}\tau_{\mu}, \qquad (6.8.11)$$

$$\hat{h}_{(\chi)\mu\nu} = \delta_{ab} \hat{e}^{a}_{(\chi)\mu} \hat{e}^{b}_{(\chi)\nu'}$$
(6.8.12)

where we have replaced  $m_{\mu}$  with the gauge invariant version  $M_{\mu}$ . Note that we can relate the lower dimensional quantities to higher dimensional fields in a more covariant manner. For example, observe that

$$\Gamma^{\mu}{}_{\nu} = -v^{\mu}S^{0}_{\nu} + e^{\mu}_{a}S^{a}_{\nu} = -v^{\mu}t_{u\nu} + e^{\mu}_{a}\left(t_{\rho\nu}e^{\rho a} + t_{u\nu}m^{a}\right) + v^{\mu}j^{0}\hat{a}_{\nu} - e^{\mu}_{a}j^{a}\hat{a}_{\nu}$$
(6.8.13)

$$=\gamma_{(0)}^{\mu A}t_{A\nu} - j^{\mu}\hat{a}_{\nu} = t^{\mu}{}_{\nu} - j^{\mu}\hat{a}_{\nu}.$$
(6.8.14)

A similar analysis shows that

$$T^{\mu} = t^{\mu}{}_{u} - j^{\mu}\vartheta. \tag{6.8.15}$$

Continuing this procedure, we find

$$t^{\mu\nu} = 2\tilde{\Phi} \left( T^{\mu} + \vartheta j^{\mu} \right) - \vartheta^{\sigma} \left( T^{\mu}{}_{\sigma} + j^{\mu}\hat{a}_{\sigma} \right), \qquad t^{\mu\nu} = -\vartheta^{\mu} \left( T^{\nu} + \vartheta j^{\nu} \right) + h^{\mu\rho} \left( T^{\nu}{}_{\rho} + j^{\mu}\hat{a}_{\rho} \right).$$
(6.8.16)

This implies further that

$$t^{\mu}{}_{\nu} = 2\tilde{\Phi}T^{\mu}_{(\hat{a})}\bar{h}_{\mu\nu} - \vartheta^{\sigma}T^{\mu}_{(\hat{a})\sigma}\bar{h}_{\mu\nu}, \qquad (6.8.17)$$

where we have used that  $\gamma_{(0)uu} = 0$ .

### 6.8.2 Ward Identities

These Ward identities are obtained by requiring invariance of the renormalized on-shell action under the respective symmetries, that is to say, the expression

$$\delta S_{\rm ren} = \int_{\partial \mathcal{M}} \mathrm{d}^3 x \ e \left( -S^0_\mu \delta v^\mu + S^a_\mu \delta e^\mu_a + T^0 \delta m_0 + T^a \delta m_a + \langle O_\chi \rangle \ \delta \chi + j^a \delta a_a - j^0 \delta \tilde{\varphi} + \langle O_\vartheta \rangle \ \delta \vartheta - \mathcal{A}^{(0)}_{\rm red} \frac{\delta r}{r} \right), \tag{6.8.18}$$

should be invariant under  $Diff(\mathcal{M}) \times Sch_2(d, 1) \times U(1)$ . Note also that  $\langle O_{\theta} \rangle = 0$  above.

### 6.8.2.1 Galilean Boosts

We start with the boost Ward identity; the fields transforming under Galilean boosts G are

$$\delta_G v^\mu = \lambda^a e^\mu_a, \qquad \delta_G m_a = \lambda_a, \qquad \delta_G m_0 = -\lambda^a m_a, \qquad \delta_G \tilde{\varphi} = a_a \lambda^a, \qquad \delta_G a_a = \vartheta \lambda_a. \tag{6.8.19}$$

Requiring invariance of (6.8.18) under Galilean boosts, we obtain the Ward identity

$$-S^{0}_{\mu}e^{\mu}_{a} + T^{b}\delta_{ba} - T^{0}m_{a} - j^{0}a_{a} + j^{b}\delta_{ba}\vartheta = 0.$$
(6.8.20)

Employing techniques identical to those in section 5.3.3, we may covariantize the boost Ward identity above. Using our result for the pure Lifshitz boost Ward identity (5.3.34), we note that (6.8.20) can be written as

$$\hat{e}^{a}_{(\chi)\mu}T^{\mu} - j^{0}a^{a} + j^{a}\vartheta = \tau_{\nu}e^{\mu a}\left(T^{\nu}{}_{\mu} + T^{\nu}\partial_{\mu}\chi\right).$$
(6.8.21)

To covariantize the new part, observe that

$$j^0 = \tau_\mu j^\mu, \qquad j^b = e^b_\mu j^\mu, \qquad a_a = e^\mu_a a_\mu,$$
 (6.8.22)

implying that

$$-j^{0}a_{a}+j^{b}\delta_{ba}\vartheta=j^{\mu}\left(-\tau_{\mu}e_{a}^{\nu}a_{\nu}+\vartheta e_{\mu a}\right)=j^{\mu}\left(-\tau_{\mu}e_{a}^{\nu}\left(a_{\nu}-\vartheta M_{\nu}\right)+\vartheta \hat{e}_{(\chi)\mu a}\right).$$
(6.8.23)

We can replace  $a_{\nu} - \vartheta M_{\nu}$  with a Stückelberg invariant extension of  $\hat{a}_{\mu}$ ,

$$\hat{a}_{(\chi)\mu} = a_{\mu} - \vartheta M_{\mu} - \tilde{\varphi}\tau_{\mu}, \qquad (6.8.24)$$

due to the orthogonality property  $\tau_{\mu}e_{a}^{\mu}=0$ . All in all, this means that the boost Ward identity becomes

$$\hat{e}^{a}_{(\chi)\mu}T^{\mu} + j^{\mu} \left(\vartheta \hat{e}^{a}_{(\chi)\mu} - \tau_{\mu} e^{\nu a} \hat{a}_{(\chi)\nu}\right) = \tau_{\nu} e^{\mu a} \left(T^{\nu}_{\mu} + T^{\nu} \partial_{\mu} \chi\right),$$
(6.8.25)

which can also be written as

$$\hat{e}^{a}_{(\chi)\mu}T^{\mu}_{(\hat{a})} = \tau_{\nu}e^{\mu a}T_{(\chi,\hat{a})}{}^{\nu}{}_{\mu}.$$
(6.8.26)

Thus, we see that the boost Ward identity takes a form identical to what we find in pure Lifshitz holography in chapter 5, but instead in written in terms of the GED-augmented quantities defined in section 6.8.1.

### 6.8.2.2 Rotations

Next, we turn to rotations *J*, where the relevant transformations are

$$\delta_J a_a = \lambda_a{}^b a_b, \qquad \delta_J m_a = \lambda_a{}^b m_b, \qquad \delta_J e_a^\mu = \lambda_a{}^b e_b^\mu, \tag{6.8.27}$$

which means that the rotation Ward identity becomes

$$S^{[a}_{\mu}e^{\mu b]} + T^{[a}m^{b]} + j^{[a}a^{b]} = 0.$$
(6.8.28)

Multiplying by  $m^b$  and antisymmetrizing, this becomes

$$S^{0}_{\mu}e^{\mu[a}m^{b]} + t^{0}a^{[a}m^{b]} - t^{[a}m^{b]}\vartheta = T^{[a}m^{b]}.$$
(6.8.29)

Following the insight from section 5.3.3, we expect that in order to obtain the rotation Ward identity, we need to apply the boost Ward identity in the form

$$-S^{0}_{\mu}e^{\mu a} + T^{a} - T^{0}m^{a} - j^{0}a^{a} + j^{a}\vartheta = 0, \qquad (6.8.30)$$

so that our rotation Ward identity (6.8.28) can be recast in the form

$$S^{[a}_{\mu}e^{\mu b]} + S^{0}_{\mu}e^{\mu[a}m^{b]} + j^{0}a^{[a}m^{b]} - j^{[a}m^{b]}\vartheta + j^{[a}a^{b]} = 0.$$
(6.8.31)

Now, observe that we can rewrite the left-hand side of the expression above in the following manner

$$\hat{e}^{[a}_{(\chi)\nu}e^{b]\mu}T_{(\chi,\hat{a})}{}^{\nu}{}_{\mu} - \partial_{\mu}\chi e^{\mu b} \left(\vartheta t^{a} - j^{0}a^{a} + T^{a} - m^{a}T^{0} - e^{\rho a}S^{0}_{\rho}\right)$$
(6.8.32)

$$+ \partial_{\mu} \chi e^{\mu a} \left( \vartheta j^{b} - j^{0} a^{b} + T^{b} - m^{b} T^{0} - e^{\rho b} S^{0}_{\rho} \right).$$
(6.8.33)

Here we recognize the two last terms as two copies of the boost Ward identity<sup>5</sup> (6.8.30), implying that the rotation Ward identity can be written covariantly as

$$\hat{e}^{[a]}_{(\chi)\nu} e^{b]\mu} T_{(\chi,\hat{a})}{}^{\nu}{}_{\mu} = 0.$$
(6.8.34)

This statement is the non-relativistic analogue of symmetry of the energy-momentum tensor, and we see again that the Ward identity can be expressed compactly in terms of the GED-augmented HIM stress tensor.

## 6.8.2.3 *Gauge Transformations*

For gauge transformations  $\delta_{\Gamma}$ , the relevant transformations are

$$\delta_{\Gamma}\tilde{\varphi} = v^{\mu}\partial_{\mu}\Gamma_{(0)}, \qquad \delta_{\Gamma}a_a = e_a^{\mu}\partial_{\mu}\Gamma_{(0)}, \tag{6.8.35}$$

which means that we get

$$0 = \int d^{3}x \ e \left( -j^{0}v^{\mu}\partial_{\mu}\Gamma_{(0)} + j^{a}e^{\mu}_{a}\partial_{\mu}\Gamma_{(0)} \right) = -\int d^{3}x \ e \frac{\Gamma_{(0)}}{e}\partial_{\mu} \left[ e \left( -j^{0}v^{\mu} + j^{a}e^{\mu}_{a} \right) \right] = -\int d^{3}x \ e \frac{\Gamma_{(0)}}{e}\partial_{\mu} \left( ej^{\mu} \right) \left[ e^{-\frac{1}{2}} \left( -j^{0}v^{\mu} + j^{a}e^{\mu}_{a} \right) \right] = -\int d^{3}x \ e \frac{\Gamma_{(0)}}{e}\partial_{\mu} \left( ej^{\mu} \right) \left[ e^{-\frac{1}{2}} \left( -j^{0}v^{\mu} + j^{a}e^{\mu}_{a} \right) \right] = -\int d^{3}x \ e \frac{\Gamma_{(0)}}{e}\partial_{\mu} \left( ej^{\mu} \right) \left[ e^{-\frac{1}{2}} \left( -j^{0}v^{\mu} + j^{a}e^{\mu}_{a} \right) \right] = -\int d^{3}x \ e \frac{\Gamma_{(0)}}{e}\partial_{\mu} \left( ej^{\mu} + j^{a}e^{\mu}_{a} \right) \left[ e^{-\frac{1}{2}} \left( -j^{0}v^{\mu} + j^{a}e^{\mu}_{a} \right) \right] = -\int d^{3}x \ e \frac{\Gamma_{(0)}}{e}\partial_{\mu} \left( ej^{\mu} + j^{a}e^{\mu}_{a} \right) \left[ e^{-\frac{1}{2}} \left( -j^{0}v^{\mu} + j^{a}e^{\mu}_{a} \right) \right] = -\int d^{3}x \ e \frac{\Gamma_{(0)}}{e}\partial_{\mu} \left( ej^{\mu} + j^{a}e^{\mu}_{a} \right) \left[ e^{-\frac{1}{2}} \left( -j^{0}v^{\mu} + j^{a}e^{\mu}_{a} \right) \right] = -\int d^{3}x \ e \frac{\Gamma_{(0)}}{e}\partial_{\mu} \left( ej^{\mu} + j^{a}e^{\mu}_{a} \right) \left[ e^{-\frac{1}{2}} \left( -j^{0}v^{\mu} + j^{a}e^{\mu}_{a} \right) \right] = -\int d^{3}x \ e \frac{\Gamma_{(0)}}{e}\partial_{\mu} \left[ e^{-\frac{1}{2}} \left( -j^{0}v^{\mu} + j^{a}e^{\mu}_{a} \right) \right] \left[ e^{-\frac{1}{2}} \left( -j^{0}v^{\mu} + j^{a}e^{\mu}_{a}$$

that is to say, the  $U_{\Gamma}(1)$  Ward identity is

$$\frac{1}{e}\partial_{\mu}\left(ej^{\mu}\right) = 0. \tag{6.8.37}$$

This expresses the TNC analogue of conservation of the GED current. Note that we can also write this in the form

$$0 = \mathring{\nabla}_{\mu} j^{\mu} = \nabla_{\mu} j^{\mu} + \overline{\Gamma}^{\nu}_{[\nu\mu]} j^{\mu}.$$
 (6.8.38)

We note that [36] finds an identical Ward identity.

<sup>5</sup> Recall that the same happened in our treatment of pure Lifshitz holography, cf. section 5.3.3.

## 6.8.2.4 Stückelberg Gauge Transformations

In this case, the relevant field transformations are

$$\delta_N \chi = \sigma, \qquad \delta_N m_\mu = \partial_\mu \sigma \Rightarrow \delta_N m_0 = -v^\mu \partial_\mu \sigma, \quad \delta_N m_a = e_a^\mu \partial_\mu \sigma.$$
 (6.8.39)

Plugging this into the expression for  $\delta S$  in (6.8.18), we get

$$0 = \int d^3x \ e \left( T^{\mu} \partial_{\mu} \sigma + \langle O_{\chi} \rangle \, \sigma \right) \tag{6.8.40}$$

$$= \int d^3x \ e \left( -\frac{\sigma}{e} \partial_\mu \left( e T^\mu \right) + \langle O_\chi \rangle \right), \tag{6.8.41}$$

which implies the Ward identity

$$\langle O_{\chi} \rangle = \frac{1}{e} \partial_{\mu} (eT^{\mu}). \tag{6.8.42}$$

Since the left-hand side is generically different from zero, this breaks mass current conservation.

#### 6.8.2.5 Weyl Transformations

The relevant transformations read in this case

$$\delta_D v^\mu = -2\Lambda_D v^\mu, \qquad \delta_D e^\mu_a = -\Lambda_D e^\mu_a, \qquad \delta_D m_0 = -2\Lambda_D m_0, \qquad \delta_D m_a = -\Lambda_D m_a \quad (6.8.43)$$
  
$$\delta_D \tilde{\varphi} = -2\Lambda_D \tilde{\varphi}, \qquad \delta_D a_a = -\Lambda_D a_a, \qquad \delta_D r = -\Lambda_D r. \quad (6.8.44)$$

This produces the Weyl Ward Identity

$$\mathcal{A}_{\rm red}^{(0)} = -2S^0_{\mu}v^{\mu} + S^a_{\mu}e^{\mu}_a + 2T^0m_0 + T^am_a - 2j^0\tilde{\varphi} + j^aa_a.$$
(6.8.45)

The right-hand side can be rewritten in the following manner:

$$-2v^{\mu}\tau_{\nu}\left(T^{\nu}{}_{\mu}+j^{\nu}\hat{a}_{\mu}\right)+e^{\mu}_{a}e^{a}_{\nu}\left(T^{\mu}{}_{\nu}+j^{\nu}\hat{a}_{\mu}\right)+2\tilde{\Phi}\tau_{\mu}\left(T^{\mu}+j^{\mu}\vartheta\right)+m_{a}\left(-m^{a}T^{0}+T^{a}-m^{a}j^{0}\vartheta+\vartheta j^{a}\right).$$
(6.8.46)

Using the boost Ward identity in the form (6.8.20), the part proportional to  $m_a$  takes the form

$$m_a \left( S^0_{\mu} e^{\mu a} + j^0 \left( a^a - m^a \vartheta \right) \right) = m_a \tau_{\nu} e^{\mu a} \left[ T^{\nu}{}_{\mu} + j^{\nu} \hat{a}_{\mu} \right], \qquad (6.8.47)$$

which means that we can further rewrite the Weyl Ward identity (6.8.45) as

$$\mathcal{A}_{\rm red}^{(0)} = -2\hat{v}^{\mu}\tau_{\nu}\left(T^{\nu}{}_{\mu}+j^{\nu}\hat{a}_{\mu}\right) + e^{\mu}_{a}\hat{e}^{a}_{\nu}\left(T^{\mu}{}_{\nu}+j^{\nu}\hat{a}_{\mu}\right) + 2\tilde{\Phi}\tau_{\mu}\left(T^{\mu}+j^{\mu}\vartheta\right)$$
(6.8.48)

$$= \left(e_{a}^{\mu}\hat{e}_{\nu}^{a} - 2\hat{v}^{\mu}\tau_{\nu}\right)T_{(\hat{a})}^{\mu}{}_{\nu} + 2\tilde{\Phi}\tau_{\mu}T_{(\hat{a})}^{\mu}.$$
(6.8.49)

The first term in (6.8.49) is the z = 2 deformed trace, while the second can be interpreted as a form of potential energy since  $T^0$  is roughly the mass, while  $\tilde{\Phi}$  plays the rôle of the Newtonian potential.

## 6.8.2.6 *Diffeomorphisms*

All the fields transform under infinitesimal boundary diffeomorphisms generated by  $\xi$ , but there are only two types of transformations depending on the number of space-time indices:

$$\delta_{\xi} X = \xi^{\nu} \partial_{\nu} X, \qquad \delta_{\xi} V^{\mu} = \xi^{\nu} \partial_{\nu} V^{\mu} - V^{\nu} \partial_{\nu} \xi^{\mu}. \tag{6.8.50}$$

In particular, the only fields carrying space-time indices are  $v^{\mu}$  and  $e^{\mu}_{a}$ , and we note that due their transformation properties it will be necessary to integrate by parts in order to obtain the Ward identity. This procedure is straightforward and gives the Ward identity

$$0 = \frac{1}{e} \partial_{\mu} \left( eT^{\mu}{}_{\nu} \right) - S^{0}_{\mu} \partial_{\nu} v^{\mu} + S^{a}_{\mu} \partial_{\nu} e^{\mu}_{a} + T^{0} \partial_{\nu} m_{0} + T^{a} \partial_{\nu} m_{a} + \langle O_{\chi} \rangle \partial_{\nu} \chi + j^{a} \partial_{\nu} a_{a} - j^{0} \partial_{\nu} \tilde{\varphi} + \langle O_{\vartheta} \rangle \partial_{\nu} \vartheta$$

$$(6.8.51)$$

$$= \frac{1}{e}\partial_{\mu}\left(e\left(T^{\mu}{}_{\nu}+T^{\mu}\partial_{\nu}\chi\right)\right) - S^{0}_{\mu}\partial_{\nu}v^{\mu} + S^{a}_{\mu}\partial_{\nu}e^{\mu}_{a} + T^{0}\partial_{\nu}m_{0} + T^{a}\partial_{\nu}m_{a} + j^{a}\partial_{\nu}a_{a} - j^{0}\partial_{\nu}\tilde{\varphi}.$$
(6.8.52)

We will consider the covariant version of this result from the perspective of null reduction in the next section.

### 6.8.3 Ward Identities From Null Reduction

The five-dimensional Ward identities read

$$t^{A}{}_{A} = \mathcal{A}^{(0)}_{5\mathrm{D}}, \qquad \mathring{\nabla}_{A} t^{A}{}_{B} + \mathcal{F}_{(0)BA} \mathcal{J}^{A}_{(0)} + \langle O_{\psi} \rangle \,\partial_{B} \psi_{(0)} = 0, \qquad \mathring{\nabla}_{A} \mathcal{J}^{A}_{(0)} = 0, \tag{6.8.53}$$

where  $\mathcal{A}_{5D}^{(0)}$  is the boundary value of the Weyl anomaly, which we emphasize has nothing to with the five-dimensional gauge field  $\mathcal{A}_A$ , and  $\mathring{\nabla}$  is the covariant derivative of  $\gamma_{(0)}$ . Reducing the last of these gives us

$$\partial_{\mu}\left(ej^{\mu}\right) = 0,\tag{6.8.54}$$

as we found in (6.8.37). Note that this does not put any constraints on  $\mathcal{J}_{(0)}^{u}$ . Now, recall that

$$t^{\mu}{}_{\nu} = T^{\mu}{}_{\nu} + j^{\mu}\hat{a}_{\nu}, \qquad t^{\mu}{}_{u} = T^{\mu} + j^{\mu}\vartheta.$$
(6.8.55)

The u-component of the higher-dimensional diffeomorphism Ward identity then takes the form

$$0 = \frac{1}{e} \partial_{\mu} \left( e t^{\mu}{}_{u} \right) + \mathcal{F}_{(0)u\mu} \mathcal{J}^{\mu}_{(0)} + \langle O_{\psi} \rangle \, \partial_{u} \psi_{(0)} = \frac{1}{e} \partial_{\mu} \left( e \left[ T^{\mu} + j^{\mu} \vartheta \right] \right) - j^{\mu} \partial_{\mu} \vartheta - \langle O_{\chi} \rangle \tag{6.8.56}$$

$$=\frac{1}{e}\partial_{\mu}\left(eT^{\mu}\right)+j^{\mu}\partial_{\mu}\vartheta+\frac{\vartheta}{e}\underbrace{\partial_{\mu}\left(ej^{\mu}\right)}_{=0}-j^{\mu}\partial_{\mu}\vartheta-\left\langle O_{\chi}\right\rangle,$$
(6.8.57)

leading to the Stückelberg Ward identity (6.8.42) found previously.

We now begin our reduction of the Weyl Ward identity (using  $\mathcal{A}_{5D}^{(0)} = \mathcal{A}_{red}^{(0)}$ ):

$$\mathcal{A}_{\text{red}}^{(0)} = \gamma_{(0)}^{AB} t_{AB} = \gamma_{(0)}^{\mu\nu} t_{\mu\nu} + 2\gamma_{(0)}^{\mu\mu} t_{u\mu} + \gamma_{(0)}^{uu} t_{uu}$$
(6.8.58)

$$=h^{\mu\nu}t_{\mu\nu}-2\hat{v}^{\mu}t_{\mu\mu}+2\Phi t_{\mu\mu}$$
(6.8.59)

$$= e_a^{\mu} S_{\mu}^{a} + j^{a} \hat{a}_{\mu} e_a^{\mu} - \left(S_{\mu}^{0} + j^{0} \hat{a}_{\mu}\right) m^{a} e_a^{\mu} - 2\hat{v}^{\mu} \left(S_{\mu}^{0} + j^{0} \hat{a}_{\mu}\right) + 2\tilde{\Phi} \left(T^{0} + \vartheta j^{0}\right)$$
(6.8.60)

$$= (T^{\mu}{}_{\nu} + j^{\mu}\hat{a}_{\nu}) \left[ e^{\nu}_{a}\hat{e}^{a}_{\mu} - 2\hat{v}^{\nu}\tau_{\mu} \right] + 2\tilde{\Phi}\tau_{\mu} \left( T^{\mu} + \vartheta j^{\mu} \right)$$
(6.8.61)

$$=T^{\mu}_{(\hat{a})\nu}\left[e^{\nu}_{a}\hat{e}^{a}_{\mu}-2\hat{v}^{\nu}\tau_{\mu}\right]+2\tilde{\Phi}\tau_{\mu}T^{\mu}_{(\hat{a})},$$
(6.8.62)

in agreement with our previous result (6.8.49).

Turning to the diffeomorphism Ward identity, we observe

$$\mathring{\nabla}_{A}t^{A}{}_{\mu} = \partial_{\nu}t^{\nu}{}_{\mu} + \mathring{\Gamma}^{\nu}{}_{\nu\lambda}t^{\lambda}{}_{\mu} - \mathring{\Gamma}^{\lambda}{}_{\nu\mu}t^{\nu}{}_{\lambda} + \mathring{\Gamma}^{u}{}_{u\lambda}t^{\lambda}{}_{\mu} - \mathring{\Gamma}^{\lambda}{}_{u\mu}t^{u}{}_{\lambda}$$
(6.8.63)

$$= \nabla_{\nu} T^{\nu}_{(\hat{a})\mu} + \overline{\Gamma}^{\lambda}_{[\nu\mu]} T^{\nu}_{(\hat{a})\lambda} - 2\overline{\Gamma}^{\nu}_{[\nu\lambda]} T^{\lambda}_{(\hat{a})\mu} + h^{\lambda\rho} \partial_{(\mu} \tau_{\rho)} \left[ 2\tilde{\Phi} T^{\sigma}_{(\hat{a})} \bar{h}_{\sigma\lambda} - \hat{v}^{\eta} T^{\sigma}_{(\hat{a})\eta} \bar{h}_{\sigma\lambda} \right].$$
(6.8.64)

Now, defining  $\hat{f}_{\mu\nu} = 2\partial_{(\mu}\hat{a}_{\nu)}$  and using the relation (6.7.59) for  $\mathcal{J}_{(0)}^{\mu}$ , the rest of the diffeomorphism Ward identity can be written as

$$\mathcal{F}_{(0)\mu A}\mathcal{J}^{A}_{(0)} + \langle O_{\psi}\rangle \,\partial_{\mu}\psi_{(0)} = \hat{f}_{\mu\nu}j^{\nu} + j^{\nu}m_{\nu}\partial_{\mu}\vartheta + \langle O_{\chi}\rangle \,\partial_{\mu}\chi \tag{6.8.65}$$

$$= \hat{f}_{\mu\nu}j^{\nu} + j^{\nu}m_{\nu}\partial_{\mu}\vartheta + \mathring{\nabla}_{\nu}T^{\nu}\partial_{\mu}\chi.$$
(6.8.66)

Results from [57] indicate that the diffeomorphism Ward identity can also be covariantly expressed in the form

$$0 = e^{-1} \left[ \partial_{\nu} \left( eT^{\nu}_{(\hat{a})\mu} \right) + \partial_{\nu} \left( eT^{\nu} \right) \partial_{\mu} \chi \right] + T^{\rho}_{(\hat{a}\nu)} \left( \hat{v}^{\nu} \partial_{\mu} \tau_{\rho} - e^{\nu}_{a} \partial_{\mu} \hat{e}^{a}_{\rho} \right) + \tau_{\nu} T^{\nu}_{(\hat{a})} \partial_{\mu} \tilde{\Phi} + j^{\nu} \left( \hat{f}_{\mu\nu} + m_{\nu} \partial_{\mu} \vartheta \right).$$

$$(6.8.67)$$

#### 6.9 GENERAL-Z CONJECTURE

Based on the analysis for z = 2 above and the dimensional analysis of section 6.1.3, we can put forward a general conjecture for charged Lifshitz holography with z > 1.

# 6.9.1 The Action

Amalgamating our starting point for pure Lifshitz holography (5.2.1) in chapter (5) with our specific upliftable model (6.2.21), we take as our starting point the action

$$S = \int d^{4}x \,\sqrt{-g} \left( R - \frac{1}{4} Z_{(F)}(\Phi, Y) F^{2} - \frac{1}{2} W(\Phi, Y) B^{2} - \frac{x}{2} (\partial \Phi)^{2} - \frac{1}{4} Z_{(H)}(\Phi, Y) H^{2} \right)$$

$$- \frac{1}{2} Z_{(F,H)}(\Phi, Y) F_{MN} H^{MN} - \frac{1}{2} Z_{(Y)}(\partial Y)^{2} + V(\Phi, Y) \right).$$
(6.9.1)
(6.9.2)

We have kept the dependency on Y general in the action above. The functions  $Z_{(i)}$  and W are restricted to be positive and to be of such a form that the function  $\alpha$  does not depend on Y. If we did not do that, we would no longer find that  $\alpha \simeq \alpha_{(0)} + \cdots$ , since, based on our dimensional analysis in section 6.1.3, this field behaves as  $Y \simeq r^{z/2-1}\vartheta$ , so for 1 < z < 2, the leading component would no longer be  $\mathcal{O}(r^0)$ . Whenever  $z \ge 2$ ,  $\alpha$  can depend on Y, but we will not explicitly consider this. The potential V is negative close to a Lifshitz solution.

We expect in analogy with our analysis in this chapter that this action admits solutions of the form

$$ds^{2} = -\frac{1}{r^{2z}}dt^{2} + \frac{1}{r^{2}}\left(dr^{2} + dx^{2} + dy^{2}\right),$$
(6.9.3)

$$B = A_0 \frac{1}{r^z} dt, \qquad C = -\frac{\beta(\Phi, Y)}{\alpha(\Phi)} A_0 \frac{1}{r^z} dt, \qquad Y = r^{z/2 - 1} Y_{\star}, \qquad \Phi = \Phi_{\star}, \tag{6.9.4}$$

where  $\beta(\Phi, \Upsilon)$  is a suitable generalization of the combination  $\Upsilon \alpha(\Phi)$ , which appears in the upliftable model.

#### 6.9.2 Sources & VEVs

The metric and Proca field are expressed as in chapter 5,

$$ds^{2} = \frac{dr^{2}}{R(\Phi)r^{2}} - E^{0}E^{0} + \delta_{ab}E^{a}E^{b}, \qquad B_{M} = A_{M} - \partial_{M}\Xi,$$
(6.9.5)

Invoking the equivalence principle, we can translate the results of section 6.1.3 for the scaling weights of the GED fields on a flat background to the scaling weights of the frame components of the sources, which we saw matched our holographic results for z = 2. Our boundary conditions, suitably generalized, can therefore be expressed in the form (where  $\Delta \ge 0$ )

$$E_0^{\mu} \simeq -r^z \alpha_{(0)}^{-1/3} v^{\mu}, \tag{6.9.6}$$

$$E_a^{\mu} \simeq r \alpha_{(0)}^{1/3} e_a^{\mu}, \tag{6.9.7}$$

$$A_{\mu} - \alpha(\Phi) E^0_{\mu} \simeq -r^{z-2} \tilde{m}_{\mu}, \qquad (6.9.8)$$

$$\Phi \simeq r^{\Delta}\phi, \tag{6.9.9}$$

$$\Xi \simeq -r^{2-2}\chi,\tag{6.9.10}$$

$$A_r \simeq -(z-2)r^{z-3}\chi, \tag{6.9.11}$$

$$Y \simeq r^{2/2} \quad \psi \tag{6.9.12}$$

$$E_0^{\mu}C_{\mu} + \Upsilon\alpha(\Phi) \simeq -r^{z/2+1}\alpha_{(0)}^{-1/3}\tilde{\varphi}, \qquad (6.9.13)$$

$$E_a^{\mu}C_{\mu} \simeq r^{z/2} \alpha_{(0)}^{1/3} a_a. \tag{6.9.14}$$

Note that, as was the case in chapter 5,  $\alpha_{(0)}$  is only a function of  $\phi$  when  $\Delta = 0$ , which, as explicitly demonstrated in section 6.7.2, leads to additional contributions to the VEV  $\langle \tilde{O}_{\phi} \rangle$ .

These generalized sources transform according to

$$\delta e^a_\mu = \pounds_{\xi} e^a_\mu + \lambda^a \tau_\mu + \lambda^a{}_b e^b_\mu + \Lambda_D e^a_\mu. \tag{6.9.15}$$

$$\delta \tau_{\mu} = \pounds_{\xi} \tau_{\mu} + z \Lambda_D \tau_{\mu}, \tag{6.9.16}$$

$$\delta\chi = \pounds_{\xi}\chi + \sigma - (z-2)\Lambda_D\chi, \tag{6.9.17}$$

$$\delta v^{\mu} = \pounds_{\xi} v^{\mu} + \lambda^a e^{\mu}_a - z \Lambda_D v^{\mu}, \qquad (6.9.18)$$

$$\delta e_a^{\mu} = \pounds_{\xi} e_a^{\mu} + \lambda_a^{\ b} e_b^{\mu} - \Lambda_D, \tag{6.9.19}$$

$$\delta\phi = \pounds_{\xi}\phi - \Delta\Lambda_D\phi, \tag{6.9.20}$$

$$\delta \tilde{m}_{\mu} = \pounds_{\xi} m_{\mu} + e^{\mu}_{\mu} \lambda_a + \partial_{\mu} \sigma - (z - 2) \Lambda_D \tilde{m}_{\mu}, \qquad (6.9.21)$$

$$\delta\vartheta = \pounds_{\xi}\vartheta - (z/2 - 1)\Lambda_D\vartheta, \tag{6.9.22}$$

$$\delta\tilde{\varphi} = \pounds_{\xi}\tilde{\varphi} + a_a\lambda^a + v^\mu\partial_\mu\Gamma_{(0)} - (z/2+1)\Lambda_D\tilde{\varphi}, \qquad (6.9.23)$$

$$\delta a_{\mu} = \pounds_{\xi} a_{\mu} + \vartheta \lambda_a e^a_{\mu} + \tau_{\mu} a_{\nu} e^{\nu}_a \lambda^a + \partial_{\mu} \Gamma_{(0)} + \tau_{\mu} v^{\nu} \partial_{\nu} \Gamma_{(0)} - (z/2 - 1) \Lambda_D a_{\mu}, \tag{6.9.24}$$

$$\delta a_a = \pounds_{\xi} a_a + \vartheta \lambda_a + \lambda_a{}^b a_b - \Lambda_D a_a + e_a^{\mu} \partial_{\mu} \Gamma_{(0)} - z/2\Lambda_D a_a.$$
(6.9.25)

Similarly, as we did in chapter 5, we can assume the existence of a renormalized bulk action write the variation of the on-shell renormalized action as  $\delta S_{\text{ren}} = -\int_{\partial \mathcal{M}} d^3 x \ e \ \left( \mathcal{V} \delta \mathcal{X} - \mathcal{A} \frac{\delta r}{r} \right)$ , where the source complex is given by  $\mathcal{X} = \{E^0_{\mu}, E^{\mu}_{a}, \varphi, A_{a}, \Xi, \Phi, \Omega, C_{a}, Y\}$ , and the responses are collected in the complex  $\mathcal{V} = \{S^0_{\mu}, S^a_{\mu}, \mathcal{T}_{\varphi}, \mathcal{T}^a, \mathcal{T}_{\Xi}, \mathcal{T}_{\Phi}, \mathcal{T}_{Y}\}$ , whose leading terms are the VEVs:

$$S^0_{\mu} \simeq r^2 \alpha^{2/3}_{(0)} S^0_{\mu}, \tag{6.9.26}$$

$$S^a_{\mu} \simeq r^{z+1} S^a_{\mu}$$
, (6.9.27)

$$\mathcal{T}_{\varphi} \simeq r^{4-z} \alpha_{(0)}^{2/3} T^0, \tag{6.9.28}$$

$$\mathcal{T}^a \simeq r^3 T^a, \tag{6.9.29}$$

$$\mathcal{T}_{\Xi} \simeq r^4 \alpha_{(0)}^{1/3} \left\langle O_{\chi} \right\rangle, \tag{6.9.30}$$

$$\mathcal{T}_{\Phi} \simeq r^{z+2-\Delta} \alpha_{(0)}^{1/3} \langle O_{\phi} \rangle , \qquad (6.9.31)$$

$$\mathcal{T}_{\Omega} \simeq r^{z/2+1} \alpha_{(0)}^{2/3} j^0, \tag{6.9.32}$$

$$\mathcal{T}^a \simeq r^{z/2+2} j^a, \tag{6.9.33}$$

$$\mathcal{T}_{Y} \simeq r^{z/2+3} \alpha_{(0)}^{1/3} \left\langle O_{\vartheta} \right\rangle, \tag{6.9.34}$$

where we have, once more, used the results of the field theory dimensional analysis of section 6.1.3. Thus, the variation of the renormalized on-shell action becomes<sup>6</sup>:

$$\delta S_{\rm ren} = \int d^3 x \ e \ \left[ -S^0_\mu \delta v^\mu + S^a_\mu \delta e^\mu_a + T^0 \delta \tilde{m}_0 + T^a \delta \tilde{m}_a + \langle O_\chi \rangle \ \delta \chi + \langle \tilde{O}_\phi \rangle \ \delta \phi - j^0 \delta \tilde{\phi} + j^a \delta a_a + \langle O_\vartheta \rangle \ \delta \vartheta - \mathcal{A}_{(0)} \frac{\delta r}{r} \right],$$

$$(6.9.35)$$

where

$$\langle \tilde{O}_{\phi} \rangle = \langle O_{\phi} \rangle + \delta_{\Delta,0} \frac{1}{3} \left[ v^{\mu} (S^{0}_{\mu} + T^{0} m_{\mu}) + e^{\mu}_{a} (S^{a}_{\mu} + T^{a} m_{\mu}) + j^{a} a_{a} - j^{0} \tilde{\varphi} \right] \frac{d \log \alpha_{(0)}}{d\phi} = 0. \quad (6.9.36)$$

<sup>6</sup> In this section, we use the symbol  $A_{(0)}$  for the anomaly, which in our z = 2 was the leading part of the five-dimensional gauge field. We hope that this does not cause confusion.

The VEVs transform under the Schrödinger group in the following manner

$$\delta S^0_{\mu} = \pounds_{\xi} S^0_{\mu} - T^0 \partial_{\mu} \sigma + j^0 \vartheta \partial_{\mu} \sigma - j^0 \partial_{\mu} \Gamma_{(0)} - 2\Lambda_D S^0_{\mu} + \cdots, \qquad (6.9.37)$$

$$\delta S^a_{\mu} = \pounds_{\xi} S^a_{\mu} + \lambda^a S^0_{\mu} + \lambda^a{}_b S^b_{\mu} - T^a \partial_{\mu} \sigma + j^a \vartheta \partial_{\mu} \sigma - j^a \partial_{\mu} \Gamma_{(0)} - (z+1)\Lambda_D S^a_{\mu} + \cdots, \quad (6.9.38)$$

$$\delta T^{0} = \pounds_{\xi} T^{0} - (4 - z)\Lambda_{D} T^{0} + \cdots, \qquad (6.9.39)$$

$$\delta T^a = \pounds_{\xi} T^a + \lambda^a T^0 + \lambda^a{}_b T^b - 3\Lambda_D T^a + \cdots, \qquad (6.9.40)$$

$$\delta \langle O_{\chi} \rangle = \pounds_{\xi} \langle O_{\chi} \rangle - 4\Lambda_D \langle O_{\chi} \rangle + \cdots, \qquad (6.9.41)$$

$$\delta \langle O_{\phi} \rangle = \pounds_{\xi} \langle O_{\phi} \rangle + \delta_{\Delta,0} \frac{d \log \alpha_{(0)}}{d\phi} \left( \lambda_a T^a + \vartheta j^a \lambda_a \right) - (z + 2 - \Delta) \Lambda_D \left\langle O_{\phi} \right\rangle + \cdots, \qquad (6.9.42)$$

$$\delta \langle O_{\vartheta} \rangle = \pounds_{\xi} \langle O_{\vartheta} \rangle - (z/2+3)\Lambda_D \langle O_{\vartheta} \rangle + \cdots, \qquad (6.9.43)$$

$$\delta j^0 = \pounds_{\xi} j^0 - (z/2+2)\Lambda_D j^0 + \cdots, \qquad (6.9.44)$$

$$\delta j^{a} = \pounds_{\xi} j^{a} + \lambda^{a} j^{0} + \lambda^{a} {}_{b} j^{b} - (z/2+2)\Lambda_{D} j^{a} + \cdots, \qquad (6.9.45)$$

where the dots denote possible terms involving derivatives of  $\Lambda_D$ .

## 6.9.3 Ward Identities

As demonstrated in [29, 32] and discussed in section 5.3, the Ward identities in pure Lifshitz holography are the same in all the four cases (5.2.28), although their derivation are differ from case to case. Assuming this also applies for charged Lifshitz holography, we restrict our attention to the case 1 < z < 2 with  $\Delta > 0$ . In this case, the analysis of section 6.8.2 is nearly unchanged: the main difference is that now  $\langle O \rangle_{\vartheta}$  and  $\langle \tilde{O}_{\varphi} \rangle$  are no longer zero. Therefore, we conjecture the following Ward identities

$$\hat{e}^{a}_{(\chi)\mu}T^{\mu}_{(\hat{a})} = \tau_{\nu}e^{\mu a}T^{\nu}_{(\chi,\hat{a})\mu} \qquad \text{(boosts),}$$
(6.9.46)

$$0 = \hat{e}^{[a]}_{(\chi)\nu} e^{b]\mu} T^{\nu}_{(\chi,\hat{a})\mu} \qquad \text{(rotations)}, \tag{6.9.47}$$

$$0 = \frac{1}{e} \partial_{\mu} \left( e j^{\mu} \right) \qquad (U_{\Gamma}(1) \text{ gauge transformations}), \tag{6.9.48}$$

$$\langle O_{\chi} \rangle = \frac{1}{\rho} \partial_{\mu} (eT^{\mu})$$
 (Stückelberg gauge transformations), (6.9.49)

$$\mathcal{A}_{(0)} = -zS^{0}_{\mu}v^{\mu} + S^{a}_{\mu}e^{\mu}_{a} + (2z-2)T^{0}m_{0} + (z-1)T^{a}m_{a} + (z-2)\langle O_{\chi}\rangle\chi$$
(6.9.50)

$$+\Delta \langle \tilde{O}_{\phi} \rangle \phi - (z/2+1)j^0 \tilde{\varphi} + \frac{z}{2}j^a a_a + (z/2-1) \langle O_{\vartheta} \rangle \vartheta \qquad \text{(Weyl transformations),}$$

$$0 = \frac{1}{e} \partial_{\mu} \left( e T^{\mu}_{(\chi)\nu} \right) - S^{0}_{\mu} \partial_{\nu} v^{\mu} + S^{a}_{\mu} \partial_{\nu} e^{\mu}_{a} + T^{0} \partial_{\nu} m_{0}$$

$$(6.9.52)$$

+ 
$$T^a \partial_\nu m_a + j^a \partial_\nu a_a - j^0 \partial_\nu \tilde{\varphi} + \langle O_\vartheta \rangle \partial_\nu \vartheta + \langle \tilde{O}_\phi \rangle \partial_\nu \phi$$
 (Diffeomorphisms). (6.9.53)
# CONCLUSION & OUTLOOK

In this thesis, we have developed charged Lifshitz holography for z = 2 by Scherk-Schwarz reducing the electromagnetic uplift, which we holographically renormalized to obtain a novel counterterm. We have shown that the new sources transform as the fields of Galilean electrodynamics, and we have worked out the VEVs and the corresponding Ward identities. We have shown that the boundary geometry becomes TNC and that fixing the boundary value of the dilaton  $\phi = 0$  turns the boundary geometry into TTNC, and we have commented on the relation to the gauging of the Schrödinger algebra.

Our results open up a number of very interesting avenues for future research: one immediate thing to do would be the explicit verification of the general-*z* charged Lifshitz holography conjecture made in section 6.9 using the methods of [32]: once it has been verified that (6.9.3)–(6.9.4) provides a solution to the general model (6.9.2), the identification of sources and VEVs as well as the computation of the Ward identities will match our results.

With an understanding of charged Lifshitz holography in conjunction with a suitably generalized hydrodynamic analysis à la [57], we would be able to undertake a promising first foray into what we may dub *Lif/CMT*. In particular, given the link between TNC geometry and certain condensed matter problems, this for example paves the way for a Lifshitz holographic realization of the fractional quantum hall effect (FQHE).

Another possibly worthwhile path of exploration—also related to hydrodynamics—would be to investigate if a hydrodynamical analysis of charged z = 0 Schrödinger black branes in the uplift can be null reduced to give an interesting perspective on existing results [156] on non-relativistic hydrodynamics where the boost symmetry is intact.

In another direction, it would be interesting to develop a notion of holography with HL gravity in the bulk, especially given the connection between dynamical TNC geometry and HL gravity [53] and to see if the boundary geometry in this scenario is also described by TNC geometry. This again has ties to the FQHE [38]. In a similar vein, it would be interesting the explore the holographic implications of bulk HL gravity coupled to GED on a TNC geometry. As we have demonstrated, such an action arises from the Weyl anomaly of our z = 2 charged Lifshitz model.

It would also be fruitful to explore the holographic rôle of other non-Lorentzian geometries. In particular it was recently discovered in [157] that by null reducing the Polyakov action in target space and sending the string tension to zero, a novel type of non-Lorentzian geometry called U(1)-Galilean geometry emerges. Whereas NC geometry arises by gauging the Galilei algebra which arises as a  $c \rightarrow \infty$  contraction of the Poincaré algebra, as discussed in chapter 4, one may instead take the limit  $c \rightarrow 0$  to obtain the Carroll algebra, which was gauged in [121] to obtain Carrollian geometry. It was shown in [158] that the BMS group is the conformal extension of the Carroll group, and the relation to flat space holography was explored in [159]. The boundary geometry for Lifshitz space-times with z < 1 would be Carroll geometry, but such space-times violate the null energy condition and as such are not usually considered.

Extending the analysis of the holographic superconductor (cf. appendix E) using the version of Lifshitz holography developed in this thesis would also be worthwhile. In this context, the results of [160] could be useful.

There are other models admitting Lifshitz solutions. In particular, setting W = 0 in the EPD action (5.2.1) results in an EMD model, where the Lifshitz solutions are supported by a Maxwell field and a logarithmically running dilaton [22]. Lifshitz hydrodynamics in this theory was explored in [148, 149], and it would be interesting to also add additional charged fields to this analysis and working out the consequences.

In regards to holographic renormalization, as we described in chapter 3, it would be interesting to work out the properties of *p*-form fields in general and verify the conjecture put forward in section 3.3.

We also point out that in the Lifshitz holography literature, there is a disagreement about the scaling weight of a scalar  $\psi$  that appears in metric analyses of Einstein-Proca models [27] as well as of EPD models [28, 106]. This discrepancy carries over to the charged Lifshitz holography that we have developed in this thesis.

In particular, [27, 28, 106] argue that  $\psi$  has scaling weight  $\Delta_{-}$  given by

$$\frac{1}{2}(z+2-\beta_z), \qquad \beta_z^2 = (z+2)^2 + 8(z-1)(z-2). \tag{7.0.1}$$

This source is related to  $\tilde{\Phi}_{(\chi)} = -v^{\mu}M_{\mu} + \frac{1}{2}h^{\mu\nu}M_{\mu}M_{\nu}$  [33, 57], which, from our analysis in chapter 5, we know has scaling weight 2(z-1), which only agrees with

$$\delta_D M_\mu = -(z-2)\Lambda_D M_\mu, \qquad \delta_D v^\mu = -z\Lambda_D v^\mu, \tag{7.0.2}$$

implying that  $\tilde{\Phi}$  has scaling weight 2(z-1). Thus, the scaling weight of  $\psi$  in (7.0.1) only agrees with that of  $\tilde{\Phi}_{(\chi)}$  when z = 1. It would therefore be interesting to work out the precise relation between  $\tilde{\Phi}$  and  $\psi$ . In particular, a possible resolution could involve the scalar field  $\Phi \simeq r^{\Delta}\phi$ , where the scaling weight  $\Delta$  can in principle be computed, and it could be that  $\psi$  is related to a combination of  $\phi$  and  $\tilde{\Phi}_{(\chi)}$ .

# APPENDIX



In this appendix, we describe anti de-Sitter (AdS) spaces to provide some background for the material discussed in chapter 2. What follows is based on [42, 43, 161].

### A.1 ADS SPACE-TIMES

# A.1.1 Definition and Useful Coordinates

(d+1)-dimensional anti de-Sitter space,  $AdS_{d+1}$ , is the unique maximally symmetric Einstein space<sup>1</sup> with constant negative curvature.  $AdS_{d+1}$  can be embedded into  $\mathbb{R}^{d,2}$  with metric  $ds^2 = -(dX^0)^2 - (dX^{d+1})^2 + \sum_{i=1}^{d} (dX^i)^2$  as the quadric,

$$-(X^{0})^{2} - (X^{d+1})^{2} + \sum_{i=1}^{d} (X^{i})^{2} = -L^{2},$$
(A.1.1)

where *L* is the radius of curvature of  $\operatorname{AdS}_{d+1}$ . The embedding is clearly invariant under the "Lorentz group" for  $\mathbb{R}^{d,2}$ , SO(d, 2), which has dimension  $\frac{1}{2}(d+1)(d+2)$ —this is the number of Killing vectors associated to  $\operatorname{AdS}_{d+1}$ , leading us to conclude that  $\operatorname{AdS}_{d+1}$  is maximally symmetric. SO(d, 2) is the conformal group (see B) of *d*-dimensional Minkowski space. We now turn to describe some explicit parametrizations satisfying the AdS quadric (A.1.1); first, introduce  $t \in \mathbb{R}$ ,  $\vec{x} = (x^1, \ldots, x^{d-1}) \in \mathbb{R}^{d-1}$  and, finally,  $r \in \mathbb{R}^+$ , in terms of which we have

$$X^{0} = \frac{L^{2}}{2r} \left( 1 + \frac{r^{2}}{L^{4}} \left( \vec{x}^{2} - t^{2} + L^{2} \right) \right),$$
(A.1.2)

$$X^{i} = \frac{rx^{i}}{L}, \ i \in \{1, \dots, d-1\},$$
 (A.1.3)

$$X^{d} = \frac{L^{2}}{2r} \left( 1 + \frac{r^{2}}{L^{2}} \left( \vec{x}^{2} - t^{2} - L^{2} \right) \right),$$
(A.1.4)

$$\mathbf{X}^{d+1} = \frac{rt}{L}.\tag{A.1.5}$$

Since r > 0, we only cover half of  $AdS_{d+1}$ —the local coordinates  $t, r, \vec{x}$  define the *Poincaré patch* (see also figure A.1) coordinates, where the induced metric becomes<sup>2</sup>,

$$ds_{\text{AdS}_{d+1}}^2 = \frac{L^2}{r^2} dr^2 + \frac{r^2}{L^2} \left( -dt^2 + d\vec{x}^2 \right) = \frac{L^2}{r^2} dr^2 + \frac{r^2}{L^2} \underbrace{\eta_{\mu\nu} dx^{\mu} dx^{\nu}}_{=ds_{\mathbb{R}^{d-1,1}}^2},$$
(A.1.7)

where we have recognized the metric of *d*-dimensional Minkowski space. Using this metric, we can calculate the Ricci scalar, which becomes  $R = -\frac{d(d+1)}{L^2}$ , implying that  $L^2$  is indeed the radius of curvature. Another useful form of the Poincaré metric is obtained by inverting the radial coordinate,  $z = L^2/r$ , thus yielding the metric in Poincaré *z*-coordinates,

$$ds^{2} = \frac{L^{2}}{z^{2}} \left( dz^{2} + \eta_{\mu\nu} dx^{\mu} dx^{\nu} \right).$$
 (A.1.8)

Note that the boundary in these coordinates is located at z = 0. When doing holographic renormalization, we use another set of coordinates known as *domain-wall coordinates*, which are nothing but

$$\gamma_{xx} = \eta_{MN} \frac{\partial X^M}{\partial x} \frac{\partial X^N}{\partial x} = -\frac{\partial X^0}{\partial x} \frac{\partial X^0}{\partial x} + \frac{\partial X^d}{\partial x} \frac{\partial X^d}{\partial x} + \underbrace{\frac{\partial X^i}{\partial x} \frac{\partial X^i}{\partial x}}_{\text{only one } X^i \text{ depends on } x} = \frac{r^2}{L^2}, \quad (A.1.6)$$

where  $\eta_{MN}$  is the d + 2 dimensional Minkowski metric of signature (-, +, ..., +, -).

<sup>1</sup> That is, it solves the Einstein equations.

<sup>2</sup> For example, any one of the *xx*-components (to be understood as any of  $d - 1 x^{i'}s$ ) of the induced metric,  $\gamma$ , is calculated as follows:

Gaussian normal coordinates. They are obtained from the Poincaré coordinates of (A.1.8) by setting  $z = e^{-r}$ —which means that the boundary will now be located at  $r \to \infty$ —and they read

$$ds^{2} = dr^{2} + e^{2r} \eta_{\mu\nu} dx^{\mu} dx^{\nu}.$$
 (A.1.9)

It is also possible to introduce *global coordinates*  $\{\tau, \rho, \{\theta_i\}_{i \in \{1,...,d-1\}}\}$  for  $AdS_{d+1}$ : the following parametrization satisfies the  $AdS_{d+1}$  quadric:

$$X^0 = L\cosh\rho\cos\tau,\tag{A.1.10}$$

$$X^{d+1} = L\cosh\rho\sin\tau, \tag{A.1.11}$$

$$X^{1} = L \sinh \rho \, \cos \theta_{1}, \tag{A.1.12}$$

$$X^{2} = L \sinh \rho \, \sin \theta_{1} \cos \theta_{2}, \tag{A.1.13}$$

$$X^{d-1} = L \sinh \rho \, \sin \theta_1 \cdots \sin \theta_{d-2} \cos \theta_{d-1} \tag{A.1.15}$$

$$X^{d} = L \sinh \rho \, \sin \theta_{1} \cdots \sin \theta_{d-2} \sin \theta_{d-1}, \tag{A.1.16}$$

where the  $\{\theta_i\}_{i \in \{1,...,d-1\}}$  precisely parametrize  $S^{d-1}$ . The induced metric becomes,

$$ds_{\text{AdS}_{d+1}}^2 = L^2 \left( -\cosh^2 \rho \ d\tau^2 + d\rho^2 + \sinh^2 \rho \ d\Omega_{\text{S}^{d-1}}^2 \right), \tag{A.1.17}$$

where  $d\Omega_{S^{d-1}}^2$  is the metric of the (d-1)-sphere. Since the metric above does not depend on  $\tau$ , we infer the existence of a timelike killing vector<sup>3</sup>  $\partial_{\tau}$ , and since this killing vector is defined globally on the manifold,  $\tau$  acts as a sensible global time coordinate. Near the center of  $AdS_{d+1}$ ,  $\rho = 0$ , the metric assumes the form  $ds_{AdS_{d+1}}^2 \stackrel{\rho \sim 0}{\simeq} L^2(-d\tau^2 + d\rho^2 + \rho^2 d\Omega_{S^{d-1}}^2)$ , implying that the space-time described by (A.1.17) has topology—since  $\tau$  is periodic—of  $S^1 \times \mathbb{R}^d$ , where  $S^1$  is the periodic time; in particular, since  $\partial_{\tau}$  is everywhere timelike, keeping  $\rho$  and  $\theta_i$  fixed while varying  $\tau$  will produce closed time-like curves. This is, however, not an intrinsic property of the space-time—merely an artefact of our embedding:  $\mathbb{R}^{d,2}$  has two timelike directions, so the appearance of closed timelike curves is not so surprising after all. To get rid these, we may instead choose to define [44, 161] (although some authors, e.g. [43], do not!)  $AdS_{d+1}$  as the *universal covering space*, where we decompactify so that  $\tau$  assumes all real values with no periodicity constraints, which makes  $AdS_{d+1}$  a causal space-time. Since the universal covering of  $AdS_{d+1}$  will have topology  $\mathbb{R}^{d+1}$ . We also want to draw the conformal diagram for  $AdS_{d+1}$ , and for this, we change coordinates again:

$$\cosh \rho = \frac{1}{\cos \phi'},\tag{A.1.18}$$

implying that  $\sinh \rho = \tan \phi$ . In this manner, the metric takes the form

$$ds_{AdS_{d+1}}^2 = \frac{L^2}{\cos^2 \phi} \left( -d\tau^2 + d\phi^2 + \sin^2 \phi \ d\Omega_{S^{d-1}}^2 \right), \tag{A.1.19}$$

which we see implies that  $AdS_{d+1}$  is *conformally flat*<sup>5</sup>. The range of the new radial coordinate is  $0 \le \phi \le \frac{\pi}{2}$ .

<sup>3</sup> Timelike since the  $\tau\tau$ -component of the metric is less than zero.

<sup>4</sup> To see this [162], one can check that  $p : \mathbb{R} \to S^1$  with  $p(t) = (\cos t, \sin t)$  satisfies the properties of a covering map.

<sup>5</sup> A manifold (M,g) is conformally flat if for each  $x \in M$ , there exists a neighborhood U of x as well as a smooth function f on U such that  $(U, e^{2f}g)$  is flat. In our case, we can choose for all  $x \in AdS_{d+1}$  the function,  $e^{2f} = \frac{L^2}{\cos^2 \phi}$ , or  $f = \log L - \log \cos \phi$ , which is smooth.



Figure A.1: Conformal diagram of global  $AdS_{d+1}$  (i.e. we drop the conformal factor  $L^2/\cos\phi$  in the global AdS metric (A.1.17)) with radial direction  $\phi$  and global time  $\tau$ . In the figure, every point  $(\tau, \phi)$  contains a  $S^{d-1}$  and the boundary topology is (set  $\phi = \pi/2$  in (A.1.17)) that of a cylinder,  $\mathbb{R} \times S^{d-1}$ . The Poincaré patch, described by the metric (A.1.8), covers only part of global AdS (the pink area), and its boundary is conformal to  $\mathbb{R}^{d-1,1}$ , i.e. (d-1)-dimensional Minkowski space. See appendix A of [163] for more details.

# A.1.2 Conformally Compact Einstein Manifolds & The Boundary

We now turn to examining the boundary of  $\operatorname{AdS}_{d+1}$ . In particular, the metric in Poincaré coordinates, (A.1.7), we have in the  $r \to 0$  limit a degenerate *Killing horizon*<sup>6</sup>, known as the Poincaré horizon. On the other hand, in the limit  $r \to \infty$ , the metric has a pole of second order. As described in [79, 93, 164]—and in the more rigorous [165]—we now turn to describing a general method for taming the conformal boundary of AdS. Therefore, we let M be the interior<sup>7</sup> of a compact (d + 1)-dimensional manifold  $\overline{M}$  with boundary  $\partial M$  (sometimes denoted  $\mathcal{F}$  [166]). A metric  $g_{\mu\nu}$  on M is *conformally compact* if it has second order pole at the boundary  $\partial M$  and there exists a *defining function*  $\rho$  on  $\overline{M}$ such that the conformally equivalent metric,

$$\tilde{g}_{\mu\nu} = \rho^2 g_{\mu\nu},\tag{A.1.20}$$

smoothly extends to a metric on the compactification<sup>8</sup>  $\overline{M}$ . The defining function is a smooth nonnegative satisfying  $\rho|_{\partial M} = 0$ ,  $\partial_{\mu}\rho|_{\partial M} \neq 0$ . The induced metric  $\gamma_{\mu\nu} = \tilde{g}_{\mu\nu}|_{\partial M}$  is the boundary metric associated to the specific conformal compactification of (A.1.20)—there are generally many possible defining functions, and so many conformal compactifications of a given metric  $g_{\mu\nu}$ , which induces a conformal (equivalence) class of metrics  $[\gamma]$  on  $\partial M$ , known as conformal infinity. A *conformally compact Einstein manifold* is a conformally compact manifold that also solves Einstein's equation. We shall have more to say about this when considering the generalized spaces AAdS and AldS spaces. We can now apply this analysis the  $AdS_{d+1}$  metric in the coordinates of (A.1.19)—which indeed has a second order pole in the radial coordinate  $\phi$  at  $\phi = \pi/2$ . Taking  $\rho = \frac{\cos \phi}{L}$ , we see that it satisfies all the conditions for a defining function, and thus the boundary metric becomes  $\eta_{\mu\nu}$ . Now, let's discuss how conformal transformations act on the boundary. From the quadric, (A.1.1), the full conformal symmetry SO(d, 2)is clear, whereas from the Poincaré metric (A.1.7), only two subgroups of SO(d, 2) are manifest: the set of Poincaré transformations acting on  $t, \vec{x}$ , ISO(d-1, 1), and SO(1, 1), which has the action,

$$(t, \vec{x}, r) \mapsto (\lambda t, \lambda \vec{x}, r/\lambda). \tag{A.1.21}$$

<sup>6</sup> A Killing horizon is a null hypersurface defined by vanishing norm of a Killing vector  $k_{\mu}$ . That it is degenerate just means that the surface gravity vanishes on the hypersurface.

<sup>7</sup> In particular, *M* will be non-compact.

<sup>8</sup> Note that we now use the word in the mathematical sense: it simply refers to the process of making a topological space into a compact space.

Using a light-cone formalism for the conformal algebra and the isomorphism  $SO(d, 2) \simeq ISO(d, 1) \times SO(1, 1)$ , one can make explicit exactly how the symmetries are inherited by the boundary, see [11] for details.

# A.1.3 The Boundary: There and Back Again

In this section, we consider the behaviour of radial geodesics and investigate if they can reach the boundary. Being interested in radial rays, we may set  $d\Omega_{d-1}^2 = 0$ . Starting light rays, we immediately set  $ds^2 = 0$ , allowing us to determine the coordinate time it takes to go to the boundary. Using global coordinates (A.1.17), we find

$$0 = ds^{2} = L^{2}(-\cosh^{2}(\rho)d\tau^{2} + d\rho^{2}) \therefore d\rho^{2} = \cosh^{2}(\rho)d\tau^{2}.$$
 (A.1.22)

This is a differential equation that we can solve:

$$\int_{\rho_0}^{\rho} \frac{1}{\cosh(\rho')} d\rho' = \tau(\rho) - \tau(\rho_0) = \tau(\rho), \tag{A.1.23}$$

i.e.

$$\tau(\rho) = 2 \tan^{-1}[\tanh(\rho/2)] - 2 \tan^{-1}[\tanh(\rho_0/2)].$$
(A.1.24)

The time for a light ray to go to the boundary and back must be  $2t(\infty)$ , which is then

$$2\tau(\infty) = \lim_{\rho \to \infty} 4(\tan^{-1}[\tanh(\rho/2)] - \tan^{-1}[\tanh(\rho_0/2)]) = 4(\pi/4 - \tan^{-1}[\tanh(\rho_0/2)]),$$
(A.1.25)

so for  $\rho_0 = 0$  we find, specifically, that  $2\tau(\infty) = \pi$ .

Note that if we had instead been interested in how much *affine time* it would take the light ray, we would find that the answer diverges.

Now, turning to massive geodesics, we could in principle go on to compute the Christoffel symbols and look at the geodesic equation, but we can obtain some interesting qualitative features of the behaviour of a massive, radially-directed particle just from the normalization of 4-velocity, that is  $u^{\mu}u_{\mu} = -1$ , with  $u^{\mu} = \frac{dx^{\mu}}{d\lambda}$ , where  $\lambda$  is the proper time. We write

$$-1/L^2 = rac{1}{L^2} g_{\mu
u} u^\mu u^
u = -\cosh^2(
ho) \left(rac{\mathrm{d}t}{\mathrm{d}\lambda}
ight)^2 + \left(rac{\mathrm{d}
ho}{\mathrm{d}\lambda}
ight)^2.$$

Note that  $\partial_{\tau}$  is a (time-like) Killing vector of AdS, with components which we will denote  $K^{\mu} = \delta^{\mu}_{\tau}$ , meaning that  $p^{\mu} \nabla_{\mu} (K_{\nu} p^{\nu}) = 0$ . In other words,  $K_{\nu} p^{\nu} = g_{\nu\rho} K^{\rho} \frac{dx^{\nu}}{d\lambda} = g_{\nu\tau} \frac{dx^{\nu}}{d\lambda} = -\cosh^2(\rho) \frac{d\tau}{d\lambda}$  is conserved along the trajectory, i.e.  $\cosh^2(\rho) \frac{d\tau}{d\lambda} = C$ , which means that

$$\left(\frac{\mathrm{d}\rho}{\mathrm{d}\lambda}\right)^2 = C^2/\cosh^2(\rho) - 1/L^2.$$

Notice that  $\frac{d\rho}{d\lambda} = 0$  has a solution  $\rho = \rho^*$  given by

$$\cosh^2(\rho^*) = (LC)^2$$

which means that there is always a turning point beyond which the massive particle cannot probe. The position of this turning point is "energy-dependent" (that is, it is *C*-dependent—recall that *C* is the conserved quantity corresponding to the time-translation symmetry). We find

$$\frac{\mathrm{d}\rho}{\mathrm{d}\lambda} = \pm \frac{1}{L} \sqrt{\frac{(LC)^2}{\cosh^2(\rho)}} - 1,$$

and choosing the positive sign, we obtain

$$L\int_{\rho_0}^{\rho}\left(\frac{(LC)^2}{\cosh^2(\rho')}-1\right)^{-1/2}d\rho'=\lambda(\rho)-\lambda(\rho_0).$$

If we let  $\rho_0 = 0$  and compute  $\lambda(\rho^*) - \lambda(0)$ , we find

$$\lambda(\rho^*) - \lambda(0) = L \int_0^{\rho^*} \left( \frac{(LC)^2}{\cosh^2(\rho')} - 1 \right)^{-1/2} d\rho',$$

so, since  $\rho^* = \cosh^{-1}(LC)$ , we obtain (using Mathematica)

$$\lambda(\cosh^{-1}(LC)) - \lambda(0) = L \int_0^{\cosh^{-1}(LC)} \left(\frac{(LC)^2}{\cosh^2(\rho')} - 1\right)^{-1/2} d\rho' = \pi/2,$$

implying that the turning point  $\rho^*$  for massive particles is analogous to the boundary for massless particles, i.e. it takes proper time  $\pi$  to go from  $\rho = 0$  to the turning point and back again.

# A.2 Alads space-times & the fefferman-graham theorem

Holographically, as we discussed in chapter 2, the energy-momentum tensor of the dual field theory is sourced by the boundary value of the bulk metric. Consequently, a dynamical boundary metric is required in order for the field theory energy-momentum tensor to be well defined as the usual variational derivative. This can be achieved via so-called *asymptotically locally AdS* space-times [48]:

#### **Definition (AldS):** An asymptotically locally AdS space-time is a conformally compact Einstein manifold.

An important subset of AldS space-times are the asymptotically AdS (AAdS) space-times, the boundaries of which are equal to that of AdS; a nice example is the (d + 1)-dimensional AdS-Schwarzschild (static and spherically symmetric) black hole with metric [167]

$$ds^{2} = -f_{d}(r)dt^{2} + f_{d}(r)^{-1}dr^{2} + r^{2}d\Omega_{S^{d-1}}^{2},$$
(A.2.1)

where the blackening factor is  $f_d(r) = \frac{r^2}{L^2} + 1 - \left(\frac{r_h}{r}\right)^{d-3} \left(\frac{r_h^2}{L^2} + 1\right) \xrightarrow{r \to \infty} \frac{r^2}{L^2} + 1$ , which thus reduces to the global AdS metric of (A.1.17) in the limit  $r \to \infty$  upon doing the coordinate transformation  $r \mapsto L \sinh \rho$  and  $t \mapsto L\tau$ , since

$$ds_{r\gg1}^2 \sim -\left(\frac{r^2}{L^2} + 1\right) dt^2 + \frac{dr^2}{\frac{r^2}{L^2} + 1} + r^2 d\Omega_{S^{d-1}}^2$$
(A.2.2)

$$\xrightarrow{t\mapsto L\sinh\rho} -L^2(\sinh^2\rho+1)d\tau^2 + L^2\frac{\cosh^2\rho}{\sinh^2\rho+1}d\rho^2 + L^2\sinh^2\rho d\Omega_{S^{d-1}}^2$$
(A.2.3)

$$= L^{2} \left( -\cosh^{2} \rho \ \mathrm{d}\tau^{2} + \mathrm{d}\rho^{2} + \sinh^{2} \rho \ \mathrm{d}\Omega_{S^{d-1}}^{2} \right).$$
(A.2.4)

A concrete condition, which we can take as the defining property of AlAdS space-times was also given in [48], and reads

$$R_{\mu\nu\rho\sigma} = \frac{1}{L^2} \left( g_{\mu\rho} g_{\nu\sigma} - g_{\mu\sigma} g_{\nu\rho} \right) + \mathcal{O}(r^{-3}), \tag{A.2.5}$$

in Poincaré coordinates. The class of AlAdS metrics are subject to the Fefferman-Graham theorem [87], and a generic AlAdS metric can be written in Fefferman-Graham coordinates as follows<sup>9</sup> [48, 168],

$$ds^{2} = L^{2} \left( \frac{d\rho^{2}}{4\rho^{2}} + \frac{1}{\rho} g_{\mu\nu}(\rho, x) dx^{\mu} dx^{\nu} \right),$$
(A.2.6)

where  $g_{\mu\nu}(x,\rho)$  admits an expansion which depends on whether *d* is even or odd (that is, whether d + 1 is odd or even); if *d* is odd, the expansion takes the form (where the subscript in parentheses indicates the number of derivatives involved in the term under scrutiny) [169]

$$g_{\mu\nu}(x,\rho) = g_{(0)\mu\nu}(x) + \rho g_{(2)\mu\nu}(x) + \rho^2 g_{(4)\mu\nu}(x) + \dots,$$
(A.2.7)

<sup>9</sup> In the case of pure AdS, where  $g_{\mu\nu} = \eta_{\mu\nu}$ , the Fefferman-Graham metric can be obtained from the *z*-metric (A.1.8) by defining  $\rho = z^2$ .

while for even d, the expansion involves logarithmic terns

$$g_{\mu\nu}(x,\rho) = g_{(0)\mu\nu}(x) + \rho g_{(2)\mu\nu}(x) + \dots \rho^{d/2} g_{(d)\mu\nu}(x) + \rho^{d/2} \log(\rho) h_{(d)\mu\nu}(x) + \mathcal{O}(\rho^{\frac{d}{2}+1}).$$
(A.2.8)

The coefficients  $g_{(0)\mu\nu}, \ldots, g_{(d-2)\mu\nu}$  are determined by solving the Einstein equations order by order in  $\rho$ . In a holographic setting, the term  $h_{(d)\mu\nu}$  turns out to be equal to the metric variation of the holographic conformal anomaly [169], while  $g_{(d)\mu\nu}$  is related to the one-point function of the energymomentum tensor of the dual field theory. Note that the solution obtained in this manner is only valid near the boundary; more powerful techniques are required if one is interested in solutions that extend into the deep interior. An important ingredient of the method of holographic renormalization is a Fefferman-Graham-like expansion for generic bulk fields (see chapter 3 and in particular appendix ??)—this was first realized by Witten in [79]. Conformal invariance (see [42, 43, 170, 171] for more comprehensive reviews) generalizes Poincaré invariance by imposing scale invariance. This circumvents the Coleman-Mandula no-go theorem, since

scale invariance does not allow the existence of a non-trivial S-matrix due to the fact that asymptotic states cannot be defined in the usual manner<sup>1</sup>. Although no formal proof exists in dimensions different from two, evidence suggests that any unitary scale invariant theory is invariant under the full conformal group.

The conformal group consists of diffeomorphisms which preserve the form of the metric up to an arbitrary scale factor<sup>2</sup>,

$$g_{\mu\nu}(x) \to \Omega^2(x)g_{\mu\nu}(x), \tag{B.1.1}$$

which is the smallest group containing the Poincaré group ( $\Omega = 1$ ) as well as inversion symmetry,  $x \rightarrow \infty$  $x^{\mu}/x^{2}$ . In Minkowski space, the conformal group is generated by the usual Poincaré transformations as well as the scale transformation,

$$x^{\mu} \to \lambda x^{\mu}$$
, (B.1.2)

and the special conformal transformations,

$$x^{\mu} = \frac{x^{\mu} + a^{\mu}x^2}{1 + 2x^{\nu}a_{\nu} + a^2x^2}.$$
(B.1.3)

The generators of the corresponding Lie algebra are  $J_{\mu\nu}$  for Lorentz transformations,  $P_{\mu}$  for translations, D for the scaling transformation (dilatation), and  $K_{\mu}$  for the special conformal transformation, and the new—i.e. those different from Poincaré commutation relations (see (4.1.1))—non-vanishing commutation relations of the conformal algebra read

$$[D, P_{\mu}] = -iP_{\mu}, \ [J_{\mu\nu}, K_{\rho}] = -i(\eta_{\mu\rho}K_{\nu} - \eta_{\nu\rho}K_{\mu}), \ [D, K_{\mu}] = iK_{\mu}, \ [P_{\mu}, K_{\nu}] = 2iJ_{\mu\nu} - 2i\eta_{\mu\nu}D.$$
(B.1.4)

The conformal algebra is isomorphic to  $\mathfrak{so}(d,2)$  (with signature  $(-,+,\cdots,+,-)$ ); the isomorphism is constructed by enhancing the *J*'s in the following manner:

$$J_{\mu d} = \frac{K_{\mu} - P_{\mu}}{2}, \quad J_{\mu(d+1)} = \frac{K_{\mu} + P_{\mu}}{2}, \quad J_{(d+1)d} = D.$$
(B.1.5)

The physically interesting representations of the conformal algebra give rise to operators satisfying  $[D, \phi(0)] = -i\Delta\phi(0)$ , which—since  $\phi(x) = e^{ix^{\mu}P_{\mu}}\phi(0)$ —implies that,

$$[D,\phi(x)] = [D,e^{ix^{\mu}P_{\mu}}\phi(0)] = \left([D,e^{ix^{\mu}P_{\mu}}] + e^{ix^{\mu}P_{\mu}}D\right)\phi(0) + e^{ix^{\mu}P_{\mu}}\phi(0)D,$$
(B.1.6)

where

$$[D, e^{ix^{\mu}P_{\mu}}] = \sum_{n=0}^{\infty} \frac{i^{n}}{n!} x^{\mu_{1}} \cdots x^{\mu_{n}} \underbrace{[D, P_{\mu_{1}} \cdots P_{\mu_{n}}]}_{=:[D, P^{n}]}.$$
(B.1.7)

Then, since  $[D, P_{\mu}] = iP_{\mu}$ , we postulate the commutation relation

$$[D, P^n] = inP^n, \tag{B.1.8}$$

<sup>1</sup> In fact, the only allowed asymptotic state is  $|0\rangle$ . One way to see this is to note that since the theory is scale invariant, the interactions are always long-range forces, so asymptotic in and out states cannot be defined in the usual manner. One would think that this voids the entire amplitude program which uses (primarily)  $\mathcal{N} = 4$  SYM in the planar limit, which is, as we have seen, conformal. However, mirror symmetry relates the conformal Coulomb branch of the theory to the Higgs branch, which breaks conformal invariance and thus possesses an S-matrix [172].

<sup>2</sup> We discuss conformal symmetry in dimensions d > 2; in two dimensions, the conformal group is infinite dimensional, and the corresponding algebra is the familiar Witt algebra.

which can easily be proved by induction: assume (B.1.8) to be true and consider the commutator,

$$[D, P^{n+1}] = [D, P^n]P + P^n[D, P] = inP^{n+1} + iP^{n+1} = i(n+1)P^{n+1}.$$
 (B.1.9)

Therefore, we find that

$$[D,\phi(x)] = \sum_{n=0}^{\infty} \frac{i^{n+1}n}{n!} (x^{\mu}P_{\mu})^n \phi(0) + e^{ix^{\mu}P^{\mu}} [D,\phi(0)]$$
(B.1.10)

$$=i^{2}x^{\mu}P_{\mu}\sum_{n=1}^{\infty}\frac{i^{n-1}}{(n-1)!}(x^{\mu}P_{\mu})^{n-1}\phi(0)-i\Delta\phi(x)$$
(B.1.11)

$$= i \left( x^{\mu} \partial_{\mu} - \Delta \right) \phi(x), \tag{B.1.12}$$

where we have used the explicit form of the momentum operator,  $P_{\mu} = -i\partial_{\mu}$ . This relation can be used to fix the two-point function for scalar operators up to a constant; by rotation and translation invariance, we have

$$\langle \phi_1(x)\phi_2(y)\rangle = f(|x-y|),$$
 (B.1.13)

where *f* is an as of yet undetermined function. We now need the Ward identity corresponding to dilatation: this may be found in the "usual" Schwinger-Dyson manner, or by noting that  $D |0\rangle = 0$ , implying that, for two-point functions<sup>3</sup>,

$$0 = \langle 0 | [D, \phi_1(x)\phi_2(y)] | 0 \rangle = \langle 0 | \phi_1(x)[D, \phi_2(y)] + [D, \phi_1(x)]\phi_2(y) | 0 \rangle = (x^{\mu}\partial_{\mu}^{(x)} - \Delta_1 + y^{\mu}\partial_{\mu}^{(y)} - \Delta_2) \langle \phi_1(x)\phi_2(y) \rangle$$
(B.1.15)

where the superscript on the derivative refers to the variable that is differentiated. The differential equation  $0 = (x^{\mu}\partial_{\mu}^{(x)} - \Delta_1 + y^{\mu}\partial_{\mu}^{(y)} - \Delta_2)f(|x - y|)$  is solved by

$$f(|x-y|) = \frac{C}{|x-y|^{\Delta_1 + \Delta_2}},$$
(B.1.16)

where *C* is a constant. Invariance under special conformal transformations further fixes  $\Delta_1 = \Delta_2$ . Conformal symmetry also fixes the tree-point function uniquely (up to a constant); this takes the form

$$\langle \phi_1(x_1)\phi_2(x_2)\phi_3(x_3)\rangle = \frac{C_{123}}{x_{12}^{\Delta_1 + \Delta_2 - \Delta_3}x_{23}^{\Delta_2 + \Delta_3 - \Delta_1}x_{31}^{\Delta_3 + \Delta_1 - \Delta_2}},$$
(B.1.17)

where  $f_{123}$  is a constant and  $x_{ij} = |x_i - x_j|$ . We now derive (B.1.17) from conformal symmetry, using a slightly different approach than the one we used to obtain (B.1.16) (but the method can easily be "degeneralized" to also give the two-point function). A spin-less quasi-primary field  $\phi$  of scaling dimension  $\Delta$  transforms—by definition—under general conformal transformations as

$$\phi(x) \to \phi'(x') = \left| \frac{\partial x'}{\partial x} \right|^{-\Delta/d} \phi(x),$$
(B.1.18)

where  $\left|\frac{\partial x'}{\partial x}\right|$  is the Jacobian of the coordinate transformation and *d* the space-time dimension. For conformal rescalings,  $x \to x' = \lambda x$ , (B.1.18) implies that

$$\phi'(\lambda x) = \lambda^{-\Delta} \phi(x). \tag{B.1.19}$$

Now, translation invariance restricts the form of the three-point function

$$\langle \phi_1(x_1)\phi_2(x_2)\phi_3(x_3)\rangle = f(|x_{12}|, |x_{13}|, |x_{23}|),$$
 (B.1.20)

where  $x_{ij} = x_i - x_j$ . The behaviour of the fields under conformal rescalings (B.1.19) implies that

$$f(x, y, z) = \lambda^{\Delta_1 + \Delta_2 + \Delta_3} f(\lambda x, \lambda y, \lambda z).$$
(B.1.21)

$$0 = \sum_{j=1}^{n} \left( x_j^{\mu} \frac{\partial}{\partial x_j^{\mu}} + \Delta_j \right) \left\langle \phi(x_1) \dots \phi_j(x_j) \dots \phi_n(x_n) \right\rangle.$$
(B.1.14)

<sup>3</sup> The calculation below is easily generalized to an arbitrary number *n* of scalar fields, in which case one obtains the Ward identity

Introducing  $a + b + c = \Delta_1 + \Delta_2 + \Delta_3$ , the above implies that

$$x^{a}y^{b}z^{c}f(x,y,z) = x^{a}y^{b}z^{c}\lambda^{\Delta_{1}+\Delta_{2}+\Delta_{3}}f(\lambda x,\lambda y,\lambda z) = (\lambda x)^{a}(\lambda y)^{b}(\lambda z)^{c}f(\lambda x,\lambda y,\lambda z), \quad (B.1.22)$$

i.e. it transforms as a constant under conformal rescalings, implying that it is independent of x, z and z. Thus, we are led to conclude that

$$\langle \phi_1(x_1)\phi_2(x_2)\phi_3(x_3)\rangle = \frac{C_{123}^{abc}}{|x_{12}|^a |x_{13}|^b |x_{23}|^c},$$
 (B.1.23)

where  $\langle \phi_1(x_1)\phi_2(x_2)\phi_3(x_3) \rangle$  is a (structure) constant. Now, special conformal transformations with parameter *b* can be shown<sup>4</sup> to have Jacobian

$$\left|\frac{\partial x'}{\partial x}\right| = \frac{1}{(1 - 2b \cdot x + b^2 x^2)^d}.$$
(B.1.24)

Defining  $\gamma_i := 1 - 2b \cdot x_i + b^2 x_i^2$ , the distance  $|x_{ij}| = \sqrt{(x_i - x_j) \cdot (x_i - x_j)}$  transforms under special conformal transformations as

$$|x_{ij}| \rightarrow \left| x_{ij}' \right| = \frac{|x_{ij}|}{\sqrt{\gamma_i \gamma_j}}.$$
 (B.1.25)

Transforming both sides of (B.1.23) using (B.1.18) and (B.1.25), we see that

$$\gamma_1^{\Delta_1}\gamma_2^{\Delta_2}\gamma_3^{\Delta_3} \langle \phi_1(x_1)\phi_2(x_2)\phi_3(x_3)\rangle = (\gamma_1\gamma_2)^{a/2}(\gamma_1\gamma_3)^{b/2}(\gamma_2\gamma_3)^{c/2} \frac{C_{123}^{abc}}{|x_{12}|^a |x_{13}|^b |x_{23}|^c}, \quad (B.1.26)$$

which implies that

$$\langle \phi_1(x_1)\phi_2(x_2)\phi_3(x_3)\rangle = \gamma_1^{-\Delta_1 + \frac{a+b}{2}}\gamma_2^{-\Delta_2 + \frac{a+c}{2}}\gamma_3^{-\Delta_3 + \frac{b+c}{2}} \langle \phi_1(x_1)\phi_2(x_2)\phi_3(x_3)\rangle, \qquad (B.1.27)$$

which is true for arbitrary  $\gamma_i$ , which gives us the equations

$$2\Delta_1 = a + b, \quad 2\Delta_2 = a + c, \quad 2\Delta_3 = b + c,$$
 (B.1.28)

the solution to which is

$$a = \Delta_1 + \Delta_2, \quad b = \Delta_1 + \Delta_3 - \Delta_2, \quad c = \Delta_2 + \Delta_3 - \Delta_1,$$
 (B.1.29)

which gives (B.1.17).

Now, an infinitesimal conformal transformation  $x^{\mu} \rightarrow {x'}^{\mu} = x^{\mu} + \epsilon^{\mu}$  satisfies:

$$\frac{\partial x'^{\mu}}{\partial x^{\nu}} = \delta^{\mu}_{\nu} + \partial_{\nu}\epsilon^{\mu} = \left(1 + \frac{1}{d}\partial_{\rho}\epsilon^{\rho}\right) \left(\delta^{\mu}_{\nu} + \frac{1}{2}(\partial_{\nu}\epsilon^{\mu}) - \partial^{\mu}\epsilon_{\nu}\right), \tag{B.1.30}$$

where we have used the conformal Killing equation in the last equality. This we see to be an infinitesimal rescaling times and infinitesimal rotation, which, upon exponentiating, gives us the finite transformation

$$\frac{\partial x'^{\mu}}{\partial x^{\nu}} = \Omega'(x) R^{\mu}_{\ \nu}(x), \ R \in O(d), \tag{B.1.31}$$

where the parameter  $\Omega'$  is equal to the  $\Omega$  of (B.1.1), since

$$\eta_{\rho\sigma}\frac{\partial x^{\prime\rho}}{\partial x^{\mu}}\frac{\partial x^{\prime\sigma}}{\partial x^{\nu}} = {\Omega^{\prime}}^2\eta_{\mu\nu},\tag{B.1.32}$$

where we have used that  $R \in O(d)$ . Next, we need to find out how these guys act on the Hilbert space, and, by extension, on the primary fields  $\phi(0)$ . To this end, we construct unitary operators which are representations of the conformal generators,  $U = e^{iQ_{\epsilon}}$ , where the  $Q_{\epsilon}$  are the charges associated with

<sup>4</sup> One approach to showing this is to use the homomorphism property of determinants and to use that a special conformal transformation is equivalent to an inversion followed by a translation followed by another inversion; see e.g. [43].

infinitesimal conformal transformations [43, 171], which allows us to write the general transformation of a field  $\phi^a$  transforming in the SO(d) representation *D* as,

$$U\phi^{a}(x)U^{-1} = \Omega(x')^{\Delta}D(R(x'))_{b}^{a}\phi^{b}(x'), \qquad (B.1.33)$$

where, as we have seen,  $\frac{\partial x'^{\mu}}{\partial x^{\nu}} = \Omega'(x)R^{\mu}_{\nu}(x)$ . For the scalar representation, D(R) = 1, while for the vector representation, we have  $D(R)_{\mu}{}^{\nu} = R_{\mu}{}^{\nu}$ . For a scalar operator, as claimed, this reduces to the relation

$$\phi(\lambda x) = \lambda^{-\Delta} \phi(x), \tag{B.1.34}$$

where we have renamed  $\Omega \rightarrow \lambda$ .

Now, the commutation relations of (B.1.4) imply that  $K_{\mu}$  is a lowering operator for the conformal weight,

$$[D, K_{\mu}\phi(0)] = [D, K_{\mu}]\phi(0) + K_{\mu}[D, \phi(0)] = -i(\Delta - 1)K_{\mu}\phi(0),$$
(B.1.35)

implying that  $K_{\mu}$  reduces the conformal weight by 1. Note that  $K\phi(0)$  above is shorthand for  $[K, \phi(0)]$ —this notation is valid due to the Jacobi identity and simplifies the notation (see also [171]); we will use it in the rest of this section.

RepresentationNow, s of the four-dimensional conformal group can be labeled by  $(\Delta, j_L, j_R)$ , corresponding to the quantum numbers of the subgroup  $SO(1,1) \times SO(3,1)$ , where the *j*'s are the usual quantum numbers associated with the (algebra) isomorphism  $SO(3,1) = SU_L(2) \times SU(2)_R$ , which, since the quantum theory is required to be unitary, imposes the bounds  $\Delta \ge 1 + j_L$  for  $j_R = 0$  and the same with  $L \leftrightarrow R$ , as well as  $\Delta \ge 2 + j_L + j_R$  for both  $j_L, j_R \ne 0$ . In four dimensions, therefore, scalars<sup>5</sup> ( $j_L = 0 = j_R$ ) satisfy  $\Delta \ge 1$ , while vectors ( $j_L = 1/2 = j_R$ ) satisfy  $\Delta \ge 3$  and so forth. The unitarity bound implies the existence of fields with lowest possible conformal weight, which are consequently annihilated by  $K_\mu$ —these are known as primary fields,

$$K_{\mu}\phi(0) = 0 \stackrel{\text{DEF}}{\iff} \phi$$
 a primary operator. (B.1.36)

where we remind the reader that  $K\phi(0)$  is short-hand for  $[K, \phi(0)]$ . From such a primary operator, we can construct the conformal descendants by acting with the momentum operator, which acts as a raising operator, since

$$[D, P_{\mu}\phi(0)] = -i(\Delta + 1)P_{\mu}\phi(0), \qquad (B.1.37)$$

so the operator  $P_{\mu_1} \cdots P_{\mu_n} \phi(0)$  has conformal weight  $\Delta + n$  (provided that  $\phi(0)$  has weight  $\Delta$ ). It's amusing to note that  $\phi(x) = e^{ix^{\mu}P_{\mu}}\phi(0)$ , which we used previously, is an infinite linear combinations of descendant operators. Finally, the primary operator satisfies the relation  $[J_{\mu\nu}, \phi(0)] = -\mathcal{J}_{\mu\nu}\phi(0)$ , where  $\mathcal{J}_{\mu\nu}$  is a finite-dimensional representation of the Lorentz group, implying that a primary conformal field satisfies the relations:

$$[D,\phi(0)] = -i\Delta\phi(0), \ [J_{\mu\nu},\phi(0)] = \mathcal{J}_{\mu\nu}\phi(0), \ [K_{\mu},\phi(0)] = 0.$$
(B.1.38)

This allows us to construct a representation of the conformal algebra out of the primary  $\phi(0)$  and its descendants<sup>6</sup>.

# B.2 THE SUPERCONFORMAL ALGEBRA

The four-dimensional superconformal group  $PSU(2, 2|\mathcal{N})$ , which we will focus on, is the symmetry group of SYM with  $\mathcal{N}$  supersymmetries. The corresponding algebra,  $\mathfrak{psu}(2, 2|\mathcal{N})$ , consists of the conformal generators,  $J_{\mu\nu}$ ,  $P_{\mu}$ , D,  $K_{\mu}$  as well as the Poincaré supercharges  $Q^a_{\alpha}$ ,  $\bar{Q}^a_{\dot{\alpha}}$ —in addition to these, closure of the superconformal algebra requires further supercharges  $S^a_{\alpha'}, \bar{S}^a_{\dot{\alpha}}$ , which are the fermionic superpartners of  $K_{\mu}$  (the Poincaré supercharges are superpartners of  $P_{\mu}$ ). Furthermore, the internal  $\mathcal{R}$ -symmetry group  $U(\mathcal{N}) = U(1) \times SU(\mathcal{N}) \subset SU(2, 2|\mathcal{N})$  is generated by T (the U(1)) and  $T^i$  (the  $SU(\mathcal{N})$ ) for  $i = 1, \ldots, \mathcal{N}^2 - 1$ .

<sup>5</sup> In *d* dimensions, this generalizes to  $\Delta \geq \frac{d-2}{2}$ .

<sup>&</sup>lt;sup>6</sup> This is accomplished by using the technique of induced representations: we pass from a representation of the subalgebra generated by D,  $J_{\mu\nu}$ , K at x = 0 to a representation of the full conformal algebra by shifting the position using  $P_{\mu}$ . Mathematicians like to call such a construction a parabolic Verma module.

In order to find representations of the superconformal algebra, we do the same as for the conformal algebra above: we identity the superconformal primary operators<sup>7</sup> O, which in addition to the requirement (B.1.36) has to satisfy<sup>8</sup>

$$[S_{a\alpha}, \mathcal{O}] = 0 = [\bar{S}^a_{\dot{\alpha}}, \mathcal{O}], \tag{B.2.1}$$

for all a = 1, ..., N and  $\alpha, \dot{\alpha} = 1, 2$ , and where  $[\cdot, \cdot]$  is a commutator if  $\mathcal{O}$  is bosonic and an anticommutator if  $\mathcal{O}$  is fermionic. This is because the  $S, \overline{S}$ —just like  $K_{\mu}$ —lowers the conformal dimension by one half, since  $[D, S_{a\alpha}] = \frac{i}{2} S_{a\alpha}$ , implying that, if  $\phi$  is a fermionic primary:

$$[D, S_{a\alpha}\phi(0)] = -i(\Delta - 1/2)S_{a\alpha}\phi(0), \tag{B.2.2}$$

and similarly for  $\bar{S}^a_{\dot{\alpha}}$ . In particlar, superconformal primary operators are conformal primaries, and the formalism developed in the previous section carries over; in particular  $P_{\mu}$  acts as a raising operator for the conformal weight.

A new feature, however, is the existence of superdescendants: the commutation relation  $[D, Q^a_{\alpha}] = -\frac{i}{2}Q^a_{\alpha}$  implies that  $Q^a_{\alpha}$  (and its conjugate) raise the conformal dimension by one half, i.e. the superdescendant  $\mathcal{O}' = [\mathcal{Q}, \mathcal{O}]$  has conformal dimension  $\Delta' = \Delta + 1/2$ . Note that superdescandants of superconformal primaries are conformal primary operators, since (we take  $\mathcal{O}$  to be bosonic here since it makes the computation easier)

$$[K_{\mu}, \mathcal{O}'] = [K_{\mu}, [Q, \mathcal{O}]] = [Q, [K_{\mu}, \mathcal{O}]] + [[K_{\mu}, Q], \mathcal{O}] \sim [S, \mathcal{O}] = 0,$$
(B.2.3)

where we have used the Jacobi identity and the condition (B.2.1).

A class of particularly interesting operators are the so-called chiral primary operators that are annihilated by at least one of the  $Q^a_{\alpha'}$ 

$$[\mathcal{Q}^a_{\alpha}, \mathcal{O}] = 0, \tag{B.2.4}$$

which are 1/2 BPS operators, which, as we now show, means that the scaling dimension  $\Delta$  is protected. Let  $\mathcal{O}_{\Delta}$  be a chiral primary operator of conformal dimension  $\Delta$  and spin  $\mathcal{J}_{\mu\nu}$ , i.e.

$$[D, \mathcal{O}_{\delta}(0)] = -i\Delta \mathcal{O}_{\Delta}(0), \quad [J_{\mu\nu}, \mathcal{O}_{\Delta}(0)] = -\mathcal{J}_{\mu\nu}\mathcal{O}_{\Delta}(0). \tag{B.2.5}$$

Superconformal primarity implies that

$$[S^a_{\alpha}, \mathcal{O}_{\Delta}(x)] = 0 = [\bar{S}_{a\dot{\alpha}}, \mathcal{O}_{\Delta}(x)], \qquad (B.2.6)$$

for all  $a \in \{1, ..., N\}$  and  $\alpha, \dot{\alpha} \in \{1, 2\}$ , while the condition that  $\mathcal{O}_{\Delta}$  be chiral primary amounts to the requirement

$$[Q^a_{\alpha}, \mathcal{O}_{\Delta}(x)] = 0, \tag{B.2.7}$$

for at least one of the  $Q^a_{\alpha}$ 's, which we will denote  $Q^{\bar{a}}_{\bar{\alpha}}$ . Now, recall that the anti-commutation relation between the special conformal supercharges and the Poincaré supercharges has the form

$$\left\{Q^{a}_{\alpha},S_{\beta b}\right\} = \epsilon_{\alpha\beta}\left(\delta^{a}_{b}D + R^{a}_{\ b}\right) + \frac{1}{2}\delta^{a}_{b}J_{\mu\nu}(\sigma^{\mu\nu})_{\alpha\beta},\tag{B.2.8}$$

where  $R^a_b \in \mathfrak{u}_R(\mathcal{N})$ , i.e. the automorphism subalgebra (*R*-symmetry) of the full  $\mathfrak{su}(2, 2|\mathcal{N})$ . Using (B.2.6) and (B.2.7), we see that (schematically)

$$0 = [\{S, Q\}, \mathcal{O}_{\Delta}(0)] = [J + D + R, \mathcal{O}_{\Delta}(0)] \sim (\Delta + \mathcal{R} + \mathcal{J})\mathcal{O}_{\Delta}(0).$$
(B.2.9)

So, the dimension is given in terms of *R*-charge and spin, implying that it is indeed protected from quantum corrections.

In four-dimensional  $\mathcal{N} = 4$  SYM, the elementary fields are the fermions,  $\bar{\psi}_{a\dot{\alpha}}, \psi^a_{\alpha}$ , the six scalars,  $\phi^i$  and the gauge field  $A_{\mu}$ , all transforming in the adjoint of the gauge group, SU(N). Representations of the superalgebra  $\mathfrak{psu}(2,2|4)$  are labeled by the quantum numbers of the bosonic subgroup (just as we had for the conformal group above, but with more structure)

$$\underbrace{\mathfrak{so}(3,1)}^{(j_L,j_R)} \times \underbrace{\mathfrak{so}(1,1)}^{\Delta} \times \underbrace{\mathfrak{su}(4)}^{[r_1,r_2,r_3]} \mathfrak{su}(4), \qquad (B.2.10)$$

<sup>7</sup> We consider only local gauge invariant operators constructed from the elementary fields of the CFT.

<sup>8</sup> The latin index structure of  $Q^a_{\alpha}$  and  $S_{a\alpha}$  comes from the fact they transform in conjugated representations of  $U(\mathcal{N})$ .

where  $[r_1, r_2, r_3]$  are the Dynkin labels of the representations of  $\mathfrak{su}(4)$ , labelling a given irreducible representation. The dimension of such a representation is given in terms of the Dynkin labels by [173]

$$\operatorname{Dim}(r_1, r_2, r_3) = (r_1 + 1)(r_2 + 1)(r_3 + 1)\left(1 + \frac{r_1 + r_2}{2}\right)\left(1 + \frac{r_1 + r_2 + r_3}{3}\right), \quad (B.2.11)$$

and gives the degeneracy of the state transforming in the representation under scrutiny. In order to construct gauge-invariant local operators, we must use only gauge-invariant operators such as the fermions, the scalars and the field strength tensor evaluated at the same space-time point x. A local single-trace operator involving scalars, for example, becomes

$$\mathcal{O}(x) = \operatorname{Tr}\left[\phi^{i}\cdots\phi^{j}\right](x), \tag{B.2.12}$$

where the trace is over the adjoint representation of SU(N); i.e. we write  $\phi^i = \phi^i_a T^a$  and take the matrix trace of the generators. In general, such single-trace scalar operators have the form [45],

$$\mathcal{O}(x) = \operatorname{Str}\left[\phi^{\{i_1}\cdots\phi^{i_k\}}\right],\tag{B.2.13}$$

where the  $\{i_1 \dots i_k\}$  stands for the traceless part and the Str stands for symmetrized trace over the gauge algebra, given by

$$\operatorname{Str}\left[T_{a_{1}}\cdots T_{a_{n}}\right] = \sum_{\sigma \in S_{n}} \operatorname{Tr}\left[T_{\sigma(a_{1})}\cdots T_{\sigma_{a_{n}}}\right].$$
(B.2.14)

These operators can be shown to be 1/2 BPS and chiral primary operators with conformal dimension  $\Delta = k$ ; in particular, in the large-*N* limit of AdS/CFT, the single-trace operators are leading, so these are the most important to us. For example, the  $\Delta = 2$  single trace operator is given by Str  $\left[\phi^{\{i}\phi^{j\}}\right] =$ Tr  $\left[\phi^{i}\phi^{j}\right] - \frac{1}{6}\delta^{ij}$ Tr  $\left[\phi^{k}\phi^{k}\right]$ , where we have used that the  $\mathcal{N} = 4$  multiplet contains six scalars as well as the cyclic property of the trace. A more complete treatment [45] reveals that various BPS multiplets and non-BPS multiplets satisfy the properties listed in the table below (the column #*Q* lists the number of Poincaré supercharges left invariant by the primary operator in question):

Operator type	#Q	spin range	su(4) rep.	conformal dim. $\Delta$
identity	16	0	[0,0,0]	0
1/2 BPS	8	2	[0,k,0], k > 2	k
1/4 BPS	4	3	$[\ell,k,\ell], \ \ell \geq 1$	$k+2\ell$
1/8 BPS	2	7/2	$[\ell,k,\ell+2m], m \ge 1$	$k+2\ell+3m$
non-BPS	0	4	any	unprotected

Table B.1: Properties of superconformal BPS multiplets.

The Konishi operator is an example of a non-BPS operator; it is given by [43]

$$K = \operatorname{Tr}\left[\phi^{i}\phi^{i}\right]. \tag{B.2.15}$$

### C.1 SUPERSYMMETRIC GAUGE THEORY

In 1967, Coleman and Mandula proved a no-go theorem [174] for the S-matrix of quantum field theory: given certain reasonable assumptions, the only possible symmetry group is a direct product of the Poincaré group and an internal symmetry group<sup>1</sup>. Another nifty albeit rather long proof of the theorem using Lie algebras directly is presented by Weinberg in [176]. It is, however, possible to circumvent the Coleman-Mandula theorem in two ways: if a theory contains only massless particles, one may extend the Poincaré group to the conformal group; or, one may introduce supersymmetry [177] by enhancing the Poincaré algebra to a superalgebra (graded Lie algebra) via the introduction of a set of spinorial<sup>2</sup> supercharges  $Q^a$ , a = 1, ..., N, where N is the number of independent supersymmetries in the system—see appendix B and also e.g. [170, 178]. A particularly interesting supersymmetric theory is (planar)  $\mathcal{N} = 4$  super Yang-Mills theory (SYM) in four spacetime dimensions. It is the CFT of AdS/CFT, and it has garnered much interest as a more tractable toy model of QCD: understanding the perturbative behavior of  $\mathcal{N} = 4$  has lead to great advances in the field of scattering amplitudes, e.g. it has been realized that the off-shell gauge freedoms in Feynman diagrams in general QFTs are hugely redundant [172], and new recursion relations for tree amplitudes [179]. Great progress at loop level has also been achieved—recently, the planar  $\mathcal{N} = 4$  four-point gluon amplitude has been computed to no less than ten loops [180].



Figure C.1: (a): This particular on-shell diagram corresponds to the four-gluon tree amplitude of  $\mathcal{N} = 4$  SYM. The vertices are equivalent to three-particle on-shell amplitudes which are fixed uniquely by little group scaling—specifically, the filled vertex corresponds to the MHV three-point amplitude, given by the Parke-Taylor expression  $\frac{\langle 23 \rangle^3}{\langle 12 \rangle \langle 31 \rangle} \delta^{2 \times 2} (\lambda \cdot \tilde{\lambda})$ , where  $\delta$ -function imposes momentum conservation expressed in spinor-helicity variables. Similarly, the  $\overline{\text{MHV}}$  vertex (the non-filled), is given by  $\frac{[23]^3}{[12][31]} \delta^{2 \times 2} (\lambda \cdot \tilde{\lambda})$ . It is worth noting that this one on-shell diagram corresponds to the sum of three Feynman diagrams. (b): Illustration of BCFW recursion: an *n*-point tree amplitude  $\mathcal{A}_n(\{h, p\})$  (LHS) is obtained by shifting the momentum of two lines, *i* and *j*, say, by two C-valued vectors  $zr_i^{\mu}, zr_j^{\mu}$ , for a complex parameter *z* in such a way that the shifted lines  $\hat{i}, \hat{j}$  are on-shell. Setting z = 0, the shifted amplitude,  $\hat{\mathcal{A}}_n(z)$  reduces to  $\mathcal{A}_n$ , so considering  $\hat{\mathcal{A}}_n/z$ , we can use Cauchy's theorem (disregarding the possibility of a pole at infinity) to relate  $\mathcal{A}_n = \sum \operatorname{Res}_{z\neq 0} \frac{\mathcal{A}_n(z)}{z}$ . Now, since we are considering tree-level diagrams, the poles occur when propagators between legs *a*, *b*, say, go on-shell, for  $z = z_{ab}$ , which results in a residue of the form  $\hat{\mathcal{A}}_L(z_{ab}) \frac{1}{p_{ab}^2} \hat{\mathcal{A}}_R(z_{ab})$ , where the shifted left/right amplitudes are on-shell evaluated at  $z = z_{ab}$ . Summing over all possible lines that can go on shell and all helicity configurations  $\lambda$  then gives the total sum of residues, which is the sought-after tree amplitude.

<sup>1</sup> In [175], Witten presents a delightful argument as to why the Coleman-Mandula theorem should be true: it turns out that extending the Poincaré symmetry *overconstrains* scattering amplitudes, which can be non-vanishing only for a discrete set of scattering angles.

<sup>2</sup> That is, they transform as spinors under Lorentz transformations and they are an odd element of a  $\mathbb{Z}_2$ -graded symmetry algebra.

Roughly following [181], we now describe  $\mathcal{N} = 4$  in four spacetime dimensions. The spinorial supercharges—in four dimensions—are Weyl spinors, consisting of two complex components,  $Q^a =$ 

 $\begin{pmatrix} Q^a_{\alpha} \\ Q^{a\dot{\alpha}} \end{pmatrix}$ , implying that  $\mathcal{N} = 4$  SYM has 16 real supercharges, which is also the dimension of a Majorana-Weyl spinor in ten dimensions, which, as we shall see, is quite significant.  $\mathcal{N}=4$  is the maximum number of supercharges we can have, since they raise/lower the helicity by  $\Delta \lambda = \frac{1}{2}$  when acting on the fields, so—since we require that no particles have spin greater than one (otherwise we'd have a theory of gravity)—the  $\mathcal{N} = 4$  is the maximal degree of SUSY possible. The supersymmetry algebra has a global U(N) symmetry rotating the supercharges into each other: this is known as  $\mathcal{R}$ -symmetry. The extended supersymmetry algebra read [43],

$$[\mathcal{Q}^{a}_{\alpha}, J^{\mu\nu}] = (\sigma^{\mu\nu})_{\alpha}{}^{\beta}\mathcal{Q}^{a}_{\beta}, \qquad \qquad [\bar{\mathcal{Q}}^{a}_{\dot{\alpha}}, J^{\mu\nu}] = \epsilon_{\dot{\alpha}\dot{\beta}}(\sigma^{\mu\nu})^{\beta}{}_{\dot{\gamma}}\bar{\mathcal{Q}}^{a\dot{\gamma}}, \qquad (C.1.1)$$

$$[\mathcal{Q}^{a}_{\alpha}, P^{\mu}] = 0, \qquad [\bar{\mathcal{Q}}^{a}_{\alpha}, P^{\mu}] = 0, \qquad (C.1.2)$$

$$\{\mathcal{Q}^{a}_{\alpha}, \bar{\mathcal{Q}}_{b\dot{\beta}}\} = 2\sigma^{\mu}_{\alpha\dot{\beta}}P_{\mu}\delta^{a}_{b}$$

$$\{\mathcal{Q}^{a}_{\alpha}, \mathcal{Q}^{b}_{\beta}\} = \epsilon_{\alpha\beta}Z^{ab}$$

$$\{\bar{\mathcal{Q}}^{a}_{\alpha}, \bar{\mathcal{Q}}^{b}_{,\dot{\beta}}\} = \epsilon_{\alpha\dot{\beta}}\bar{Z}_{ab},$$

$$(C.1.4)$$

$$\{\bar{\mathcal{Q}}_{a\dot{\alpha}}, \bar{\mathcal{Q}}_{,b\dot{\alpha}}\} = \epsilon_{\dot{\alpha}\dot{\beta}}\bar{Z}_{ab},$$

$$(C.1.5)$$

$$\{\mathcal{Q}^a_{\alpha}, \mathcal{Q}^b_{\beta}\} = \epsilon_{\alpha\beta} Z^{ab} \tag{C.1.4}$$

$$\{\bar{\mathcal{Q}}_{a\dot{\alpha}}, \bar{\mathcal{Q}}_{b\dot{\beta}}\} = \epsilon_{\dot{\alpha}\dot{\beta}}\bar{Z}_{ab},\tag{C.1.5}$$

where the anti-symmetric  $Z^{ab}$  and its barred cousin are the central charges of the SUSY algebra, which generate its center (i.e., they commute with all other generators of the algebra). They also satisfy  $Z^{ab} = (\bar{Z}^{\dagger})_{ab}$ , since  $\bar{Q}_{a\dot{\alpha}} = (\bar{Q}^a_{\alpha})^*$ . We can now explicitly see the emergence of the  $\mathcal{R}$ -symmetry: under global phase rotations ("reshufflings") of the supercharges,

$$\mathcal{Q}^a_{\alpha} \mapsto (\mathcal{Q}^a)'_{\alpha} = R^a{}_b \mathcal{Q}^b_{\alpha'}, \qquad \bar{\mathcal{Q}}_{a\dot{\alpha}} \mapsto \bar{\mathcal{Q}}'_{a\dot{\alpha}} = \bar{\mathcal{Q}}'_{b\dot{\alpha}} (R^\dagger)^b{}_a.$$
(C.1.6)

Under *R*-symmetry, the first two lines of the SUSY algebra are clearly unchanged, while invariance of (C.1.3) implies unitarity of  $R^a_{\ b}$ —to see this, take the trace of (C.1.3) and  $\mathcal{R}$ -transform the resulting expression,

$$\{\mathcal{Q}^{a}_{\alpha}, \bar{\mathcal{Q}}_{a\dot{\beta}}\} = 2\mathcal{N}\sigma^{\mu}_{\alpha\dot{\beta}}P_{\mu} \mapsto 2\mathcal{N}\sigma^{\mu}_{\alpha\dot{\beta}}P_{\mu} \stackrel{(!)}{=} (R^{\dagger})^{c}{}_{a}R^{a}{}_{b}\{\mathcal{Q}^{b}_{\alpha}, \bar{\mathcal{Q}}_{c\dot{\beta}}\}$$
(C.1.7)

$$\therefore (R^{\dagger})^c_{\ a} R^a_{\ b} = \delta^c_b, \tag{C.1.8}$$

implying unitarity of  $R^a_{\ b}$ . Thus—as long as the central charges vanish—we conclude that the  $\mathcal{R}$ symmetry group is U(N). It turns that when the central charges do not vanish, the  $\mathcal{R}$ -symmetry group (which is of course still required to be unitary by the argument just presented) is given by the compact symplectic unitary group,  $PSU(\mathcal{N})$ . It turns out that for D = 4 the central charges do vanish, and for  $\mathcal{N} = 4$  SYM in four space-time dimensions, the  $\mathcal{R}$ -symmetry group is U(4)—however, for  $\mathcal{N} = 4$  SYM, the superconformal algebra (as we shall see) breaks this to SU(4). We now turn to the classification of massless particle representations of the SUSY algebra in four spacetime dimensions. Since the particles are massless, we choose a Lorentz frame in which the momentum takes the form  $P^{\mu} = (E, 0, 0, E)$  for E > 0, and so the relation (C.1.3) takes the explicit form<sup>3</sup>—where the Q are unitary reps of the SUSY algebra on the relevant Hilbert space  $\mathcal{H}$ :

$$\{\mathcal{Q}^{a}_{\alpha}, \bar{\mathcal{Q}}_{b\dot{\beta}}\} = 2(\sigma^{\mu}P_{\mu})_{\alpha\dot{\beta}}\delta^{a}_{b} = \begin{pmatrix} 4E & 0\\ 0 & 0 \end{pmatrix}_{\alpha\dot{\beta}}\delta^{a}_{b}.$$
(C.1.10)

Setting  $\alpha = \dot{\beta} = 2$  and a = b, we get (no sum on *a*)  $\{\mathcal{Q}_2^a, \bar{\mathcal{Q}}_{a2}\} = \{\mathcal{Q}_2^a, (\mathcal{Q}_2^b)^{\dagger}\} = 0$ , so for any (normalized)  $|\phi\rangle \in \mathcal{H}$ , we have that

$$0 = \langle \phi | \{ \mathcal{Q}_2^a, (\mathcal{Q}_2^a)^{\dagger} \} | \phi \rangle = 2 ||\mathcal{Q}_2^a||^2 \quad (\text{no sum on } a), \tag{C.1.11}$$

implying that  $Q_2^a = 0$ . Furthermore, using the relation  $\{Q_{\alpha}^a, Q_{\beta}^b\} = \epsilon_{\alpha\beta} Z^{ab}$  with  $\alpha = 1$  and  $\beta = 2$  gives us that  $0 = Z^{ab}$  for all a, b since  $\epsilon_{12} = 1$  and  $Q_2^b$  vanishes for all b; that is, central charges are absent

$$v_{\mu}\sigma^{\mu}_{\alpha\dot{\beta}} = \begin{pmatrix} v^{0} + v^{3} & v^{1} - iv^{2} \\ v^{1} + iv^{2} & v^{0} - v^{3} \end{pmatrix}.$$
 (C.1.9)

<sup>3</sup> Note that for any vector  $v^{\mu}$ , we have that

from  $\mathcal{N} = 4$  SYM. The commutation relations also imply that  $\{\mathcal{Q}_1^a, \mathcal{Q}_{b\dot{1}}\} = 4E\delta_b^a$ , which allows us to define creation and annihilation operators realising the fermionic harmonic oscillator algebra:

$$a^{b} = \frac{\mathcal{Q}_{1}^{b}}{2\sqrt{E}}, \quad a_{b}^{\dagger} = \frac{\bar{\mathcal{Q}}_{b1}}{2\sqrt{E}}, \quad (C.1.12)$$

satisfying the anti-commutation relations,

$$\{a^{b}, a^{\dagger}_{c}\} = \delta^{b}_{c}, \quad \{a^{b}, a^{c}\} = 0 = \{a^{\dagger}_{b}, a^{\dagger}_{c}\}.$$
 (C.1.13)

Now, the helicity  $\lambda$  of a given single particle state  $|p^{\mu}, \lambda\rangle \in \mathcal{H}^{\otimes 1}$  is the eigenvalue<sup>4</sup> of  $J_{12}$ , and thus, using (C.1.1), we find that (taking  $\alpha = 1$ )

$$[\mathcal{Q}_1^a, J_{12}] = (\sigma_{12})_1^{\ 1} \mathcal{Q}_1^a = \frac{1}{2} (\sigma_3)_1^{\ 1} \mathcal{Q}_1^a = \frac{1}{2} \mathcal{Q}_1^a, \tag{C.1.16}$$

and thus,

$$J_{12}\mathcal{Q}_1^a | p^\mu, \lambda \rangle = \lambda \mathcal{Q}_1^a | p^\mu, \lambda \rangle - [\mathcal{Q}_1^a, J_{12}] | p^\mu, \lambda \rangle \tag{C.1.17}$$

$$= (\lambda - 1/2)\mathcal{Q}_1^a | p^\mu, \lambda \rangle, \qquad (C.1.18)$$

implying that  $Q_1^a$ —and by extension  $a^b$ —lowers the helicity by 1/2. A completely analogous argument can be invoked to see that  $\bar{Q}_{1b}$  and  $a_b^{\dagger}$  increases the helicity by 1/2. Now, take  $\mathcal{N} = 1$  and note that the spinorial nature of  $\bar{Q}_1$  implies its nilpotency. To form a supermultiplet, we take a vacuum state  $|\Omega\rangle$ —which by definition is annihilated by a, i.e.  $a |\Omega\rangle = 0$ — of lowest helicity,  $\lambda_0$ , and act with  $a^{\dagger}$ , and so, by nilpotency of  $a^{\dagger}$ , the supermultiplet (since we require that no helicity be greater than one) consists only of two states,  $|\Omega\rangle = |p^{\mu}, \lambda_0\rangle$  and  $a^{\dagger} |\Omega\rangle = |p^{\mu}, \lambda_0 + 1/2\rangle$ . However, in order to make the multiplet CPT self-conjugate, we have—in general—to add the states with opposite chirality, implying that the full multiplet contains the states  $|p^{\mu}, \pm \lambda_0\rangle$ ,  $|p^{\mu}, \pm (\lambda_0 + 1/2)\rangle$ . This makes possible only two kinds of multiplets: starting from  $\lambda_0 = 0$ , we get helicities  $\{-1/2, 0, 0, +1/2\}$ —this is the chiral multiplet, consisting of two fermionic degrees of freedom and a complex scalar (two real scalar fields). Taking  $\lambda_0 = 1/2$ , we obtain the vector multiplet with helicities  $\{-1, -1/2, +1/2, +1\}$ , the on-shell field content of which consists of two spinors and one vector.

Now, for  $\mathcal{N} = 4$ , however, starting with the lowest helicity state  $\lambda_0 = -1$ , we get a supermultiplet of  $16 = 2^{\mathcal{N}}$  states which is already CPT self-conjugate<sup>5</sup>. For the excited states (obtained by acting with a combination of the four creation operators on the chosen minimal helicity state) themselves—which translates to the field content of the supermultiplet—this produces a *Pascal's triangle structure*: There are  $\binom{4}{4} = 1$  ways of of selecting four operators out of four,  $\binom{4}{3} = 4$  of selecting three out of four operators etc. All in all, this produces a vector multiplet containing a gauge boson  $A_{\mu}$ , which takes values in  $\mathfrak{su}(N)$ , four left-handed fermions,  $\psi_A$ ,  $A \in \{1, \ldots, 4\}$ , and six real scalars, which we can group together to form three complex scalars,  $\Phi_k$ ,  $k \in \{1, 2, 3\}$ . Further, since  $A_{\mu}$  transforms in the adjoint representation of SU(N), the fact that all the other fields are related to  $A_{\mu}$  by supersymmetry implies that they also transform in the adjoint of SU(N). Under the SU(4)  $\mathcal{R}$ -symmetry, the fields transform in more interesting ways: the gauge field  $A_{\mu}$  is a singlet, 1; scalars are in the **6**, whereas the fermions transform in **4**.

The Lagrangian for  $\mathcal{N} = 4$  can be obtained by dimensional reduction: starting from the unique  $\mathcal{N} = 1$  SYM in ten dimensions,

$$S_{10D}^{\mathcal{N}=1} = \int d^{10}x \, \mathrm{Tr} \left[ -\frac{1}{2} F_{MN} F^{MN} + \frac{i}{2} \bar{\Psi} \Gamma^M D_M \Psi \right], \quad M, N \in \{0, \dots, 9\}$$
(C.1.19)

4 To see this, recall that helicity is defined as the projection of angular momentum along the direction of the momentum,

$$\lambda = \frac{\vec{J} \cdot \vec{P}}{\left| \vec{P} \right|}.$$
(C.1.14)

For our concrete massless momentum, we have  $\left|\vec{P}\right| = \sqrt{2}E$ , and since  $J_i = \frac{1}{2}\epsilon_{ijk}J^{jk}$ , this implies that

$$\lambda = \frac{1}{\sqrt{2}}J^{12} = \frac{1}{\sqrt{2}}J_{12}.$$
(C.1.15)

<sup>5</sup> One can do the same exercise for N = 3, but to ensure CPT self-conjugacy, one has to double the field content which makes it identical to N = 4.

where  $\Gamma^{M}$  are the 32 × 32 (since  $2^{\lfloor 5 \rfloor} = 32$ ) Dirac matrices in ten spacetime dimensions, and the  $\Psi$ 's are Majorana-Weyl spinors. The field strength is given by  $F_{MN} = \partial_M A_N - \partial_N A_M + ig[A_M, A_N]$  whereas the covariant derivative is given by  $D_M \Psi = \partial_M \Psi + ig[A_M, \Psi]$ . The actual procedure, which is done in some detail in [44], involves dimensional reduction on a  $T^6$ . Explicitly, the ten-dimensional gauge field decomposes in the following manner,

$$A_M = (A_\mu(x^\nu), \phi_i(x^\nu)), \quad \mu, \nu \in \{0, \dots, 3\}, \quad i \in \{1, \dots, 6\},$$
(C.1.20)

where—by assumption—nothing depends on the compactified directions. So, for example,  $F_{\mu i} = \partial_{\mu}\phi_i + ig[A_{\mu},\phi_i] = D_{\mu}\phi_i$ , where we have recognized the covariant derivative. Similarly,  $F_{ij} = ig[\phi_i,\phi_j]$ , so for the dimensionally reduced  $F_{MN}F^{MN}$ , we obtain,

$$\operatorname{Tr}\left[F_{MN}F^{MN}\right] = \operatorname{Tr}\left[F_{\mu\nu}F^{\mu\nu} + F_{\mu i}F^{\mu i} + F_{ij}F^{ij}\right] = \operatorname{Tr}\left[F_{\mu\nu}F^{\mu\nu} + D_{\mu}\phi_{i}D^{\mu}\phi_{i} - g^{2}[\phi_{i},\phi_{j}][\phi_{i},\phi_{j}]\right].$$
(C.1.21)

The full Lagrangian reads<sup>6</sup>:

$$\mathcal{L} = \operatorname{Tr}\left[-\frac{1}{2g_{\mathrm{YM}}^2}F_{\mu\nu}F^{\mu\nu} - i\bar{\psi}^a\bar{\sigma}^\mu D_\mu\psi_a - D_\mu\phi_i D^\mu\phi_i + g_{\mathrm{YM}}\left(C^{ab}_{\ i}\psi_a[\phi_i,\psi_b] + \mathrm{h.c.}\right) + \frac{g_{\mathrm{YM}}^2}{2}[\phi_i,\phi_j]^2\right],\tag{C.1.22}$$

where the  $C_i^{ab}$  are the Clebsch-Gordan coefficients coupling two **4** representations of SO(6) to a **6**. Now, we see that the reduction of (C.1.20) breaks the original SO(9,1) Lorentz invariance: it gets broken to  $SO(3,1) \times SO(6)$ , where the six scalars  $\phi_i$  transform as a vector under SO(6), whereas the eight chiral spinors  $\psi_a$  and  $\bar{\psi}_a$  transform in **4** and  $\bar{\mathbf{4}}$ , respectively, of SO(6). Recall that the  $\mathcal{R}$ -symmetry group of  $\mathcal{N} = 4$  is SU(4), which is isomorphic to SO(6)—so this is nothing but the  $\mathcal{R}$ -symmetry group.

Further, one can show that the β-function of  $\mathcal{N} = 4$  vanishes, implying conformal invariance of the theory, which enhances the Lorentz group SO(3, 1) to the conformal group, SO(4, 2). We now see the first glimpse of the AdS/CFT correspondence: as we shall see, the isometry group of AdS<sub>5</sub> ×  $S^5$  is *precisely*  $SO(4, 2) \times SO(6)$ , so the global symmetries on either side of the correspondence match. In fact, the full symmetry group of  $\mathcal{N} = 4$  is obtained by combining the conformal symmetry with supersymmetry: in this manner, one obtains the supergroup PSU(2, 2|4). The corresponding Lie superalgebra  $\mathfrak{psu}(2, 2|4)$  contains, somewhat surprisingly, two extra superconformal spinorial generators,  $S^a_{\alpha}$  and  $\bar{S}^a_{\alpha}$ , which are needed for the algebra to close. The total set of generators of  $\mathfrak{psu}(2, 2|4)$  consists of the subalgebra, where T generates  $\mathfrak{u}(1)$  and the generators T and  $T^j$  from the  $\mathfrak{u}(\mathcal{N}) = \mathfrak{u}(1) \times \mathfrak{su}(\mathcal{N})$  subalgebra, where T generates  $\mathfrak{u}(1)$  and the  $T^j$  generate  $\mathfrak{su}(\mathcal{N})$ . The fermionic generators are  $Q^a_{\alpha}$ ,  $\bar{Q}_{a\dot{\alpha}}$ ,  $\bar{S}^a_{\alpha}$ . The full algebra is somewhat formidable, but it turns out that for  $\mathcal{N} = 4$ , the  $\mathfrak{u}(1)$  generator T commutes with all other generators, and so, by Schur's lemma, is a multiple of the identity. It can be made to vanish [43], and thus, the  $\mathcal{R}$ -symmetry group is not U(4) but rather SU(4).

# C.1.1 The Planar Limit of SU(N) Gauge Theories

In particular, the prototypical gauge theory is SU(N) Yang-Mills theory, which has the Lagrangian,

$$\mathcal{L}_{\rm YM} = -\frac{1}{2g^2} \text{Tr} \left[ F_{\mu\nu} F^{\mu\nu} \right], \qquad (C.1.23)$$

which is identical to the gluonic part of  $\mathcal{N} = 4$  SYM we considered previously. The  $\mathfrak{su}(N)$ -valued gauge field  $A_{\mu} = A_{\mu}^{a}T_{a}$  is written in terms of the generators  $\{T^{a}\}_{a \in \{1,...,N^{2}-1\}}$  of  $\mathfrak{su}(N)$  satisfying  $[T_{a}, T_{b}] = if_{ab}^{\ c}T_{c}$  and  $\operatorname{Tr}[T_{a}T_{b}] = C(\mathbf{R})\delta^{ab}$ , where  $C(\mathbf{R})$  is a number that depends on the representation that turns out to be 1/2. Thus, there are two "perspectives" on  $A_{\mu}$ : it can either be considered a *N*-vector,  $A_{\mu}^{a}$ , or it can be written in terms of the generators  $\{T_{a}\}$ . Taking the gauge group to be U(N) rather than SU(N) (which in the limit of large *N* give the same theory) and thinking of  $A_{\mu}$  as a

<sup>6</sup> We exclude the topological  $\theta$ -term, proportional *in the action* to  $\int F \wedge F$ , which breaks CP invariance.

(C.1.26)

 $N \times N$  matrix  $(A_{\mu})^{i}_{j} = A^{a}_{\mu}(T_{a})^{i}_{j}$ , we can write the (Lorenz gauge) Feynman rules of the theory in the following manner [11],

$$\langle (A_{\mu})^{i}{}_{j}(p)(A_{\nu})^{k}{}_{l}(k) \rangle = g^{2} \delta^{i}_{j} \delta^{k}_{l} \frac{\eta_{\mu\nu}}{p^{2}} (2\pi)^{d} \delta^{d}(p+k),$$

$$\langle (A_{\mu})^{i}{}_{j}(p)(A_{\nu})^{k}{}_{l}(k)(A_{\rho})^{m}{}_{n}(q) \rangle = \frac{1}{g^{2}} \delta^{i}_{n} \delta^{k}_{j} \delta^{m}_{l}(p_{\mu}+k_{\nu}+q_{\rho})(2\pi)^{d} \delta^{d}(p+k+q),$$

$$(C.1.24)$$

$$\langle (A_{\mu})^{i}{}_{j}(p)(A_{\nu})^{k}{}_{l}(k)(A_{\rho})^{m}{}_{n}(q)(A_{\sigma})^{h}{}_{g}(r) \rangle = \frac{1}{g^{2}} \delta^{i}_{g} \delta^{k}_{j} \delta^{m}_{l} \delta^{h}_{n} \eta_{\mu\nu} \eta_{\rho\sigma} (2\pi)^{d} \delta^{d}(p+k+q+r),$$

$$\langle (A_{\mu})^{i}{}_{j}(p)(A_{\nu})^{k}{}_{l}(k)(A_{\rho})^{m}{}_{n}(q)(A_{\sigma})^{h}{}_{g}(r) \rangle = \frac{1}{g^{2}} \delta^{i}_{g} \delta^{k}_{j} \delta^{m}_{l} \delta^{h}_{n} \eta_{\mu\nu} \eta_{\rho\sigma} (2\pi)^{d} \delta^{d}(p+k+q+r),$$

which can represented by the 't Hooft the double line notation diagrams [182] of figure C.2.



Figure C.2: The propagator and the vertices for U(N) Yang-Mills theory. Their values are given in eqs. (C.1.24) to (C.1.26).

Clearly, each closed loop contributes a factor N, since  $\delta^{ij}\delta^{ji} = \text{Tr}[\mathbb{1}_{N\times N}] = N$ ; thus, a Feynman diagram with F loops, or *faces*, will carry a factor of  $N^F$ , whereas for each propagator (edge) (P), there will be a factor  $g^2$ , while each vertex (whether it be cubic or quartic) contributes a factor  $g^{-2}$ , implying that a generic Feynman diagram will contribute a factor,

diagram 
$$\sim g^{2P-2V} N^F = (g^2 N)^{P-V} N^{F-P+V} = \lambda^{P-V} N^{\chi}$$
, (C.1.27)

where we have defined the 't Hooft coupling  $\lambda = g^2 N$ —which becomes the effective coupling of the theory at large *N*—and identified the Euler characteristic  $\chi$ . which is also related to the genus g of the surface—in the absence of boundaries—via the relation  $\chi = 2 - 2g$ . Thus, in the 't Hooft (or *planar*) limit, where  $N \rightarrow \infty$  with  $\lambda$  held fixed, only planar diagrams, that is, diagrams that can be drawn on  $S^2$  contribute. It is also worth noting that taking the limit  $N \rightarrow \infty$  naïvely leads to a divergent  $\beta$ -function, while taking the limit with fixed 't Hooft coupling leads to a renormalization equation with finite coefficients [43].



Figure C.3: Two vacuum diagrams: the left is planar; it can be *folded* over the sphere  $S^2$ , whereas the other cannot: it has genus one, and thus maps to the 2-torus,  $T^2$ . Figure inspired by [183].

Crucially, the large *N* expansion of our field theory is formally the same as the topological expansion of strings with string coupling 1/N, which is also a sum over genera. For  $\mathcal{N} = 4$ , the string theory that leads to the correct expansion is, as we explore in appendix D, ten-dimensional type IIB string theory on  $AdS_5 \times S^5$ , which we now briefly describe.

### C.2 TYPE IIB SUPERGRAVITY

Following [184] (see also [45]), we take the limit  $\alpha' \rightarrow 0$ , thereby retaining only the massless modes of the type IIB superstring—this is precisely type IIB supergravity. The bosonic spectrum from the NS-NS sector thus consists of the dilaton,  $\Phi$ , the graviton  $G_{MN}$  and the Kalb-Ramond two-form,  $B_2$ , and its associated field strength  $H_3 = dB_2$ . The R-R sector contributes form fields,  $C_0$ ,  $C_2$  and  $C_5$ , with their associated odd-form field strength, of which  $\tilde{F}_5$  is particularly interesting since it is self-dual. Finally, the fermionic part of the spectrum is provided by the NS-R (and its cousin) sector, which includes two left-handed Majorana-Weyl gravitinos and two right-handed Majorana-Weyl gravitinos.

However, it turns out that the self-duality of  $\tilde{F}_5$  makes writing down a classical action for type IIB SUGRA quite complicated—the issue is that the 'usual'  $F^2$  action, i.e. something like  $\int F_5 \wedge *F_5$ , does not incorporate the self-duality constraint, and thus describes twice the number of degrees of freedom. One solution to this problem is the PST approach [185] makes use of an auxiliary scalar field and an extra gauge symmetry, which allows the scalar field to be set equal to one of the space-time coordinates (this results in a loss of general covariance in the chosen direction).

Another way to deal with this is to produce an action which gives the correct equations of motion when one imposes the self-duality constrain. The action—which is not supersymmetric due to the overcounting that has to be fixed by the self-duality constraint—turns out to be,

$$S_{\text{IIB SUGRA}} = \frac{1}{2\kappa^2} \left( \int d^{10}x \,\sqrt{-G} \left\{ e^{-2\Phi} \left[ R + 4\partial_M \Phi \partial^M \Phi - \frac{1}{2} \,|H_3|^2 \right] - \frac{1}{2} \,|F_1|^2 - \frac{1}{2} \,|\tilde{F}_3|^2 + \frac{1}{2} \,|\tilde{F}_5|^2 \right\} \right) \tag{C.2.1}$$

$$-\frac{1}{\kappa^2}\int C_4\wedge H_3\wedge F_3,\tag{C.2.2}$$

where  $F_{n+1} = dC_n$ ,  $H_3 = dB_2$ ,  $\tilde{F}_3 = F_3 - C_0H_3$  and  $\tilde{F}_5 = F_5 - \frac{1}{2}C_2 \wedge H_3 + \frac{1}{2}B_2 \wedge F_3$ . Here, the R-R fields differ by field redefinitions from the ones that couple simply to the the D-brane world volumes. The self-duality constraint that makes the action above "correct" reads

$$\tilde{F}_5 = \star \tilde{F}_5. \tag{C.2.3}$$

Furthermore, the ten-dimensional gravitational constant is given by.

$$2\kappa = (2\pi)^7 {\alpha'}^4. \tag{C.2.4}$$

Interestingly, type IIB SUGRA has a global  $SL(2, \mathbb{R})$  (Möbius) symmetry which is not manifest in the action (C.2.2); in order to identify it, one has to rewrite to Einstein frame and perform a couple of field redefinitions (see [45, 184]). This  $SL(2, \mathbb{R})$  invariance is holographically related to the Montonen-Olive duality of  $\mathcal{N} = 4$  SYM [186].

In this appendix, which serves as supplementary material to chapter 2, we provide a "derivation" of the AdS/CFT correspondence from string theory by considering a stack of N D<sub>3</sub> branes from the perspectives of open and closed strings, following [42–44].

# D.1 D3 DUALITY AND ADS5/CFT4

Consider a stack of *N* D<sub>3</sub> branes in  $\mathbb{R}^{9,1}$ , which extend along a (3 + 1)-dimensional hyperplane. Perturbative string theory on this background contains both open and closed strings: the closed strings enter as excitations of the vacuum, while the open strings end on the D<sub>3</sub> branes and, as such, act as excitations of the branes. At low energies,  $E \ll l_s^{-1}$ , only massless string modes are excited. When the string coupling is small,  $g_s \ll 1$ , the Dirac-Born-Infeld (DBI) action describes the string dynamics accurately: the dynamics of open strings are then described by a supersymmetric gauge theory living on the worldvolume of the D branes, where the gauge field  $A_{\mu}$  corresponds to open string excitations parallel to the D brane, while excitations transverse to the D brane—which are interpreted as fluctuations in the position of the D-brane—behave as scalar fields. If we consider only a single D brane, we get a U(1) gauge theory (see e.g. [187]), but if we consider N coincident branes, we have to introduce Chan-Paton factors, which are non-dynamical degrees of freedom assigned to the endpoints of the string: the Chan-Paton factor  $\lambda_{ij}$  labels strings that go between the *i*'th and the *j*'th brane. One can show that the  $\lambda_{ii}$  form the Lie algebra U(N), resulting in U(N) gauge theory with effective coupling  $g_s N$ . On the other hand, D-branes are also solitonic solutions to supergravity, where they act as sources of the gravitational field. In order for supergravity to be a good description, we require that the characteristic scale L of the spacetime under scrutiny to be large implying weak curvature and, by extension, low energy, so that supergravity is a good description of the dynamics. For *N* coincident D-branes, we have  $\frac{L^4}{{\alpha'}^2} \propto g_s N$ , leading us to conclude that the closed string perspective is valid for  $g_s N \gg 1$ . In summary, the (schematic) action describing the configuration of *N* D<sub>3</sub> branes is given by

$$S = S_{\text{closed}} + S_{\text{open}} + S_{\text{int}}, \tag{D.1.1}$$

where  $S_{\text{closed}}$  is the the action of type IIB supergravity (SUGRA) with some additional corrections since we're in the low-energy limit—and  $S_{\text{open}}$  describes the dynamics on the (3+1)-dimensional worldvolume constituted by the brane-stack, and  $S_{\text{int}}$  describes the interactions (which we will largely ignore; see [42] for details). The (effective) action above involves only the massless modes, but in general takes into account the effects of integrating out the massive modes. The action is not renormalizable.

We will consider the action above *from two different points of view*, and—from the physical equivalence of these perspectives—arrive at the AdS/CFT correspondence. Note that in both perspectives, we will be in the low energy limit,  $E \ll \frac{1}{l_s}$ , while the two perspectives will be valid for opposite extremes of the string coupling.

#### D.1.1 Open Strings

We start by taking the point of view of the open strings; valid in the regime  $g_s N \ll 1$ . While we should really consider a stack of N coincident D<sub>3</sub>-branes in  $\mathbb{R}^{1,9}$  extending into the directions  $X^{\mu}$  for  $\mu \in \{0, 1, 2, 3\}$  with  $X^I = 0$  for  $I \in \{4, \ldots, 9\}$ , we will, as is standard, focus on the case N = 1 and then hand-wavily generalize our results. The dynamics on the D<sub>3</sub> brane is described by the DBI action,

$$S_{\text{DBI}} = -\frac{1}{(2\pi)^3 l_s g_s} \int d^4 \sigma \, \sqrt{-\det\left[\gamma_{ab} + 2\pi \alpha' F_{ab}\right]} + \text{fermions},\tag{D.1.2}$$

where  $\gamma_{ab} = \eta_{MN} \partial_a X^M \partial_b X^N$  is the induced metric, and F = dA is the field strength two-form of the U(1) gauge field. The  $\sigma^a$  with  $a \in \{0, ..., 3\}$  are the coordinates of the wold-volume, which we can choose by specifying our embedding such that  $X^a(\sigma) = \sigma^a$ , whereas the directions transverse to the

brane—which we center at the origin—are described in terms of six scalar fields, which are essentially fluctuations in brane position,

$$X^{i+3}(\sigma) = 2\pi \alpha' \phi_i(\sigma), \ i \in \{1, \dots, 6\}.$$
 (D.1.3)

Hence, the induced metric takes the form,

$$\gamma_{ab} = \eta_{ab} + (2\pi\alpha')^2 \partial_a \phi^i \partial_b \phi_i, \tag{D.1.4}$$

which means that the determinant featuring in the DBI action (D.1.2) takes the form:

$$\det := \det \left[ \eta_{ab} + 2\pi \alpha' F_{ab} + (2\pi \alpha')^2 \partial_a \phi^i \partial_b \phi_i \right].$$
(D.1.5)

The low-energy limit  $l_s E \to 0$  implies  $\alpha' \to 0$  and allows us to expand the determinant in the by-now small parameter  $\alpha'$ , which we will do shortly. Consider first the general expression  $\eta_{ab} + \epsilon B_{ab}$  where  $\epsilon$  is some small parameter and perform the rewriting

$$\eta_{ab} + \epsilon B_{ab} = \eta_{ac} \left( \delta_b^c + \epsilon B^c_{\ b} \right). \tag{D.1.6}$$

(D.1.9)

The determinant (D.1.6) can now be computed by using the homomorphism property of the determinant,

$$\det\left(\eta_{ac}\left(\delta_{b}^{c}+\epsilon B^{c}_{b}\right)\right) = \det\left(\eta_{ac}\right)\det\left(\delta_{b}^{c}+\epsilon B^{c}_{b}\right) = -\det\left(\delta_{b}^{c}+\epsilon B^{c}_{b}\right). \tag{D.1.7}$$

Now, since det  $\mathbf{A} = \exp(\operatorname{Tr}[\log \mathbf{A}])$  for  $\mathbf{A}$  an  $n \times n$  matrix, setting  $\mathbf{A} = 1 + \epsilon \mathbf{B}$ , we get

$$\det(\mathbb{1} + \epsilon \mathbf{B}) = \exp\left(\operatorname{Tr}\left[\log(\mathbb{1} + \epsilon \mathbf{B})\right]\right) = \exp\left(\sum_{n=1}^{\infty} \frac{(-1)^{n+1}}{n} \epsilon^n \operatorname{Tr}\left[\mathbf{B}^n\right]\right)$$
(D.1.8)  
$$= 1 + \sum_{n=1}^{\infty} \frac{(-1)^{n+1}}{n} \epsilon^n \operatorname{Tr}\left[\mathbf{B}^n\right] + \frac{1}{2} \left(\sum_{n=1}^{\infty} \frac{(-1)^{n+1}}{n} \epsilon^n \operatorname{Tr}\left[\mathbf{B}^n\right]\right) \left(\sum_{m=1}^{\infty} \frac{(-1)^{m+1}}{m} \epsilon^m \operatorname{Tr}\left[\mathbf{B}^m\right]\right) + \dots$$

$$= 1 + \epsilon \operatorname{Tr} \left[\mathbf{B}\right] - \frac{1}{2} \epsilon^{2} \operatorname{Tr} \left[\mathbf{B}^{2}\right] + \frac{1}{2} \epsilon^{2} \operatorname{Tr} \left[\mathbf{B}\right]^{2}.$$
(D.1.10)

Combining this with the expansion  $\sqrt{1 + x} = 1 + x/2 + O(x^2)$ , we find that the expansion of the DBI determinant (D.1.5) becomes

$$\sqrt{-\det} = 1 + (2\pi\alpha')^2 \left( -\frac{1}{2} F^a{}_b F^b{}_a + \frac{1}{2} \partial_a \phi^i \partial^a \phi_i \right) + \mathcal{O}(\alpha'^3) \tag{D.1.11}$$

$$= 1 + (2\pi\alpha')^2 \left(\frac{1}{2}F_{ab}F^{ab} + \frac{1}{2}\partial_a\phi^i\partial^a\phi_i\right) + \mathcal{O}(\alpha'^3), \tag{D.1.12}$$

where the "matrix index structure" (i.e. one raised and one lowered index) we impose in determinant relation (D.1.7) incurs an additional minus due to antisymmetry of the field strength. This means that the DBI action assumes the form<sup>1</sup>

$$S_{\text{DBI}} = -\frac{1}{(4\pi)g_s} \int d^4\sigma \,\left(\frac{1}{4}F_{ab}F^{ab} + \partial_a\phi^i\partial^a\phi_i\right) + \text{fermions,} \tag{D.1.13}$$

which is the action for four-dimensional  $\mathcal{N} = 4$  SYM with gauge group U(1) provided that we identify (compare (C.1.23))

$$g_{\rm YM}^2 = 4\pi g_{\rm s}.$$
 (D.1.14)

Generalizing the above to a stack of *N* coincident D<sub>3</sub>-branes, the gauge fields and scalars are valued in  $\mathfrak{u}(N)$ ,  $\phi^i = \phi^{ia}T_a$ ,  $A_\mu = A^a_\mu T_a$ , where the  $T_a$  are the  $N^2 - 1$  generators of the Lie algebra satisfying  $[T_a, T_b] = f_{ab}{}^c T_c$ . This complicates matters somewhat: the partial derivatives turn into covariant derivatives, but the end result is the same: the identification of (D.1.14) makes equivalent the DBI action and that of four-dimensional U(N) SYM theory [43]. Furthermore, the limit  $\alpha' \to 0$  reduces the type IIB action—i.e.  $S_{closed}$  of the full action (D.1.1)—to free supergravity in the  $\mathbb{R}^{1,9}$  bulk. Similarly,

<sup>1</sup> We ignore the constant contribution from the 1, which will just integrate to the world-volume.

we see that  $S_{int} \xrightarrow{\alpha' \to 0} 0$ , so the  $\mathcal{N} = 4 U(N)$  gauge theory living on the brane world-volume decouples (hence the  $\alpha' \to 0$  is sometimes called the *decoupling limit*) from the free SUGRA theory in the flat bulk. Now, let's examine the limit that we're taking in some detail: starting with N + 1 D3-branes in  $\mathbb{R}^{1,10}$ , we take N of them to be coincident with  $X^{i+3} = 0$  for  $i \in \{1, \dots, 6\}$ , while we take the last brane to be separated from the stack in the  $X^9$ -direction by a distance r; i.e. the last brane has  $X^9 = r$ . Considering only the massless modes, the fields living on the brane are now described by a  $U(N) \times U(1)$  gauge theory. This separation is characterized by a Higgs expectation value  $\langle X^9 \rangle = \frac{r}{2\pi \alpha'}$ , which has to be kept fixed when bringing together the stack and the rogue brane. This results in the *Maldacena limit* [1],

$$\alpha' \to 0, \quad U := \frac{r}{\alpha'} = \text{fixed.}$$
 (D.1.15)

In particular, keeping U fixed implies that the mass of the stretched strings remains fixed. Seemingly, then, we get U(N) four-dimensional  $\mathcal{N} = 4$  SYM living in the branes; it turns out, however, that the  $U(1) \subset U(N)$  corresponds to singleton fields living on the boundary in the gravity theory that cannot propagate into the bulk and thus decouple, leaving us with SU(N) four-dimensional  $\mathcal{N} = 4$  SYM. Note also that what we have done so far is valid for any N!

# D.1.2 Closed Strings

From the point of view of closed strings, a stack of *N* D<sub>3</sub>-branes in the strongly coupled limit<sup>2</sup>,  $g_s N \gg 1$ , arise as a BPS solution to type IIB supergravity preserving  $SO(3,1) \times SO(6)$  isometries of  $\mathbb{R}^{9,1}$ . Explicitly, this solution is given by [43]

$$ds^{2} = H^{-1/2} \eta_{\mu\nu} dX^{\mu} dX^{\nu} + H^{1/2} (dr^{2} + r^{2} d\Omega_{S^{5}}^{2}), \qquad (D.1.16)$$

$$H = 1 + \frac{L^4}{r^4}, \quad L^4 = 4\pi g_s N {\alpha'}^2, \tag{D.1.17}$$

$$C_4 = (1/H - 1) dX^0 \wedge dX^1 \wedge dX^2 \wedge dX^3 + \text{ terms that ensure self-duality of } F_{(5)}, \quad (D.1.18)$$

$$e^{\varphi} = g_s, \quad B_{MN} = 0,$$
 (D.1.19)

where  $r^2 = X_i X^i$  and  $C_4$  is the four-form gauge field, which is related to the self-dual five-form R-R field via  $F_{(5)} = dC_{(4)}$ . Further, since  $g_{00}$  is non-constant, the energy  $E_r$  of a string excitation as measured by an observer at a constant position r and the energy  $E_{\infty}$  measured by an observer at infinity are related by [42]

$$E_{\infty} = H^{-1/4} E_r. \tag{D.1.20}$$

The D<sub>3</sub>-branes act as sources for the self-dual five-form  $F_{(5)}$ , which has a flux on the five-sphere. The background of the D<sub>3</sub> SUGRA solution has two distinct regions: for large  $r \gg L$ , we have  $H \sim 1$ , and we're left with ten-dimensional Minkowski space, while  $r \ll L$  corresponds to the near-horizon or throat region, where the 1 in the harmonic function H can be neglected, leaving us with

$$ds_{\text{throat}}^2 = \frac{r^2}{L^2} \eta_{\mu\nu} dX^{\mu} dX^{\nu} + \frac{L^2}{r^2} (dr^2 + r^2 d\Omega_{S^5}^2)$$
(D.1.21)

$$= \underbrace{\left(\frac{r^2}{L^2}\eta_{\mu\nu}dX^{\mu}dX^{\nu} + \frac{L^2}{r^2}dr^2\right)}_{\text{AdS}_5 \text{ with radius }L} + \underbrace{L^2d\Omega_{S^5}^2}_{S^5 \text{ with radius }L} \tag{D.1.22}$$

which we recognize as the metric of  $AdS_5 \times S^5$  in Poincaré patch coordinates (cf. eq. (A.1.7)). This leads us to conclude that we have two different types of closed strings; closed strings propagating in the ten-dimensional Minkowski space (the asymptotically flat region) far away from the horizon and closed strings propagating in the near-horizon geometry  $AdS_5 \times S^5$ . In the Maldacena limit, however, these regions decouple, as we shall now argue, following [42]: consider a string state near the throat

<sup>2</sup> Note that  $g_s N$ , while large, is kept *fixed*, which in particular implies that the 't Hooft coupling  $\lambda = g_{YM}^2 N \sim g_s N$  is kept fixed from the open string perspective. Note further that this assumes that  $g_s < 1$ ; if  $g_s > 1$ , we'd have instead  $N/g_s \gg 1$ —that is, we need large N, *not* large g. This requirement arises because SUGRA is only a good description when the background curvature is much larger than the string length,  $L \gg l_s$ , because we need to avoid string worldsheet quantum corrections. Further, quantum string corrections are avoided when  $g_s \rightarrow 0$ .

with (fixed) energy  $E_r \gg 1$  (at fixed radial position *r*). The energy  $E_{\infty}$  of this string state as measured by an observer at infinity is then given by (D.1.20)

$$E_{\infty} \sim \frac{r}{L} E_r, \tag{D.1.23}$$

where we have used that *r* is in the throat region, i.e.  $r \sim 0$ , implying that  $E_{\infty}$  is negligible since  $\frac{r}{L}E_r \rightarrow 0$  for  $E_r$  fixed and  $r \ll L$ . Thus, the observer at infinity sees two different low-energy contributions: closed strings in the asymptotically flat region described by type IIB SUGRA in  $\mathbb{R}^{1,9}$ , while the closed strings in the throat region are described by fluctuations around the  $AdS_5 \times S^5$  solution of IIB SUGRA, and since these fluctuations can be arbitrarily large even in the low energy limit, we are dealing with *the full quantum string theory* on  $AdS_5 \times S^5$ . When taking the low energy limit, the two regions decouple from each other<sup>3</sup>. To summarize, we have found that in the low-energy large-*N* limit, the open and closed string perspectives, which are equivalent, give rise to two different decoupled effective theories:

- Open string perspective: four-dimensional SU(N)  $\mathcal{N} = 4$  SYM on  $\mathbb{R}^{1,3}$  and type IIB SUGRA on  $\mathbb{R}^{1,9}$ .
- Closed string perspective: Type IIB string theory on  $AdS_5 \times S^5$  and type IIB SUGRA  $\mathbb{R}^{1,9}$ .

This leads us to conclude that four-dimensional SU(N)  $\mathcal{N} = 4$  SYM on  $\mathbb{R}^{1,3}$  is dual to type IIB string theory on AdS<sub>5</sub> × S<sup>5</sup> with the identification:

$$g_{\rm YM}^2 = 4\pi g_s, \quad L^4 = 4\pi g_s N(\alpha')^2, \quad \lambda = g_{\rm YM}^2 N = 2\pi g_s N = \frac{L^4}{2(\alpha')^2},$$
 (D.1.26)

where  $\lambda$  is the 't Hooft coupling that we introduced previously. Of course, our derivation is valid only for large *N* and fixed  $\lambda$ , but it is natural to conjecture that the duality holds for any values of  $\lambda$  and *N*.

$$ds_{\text{throat}}^2 = {\alpha'}^2 \frac{U^2}{L^2} \eta_{\mu\nu} dX^{\mu} dX^{\nu} + \frac{L^2}{U^2} (dU^2 + U^2 d\Omega_{S^5}^2)$$
(D.1.24)

$$= \alpha' \left[ \frac{U^2}{\sqrt{4\pi g_s N}} \eta_{\mu\nu} dX^{\mu} dX^{\nu} + \frac{\sqrt{4\pi g_s N}}{U^2} dU^2 + \sqrt{4\pi g_s N} d\Omega_{55}^2 \right],$$
(D.1.25)

which—up to an overall factor of  $\alpha'$ —again produces the metric of  $AdS_5 \times S^5$  in Poincaré patch coordinates. This is the procedure of the original paper [1].

<sup>3</sup> When writing the throat metric (D.1.22), we have used the coordinate r, which is *not* fixed in the Maldacena limit (D.1.15). Instead, we should replace our radial coordinate with U, which *is* kept fixed in the Maldacena limit,  $r = U\alpha'$ , yielding the near-horizon metric:

# e.1 holography at finite temperature, density & chemical potential

# E.1.1 Gravity Dual of Finite Temperature Field Theory I: Black D3-Branes

The gravity dual of a finite temperature field theory is obtained by considering a black brane<sup>1</sup> in AdS space (cf. the dictionary in 2.3)—thus, in this case, making the spacetime AAdS; see the metric in (E.1.1). These black branes radiate and have a temperature,  $T_H$ —the Hawking temperature—associated to them. This identification was first realized by Witten in [188], where he considers the AdS Schwarzschild black hole; we shall take this approach shortly. Our starting point for now—to make contact with our heuristic derivation of the AdS/CFT correspondence—will be a black D3-brane, which is a non-extremal D3-brane (see [189, 190]) described by the metric (compare (D.1.16))

$$ds^{2} = H(r)^{-1/2} \left( -f(r)dt^{2} + d\vec{x}^{2} \right) + H(r)^{1/2} \left( \frac{dr^{2}}{f(r)} + r^{2}d\Omega_{S^{5}}^{2} \right),$$
(E.1.1)

with blackening factor  $f(r) = 1 - \left(\frac{r_h}{r}\right)^4$  and harmonic function  $H(r) = 1 + \left(\frac{L}{r}\right)^4$ . The black brane horizon  $r_h$  is related to the Hawking temperature of the black brane—we will have more to say about this connection later; in particular,  $r_h = 0$  reduces the black D3-brane to an ordinary extremal D3-brane with metric (D.1.16). Proceeding in the same manner as we did when motivating the AdS/CFT correspondence in appendix D, we—following [43]—focus on the near-horizon (throat) region of the black D3-brane, where  $r/L \ll 1$ . Upon introducing the coordinate  $z = L^2/r$ , the black D3-brane in the throat limit takes the form

$$ds_{\text{throat}}^{2} = \frac{r^{2}}{L^{2}} \left( -\left[1 - \left(\frac{r_{h}}{r}\right)^{4}\right] dt^{2} + d\vec{x}^{2} \right) + \frac{L^{2}}{r^{2} \left(1 - \left(\frac{r_{h}}{r}\right)^{4}\right)} dr^{2} + L^{2} d\Omega_{S^{5}}^{2}$$
(E.1.2)

$$= \frac{L^2}{z^2} \left[ -\left(1 - \left(\frac{z}{z_h}\right)^4\right) dt^2 + d\vec{x}^2 + \frac{1}{1 - \left(\frac{z}{z_h}\right)^4} dz^2 \right] + L^2 d\Omega_{S^5}^2$$
(E.1.3)

$$= \frac{L^2}{z^2} \left[ \left( 1 - \left(\frac{z}{z_h}\right)^4 \right) d\tau^2 + d\vec{x}^2 + \frac{1}{1 - \left(\frac{z}{z_h}\right)^4} dz^2 \right] + L^2 d\Omega_{S^5}^2$$
(E.1.4)

where  $z_h = L^2/r_h$ , and where, in the last equality, we have Wick rotated to Euclidean time,  $\tau = it$ . Further, we define a radial variable  $\rho = L\sqrt{1-\frac{z}{z_h}}$ , which measures the distance from the black brane horizon at  $z_h$ , in terms of which the Euclidean metric close to the black brane horizon (that is, to lowest order in  $\rho$ ) takes the form

$$ds_{\text{throat-NBBH}}^2 = \frac{4\rho^2}{z_h^2} d\tau^2 + \frac{L^2}{z_h^2} d\vec{x}^2 + d\rho^2 + L^2 d\Omega_{S^5}^2, \qquad (E.1.5)$$

where NBBH stands for *near black brane horizon*. Next, we focus on the  $(\tau, \rho)$  plane and rescale the Euclidean time coordinate,  $\phi = 2\tau/z_h$ , which yields the  $(\phi, \rho)$  plane metric

$$ds^2 = d\rho^2 + \rho^2 d\phi^2,$$
(E.1.6)

and thus, in order to avoid a conical singularity, we need periodicity; that is, we must identify  $\phi \sim \phi + 2\pi$  (the  $\sim$  is to be read as *identified with*), implying that  $\tau \sim \tau + z_h \pi$ . The periodicity of the Euclidean time is precisely the inverse temperature (see table 2.3), leading us to conclude that

$$T = \frac{1}{z_h \pi}.\tag{E.1.7}$$

<sup>1</sup> In contradistinction to black holes, the spatial topology of a black brane is not compact: this is required in order for the spatial geometry of the dual field theory to be  $\mathbb{R}^3$ . In particular, the black D<sub>3</sub>-brane has horizon topology  $\mathbb{R}^3 \times S^5$ , i.e. planar with five-spheres at every point.

Note that we have to use the periodicity of the time coordinate  $\tau$  and *not* its diffeomorphic cousin  $\phi$ , since we identify  $\tau$  as the (Euclidean) time coordinate in the dual  $\mathcal{N} = 4$  theory. We now want to apply the Bekenstein-Hawking formula for the black brane entropy,  $S = \frac{A}{4G}$ . First, we determine the horizon area A of the black D3-brane, for which we introduce the metric  $\tilde{g}_{ij}$ — derived from (E.1.5)—describing a hyperplane with  $\rho = 0$  (equivalently,  $z = z_h$ ) and  $\tau$  fixed, but arbitrary. The determinant, owing to the diagonality of the metric (E.1.5), is consequently given by  $\tilde{g} = \frac{L^6}{z_h^6} \times L^{10}g_{S^5}$ , with  $g_{S^5}$  the determinant of the metric of  $S^5$ , implying that the area becomes<sup>2</sup>,

$$A = \int_{\mathbb{R}^3 \times S^5} d^8 x \ \sqrt{\tilde{g}} = \frac{L^8}{z_h^3} \int_{\mathbb{R}^3} d^3 x \times \int_{S^5} d^5 x \ \sqrt{g_{S^5}} = \frac{L^8}{z_h^3} \operatorname{Vol}(\mathbb{R}^3) \operatorname{Vol}(S^5) = \frac{\pi^3 L^8}{z_h^3} \operatorname{Vol}(\mathbb{R}^3)$$

$$= \pi^6 T^3 L^8 \operatorname{Vol}(\mathbb{R}^3), \qquad (E.1.9)$$

where we have used (E.1.7). Now, a nice holographic calculation [43] shows that the ten-dimensional Newton constant G is given by

$$G = \frac{\pi^4 L^8}{2N^2},$$
 (E.1.10)

implying that the entropy of the black D<sub>3</sub>-brane, and—by extension—the field theory, takes the form<sup>3</sup>

$$S = \frac{A}{4G} = \frac{\pi^2}{2} N^2 T^3 \text{Vol}(\mathbb{R}^3).$$
(E.1.12)

This implies that the free energy density takes the form  $f = -\int dT S(T)/Vol(\mathbb{R}^3) = -\frac{1}{8}N^2\pi^2T^4$ , which, as noted in [12], is precisely the Stefan-Boltzmann law. Finally, note that (E.1.12) is a highly non-trivial result: it is impossible to calculate the entropy  $\mathcal{N} = 4$  SYM via the usual route—attempting to do so would require summing infinitely many Feynman diagrams to arbitrarily high loop order, which is not presently feasible. It is, however, instructive to compare this with the entropy of free  $\mathcal{N} = 4$  SYM. This is given by [191]

$$S_{\text{Free}} = \frac{2\pi^2}{3} N^2 T^3 \text{Vol}(\mathbb{R}^3),$$
 (E.1.13)

implying that the entropy varies extremely slowly with coupling (see also [11]).

# E.1.2 Gravity Dual of Finite Temperature Field Theory II: AdS Schwarzschild Black Hole

Rather than starting from the black D3-brane solution of type IIB SUGRA, we can instead, as Witten does in [188]—and as is done in [11–13]—focus on the ordinary planar AdS-Schwarzschild Black hole, which in (d + 1) dimensions<sup>4</sup> has the form

$$ds^{2} = \frac{L^{2}}{z^{2}} \left( -f(z)dt^{2} + d\vec{x}^{2} + f(z)^{-1}dz^{2} \right),$$
 (E.1.14)

in Poincaré-like coordinates, where  $f(z) = 1 - \frac{z^d}{z_h^d}$ . For d = 4, this metric is identical to the black D3brane throat metric, and we may apply the same general method to compute the Hawking temperature of the  $\operatorname{Ads}_{d+1}$  Schwarzschild black hole; in particular, we apply the coordinate transformation  $\rho = \frac{2L}{\sqrt{d}}\sqrt{1-\frac{z}{z_h}}$  and Wick rotate,  $\tau = it$ , which puts the metric into the form  $ds^2 = \frac{d^2\rho^2}{4z_h^2}d\tau^2 + d\rho^2 + \frac{L^2}{z_h^2}d\vec{x}^2$ . Again, we focus on the  $(\tau, \rho)$  plane and rescale Euclidean time,  $\phi = d/(2z_h)\tau$ , to bring the  $(\tau, \rho)$  plane metric to the form  $ds^2 = d\rho^2 + \rho^2 d\phi^2$  which, to avoid a conical singularity, requires  $\phi \sim \phi + 2\pi$ , or, equivalently,  $\tau \sim \tau + \frac{4z_h\pi}{d}$ , implying that the Hawking temperature is

$$T = \frac{d}{4z_h \pi}.\tag{E.1.15}$$

<sup>2</sup> The volume of  $S^d$  is given by  $\operatorname{Vol}(S^d) = \frac{2\pi^{(d+1)/2}}{\Gamma(\frac{d+1}{2})}$ .

$$T = \alpha \frac{1}{d+1}, \ U = \alpha \frac{d}{d+1}, \ S = \alpha T^d.$$
 (E.1.11)

in *d* dimensions, where  $\alpha$  is a constant. Our findings above, e.g. (E.1.12), confirm this behavior. For more details see [11].

F

<sup>3</sup> We note in passing that for a conformal theory, which possesses no internal energy scales, the temperature dependence of thermodynamic variables is entirely determined by scaling,

<sup>4</sup> Since we forget about the supergravity origins of the objects under scrutiny, we can keep the dimensionality general.

It is useful to generalize this argument to generic AAdS manifolds, which in particular include the static black holes that we are currently dealing with. These have the form of (E.1.14) for some function f(z) with a first order zero at the black hole horizon  $z_h$ . Now, we can ignore the conformal factor  $L^2/z^2$ , since the Hawking temperature is invariant under conformal transformations (see [192]), so, after Wick rotating, we introduce  $\rho^2 = \frac{4}{f'(z_h)}(z - z_h)$  and  $\frac{1}{2}f'(z_h)$ , which brings the (conformal) metric—to lowest order in  $\rho$ —into the form in the  $(\phi, \rho)$  plane,  $ds^2 = d\rho^2 + \rho^2 d\phi^2$ , so, to avoid the usual conical singularity, we must identify  $\phi \sim \phi + 2\pi$ , which leads to the periodicity  $\tau \sim \tau + \frac{4\pi}{|f'(z_h)|}$ , so that the Hawking temperature takes the form:

$$T = \frac{|f'(z_h)|}{4\pi}.$$
 (E.1.16)

Thus, considering a planar Schwarzschild black hole reproduces the results obtained from the black D3-brane (for appropriate values of the dimension). This represents a more heuristic approach to the AdS/CFT correspondence: rather than referring to the supergravity roots of the correspondence, we can instead take the correspondence for granted and consider various field configurations in AdS to see what consequences they have in the field theory. This *bottom-up* approach is very useful, but lacking an explicit string theory embedding comes at a cost: determining relations between certain parameters (such as the Newton constant in (E.1.10)) is no longer possible.

# E.1.3 The D<sub>3</sub>/D<sub>7</sub> Brane Configuration and the AdS Reissner-Nordström Black Hole

Referring to the holographic dictionary of 2.3, we observe that a finite chemical potential  $\mu$  in the field theory requires an electric monopole in the bulk: the time component of the bulk U(1) gauge field is related to  $\mu$ . To motivate this, we consider a theory with global U(1) gauge containing a massless complex scalar and a massless Dirac fermion charged under the gauge symmetry. The Lagrangian of such a theory reads

$$\mathcal{L} = -(D_{\mu}\varphi)^{*}D^{\mu}\varphi + i\bar{\psi}\mathcal{D}_{\mu}\psi - \frac{1}{4g^{2}}F^{\mu\nu}F_{\mu\nu}, \qquad (E.1.17)$$

where the covariant derivative is given by  $D_{\mu} = \partial_{\mu} - iA_{\mu}$ . Now, introduce a non-vanishing background field  $\mu \in \mathbb{R}$  for the time component of the gauge field around which the dynamic field fluctuates,  $\tilde{A}_0 = \mu + A_0$ . This constant shift will not affect the field strengths and thus generates an extra potential  $V = \mu^2 \varphi^* \varphi - \mu \psi^{\dagger} \psi$ , where we have used the fact that  $\bar{\psi} \gamma^0 \psi = \psi^{\dagger} \beta \gamma^0 \psi = \psi^{\dagger} \psi$ , since the  $\beta$ -matrix and  $\gamma^0$  are numerically equivalent (though they differ in van der Waerden indices) and square to one (this is why we need the time component—none of the other  $\gamma$ -matrices would produce the pure number operator). The modified Lagrangian becomes  $\tilde{\mathcal{L}} = \mathcal{L} + V$ , and we recognize  $\psi^{\dagger} \psi$  as the number density operator. Further, we see that the scalar acquires a negative mass squared,  $-\mu^2$ , which is allowed by the BF bound (2.4.10) as long as it isn't too negative. This analysis naturally leads us to propose that a finite chemical potential is holographically encoded in a bulk one-form  $A = A_t(r) dt$  with near-boundary  $(r \to \infty)$ 

$$A_t(r) \sim \mu + \frac{\tilde{\rho}}{r^{d-2}},\tag{E.1.18}$$

in *d* dimensions, where  $\tilde{\rho}$  is related to the density  $\rho$ . There is one obstacle, however: there is no one-form field in type IIB SUGRA! There are two ways to get around this, which we now describe.

# E.1.3.1 D3/D7 Brane Configuration

As described in [43, 193], one can introduce a stack of  $N_f$  (with f standing for *flavor*) D7-branes in addition to the usual stack of N D3-branes. One can show that the theory dual to this configuration is  $\mathcal{N} = 4$  SYM coupled to  $N_f$  multiplets of  $\mathcal{N} = 2$  with gauge group  $U(N_f)$ . In particular, the  $\mathcal{N} = 2$  multiplets transform in the fundamental representation of the gauge group—just as quarks in ordinary QCD. Collectively, we dub the fields in such a multiplet *flavor fields*. The mass of the flavor fields turns out to be given by the separation  $\ell$  between the brane-stacks,  $m_f = \frac{\ell}{2\pi\alpha'}$ . To get the desired U(1) symmetry of the field theory, we need just a single flavor multiplet,  $N_f$ , which is well within the so-called *probe limit*, where any back-reaction to the supergravity solution can be neglected; that is: the AdS<sub>5</sub> × S<sup>5</sup> geometry of the D3-branes will still be the near-horizon geometry. In this probe limit, the

combined action becomes  $S_{\text{probe}} = S_{\text{IIB-SUGRA}} + S_{\text{D7}}$ , where  $S_{\text{D7}}$  at T = 0 reduces to the DBI action<sup>5</sup>. In this case, we have

$$S_{\rm D7} = -\tau_7 \int d^8 \xi \sqrt{-\det(\gamma_{ab} + (2\pi\alpha')F_{ab})},$$
 (E.1.19)

where  $\tau_7$  is the D7-brane tension. Now, the rest of the argument proceeds as follows: the field theory has, by Noether's theorem, a conserved current  $J^{\mu}$  corresponding to the global U(1) symmetry, the (expectation value of) time component of which is precisely the charge density,  $\rho = \langle J^0 \rangle$ . Obtaining and solving the equations of motion of (E.1.19), we can find the on-shell action  $S_{D7}^{\text{on-shell}}$ , which we may then renormalize and identify with the grand canonical potential,  $\Omega \sim S_{D7,\text{ren}}^{\text{on-shell}}$ . Noting that the density satisfies  $\langle J^0 \rangle \sim \frac{\partial \Omega}{\partial u}$ , we obtain

$$\langle J^0 \rangle = \frac{\delta S_{\text{D7,ren}}^{\text{on-shell}}}{\delta A_t (r = \infty)}.$$
(E.1.20)

Performing this variation, we obtain

$$\langle J^0 \rangle \sim \tilde{
ho},$$
 (E.1.21)

up to string-theoretical factors.

### E.1.3.2 Bottoms Up & The AdS Reissner-Nordström Black Hole

As described in [11–13, 194], another—perhaps more useful method—of obtaining a finite chemical potential and a global U(1) symmetry in the field theory is to simply add a Maxwell gauge field in the bulk, in a "bottom-up approach" to holography. The bulk is consequently described by the negative cosmological constant Einstein-Maxwell action in (d + 1) dimensions,

$$S_{\rm EM} = \int d^d x \, \sqrt{-g} \left[ \frac{1}{2\kappa^2} \left( R + \frac{d(d+1)}{L^2} \right) - \frac{1}{4g_F^2} F_{\mu\nu} F^{\mu\nu} \right], \tag{E.1.22}$$

where  $g_F$  is the electromagnetic coupling and the field strength is  $F_{\mu\nu} = \partial_{\mu}A_{\nu} - \partial_{\nu}A_{\mu}$ . Varying this action produces the equations of motion,

$$R_{\mu\nu} - \frac{R}{2}g_{\mu\nu} - \frac{d(d-1)}{L^2}g_{\mu\nu} = \frac{\kappa^2}{2g^2} \left(2F_{\mu\sigma}F_{\rho\nu}g^{\sigma\rho} - \frac{1}{2}g_{\mu\nu}F_{\sigma\rho}F^{\sigma\rho}\right), \quad \nabla_{\mu}F^{\mu\nu} = 0, \quad (E.1.23)$$

which are solved by the AdS Reissner-Nordström planar black hole,

$$ds^{2} = \frac{L^{2}}{z^{2}} \left( -f(z)dt^{2} + f(z)^{-1}dz^{2} + d\vec{x}^{2} \right),$$
 (E.1.24)

where  $f(z) = 1 - Mz^d + Q^2 z^{2(d-1)}$ . The horizon, determined by the condition  $f(z_0) = 0$ , has two solutions; we identify the larger of the two, located at  $z_+$ , with the horizon. The parameters involved in f(z) are explicitly given by [13]

$$M = z_{+}^{-d} + \frac{z_{+}^{2-d}\mu^2}{\gamma^2}, \quad Q^2 = \frac{z_{+}^{4-2d}\mu^2}{\gamma^2}, \quad (E.1.25)$$

where

$$\gamma^2 = \frac{(d-1)g_F^2 L^2}{(d-2)\kappa^2},\tag{E.1.26}$$

Finally, the Maxwell one-form becomes

$$A = \mu \left( 1 - \left(\frac{z}{z_+}\right)^{d-2} \right) \mathrm{d}t. \tag{E.1.27}$$

<sup>5</sup> In principle, there is also a contribution from a Chern-Simons term, but since it vanishes at zero temperature we'll ignore it.

We now proceed to compute the Hawking temperature of the RN-AdS black hole:

$$T = \frac{|f'(z_h)|}{4\pi} = \frac{\left|(2d-2)Q^2 z_+^{2d-3} - dM z_+^{d-1}\right|}{4\pi} = \frac{1}{4\pi z_+} \left(d - \frac{(d-2)z_+^2 \mu^2}{\gamma^2}\right).$$
 (E.1.28)

We may then determine  $\rho = \langle J^t \rangle$  in the same way as previously outlined: we compute the grand canonical ensemble potential at the boundary as  $\Omega = -T \log \mathcal{Z}$ , where  $\mathcal{Z}$  is the gravity partition function in the saddle-point approximation,  $\mathcal{Z} = \exp \left[-S_E^{\text{on-shell}}\right]$  with E standing for Euclidean. We may then identify  $\Omega = TS_E^{\text{on-shell}}$ , and the density relation becomes  $\rho = \langle J^t \rangle = -\frac{1}{\operatorname{Vol}(\mathbb{R}^{d-1})} \frac{\partial \Omega}{\partial \mu}$ .

#### E.2 THE S-WAVE HOLOGRAPHIC SUPERCONDUCTOR

First proposed in [5], where the "vanilla" *s*-wave superconductor was constructed in a bottom-up manner, holographic superconductors has since grown into a large area of research, with the more complicated varieties, the *p*- and *d*-wave superconductors, which carry angular momenta of  $\ell = 1$  and  $\ell = 2$ , respectively, now having been embedded in a holographic framework [195–197]. To get to grips with the formalism, we now describe the simplest holographic superconductor: the *s*-wave holographic superconductor, following the reviews [11, 13, 198–200].

As we have seen, systems with finite U(1) charge density are gravitationally encoded in AdS-Reissner-Nordström black holes, which arise as solutions to the Einstein-Maxwell action with negative cosmological constant. Superconductivity is a consequence of spontaneous breaking of a (gauged) U(1) symmetry to  $\mathbb{Z}_2$ . For our purposes, however, we focus on *global* U(1) symmetry, since we cannot holographically obtain a gauged U(1) symmetry on the boundary. This implies that what we are *really* describing is a holographic superfluid.

In this scenario, photons are effectively treated as an external electromagnetic field, corresponding to non-dynamical photons<sup>6</sup>. This global symmetry is then taken to be spontaneously broken: a charged operator—in this case, the Cooper pair order parameter—acquires a finite VEV.

The holographic dictionary (cf. 2.3) tells us that such a charged operator is itself dual to charged operators (with gauged U(1) symmetry) in the bulk—in particular, this implies that a superfluid on the boundary is dual to a form of superconductivity in the bulk. In order to simplify the analysis, we take the charged field to be a complex scalar (Higgs field)  $\varphi$  coupled to a gauge field  $A_{\mu}$ , so the bulk action describing the holographic *s*-wave superconductor takes the form

$$S = \int d^{d+1}x \sqrt{-g} \left[ \frac{1}{2\kappa_2} \left( R + \frac{d(d-1)}{L^2} \right) - \frac{1}{4g_F} F_{\mu\nu} F^{\mu\nu} - \left| D_{\mu} \varphi \right|^2 - V(|\varphi|) \right], \quad (E.2.1)$$

where  $D_{\mu} = \partial_{\mu} - iqA_{\mu}$  is the gauge covariant derivative and q the charge of the scalar field. In what follows, we'll specialize to d = 3 and take the potential to be a mass term,  $V(|\varphi|) = m^2 |\varphi|^2$ . Interestingly, this model has been realized as a Freund-Rubin compactification if M-theory in [8]. Furthermore, before the advent of holographic superconductors, this model was studied quite thoroughly by Gubser in [201]. To proceed, we note that the normal—non-superconducting—state of the theory, which satisfies  $\varphi = 0$ , is given AdS-RN planar black hole. We now wish to lower the temperature see if we can obtain a transition to a charged condensate at a critical temperature,  $T_c$ .

In order to obtain such a charged condensate, we need a finite VEV,  $\langle \mathcal{O} \rangle$ , where  $\mathcal{O}$  is the operator dual to  $\varphi$ . In particular, this implies the existence of a stability of the AdS-RN black hole with respect to perturbations of  $\varphi$ . Now, the equation of motion for  $\varphi$  is obtained by varying the action with respect to  $\varphi^*$ ,  $\frac{\delta S}{\delta \sigma^*} = 0$ , leading to

$$\delta_{\varphi^*} S = -\int d^4 x \,\sqrt{-g} \left[ \left( \partial_\mu \delta \varphi^* + iq A_\mu \delta \varphi^* \right) D^\mu \varphi - m^2 \varphi \delta \varphi^* \right] \tag{E.2.2}$$

$$\stackrel{\text{IBP}}{=} \int d^4x \, \left[ \partial_\mu \left( \sqrt{-g} D^\mu \varphi \right) - iq \sqrt{-g} A_\mu D^\mu \varphi - \sqrt{-g} m^2 \varphi \right] \delta \varphi^*, \tag{E.2.3}$$

implying that the equations of motion take the form

$$0 = \frac{1}{\sqrt{-g}} \partial_{\mu} \left( \sqrt{-g} D^{\mu} \varphi \right) - iq A_{\mu} D^{\mu} \varphi - m^2 \varphi = \nabla_{\mu} (D^{\mu} \varphi) - iq A_{\mu} D^{\mu} \varphi - m^2 \varphi.$$
(E.2.4)

<sup>6</sup> This neglecting of effectively virtual photons is also made in BCS theory, so saying that we are describing a superconductor is less of a stretch than it may seem at first.

where we have used that  $\nabla_{\mu}V^{\mu} = \frac{1}{\sqrt{-g}}\partial_{\mu}(\sqrt{-g}V^{\mu})$ . Further, using  $\nabla_{\mu}\varphi = D_{\mu}\varphi$ , the above can be recast in the form

$$(\nabla_{\mu} - iqA_{\mu})(\nabla^{\mu} - iqA^{\mu})\varphi - m^{2}\varphi = 0.$$
(E.2.5)

Alternatively, the equations of motion can be obtained from doing the minimal coupling procedure twice: starting from the flat space non-charged coupled scalar Lagrangian,

$$\mathcal{L} = -\partial_{\mu}\varphi^*\partial^{\mu}\varphi - m^2\varphi^*\varphi, \qquad (E.2.6)$$

the Euler-Lagrange equations of motion  $\frac{\partial \mathcal{L}}{\partial \varphi} - \partial_{\mu} \frac{\partial \mathcal{L}}{\partial(\partial_{\mu} \varphi)}$  lead to the Klein-Gordon equation,

$$\partial_{\mu}\partial^{\mu}\varphi - m^{2}\varphi = 0, \qquad (E.2.7)$$

which we can "gauge minimally couple",  $\partial_{\mu} \rightarrow D_{\mu}$ 

$$D_{\mu}D^{\mu}\varphi - m^{2}\varphi = 0, \qquad (E.2.8)$$

which we can again "gravitationally minimally couple", where the  $\partial_{\mu}$  that sits inside  $D_{\mu}$  is sent to  $\nabla_{\mu}$ , in which case we end up with

$$(\nabla_{\mu} - iqA_{\mu})(\nabla^{\mu} - iqA^{\mu})\varphi - m^{2}\varphi = 0.$$
(E.2.9)

Moving on, we see that this induces a mass correction to the scalar:

$$m_{\rm eff}^2 = m^2 + q^2 A_\mu A^\mu, \tag{E.2.10}$$

which in the presence of an exclusively electrostatic gauge field,  $A_i = 0$ , becomes,

$$m_{\rm eff}^2 = m^2 - |g^{tt}| q^2 A_t^2.$$
 (E.2.11)

### E.2.1 Holographic Superconductivity in the Probe Limit (d = 3)

We now turn to the simplest model for holographic superconductivity first proposed in the seminal paper [5]. The key simplifying assumption is to ignore backreaction on the metric, that is, we take  $\kappa^2 \ll g^2 L^2$ , which decouples the Maxwell-Higgs sector from the gravity sector. This results in an AdS-Schwarzschild background described by

$$ds^{2} = -f(r)dt^{2} + \frac{dr^{2}}{f(r)} + r^{2}\left(dx^{2} + dy^{2}\right),$$
(E.2.12)

where

$$f = \frac{r^2}{L^2} - \frac{M}{r}.$$
 (E.2.13)

The Hawking temperature is determined in the usual fashion (cf. (E.1.16)),

$$T = \frac{|f'(r_h)|}{4\pi},$$
 (E.2.14)

where  $r_h$  is determined by the condition  $f(r_h) = 0$ , that is,  $r_h = L^{2/3} M^{1/3}$ , implying that

$$T = \frac{3M^{1/3}}{4L^{4/3}\pi'},\tag{E.2.15}$$

taking the potential to be  $V(|\varphi|) = -\frac{2|\varphi|^2}{L^2}$ , and employing a radial ansatz,  $\varphi = \phi(r)$ , we find that the equations of motion take the form—assuming q = 1 and that the spatial part of A vanishes,

$$0 = \nabla_{\mu}\nabla^{\mu}\phi - g^{tt}A_{t}^{2}\phi + \frac{2}{L^{2}}\phi = \frac{1}{\sqrt{-g}}\partial_{r}\left(\sqrt{-g}g^{rr}\partial_{r}\phi\right) + \frac{A_{t}^{2}}{f} + \frac{2}{L^{2}}\phi$$
(E.2.16)

$$= \frac{1}{r^2} \partial_r \left( r^2 f \phi' \right) + \frac{A_t^2}{f} \phi + \frac{2}{L^2} \phi \tag{E.2.17}$$

$$= f\phi'' + \left(f' + \frac{2f}{r}\right)\phi' + \frac{A_t^2}{f}\phi + \frac{2}{L^2}\phi$$
(E.2.18)

where we have used that  $\sqrt{-g} = r^2$ . Dividing by *f* yields the equation of motion:

$$\phi'' + \left(\frac{f'}{f} + \frac{2}{r}\right)\phi' + \frac{A_t^2}{f^2}\phi + \frac{2}{fL^2}\phi = 0,$$
(E.2.19)

in agreement with [5].

Doing the same for  $A_t$ , we find

$$A_t'' + \frac{2}{r}A_t' - \frac{2\phi^2}{f}A_t = 0.$$
 (E.2.20)

while the equation of motion for  $\phi$  implies that it is a constant phase—these findings are in agreement with those presented in the original article, [5]. We note that  $\phi$  provides a mass to the bulk theory gauge field via a Higgs mechanism.

Let's now discuss boundary conditions. A priori, we have a four-parameter family of solutions; in order to remedy this, we impose boundary conditions. First, requiring  $A_t$  to have finite norm at the horizon,  $r_0$ , defined as the value of r satisfying  $f(r_0) = 0 \Rightarrow r_0 = (ML^2)^{1/3}$ , we see that  $g^{tt}(r_0)A_t(r_0) < 0$ , implying that  $A_t(r_0) = 0$  (since  $\lim_{r \to r_0} g^{tt} = \infty$ ). In this way, we infer that  $A'_t \sim A''_t$  at the horizon; that is, they are not independent. Plugging  $A_t(r = r_0) = 0$  into (E.2.19), which we then proceed to multiply by f, gives us,

$$0 = f'(r_0)\phi'(r_0)(r_0)\phi'(r_0) + \frac{2}{L^2}\phi(r_0) = \frac{3r_0}{L^2}\phi'(r_0) + \frac{2}{L^2}\phi,$$
 (E.2.21)

where we have used that  $M = \frac{r_0^2}{L^2}$ . This can be rewritten as,

$$\phi(r_0) = -\frac{3}{2}r_0\phi'(r_0), \qquad (E.2.22)$$

thus leading us to conclude that at the horizon  $r = r_0$ , both  $\phi$  and  $A_t$  are described by one-parameter families of solutions. A similar asymptotic analysis can be carried out for the boundary: assuming the leading behaviour (at  $r \to \infty$ ) for the fields to be of the general form,

$$\phi(r) = K_{\phi} r^{\alpha_{\phi}} + \cdots, \quad A_t(r) = K_{A_t} r^{\alpha_{A_t}} + \cdots, \quad (E.2.23)$$

where  $K_{\phi}$ ,  $K_{A_t}$  are constants. We then insert the above into the appropriate equations of motion; for  $\phi$ , (E.2.19) gives us (where we have expanded the coefficients around  $r = \infty$ ),

$$\alpha_{\phi}(\alpha_{\phi}-1)K_{\phi}r^{\alpha_{\phi}-2} + \alpha_{\phi}K_{\phi}r^{\alpha_{\phi}-1}\left(\frac{4}{r}+\cdots\right) + r^{\alpha_{\phi}+2\alpha_{A_{t}}}\left(\frac{K_{\phi}K_{A_{t}}^{2}L^{4}}{r^{4}}+\cdots\right) + K_{\phi}r^{\alpha_{\phi}}\left(\frac{2}{r^{2}}+\cdots\right) = 0,$$
(E.2.24)

so, to leading order, we obtain

$$\alpha_{\phi}(\alpha_{\phi}-1) + 4\alpha_{\phi} + K_{A_t}^2 L^4 r^{2\alpha_{A_t}-2} + 2 = 0, \qquad (E.2.25)$$

which means that  $\alpha_{\phi} = \frac{1}{2} \left( -3 \pm \sqrt{1 - 4B^2 L^4 r^{2\alpha_{A_t} - 2}} \right)$ . Since we have constrained  $\phi \in \mathbb{R}$ , we must demand that  $\alpha_{A_t} < 1$ , implying that (E.2.25) becomes

$$\alpha_{\phi}(\alpha_{\phi} - 1) + 4\alpha_{\phi} + 2 = 0. \tag{E.2.26}$$

which means that

$$\alpha_{\phi} = -1, -2,$$
(E.2.27)

so, since the equations of motion are linear, we can write the solution as:

$$\phi = \frac{\phi^{(1)}}{r} + \frac{\phi^{(2)}}{r^2} + \dots$$
(E.2.28)

Similarly, the equation (E.2.20) implies (after plugging in the solution (E.2.28) and getting rid of subleading terms)

$$\alpha_{A_t}(\alpha_{A_t} - 1) + 2\alpha_{A_t} = 0, \tag{E.2.29}$$

so that  $\alpha_{A_t} = 0, -1$ , and so

$$A_t = \mu - \frac{\rho}{r} + \cdots, \qquad (E.2.30)$$

where the density is determined by the condition  $A_t(r_0) = 0$ , so that the charge density is given by  $\rho = \mu r_0$ . Both terms in  $\phi$  above are normalizable<sup>7</sup>. Now, recall that (cf. 2.2)  $m^2 L^2 = \Delta(\Delta - 3)$ . Since we have chosen  $m^2 L^2 = -2$ , this corresponds to either  $\Delta = 1$  or  $\Delta = 2$ . Recalling our discussion in section 2.4, we see that if  $\Delta = 2$ ,  $\phi^{(1)}$  is interpreted as a source for the dual operator in the field theory, whereas  $\phi^{(2)} \sim \langle \mathcal{O}_2 \rangle$  is the VEV.

Similarly, if  $\Delta = 1$ , the discussion of section 2.4 implies that the rôles are switched: now  $\phi^{(1)} \sim \langle O_1 \rangle$  is the VEV, whereas  $\phi^{(2)}$  acts as a source. In order to study phase transitions, we turn off the source and look for solutions where the VEV acquires a non-zero value at some critical temperature—this leaves a one-parameter family of solutions for both  $\phi$  and  $A_t$ . In summary, then, we have

$$\langle \mathcal{O}_i \rangle = \sqrt{2} \phi^{(i)}, \ i = 1, 2, \ \varepsilon_{ij} \phi^{(j)} = 0.$$
 (E.2.31)

where, following [5], we have chosen the  $\sqrt{2}$ -normalization, which turns out to simplify subsequent calculations. From now on, we also set M = 1 = L. Next, note that  $[\rho] = 2$  and [T] = 1, implying that, first off,  $\frac{\langle O_i \rangle}{T^i}$  are dimensionless quantities, and that  $T_c \sim \rho^{1/2}$ . Solving the equations numerically<sup>8</sup> yields the curves of figure E.1.



Figure E.1: The condensate as a function of  $O_i$  exhibiting the usual second order Ginzburg-Landau behaviour near the phase transition,  $\langle O \rangle_i \sim (1 - T/T_c)^{1/2}$ .

Fitting the curves of figure E.1, we get:

$$T_c^{(1)} \simeq 0.2255 \rho^{1/2}, \ T_c^{(2)} \simeq 0.1184 \rho^{1/2}.$$
 (E.2.32)

Note further that at low temperatures, the condensate VEV  $\langle O_1 \rangle$  begins to diverge. This is an artifact of the probe approximation that we're employing: for large values of the condensate, we can no longer ignore backreaction to the bulk metric, so our scheme is invalid in this region. We shall see, by studying the conductivity of the system, that  $\langle O_1 \rangle$  and  $\sqrt{\langle O_2 \rangle}$  can be interpreted as twice the superconducting gap (which is conventionally denoted  $\Delta$ , but since this clashes with the notation we use for the conformal dimension, we'll simply write it as "gap"). BCS theory predicts that at T = 0, we have  $2 \times \text{gap} = 3.54T_c$  [202]; for our holographic superconductor, twice the gap is either infinite for  $\Delta = 1$  or  $2 \times \text{gap} \simeq 8T_c$  for  $\Delta = 2$ , which is a characteristic of strongly interacting high- $T_c$ superconductors (see [202] for more details).

<sup>7</sup> That is, their action is finite.

<sup>8</sup> Christopher Herzog has provided a nice Mathematica document that does this on his web-page: click here.

#### E.2.2 Conductivity in the Probe Limit

#### E.2.2.1 Conductivity and the Drude Model

In this section we discuss the physics of conductivity as it appears in condensed matter systems, following [76, 203]. Consider a spatially constant electric field varying in time. After Fourier transforming, Ohm's law reads

$$\mathbf{j}(\omega) = \sigma(\omega)\mathbf{E}(\omega),\tag{E.2.33}$$

where  $\sigma(\omega)$ , which in general will be complex, is the optical conductivity. Note in particular that if we shake the electric field at some frequency  $\omega$ , the "system" responds at the same frequency: this where linear response can be applied. The real part of the optical conductivity,  $\text{Re}[\sigma(\omega)]$  describes the dissipation of the current; in other words, it behaves as an actual conductivity, whereas the imaginary part,  $\text{Im}[\sigma(\omega)]$ , describes the so-called reactive part. Next, let us briefly discuss the Drude model: consider, therefore, a particle of mass *m*, charge *q* and velocity **v**, described by Newton's second law with a friction term linear in the velocity,

$$m\frac{\mathrm{d}\mathbf{v}}{\mathrm{d}t} + \frac{m}{\tau}\mathbf{v} = q\mathbf{E},\tag{E.2.34}$$

where  $\tau$  is the scattering time (or mean free path); the average time the particle travels unhindered before bumping into something. If we then turn on an AC electric field with frequency  $\omega$ ,  $\mathbf{E}(t) = \text{Re} \left[\mathbf{E}_0 e^{-i\omega t}\right]$ , the solution to the Drude differential equation (E.2.34) becomes

$$\mathbf{v}(t) = \frac{q\tau}{m} \frac{\mathbf{E}(t)}{1 - i\omega\tau} + \underbrace{\mathsf{Const.} \times e^{-t/\tau}}_{\mathsf{Const.} \times e^{-t/\tau}}, \qquad (E.2.35)$$

so, assuming we're in the steady state the regime where the second term can be safely neglected, the optical conductivity becomes; according to (E.2.33)

$$\sigma(\omega) = \frac{\sigma_0}{1 - i\omega\tau}, \quad \sigma_0 = \frac{nq^2\tau}{m}, \quad (E.2.36)$$

where we have used that  $\mathbf{j} = nq\mathbf{v}$ , where *n* is the electron density and defined the DC ( $\omega = 0$ ) conductivity,  $\sigma_0$ . At high frequencies, particle-anti-particle creation contributes to the conductivity, but for low frequencies, the Drude model captures the physics nicely. In the superconducting limit, where the mean free path is infinite,  $\tau \rightarrow \infty$ , we may use L'Hôpital's rule to obtain:

$$\lim_{\tau \to \infty} \sigma(\omega) = \lim_{\tau \to \infty} \frac{nq^2}{-i\omega m} = i \frac{nq^2}{m\omega},$$
(E.2.37)

but since the conductivity is analytic, the real part can be determined<sup>9</sup> via the Kramers-Kronig relation,

$$\operatorname{Re}\left[\sigma(\omega)\right] = \frac{1}{\pi} \mathcal{P} \int_{-\infty}^{\infty} d\omega' \frac{\operatorname{Im}\left[\sigma(\omega')\right]}{\omega' - \omega}$$
(E.2.38)

$$=\frac{nq^2}{m\pi}\mathcal{P}\int_{-\infty}^{\infty}\mathrm{d}\omega'\frac{1}{\omega'\left(\omega'-\omega\right)}\tag{E.2.39}$$

$$=\frac{\pi ne^2}{m}\delta(\omega). \tag{E.2.40}$$

Let's quickly prove this: consider the original integral

$$I = \mathcal{P} \int_{-\infty}^{\infty} d\omega' \frac{1}{\omega'(\omega' - \omega)},$$
 (E.2.41)

and define for infinitesimal  $\epsilon$  and  $\eta$ 

$$I_{\varepsilon,\eta} = \overbrace{\int_{-\infty}^{-\eta} d\omega' \frac{1}{\omega'(\omega' - \omega + i\varepsilon)}}^{=:(*)} + \int_{\eta}^{\infty} d\omega' \frac{1}{\omega'(\omega' - \omega + i\varepsilon)}, \quad (E.2.42)$$

<sup>9</sup> Alternatively, one can observe that the integral  $\int_{-\infty}^{\infty} \operatorname{Re} \left[\sigma(\omega)\right] = \frac{\pi n q^2}{m}$  is independent of  $\tau$ , and thus infer the presence of the  $\delta$ -function in the real part of the optical conductivity in the superconducting limit. This is an example of a sum rule; this particular one is known as the Ferrell-Glover-Tinkham sum rule.

which we have chosen such that  $I = \lim_{\epsilon, \eta \downarrow 0} \operatorname{Re}[I_{\epsilon, \eta}]$ . Next, note that under a change of variables  $\chi' = -\omega'$ , we have

$$(*) = \int_{\eta}^{\infty} \mathrm{d}\chi' \, \frac{1}{\chi'(\chi' + \omega - i\epsilon)},\tag{E.2.43}$$

so that

$$I_{\epsilon,\eta} = \int_{\eta}^{\infty} d\omega' \left[ \frac{1}{\omega'(\omega' - \omega + i\epsilon)} + \frac{1}{\omega'(\omega' + \omega - i\epsilon)} \right]$$
(E.2.44)

$$= \int_{\eta}^{\omega} d\omega' \frac{2}{\omega'^2 - (\omega - i\epsilon)^2}$$
(E.2.45)

$$= - \frac{\tanh^{-1}\left(\frac{\omega'}{\omega - i\epsilon}\right)}{\omega - i\epsilon} \bigg|_{\omega' = \eta}^{\omega = \infty}.$$
(E.2.46)

Taking the limit  $\eta \to 0$  and using  $\lim_{x \to \infty} \tanh^{-1}(x) = -\frac{i\pi}{2}$ , we find that

$$I_{\epsilon} := \lim_{\eta \downarrow 0} I_{\epsilon,\eta} = \frac{\pi}{i(\omega - i\epsilon)}, \tag{E.2.47}$$

the real part of which is

$$\operatorname{Re}\left[I_{\epsilon}\right] = \frac{\pi\epsilon}{\omega^2 + \epsilon^2},\tag{E.2.48}$$

which, as  $\epsilon \to 0$ , converges to  $I = \pi^2 \delta(\omega)$ —this can be seen as follows for some smooth test function f,

$$\int_{-\infty}^{\infty} dt f(t) \frac{\epsilon}{t^2 + \epsilon^2} \stackrel{t=\epsilon u}{=} \int_{-\infty}^{\infty} dx f(\epsilon x) \frac{1}{1 + x^2}$$
(E.2.49)

$$\stackrel{\epsilon \to 0}{\longrightarrow} f(0) \int_{-\infty}^{\infty} \mathrm{d}x \; \frac{1}{1+x^2} \tag{E.2.50}$$

$$=\pi f(0).$$
 (E.2.51)

This characteristic  $\delta$ -function behaviour of the conductivity in the superconducting limit is something we will see reproduced by our holographic model.

### E.2.2.2 Maxwell Perturbations: Fluctuations of $A_x$ in the Bulk

Consulting again the dictionary of 2.3, we see that if we want a current in our system, we need to introduce a spatial component of our gauge field with some finite frequency  $\omega$ ; following [5] we pick  $A_x$ —this is of no consequence due to the symmetry of the problem. We are interested in the zero-momentum (in the AdS-Sch geometry, the zero-momentum assumption decouples  $A_x$  from the other polarizations; if we did not make this assumption there would be interactions between the fluctuations that would have to be diagonalized, see [15]) conductivity and as such we take  $A_x(t,r) = a_x(r)e^{-i\omega t}$ , implying that the Maxwell equation for  $A_x$  reads:

$$\nabla_{\mu}F^{\mu x} = 2g^{xx}A_{x}\phi^{2} = \frac{2A_{x}\phi^{2}}{r^{2}},$$
(E.2.52)

where

$$\nabla_{\mu}F^{\mu x} = \frac{1}{\sqrt{-g}}\partial_{\mu}\left[\sqrt{-g}g^{\mu\nu}g^{\sigma x}F_{\nu\sigma}\right]$$
(E.2.53)

$$=\frac{1}{r^2}\partial_{\mu}\left[r^2g^{\mu\nu}g^{xx}F_{\nu x}\right] \tag{E.2.54}$$

$$=\frac{1}{r^2}\partial_r(fA'_x) + \frac{\omega^2 A_x}{r^2 f},$$
 (E.2.55)

so the full equation of motion becomes (where we have divided out the common exponential)

$$a_x'' + \frac{f'}{f}a_x' + \left(\frac{\omega^2}{f^2} - \frac{2\phi^2}{f}\right)a_x = 0.$$
 (E.2.56)
To ensure causality, we must impose in-falling boundary conditions at the horizon, r = 1; this imposition of boundary conditions at the horizon implies that at the horizon, we have  $a_x \sim f^{-i\omega/3}$  (see [43, 204]). Similarly, performing the near-boundary analysis of  $a_x$ , we obtain the equation  $(\alpha_{A_x} + 1)\alpha_{A_x} = 0$  implying that  $\alpha_{A_x} = 0, 1$ , and thus,

$$a_x = A^{(0)} + \frac{A^{(1)}}{r} + \dots,$$
 (E.2.57)

where  $A_x^{bd} = A_x^{(0)}$  is the gauge field on the boundary (the source), satisfying  $\partial_t A_x^{bd} = -E_x = -i\omega A_x^{bd}$ , where  $E_x$  is the electric field on the boundary. The other term,  $\langle J_x \rangle = A_x^{(1)}$ , represents the expectation value of the current. From Ohm's law, then, we obtain the following optical conductivitys

$$\sigma(\omega) = \frac{\langle J_x \rangle}{E_x} = -\frac{iA_x^{(1)}}{\omega A_x^{(0)}}.$$
(E.2.58)

Below, we plot the conductivity as a function of the frequency. At  $\omega = 0$ , a  $\delta$ -function appears when  $T < T_c$ : although it cannot be seen by the numerics, its existence may again be inferred by use of the Kramers-Kronig relation (or, alternatively—as before—by the Ferrell-Glover-Tinkham sum rule),

$$\operatorname{Im}\left[\sigma(\omega)\right] = -\frac{1}{\pi} \mathcal{P} \int_{-\infty}^{\infty} d\omega' \frac{\operatorname{Re}\left[\sigma(\omega')\right]}{\omega' - \omega}, \qquad (E.2.59)$$

from which we infer, from our previous considerations, that the real part of the conductivity contains a  $\delta$ -function, Re[ $\sigma(\omega)$ ] ~  $\pi \delta \omega$ , provided that Im[ $\sigma(\omega)$ ] has a simple pole, Im[ $\sigma(\omega)$ ] ~  $-\frac{1}{\omega}$ ; this is indeed the case [5, 6]. It is interesting to note that had we *not* simplified our problem by considering the probe limit but instead considered the full AdS-RN spacetime, the  $\delta$ -function is present even when  $T > T_c$ , which seemingly presents a problem: the DC conductivity is infinite even in the normal phase! However, this infinite conductivity is not true superconductivity but rather an artefact of the translation invariance of the system; fixing the background the AdS-Schw in the probe limit implicitly breaks translation invariance by decoupling electric and energy currents [6]. It is, however, possible to break the translation invariance of the AdS-RN setup by introducing impurities—this was considered in [205].



Figure E.2: Conductivities for  $\mathcal{O}_1$  (left) and  $\mathcal{O}_2$  (right). At the origin, there is a  $\delta$ -function unseen by the numerics, hile the horizontal line is the conductivity for  $T = T_c$ .

The horizontal line in both figures represents the conductivity in the case  $T \ge T_c$ —this is interesting because it was shown in [15] the conductivity is independent of frequency in the normal phase is characteristic of field theories with AdS<sub>4</sub> duals. The following curves describe successively lower values of the temperature with fixed charge density.

This appendix deals with additional approaches to holographic renormalization, as advertised in the main text. We begin by reviewing the original FG approach in section F.1 and then proceed to discuss the dBVV ansatz method in section F.2.

#### F.1 FEFFERMAN-GRAHAM HOLOGRAPHIC RENORMALIZATION

Following the review article [48], we describe the FG approach to holographic renormalization. In this section, we will use Euclidean signature AlAdS space (see appendix A), where the Fefferman-Graham metric (A.2.6) assumes the form

$$ds^{2} = L^{2} \left( \frac{d\rho^{2}}{4\rho^{2}} + \frac{1}{\rho} g_{ij}(x,\rho) dx^{i} dx^{j} \right),$$
(F.1.1)

with an expansion for  $g_{ij}$  entirely analogous to (A.2.8). Suppressing all spacetime and internal indices, a generic bulk field  $\mathcal{F}(\rho, x)$  has a near-boundary Fefferman-Graham asymptotic expansion of the form

$$\mathcal{F}(x,\rho) = \rho^m \left( f_{(0)}(x) + \rho f_{(2)}(x) + \dots + \rho^n \left[ f_{(2n)} + \log(\rho) \tilde{f}_{(2n)}(x) \right] + \dots \right).$$
(F.1.2)

Generally, the equations of motion for  $\mathcal{F}(x, \rho)$  are second order differential equations, which gives two independent solutions with asymptotic behaviors  $\rho^m$  and  $\rho^{m+n}$ —this is just a generalization of what we saw in (2.4.9), where a similar solution arises for the scalar. Just as for the scalar case, we identify  $f_{(0)}(x)$ , which multiplies the leading behavior, with the source for the dual operator. By solving the equations of motion order by order in  $\rho$  (which is treated as a small parameter in the near-boundary analysis), one obtains algebraic equations for  $f_{(2k)}(x)$  for k < n in terms of  $f_{(0)}(x)$  and its derivatives up to order 2k. This procedure, however, does not determine  $f_{(2n)}$ , which is associated with the VEV of the dual operator. In summary, the equations of motion imply that [48]

- $f_{(0)}(x)$  is the source of the dual field theory operator,
- $f_{(2)}, \ldots, f_{(2n-2)}$  and  $\tilde{f}_{(2n)}$  are uniquely determined by the equations of motion and are local functions of  $f_{(0)}$ ,
- $\tilde{f}_{(2n)}$  is related to conformal anomalies,
- $f_{(2n)}$ , which is related to the VEV of the dual field theory operator, is undetermined by the near-boundary analysis.

Thus equipped with a generic asymptotic solution to the given field equations of motion, we proceed to calculate the on-shell value of the bulk action. In order to do so, we need to regularize: this is implemented by restricting the range of the  $\rho$  integration,  $\rho \ge \epsilon$  for  $\epsilon \ll 1$ . We then evaluate the boundary terms at  $\rho = \epsilon$ , a finite number of which will diverge in the limit  $\epsilon \rightarrow 0$ —thus, we can write the regulated on-shell action in the following manner:

$$S_{\text{reg}}^{\text{on-shell}}[f_{(0)},\epsilon] = \int_{\rho=\epsilon} d^d x \,\sqrt{g_{(0)}} \left[ \epsilon^{-\nu} a_{(0)} + \epsilon^{-(\nu+1)} a_{(2)} + \dots - \log(\epsilon) \,a_{(2\nu)} + \mathcal{O}(\epsilon^0) \right], \quad (F.1.3)$$

where  $\nu > 0$  depends only on the conformal dimension on the dual operator, and the coefficients  $a_{(k)}$  are local functions of the source  $f_{(0)}$ , whereas  $a_{(2\nu)}$  is directly related to the conformal anomaly [169]. The counterterm action is then defined as the divergent (in the  $\epsilon \rightarrow 0$  limit) part of the regularized action,

$$S_{\rm ct}[\mathcal{F}(x,\epsilon);\epsilon] = -\text{divergent terms of } S_{\rm reg}^{\rm on-shell}[f_{(0)}(\mathcal{F}(x,\epsilon)),\epsilon], \tag{F.1.4}$$

where the divergent terms are expressed in terms of the fields  $\mathcal{F}(x, \epsilon)$  living at the regulated hypersurface  $\rho = \epsilon$ . This is required for covariance of the counterterm action, and consequently it is necessary

to invert the relation (F.1.2) up to the required order—this inversion is precisely what makes the FG approach so cumbersome. When the counterterm has been covariantized, we may define the subtracted action,

$$S_{\text{sub}}[\mathcal{F}(x,\epsilon);\epsilon] = S_{\text{reg}}^{\text{on-shell}}[f_{(0)},\epsilon] + S_{\text{ct}}[\mathcal{F}(x,\epsilon);\epsilon], \qquad (F.1.5)$$

in terms of which the renomarlized on-shell action is obtained as the limit  $\epsilon 
ightarrow 0$ ,

$$S_{\text{ren}}^{\text{on-shell}}[f_{(0)}] = \lim_{\epsilon \to 0} S_{\text{sub}}[\mathcal{F}(x,\epsilon);\epsilon].$$
(F.1.6)

Using the renormalized on-shell action, we can define exact one-point functions (VEVs) of the field theory operator  $\mathcal{O}_{\mathcal{F}}$  dual to the bulk field  $\mathcal{F}$  as the variation of  $S_{\text{ren}}^{\text{on-shell}}$  with respect to the boundary source, and since  $g_{(0)ij}$  is in general non-trivial, we obtain (the subscript *s* stands for source)

$$\left\langle \mathcal{O}_{\mathcal{F}} \right\rangle_{s} = \frac{1}{\sqrt{g_{(0)}}} \frac{\delta S_{\text{ren}}^{\text{on-shell}}[f_{(0)}]}{\delta f_{(0)}}.$$
(F.1.7)

It can also be computed in terms of fields living at the regulated boundary,

$$\langle \mathcal{O}_{\mathcal{F}} \rangle_{s} = \lim_{\epsilon \to 0} \left( \frac{1}{\epsilon^{d/2 - m}} \frac{1}{\sqrt{\gamma}} \frac{\delta S_{\text{sub}}[\mathcal{F}(x,\epsilon);\epsilon]}{\delta \mathcal{F}(x,\epsilon)} \right), \tag{F.1.8}$$

where  $\gamma_{ij} = g_{ij}(x, \epsilon)/\epsilon$  is the induced metric on the regulated surface  $\rho = \epsilon$ . One can explicitly evaluate this limit, which gives the result [48]

$$\langle \mathcal{O}_{\mathcal{F}} \rangle_s \sim f_{2n} + \mathcal{C}(f_{(0)}),$$
 (F.1.9)

where the scheme-dependent<sup>1</sup> function  $C(f_{(0)})$  depends locally on  $f_{(0)}$ , while the coefficient in front of  $f_{(2n)}$  is scheme-independent.

## F.1.1 FG Holographic Renormalization of Pure Gravity in AlAdS<sub>5</sub>

To illustrate the application of the method described above, The Einstein-Hilbert action with the Gibbons-Hawking term<sup>2</sup> for  $\mathcal{M} = AlAdS_5$  with  $\Lambda = -\frac{d(d-1)}{2} = -6$ ,

$$S = \int_{\mathcal{M}} d^5 x \sqrt{-G} \left( R^{(G)} + 12 \right) + 2 \oint_{\partial \mathcal{M}} d^4 x \sqrt{-\gamma} K, \tag{F.1.10}$$

where  $\gamma_{\mu\nu}$  is the metric induced on the boundary, while *K* is the trace of the extrinsic curvature,  $K = \nabla_m n^m$ , where  $n^m$  is a unit normal vector to the boundary  $\partial \mathcal{M}$ . The AlAdS<sub>5</sub> (with L = 1) metric has the Fefferman-Graham form of (A.2.6),

$$G_{mn} dx^m dx^n = \frac{d\hat{\rho}^2}{4\hat{\rho}^2} + \frac{1}{\hat{\rho}} g_{\mu\nu}(\hat{\rho}, x) dx^\mu dx^\nu, \qquad (F.1.11)$$

where for d = 4, the expansion (A.2.8) reads

$$g_{\mu\nu}(x,\hat{\rho}) = g_{(0)\mu\nu}(x) + \hat{\rho}g_{(2)\mu\nu}(x) + \hat{\rho}^2 g_{(4)\mu\nu} + \hat{\rho}^2 \log(\hat{\rho})h_{(4)\mu\nu}(x) + \mathcal{O}(\hat{\rho}^3).$$
(F.1.12)

The boundary is located at  $\hat{\rho} = 0$ , and the corresponding boundary metric<sup>3</sup>, therefore, is  $g_{(0)\mu\nu}(x)$ . The reason why we can't use this metric immediately is, of course, that in the process of holographic renormalization we need to introduce a cut-off hypersurface at  $\hat{\rho} = \epsilon$ , so we need the full expansion. The equations of motion from varying the action (F.1.10) produces the usual Einstein equation,

$$R_{mn} - \frac{1}{2}RG_{mn} - 6G_{mn} = 0. (F.1.13)$$

In our calculations, we also require the inverse metric. In order to obtain it, we start with the ansatz

$$g^{\mu\nu} = \tilde{g}^{\mu\nu}_{(0)} + \hat{\rho}\tilde{g}^{\mu\nu}_{(2)} + \hat{\rho}^2\tilde{g}^{\mu\nu}_{(4)} + \cdots, \qquad (F.1.14)$$

<sup>1</sup> Since we have only subtracted the divergences, the scheme we have used to obtain (F.1.6) is, by analogy to the procedure in quantum field theory, known as minimal subtraction.

<sup>2</sup> The Gibbons-Hawking term is required to make the variational problem for gravity well posed, i.e. to ensure that we can impose Dirichlet boundary conditions consistently; see [206].

<sup>3</sup> Setting  $g_{(0)\mu\nu} = \eta_{\mu\nu}$  corresponds to imposing AAdS<sub>5</sub> boundary conditions.

### 138 HOLOGRAPHIC RENORMALIZATION: SUPPLEMENTARY MATERIAL

where  $\tilde{g}_{(4)}^{\mu\nu}$  contains logarithmic terms in  $\hat{\rho}$ . We have that  $G_{mn}$  is block diagonal

$$G_{mn} = \begin{pmatrix} \frac{1}{4\hat{\rho}^2} & 0\\ 0 & \frac{1}{\hat{\rho}}g_{\mu\nu} \end{pmatrix}, \qquad (F.1.15)$$

so its inverse is

$$G^{mn} = \begin{pmatrix} 4\hat{\rho}^2 & 0\\ 0 & \hat{\rho}g^{\mu\nu} \end{pmatrix}, \qquad (F.1.16)$$

which shows that  $g_{\mu\nu}(\hat{\rho} = \text{cst.})$  acts as a metric on slices of constant  $\hat{\rho}$ . In particular, it works for  $\hat{\rho} = 0$ , implying that the expansion component  $g_{(0)\mu\nu}$  is the metric on the boundary. Therefore, we can write

$$g^{\mu\nu} = g^{\mu\lambda}g^{\nu\sigma}g_{\lambda\sigma},\tag{F.1.17}$$

we—by inserting the metric expansion (F.1.12) and the ansatz (F.1.14)—get to lowest order in  $\hat{\rho}$ :

$$\mathcal{O}(\hat{\rho}^{0}):\tilde{g}_{(0)}^{\mu\nu} = \tilde{g}_{(0)}^{\mu\lambda}\tilde{g}_{(0)}^{\nu\sigma}g_{(0)\lambda\sigma'}$$
(F.1.18)

implying that  $\tilde{g}_{(0)}^{\nu\sigma}g_{(0)\lambda\sigma} = \delta_{\lambda}^{\nu}$ ; that is,  $\tilde{g}_{(0)}^{\mu\nu}$  is the inverse of  $g_{(0)\mu\nu}$  (as expected, since we argued that it works as proper metric), so we'll drop the tilde on this component. Similarly, to first order in  $\hat{\rho}$ , we get

$$\mathcal{O}(\hat{\rho}): \tilde{g}_{(2)}^{\mu\nu} = \tilde{g}_{(2)}^{\mu\lambda} \underbrace{\overline{g}_{(0)}^{\nu\sigma} g_{(0)\lambda\sigma}}_{(0)\lambda\sigma} + \tilde{g}_{(2)}^{\nu\sigma} \underbrace{\overline{g}_{(0)}^{\mu\lambda}}_{(0)\delta\sigma} + g_{(0)}^{\mu\lambda} g_{(0)}^{\nu\sigma} g_{(2)\lambda\sigma}^{(2)}, \tag{F.1.19}$$

so that

$$\tilde{g}_{(2)}^{\mu\nu} = -g_{(0)}^{\mu\lambda}g_{(2)\lambda\sigma}g_{(0)}^{\sigma\nu}, \tag{F.1.20}$$

and finally,

$$\mathcal{O}(\hat{\rho}^{2}):\tilde{g}_{(4)}^{\mu\nu} = 2\tilde{g}_{(4)}^{\mu\nu} + g_{(0)}^{\mu\lambda}g_{(4)\lambda\sigma}g_{(0)}^{\sigma\nu} + g_{(0)}^{\mu\lambda}g_{(0)}^{\nu\sigma}\log(\hat{\rho})h_{(4)\lambda\sigma} + \tilde{g}_{(2)}^{\mu\lambda}\tilde{g}_{(2)}^{\nu\sigma}g_{(0)\lambda\sigma} + g_{(0)}^{\mu\lambda}\tilde{g}_{(2)}^{\nu\sigma}g_{(2)\lambda\sigma} + \tilde{g}_{(2)}^{\mu\sigma}g_{(2)\lambda\sigma}^{\nu\sigma}g_{(2)\lambda\sigma} + \tilde{g}_{(2)}^{\mu\sigma}g_{(2)\lambda\sigma}^{\nu\sigma}g_{(2)\lambda\sigma} + \tilde{g}_{(2)}^{\mu\sigma}g_{(2)\lambda\sigma}^{\nu\sigma}g_{(2)\lambda\sigma} + \tilde{g}_{(2)}^{\mu\nu}g_{(2)\lambda\sigma}^{\nu\sigma}g_{(2)\lambda\sigma} + \tilde{g}_{(2)}^{\mu\nu}g_{(2)\lambda\sigma}^{\nu\sigma}g_{(2)\lambda\sigma}^{\nu\sigma}g_{(2)\lambda\sigma} + \tilde{g}_{(2)}^{\mu\nu}g_{(2)\lambda\sigma}^{\nu\sigma}g_{(2)\lambda\sigma}^{\nu\sigma}g_{(2)\lambda\sigma} + \tilde{g}_{(2)}^{\mu\nu}g_{(2)\lambda\sigma}^{\nu\sigma}g_{(2$$

which, upon insertion of the previous result for  $\tilde{g}_{(2)}^{\mu\nu}$ , gives:

$$\tilde{g}_{(4)}^{\mu\nu} = -g_{(0)}^{\mu\lambda}g_{(0)}^{\nu\sigma}\log(\hat{\rho})h_{(4)\lambda\sigma} - g_{(0)}^{\mu\lambda}g_{(4)\lambda\sigma}g_{(0)}^{\sigma\nu} - g_{(0)}^{\mu\alpha}g_{(2)\alpha\beta}g_{(0)}^{\beta\lambda}g_{(0)}^{\nu\gamma}g_{(0)}^{\beta\sigma}g_{(0)}^{\beta\sigma}g_{(0)\lambda\sigma}$$
(F.1.22)

$$+g_{(0)}^{\mu\lambda}g_{(0)}^{\nu\alpha}g_{(2)\alpha\beta}g_{(0)}^{\rho\nu}g_{(2)\lambda\sigma} + g_{(0)}^{\mu\alpha}g_{(2)\alpha\beta}g_{(0)}^{\rho\lambda}g_{(2)\lambda\sigma}$$
(F.1.23)

$$= -\log(\hat{\rho})g_{(0)}^{\mu\lambda}h_{(4)\lambda\sigma}g_{(0)}^{\sigma\nu} + g_{(0)}^{\mu\lambda}g_{(2)\lambda\sigma}g_{(0)}^{\sigma\beta}g_{(2)\beta\alpha}g_{(0)}^{\alpha\nu} - g_{(0)}^{\mu\lambda}g_{(4)\lambda\sigma}g_{(0)}^{\sigma\nu},$$
(F.1.24)

and so the inverse metric takes the full form:

$$g^{\mu\nu} = g^{\mu\nu}_{(0)} - \hat{\rho}g^{\mu\lambda}_{(0)}g_{(2)\lambda\sigma}g^{\sigma\nu}_{(0)} - \hat{\rho}^2\log(\hat{\rho})g^{\mu\lambda}_{(0)}h_{(4)\lambda\sigma}g^{\sigma\nu}_{(0)} + \hat{\rho}^2\left(g^{\mu\lambda}_{(0)}g_{(2)\lambda\sigma}g^{\sigma\beta}_{(0)}g_{(2)\beta\alpha}g^{\alpha\nu}_{(0)} - g^{\mu\lambda}_{(0)}g_{(4)\lambda\sigma}g^{\sigma\nu}_{(0)}\right) + \mathcal{O}(\hat{\rho}^4)$$
(F.1.25)

Using the trace-reversed form of Einstein's equations, the equations of motion take the form

$$R_{mn}^{(G)} = -4G_{mn}. (F.1.26)$$

We begin by computing the Christoffel symbols  $\Gamma_{mn}^{\ell}$ ; there are six classes of these distinguished by their index structure

$$\Gamma^{\lambda}_{\mu\nu}, \ \Gamma^{\hat{\rho}}_{\mu\nu}, \ \Gamma^{\lambda}_{\hat{\rho}\nu}, \ \Gamma^{\hat{\rho}}_{\hat{\rho}\nu}, \ \Gamma^{\lambda}_{\hat{\rho}\hat{\rho}}, \ \Gamma^{\hat{\rho}}_{\hat{\rho}\hat{\rho}}.$$
(F.1.27)

Now, since *G* is block diagonal,  $\Gamma^{\lambda}_{\mu\nu} = \Gamma^{(g)\lambda}_{\mu\nu}$ , where  $\Gamma^{(g)\lambda}_{\mu\nu}$  is given entirely in terms of *g*. Block diagonality also implies that  $\Gamma^{\hat{\rho}}_{\hat{\rho}\mu} = 0$ ,  $\Gamma^{\lambda}_{\hat{\rho}\hat{\rho}} = 0$ . The others evaluate to

$$\Gamma^{\hat{\rho}}_{\mu\nu} = 2g_{\mu\nu} - 2\hat{\rho}g'_{\mu\nu}, \quad \Gamma^{\lambda}_{\hat{\rho}\nu} = \frac{1}{2}g^{\lambda\sigma}\left(g'_{\sigma\nu} - \frac{1}{\hat{\rho}}g_{\mu\sigma}\right), \quad \Gamma^{\hat{\rho}}_{\hat{\rho}\hat{\rho}} = -\frac{1}{\hat{\rho}}, \quad (F.1.28)$$

where we have defined  $g'_{\mu\nu} =: \frac{\partial}{\partial \hat{\rho}} g_{\mu\nu}$ . The  $(\mu\nu)$ -component of the Ricci tensor (for *G*) then assumes the form

$$=\partial_{\hat{\rho}}\Gamma^{\hat{\rho}}_{\mu\nu} - \partial_{\nu}\widetilde{\Gamma^{\hat{\rho}}_{\mu\hat{\rho}}} + \Gamma^{\hat{\rho}}_{\mu\nu}\Gamma^{\ell}_{\ell\hat{\rho}} + \Gamma^{\sigma}_{\mu\nu}\widetilde{\Gamma^{\hat{\rho}}_{\hat{\rho}\sigma}} - \Gamma^{\hat{\rho}}_{\mu\ell}\Gamma^{\ell}_{\nu\hat{\rho}} - \Gamma^{\sigma}_{\mu\hat{\rho}}\Gamma^{\hat{\rho}}_{\nu\sigma} + R^{(g)}_{\mu\nu}$$
(F.1.30)

$$= g_{\mu\nu}g^{\sigma\lambda}g'_{\lambda\sigma} - 2g'_{\mu\nu} + \hat{\rho}\left(2g^{\lambda\sigma}g'_{\sigma\mu}g'_{\lambda\nu} - 2g''_{\mu\nu} - g'_{\mu\nu}g^{\sigma\lambda}g'_{\lambda\sigma}\right) + R^{(g)}_{\mu\nu} - 4\hat{\rho}^{-1}g_{\mu\nu}.$$
 (F.1.31)

This must be set equal to  $-4G_{\mu\nu}$ , which will just eat the last term, producing the equation

$$g_{\mu\nu}g^{\sigma\lambda}g'_{\lambda\sigma} + 2g'_{\mu\nu} + \hat{\rho}\left(2g^{\lambda\sigma}g'_{\sigma\mu}g'_{\lambda\nu} - 2g''_{\mu\nu} - g'_{\mu\nu}g^{\sigma\lambda}g'_{\lambda\sigma}\right) + R^{(g)}_{\mu\nu} = 0, \qquad (F.1.32)$$

There are two other equations corresponding to the  $R^{(G)}_{\hat{\rho}\hat{\rho}}$  and  $R^{(G)}_{\hat{\rho}\mu}$  components. They read

$$g^{\mu\nu}g''_{\mu\nu} - \frac{1}{2}g^{\mu\lambda}g'_{\lambda\sigma}g^{\sigma\nu}g'_{\nu\mu} = 0 \text{ (from } R^{(G)}_{\hat{\rho}\hat{\rho}}\text{),}$$
(F.1.33)

$$g^{\nu\lambda}\left(\nabla^{(g)}_{\mu}g'_{\nu\lambda} - \nabla^{(g)}_{\lambda}g'_{\mu\nu}\right) = 0 \text{ (from } R^{(G)}_{\hat{\rho}\mu}\text{).}$$
(F.1.34)

Considering the Fefferman-Graham expansion to be a perturbation in the small parameter (we're eventually interested in the limit  $\hat{\rho} \rightarrow 0$ )  $\hat{\rho}$  around  $g_{(0)\mu\nu}$ , all indices are raised and lowered with  $g_{(0)}$ . Plugging the Fefferman-Graham expansion into the equations of motion using the *Mathematica* package *xAct* [88] with the subroutine *xTras* [108], the coefficients work out to be

$$g_{(2)\mu\nu} = \frac{1}{12} R_{(0)} g_{(0)\mu\nu} - \frac{1}{2} R_{(0)\mu\nu}, \qquad (F.1.35)$$

where  $R_{(0)\mu\nu}$  is the Ricci tensor of the boundary metric  $g_{(0)\mu\nu}$ . The precise form of  $g_{(4)\mu\nu}$  is undetermined by this perturbative analysis, but we *can* say something about the trace and the divergence using the equation of motion (F.1.34); the full expressions can be found in [86]. The trace, however, will be important later—it is given by

$$g_{(0)}^{\mu\nu}g_{(4)\mu\nu} = g_{(4)\mu}^{\mu} = \frac{1}{4}g_{(2)\mu\nu}g_{(2)}^{\mu\nu}.$$
 (F.1.36)

The coefficient  $h_{(4)\mu\nu}$  satisfies

$$g_{(0)}^{\mu\nu}h_{(4)\mu\nu} = 0, \tag{F.1.37}$$

$$\nabla^{(0)\mu}h_{(4)\mu\nu} = 0. \tag{F.1.38}$$

The next step in the renormalization procedure is to compute the regularized on-shell action; imposing  $R^{(G)} = -20$  and  $K = \nabla_m n^m$  with  $n^\mu = n\delta^m_{\hat{\rho}}$ , so normality requires  $n = 2\hat{\rho}$ , so that  $\nabla_m n^m = \frac{2}{\sqrt{-G}}\partial_{\hat{\rho}}(\hat{\rho}\sqrt{-G})$ . Further, block diagonality of *G* implies that  $\sqrt{-G} = \frac{1}{2\hat{\rho}^3}\sqrt{-g}$ , so that the regularized Gibbons-Hawking term becomes

$$2\sqrt{-\gamma}K\big|_{\hat{\rho}=\epsilon} = 2\hat{\rho}^{-2}\sqrt{-g}K\Big|_{\hat{\rho}=\epsilon} = \sqrt{-g}\frac{4\hat{\rho}^{-2}}{\sqrt{-G}}\partial_{\hat{\rho}}\left(\hat{\rho}\sqrt{-G}\right)\Big|_{\hat{\rho}=\epsilon}$$
(F.1.39)

$$= 4\hat{\rho}\partial_{\hat{\rho}}\left(\frac{\sqrt{-g}}{\hat{\rho}^2}\right)\Big|_{\hat{\rho}=\epsilon}$$
(F.1.40)

$$=4\hat{\rho}^{-1}\partial_{\hat{\rho}}\left(\sqrt{-g}\right)\Big|_{\hat{\rho}=\epsilon}-4\hat{\rho}^{-2}\sqrt{-g}\Big|_{\hat{\rho}=\epsilon},$$
(F.1.41)

which produces the on-shell action

$$S_{\text{reg}}^{\text{on-shell}}[g,\epsilon] = \oint_{\partial \mathcal{M}_{\epsilon}} d^4 x \left[ -\int_{\infty}^{\epsilon} d\hat{\rho} \ 4\hat{\rho}^{-3}\sqrt{-g} + 4\hat{\rho}^{-1}\partial_{\hat{\rho}} \left(\sqrt{-g}\right) \Big|_{\hat{\rho}=\epsilon} - 8\hat{\rho}^{-2}\sqrt{-g} \Big|_{\hat{\rho}=\epsilon} \right].$$
(F.1.42)

In order to identify the divergent terms, we need to expand the determinant  $\sqrt{-g}$ , something we already did for the DBI action when "deriving" the AdS/CFT correspondence in section 2.3; replicating our method, we obtain

$$g_{\mu\nu} = g_{(0)\mu\lambda} \left[ \delta^{\lambda}_{\nu} + \hat{\rho} g^{\lambda}_{(2)\nu} + \hat{\rho}^2 g^{\lambda}_{(4)\nu} + \hat{\rho}^2 \log(\hat{\rho}) h^{\lambda}_{(4)\nu} \right],$$
(F.1.43)

implying that

$$\sqrt{-g} = \sqrt{-g_{(0)}} \sqrt{1 + \hat{\rho} g_{(2)\lambda}^{\lambda} + \hat{\rho}^2 \left[ g_{(4)\lambda}^{\lambda} + \frac{1}{2} \left\{ g_{(2)\lambda}^{\lambda} g_{(2)\nu}^{\nu} - g_{(2)\nu}^{\lambda} g_{(2)\lambda}^{\nu} \right\} \right] + \hat{\rho}^2 \log(\hat{\rho}) h_{(4)\lambda}^{\lambda} + \mathcal{O}(\hat{\rho}^4)}$$
(F.1.44)

$$= \sqrt{-g_{(0)}} \sqrt{1 + \hat{\rho}g_{(2)\lambda}^{\lambda} + \hat{\rho}^2 \left(\frac{1}{2}g_{(2)\lambda}^{\lambda}g_{(2)\nu}^{\nu} - \frac{1}{4}g_{(2)\lambda\nu}g_{(2)}^{\lambda\nu}\right)} + \mathcal{O}(\hat{\rho}^4)$$
(F.1.45)

$$= \sqrt{-g_{(0)}} \left( 1 + \frac{\hat{\rho}}{2} g_{(2)\lambda}^{\lambda} + \underbrace{\frac{\hat{\rho}^2}{8} \left[ g_{(2)\lambda}^{\lambda} g_{(2)\nu}^{\nu} - g_{(2)}^{\lambda\nu} g_{(2)\lambda\nu} \right]}_{(*)} \right) + \mathcal{O}(\hat{\rho}^4)$$
(F.1.46)

where we have used the relations (F.1.37) and (F.1.36) for  $h_{(4)\mu\nu}$  as well as the series expansion  $\sqrt{1+x} = 1 + x/2 - x^2/8 + O(x^3)$ . The integration over  $\hat{\rho}$  can now be performed, which leads to the structure of the generic regularized action (F.1.3),

$$S_{\text{reg}}^{\text{on-shell}}[g_{(0)},\epsilon] = \int_{\partial \mathcal{M}_{\epsilon}} d^4x \,\sqrt{g_{(0)}} \left[\epsilon^{-2}a_{(0)} + \epsilon^{-1}a_{(2)} - \log(\epsilon) \,a_{(4)} + \mathcal{O}(\epsilon^0)\right].$$
(F.1.47)

In particular, the  $log(\epsilon)$  contribution comes from the term (\*) when plugged into the on-shell action (F.1.42), from which we infer that

$$a_{(4)} = \frac{1}{2} \left( g_{(2)\lambda}^{\lambda} g_{(2)\nu}^{\nu} - g_{(2)}^{\lambda\nu} g_{(2)\lambda\nu} \right).$$
(F.1.48)

Performing the remaining integrations and derivatives and collecting orders of  $\epsilon$ , we find that

$$a_{(0)} = -6, \ a_{(2)} = 0.$$
 (F.1.49)

This is in agreement with [86]. Next, we determine the counterterm action, which, in order to ensure covariance, requires us to express quantities at the boundary in terms of quantities on the regularized hypersurface  $\hat{\rho} = \epsilon$ , which has metric  $\gamma_{\mu\nu} = \epsilon^{-1}g_{\mu\nu}(x,\epsilon)$ . Inverting the expression for  $\sqrt{-g}$  (F.1.46), we find that

$$\sqrt{-g_{(0)}} = \sqrt{-g} \left( 1 - \frac{\epsilon}{2} g_{(2)\lambda}^{\lambda} + \frac{\epsilon^2}{8} \left( g_{(2)\lambda}^{\lambda} g_{(2)\nu}^{\nu} + g_{(2)\nu}^{\lambda} g_{(2)\lambda}^{\nu} \right) \right) + \mathcal{O}(\epsilon^4)$$
(F.1.50)

$$=\sqrt{-\gamma}\epsilon^{2}\left(1-\frac{\epsilon}{2}g_{(2)\lambda}^{\lambda}+\frac{\epsilon^{2}}{8}\left(g_{(2)\lambda}^{\lambda}g_{(2)\nu}^{\nu}+g_{(2)\nu}^{\lambda}g_{(2)\lambda}^{\nu}\right)\right)+\mathcal{O}(\epsilon^{4}).$$
 (F.1.51)

Furthermore, taking the trace of (F.1.35) gives us the relation

$$g_{(2)\lambda}^{\lambda} = -\frac{1}{6}R_{(0)},$$
 (F.1.52)

so we now need to express  $R_{(0)}$  in terms of  $R^{(\gamma)}$ . We can now do a series expansion in  $\hat{\rho}$ , and contract the resulting expression using the inverse hypersurface metric (F.1.25), in which case we find that

$$g_{(2)\lambda}^{\lambda} = \frac{1}{6} \frac{1}{\epsilon} \left( R_{(\gamma)} + \frac{1}{2} \left[ R_{\mu\nu}^{(\gamma)} R^{(\gamma)\mu\nu} - \frac{1}{6} \left( R^{(\gamma)} \right)^2 \right] \right) + \mathcal{O}\left( \left( R^{(\gamma)} \right)^3 \right), \tag{F.1.53}$$

where terms cubic in curvatures do not contribute<sup>4</sup>. A similar analysis reveals that

$$g_{(2)\nu}^{\lambda}g_{(2)\lambda}^{\nu} = \frac{1}{\epsilon^2} \frac{1}{4} \left( R_{\mu\nu}^{(\gamma)} R^{(\gamma)\mu\nu} - \frac{2}{9} \left( R^{(\gamma)} \right)^2 \right) + \mathcal{O}\left( \left( R^{(\gamma)} \right)^3 \right).$$
(F.1.54)

Using this, the counterterm action assumes the form

$$S_{\rm ct}[g(x,\epsilon);\epsilon] = -\int_{\partial\mathcal{M}_{\epsilon}} \mathrm{d}^4 x \,\sqrt{-\gamma} \left[ -6 + \frac{1}{2}R^{(\gamma)} - \log(\epsilon) \left( \frac{R^{(\gamma)^2}}{12} - \frac{R^{(\gamma)\mu\nu}R^{(\gamma)}_{\mu\nu}}{4} \right) \right]. \quad (F.1.55)$$

This allows us to immediately compute various quantities such as one-point functions (VEVs) and Ward identities (see e.g. [95] for details on how to do this in a FG framework).

<sup>4</sup> This statement is valid in dimensions below six.

### F.2 THE DBVV ANSATZ APPROACH

Developed in [49] by de Boer and the Verlinde twins (hence the name, *dBVV*) and reviewed in e.g. [89, 207, 208], this method is the precursor of the approach discussed in chapter **3**. It also makes use of the fact that the radial direction of AlAdS normal to the boundary allows us to foliate the space into slices of constant radius (at least near the boundary), which can be used to set up a radial analogue of the ADM formalism [98, 209]. This method can be extended to Lifshitz space-times: starting from a Einstein-Proca model, this was pursued in [25]. The drawback of this method, however, is the need to specify a counterterm ansatz, and it is generically very difficult to specify a sufficiently general ansatz.

#### F.2.1 Setting up the ADM Hamiltonian and the Hamilton-Jacobi Equation

For AlAdS manifolds (see appendix A, in particular (F.2.14)), we use the radial coordinate r, which is normal to the boundary, to foliate the manifold into radial hypersurfaces (leaves)  $\Sigma_r$  defined by r = const.; these slices are diffeomorphic to the conformal boundary. Disregarding pathological topological complications, this is always possible (at least in the neighbourhood of the conformal boundary). The normal vector to  $\Sigma_r$ , which points in the direction of increasing r, can be written as  $n^m = n\delta_r^m$ , where the normalization is fixed by requiring  $g_{mn}n^mn^n = 1$ , allowing us to define a hypersurface projector  $h_{mn} = g_{mn} - n_m n_n$  projecting onto  $\Sigma_r$ . Restricting the indices on the projector also gives us the induced metric, we may decompose the (Euclidean) metric as

$$ds^{2} = g_{mn} dx^{m} dx^{n} = N^{2} dr^{2} + h_{\mu\nu} \left( dx^{\mu} + N^{\mu} dr \right) \left( dx^{\nu} + N^{\nu} dr \right),$$
(F.2.1)

where *N* is the shift and  $N^{\mu}$  the lapse. The (Euclidean) Einstein-Hilbert action on a (d + 1)-dimensional manifold  $\mathcal{M}$  (with AlAdS boundary condition) and boundary  $\Sigma_r = \partial \mathcal{M}|_r$  is given by,

$$S[g,h] = \int_{\mathcal{M}} \mathrm{d}^{d+1}x \,\sqrt{g} \left( R^{(g)} - 2\Lambda \right) - 2 \int_{\Sigma_r} \mathrm{d}^d x \,\sqrt{h}K,\tag{F.2.2}$$

where K is the trace of the extrinsic curvature on  $\Sigma_r$ . The ADM-parametrized version of this reads,

$$S[h,N] = \int_{r}^{\infty} dr' \int d^{d}x \,\sqrt{h}N\left(R^{(h)} - 2\Lambda + K_{\mu\nu}K^{\mu\nu} - K^{2}\right),\tag{F.2.3}$$

where the Gibbons-Hawking cancels out by partial integration of second derivative terms in the expansion of the Ricci scalar. Classically, the action above is equivalent to the following action,

$$S[\pi^{\mu\nu}, N, N^{\mu}] = \int \mathrm{d}r \int \mathrm{d}^{d}x \,\sqrt{h} \left[\pi^{\mu\nu}\partial_{r}h_{\mu\nu} + N\mathcal{H} - N^{\mu}\mathcal{H}_{\mu}\right],\tag{F.2.4}$$

where  $\sqrt{h}\pi^{\mu\nu}$  is the canonical momentum conjugate to the induced metric  $h_{\mu\nu}$  on  $\Sigma_r$ , whereas

$$\mathcal{H} = R^{(h)} - 2\Lambda + \pi^{\mu\nu}\pi_{\mu\nu} - \frac{1}{d-1}\pi^2, \quad \mathcal{H}^{\mu} = 2D_{\nu}\pi^{\mu\nu}, \quad (F.2.5)$$

where  $D_{\nu}$  is the covariant derivative on the hypersurface  $\Sigma_r$ . On a general hypersurface  $\Sigma$ , such a derivative is defined via  $D_m T^{n...}_{\ell...} = h^n_k \cdots h^j_\ell \cdots h^i_m \nabla_i T^{k...}_{j...}$  for  $T^{k...}_{j...} \in \Sigma$  and satisfies, in particular, metric compatibility with the hypersurface projector,  $D_m h_{n\ell} = 0$ . Upon noting that the Wick rotated Legendre transformation changes sign—schematically  $H = \sum_i p_i \dot{q}_i + L$ —we may construct the radial Hamiltonian on an equal-*r* slice  $\Sigma_r$  as follows,

$$H = \int_{\Sigma_r} d^d x \,\sqrt{h} \left( N \mathcal{H} + N^{\mu} \mathcal{H}_{\mu} \right), \tag{F.2.6}$$

The radial Hamilton-Jacobi equation then takes the form

$$\partial_r S + H = 0, \quad \pi^{\mu\nu} = \frac{1}{\sqrt{h}} \frac{\delta S}{\delta h_{\mu\nu}},$$
 (F.2.7)

where *S* is the on-shell action—which is known as Hamilton's principal function in analytical mechanics (see [91])—where it is understood that the value of  $\pi^{\mu\nu}$  given above is inserted into the HJ equation. Since *S* cannot depend on *r* (see [207]), the HJ equation reduces to

$$H = 0,$$
 (F.2.8)

implying the Hamiltonian constraint and the momentum constraint

$$\mathcal{H} = 0, \quad \mathcal{H}_{\mu} = 0. \tag{F.2.9}$$

Alternatively, these are the equations of motion obtained from (F.2.4) by varying with respect the Lagrange multiplicators<sup>5</sup> N and  $N^{\mu}$ —a point of view favoured in [51]. Note that the momentum  $\pi^{\mu\nu}$  is directly related to the Brown-York stress tensor (see [210]) on  $\Sigma_r$ ,

$$\hat{T}^{\mu\nu} = \pi^{\mu\nu}.\tag{F.2.10}$$

## F.2.2 Counterterm Action Ansatz

As in the FG approach, we write the renormalized on-shell action as  $S_{ren} = S + S_{ct}$ , where the counterterm action is specified by the following locally covariant ansatz<sup>6</sup>

$$S_{\rm ct} = \int_{\Sigma_r} \mathrm{d}^d x \; \sqrt{h} \Big[ c_0 + c_1 R^{(h)} + c_2^{(1)} R^{(h)^2} + c_2^{(2)} R^{(h)}_{\mu\nu} R^{(h)\mu\nu} \tag{F.2.11}$$

$$+ c_2^{(3)} R^{(h)}_{\mu\nu\rho\sigma} R^{(h)\mu\nu\rho\sigma} + \mathcal{O}(R^3) \Big],$$
 (F.2.12)

which, by (F.2.7), implies that the momentum splits up as  $\pi^{\mu\nu} = \pi^{\mu\nu}_{ren} - \pi^{\mu\nu}_{ct}$ , which is useful since covariance of our ansatz means that the momentum constraint associated with the counterterm vanishes identically by covariant conservation of the associated Brown-York (quasi.local) stress tensor,  $D_{\mu}\pi^{\mu\nu}_{ct} = D_{\mu}\hat{T}^{\mu\nu}_{ct}$ . Thus, the momentum constraint reduces to the *d*-dimensional diffeomorphism Ward identity on  $\Sigma_r$ ,

$$0 = \mathcal{H}^{\mu} = D_{\mu} \hat{T}^{\mu\nu}_{\rm ren}.$$
 (F.2.13)

Employing domain-wall coordinates<sup>7</sup>, where the pure AdS metric reads  $ds^2 = dr^2 + e^{2r} (\eta_{\mu\nu} dx^{\mu} dx^{\nu})$ , we see that the AlAdS boundary condition on the induced metric reads

$$\partial_r h_{\mu\nu} = 2h_{\mu\nu} + \mathcal{O}\left(e^{-r}\right). \tag{F.2.14}$$

Now, following [49], we use the diffeomorphism invariance to gauge fix N = 1 and  $N^{\mu} = 0$ ; when this is done, the imposition of the asymptotic behaviour (F.2.14) is achieved via the Hamilton equation,

$$\partial_r h_{\mu\nu} = \frac{1}{\sqrt{g}} \frac{\delta H}{\delta \pi^{\mu\nu}} = 2\pi_{\mu\nu} - \frac{2}{d-1}\pi h_{\mu\nu}.$$
 (F.2.15)

Now, want to impose (F.2.14), which is, of course, an asymptotic condition. Near the boundary, therefore—by assumption—only the counterterm action gives a contribution to  $\pi_{\mu\nu}$ ; in particular, by varying (F.2.12), we obtain:

$$\pi_{\mu\nu} = -\frac{1}{\sqrt{h}} \frac{\delta S_{\text{ren}}}{\delta h^{\mu\nu}} + \mathcal{O}\left(e^{-r}\right)$$
(F.2.16)

$$=\frac{1}{2}h_{\mu\nu}c_{0}+\mathcal{O}\left(e^{-r}\right),$$
(F.2.17)

where none of the curvatures contribute, that is, we effectively only vary  $\int d^d x \sqrt{h}c_0$ . This means that

$$\partial_r h_{\mu\nu} = h_{\mu\nu} c_0 - \frac{c_0}{d-1} h_{\lambda\rho} h^{\lambda\rho} h_{\mu\nu} = h_{\mu\nu} c_0 \left(\frac{d-1-d}{d-1}\right) = -\frac{c_0}{d-1} h_{\mu\nu}.$$
 (F.2.18)

The asymptotic boundary condition of (F.2.14) thus requires that:

$$c_0 = -2(d-1).$$
 (F.2.19)

Let's just check the variation of the next term,  $\int d^d x \sqrt{hR}$ , which becomes

$$\delta\left(\sqrt{h}R^{(h)}\right) = -\frac{1}{2}\sqrt{h}h_{\mu\nu}\delta h^{\mu\nu}R^{(h)} + R^{(h)}_{\mu\nu}\delta h^{\mu\nu}\sqrt{g},$$
 (F.2.20)

implying that the contribution to  $\pi^{\mu\nu}$  takes the form

$$\pi_{\mu\nu}^{(R)} = -\frac{1}{2} R^{(h)} h_{\mu\nu} + R^{(h)}_{\mu\nu} = (R + \mathcal{K}) h_{\mu\nu} \sim e^{-2r} h_{\mu\nu}, \qquad (F.2.21)$$

where  $\mathcal{K}$  is a constant, so this will not contribute. The same goes for the higher order curvatures.

<sup>5</sup> That is, non-dynamical (no kinetic terms).

<sup>6</sup> This expression becomes more involved if we were to include more fields in our model.

<sup>7</sup> The boundary in domain-wall coordinates is located at  $r \to \infty$  since it is obtained from the usual Poincaré patch metric by  $r \to e^{-r}$ ; in Poincaré coordinates, the boundary is located at r = 0.

# F.2.3 The DeWitt Bracket

Having fixed the asymptotic boundary condition, which determined  $c_0$ , we're now ready to solve the Hamiltonian constraint,  $\mathcal{H} = 0$ . In order to accomplish this, it is convenient to introduce the following piece of notation—which we shall call the *DeWitt bracket* which is symmetric and bilinear—for the kinetic part of  $\mathcal{H}$ ,

$$\{S,S\}_{\rm DW} := G_{\mu\nu\rho\sigma} \frac{\delta S}{\delta h_{\mu\nu}} \frac{\delta S}{\delta h_{\rho\sigma}},\tag{F.2.22}$$

where  $G_{\mu\nu\rho\sigma}$  is the DeWitt metric, given by

$$G_{\mu\nu\rho\sigma} = \frac{1}{h} \left( h_{\mu(\rho} h_{\sigma)\nu} - \frac{1}{d-1} h_{\mu\nu} h_{\rho\sigma} \right).$$
(F.2.23)

In terms of the DeWitt bracket, the Hamiltonian constraint in our gauge takes the form

$$\mathcal{H} = \{S, S\}_{\text{DW}} + R^{(h)} - 2\Lambda.$$
(F.2.24)

Bilinearity and symmetry of the DeWitt bracket implies that upon writing  $S = S_{ren} + S_{ct}$ , the Hamiltonian constraint becomes

$$\mathcal{H} = \{S_{\text{ren}}, S_{\text{ren}}\}_{\text{DW}} + \{S_{\text{ct}}, S_{\text{ct}}\}_{\text{DW}} + 2\{S_{\text{ct}}, S_{\text{ren}}\}_{\text{DW}} + R^{(h)} - 2\Lambda = 0.$$
(F.2.25)

The next step is to sort the Hamiltonian  $\mathcal{H}$  by the number of radial derivatives *n*; schematically

$$\mathcal{H} = \sum_{n \ge 0} \mathcal{H}^{(2n)}.$$
 (F.2.26)

As before, this means that each  $\mathcal{H}^{(n)}$  vanishes in isolation. An analogous derivative expansion for  $\sqrt{g}\mathcal{L}_{ct}$  gives us that, in domain-wall coordinates, the asymptotic radial scaling for AlAdS geometries depends on the number of derivatives *n* as

$$\sqrt{g}\mathcal{L}_{\mathrm{ct}}^{(n)} \sim e^{(d-n)r}.$$
 (F.2.27)

As before—by assumption—the renormalized action  $S_{ren}$  has a finite limit as  $r \to \infty$ ; thus the relevant DeWitt brackets scale asymptotically as

$$\left\{S_{\rm ct}^{(m)}, S_{\rm ct}^{(n)}\right\}_{\rm DW} \sim e^{-(m+n)r},$$
 (F.2.28)

$$\left\{S_{\rm ct}^{(n)}, S_{\rm ren}\right\}_{\rm DW} \sim e^{-(d+n)r},$$
 (F.2.29)

$$\{S_{\rm ren}, S_{\rm ren}\}_{\rm DW} \sim e^{-2dr}.$$
 (F.2.30)

The lowest-order Hamiltonian constraint, therefore, takes the form

$$0 = \mathcal{H}^{(0)} = \left\{ S_{ct}^{(0)}, S_{ct}^{(0)} \right\}_{DW} - 2\Lambda.$$
(F.2.31)

In particular, we find that

$$\left\{S_{\rm ct}^{(0)}, S_{\rm ct}^{(0)}\right\}_{\rm DW} = \frac{c_0^2}{h} \left(h_{\mu(\rho} h_{\sigma)\nu} - \frac{1}{d-1} h_{\mu\nu} h_{\rho\sigma}\right) \frac{1}{4} h h^{\mu\nu} h^{\rho\sigma}$$
(F.2.32)

$$= \frac{c_0^2}{4} \left( h_{\mu(\rho} h_{\sigma)\nu} - \frac{1}{d-1} h_{\mu\nu} h_{\rho\sigma} \right) h^{\mu\nu} h^{\rho\sigma}$$
(F.2.33)

$$= \frac{c_0^2}{4} \left( \frac{1}{2} \left( h_{\mu\rho} h_{\sigma\nu} + h_{\mu\sigma} h_{\rho\nu} \right) h^{\mu\nu} h^{\rho\sigma} - \frac{d^2}{d-1} \right)$$
(F.2.34)

$$=\frac{c_0^2}{4}\left(d-\frac{d^2}{d-1}\right)$$
(F.2.35)

$$=\frac{c_0^2}{4}\left(\frac{d^2-d}{d-1}-\frac{d^2}{d-1}\right)$$
(F.2.36)

$$= -\frac{c_0^2 d}{4(d-1)}.$$
 (F.2.37)

This should be equal to

$$2\Lambda = -d(d-1),$$
 (F.2.38)

implying that

$$c_0 = \pm 2(d-1),$$
 (F.2.39)

which agrees with our previous boundary condition analysis, provided we choose the negative solution. It is amusing to note that this only fixes the square of  $c_0$ ; we need the boundary conditions to fix the sign. The next Hamiltonian constraint (assuming d > 2—if we did not assume this, there'd be additional contributions from  $\{S_{ct}^{(0)}, S_{ren}\}_{DW}$ ; we'll have more to say about this later when discussing the holographic Weyl anomaly), which has two derivatives, consequently reads

$$0 = \mathcal{H}^{(2)} = 2 \left\{ S_{\rm ct}^{(0)}, S_{\rm ct}^{(2)} \right\}_{\rm DW} + R^{(h)}, \tag{F.2.40}$$

where<sup>8</sup>

$$2\left\{S_{ct}^{(0)}, S_{ct}^{(2)}\right\}_{DW} = -\frac{c_0 c_1 (d-2) R^{(h)}}{2(d-1)} = c_1 (d-2) R^{(h)},$$
(F.2.41)

implying that

$$c_1 = \frac{1}{d-2}.$$
 (F.2.42)

The next constraint reads:

$$0 = \mathcal{H}^{(4)} = 2 \left\{ S_{ct}^{(0)}, S_{ct}^{(4)} \right\}_{DW} + \left\{ S_{ct}^{(2)}, S_{ct}^{(2)} \right\}_{DW}.$$
 (F.2.43)

The first of these brackets work out to be,

$$2\left\{S_{ct}^{(0)}, S_{ct}^{(4)}\right\}_{DW} = c_0 \frac{(4-d)}{2(d-1)} \left(c_2 R^{(h)^2} + c_3 R_{\mu\nu}^{(h)} R^{(h)\mu\nu} + c_4 R_{\mu\nu\rho\sigma}^{(h)} R^{(h)\mu\nu\rho\sigma}\right)$$
(F.2.44)  
+  $\frac{c_0}{d-1} \left(\left[2(1+d)c_2 + c_3\right] \nabla_{\mu}^{(h)} \nabla^{(h)\mu} R^{(h)} + \left[4c_4 + c_3(d-2)\right] \nabla_{\mu}^{(h)} \nabla_{\nu}^{(h)} R^{(h)\mu\nu}\right),$ (F.2.45)

while the second takes the form:

$$\left\{S_{\rm ct}^{(2)}, S_{\rm ct}^{(2)}\right\}_{\rm DW} = c_1^2 \left(R_{\mu\nu}^{(h)} R^{(h)\mu\nu} - \frac{dR^{(h)^2}}{4(d-1)}\right).$$
(F.2.46)

From these, we immediately infer the equations

$$c_2 = \frac{d}{4(d-4)(d-1)(d-2)^2}, \ c_3 = -\frac{1}{(d-4)(d-2)^2}, \ c_4 = 0.$$
 (F.2.47)

Setting d = 4, we reassuringly obtain the result we also got the FG approach to holographic renormalization. It is worth noting, however, that the above is *not* the whole story. We have ignored certain additional terms involving covariant derivatives (produced by *xAct*), which do not cancel, suggesting that the ansatz employed is not sufficiently general. Curiously, the authors of [207] find similar terms, but claim that they cancel precisely.

### F.2.4 Holographic Weyl Anomalies

In the Hamilton-Jacobi approach to holographic renormalization, a *d*-dimensional Weyl anomaly<sup>10</sup> appears when  $\left\{S_{ct}^{(0)}, S_{ct}^{(d)}\right\}_{DW}$  vanishes identically, which is the case when *d* is even. However, since

<sup>8</sup> These variations are now somewhat tedious to perform, but they can be evaluated efficiently in *xAct*, which turns out to be admirably suited for this.

<sup>10</sup> For a cool cohomological approach to Weyl anomalies, see [211, 212].

 $\{S_{ct}^{(0)}, S_{ren}\}_{DW} \sim e^{-dr}$ , the Hamiltonian constraint  $\mathcal{H}^{(d)} = 0$  is non-trivial, and, in particular, involves the Weyl anomaly via

$$-2\left\{S_{\rm ct}^{(0)}, S_{\rm ct}^{(d)}\right\}_{\rm DW} = \frac{2}{\sqrt{h}} \frac{\delta S_{\rm ren}}{\delta h_{\mu\nu}} h_{\mu\nu} = h_{\mu\nu} T_{\rm ren}^{\mu\nu}.$$
 (F.2.48)

In two and four dimensions, the Hamiltonian constraints read

$$0 = \mathcal{H}^{(2)} = R - 2 \left\{ S_{\rm ct}^{(0)}, S_{\rm ren} \right\}_{\rm DW} \qquad (d = 2) \tag{F.2.49}$$

$$0 = \mathcal{H}^{(4)} = \left\{ S_{ct}^{(2)}, S_{ct}^{(2)} \right\}_{DW} - 2 \left\{ S_{ct}^{(0)}, S_{ren} \right\}_{DW} \qquad (d = 4),$$
(F.2.50)

which gives us, straightforwardly, the Weyl anomaly in the dimensions considered above,

$$h_{\mu\nu}T_{\rm ren}^{\mu\nu} = -R^{(h)}$$
 (f.2.51) (F.2.51)

$$h_{\mu\nu}T_{\rm ren}^{\mu\nu} = -\frac{1}{4} \left( R_{\mu\nu}^{(h)} R^{(h)\mu\nu} - \frac{R^{(h)^2}}{3} \right) \qquad (d=4).$$
(F.2.52)

(F.2.53)

# G.1 HOLOGRAPHIC RENORMALIZATION OF EINSTEIN-PROCA THEORY WITH LIFSHITZ BOUND-ARY CONDITIONS

It is the purpose of this appendix to describe the process of holographic renormalization applied to Lifshitz space-times arising from Einstein-Proca models. The notation differs slightly from the one used in the main text (cf. chapter 3); we hope this will not cause any confusion.

## G.1.1 The Einstein-Proca ADM Hamiltonian

The Lifshitz space-time arises as a solution to the Einstein-Proca action with a Gibbons-Hawking term,

$$S = \int d^{d+1}x \,\sqrt{g} \left(\hat{R} - 2\Lambda - \frac{1}{4}\hat{F}^2 - \frac{1}{2}M^2\hat{A}^2\right) - 2\int_{\Sigma_r} d^d x \,\sqrt{h}K,\tag{G.1.1}$$

where we have used hats to distinguish (d + 1)-dimensional quantities from *d*-dimensional ones. In domain-wall coordinates, the pure Lifshitz solution has the form

$$ds^{2} = dr'^{2} - e^{2zr'}dt^{2} + e^{2r'}d\vec{x}^{2}.$$
 (G.1.2)

Since the FG theorem does not apply to asymptotically locally Lifshitz, we must use either (i) the dBVV-deWitt method, which was done in [25], or (ii) the HJ approach described in chapter 3—this was done in [27, 96]. It is to this second approach that we will now turn our attention.

First, we sketch the derivation of the Einstein-Proca ADM Hamiltonian, starting from (G.1.1). First, we'll need to set up a codimension-1 foliation of  $\mathcal{M} = \text{AlLif}$  in terms of radial hypersurfaces (as we pointed out when renormalizing AlAdS, it is sufficient to assume this can be done in a neighbourhood of the boundary). In general [213, 214], a non-zero one-form  $\omega$  on  $\mathcal{M}$  generates a codimension-1 foliation of  $\mathcal{M}$  if and only if it is integrable, that is, it satisfies the Frobenius condition,

$$\omega \wedge \mathrm{d}\omega = 0. \tag{G.1.3}$$

which, in particular is satisfied by vectors *n* normal to some hyperplane,  $n \wedge dn = 0$ . In our case, the vector normal to  $\Sigma_r$ —which is defined by r = const.—is<sup>1</sup>  $n^{\hat{\mu}} = \frac{\hat{\nabla}^{\hat{\mu}}r}{|\hat{\nabla}^{\hat{\nu}}r\hat{\nabla}_{\hat{\nu}}r|^{1/2}}$ , and the foliation one-form constructed from the normal vector is

$$n_{\hat{\mu}} = g_{\hat{\mu}\hat{\nu}} n^{\hat{\nu}} = N \hat{\nabla}_{\hat{\mu}} r, \qquad (G.1.4)$$

where the normalization *N* is the lapse function. The hypersurface projector, which pulls back tensors on  $\mathcal{M}$  onto  $\Sigma_r$ , is given by,

$$h_{\hat{\mu}\hat{\nu}} = g_{\hat{\mu}\hat{\nu}} - n_{\hat{\mu}}n_{\hat{\nu}}, \tag{G.1.5}$$

whereas  $n_{\hat{\mu}}n_{\hat{\nu}}$  projects onto  $\Sigma_r^{\perp}$ . The Proca field is thus decomposable in the following manner,

$$\hat{A}_{\hat{\mu}} = \Phi n_{\hat{\mu}} + A_{\hat{\mu}}, \quad \Phi = n^{\hat{\nu}} \hat{A}_{\hat{\nu}}, \quad A_{\hat{\mu}} = h_{\hat{\mu}}{}^{\hat{\nu}} \hat{A}_{\hat{\nu}}.$$
 (G.1.6)

Next, we introduce the flow vector  $r^{\hat{\mu}}$  defined implicitly via  $r^{\hat{\mu}}n_{\hat{\mu}} = N$ . The hypersurface tangential part of the flow vector is called the shift function,  $N^{\hat{\mu}} = h^{\hat{\mu}}{}_{\hat{\nu}}r^{\hat{\nu}}$ , which means that the flow vector is decomposable as

$$r^{\hat{\mu}} = Nn^{\hat{\mu}} + N^{\hat{\mu}}.$$
 (G.1.7)

Eventually, we'll want to express the action (G.1.1) in terms of canonical velocities. We define these as the Lie derivative along  $r^{\hat{\mu}}$ ,

$$\dot{h}_{\hat{\mu}\hat{\nu}} = \pounds_r h_{\hat{\mu}\hat{\nu}}, \qquad \dot{A}_{\hat{\mu}} = \pounds_r A_{\hat{\mu}}. \tag{G.1.8}$$

<sup>1</sup> In principle, one has to be careful about the sign, but  $\Sigma_r$  is time-like, i.e.  $n_m n^m = +1$ .

If  $r^{\hat{\mu}} = \delta_r^{\hat{\mu}}$ , as will be the case eventually when imposing radial gauge on the shift and lapse, the Lie derivative reduces to the radial derivative. The extrinsic curvature  $K_{\hat{\mu}\hat{\nu}}$  of  $\Sigma_r$  is given by

$$K_{\hat{\mu}\hat{\nu}} = \frac{1}{2} \pounds_n h_{\hat{\mu}\hat{\nu}} = h_{\hat{\mu}}{}^{\hat{\lambda}} \hat{\nabla}_{\hat{\lambda}} n_{\hat{\nu}}. \tag{G.1.9}$$

Carrying on these calculations (see Appendix A of [25]) to the end and imposing the Gauss-Codazzi equations, we get the familiar expression

$$H = \int_{\Sigma_r} \mathrm{d}^d x \,\sqrt{h} \left( N \mathcal{H} + N^{\mu} \mathcal{H}_{\mu} \right). \tag{G.1.10}$$

Diffeomorphism invariance allows us to gauge fix  $\mathcal{N} = 1$ ,  $\mathcal{N}^{\mu} = 0$ , leading to the result

$$H = \int_{\Sigma_r} \mathrm{d}^d x \,\sqrt{h} \left[ \left( K_{\mu\nu} \pi^{\mu\nu} + \frac{1}{2} Q_\mu Q^\mu \right) + R - 2\Lambda - \frac{1}{4} F_{\mu\nu} F^{\mu\nu} + \frac{M^2}{2} \left( \Phi^2 - A_\mu A^\mu \right) \right], \,\,(\text{G.1.11})$$

where  $\Phi$  is the radial component of the Proca one-form,  $\hat{A} = \Phi dr + A_{\mu} dx^{\mu}$  and is constrained by the Proca constraint,

$$\Phi = -\frac{1}{M^2} D_\mu Q^\mu. \tag{G.1.12}$$

Note that in domain-wall coordinates, the Proca field supporting the vanilla Lifshitz solution has the form [22]

$$\hat{A} = \alpha e^{zr} \mathrm{d}t,\tag{G.1.13}$$

for some constant  $\alpha$ , which depends on both d and z. When considering the Proca field on a radial hypersurface  $\Sigma_r$ , the hypersurface tangential part of has the form  $A = \tilde{\alpha} dt$ . Restricting our attention to timelike Proca fields, we introduce an auxiliary field  $\psi$ , which measures the deformation away from the pure Lifshitz Proca field value, that is,

$$A = (\tilde{\alpha} + \psi) \,\mathrm{d}t. \tag{G.1.14}$$

 $\psi$  corresponds to the source of a relevant operator for  $1 < z < d_s$ ; for  $d_s = z$ , it becomes marginally relevant [215]. The canonical momenta are given by the usual relations:

$$\pi^{\mu\nu} = \frac{1}{\sqrt{h}} \frac{\delta S}{\delta h_{\mu\nu}}, \quad Q^{\mu} = \frac{1}{\sqrt{h}} \frac{\delta S}{\delta A_{\mu}}.$$
 (G.1.15)

Note that any AlLif metric may be written as (in domain-wall coordinates)

$$ds^{2} = dr^{2} + h_{\mu\nu}(r, x)dx^{\mu}dx^{\nu}, \qquad (G.1.16)$$

which, in order to set up the ADM Hamiltonian (G.1.11), was decomposed in a radial ADM-manner, i.e. by writing

$$ds^{2} = \mathcal{N}^{2} dr^{2} + h_{\mu\nu}(r, x) \left( dx^{\mu} + \mathcal{N}^{\mu} dr \right) \left( dx^{\nu} + \mathcal{N}^{\nu} dr \right), \tag{G.1.17}$$

where  $\mathcal{N}$  is the radial lapse function, whereas  $\mathcal{N}^{\mu}$  is the radial shift. We shall as in the case of AlAdS, we will gauge-fix—by means of foliation preserving diffeomorphisms—the shift and set the lapse equal to one. On the leaves  $\Sigma_r$  of the radial foliation, we introduce a temporal foliation, allowing us to write the induced metric on  $\Sigma_r$  as

$$h_{\mu\nu}(r,x)dx^{\mu}dx^{\nu} = -N^{2}dt^{2} + \gamma_{ij}\left(dx^{i} + N^{i}dt\right)\left(dx^{j} + N^{j}dt\right),$$
(G.1.18)

where *N* is the temporal lapse on  $\Sigma_r$  and  $N^i$  the temporal shift. We use indices from the middle of the Latin alphabet, *i*, *j*, *k*, *l*, *m*, *n* to denote spatial directions on  $\Sigma_r$ . The validity of such a foliation in terms of leaves of constant (absolute) time ties in nicely with our non-relativistic interpretation of the boundary theory. The gravity field multiplet then consists of the fields,

$$N, N', \gamma_{ij},$$
 (G.1.19)

More information about this double-foliation scheme may be found in [96]. Note that foliation preserving diffeomorphisms on  $\Sigma_r$  allows us to set  $N^i = 0$ , while we choose to keep N general. This, then, allows us to introduce the following vielbeine, where we use lower-case Latin letters from the beginning of the alphabet to denote flat indices, i.e. *a*, *b*, *c*, *d*. Since we'll be working with vielbeine for the most part, it is convenient to introduce the following objects

$$T_a^{\mu} = \frac{1}{\sqrt{h}} \frac{\delta S}{\delta e_{\mu}^a}, \quad \pi_{\psi} = \frac{1}{\sqrt{h}} \frac{\delta S}{\delta \psi}.$$
 (G.1.20)

We now proceed to derive a couple of useful relations involving the objects above. Writing  $e = \sqrt{h}$ , we see that

$$T^{a}{}_{b} = e^{a}_{\mu}T^{\mu}{}_{a} = e^{a}_{\mu}\frac{1}{e}\frac{\delta S}{\delta e^{b}_{\mu}},$$
(G.1.21)

so, writing

$$A_{\rho} = e_{\rho}^{c} A_{c}, \qquad (G.1.22)$$

we find by virtue of the functional chain rule (note that *r* is fixed!),

$$e^{a}_{\mu}\frac{1}{e}\frac{\delta S}{\delta e^{b}_{\mu}} = e^{a}_{\mu}(x,r)\frac{1}{e(x,r)}\int d^{d}x' \left(\frac{\delta S}{\delta h_{\rho\sigma}(x')}\frac{\delta h_{\rho\sigma}(x')}{\delta e^{b}_{\mu}(x)} + \frac{\delta S}{A_{\rho}(x')}\frac{A_{\rho}(x')}{\delta e^{b}_{\mu}(x)}\right)$$
(G.1.23)  
$$= \int d^{d}x'\delta^{(d)}(x'-x) \left(e^{a}_{\mu}(x)\pi^{\rho\sigma}(x')\eta_{cd}\left(\delta^{c}_{b}\delta^{\mu}_{\rho}e^{d}_{\sigma}(x') + \delta^{d}_{b}\delta^{\mu}_{\sigma}e^{c}_{\rho}\right) + e^{a}_{\mu}(x)Q^{\rho}(x')A_{c}(x')\delta^{c}_{b}\delta^{\mu}_{\rho}\right)$$
(G.1.24)

$$=2\pi^{ad}\eta_{db} + Q^a A_b \tag{G.1.25}$$

$$=2\pi^{a}{}_{b}+Q^{a}A_{b}, (G.1.26)$$

so that,

$$T^{a}_{\ b} = 2\pi^{a}_{\ b} + Q^{a}A_{b}. \tag{G.1.27}$$

Now, since  $\delta A_{\mu} = \delta \psi \delta_{\mu}^{0}$ , we get

$$Q^{\mu} = \pi_{\psi} \delta^{\mu}_{0}, \qquad (G.1.28)$$

which means that

$$Q^{a} = \pi_{\psi} e^{a}_{\mu} \delta^{\mu}_{0} = \pi_{\psi} e^{a}_{0}. \tag{G.1.29}$$

### G.1.2 Boundary Conditions & the $\delta_D$ Expansions

We now introduce spatial flat indices as  $\bar{a}$ . There are now two ways of imposing boundary conditions. Either we linearise the field equations as done in [27, 131], in which case we find the asymptotic behaviour for the vielbeine and the Proca deformation field<sup>2</sup>,

$$e^0_\mu = e^{zr} e^{(0)0}_\mu + \cdots,$$
 (G.1.30)

$$e^{\bar{a}}_{\mu} = e^r e^{(0)\bar{a}}_{\mu} + \cdots$$
, (G.1.31)

$$\psi = e^{-\Delta_{-}r}\psi^{(0)} + \cdots$$
, (G.1.32)

where

$$\Delta_{-} = \frac{1}{2} \left( z + D - \sqrt{(z + d - 1)^2 + 8(z - 1)(z - d + 1)} \right).$$
(G.1.33)

Using the chain rule, we find that

$$\delta_r = \int_{\Sigma_r} \mathrm{d}^d x \, \left( \dot{e}^0_\mu \frac{\delta}{\delta e^0_\mu} + \dot{e}^{\bar{a}}_\mu \frac{\delta}{\delta e^{\bar{a}}_\mu} + \dot{\psi} \frac{\delta}{\delta \psi} \right), \tag{G.1.34}$$

<sup>2</sup> Note that these conditions differ from those in [27, 96] due to our usage of domain-wall coordinates.

where a dot denotes radial differentiation. Similarly, the dilatation operator becomes

$$\delta_D = \int_{\Sigma_r} \mathrm{d}^d x \, \left( \lambda_1 e^0_\mu \frac{\delta}{\delta e^0_\mu} + \lambda_2 e^{\bar{a}}_\mu \frac{\delta}{\delta e^{\bar{a}}_\mu} - \Delta_- \psi \frac{\delta}{\delta \psi} \right), \tag{G.1.35}$$

where the coefficients in front of the various terms in  $\delta_D$  are the leading scaling behaviour of the field, and will be fixed by imposing the HJ equation. Now we'll want to impose the asymptotic boundary condition  $\delta_r = \delta_D$ , but before we do that we'll rewrite both  $\delta_r$  and  $\delta_D$ , using that  $2K_{\mu\nu} = \frac{1}{2}\partial_r h_{\mu\nu}$ 

$$\delta_r = \int_{\Sigma_r} \mathrm{d}^d x \; \sqrt{h} \left( 2K_{ab} \frac{e^a_{\rho} e^b_{\sigma}}{\sqrt{h}} \frac{\delta}{\delta h_{\rho\sigma}} + Q_a \frac{e^a_{\rho}}{\sqrt{h}} \frac{\delta}{\delta A_{\rho}} \right), \tag{G.1.36}$$

which is identical to (G.1.34), since

$$2K_{ab}e^{a}_{\rho}e^{b}_{\sigma}\frac{\delta}{\delta h_{\rho\sigma}} = \left(e^{\mu}_{a}\dot{e}_{\mu b} + e^{\mu}_{b}\dot{e}_{\mu a}\right)e^{a}_{\rho}e^{b}_{\sigma}\frac{\delta}{\delta h_{\rho\sigma}} = \left(\delta^{\mu}_{\rho}\dot{e}_{\mu b}e^{b}_{\sigma} + \delta^{\mu}_{\sigma}\dot{e}_{\mu a}e^{a}_{\rho}\right)\frac{\delta}{\delta h_{\rho\sigma}} \tag{G.1.37}$$

$$=2\dot{e}_{\mu a}e^{a}_{\rho}\frac{\delta}{\delta h_{\mu\rho}}=\eta_{ab}\dot{e}^{a}_{\mu}2e^{b}_{\rho}\frac{\delta}{\delta h_{\mu\rho}}=\dot{e}^{a}_{\mu}(x)\int_{\Sigma_{r}}\mathrm{d}^{d}x'\frac{\delta h_{\nu\rho}(x')}{\delta e^{a}_{\mu}(x)}\frac{\delta}{\delta h_{\nu\rho}(x')}\tag{G.1.38}$$

$$= \dot{e}^a_\mu \frac{\delta}{\delta e^a_\mu}.\tag{G.1.39}$$

Note also that

$$2\dot{e}_{\mu a}e^{a}_{\rho}\frac{\delta}{\delta h_{\mu \rho}}=\dot{h}_{\mu \rho}\frac{\delta}{\delta h_{\mu \rho}},\tag{G.1.40}$$

which shows how  $\delta_r$  in (G.1.34) is obtained from the corresponding expression involving the metric rather than vielbeine in the first place. Now, the relation

$$A_a = (\alpha + \psi) \,\delta_a^0,\tag{G.1.41}$$

implies that the implicit radial derivative reads

$$\delta_r = \int_{\Sigma_r} \mathrm{d}^d x \; \sqrt{h} \left( 2K_{ab} e^a_\rho e^b_\sigma \frac{1}{\sqrt{h}} \frac{\delta}{\delta h_{\rho\sigma}} + \dot{A}_\mu \frac{1}{\sqrt{h}} \frac{\delta}{\delta A_\mu} \right) \tag{G.1.42}$$

$$= \int_{\Sigma_r} \mathrm{d}^d x \; N \sqrt{\gamma} \left( 2K_{ab} e^a_\rho e^b_\sigma \frac{1}{\sqrt{h}} \frac{\delta}{\delta h_{\rho\sigma}} + \frac{1}{\sqrt{h}} Q_\mu \frac{\delta}{\delta A_\mu} \right) \tag{G.1.43}$$

$$= \int_{\Sigma_r} \mathrm{d}^d x \, N \sqrt{\gamma} \left( 2K_{ab} e^a_\rho e^b_\sigma \frac{1}{\sqrt{h}} \frac{\delta}{\delta h_{\rho\sigma}} + \frac{1}{\sqrt{h}} Q_a \frac{\delta}{\delta A_a} \right) \tag{G.1.44}$$

$$= \int_{\Sigma_r} d^d x \ N \sqrt{\gamma} \left( 2K_{ab} e^a_\rho e^b_\sigma \frac{1}{\sqrt{h}} \frac{\delta}{\delta h_{\rho\sigma}} + Q_0 \frac{1}{\sqrt{h}} \frac{\delta}{\delta \psi} \right) \tag{G.1.45}$$

$$= \int_{\Sigma_r} d^d x \ N \sqrt{\gamma} \left( 2K_{ab} \hat{\pi}^{ab} + Q_0 \hat{\pi}_{\psi} \right), \tag{G.1.46}$$

and thus,

$$\hat{\pi}^{ab} = \frac{1}{2} \left( \hat{T}^{ab} - A^b \hat{Q}^a \right) = \frac{1}{2} \left( \hat{T}^{ab} - \eta^{bc} A_c \delta^a_0 \hat{\pi}_\psi \right) \tag{G.1.47}$$

$$= \frac{1}{2} \left( \hat{T}^{ab} - \eta^{bc} (\alpha + \psi) \delta_c^0 \delta_0^a \hat{\pi}_{\psi} \right)$$
(G.1.48)

$$= \frac{1}{2} \left( \hat{T}^{ab} + (\alpha + \psi) \delta^b_0 \delta^a_0 \hat{\pi}_\psi \right). \tag{G.1.49}$$

With this, we obtain the result with all indices *flat*:

$$\delta_r = \int_{\Sigma_r} \mathrm{d}^d x \, N \sqrt{\gamma} \left( K_{ab} \hat{T}^{ab} + \left[ Q_0 + (\alpha + \psi) K_{00} \right] \hat{\pi}_{\psi} \right). \tag{G.1.50}$$

Now, the asymptotic boundary conditions are implemented as

$$\delta_r = \delta_D. \tag{G.1.51}$$

The non-derivative part of the counterterm action must [25], due to general covariance, necessarily be of the form

$$S = \int_{\Sigma_r} d^d x \,\sqrt{h} U(\psi) + \text{derivative terms.}$$
(G.1.52)

Now, the next thing we require is a way to relate the quantities above to the extrinsic curvature. This is achieved through the relation

$$\pi_{\mu\nu} = K_{\mu\nu} - h_{\mu\nu}K, \tag{G.1.53}$$

the flat version of which simply reads

$$\pi_{ab} = K_{ab} - \eta_{ab}K. \tag{G.1.54}$$

In particular, we have that

$$\pi = K(1 - (d_s + 1)) = d_s K, \tag{G.1.55}$$

and thus

$$\pi^{ab} = K^{ab} - \eta^{ab} \frac{\pi}{d_s} \Rightarrow K^{ab} = \pi^{ab} + \eta^{ab} \frac{\pi}{d_s}.$$
(G.1.56)

Now, we use that

$$\pi^{ab} = \frac{1}{2} \left( T^{ab} - Q^a A^b \right) = \frac{1}{2} \left( T^{ab} + U'(\psi)(\alpha + \psi) \delta^a_0 \delta^b_0 \right).$$
(G.1.57)

In particular, we have that

$$T^{\mu}_{\ a} = \frac{1}{\sqrt{h}} \frac{\delta S}{\delta e^{a}_{\mu}} = e^{\mu}_{a} U(\psi), \tag{G.1.58}$$

where we have used that  $\delta e = e e_a^{\mu} \delta e_{\mu}^a$ , implying that

$$T^{ab} = \eta^{ab} U(\psi), \tag{G.1.59}$$

and thus

$$\pi^{ab} = \frac{1}{2} \left( \eta^{ab} U(\psi) + U'(\psi)(\alpha + \psi) \delta_0^a \delta_0^b \right).$$
(G.1.60)

Now, we can find the extrinsic curvature,

$$\pi^{ab} = K^{ab} + \eta^{ab} \frac{\pi}{d_s},\tag{G.1.61}$$

where

$$\pi = \frac{1}{2} \left( (d_s + 1)U(\psi) - U'(\psi)(\alpha + \psi) \right),$$
 (G.1.62)

so that

$$K^{ab} = \pi^{ab} - \eta^{ab} \frac{\pi}{d_s} = -\frac{\eta^{ab}}{2d_s} U(\psi) + \frac{U'(\psi)(\alpha + \psi)}{2} \left(\delta_0^a \delta_0^b + \frac{\eta^{ab}}{d_s}\right)$$
(G.1.63)

$$=\frac{U(\psi)+(d_{s}-1)U'(\psi)(\alpha+\psi)}{2d_{s}}\delta_{0}^{a}\delta_{0}^{b}+\frac{-U(\psi)+(\alpha+\psi)U'(\psi)}{2d_{s}}\delta_{\bar{a}}^{\bar{a}}\delta_{\bar{b}}^{a}.$$
 (G.1.64)

Next, note that

$$Q^{a} = \frac{1}{\sqrt{h}} \frac{\delta S}{\psi} \delta^{a}_{0} = U'(\psi) \delta^{a}_{0} \Rightarrow Q_{0} = -U'(\psi).$$
(G.1.65)

We now want impose the boundary conditions, which implies that

$$\overbrace{K_{ab}\hat{T}^{ab} + (Q_0 + (\alpha + \psi)K_{00})\hat{\pi}_{\psi}}^{r \to \infty} = z\hat{T}^0_{\ 0} + \hat{T}^{\bar{a}}_{\ \bar{a}} - \Delta_-\psi\hat{\pi}_{\psi}.$$
(G.1.66)

Comparing scaling weights and writing  $K_{ab}\hat{T}^{ab} = K^b_a\hat{T}^a_b$  writing out scaling expansions in terms of dilatation weights, e.g.  $Q_0 = \sum_w Q_0^{(w)}$  with  $\delta_D Q_0^{(w)} = -wQ_0^{(w)}$ . This procedure immediately gives us

$$K^{(0)b}{}_{a}\hat{T}^{a}{}_{b} = z\hat{T}^{0}{}_{0} + \hat{T}^{\bar{a}}{}_{\bar{a}} \Rightarrow K^{(0)}_{ab} = -z\delta^{0}_{a}\delta^{0}_{b} + \delta_{\bar{a}\bar{b}}\delta^{\bar{a}}_{a}\delta^{\bar{b}}_{b},$$
(G.1.67)

and, similarly,

$$Q_0^{(0)} + \alpha K_{00}^{(0)} = 0, \tag{G.1.68}$$

$$\psi K_{00}^{(0)} + \alpha K_{00}^{(\Delta_{-})} + Q_0^{(\Delta_{-})} = -\Delta_{-}\psi.$$
(G.1.69)

To proceed, we write  $U(\psi)$  as a series in  $\psi$ :

$$U(\psi) = \sum_{n=0}^{n_{\text{max}}} u_n \psi^n, \qquad (G.1.70)$$

and thus we find for the quantities above, using (G.1.64)

$$K_{00}^{(0)} = \frac{u_0 + \alpha(d_s - 1)u_1}{2d_s}, \quad K_{00}^{(\Delta_-)} = \psi \frac{u_1 + (d_s - 1)(2\alpha u_2 + u_1)}{2d_s}$$
(G.1.71)

$$K_{\bar{a}\bar{b}}^{(0)} = \frac{-u_0 + \alpha u_1}{2d_s} \delta_{\bar{a}\bar{b}}, \quad Q_0^{(0)} = -u_1, \quad Q_0^{(\Delta_-)} = -2u_2\psi.$$
(G.1.72)

Combining the results above with eqs. (G.1.67)–(G.1.69) then gives us—since  $K_{00}^{(0)} = -z$ 

$$Q_0^{(0)} + \alpha K_{00}^{(0)} = 0 \Rightarrow u_1 = -\alpha z, \tag{G.1.73}$$

which we can plug into the relation  $K_{\bar{a}\bar{b}}^{(0)} = \delta_{\bar{a}\bar{b}}$  to obtain for  $u_0$  the expression

$$u_0 = -\alpha^2 z - 2d_s. \tag{G.1.74}$$

Similarly, plugging the value of  $u_1$  into the relation  $K_{00}^{(0)} = -z$  gives us the equality  $u_0 = -2zd_s + \alpha^2(d_s - 1)z$ . Equating the two expression for  $u_0$  gives us the following result for  $\alpha$ :

$$\alpha = \sqrt{\frac{2(z-1)}{z}},\tag{G.1.75}$$

which we in turn can plug into (G.1.74) to obtain:

$$u_0 = -2(z+d_s-1). \tag{G.1.76}$$

Finally, plugging all the above results into (G.1.69), we obtain

$$u_2 = -\frac{zd_s \left(2z - 1 - \Delta_{-}\right)}{2 \left(z + d_s - 1\right)},\tag{G.1.77}$$

which implies that the first couple of (non-derivative) counterterms are given by

$$S_{\rm ct} = \int_{\Sigma_r} d^d x \,\sqrt{h} \left( -2(z+d_s-1) - z\alpha\psi - \frac{zd_s \left(2z-1-\Delta_{-}\right)}{2 \left(z+d_s-1\right)}\psi^2 + \cdots \right). \tag{G.1.78}$$

Expanding the Hamiltonian density  $\mathcal{H}$  extracted from (G.1.11), we can expand  $\mathcal{H}$  in scaling weights,  $\mathcal{H} = \sum_{w} \mathcal{H}^{(w)}$  with  $\delta_D \mathcal{H}^{(w)} = -w \mathcal{H}^{(w)}$ . Note the symmetry property (which we derived in chapter 3)

$$K_{ab}^{(n)}\pi^{(m)ab} = K_{ab}^{(m)}\pi^{(n)ab},$$
(G.1.79)

which has to be taken into account when writing

$$\mathcal{H}^{(w)} = \sum_{n+m=w} \left( K_{ab}^{(m)} \pi^{(n)ab} + \frac{1}{2} Q_a^{(n)} Q^{(m)a} - \frac{M^2}{2} \Phi^{(n)} \Phi^{(m)} \right) + \mathcal{V}^{(w)}, \tag{G.1.80}$$

where  $\mathcal{V}^{(w)}$  is the part of

$$\mathcal{V} = R - 2\Lambda - \frac{1}{2}F_{ab}F^{ab} - \frac{M^2}{2}A_aA^a$$
(G.1.81)

with dilatonic scaling weight w. With this at hand, we can solve the Hamiltonian constraint recursively as we did for AlAdS boundary conditions<sup>3</sup>. The next step will be to determine a *useful relation*, which allows us to find the terms of the on-shell Lagrangian  $\mathcal{L}$ . The idea is to do exactly the same as we did for AlAdS boundary conditions in chapter 3: we expand the (unknown) counterterm Lagrangian in eigenmodes of  $\delta_D$  and relate the terms to the Hamiltonian constraint. We follow the approaches taken by [27, 96]. By the same arguments as for AlAdS boundary conditions, we must now include a linear term (logarithmic in Poincaré-type coordinates), i.e.

$$\mathcal{L}_{\rm ct} = -\sum_{0 \le \lambda < z+d_s} \mathcal{L}^{(\lambda)} - \widetilde{\mathcal{L}}^{(z+d_s)} r.$$
(G.1.83)

As before, the individual terms satisfy the scaling relations:

$$\delta_D \mathcal{L}^{(w)} = -w \mathcal{L}^{(w)}, \quad w \neq z + d_s, \tag{G.1.84}$$

$$\delta_D \mathcal{L}^{(z+d_s)} = -(z+d_s)\mathcal{L}^{(z+d_s)} + \widetilde{\mathcal{L}}^{(z+d_s)}, \qquad (G.1.85)$$

$$\delta_D \widetilde{\mathcal{L}}^{(z+d_s)} = -(z+d_s) \widetilde{\mathcal{L}}^{(z+d_s)}. \tag{G.1.86}$$

As we have seen:

$$(z + d_s + \delta_D)\mathcal{L} = zT^0_{\ 0} + T^{\bar{a}}_{\ \bar{a}} - \Delta_-\psi\pi_{\psi}, \tag{G.1.87}$$

which we can expand in scaling weights to get

$$(z+d_s-w)\mathcal{L}^{(w)} = zT_0^{(w)0} + T_{\bar{a}}^{(w)\bar{a}} - \Delta_-\psi\pi_{\psi}^{(w-\Delta_-)}, \quad w \neq z+d_s,$$
(G.1.88)

while for  $w = z + d_s$ , we get instead

$$(z+d_s+\delta_D) \mathcal{L}^{(z+d_s)} = \widetilde{\mathcal{L}}^{(z+d_s)} = zT_0^{(z+d_s)0} + T_{\bar{a}}^{(z+d_s)\bar{a}} - \Delta_-\psi\pi_{\psi}^{(z+d_s-\Delta_-)}.$$
 (G.1.89)

Now the Hamiltonian constraint, which we have now seen multiple times, is obtained by varying the bulk action with respect to  $\mathcal{N}$  (before setting it equal to one). In particular, using  $\pi_{ab} = K_{ab} - \eta_{ab}K$  and the Proca constraint for  $\Phi$  (recall  $\hat{A} = \Phi dr + A_{\mu} dx^{\mu}$ ), which reads  $\Phi = -\frac{1}{M^2} \nabla_{\mu} Q^{\mu}$ , we can write rewrite the Hamiltonian constraint as

$$K^{2} - K_{ab}K^{ab} - \frac{1}{2}Q_{a}Q^{a} - \frac{1}{2M^{2}}\left(\nabla_{a}Q^{a}\right)^{2} = R - 2\Lambda - \frac{1}{2}F_{ab}F^{ab} - \frac{M^{2}}{2}A_{a}A^{a}, \qquad (G.1.90)$$

which we can expand dilatation weights and plus into (G.1.89), culminating in the following expression, valid for all  $w \notin \{0, \Delta_-, 2\Delta_-\}$ 

$$(z+d_s-w)\mathcal{L}^{(w)}=\mathfrak{Q}^{(w)}+\mathfrak{S}^{(w)}, \qquad (G.1.91)$$

where the quadratic term  $\mathfrak{Q}^{(w)}$  has the form

$$\mathfrak{Q}^{(w)} = \sum_{\substack{0 < \lambda < w/2\\ w \notin \{\Delta_{-}, 2\Delta_{-}\}}} \left[ 2K_{ab}^{(\lambda)} \pi^{(w-\lambda)ab} + Q_{a}^{(\lambda)} Q^{(w-\lambda)a} + \frac{1}{M^{2}} \left( \nabla_{a} Q^{a} \right)^{(\lambda)} \left( \nabla_{a} Q^{a} \right)^{(w-\lambda)} \right] \quad (G.1.92)$$

$$+ \left[ K_{ab}^{(\Delta_{-})} T^{(w-\Delta_{-})ab} + K_{00}^{(\Delta_{-})} Q^{(w-2\Delta_{-})0} \psi + Q_{\bar{a}}^{(\Delta_{-})} Q^{(w-\Delta_{-})\bar{a}} \right]$$
(G.1.93)

$$+\left[K_{ab}^{(w/2)}\pi^{(w/2)ab} + \frac{1}{2}Q_{a}^{(w/2)}Q^{(w/2)a} + \frac{1}{2M^{2}}\left(\nabla_{a}Q^{a}\right)^{(w/2)}\left(\nabla_{a}Q^{a}\right)^{(w/2)}\right]$$
(G.1.94)

where we have conveniently grouped terms in accordance to general scaling behaviour; i.e. the terms in the first bracket involve  $\lambda$  but not  $\Delta_-$ , the next involves  $\Delta_-$  but not  $\lambda$ , while the last involves neither. The contribution from the *source* term is the part of

$$\mathfrak{S} = R - 2\Lambda - \frac{1}{2}F_{ab}F^{ab} - \frac{M^2}{2}A_aA^a, \tag{G.1.95}$$

$$\Delta_{\pm} = \frac{1}{2} \left( z + d_s \pm \sqrt{(z + d_s)^2 + 8(z - 1)(z - d_s)} \right), \tag{G.1.82}$$

where we must pick  $max(-\Delta_{\pm}) = \Delta_{-}$ , since it will be leading.

<sup>3</sup> A good consistency would have been to take  $\Delta_{-} = \Delta$  to be general and then combining  $\delta_r = \delta_D$  along with the Hamiltonian constraint. Doing this, one finds that  $\mathcal{H}^{(0)}$  and  $\mathcal{H}^{(\Delta)}$  vanish identically, while  $\mathcal{H}^{2\Delta}$  only vanishes when

with scaling weight *w*. For  $w \in \{0, \Delta_-, 2\Delta_-\}$ , we find instead

$$(z+d_s)\mathcal{L}^{(0)} = 2\mathcal{L}^{(0)},$$
 (G.1.96)

$$(z + d_s - \Delta_-)\mathcal{L}^{(\Delta_-)} = (\Delta_- - z)\psi\pi_{\psi}^{(0)} + \mathfrak{S}^{(\Delta_-)}, \tag{G.1.97}$$

$$(z+d_s-2\Delta_{-})\mathcal{L}^{(2\Delta_{-})} = (\Delta_{-}-z)\psi\pi_{\psi}^{(\Delta_{-})} + K_{ab}^{(\Delta_{-})}\pi^{(\Delta_{-})ab} + \frac{1}{2}Q_a^{(\Delta_{-})}Q^{(\Delta_{-})a} + \mathfrak{S}^{(2\Delta_{-})}.$$
 (G.1.98)

Now, expanding  $\mathfrak{S}$  in dilatation weights, we find that *R* has terms of weight 2 and 2*z*,  $F_{ab}F^{ab}$  has terms of order 2, 2 +  $\Delta_-$ , 2 + 2 $\Delta_-$ , whereas  $A_a A^a$  has components of order 0,  $\Delta_-$ , 2 $\Delta_-$ :

$$\mathfrak{S}^{(0)} = -2\Lambda + \frac{M^2}{2}\alpha^2 = (z+d_s)(z+d_s-1), \tag{G.1.99}$$

$$\mathfrak{S}^{(2)} = R^{(2)} - \frac{1}{4} \left( F_{ab} F^{ab} \right)^{(2)} = R^{(\gamma)} - \frac{2\nabla^2 N}{N} + \frac{\alpha^2}{2} \left( \frac{\nabla N}{N} \right)^2, \tag{G.1.100}$$

$$\mathfrak{S}^{(2z)} = R^{(2z)} = K_{ij} K^{ij} - (K)^2 + \text{total derivatives}, \qquad (G.1.101)$$

$$\mathfrak{S}^{(\Delta_{-})} = M^2 \alpha \psi = d_s z \alpha \psi, \tag{G.1.102}$$

$$\mathfrak{S}^{(2\Delta_{-})} = \frac{M^2}{2}\psi^2 = \frac{d_s z}{2}\psi^2,\tag{G.1.103}$$

$$\mathfrak{S}^{(2+\Delta_{-})} = -\frac{1}{4} \left( F_{ab} F^{ab} \right)^{(2+\Delta_{-})} = \frac{\alpha \nabla^i N \nabla_i (N\psi)}{N}, \tag{G.1.104}$$

$$\mathfrak{S}^{(2+2\Delta_{-})} = -\frac{1}{4} \left( F_{ab} F^{ab} \right)^{(2+2\Delta_{-})} = \frac{\nabla^{i}(N\psi) \nabla_{i}(N\psi)}{N}.$$
(G.1.105)

Note that we have already determined  $\mathcal{L}^{(\Delta_{-})}$  and  $\mathcal{L}^{(2\Delta_{-})}$  by imposing the asymptotic equality  $\delta_r = \delta_D$ .

# G.1.3 Finding the Counterterms

The considerations above have enabled us to finally begin extracting the actual counterterms. In particular, at the non-derivative level at zero weight, we get

$$\mathcal{L}^{(0)} = \frac{2\mathfrak{S}^{(0)}}{z+d_s} = 2(z+d_s-1), \tag{G.1.106}$$

in agreement with our previous findings. Reassuringly, the terms  $\mathcal{L}^{(\Delta_{-})}$  and  $\mathcal{L}^{(2\Delta_{-})}$  also give what we found by imposing the asymptotic equality  $\delta_r = \delta_D$ , although this way of determining them is rather more cumbersome. Noting that a weight two, the quadratic term vanishes, i.e.  $\mathfrak{Q}^{(0)} = 0$ , we find that

$$\mathcal{L}^{(2)} = \frac{1}{z + d_s - 2} \left( R^{(\gamma)} + \frac{\alpha^2 \nabla^i N \nabla_i N}{2N^2} \right).$$
(G.1.107)

In this manner, all the counterterms can be calculated. For details, see the appendices of [96], which presents extensive calculations of the counterterms.

In this appendix, we describe Kaluza-Klein reduction and Scherk-Schwarz reduction. We then apply the Scherk-Schwarz reduction to the *uplift*: a five-dimensional Einstein-Dilaton theory that reduces to a four-dimensional EPD model admitting z = 2 Lifshitz solutions. The uplift is the precursor of the electromagnetic uplift discussed in chapter (6).

### H.1 KALUZA-KLEIN REDUCTION

Kaluza-Klein (KK) reduction was originally introduced by Kaluza and Klein [216, 217] as a way to unify gravity and electromagnetism. They showed that by compactifying a fifth dimension, it was possible to obtain gravity coupled to electromagnetism with a small caveat: this produced an additional scalar particle, the dilaton. In this section, we describe the generalities of the method, following [218–220].

Starting from a generic theory containing gravity  $\hat{g}_{\hat{\mu}\hat{\nu}}$  (indices  $\hat{\mu}, \hat{\nu} = 1, ..., D+1$ ) and matter fields collectively denoted  $\hat{\Phi}$ , described by the Lagrangian

$$\mathcal{L} = \sqrt{-\hat{g}}\hat{R} + \cdots, \qquad (H.1.1)$$

we assume that (D + 1)-dimensional spacetime can be written as a direct product  $\mathcal{M}_{D+1} = \mathcal{Q}_{d+1} \times \mathcal{C}_{D-d}$  with  $\mathcal{C}_{D-d}$  a compact manifold. This product manifold will thus be a solution to the equations of motion obtained from (H.1.1), exhibiting what [220] terms *spontaneous compactification*, which just means that the metric respects the product structure

$$\hat{g}(x,y) = \begin{pmatrix} \mathring{g}_{\mu\nu}(x) & 0\\ 0 & \mathring{g}_{mn}(y), \end{pmatrix}$$
(H.1.2)

where  $x^{\mu}$  are the coordinates on the Lorentzian  $Q_{d+1}$  and  $y^{m}$  the coordinates on the compact Riemannian manifold  $C_{D-d}$ . The metric is naturally dynamical, so the metric above is the background *ground state* solution of the theory. Generally, the fields on  $Q_{d+1}$  are taken to depend trivially on the coordinates of  $C_{D-d}$ , as we see happens automatically in (H.1.2). The fields on  $C_{D-d}$ , on the other hand, must be truncated in a consistent manner. Typically, this involves keeping only certain modes of a Fourier-like expansion. To be specific, let the compact manifold be  $S^1$ , in which case we have the decomposition  $\mathcal{M}_{D+1} = \mathcal{Q}_D \times S^1$ . Let y be the compactified coordinate, which is thus periodically identified,

$$y = y + 2\pi L, \tag{H.1.3}$$

with *L* the radius of the circle, which is usually assumed to be small. In this case, a generic field (including the metric) of our theory can be expanded in Fourier-like modes on  $S^1$ ,

$$\hat{\Phi}(x,y) = \sum_{n} \Phi^{(n)}(x) e^{iny/L}.$$
(H.1.4)

The modes  $n \neq 0$  correspond to massive fields, while the mode n = 0 represents a massless field, as can be realized by considering a massless scalar  $\hat{\phi}$  on *flat* (D + 1)-dimensional space, which satisfies the Klein-Gordon equation,

$$\hat{\Box}\hat{\phi} = 0, \tag{H.1.5}$$

where  $\hat{\Box} = \partial^{\hat{\mu}} \partial_{\hat{\mu}}$ , which, when compactified on  $S^1$ , admits an expansion of the form (H.1.4),  $\hat{\phi} = \sum_n \phi^{(n)} e^{iny/L}$ , so the Klein-Gordon equation implies that

$$0 = \partial^{\hat{\mu}} \partial_{\hat{\mu}} \hat{\phi} = \sum_{n} \left( \Box \phi^{(n)} - \frac{n^2}{L^2} \right) e^{iny/L}, \tag{H.1.6}$$

so linear independence of the exponentials implies that the mass of the n'th mode is

$$m_n = \frac{|n|}{L},\tag{H.1.7}$$

which, due to the infinity of every increasing masses, is known as the Kaluza-Klein tower of states. Thus, when  $L \ll 1$  (in practice, one takes L to be of the order of the Planck length), the non-zero modes will be immensely heavy and can be safely neglected. This corresponds, as we claimed, to only keeping the zero-mode of the generalized Fourier expansion and thus truncates the Kaluza-Klein spectrum. This is sometimes known as the Kaluza-Klein reduction ansatz, since, generally<sup>1</sup>, we have to worry about the consistency of our truncation, that is, whether ours setting all massive modes equal to zero solves the (D+1)-dimensional equations of motion. In general, if the fields in the Kaluza-Klein expansion (H.1.4) have a global symmetry group G, then the truncation will be consistent if we keep all G-singlets (that is, modes that do not transform under G). Clearly, in the case of the massless field on flat space, the expansion  $\hat{\phi} = \sum_{n} \phi^{(n)} e^{iny/L}$  and spectrum (H.1.7) has a G = U(1) symmetry rotating modes  $n = \pm m$  into each other, implying that only the n = 0 mode is invariant under G, and thus our truncation is consistent. See [44] for more details.

### H.1.1 Kaluza-Klein Reduction of Pure Gravity & the EMD Model

Now, let's try our hand at an explicit example: let's compactify (d + 1)-dimensional pure gravity over  $S^1$ , following the analysis in [218]. Heuristically, the reduction of  $\hat{g}_{\hat{\mu}\hat{\nu}}$  results in the following fields

- $\hat{g}_{\mu\nu}$ , which is the metric of the reduced theory,
- $\hat{g}_{\mu\nu}$ , a one-form, which seems likely to become a U(1) gauge field,
- a scalar  $\hat{g}_{yy}$ .

In order to perform the reduction, we employ the following KK ansatz for the (D + 1)-dimensional metric,

$$d\hat{s}^{2} = e^{2\alpha\phi} ds^{2} + e^{2\beta\phi} (dy + A)^{2},$$
(H.1.8)

where  $\alpha$ ,  $\beta$  are constants that we'll choose in a convenient matter shortly, and  $\mathcal{A} = \mathcal{A}_{\mu} dx^{\mu}$ . The reduced fields we discussed earlier may then be expressed in terms of the quantities appearing in (H.1.8) in the following manner,

$$\hat{g}_{\mu\nu} = e^{2\alpha\phi}g_{\mu\nu} + e^{2\beta\phi}\mathcal{A}_{\mu}\mathcal{A}_{\nu}, \quad \hat{g}_{\mu y} = e^{2\beta\phi}\mathcal{A}_{\mu}, \quad \hat{g}_{yy} = e^{2\beta\phi}. \tag{H.1.9}$$

The metric (H.1.8) is admits the following description in terms of vielbeine,

$$\hat{e}^{a} = \hat{e}^{a}_{\mu} dx^{\mu} = e^{\alpha \phi} e^{a}, \qquad \hat{e}^{z} = e^{\beta \phi} (dy + \mathcal{A}).$$
 (H.1.10)

where Latin letters represent *D*-dimensional flat indices. Introducing capital Latin letters for (D + 1)-dimensional flat indices, we can write our vielbeine in matrix form,

$$\hat{e}^{A}_{\hat{\mu}} = \begin{pmatrix} e^{\beta\phi} & e^{\beta\phi}A_{\mu} \\ 0 & e^{\alpha\phi}e^{a}_{\mu} \end{pmatrix}.$$
(H.1.11)

Noting that this matrix is triangular, the determinant is just the product of the diagonal elements, i.e.  $\sqrt{-\hat{g}} = \hat{e} = e^{(D\alpha+\beta)}e$ . Requiring that we be in Einstein-Frame (i.e., that the Einstein-Hilbert action retains its canonical form,  $S \sim \int \sqrt{-g}R$ ), we should choose  $\beta = -\alpha(D-2)$ , since, as we shall shortly, the Ricci scalar carries a factor  $e^{-2\alpha\phi}$ . Similarly, by requiring that the kinetic scalar term, which comes from the reduced Ricci scalar, is canonically normalized, we get

$$\alpha^2 = \frac{1}{2(D-1)(D-1)}, \quad \beta = -(D-2)\alpha.$$
 (H.1.12)

Now we proceed to calculate the Ricci scalar. Since our connection is symmetric, the spin connection is determined via

$$\mathrm{d}\hat{e}^{A} + \hat{\omega}^{A}_{\ B} \wedge \hat{e}^{B} = 0, \tag{H.1.13}$$

<sup>1</sup> Except for the case of compactification over tori, where keeping only the massless mode is always consistent, note in particular the (trivial) isomorphism  $T^1 \simeq S^1$ , so that keeping only massless modes

which gives us

$$\hat{\omega}^{ab} = \omega^{ab} + \alpha e^{-\alpha\phi} \left( \partial^b \phi \hat{e}^a - \partial^a \phi \hat{e}^b \right) - \frac{1}{2} \mathcal{F}^{ab} e^{(\beta - 2\alpha)\phi} \hat{e}^y, \tag{H.1.14}$$

$$\hat{\omega}^{ay} = -\beta e^{-\alpha\phi} e^{-\alpha\phi} \partial^a \phi \hat{e}^y - \frac{1}{2} \mathcal{F}^a{}_b e^{(\beta - 2\alpha)\phi} \hat{e}^b, \qquad (H.1.15)$$

where  $\mathcal{F} = d\mathcal{A}$ . Now, the curvature two form is given by

$$\hat{R}_{AB} = \mathrm{d}\hat{\omega}_{AB} + \hat{\omega}_{AC} \wedge \hat{\omega}^{C}{}_{B}, \qquad (\mathrm{H.1.16})$$

the trace of which can be computed using the explicit results for the spin connections (H.1.14)–(H.1.15); the result is [218]

$$\hat{R} = e^{-2\alpha\phi} \left[ R - \frac{1}{2} (\partial\phi)^2 + (D - 3)\alpha \Box\phi \right] - \frac{1}{4} e^{-2D\alpha\phi} \mathcal{F}^2, \tag{H.1.17}$$

and thus, combining our findings, we obtain the reduced action<sup>2</sup>

$$S = \int \mathrm{d}^D \sqrt{-g} \left( R - \frac{1}{2} (\partial \phi)^2 - \frac{1}{4} e^{-2(d-1)\alpha\phi} \mathcal{F}_{\mu\nu} \mathcal{F}^{\mu\nu} \right), \tag{H.1.18}$$

for which we can readily determine the equations of motion; they read

$$R_{\mu\nu} = \frac{1}{2} \partial_{\mu} \phi \partial_{\nu} \phi + \frac{1}{2} e^{-2(D-1)\alpha\phi} \left( \mathcal{F}_{\mu\lambda} \mathcal{F}_{\nu\rho} g^{\rho\lambda} - \frac{1}{2(D-2)} \mathcal{F}^2 g_{\mu\nu} \right), \tag{H.1.19}$$

$$0 = \nabla^{\mu} \left( e^{-2(D-1)\alpha\phi} \mathcal{F}_{\mu\nu} \right), \tag{H.1.20}$$

$$\Box \phi = -\frac{1}{2} (D-1) \alpha e^{-2(D-1)\alpha \phi} \mathcal{F}^2.$$
(H.1.21)

In particular, we note that the equation of motion for  $\phi$ , (H.1.21) implies that setting  $\phi = 0$  would be an inconsistent truncation.

Thus, reducing pure gravity in (D + 1) dimensions results in *D*-dimensional Einstein-Maxwell-Dilaton theory.

### H.2 SCHERK-SCHWARZ REDUCTION

The Scherk-Schwarz reduction<sup>3</sup>, which was introduced in [221, 222], is a generalization of the KK method. The main idea of Scherk-Schwarz reduction is use certain symmetries of the higher-dimensional theory to generate masses in the reduced theory.

This is achieved by allowing the higher-dimensional fields to depend on the compact direction<sup>4</sup> u in a manner consistent with the symmetries of the higher-dimensional action, which ensures that the u-dependence drops out of the equations of motion of the reduced action. These symmetries fall into two categories

- Global/internal symmetries: phase, scale and shift symmetries etc.
- Local/external symmetries: space-time symmetries, e.g. translations or rotations in the compact manifold.

We will deal only with symmetries of the first kind.

Consider a global U(1) symmetry acting on the fields as  $\hat{\Phi} \to e^{i\Lambda}\hat{\Phi}$  for some  $\Lambda \in \mathbb{R}$ . The generalization of the periodicity condition (H.1.3) then reads

$$\hat{\Phi}(x,u+2\pi L) = \overbrace{e^{2\pi i m L}}^{\in U(1)} \hat{\Phi}(x,u), \quad m \in \mathbb{R},$$
(H.2.1)

i.e. we identify the two fields up to a global phase transformation, which is sometimes known as a *twist*. By using this twisted periodicity, we may obtain the following Fourier decomposition

$$\hat{\Phi}(x,u) = e^{imu} \sum_{n} \Phi_n(x) e^{inu/L}.$$
(H.2.2)

<sup>&</sup>lt;sup>2</sup> We ignore the extra factor of  $2\pi L$  arising from the integration over the compact dimension.

<sup>3</sup> Sometimes the procedure is known as generalized dimensional reduction or twisted reduction, for reasons we will see shortly.

<sup>4</sup> We change our notation in this section so as to match the one used in chapter 6—we hope that this will not cause any confusion.

In the limit  $L \rightarrow 0$ —just as for the KK case—the massive modes decouple (i.e. become infinitely massive and can be ignored), leaving us with the ansatz,

$$\hat{\Phi}(x,u) = e^{imu}\Phi_0(x),\tag{H.2.3}$$

which is a *local* U(1) transformation ( $\Lambda = mu$ ) acting on the zero-mode; in this way, the global U(1) has been gauged. This allows us to unravel the general structure (see also [223]): if the higherdimensional action has a global symmetry group G acting on the fields as  $\hat{\Phi} \rightarrow g(\hat{\Phi})$  for  $g \in G$ , The generalization of the reduction ansatz (H.2.3) then involves an element  $g_u = g(u) \in G$  with an explicit dependence on the compact direction u, i.e., schematically,

$$\hat{\Phi}(x,u) = g_z(\Phi(x)), \tag{H.2.4}$$

where it can (and usually will; cf. (H.2.3)) happen that only certain modes are included on the righthand side. This results in a reduced theory independent of u and, as we saw in the case of global U(1)symmetry, gauges the group G. Now, since g(u) is not periodic, going around the compact coordinate results in a twist, also in this context known as the *monodromy*,

$$G \ni \mathcal{M}(g_u) := g(2\pi L) \tag{H.2.5}$$

which is just the generalization of the twist in (H.2.2). The exponential map allows us to express  $g_u$  in terms an element M of the Lie algebra  $\mathfrak{g}$  of G,

$$g(u) = e^{\frac{Mu}{2\pi L}}, \quad M \in \mathfrak{g}. \tag{H.2.6}$$

 $M \in \mathfrak{g}$  spans a one-dimensional subgroup H of G. These considerations imply that the monodromy may be expressed as

$$\mathcal{M}(g) = e^M,\tag{H.2.7}$$

whereas *M*, which is in fact the mass matrix of the reduced theory, can be written as

$$M = 2\pi L g^{-1} \partial_u g. \tag{H.2.8}$$

By demanding that M does not depend on u, it is possible to determine the function g.

#### H.2.1 Scherk-Schwarz Reduction of the Uplift

In this section, we give the details of the reduction alluded to in section 6.2.1. Consider a fivedimensional Einstein-Dilaton model of the form

$$S_{5d} = \frac{1}{2\kappa_5^2} \int d^5x \sqrt{-\mathcal{G}} \left( R^{(\mathcal{G})} + 12 - \frac{1}{2} \partial_{\mathcal{M}} \psi \partial^{\mathcal{M}} \psi \right), \qquad (H.2.9)$$

with  $\kappa_5^2 = 8\pi G_5$  and  $\mathcal{M} = (u, M)$ . The reduction ansatz is the one we used in chapter 6; it reads

$$ds_{5}^{2} = \mathcal{G}_{\mathcal{M}\mathcal{N}} dx^{\mathcal{M}} dx^{\mathcal{N}} = \frac{dr^{2}}{r^{2}} + \gamma_{AB} dx^{A} dx^{B} = e^{-\Phi} g_{MN} dx^{M} dx^{N} + e^{2\Phi} \left( du + A_{M} dx^{M} \right)^{2}$$
(H.2.10)

$$= e^{-\Phi} \left( e^{\Phi} \frac{\mathrm{d}r^2}{r^2} + h_{\mu\nu} \mathrm{d}x^{\mu} \mathrm{d}x^{\nu} \right) + e^{2\Phi} \left( \mathrm{d}u + A_{\mu} \mathrm{d}x^{\mu} \right)^2, \tag{H.2.11}$$

$$\psi = 2u + 2\Xi, \tag{H.2.12}$$

where the four dimensional fields  $g_{MN}$ ,  $A_M$ ,  $\Xi$  and  $\Phi$  do not depend on the compactified *u*-direction, which—since we're reducing on a circle of radius *L*—is periodically identified,  $u \sim u + 2\pi L$ . Note also that since our normalization is such that  $\frac{1}{16\pi G_4} = 1$ , the five-dimensional Newton constant satisfies  $\frac{2\pi L}{16\pi G_5} = 1$ .

The renormalized on-shell four-dimensional EPD z = 2 action has the schematic form

$$S_{\rm ren} = S + S_{\rm gh} + S_{\rm ct},$$
 (H.2.13)

where  $S_{\text{gh}} = 2 \int d^3x \sqrt{-hK}$  is the Gibbons-Hawking boundary, which, as per usual, is required to make the variational problem well posed, while the counterterm is the dimensionally reduction of the expression found in section 3.2.1 with the additional Maxwell field set to zero.

The reduction ansatz reads in components:

$$\mathcal{G}_{MN} = e^{-\Phi}g_{MN} + e^{2\Phi}A_MA_N, \quad \mathcal{G}_{Mu} = e^{2\Phi}A_M, \quad \mathcal{G}_{uu} = e^{2\Phi},$$
 (H.2.14)

while the inverse becomes

$$\mathcal{G}^{MN} = e^{\Phi}g^{MN}, \quad \mathcal{G}^{Mu} = -e^{\Phi}A^M, \quad \mathcal{G}^{uu} = e^{-2\Phi} + e^{\Phi}A^M A_M,$$
(H.2.15)

as we now demonstrate. The requirement  $\mathcal{G}_{\mathcal{MR}}\mathcal{G}^{\mathcal{RN}} = \delta_{\mathcal{M}}^{\mathcal{N}}$  is equivalent to

$$\mathcal{G}_{M\mathcal{M}}\mathcal{G}^{\mathcal{M}N} = \delta^N_M, \quad \mathcal{G}_{u\mathcal{M}}\mathcal{G}^{\mathcal{M}u} = 1, \quad \mathcal{G}_{u\mathcal{M}}\mathcal{G}^{\mathcal{M}M} = 0.$$
 (H.2.16)

To check this, we observe that

4

$$\mathcal{G}_{M\mathcal{M}}\mathcal{G}^{\mathcal{M}N} = \mathcal{G}_{MR}\mathcal{G}^{RN} + \mathcal{G}_{Mu}\mathcal{G}^{uN} \tag{H.2.17}$$

$$= \left(e^{-\Phi}g_{MR} + e^{2\Phi}A_M A_R\right)e^{\Phi}g^{RN} - e^{3\Phi}A_N A^N$$
(H.2.18)

$$=g_{MR}g^{RN} (H.2.19)$$

$$=\delta_{M}^{N},\tag{H.2.20}$$

and

$$\mathcal{G}_{u\mathcal{M}}\mathcal{G}^{\mathcal{M}u} = \mathcal{G}_{uu}\mathcal{G}^{uu} + \mathcal{G}_{u\mathcal{M}}\mathcal{G}^{\mathcal{M}u} \tag{H.2.21}$$

$$=e^{2\Phi}\left(e^{-2\Phi}+\underline{e}^{\Phi}A_{M}A^{N}\right)-\underline{e}^{3\Phi}A_{M}A^{M}$$
(H.2.22)

and finally,

$$\mathcal{G}_{u\mathcal{M}}\mathcal{G}^{\mathcal{M}M} = \mathcal{G}_{uu}\mathcal{G}^{uM} + \mathcal{G}_{uN}\mathcal{G}^{NM} \tag{H.2.24}$$

$$= -e^{3\Phi}A^{M} + e^{3\Phi}A_{N}g^{MN}$$
(H.2.25)

$$= 0.$$
 (H.2.26)

So what we have found is indeed the inverse. Alternatively, we could have used the block matrix inversion identity

$$\begin{pmatrix} \mathbf{A} & \mathbf{B} \\ \mathbf{C} & \mathbf{D} \end{pmatrix}^{-1} = \begin{pmatrix} \left( \mathbf{A} - \mathbf{B}\mathbf{D}^{-1}\mathbf{C} \right)^{-1} & -(\mathbf{A} - \mathbf{B}\mathbf{D}^{-1}\mathbf{C})^{-1}\mathbf{B}\mathbf{D}^{-1} \\ -\mathbf{D}^{-1}\mathbf{C}(\mathbf{A} - \mathbf{B}\mathbf{D}^{-1}\mathbf{C})^{-1} & \mathbf{D}^{-1} + \mathbf{D}^{-1}\mathbf{C}(\mathbf{A} - \mathbf{B}\mathbf{D}^{-1}\mathbf{C})^{-1}\mathbf{B}\mathbf{D}^{-1} \end{pmatrix}.$$
 (H.2.27)

In a similar vein,  $2 \times 2$  block matrices for invertible **D** satisfy the following determinant relation,

$$\det \begin{pmatrix} \mathbf{A} & \mathbf{B} \\ \mathbf{C} & \mathbf{D} \end{pmatrix} = \det(\mathbf{D}) \det \left( \mathbf{A} - \mathbf{B}\mathbf{D}^{-1}\mathbf{C} \right), \qquad (H.2.28)$$

implying that

$$\det(-\mathcal{G}) = e^{2\Phi} \det\left(-e^{-\Phi}g_{MN} - e^{2\Phi}A_{\mathcal{M}}A_N + e^{2\Phi}A_{\mathcal{M}}A_N\right) = e^{-2\Phi} \det(-g), \qquad (\text{H.2.29})$$

i.e.

$$\sqrt{-\mathcal{G}} = e^{-\Phi} \sqrt{-g}.\tag{H.2.30}$$

From this, we may immediately identify the KK scalar *k* (in the language of [219]),

,

$$k = e^{-\Phi}.\tag{H.2.31}$$

Following [219], we now proceed with the reduction of the Ricci scalar; starting off with the Christoffel symbols, we find that

$$\Gamma_{\mathcal{M}\mathcal{N}}^{\mathcal{R}} = \frac{1}{2} \mathcal{G}^{\mathcal{R}\mathcal{S}} \left( \partial_{\mathcal{M}} \mathcal{G}_{\mathcal{N}\mathcal{S}} + \partial_{\mathcal{N}} \mathcal{G}_{\mathcal{M}\mathcal{S}} - \partial_{\mathcal{S}} \mathcal{G}_{\mathcal{M}\mathcal{N}} \right), \tag{H.2.32}$$

$$\Gamma_{MN}^{R} = \frac{1}{2} \mathcal{G}^{RS} \left( \partial_{M} \mathcal{G}_{NS} + \partial_{N} \mathcal{G}_{MS} - \partial_{S} \mathcal{G}_{MN} \right)$$
(H.2.33)

$$=\frac{1}{2}\mathcal{G}^{RS}\left(\partial_{M}\mathcal{G}_{NS}+\partial_{N}\mathcal{G}_{MS}-\partial_{S}\mathcal{G}_{MN}\right)+\frac{1}{2}\mathcal{G}^{Ru}\left(\partial_{M}\mathcal{G}_{Nu}+\partial_{N}\mathcal{G}_{Mu}\right).$$
(H.2.34)

This means that the five-dimensional Ricci can be decomposed in the following manner [219],

$$R_{uu}^{(\mathcal{G})} = -e^{3\Phi} \Box_{(g)} \Phi + \frac{1}{4} e^{6\Phi} F_{MN} F^{MN}, \tag{H.2.35}$$

$$R_{uM}^{(\mathcal{G})} = R_{uu}^{(\mathcal{G})} A_M + \frac{1}{2} \nabla^N \left( e^{3\Phi} F_{MN} \right), \tag{H.2.36}$$

$$R_{MN}^{(\mathcal{G})} = A_M R_{uN}^{(\mathcal{G})} + A_N R_{uM}^{(\mathcal{G})} - A_M A_N R_{uu}^{(\mathcal{G})} + R_{MN}^{(g)} - \frac{3}{2} \partial_M \Phi \partial_N \Phi + \frac{1}{2} g_{MN} \Box_{(g)} \Phi - \frac{1}{2} e^{3\Phi} F_{MP} F_N^{P}.$$
(H.2.37)

Hence, the five-dimensional Ricci scalar can be written as

$$R^{(\mathcal{G})} = \mathcal{G}^{\mathcal{M}\mathcal{N}} R^{(\mathcal{G})}_{\mathcal{M}\mathcal{N}} = \mathcal{G}^{uu} R^{(\mathcal{G})}_{uu} + 2\mathcal{G}^{uM} R^{(\mathcal{G})}_{uM} + \mathcal{G}^{MN} R^{(\mathcal{G})}_{MN}.$$
 (H.2.38)

We compute these contributions individually. First up is

$$\mathcal{G}^{uu}R^{(\mathcal{G})}_{uu} = \left(e^{-2\Phi} + e^{\Phi}A^{M}A_{M}\right)\left(-e^{3\Phi}\Box_{(g)}\Phi + \frac{1}{4}e^{6\Phi}F_{MN}F^{MN}\right),\tag{H.2.39}$$

whereas

$$2\mathcal{G}^{uM}R_{uM}^{(\mathcal{G})} = -2e^{\Phi}A^{M}A_{M}\left(-e^{3\Phi}\Box_{(g)}\Phi + \frac{1}{4}e^{6\Phi}F_{MN}F^{MN}\right) - e^{\Phi}A^{M}\nabla^{N}\left(e^{3\Phi}F_{MN}\right), \quad (\text{H.2.40})$$

and, finally,

$$\mathcal{G}^{MN}R_{MN}^{(\mathcal{G})} = \mathbf{Z}e^{\Phi}\left(-e^{3\Phi}\Box_{(g)}\Phi + \frac{1}{4}e^{6\Phi}F_{MN}F^{MN}\right)A^{M}A_{M} + e^{\Phi}A^{M}\nabla^{N}\left(e^{3\Phi}F_{MN}\right)$$
(H.2.41)

$$-e^{\Phi}A^{M}A_{M}\left(-e^{3\Phi}\Box_{(g)}\Phi + \frac{1}{4}e^{6\Phi}F_{MN}F^{MN}\right) + e^{\Phi}R^{(g)} - \frac{3}{2}e^{\Phi}\partial^{M}\Phi\partial_{M}\Phi \quad (\text{H.2.42})$$

$$+2e^{\Phi}\Box_{(g)}\Phi - \frac{1}{2}e^{4\Phi}F_{MN}F^{MN}$$
(H.2.43)

$$= e^{\Phi} \left( -e^{3\Phi} \Box_{(g)} \Phi + \frac{1}{4} e^{6\Phi} F_{MN} F^{MN} \right) A^{M} A_{M} + e^{\Phi} A^{M} \nabla^{N} \left( e^{3\Phi} F_{MN} \right)$$
(H.2.44)

$$+e^{\Phi}R^{(g)} - \frac{3}{2}e^{\Phi}\partial^{M}\Phi\partial_{M}\Phi + 2e^{\Phi}\Box_{(g)}\Phi - \frac{1}{2}e^{4\Phi}F_{MN}F^{MN}.$$
 (H.2.45)

Combining everything, we obtain

$$R^{(\mathcal{G})} = e^{\Phi}R^{(g)} - \frac{3}{2}e^{\Phi}\partial^{M}\Phi\partial_{M}\Phi - \frac{1}{4}e^{4\Phi}F_{MN}F^{MN} + e^{\Phi}\Box_{(g)}\Phi.$$
 (H.2.46)

Next, we consider the reduction of the scalar term, for which the ansatz (H.2.12) gives us

$$-\frac{1}{2}\partial_{\mathcal{M}}\psi\partial^{\mathcal{M}}\psi = -\frac{1}{2}\mathcal{G}^{\mu\nu}\partial_{\mu}\psi\partial^{\mu}\psi - \mathcal{G}^{\mu\mathcal{M}}\partial_{\mu}\psi\partial_{\mathcal{M}}\psi - \frac{1}{2}\mathcal{G}^{\mathcal{M}\mathcal{N}}\partial_{\mathcal{M}}\psi\partial_{\mathcal{N}}\psi \tag{H.2.47}$$

$$= -2\mathcal{G}^{uu} - 4\mathcal{G}^{uM}\partial_M\psi - 2\mathcal{G}^{MN}\partial_M\psi\partial_N\psi \qquad (\text{H.2.48})$$

$$= -2\left(e^{-2\Phi} + 4e^{\Phi}A^{M}A_{M}\right) + 4e^{\Phi}A^{M}\partial_{M}\Xi - 2e^{\Phi}\partial^{M}\Xi\partial_{M}\Xi \qquad (\text{H.2.49})$$

$$= -2B^{M}B_{M} - 2e^{-2\Phi}, (H.2.50)$$

where  $B_M = A_M - \partial_M \Xi$ .

Turning our attention to the Gibbons-Hawking term, we note that, first of all

$$K_{AB} = -\frac{1}{2} \pounds_n \gamma_{AB}, \tag{H.2.51}$$

where  $n^{\mathcal{M}} = -r\delta_r^{\mathcal{M}}$  is the normal vector to radial slices. From the definition of the Lie derivative, this gives us

$$K_{AB} = \frac{1}{2} n^{\mathcal{M}} \partial_{\mathcal{M}} \gamma_{AB} = \frac{r}{2} \partial_r \gamma_{AB}. \tag{H.2.52}$$

The trace of the extrinsic curvature is given by

$$K = \gamma^{AB} K_{AB}. \tag{H.2.53}$$

To determine the inverse metric  $\gamma^{AB}$ , we apply the inversion identity (H.2.27): since

$$\gamma_{AB} = \begin{pmatrix} e^{-\Phi} h_{\mu\nu} + e^{2\Phi} A_{\mu} A_{\nu} & e^{2\Phi} A_{\nu} \\ e^{2\Phi} A_{\mu} & e^{2\Phi} \end{pmatrix},$$
(H.2.54)

we get

$$\gamma^{AB} = \begin{pmatrix} e^{\Phi} h^{\mu\nu} & -e^{\Phi} A^{\nu} \\ -e^{\Phi} A^{\mu} & e^{-2\Phi} + e^{\Phi} A^{\mu} A_{\mu} \end{pmatrix}.$$
 (H.2.55)

Therefore, we find that

$$K = \frac{r}{2} \gamma^{AB} \partial_r \gamma_{AB} = \left[ \frac{r}{2} \gamma^{\mu\nu} \partial_r \gamma_{\mu\nu} \right] + \left\{ r \gamma^{\mu u} \partial_r \gamma_{\mu u} \right\} + \left( \frac{r}{2} \gamma^{u u} \partial_r \gamma_{u u} \right)$$
(H.2.56)

$$= \left[ -\frac{3r}{2}\partial_r \Phi + \frac{r}{2}h^{\mu\nu}\partial_r h_{\mu\nu} + \frac{re^{3\Phi}A^{\mu}A_{\mu}\partial_r \Phi}{r} + \frac{re^{3\Phi}A^{\mu}\partial_r A_{\mu}}{r} \right]$$
(H.2.57)

$$-\left\{\underline{2re^{3\Phi}A^{\mu}A_{\mu}\partial_{r}\Phi + re^{3\Phi}A^{\mu}\partial_{r}A_{\mu}}\right\}$$
(H.2.58)

$$+\left(r\partial_{r}\Phi + re^{3\Phi}A^{\mu}A_{\mu}\partial_{r}\Phi\right) \tag{H.2.59}$$

$$= -\frac{r}{2}\partial_r \Phi + \frac{r}{2}h^{\mu\nu}\partial_r h_{\mu\nu}, \qquad (H.2.60)$$

and so

$$2\sqrt{-\gamma}K = \sqrt{-h} \left( -n^M \partial_M \Phi + 2K^{(h)} \right), \tag{H.2.61}$$

where  $n^M = e^{-\Phi/2} \delta_r^M$ . We note that the first term precisely cancels out out the boundary term produced by the Laplacian term present in the expression for the reduced Ricci scalar (H.2.46).

The reduction of the counterterm proceeds completely analogously, and gives the result

$$S_{\rm ct} = 2 \int_{\partial \mathcal{M}} d^3 x \, \sqrt{-h} \left( -\frac{1}{4} e^{\Phi/2} \left[ R[h] - \frac{3}{2} \partial_\mu \Phi \partial^\mu \Phi - \frac{1}{4} e^{3\Phi} F_{\mu\nu} F^{\mu\nu} - 2B_\mu B^\mu + 10e^{-\Phi} \right] \right)$$
(H.2.62)

$$-\log r \int_{\partial \mathcal{M}} \mathrm{d}^3 x \,\sqrt{-h} e^{-\Phi/2} \mathcal{A}.\tag{H.2.63}$$

Combining our findings, the reduced renormalized action becomes—after integration over the compact direction *u* and using  $\frac{2\pi L}{2\kappa_5^2} = 1$ 

$$S_{\rm ren} = \int d^4x \,\sqrt{-g} \left( R[g] - \frac{3}{2} \partial^M \Phi \partial_M \Phi - \frac{1}{4} e^{3\Phi} F_{MN} F^{MN} - 2B^M B_M - V(\Phi) \right) \tag{H.2.64}$$

$$+2\int d^3x \sqrt{-h}K + S_{ct},$$
 (H.2.65)

where

$$V(\Phi) = 2e^{-3\Phi} - 12e^{-\Phi}.$$
 (H.2.66)

- [1] Juan Martin Maldacena. "The Large N limit of superconformal field theories and supergravity." In: *Int. J. Theor. Phys.* 38 (1999). [Adv. Theor. Math. Phys.2,231(1998)], pp. 1113–1133. DOI: 10.1023/A:1026654312961. arXiv: hep-th/9711200 [hep-th] (cit. on pp. 1, 123, 124).
- [2] Gerard 't Hooft. "Dimensional reduction in quantum gravity." In: Salamfest 1993:0284-296. 1993, pp. 0284-296. arXiv: gr-qc/9310026 [gr-qc] (cit. on pp. 1, 4).
- [3] Leonard Susskind. "The World as a hologram." In: J. Math. Phys. 36 (1995), pp. 6377–6396. DOI: 10.1063/1.531249. arXiv: hep-th/9409089 [hep-th] (cit. on pp. 1, 4).
- [4] Edward Witten. "String theory dynamics in various dimensions." In: Nucl. Phys. B443 (1995), pp. 85–126. DOI: 10.1016/0550-3213(95)00158-0. arXiv: hep-th/9503124 [hep-th] (cit. on p. 1).
- [5] Sean A. Hartnoll, Christopher P. Herzog, and Gary T. Horowitz. "Building a Holographic Superconductor." In: *Phys. Rev. Lett.* 101 (2008), p. 031601. DOI: 10.1103/PhysRevLett.101.031601. arXiv: 0803.3295 [hep-th] (cit. on pp. 1, 14, 129, 130, 131, 132, 134, 135).
- [6] Sean A. Hartnoll, Christopher P. Herzog, and Gary T. Horowitz. "Holographic Superconductors." In: *JHEP* 12 (2008), p. 015. DOI: 10.1088/1126-6708/2008/12/015. arXiv: 0810.1563 [hep-th] (cit. on pp. 1, 14, 135).
- [7] C. P. Herzog, P. K. Kovtun, and D. T. Son. "Holographic model of superfluidity." In: *Phys. Rev.* D79 (2009), p. 066002. DOI: 10.1103/PhysRevD.79.066002. arXiv: 0809.4870 [hep-th] (cit. on p. 1).
- [8] Frederik Denef and Sean A. Hartnoll. "Landscape of superconducting membranes." In: *Phys. Rev.* D79 (2009), p. 126008. DOI: 10.1103/PhysRevD.79.126008. arXiv: 0901.1160 [hep-th] (cit. on pp. 1, 129).
- [9] G. Policastro, Dan T. Son, and Andrei O. Starinets. "The Shear viscosity of strongly coupled N=4 supersymmetric Yang-Mills plasma." In: *Phys. Rev. Lett.* 87 (2001), p. 081601. DOI: 10.1103/ PhysRevLett.87.081601. arXiv: hep-th/0104066 [hep-th] (cit. on pp. 1, 14).
- [10] Dam T. Son and Andrei O. Starinets. "Viscosity, Black Holes, and Quantum Field Theory." In: Ann. Rev. Nucl. Part. Sci. 57 (2007), pp. 95–118. DOI: 10.1146/annurev.nucl.57.090506.123120. arXiv: 0704.0240 [hep-th] (cit. on p. 1).
- [11] Jan Zaanen, Ya-Wen Sun, Yan Liu, and Koenraad Schalm. Holographic Duality in Condensed Matter Physics. Cambridge Univ. Press, 2015. ISBN: 9781107080089. URL: http://www.cambridge. org/mw/academic/subjects/physics/condensed-matter-physics-nanoscience-and-mesoscopicphysics/holographic-duality-condensed-matter-physics?format=HB (cit. on pp. 1, 2, 4, 5, 9, 10, 14, 106, 119, 126, 128, 129).
- [12] John McGreevy. "Holographic duality with a view toward many-body physics." In: Adv. High Energy Phys. 2010 (2010), p. 723105. DOI: 10.1155/2010/723105. arXiv: 0909.0518 [hep-th] (cit. on pp. 1, 2, 4, 5, 7, 8, 126, 128).
- [13] Sean A. Hartnoll. "Lectures on holographic methods for condensed matter physics." In: *Class. Quant. Grav.* 26 (2009), p. 224002. DOI: 10.1088/0264-9381/26/22/224002. arXiv: 0903.3246
   [hep-th] (cit. on pp. 1, 126, 128, 129).
- [14] John McGreevy. "TASI 2015 Lectures on Quantum Matter (with a View Toward Holographic Duality)." In: Proceedings, Theoretical Advanced Study Institute in Elementary Particle Physics: New Frontiers in Fields and Strings (TASI 2015): Boulder, CO, USA, June 1-26, 2015. 2017, pp. 215–296. DOI: 10.1142/9789813149441\_0004. arXiv: 1606.08953 [hep-th]. URL: http://inspirehep.net/record/1472992/files/arXiv:1606.08953.pdf (cit. on p. 1).
- [15] Christopher P. Herzog, Pavel Kovtun, Subir Sachdev, and Dam Thanh Son. "Quantum critical transport, duality, and M-theory." In: *Phys. Rev.* D75 (2007), p. 085020. DOI: 10.1103/PhysRevD. 75.085020. arXiv: hep-th/0701036 [hep-th] (cit. on pp. 1, 134, 135).
- [16] Philip W Anderson. "Strange connections to strange metals." In: *Physics Today* 66.4 (2013), p. 9 (cit. on p. 1).

- [17] Nabil Iqbal and Hong Liu. "Universality of the hydrodynamic limit in AdS/CFT and the membrane paradigm." In: *Phys. Rev.* D79 (2009), p. 025023. DOI: 10.1103/PhysRevD.79.025023. arXiv: 0809.3808 [hep-th] (cit. on p. 1).
- [18] Veronika E. Hubeny, Shiraz Minwalla, and Mukund Rangamani. "The fluid/gravity correspondence." In: *Black holes in higher dimensions*. [,817(2011)]. 2012, pp. 348–383. arXiv: 1107.5780 [hep-th]. URL: https://inspirehep.net/record/920982/files/arXiv:1107.5780.pdf (cit. on p. 1).
- [19] D. T. Son. "Toward an AdS/cold atoms correspondence: A Geometric realization of the Schrodinger symmetry." In: *Phys. Rev.* D78 (2008), p. 046003. DOI: 10.1103/PhysRevD.78.046003. arXiv: 0804.3972 [hep-th] (cit. on p. 1).
- [20] Koushik Balasubramanian and John McGreevy. "Gravity duals for non-relativistic CFTs." In: *Phys. Rev. Lett.* 101 (2008), p. 061601. DOI: 10.1103/PhysRevLett.101.061601. arXiv: 0804.4053 [hep-th] (cit. on p. 1).
- [21] Shamit Kachru, Xiao Liu, and Michael Mulligan. "Gravity duals of Lifshitz-like fixed points." In: *Phys. Rev.* D78 (2008), p. 106005. DOI: 10.1103/PhysRevD.78.106005. arXiv: 0808.1725
   [hep-th] (cit. on p. 1).
- [22] Marika Taylor. "Non-relativistic holography." In: None (2008). arXiv: 0812.0530 [hep-th] (cit. on pp. 1, 99, 147).
- [23] Morten H. Christensen, Jelle Hartong, Niels A. Obers, and Blaise Rollier. "Torsional Newton-Cartan Geometry and Lifshitz Holography." In: *Phys. Rev.* D89 (2014), p. 061901. DOI: 10.1103/ PhysRevD.89.061901. arXiv: 1311.4794 [hep-th] (cit. on pp. 1, 3, 37, 46, 47, 51, 53, 63, 67, 71, 75).
- [24] Morten H. Christensen, Jelle Hartong, Niels A. Obers, and Blaise Rollier. "Boundary Stress-Energy Tensor and Newton-Cartan Geometry in Lifshitz Holography." In: *JHEP* 01 (2014), p. 057. DOI: 10.1007/JHEP01(2014)057. arXiv: 1311.6471 [hep-th] (cit. on pp. 1, 3, 37, 46, 47, 48, 49, 51, 55, 56, 63, 67, 71, 74, 75, 76, 78, 84, 89).
- [25] Marco Baggio, Jan de Boer, and Kristian Holsheimer. "Hamilton-Jacobi Renormalization for Lifshitz Spacetime." In: *JHEP* 01 (2012), p. 058. DOI: 10.1007/JHEP01(2012)058. arXiv: 1107. 5562 [hep-th] (cit. on pp. 1, 35, 141, 146, 147, 150).
- [26] Simon F. Ross and Omid Saremi. "Holographic stress tensor for non-relativistic theories." In: JHEP 09 (2009), p. 009. DOI: 10.1088/1126-6708/2009/09/009. arXiv: 0907.1846 [hep-th] (cit. on pp. 1, 64).
- [27] Simon F. Ross. "Holography for asymptotically locally Lifshitz spacetimes." In: *Class. Quant. Grav.* 28 (2011), p. 215019. DOI: 10.1088/0264-9381/28/21/215019. arXiv: 1107.4451 [hep-th] (cit. on pp. 1, 15, 35, 51, 53, 64, 99, 100, 146, 148, 152).
- [28] Wissam Chemissany and Ioannis Papadimitriou. "Lifshitz holography: The whole shebang." In: JHEP 01 (2015), p. 052. DOI: 10.1007/JHEP01(2015)052. arXiv: 1408.0795 [hep-th] (cit. on pp. 1, 23, 25, 35, 36, 51, 99, 100).
- [29] Jelle Hartong, Elias Kiritsis, and Niels A. Obers. "Lifshitz spacetimes for Schrodinger holography." In: *Phys. Lett.* B746 (2015), pp. 318–324. DOI: 10.1016/j.physletb.2015.05.010. arXiv: 1409.1519 [hep-th] (cit. on pp. 1, 2, 3, 37, 46, 51, 52, 55, 59, 60, 62, 63, 64, 66, 98).
- [30] E. Cartan. "Sur les variétés à connexion affine et la théorie de la relativité généralisée. (première partie) (Suite)." In: *Annales Sci. Ecole Norm. Sup.* 40 (1923), pp. 325–412 (cit. on pp. 2, 37).
- [31] E. Cartan. "Sur les variÃl'tÃl's Ãă connexion affine et la thÃl'orie de la relativitÃl' gÃl'nÃl'ralisÃl'e. (premiÃlre partie) (Suite)." In: Annales Sci. Ecole Norm. Sup. 41 (1924), pp. 1–25 (cit. on pp. 2, 37).
- [32] Niels A. Obers Jelle Hartong Elias Kiritsis. "In preparation..." (cit. on pp. 2, 3, 37, 51, 52, 53, 54, 55, 59, 64, 83, 98, 99).
- [33] Jelle Hartong, Elias Kiritsis, and Niels A. Obers. "Field Theory on Newton-Cartan Backgrounds and Symmetries of the Lifshitz Vacuum." In: *JHEP* 08 (2015), p. 006. DOI: 10.1007/JHEP08(2015) 006. arXiv: 1502.00228 [hep-th] (cit. on pp. 2, 37, 46, 47, 48, 51, 52, 59, 60, 75, 80, 100).
- [34] Eric A. Bergshoeff, Jelle Hartong, and Jan Rosseel. "Torsional Newton-Cartan geometry and the Schrödinger algebra." In: *Class. Quant. Grav.* 32.13 (2015), p. 135017. DOI: 10.1088/0264-9381/32/13/135017. arXiv: 1409.5555 [hep-th] (cit. on pp. 2, 3, 37, 38, 42, 46, 47, 59, 60, 61, 67, 80, 81, 82, 83).

- [35] Dam Thanh Son. "Newton-Cartan Geometry and the Quantum Hall Effect." In: (2013). arXiv: 1306.0638 [cond-mat.mes-hall] (cit. on pp. 2, 37, 50).
- [36] Michael Geracie, Dam Thanh Son, Chaolun Wu, and Shao-Feng Wu. "Spacetime Symmetries of the Quantum Hall Effect." In: *Phys. Rev.* D91 (2015), p. 045030. DOI: 10.1103/PhysRevD.91. 045030. arXiv: 1407.1252 [cond-mat.mes-hall] (cit. on pp. 2, 37, 50, 93).
- [37] Michael Geracie and Dam Thanh Son. "Hydrodynamics on the lowest Landau level." In: JHEP o6 (2015), p. 044. DOI: 10.1007/JHEP06(2015)044. arXiv: 1408.6843 [cond-mat.mes-hall] (cit. on pp. 2, 50).
- [38] Chaolun Wu and Shao-Feng Wu. "Horava-Lifshitz gravity and effective theory of the fractional quantum Hall effect." In: *JHEP* 01 (2015), p. 120. DOI: 10.1007/JHEP01(2015)120. arXiv: 1409. 1178 [hep-th] (cit. on pp. 2, 99).
- [39] Andrey Gromov and Alexander G. Abanov. "Thermal Hall Effect and Geometry with Torsion." In: *Phys. Rev. Lett.* 114 (2015), p. 016802. DOI: 10.1103/PhysRevLett.114.016802. arXiv: 1407. 2908 [cond-mat.str-el] (cit. on p. 2).
- [40] Michael Geracie, Siavash Golkar, and Matthew M. Roberts. "Hall viscosity, spin density, and torsion." In: (2014). arXiv: 1410.2574 [hep-th] (cit. on p. 2).
- [41] Jelle Hartong, Emil Have, Cynthia Keeler, and Niels A. Obers. "To appear…" In: () (cit. on pp. 2, 67).
- [42] Ofer Aharony, Steven S. Gubser, Juan Martin Maldacena, Hirosi Ooguri, and Yaron Oz. "Large N field theories, string theory and gravity." In: *Phys. Rept.* 323 (2000), pp. 183–386. DOI: 10. 1016/S0370-1573(99)00083-6. arXiv: hep-th/9905111 [hep-th] (cit. on pp. 2, 4, 6, 103, 109, 121, 123).
- [43] Martin Ammon and Johanna Erdmenger. *Gauge/gravity duality*. Cambridge, UK: Cambridge Univ. Pr., 2015. ISBN: 9781107010345. URL: http://www.cambridge.org/de/academic/subjects/physics/theoretical-physics-and-mathematical-physics/gaugegravity-duality-foundations-and-applications (cit. on pp. 2, 4, 7, 8, 9, 11, 12, 13, 103, 104, 109, 111, 112, 114, 116, 118, 119, 121, 122, 123, 125, 126, 127, 135).
- [44] Horatiu Nastase. *Introduction to the ADS/CFT Correspondence*. Cambridge: Cambridge University Press, 2015. ISBN: 9781107085855 (cit. on pp. 2, 4, 11, 104, 118, 121, 155).
- [45] Eric D'Hoker and Daniel Z. Freedman. "Supersymmetric gauge theories and the AdS / CFT correspondence." In: *Strings, Branes and Extra Dimensions: TASI 2001: Proceedings.* 2002, pp. 3–158. arXiv: hep-th/0201253 [hep-th] (cit. on pp. 2, 4, 8, 11, 13, 114, 120).
- [46] Raphael Bousso. "The Holographic principle." In: *Rev. Mod. Phys.* 74 (2002), pp. 825–874. DOI: 10.1103/RevModPhys.74.825. arXiv: hep-th/0203101 [hep-th] (cit. on pp. 2, 4, 5).
- [47] Wolfgang Mueck and K. S. Viswanathan. "Conformal field theory correlators from classical scalar field theory on AdS(d+1)." In: *Phys. Rev.* D58 (1998), p. 041901. DOI: 10.1103/PhysRevD. 58.041901. arXiv: hep-th/9804035 [hep-th] (cit. on pp. 2, 4, 13).
- [48] Kostas Skenderis. "Lecture notes on holographic renormalization." In: Class. Quant. Grav. 19 (2002), pp. 5849–5876. DOI: 10.1088/0264-9381/19/22/306. arXiv: hep-th/0209067 [hep-th] (cit. on pp. 2, 17, 107, 136, 137).
- [49] Jan de Boer, Erik P. Verlinde, and Herman L. Verlinde. "On the holographic renormalization group." In: JHEP 08 (2000), p. 003. DOI: 10.1088/1126-6708/2000/08/003. arXiv: hep-th/ 9912012 [hep-th] (cit. on pp. 2, 15, 141, 142).
- [50] Ioannis Papadimitriou. "Lectures on Holographic Renormalization." In: *Springer Proc. Phys.* 176 (2016), pp. 131–181. DOI: 10.1007/978-3-319-31352-8\_4 (cit. on pp. 2, 15, 17, 18, 25).
- [51] Ioannis Papadimitriou. "Holographic Renormalization of general dilaton-axion gravity." In: *JHEP* 08 (2011), p. 119. DOI: 10.1007/JHEP08(2011)119. arXiv: 1106.4826 [hep-th] (cit. on pp. 2, 15, 21, 25, 29, 33, 142).
- [52] Kostas Skenderis and Ioannis Papadimitriou. "In preparation..." (cit. on pp. 2, 15, 34).
- [53] Jelle Hartong and Niels A. Obers. "Horava-Lifshitz gravity from dynamical Newton-Cartan geometry." In: *JHEP* 07 (2015), p. 155. DOI: 10.1007/JHEP07(2015)155. arXiv: 1504.07461
   [hep-th] (cit. on pp. 2, 37, 38, 47, 66, 67, 75, 84, 99).

- [54] Roel Andringa, Eric Bergshoeff, Sudhakar Panda, and Mees de Roo. "Newtonian Gravity and the Bargmann Algebra." In: *Class. Quant. Grav.* 28 (2011), p. 105011. DOI: 10.1088/0264-9381/ 28/10/105011. arXiv: 1011.1145 [hep-th] (cit. on pp. 2, 37, 38, 42, 43, 44, 82).
- [55] E. A. Ivanov and J. Niederle. "Gauge Formulation of Gravitation Theories. 1. The Poincare, De Sitter and Conformal Cases." In: *Phys. Rev.* D25 (1982), p. 976. DOI: 10.1103/PhysRevD.25.976 (cit. on pp. 2, 37).
- [56] Guido Festuccia, Dennis Hansen, Jelle Hartong, and Niels A. Obers. "Torsional Newton-Cartan Geometry from the Noether Procedure." In: *Phys. Rev.* D94.10 (2016), p. 105023. DOI: 10.1103/ PhysRevD.94.105023. arXiv: 1607.01926 [hep-th] (cit. on pp. 2, 37, 47, 48).
- [57] Jelle Hartong, Niels A. Obers, and Marco Sanchioni. "Lifshitz Hydrodynamics from Lifshitz Black Branes with Linear Momentum." In: *JHEP* 10 (2016), p. 120. DOI: 10.1007/JHEP10(2016) 120. arXiv: 1606.09543 [hep-th] (cit. on pp. 2, 3, 29, 33, 37, 47, 48, 49, 51, 52, 66, 67, 71, 74, 75, 76, 95, 99, 100).
- [58] Guido Festuccia, Dennis Hansen, Jelle Hartong, and Niels A. Obers. "Symmetries and Couplings of Non-Relativistic Electrodynamics." In: *JHEP* 11 (2016), p. 037. DOI: 10.1007/JHEP11(2016) 037. arXiv: 1607.01753 [hep-th] (cit. on pp. 3, 68, 69, 70, 75).
- [59] Raphael Bousso. "A Covariant entropy conjecture." In: JHEP 07 (1999), p. 004. DOI: 10.1088/ 1126-6708/1999/07/004. arXiv: hep-th/9905177 [hep-th] (cit. on pp. 4, 5).
- [60] J. D. Bekenstein. "Black holes and the second law." In: Lett. Nuovo Cim. 4 (1972), pp. 737–740.
   DOI: 10.1007/BF02757029 (cit. on p. 4).
- [61] Ali Masoumi and Samir D. Mathur. "A violation of the covariant entropy bound?" In: *Phys. Rev.* D91.8 (2015), p. 084058. DOI: 10.1103/PhysRevD.91.084058. arXiv: 1412.2618 [hep-th] (cit. on p. 5).
- [62] Eanna E. Flanagan, Donald Marolf, and Robert M. Wald. "Proof of classical versions of the Bousso entropy bound and of the generalized second law." In: *Phys. Rev.* D62 (2000), p. 084035. DOI: 10.1103/PhysRevD.62.084035. arXiv: hep-th/9908070 [hep-th] (cit. on p. 5).
- [63] Raphael Bousso, Horacio Casini, Zachary Fisher, and Juan Maldacena. "Proof of a Quantum Bousso Bound." In: *Phys. Rev.* D90.4 (2014), p. 044002. DOI: 10.1103/PhysRevD.90.044002. arXiv: 1404.5635 [hep-th] (cit. on p. 5).
- [64] Raphael Bousso. "Light sheets and Bekenstein's bound." In: *Phys. Rev. Lett.* 90 (2003), p. 121302.
   DOI: 10.1103/PhysRevLett.90.121302. arXiv: hep-th/0210295 [hep-th] (cit. on p. 5).
- [65] Gary T. Horowitz and Joseph Polchinski. "Gauge/gravity duality." In: *ArXiv* (2006). arXiv: gr-qc/0602037 [gr-qc] (cit. on p. 5).
- [66] Joseph Polchinski. "Introduction to Gauge/Gravity Duality." In: Proceedings, Theoretical Advanced Study Institute in Elementary Particle Physics (TASI 2010). String Theory and Its Applications: From meV to the Planck Scale: Boulder, Colorado, USA, June 1-25, 2010. 2010, pp. 3–46. DOI: 10.1142/9789814350525\_0001. arXiv: 1010.6134 [hep-th]. URL: http://inspirehep.net/record/875017/files/arXiv:1010.6134.pdf (cit. on p. 5).
- [67] Zvi Bern. "Perturbative quantum gravity and its relation to gauge theory." In: *Living Rev. Rel.* 5 (2002), p. 5. DOI: 10.12942/lrr-2002-5. arXiv: gr-qc/0206071 [gr-qc] (cit. on p. 5).
- [68] Steven Weinberg and Edward Witten. "Limits on Massless Particles." In: *Phys. Lett.* B96 (1980), pp. 59–62. DOI: 10.1016/0370-2693(80)90212-9 (cit. on p. 5).
- [69] M. Srednicki. *Quantum field theory*. Cambridge University Press, 2007. ISBN: 9780521864497, 9780511267208 (cit. on pp. 6, 12, 27).
- [70] Michael E. Peskin and Daniel V. Schroeder. An Introduction to quantum field theory. 1995. ISBN: 9780201503975, 0201503972. URL: http://www.slac.stanford.edu/spires/find/books/www? cl=QC174.45%3AP4 (cit. on pp. 6, 12).
- [71] Edward Witten. "Geometric Langlands From Six Dimensions." In: (2009). arXiv: 0905.2720 [hep-th] (cit. on p. 6).
- [72] Elias Kiritsis. *String theory in a nutshell*. 2007 (cit. on pp. 7, 13).
- [73] Leonard Susskind and Edward Witten. "The Holographic bound in anti-de Sitter space." In: (1998). arXiv: hep-th/9805114 [hep-th] (cit. on p. 7).
- [74] L. Susskind and J. Lindesay. *An introduction to black holes, information and the string theory revolution: The holographic universe.* 2005 (cit. on p. 7).

- [75] K. Konishi. "Anomalous Supersymmetry Transformation of Some Composite Operators in SQCD." In: *Phys. Lett.* 135B (1984), pp. 439–444. DOI: 10.1016/0370-2693(84)90311-3 (cit. on p. 8).
- [76] David Tong. "Lectures on holographic conductivity." In: Presented at Cracow School of Theoretical Physics (2013), p. 1 (cit. on pp. 8, 133).
- [77] Peter Breitenlohner and Daniel Z. Freedman. "Positive Energy in anti-De Sitter Backgrounds and Gauged Extended Supergravity." In: *Phys. Lett.* B115 (1982), pp. 197–201. DOI: 10.1016/ 0370-2693(82)90643-8 (cit. on p. 9).
- S. S. Gubser, Igor R. Klebanov, and Alexander M. Polyakov. "Gauge theory correlators from noncritical string theory." In: *Phys. Lett.* B428 (1998), pp. 105–114. DOI: 10.1016/S0370-2693(98) 00377-3. arXiv: hep-th/9802109 [hep-th] (cit. on p. 10).
- [79] Edward Witten. "Anti-de Sitter space and holography." In: Adv. Theor. Math. Phys. 2 (1998), pp. 253–291. arXiv: hep-th/9802150 [hep-th] (cit. on pp. 10, 12, 105, 108).
- [80] Eliot Hijano, Per Kraus, Eric Perlmutter, and River Snively. "Semiclassical Virasoro blocks from AdS<sub>3</sub> gravity." In: JHEP 12 (2015), p. 077. DOI: 10.1007/JHEP12(2015)077. arXiv: 1508.04987 [hep-th] (cit. on p. 13).
- [81] Heng-Yu Chen, En-Jui Kuo, and Hideki Kyono. "Anatomy of Geodesic Witten Diagrams." In: *JHEP* 05 (2017), p. 070. DOI: 10.1007/JHEP05(2017)070. arXiv: 1702.08818 [hep-th] (cit. on p. 13).
- [82] Joao Penedones. "Writing CFT correlation functions as AdS scattering amplitudes." In: JHEP 03 (2011), p. 025. DOI: 10.1007/JHEP03(2011)025. arXiv: 1011.1485 [hep-th] (cit. on p. 13).
- [83] Dhritiman Nandan, Anastasia Volovich, and Congkao Wen. "On Feynman Rules for Mellin Amplitudes in AdS/CFT." In: JHEP 05 (2012), p. 129. DOI: 10.1007/JHEP05(2012)129. arXiv: 1112.0305 [hep-th] (cit. on p. 13).
- [84] Suvrat Raju. "BCFW for Witten Diagrams." In: *Phys. Rev. Lett.* 106 (2011), p. 091601. DOI: 10. 1103/PhysRevLett.106.091601. arXiv: 1011.0780 [hep-th] (cit. on p. 13).
- [85] Daniel Z. Freedman, Samir D. Mathur, Alec Matusis, and Leonardo Rastelli. "Correlation functions in the CFT(d) / AdS(d+1) correspondence." In: *Nucl. Phys.* B546 (1999), pp. 96–118. DOI: 10.1016/S0550-3213(99)00053-X. arXiv: hep-th/9804058 [hep-th] (cit. on p. 13).
- [86] Sebastian de Haro, Sergey N. Solodukhin, and Kostas Skenderis. "Holographic reconstruction of space-time and renormalization in the AdS / CFT correspondence." In: *Commun. Math. Phys.* 217 (2001), pp. 595–622. DOI: 10.1007/s002200100381. arXiv: hep-th/0002230 [hep-th] (cit. on pp. 15, 33, 139, 140).
- [87] Charles Fefferman and C Robin Graham. "Conformal invariants." In: Asterisque 95 (1985) (cit. on pp. 15, 107).
- [88] Jose-Martin Garcia. "xAct: Efficient tensor computer algebra for the Wolfram Language." In: Mathematica Package (2016). URL: http://xact.es/ (cit. on pp. 15, 33, 139).
- [89] Jan de Boer. "The Holographic renormalization group." In: *Fortsch. Phys.* 49 (2001), pp. 339–358. DOI: 10.1002/1521-3978(200105)49:4/6<339::AID-PROP339>3.0.C0;2-A. arXiv: hep-th/0101026 [hep-th] (cit. on pp. 15, 19, 141).
- [90] Alexander L Fetter and John Dirk Walecka. *Theoretical mechanics of particles and continua*. Courier Corporation, 2003 (cit. on pp. 15, 16).
- [91] Herbert Goldstein. *Classical mechanics*. Addison-Wesley Pub. Co., 1980 (cit. on pp. 15, 16, 141).
- [92] Vladimir Igorevich Arnol'd. *Mathematical methods of classical mechanics*. Vol. 60. Springer Science & Business Media, 2013 (cit. on pp. 15, 16).
- [93] Ioannis Papadimitriou and Kostas Skenderis. "Thermodynamics of asymptotically locally AdS spacetimes." In: JHEP 08 (2005), p. 004. DOI: 10.1088/1126-6708/2005/08/004. arXiv: hep-th/0505190 [hep-th] (cit. on pp. 15, 22, 105).
- [94] Ioannis Papadimitriou and Kostas Skenderis. "AdS / CFT correspondence and geometry." In: IRMA Lect. Math. Theor. Phys. 8 (2005), pp. 73–101. DOI: 10.4171/013-1/4. arXiv: hep-th/0404176 [hep-th] (cit. on p. 15).
- [95] Ioannis Papadimitriou and Kostas Skenderis. "Correlation functions in holographic RG flows." In: *JHEP* 10 (2004), p. 075. DOI: 10.1088/1126-6708/2004/10/075. arXiv: hep-th/0407071 [hep-th] (cit. on pp. 15, 140).

- [96] Tom Griffin, Petr Horava, and Charles M. Melby-Thompson. "Conformal Lifshitz Gravity from Holography." In: *JHEP* 05 (2012), p. 010. DOI: 10.1007/JHEP05(2012)010. arXiv: 1112.5660 [hep-th] (cit. on pp. 15, 51, 146, 147, 148, 152, 153).
- [97] Ana Cannas Da Silva. Lectures on symplectic geometry. Vol. 3575. Springer, 2001 (cit. on p. 16).
- [98] Richard L. Arnowitt, Stanley Deser, and Charles W. Misner. "The Dynamics of general relativity." In: *Gen. Rel. Grav.* 40 (2008), pp. 1997–2027. DOI: 10.1007/s10714-008-0661-1. arXiv: gr-qc/0405109 [gr-qc] (cit. on pp. 18, 141).
- [99] Ioannis Papadimitriou. "Multi-Trace Deformations in AdS/CFT: Exploring the Vacuum Structure of the Deformed CFT." In: *JHEP* 05 (2007), p. 075. DOI: 10.1088/1126-6708/2007/05/075. arXiv: hep-th/0703152 [hep-th] (cit. on p. 18).
- [100] M. Henneaux and C. Teitelboim. *Quantization of gauge systems*. Princeton, USA: Univ. Pr. (1992) 520 p, 1992. ISBN: 0691037698, 9780691037691 (cit. on p. 19).
- [101] Bryce S. DeWitt. "Quantum Theory of Gravity. I. The Canonical Theory." In: *Phys. Rev.* 160 (5 1967), pp. 1113–1148. DOI: 10.1103/PhysRev.160.1113. URL: http://link.aps.org/doi/10.1103/PhysRev.160.1113 (cit. on p. 20).
- [102] Jelle Hartong, Yang Lei, and Niels A. Obers. "Nonrelativistic Chern-Simons theories and threedimensional Hoava-Lifshitz gravity." In: *Phys. Rev.* D94.6 (2016), p. 065027. DOI: 10.1103/ PhysRevD.94.065027. arXiv: 1604.08054 [hep-th] (cit. on p. 20).
- [103] Ingmar Kanitscheider, Kostas Skenderis, and Marika Taylor. "Precision holography for non-conformal branes." In: *JHEP* 09 (2008), p. 094. DOI: 10.1088/1126-6708/2008/09/094. arXiv: 0807.3324 [hep-th] (cit. on p. 21).
- [104] Bom Soo Kim. "Holographic Renormalization of Einstein-Maxwell-Dilaton Theories." In: *JHEP* 11 (2016), p. 044. DOI: 10.1007/JHEP11(2016)044. arXiv: 1608.06252 [hep-th] (cit. on p. 22).
- [105] Blaise Gouteraux, Jelena Smolic, Milena Smolic, Kostas Skenderis, and Marika Taylor. "Holog-raphy for Einstein-Maxwell-dilaton theories from generalized dimensional reduction." In: JHEP 01 (2012), p. 089. DOI: 10.1007/JHEP01(2012)089. arXiv: 1110.2320 [hep-th] (cit. on p. 22).
- [106] Wissam Chemissany and Ioannis Papadimitriou. "Generalized dilatation operator method for non-relativistic holography." In: *Phys. Lett.* B737 (2014), pp. 272–276. DOI: 10.1016/j.physletb. 2014.08.057. arXiv: 1405.3965 [hep-th] (cit. on pp. 25, 35, 36, 99, 100).
- [107] Wissam Chemissany, David Geissbuhler, Jelle Hartong, and Blaise Rollier. "Holographic Renormalization for z=2 Lifshitz Space-Times from AdS." In: *Class. Quant. Grav.* 29 (2012), p. 235017.
   DOI: 10.1088/0264-9381/29/23/235017. arXiv: 1205.5777 [hep-th] (cit. on pp. 29, 51, 67, 84).
- [108] Teake Nutma. "xTras : A field-theory inspired xAct package for mathematica." In: *Comput. Phys. Commun.* 185 (2014), pp. 1719–1738. DOI: 10.1016/j.cpc.2014.02.006. arXiv: 1308.3493
   [cs.SC] (cit. on pp. 33, 139).
- [109] Monica Guica, Kostas Skenderis, Marika Taylor, and Balt C. van Rees. "Holography for Schrodinger backgrounds." In: JHEP 02 (2011), p. 056. DOI: 10.1007/JHEP02(2011)056. arXiv: 1008.1991 [hep-th] (cit. on p. 35).
- [110] Tomas Andrade, Cynthia Keeler, Alex Peach, and Simon F. Ross. "Schrödinger holography with z = 2." In: *Class. Quant. Grav.* 32.8 (2015), p. 085006. DOI: 10.1088/0264-9381/32/8/085006. arXiv: 1412.0031 [hep-th] (cit. on p. 35).
- [111] Tomas Andrade, Cynthia Keeler, Alex Peach, and Simon F. Ross. "Schrödinger holography for z < 2." In: *Class. Quant. Grav.* 32.3 (2015), p. 035015. DOI: 10.1088/0264-9381/32/3/035015. arXiv: 1408.7103 [hep-th] (cit. on p. 35).
- [112] Xavier Bekaert and Kevin Morand. "Connections and dynamical trajectories in generalised Newton-Cartan gravity I. An intrinsic view." In: J. Math. Phys. 57.2 (2016), p. 022507. DOI: 10.1063/1.4937445. arXiv: 1412.8212 [hep-th] (cit. on pp. 37, 41).
- [113] Steven Weinberg. *The Quantum theory of fields. Vol. 1: Foundations*. Cambridge University Press, 2005. ISBN: 9780521670531, 9780511252044 (cit. on p. 37).
- [114] J. Baez and J. P. Muniain. *Gauge fields, knots and gravity*. Singapore, Singapore: World Scientific (1994) 465 p. (Series on knots and everything, 4), 1995 (cit. on p. 37).
- [115] Christian Duval and Peter A. Horvathy. "Non-relativistic conformal symmetries and Newton-Cartan structures." In: J. Phys. A42 (2009), p. 465206. DOI: 10.1088/1751-8113/42/46/465206. arXiv: 0904.0531 [math-ph] (cit. on pp. 41, 80, 81).

- [116] Antonio N. Bernal and Miguel Sanchez. "Leibnizian, Galilean and Newtonian structures of space-time." In: J. Math. Phys. 44 (2003), pp. 1129–1149. DOI: 10.1063/1.1541120. arXiv: grqc/0211030 [gr-qc] (cit. on p. 41).
- [117] Michael Geracie, Kartik Prabhu, and Matthew M. Roberts. "Curved non-relativistic spacetimes, Newtonian gravitation and massive matter." In: J. Math. Phys. 56.10 (2015), p. 103505. DOI: 10.1063/1.4932967. arXiv: 1503.02682 [hep-th] (cit. on p. 41).
- [118] Robert M. Wald. *General Relativity*. 1984. DOI: 10.7208/chicago/9780226870373.001.0001 (cit. on p. 41).
- [119] E. Inonu and Eugene P. Wigner. "On the Contraction of groups and their representations." In: *Proc. Nat. Acad. Sci.* 39 (1953), pp. 510–524. DOI: 10.1073/pnas.39.6.510 (cit. on p. 42).
- [120] Jelle Hartong, Elias Kiritsis, and Niels A. Obers. "Schrödinger Invariance from Lifshitz Isometries in Holography and Field Theory." In: *Phys. Rev.* D92 (2015), p. 066003. DOI: 10.1103/ PhysRevD.92.066003. arXiv: 1409.1522 [hep-th] (cit. on p. 47).
- [121] Jelle Hartong. "Gauging the Carroll Algebra and Ultra-Relativistic Gravity." In: *JHEP* 08 (2015), p. 069. DOI: 10.1007/JHEP08(2015)069. arXiv: 1505.05011 [hep-th] (cit. on pp. 47, 49, 99).
- [122] B. Julia and H. Nicolai. "Null Killing vector dimensional reduction and Galilean geometrodynamics." In: *Nucl. Phys.* B439 (1995), pp. 291–326. DOI: 10.1016/0550-3213(94)00584-2. arXiv: hep-th/9412002 [hep-th] (cit. on p. 47).
- [123] Roel Andringa, Eric A. Bergshoeff, Jan Rosseel, and Ergin Sezgin. "3D Newton-Cartan super-gravity." In: *Class. Quant. Grav.* 30 (2013), p. 205005. DOI: 10.1088/0264-9381/30/20/205005. arXiv: 1305.6737 [hep-th] (cit. on p. 50).
- [124] Eric Bergshoeff, Jan Rosseel, and Thomas Zojer. "Newton-Cartan supergravity with torsion and Schrödinger supergravity." In: *JHEP* 11 (2015), p. 180. DOI: 10.1007/JHEP11(2015)180. arXiv: 1509.04527 [hep-th] (cit. on p. 50).
- [125] Jérémie Unterberger and Claude Roger. *The Schrödinger-Virasoro Algebra*. Berlin: Springer, 2012. ISBN: 9783642227165, 9783642227172. DOI: 10.1007/978-3-642-22717-2 (cit. on pp. 50, 80).
- [126] Maciej Dunajski and James Gundry. "Non-Relativistic Twistor Theory and NewtonCartan Geometry." In: *Commun. Math. Phys.* 342.3 (2016), pp. 1043–1074. DOI: 10.1007/s00220-015-2557-8. arXiv: 1502.03034 [hep-th] (cit. on p. 50).
- [127] James Gundry. "Higher symmetries of the Schrödinger operator in NewtonCartan geometry." In: J. Geom. Phys. 113 (2017), pp. 73–85. DOI: 10.1016/j.geomphys.2016.06.003. arXiv: 1601.
  02797 [math-ph] (cit. on p. 50).
- [128] James Gundry. "Kodaira Families and Newton-Cartan Structures with Torsion." In: (2017). arXiv: 1704.00759 [math.DG] (cit. on p. 50).
- [129] Eric Bergshoeff, Athanasios Chatzistavrakidis, Luca Romano, and Jan Rosseel. "Newton-Cartan Gravity and Torsion." In: (2017). arXiv: 1708.05414 [hep-th] (cit. on p. 50).
- [130] Eddy Ardonne, Paul Fendley, and Eduardo Fradkin. "Topological order and conformal quantum critical points." In: Annals Phys. 310 (2004), pp. 493–551. DOI: 10.1016/j.aop.2004.01.004. arXiv: cond-mat/0311466 [cond-mat] (cit. on pp. 51, 52).
- [131] Marika Taylor. "Lifshitz holography." In: *Class. Quant. Grav.* 33.3 (2016), p. 033001. arXiv: 1512.
   03554 [hep-th] (cit. on pp. 51, 148).
- [132] Paul M Chaikin and Tom C Lubensky. *Principles of condensed matter physics*. Cambridge university press, 2000 (cit. on p. 52).
- [133] G Grinstein. "Anisotropic sine-Gordon model and infinite-order phase transitions in three dimensions." In: *Physical Review B* 23.9 (1981), p. 4615 (cit. on p. 52).
- [134] Daniel S. Rokhsar and Steven A. Kivelson. "Superconductivity and the Quantum Hard-Core Dimer Gas." In: *Phys. Rev. Lett.* 61 (20 1988), pp. 2376–2379. DOI: 10.1103/PhysRevLett.61.2376. URL: http://link.aps.org/doi/10.1103/PhysRevLett.61.2376 (cit. on p. 52).
- Benjamin Hsu, Michael Mulligan, Eduardo Fradkin, and Eun-Ah Kim. "Universal entanglement entropy in 2D conformal quantum critical points." In: *Phys. Rev.* B79 (2009), p. 115421.
   DOI: 10.1103/PhysRevB.79.115421. arXiv: 0812.0203 [cond-mat.stat-mech] (cit. on p. 52).

- [136] Tianci Zhou, Xiao Chen, Thomas Faulkner, and Eduardo Fradkin. "Entanglement entropy and mutual information of circular entangling surfaces in the 2+1-dimensional quantum Lifshitz model." In: J. Stat. Mech. 1609.9 (2016), p. 093101. DOI: 10.1088/1742-5468/2016/09/093101. arXiv: 1607.01771 [cond-mat.stat-mech] (cit. on p. 52).
- [137] Shira Chapman, Yaron Oz, and Avia Raviv-Moshe. "On Supersymmetric Lifshitz Field Theories." In: *JHEP* 10 (2015), p. 162. DOI: 10.1007/JHEP10(2015)162. arXiv: 1508.03338 [hep-th] (cit. on p. 52).
- [138] Gaetano Bertoldi, Benjamin A. Burrington, and Amanda Peet. "Black Holes in asymptotically Lifshitz spacetimes with arbitrary critical exponent." In: *Phys. Rev.* D80 (2009), p. 126003. DOI: 10.1103/PhysRevD.80.126003. arXiv: 0905.3183 [hep-th] (cit. on p. 52).
- [139] Gaetano Bertoldi, Benjamin A. Burrington, and Amanda W. Peet. "Thermodynamics of black branes in asymptotically Lifshitz spacetimes." In: *Phys. Rev.* D80 (2009), p. 126004. DOI: 10. 1103/PhysRevD.80.126004. arXiv: 0907.4755 [hep-th] (cit. on p. 52).
- [140] Henri Ruegg and Marti Ruiz-Altaba. "The Stueckelberg field." In: Int. J. Mod. Phys. A19 (2004), pp. 3265–3348. DOI: 10.1142/S0217751X04019755. arXiv: hep-th/0304245 [hep-th] (cit. on p. 53).
- [141] Kristan Jensen. "On the coupling of Galilean-invariant field theories to curved spacetime." In: *arXiv preprint arXiv:1408.6855* (2014) (cit. on p. 59).
- [142] Stefan Hollands, Akihiro Ishibashi, and Donald Marolf. "Counter-term charges generate bulk symmetries." In: *Phys. Rev.* D72 (2005), p. 104025. DOI: 10.1103/PhysRevD.72.104025. arXiv: hep-th/0503105 [hep-th] (cit. on p. 64).
- [143] Yuya Kusuki, Tadashi Takayanagi, and Koji Umemoto. "Holographic Entanglement Entropy on Generic Time Slices." In: (2017). arXiv: 1703.00915 [hep-th] (cit. on p. 66).
- [144] Simon A. Gentle and Cynthia Keeler. "On the reconstruction of Lifshitz spacetimes." In: *JHEP* 03 (2016), p. 195. DOI: 10.1007/JHEP03(2016)195. arXiv: 1512.04538 [hep-th] (cit. on p. 66).
- [145] Temple He, Javier M. Magan, and Stefan Vandoren. "Entanglement Entropy in Lifshitz Theories." In: (2017). arXiv: 1705.01147 [hep-th] (cit. on p. 66).
- [146] M. Reza Mohammadi Mozaffar and Ali Mollabashi. "Entanglement in Lifshitz-type Quantum Field Theories." In: *JHEP* 07 (2017), p. 120. DOI: 10.1007/JHEP07(2017)120. arXiv: 1705.00483
   [hep-th] (cit. on p. 66).
- [147] Tom Griffin, Petr Hoava, and Charles M. Melby-Thompson. "Lifshitz Gravity for Lifshitz Holography." In: *Phys. Rev. Lett.* 110.8 (2013), p. 081602. DOI: 10.1103/PhysRevLett.110.081602. arXiv: 1211.4872 [hep-th] (cit. on p. 66).
- [148] Elias Kiritsis and Yoshinori Matsuo. "Charge-hyperscaling violating Lifshitz hydrodynamics from black-holes." In: *JHEP* 12 (2015), p. 076. DOI: 10.1007/JHEP12(2015)076. arXiv: 1508.02494 [hep-th] (cit. on pp. 66, 99).
- [149] Elias Kiritsis and Yoshinori Matsuo. "Hyperscaling-Violating Lifshitz hydrodynamics from black-holes: Part II." In: *JHEP* 03 (2017), p. 041. DOI: 10.1007/JHEP03(2017)041. arXiv: 1611.
   04773 [hep-th] (cit. on pp. 66, 99).
- [150] J. de Boer, J. Hartong, N. Obers, W. Sybesma, and S. Vandoren. "In preparation..." In: () (cit. on p. 66).
- [151] M. Le Bellac and J.-M. Levy-Leblond. "Galilean Electromagnetism." In: *Il Nuovo Cimento*. 14B (1973) (cit. on p. 68).
- [152] Germain Rousseaux. "Forty years of Galilean Electromagnetism (1973–2013)." In: *The European Physical Journal Plus* 128.8 (2013), p. 81 (cit. on p. 68).
- [153] E. S. Santos, M. de Montigny, F. C. Khanna, and A. E. Santana. "Galilean covariant Lagrangian models." In: *J. Phys.* A37 (2004), pp. 9771–9789. DOI: 10.1088/0305-4470/37/41/011 (cit. on p. 68).
- [154] Koushik Balasubramanian and K. Narayan. "Lifshitz spacetimes from AdS null and cosmological solutions." In: *JHEP* 08 (2010), p. 014. DOI: 10.1007/JHEP08(2010)014. arXiv: 1005.3291
   [hep-th] (cit. on p. 74).
- [155] R. N. Caldeira Costa and Marika Taylor. "Holography for chiral scale-invariant models." In: JHEP 02 (2011), p. 082. DOI: 10.1007/JHEP02(2011)082. arXiv: 1010.4800 [hep-th] (cit. on p. 74).

- [156] Christopher P. Herzog, Mukund Rangamani, and Simon F. Ross. "Heating up Galilean holog-raphy." In: *JHEP* 11 (2008), p. 080. DOI: 10.1088/1126-6708/2008/11/080. arXiv: 0807.1099
   [hep-th] (cit. on p. 99).
- [157] Troels Harmark, Jelle Hartong, and Niels A. Obers. "Non-Relativistic Strings and Limits of the AdS/CFT Correspondence." In: (2017). arXiv: 1705.03535 [hep-th] (cit. on p. 99).
- [158] C. Duval, G. W. Gibbons, and P. A. Horvathy. "Conformal Carroll groups and BMS symmetry." In: *Class. Quant. Grav.* 31 (2014), p. 092001. DOI: 10.1088/0264-9381/31/9/092001. arXiv: 1402.5894 [gr-qc] (cit. on p. 99).
- [159] Jelle Hartong. "Holographic Reconstruction of 3D Flat Space-Time." In: *JHEP* 10 (2016), p. 104.
   DOI: 10.1007/JHEP10(2016)104. arXiv: 1511.01387 [hep-th] (cit. on p. 99).
- [160] E. J. Brynjolfsson, U. H. Danielsson, L. Thorlacius, and T. Zingg. "Holographic Superconductors with Lifshitz Scaling." In: *J. Phys.* A43 (2010), p. 065401. DOI: 10.1088/1751-8113/43/6/065401. arXiv: 0908.2611 [hep-th] (cit. on p. 99).
- [161] Sean M. Carroll. Spacetime and geometry: An introduction to general relativity. San Francisco, USA: Addison-Wesley (2004) 513 p, 2004. ISBN: 0805387323, 9780805387322. URL: http://www.slac. stanford.edu/spires/find/books/www?cl=QC6:C37:2004 (cit. on pp. 103, 104).
- [162] Edwin H Spanier. Algebraic topology. Vol. 55. 1. Springer Science & Business Media, 1994 (cit. on p. 104).
- [163] Vijay Balasubramanian, Per Kraus, and Albion E. Lawrence. "Bulk versus boundary dynamics in anti-de Sitter space-time." In: *Phys. Rev.* D59 (1999), p. 046003. DOI: 10.1103/PhysRevD.59.046003. arXiv: hep-th/9805171 [hep-th] (cit. on p. 105).
- [164] R. Penrose and W. Rindler. SPINORS AND SPACE-TIME. VOL. 2: SPINOR AND TWISTOR METHODS IN SPACE-TIME GEOMETRY. Cambridge University Press, 1988. ISBN: 9780521347860, 9780511868429 (cit. on p. 105).
- [165] M. T. Anderson. "Topics in conformally compact Einstein metrics." In: ArXiv Mathematics eprints (Mar. 2005). eprint: math/0503243 (cit. on p. 105).
- S. W. Hawking and G. F. R. Ellis. *The Large Scale Structure of Space-Time*. Cambridge Monographs on Mathematical Physics. Cambridge University Press, 2011. ISBN: 9780521200165, 9780521099066, 9780511826306, 9780521099066. DOI: 10.1017/CB09780511524646 (cit. on p. 105).
- [167] Veronika E. Hubeny, Donald Marolf, and Mukund Rangamani. "Hawking radiation from AdS black holes." In: *Class. Quant. Grav.* 27 (2010), p. 095018. DOI: 10.1088/0264-9381/27/9/095018. arXiv: 0911.4144 [hep-th] (cit. on p. 107).
- [168] Kostas Skenderis and Sergey N. Solodukhin. "Quantum effective action from the AdS / CFT correspondence." In: *Phys. Lett.* B472 (2000), pp. 316–322. DOI: 10.1016/S0370-2693(99)01467-7. arXiv: hep-th/9910023 [hep-th] (cit. on p. 107).
- [169] M. Henningson and K. Skenderis. "The Holographic Weyl anomaly." In: *JHEP* 07 (1998), p. 023.
   DOI: 10.1088/1126-6708/1998/07/023. arXiv: hep-th/9806087 [hep-th] (cit. on pp. 107, 108, 136).
- [170] P. Di Francesco, P. Mathieu, and D. Senechal. *Conformal Field Theory*. Graduate Texts in Contemporary Physics. New York: Springer-Verlag, 1997. ISBN: 9780387947853, 9781461274759. DOI: 10.1007/978-1-4612-2256-9. URL: http://www-spires.fnal.gov/spires/find/books/www?cl=QC174.52.C66D5::1997 (cit. on pp. 109, 115).
- [171] David Simmons-Duffin. TASI Lectures on the Conformal Bootstrap. TASI, 2016. arXiv: 1602.07982
   [hep-th] (cit. on pp. 109, 112).
- [172] Henriette Elvang and Yu-tin Huang. *Scattering Amplitudes in Gauge Theory and Gravity*. Cambridge University Press, 2015. ISBN: 9781316191422, 9781107069251 (cit. on pp. 109, 115).
- [173] R. Slansky. "Group Theory for Unified Model Building." In: *Phys. Rept.* 79 (1981), pp. 1–128.
   DOI: 10.1016/0370-1573(81)90092-2 (cit. on p. 114).
- [174] Sidney R. Coleman and J. Mandula. "All Possible Symmetries of the S Matrix." In: *Phys. Rev.* 159 (1967), pp. 1251–1256. DOI: 10.1103/PhysRev.159.1251 (cit. on p. 115).
- [175] Edward Witten. "Introduction to Supersymmetry." In: THE UNITY OF THE FUNDAMENTAL INTERACTIONS. PROCEEDINGS, 19TH COURSE OF THE INTERNATIONAL SCHOOL OF SUBNUCLEAR PHYSICS, ERICE, ITALY, JULY 31 - AUGUST 11, 1981. 1983, pp. 305–371. DOI: 10.1007/978-1-4613-3655-6\_7. URL: http://link.springer.com/chapter/10.1007/978-1-4613-3655-6\_7 (cit. on p. 115).

- [176] Steven Weinberg. *The quantum theory of fields. Vol. 3: Supersymmetry.* Cambridge University Press, 2013. ISBN: 9780521670555, 9781139632638, 9780521670555 (cit. on p. 115).
- [177] J. Wess and J. Bagger. *Supersymmetry and supergravity*. Princeton, USA: Univ. Pr. (1992) 259 p, 1992 (cit. on p. 115).
- [178] Zoubos K. An Introduction to Supersymmetry. unpublished, 2012 (cit. on p. 115).
- [179] Ruth Britto, Freddy Cachazo, and Bo Feng. "New recursion relations for tree amplitudes of gluons." In: Nucl. Phys. B715 (2005), pp. 499–522. DOI: 10.1016/j.nuclphysb.2005.02.030. arXiv: hep-th/0412308 [hep-th] (cit. on p. 115).
- [180] Jacob L. Bourjaily, Paul Heslop, and Vuong-Viet Tran. "Amplitudes and Correlators to Ten Loops Using Simple, Graphical Bootstraps." In: JHEP (2016). arXiv: 1609.00007 [hep-th] (cit. on p. 115).
- [181] Zarembo K. Lectures on AdS/CFT Correspondence. unpublished, 2015 (cit. on p. 116).
- [182] Gerard 't Hooft. "A Planar Diagram Theory for Strong Interactions." In: *Nucl. Phys.* B72 (1974), p. 461. DOI: 10.1016/0550-3213(74)90154-0 (cit. on p. 119).
- [183] Alfonso V. Ramallo. "Introduction to the AdS/CFT correspondence." In: Springer Proc. Phys. 161 (2015), pp. 411–474. DOI: 10.1007/978-3-319-12238-0\_10. arXiv: 1310.4319 [hep-th] (cit. on p. 119).
- [184] Katrin Becker, Melanie Becker, and John H Schwarz. *String theory and M-theory: A modern introduction*. Cambridge University Press, 2006 (cit. on p. 120).
- [185] Paolo Pasti, Dmitri P. Sorokin, and Mario Tonin. "Covariant action for a D = 11 five-brane with the chiral field." In: *Phys. Lett.* B398 (1997), pp. 41–46. DOI: 10.1016/S0370-2693(97)00188-3. arXiv: hep-th/9701037 [hep-th] (cit. on p. 120).
- [186] C. Montonen and David I. Olive. "Magnetic Monopoles as Gauge Particles?" In: *Phys. Lett.* 72B (1977), pp. 117–120. DOI: 10.1016/0370-2693(77)90076-4 (cit. on p. 120).
- [187] Troels Harmark. Introduction to String Theory: Blok 3, 2015 version 11. Unpublished, 2015 (cit. on p. 121).
- [188] Edward Witten. "Anti-de Sitter space, thermal phase transition, and confinement in gauge theories." In: Adv. Theor. Math. Phys. 2 (1998), pp. 505–532. arXiv: hep-th/9803131 [hep-th] (cit. on pp. 125, 126).
- [189] S. S. Gubser, Igor R. Klebanov, and A. W. Peet. "Entropy and temperature of black 3-branes." In: *Phys. Rev.* D54 (1996), pp. 3915–3919. DOI: 10.1103/PhysRevD.54.3915. arXiv: hep-th/9602135
   [hep-th] (cit. on p. 125).
- [190] Gary T. Horowitz and Andrew Strominger. "Black strings and P-branes." In: *Nucl. Phys.* B360 (1991), pp. 197–209. DOI: 10.1016/0550-3213(91)90440-9 (cit. on p. 125).
- [191] C. P. Burgess, N. R. Constable, and Robert C. Myers. "The Free energy of N=4 superYang-Mills and the AdS / CFT correspondence." In: *JHEP* 08 (1999), p. 017. DOI: 10.1088/1126-6708/1999/08/017. arXiv: hep-th/9907188 [hep-th] (cit. on p. 126).
- [192] Ted Jacobson and Gungwon Kang. "Conformal invariance of black hole temperature." In: *Class. Quant. Grav.* 10 (1993), pp. L201–L206. DOI: 10.1088/0264-9381/10/11/002. arXiv: gr-qc/9307002 [gr-qc] (cit. on p. 127).
- [193] Johanna Erdmenger, Nick Evans, Ingo Kirsch, and Ed Threlfall. "Mesons in Gauge/Gravity Duals A Review." In: *Eur. Phys. J.* A35 (2008), pp. 81–133. DOI: 10.1140/epja/i2007-10540-1. arXiv: 0711.4467 [hep-th] (cit. on p. 127).
- [194] Subir Sachdev. "What can gauge-gravity duality teach us about condensed matter physics?" In: Ann. Rev. Condensed Matter Phys. 3 (2012), pp. 9–33. DOI: 10.1146/annurev-conmatphys-020911-125141. arXiv: 1108.1197 [cond-mat.str-el] (cit. on p. 128).
- [195] Steven S. Gubser and Silviu S. Pufu. "The Gravity dual of a p-wave superconductor." In: *JHEP* 11 (2008), p. 033. DOI: 10.1088/1126-6708/2008/11/033. arXiv: 0805.2960 [hep-th] (cit. on p. 129).
- [196] Francesco Benini, Christopher P. Herzog, Rakibur Rahman, and Amos Yarom. "Gauge gravity duality for d-wave superconductors: prospects and challenges." In: *JHEP* 11 (2010), p. 137. DOI: 10.1007/JHEP11(2010)137. arXiv: 1007.1981 [hep-th] (cit. on p. 129).
- [197] Keun-Young Kim and Marika Taylor. "Holographic d-wave superconductors." In: *JHEP* 08 (2013), p. 112. DOI: 10.1007/JHEP08(2013)112. arXiv: 1304.6729 [hep-th] (cit. on p. 129).
- [198] Daniele Musso. "Introductory notes on holographic superconductors." In: *PoS* Modave2013 (2013), p. 004. arXiv: 1401.1504 [hep-th] (cit. on p. 129).
- [199] Christopher P. Herzog. "Lectures on Holographic Superfluidity and Superconductivity." In: J. Phys. A42 (2009), p. 343001. DOI: 10.1088/1751-8113/42/34/343001. arXiv: 0904.1975
   [hep-th] (cit. on p. 129).
- [200] Rong-Gen Cai, Li Li, Li-Fang Li, and Run-Qiu Yang. "Introduction to Holographic Superconductor Models." In: *Sci. China Phys. Mech. Astron.* 58.6 (2015), p. 060401. DOI: 10.1007/s11433-015-5676-5. arXiv: 1502.00437 [hep-th] (cit. on p. 129).
- [201] Steven S. Gubser. "Breaking an Abelian gauge symmetry near a black hole horizon." In: *Phys. Rev.* D78 (2008), p. 065034. DOI: 10.1103/PhysRevD.78.065034. arXiv: 0801.2977 [hep-th] (cit. on p. 129).
- [202] Michael Tinkham. Introduction to superconductivity. Courier Corporation, 1996 (cit. on p. 132).
- [203] Piers Coleman. *Introduction to many-body physics*. Cambridge University Press, 2015 (cit. on p. 133).
- [204] Dam T. Son and Andrei O. Starinets. "Minkowski space correlators in AdS / CFT correspondence: Recipe and applications." In: *JHEP* 09 (2002), p. 042. DOI: 10.1088/1126-6708/2002/09/042. arXiv: hep-th/0205051 [hep-th] (cit. on p. 135).
- [205] Sean A. Hartnoll and Christopher P. Herzog. "Impure AdS/CFT correspondence." In: *Phys. Rev.* D77 (2008), p. 106009. DOI: 10.1103/PhysRevD.77.106009. arXiv: 0801.1693 [hep-th] (cit. on p. 135).
- [206] G. W. Gibbons and S. W. Hawking. "Action Integrals and Partition Functions in Quantum Gravity." In: *Phys. Rev.* D15 (1977), pp. 2752–2756. DOI: 10.1103/PhysRevD.15.2752 (cit. on p. 137).
- [207] Masafumi Fukuma, So Matsuura, and Tadakatsu Sakai. "Holographic renormalization group." In: *Prog. Theor. Phys.* 109 (2003), pp. 489–562. DOI: 10.1143/PTP.109.489. arXiv: hep-th/0212314
   [hep-th] (cit. on pp. 141, 144).
- [208] O. Biquard, ed. AdS/CFT correspondence: Einstein metrics and their conformal boundaries. Proceedings, 73rd Meeting of Theoretical Physicists and Mathematicians, Strasbourg, France, September 11-13, 2003. Vol. 8. 8. IRMA Lectures in Mathematics and Theoretical Physics. Eur. Math. Soc. Zurich, Switzerland: Eur. Math. Soc., 2005 (cit. on p. 141).
- [209] Matthias Blau. Lecture notes on general relativity. Albert Einstein Center for Fundamental Physics Bern Germany, 2011. URL: http://www.blau.itp.unibe.ch/GRLecturenotes.html (cit. on p. 141).
- [210] J. David Brown and James W. York Jr. "Quasilocal energy and conserved charges derived from the gravitational action." In: *Phys. Rev.* D47 (1993), pp. 1407–1419. DOI: 10.1103/PhysRevD.47. 1407. arXiv: gr-qc/9209012 [gr-qc] (cit. on p. 142).
- [211] L. Bonora, P. Pasti, and M. Bregola. "WEYL COCYCLES." In: Class. Quant. Grav. 3 (1986), p. 635.
  DOI: 10.1088/0264-9381/3/4/018 (cit. on p. 144).
- [212] Stanley Deser and A. Schwimmer. "Geometric classification of conformal anomalies in arbitrary dimensions." In: *Phys. Lett.* B309 (1993), pp. 279–284. DOI: 10.1016/0370-2693(93)90934-A. arXiv: hep-th/9302047 [hep-th] (cit. on p. 144).
- [213] H Blaine Lawson Jr. "Foliations." In: Bulletin of the American Mathematical Society 80.3 (1974), pp. 369–418 (cit. on p. 146).
- [214] Vladimir Rovenski. Foliations on Riemannian manifolds and submanifolds. Springer Science & Business Media, 2012 (cit. on p. 146).
- [215] Miranda C. N. Cheng, Sean A. Hartnoll, and Cynthia A. Keeler. "Deformations of Lifshitz holography." In: JHEP 03 (2010), p. 062. DOI: 10.1007/JHEP03(2010)062. arXiv: 0912.2784 [hep-th] (cit. on p. 147).
- [216] Theodor Kaluza. "Zum unitätsproblem der physik." In: *Sitzungsber. Preuss. Akad. Wiss. Berlin.(Math. Phys.)* 1921.966972 (1921), p. 45 (cit. on p. 154).
- [217] Oskar Klein. "Quantentheorie und fünfdimensionale Relativitätstheorie." In: Zeitschrift für Physik 37.12 (1926), pp. 895–906 (cit. on p. 154).

- [218] Christopher N Pope. *Kaluza-klein theory*. Centre Emile Borel, Institut Henri Poincaré, 2000 (cit. on pp. 154, 155, 156).
- [219] Tomas Ortin. Gravity and strings. Cambridge Univ. Press, 2004. ISBN: 0521824753. URL: http: //www.cambridge.org/uk/catalogue/catalogue.asp?isbn=0521824753 (cit. on pp. 154, 158, 159).
- [220] M. J. Duff, B. E. W. Nilsson, and C. N. Pope. "Kaluza-Klein Supergravity." In: *Phys. Rept.* 130 (1986), pp. 1–142. DOI: 10.1016/0370-1573(86)90163-8 (cit. on p. 154).
- [221] Joel Scherk and John H. Schwarz. "How to Get Masses from Extra Dimensions." In: *Nucl. Phys.* B153 (1979), pp. 61–88. DOI: 10.1016/0550-3213(79)90592-3 (cit. on p. 156).
- [222] Joel Scherk and John H. Schwarz. "Spontaneous Breaking of Supersymmetry Through Dimensional Reduction." In: *Phys. Lett.* B82 (1979), pp. 60–64. DOI: 10.1016/0370-2693(79)90425-8 (cit. on p. 156).
- [223] Atish Dabholkar and Chris Hull. "Duality twists, orbifolds, and fluxes." In: JHEP 09 (2003), p. 054. DOI: 10.1088/1126-6708/2003/09/054. arXiv: hep-th/0210209 [hep-th] (cit. on p. 157).